West Burton Solar Project

Planning Statement: West Burton Solar Farm Revision CD

Prepared by Lanpro Services

AprilMay 2024

PINS reference: EN010132

Document reference: **EX6EX7**/WB7.5_**CD**

APFP Regulation 5(2)(q)





Contents

<u>1</u>	<u>INTRODUCTION</u>	<u>4</u>
1.1	INTRODUCTION	4
1.2	THE APPLICANT	6
1.3	LEGISLATIVE CONTEXT REVIEW	6
1.4	Pre-Application Consultation	8
1.5	PURPOSE AND STRUCTURE OF THIS PLANNING STATEMENT	9
2	THE ORDER LIMITS	11
2.1		11
2.2	——SITE DESCRIPTION	11
2.3	SITE SURROUNDINGS	13
2.4 2.4	RELEVANT PLANNING HISTORY	<u>15</u>
3	THE SCHEME	17
3.1	INTRODUCTION	17
	— Components of the Scheme	17
3.3	Construction Period Activities	18
4	— NEED AND BENEFITS	23
<u>.</u> 		23
4.1	——Introduction ——Meeting an Increasing Demand for Electricity	23
	— WEED FOR DECARBONISATION	23 24
	THE NEED TO PROVIDE SECURITY OF SUPPLY	——————————————————————————————————————
	——THE NEED FOR LARGE SCALE SOLAR TO DELIVER LOW-COST ENERGY	25 26
	OTHER BENEFITS OF THE SCHEME	27
	— Community Liaison Group	<u>27</u>
	— Community Fund	
5	LEGISLATIVE AND POLICY CONTEXT	<u>29</u>
<u>-</u> 5.1	INTRODUCTION	
J	Legislative Context	29
5.3	— Policy Context	30
5.4	— NATIONAL PLANNING POLICY	31
5.5	— NATIONAL PLANNING POLICY FRAMEWORK	3 6
5.6——	——SUMMARY	36
5.7—	——NATIONAL INFRASTRUCTURE PLANNING GUIDANCE	36
5.8——	——PLANNING PRACTICE GUIDANCE (PPG)	<u>37</u>
5.9	LOCAL PLANNING POLICY	37 ——37
5.10—	——Supplementary Planning Documents and other local strategies	38
5.11	OTHER POLICY AND LEGISLATION	<u>39</u>
5.12	——Summary of the Main Planning Policy Requirements	4 <u>3</u>



<u>6</u>	PLANNING APPRAISAL	45
6.1	Introduction	45
6.2	MEETING THE RENEWABLE ENERGY NEED	47
6.3	ALTERNATIVE SITES AND SITE SELECTION	53
6.4	GOOD DESIGN	64
6.5	LANDSCAPE AND VISUAL ASSESSMENT	76
6.6	——HERITAGE	83
6.7	AGRICULTURE	93
6.8	——MINERAL SAFEGUARDING	100
6.9	ECOLOGY AND BIODIVERSITY	103
6.10-	WATER AND DRAINAGE	115
6.11	Noise and Vibration	124
6.12-	GLINT AND GLARE	128
6.13—	TRANSPORT AND ACCESS	130
6.14—	WASTE	136
6.15—	SOCIO-ECONOMICS, TOURISM AND RECREATION	140
6.16-	EFFECTS ON HUMAN HEALTH	144
6.17—	MAJOR ACCIDENTS AND DISASTERS	145
6.18—	AIR QUALITY	146
6.19—	——GROUND CONDITIONS	149
Z	CONCLUSIONS AND PLANNING BALANCE	152
1	INTRODUCTION	5
1.1	Introduction	5
1.2	THE APPLICANT	
1.3	LEGISLATIVE CONTEXT REVIEW	 7
1.4	Pre-Application Consultation	9
1.5	Purpose and structure of this Planning Statement	10
2	THE ORDER LIMITS	12
2.1	Introduction	12
2.2	SITE DESCRIPTION	12 12
2.3	SITE SURROUNDINGS	12 14
2.4	RELEVANT PLANNING HISTORY	1 - 16
3	THE SCHEME	<u>18</u>
3.1	Introduction	18
3.2	COMPONENTS OF THE SCHEME	18
3.3	CONSTRUCTION PERIOD ACTIVITIES	19
4	NEED AND BENEFITS	24
4.1	Introduction	24
4.2	MEETING AN INCREASING DEMAND FOR ELECTRICITY	24
<u>4.3</u>	NEED FOR DECARBONISATION	25
4.4	THE NEED TO PROVIDE SECURITY OF SUPPLY	26



4.5	THE NEED FOR LARGE SCALE SOLAR TO DELIVER LOW-COST ENERGY	27
4.6	OTHER BENEFITS OF THE SCHEME	28
4.7	COMMUNITY LIAISON GROUP	28
4.8	COMMUNITY FUND	29
<u>5</u>	LEGISLATIVE AND POLICY CONTEXT	30
5.1	Introduction	30
5.2	LEGISLATIVE CONTEXT	30
5.3	POLICY CONTEXT	31
5.4	NATIONAL PLANNING POLICY	32
5.5	NATIONAL PLANNING POLICY FRAMEWORK	37
5.6	Summary	37
5.7	NATIONAL INFRASTRUCTURE PLANNING GUIDANCE	37
<u>5.8</u>	PLANNING PRACTICE GUIDANCE (PPG)	38
<u>5.9</u>	Local Planning Policy	38
<u>5.10</u>	SUPPLEMENTARY PLANNING DOCUMENTS AND OTHER LOCAL STRATEGIES	39
5.11	OTHER POLICY AND LEGISLATION	40
5.12	SUMMARY OF THE MAIN PLANNING POLICY REQUIREMENTS	44
6	PLANNING APPRAISAL	<u>46</u>
6.1	Introduction	46
6.2	MEETING THE RENEWABLE ENERGY NEED	48
6.3	ALTERNATIVE SITES AND SITE SELECTION	54
6.4	GOOD DESIGN	65
6.5	LANDSCAPE AND VISUAL ASSESSMENT	77
6.6	HERITAGE	84
6.7	AGRICULTURE	94
6.8	Mineral Safeguarding	101
6.9	ECOLOGY AND BIODIVERSITY	104
6.10	Water and Drainage	116
6.11	Noise and Vibration	125
6.12	GLINT AND GLARE	129
6.13	Transport and Access	131
6.14	Waste	137
6.15	SOCIO-ECONOMICS, TOURISM AND RECREATION	141
6.16	Effects on Human Health	145
6.17	MAJOR ACCIDENTS AND DISASTERS	146
6.18	AIR QUALITY	 147
6.19	GROUND CONDITIONS	150
7	CONCLUSIONS AND PLANNING BALANCE	<u>153</u>



Issue Sheet

Report Prepared for: West Burton Solar Project Ltd. Examination Deadline 67

Planning Statement

Prepared by:

Name: Beccy Rejzek

Title: Associate Director MRTPI

Approved by:

Name: Ian Douglas

Title: Director MRTPI

Date Original: January 2023

Date Original: January 2023

Revision	Date	Prepared by:	Approved by:
А	3 January 2024	GV	JC
В	28 February 2024	GV	JC
С	30 April 2024	GV	JC
<u>D</u>	8 May 2024	<u>SF</u>	<u>JC</u>



1 Introduction

1.1 Introduction

- 1.1.1 West Burton Solar Project Limited (the "Applicant") has prepared this Planning Statement (the "Statement") as part of an application for a Development Consent Order (DCO) to construct, operate, maintain and decommission the West Burton Solar Project (the "Scheme").
- 1.1.2 The Scheme comprises a number of land parcels (the 'Site' or 'Sites') described as West Burton 1, 2 and 3 for the solar arrays, grid connection infrastructure and Energy Storage; and the Cable Route Corridors. The Sites are located approximately 7km southeast of Gainsborough. See the Site Location Plan [REP4-004] for the Site locations.
- 1.1.3 The Scheme is described in full in Chapter 4 of the Environmental Statement (ES), Scheme Description [APP-042] supporting the application.
- 1.1.4 The DCO application is for the construction, operation (including maintenance) and decommissioning of the Scheme. The Scheme consists of a solar photovoltaic (PV) array electricity generating station, energy storage facility and grid connection to the national electricity transmission network (NETS). The majority of the Scheme will be located within the administrative boundary of West Lindsey District Council and Lincolnshire County Council; with part of the grid connection infrastructure located within the administrative boundary of Bassetlaw District Council and Nottinghamshire County Council.
- 1.1.5 The Scheme would generate large amounts of electricity from a renewable source and so it would assist the Government in meeting its targets to decarbonise our electricity supply and reduce overall carbon emissions.
- 1.1.6 The Government expects large scale solar generation to make an important contribution to achieving its objectives for the UK's power system which are to ensure the supply of energy always remains secure, reliable, affordable, and enables the UK to meet its carbon emission reduction commitments. These include the achievement of net zero carbon emissions by 2050 and delivery of carbon budgets in the intervening years. Further details are set out in the Statement of Need [APP-320].
- 1.1.7 Overarching National Policy Statement for Energy (NPS EN-1 2011) states at paragraph 3.3.15 that new low carbon energy NSIPs are required urgently in the next 10-15 years (from its publication date in July 2011):
 - "In order to secure energy supplies that enable us to meet our obligations for 2050, there is an urgent need for new (and particularly low carbon) energy NSIPs to be brought forward as soon as possible, and certainly in the next 10 to 15 years, given the crucial role of electricity as the UK decarbonises its energy sector."
- 1.1.8 Draft Overarching National Policy Statement for Energy (NPS) were published for consultation in September 2021 and updated in March 2023. The updated National



Policy Statements (NPS) for Energy were published on 22 November 2023 and designated on 17 January 2024.

1.1.9 NPS EN-1 (November 2023) provides an update to NPS EN-1 (2011). This sets out at paragraph 3.3.65:

"There is an urgent need for new electricity network infrastructure to be brought forward at pace to meet our energy objectives."

1.1.10 It also sets out at paragraph 3.3.20 that solar, along with wind, is expected to be the main form of electricity generation in an energy system that meets the Government's objectives for delivering secure, affordable energy and meets its climate change commitments:

"Wind and solar are the lowest cost ways of generating electricity, helping reduce costs and providing a clean and secure source of electricity supply (as they are not reliant on fuel for generation). Our analysis shows that a secure, reliable, affordable, net zero consistent system in 2050 is likely to be composed predominantly of wind and solar."

1.1.11 Paragraph 3.3.21 of NPS EN-1 (November 2023) goes on to state that there is a requirement for sustained growth in capacity in onshore solar in the next decade. This updates the position compared to NPS EN-1 (2021):

"As part of delivering this, UK government announced in the British Energy Security Strategy and ambition to deliver up to 50 gigawatts (GW) of offshore wind by 2030, including up to 5GW of floating wind, and the requirement in the Energy White Paper for sustained growth in the capacity of onshore wind and solar in the next decade."

1.1.12 Paragraph 3.3.22 of NPS EN-1 (November 2023) (above) reflects the revised position compared to NPS EN-1 (2011), where Paragraph 3.3.22 states that:

"around 33 GW of the new capacity by 2025 would need to come from renewable sources to meet renewable energy commitments as set out in Section 3.4"

- 1.1.13 The Scheme represents an excellent opportunity to deliver a critical part of the portfolio of renewable energy generation that is urgently required by 2030.
- 1.1.14 The Scheme would also deliver biodiversity net gain (BNG) through the commitments set out in the Outline Landscape and Ecological Management Plan (OLEMP) [EN010132/EX6/WB7.3_E].REP6-025]. These include habitat management areas for biodiversity mitigation and enhancements, and will deliver the following from a BNG perspective:
 - Habitat Units 86.80%;
 - Hedgerow Units 54.71%; and
 - River Units 33.25%.
- 1.1.15 Further detail on this can be found within ES Appendix 9.12 Biodiversity Net Gain Report [APP-088].



- 1.1.16 The site selection and Scheme design has been developed at every stage to minimise the impact on the local area. Areas of the Scheme that were included at the non-statutory and statutory consultation stages have since been removed to reduce or remove impacts on the nearest residents, designated heritage assets and for ecological reasons. Other areas have been removed to reduce the impact on Best and Most Versatile (BMV) grade agricultural land. The Sites' layouts have also been designed so that larger structures such as substations and the Energy Storage Facility are located based upon landscape assessment and archaeological investigation works so that their impacts are minimised. Chapter 5: Alternatives and Design Evolution, of the Environmental Statement (ES) [APP-043] and the Design and Access Statement [APP-314 and APP-315] explain the design evolution of the Scheme in detail.
- 1.1.17 Overall, the proposals are considered to comply with planning policies, and deliver much needed large-scale energy-generating infrastructure in a way that is sensitive to its surrounding area and delivers additional benefits. Compliance with relevant National and Local Planning Policies is set out respectively in Appendices C and D.

1.2 The Applicant

- 1.2.1 The Scheme is being developed by the Applicant. The Applicant is part of Island Green Power Limited (IGP) which was established in 2013 and is a leading international developer of renewable energy projects. Further information on the Applicant can be found in the Funding Statement [REP4-030] that has been submitted as part of the DCO Application.
- 1.2.2 IGP has delivered 26 solar projects worldwide totalling more than 1GW of capacity. This includes 14 solar projects in the UK and Republic of Ireland. Their mission is to increase solar energy usage, making more renewable energy possible whilst reducing carbon emissions by thousands of tonnes in the process.
- 1.2.3 IGP is also progressing the Cottam Solar Project, which is within the same locality as the Scheme. It is the subject of a separate DCO application (PINS application Ref: EN010133) which was submitted on 12th January 2023 and is therefore the subject of a separate Planning Statement.

1.3 Legislative context review

- 1.3.1 The Scheme is defined as a Nationally Significant Infrastructure Project (NSIP) under Sections 14(1)(a), 15(1) and 15(2) of the Planning Act 2008 (PA 2008) as it is for the construction of an onshore generating station in England with a capacity exceeding 50 megawatts (MW). The PA 2008 requires a DCO to be obtained for the development of NSIPs.
- 1.3.2 The PA 2008 prescribes that the Secretary of State (SoS) is responsible for determining an application for development consent, with the power to appoint an Examining Authority (ExA) of appointed person(s) to manage and examine the application. The ExA, appointed through the Planning Inspectorate, will make procedural decisions and examine the application. Following their examination of



the application, the ExA will make a recommendation to the SoS who will then decide whether to grant a DCO.

- 1.3.3 DCO applications are determined in line with Section 104 of the PA 2008 where a relevant National Policy Statement (NPS) is in place, or Section 105 where one is not. NPSs set out the policy basis upon which NSIPs are determined. There is currently no NPS designated for solar generating stations. There is an Overarching NPS for Energy (EN-1 2011), but it does not provide specific guidance on solar technologies and therefore Section 104 of PA 2008 does not apply to the Scheme.
- 1.3.4 The updated NPS for Energy were published on 22 November 2023 and were laid before Parliament under section 9(8) of the Planning Act 2008.
- 1.3.5 The NPSs for Energy (November 2023) were designated on 17 January 2024 and are applicable to all new DCO applications for energy NSIPs under s104 of the Planning Act 2008 from early 2024.
- 1.3.6 These include the National Policy Statement for Renewable Energy (NPS EN-3 November 2023), which includes specific policies for solar photovoltaic generation NSIPs.
- 1.3.7 Section 1.6 of NPS EN-1 (November 2023) sets out the transitional provisions and states that for DCO applications submitted prior to the designation of the November 2023 NPSs (such as the Scheme), the 2011 suite of NPSs will continue to have effect and therefore the DCO application for the Scheme will be determined under s105 of the Planning Act 2008.
- 1.3.8 Paragraph 1.6.2 of NPS EN-1 (November 2023) sets out that where an application is accepted for examination before the new Energy NPSs are designated, those newly designated NPS will not have effect. However, paragraph 1.6.3 goes on to clarify that "...any emerging draft NPSs (or those designated but not yet having effect) are potentially capable of being important and relevant considerations in the decision-making process. The extent to which they are relevant is a matter for the relevant Secretary of State to consider within the framework of the Planning Act 2008 and with regard to the specific circumstances of each Development Consent Order application."
- 1.3.9 Consequently, the DCO application for the Scheme will be determined in accordance with Section 105 of the PA 2008 as at the time of acceptance no technology specific NPS has effect. Section 105(2) of the PA 2008 sets out what the SoS must have regard to when deciding the DCO application. This includes any matters which the SoS deems to be "both important and relevant" to their decision. The Applicant considers that the following NPSs for Energy are all important and relevant to the SoS's decision:
 - Overarching National Policy Statement for Energy (EN-1) (NPS EN-1 2011);
 - National Policy Statement for Renewable Energy (EN-3) (NPS EN-3 2011); and
 - National Policy Statement for Electricity Networks Infrastructure (EN-5) (NPS EN-5 2011).



- 1.3.10 In addition, the Applicant also expects the NPSs for Energy (November 2023) listed below to be important and relevant to the SoS's decision:
 - Overarching National Policy Statement for Energy (EN-1) (NPS EN-1 November 2023);
 - National Policy Statement for Renewable Energy (EN-3) (NPS EN-3 November 2023); and
 - National Policy Statement for Electricity Networks Infrastructure (EN-5) (NPS EN-5 November 2023).
- 1.3.11 A more detailed explanation of the legislative and policy context of the Scheme is set out in Section 5 of this Planning Statement. This includes the Applicant's reasoning for both the 2011 and November 2023 Energy NPSs being important and relevant matters in the SoS's decision.
- 1.3.12 The Scheme is 'EIA development' as defined by the Infrastructure Planning (Environmental Impact Assessment) Regulations 2017 (the EIA Regulations) which means that an Environmental Impact Assessment (EIA) is required. An Environmental Statement (ES) has been prepared and has been submitted with the DCO application [APP-038 to APP-044, APP-046 to APP-060, REP1-012, REP1-073, REP1-074, REP3-010].
- A DCO may include provisions which removes the requirement to obtain other 1.3.13 consents. Details of the consents and authorisations included in the DCO are the Memorandum the draft DCO explained in Explanatory to [EN010132/EX6/WB3.2_D].REP6-013]. A Consents and Agreements Position Statement [REP4-046] explains those other consents and licenses that are, or may be required under other legislation, that will be sought separately from the DCO for the construction and operation of the Scheme.
- 1.3.14 Section 115 of the PA 2008 also states that a DCO can include consent for 'associated development', which is development that is not an NSIP in its own right but is associated with the NSIP. The NSIP and associated development works are defined in Schedule 1 of the draft DCO [EN010132/EXGEX7/WB3.1_GH] and explained in the Explanatory Memorandum referred to above.
- 1.3.15 The elements of the Scheme that constitute the NSIP and the elements that constitute associated development are summarised in Section 3 of this Planning Statement.

1.4 Pre-Application Consultation

- 1.4.1 The Applicant has undertaken extensive consultation throughout the development of the Scheme. This is described in the Consultation Report [APP-022], and includes the stages listed below.
 - Early engagement with local authorities and statutory consultees in summer / autumn 2021.



- Non-statutory public consultation during November December 2021.
- Environmental Impact Assessment Scoping January March 2022.
- Ongoing stakeholder engagement to inform design development during 2022.
- Discussion and agreement of the content of the Statement of Community Consultation (SoCC) in April May 2022.
- Statutory consultation with the public and statutory consultees during June July 2022.
- Extended statutory consultation from July to 23rd August 2022 to share detailed Agricultural Land Classification Assessment results.
- Additional statutory consultation on changes to West Burton 3 from 25th November 2022 to 8th January 2023.
- 1.4.2 The Applicant has had regard to all feedback it has received in response to its consultations when developing the Scheme. This is described in the Consultation Report referred to above.
- 1.4.3 The ongoing consultation with the West Lindsey District Council, Bassetlaw District Council, Lincolnshire County Council and Nottinghamshire County Council (the Host Authorities) has comprised regular meetings where updates have been provided on the Scheme, including the development of the design, and technical meetings with the Host Authorities' relevant technical specialists, including on the topics of noise, heritage, landscape and visual impact, water and drainage, transport, ecology, climate change and public rights of way. The discussions with the Host Authorities have played a major role in informing the development of the Scheme design and the content of the application, including the ES as shown within Sections 4 to 11 of the Consultation Report [APP-022]. These detail how the engagement with Local Authorities and others has been undertaken from the early consultation stage in Summer-Autumn 2021 through to submission of the application. Table 1.1 of the Consultation Report [APP-022] presents a summary of the changes made to the scheme in response to consultation feedback.

1.5 Purpose and structure of this Planning Statement

- 1.5.1 The purpose of the Planning Statement is to provide an overview of the Scheme, its impacts and the DCO application as a whole, in a way that is easy to understand. It considers and assesses the Scheme against relevant planning policy and other matters the Applicant considers are likely to be important and relevant to the SoS's decision.
- 1.5.2 The remainder of the Planning Statement is structured as follows:
 - Section 2 describes the existing land uses and characteristics of the Sites and their surroundings and the Cable Route Corridor, including planning history and local plan designations. The reasons for selecting the Sites and the extent to which alternatives may be considered important and relevant to the



decision is set out within ES Chapter 5: Alternatives and Design Evolution [APP-043] and ES Appendix 5.1 Site Selection Assessment [AS-004].

- Section 3 provides a summary of the Scheme.
- Section 4 summarises the need and benefits of the Scheme.
- Section 5 outlines the decision-making framework; the planning policy context for the Scheme; and other legislation and policy considered by the Applicant to be important and relevant.
- Section 6 explains the Scheme's compliance with planning policy that the Applicant expects to be important and relevant to the decision.
- Section 7 presents the overall planning balance and conclusions of this Planning Statement.
- 1.5.3 There are four appendices appended to this Planning Statement. These appendices are to be read in conjunction with the Planning Statement and are as follows:
 - Appendix A: Planning Application History Search West Burton Sites;
 - Appendix B: Planning Application History Search Cable Route Corridor;
 - Appendix C: National Policy Accordance Table; and
 - Appendix D: Local Planning Policy Accordance Table.



2 The Order Limits

2.1 Introduction

- 2.1.1 The Order Limits, which include all land falling within the DCO application and cover an area of 886.42 hectares (ha), are located within the administrative areas of West Lindsey District Council, Lincolnshire County Council, Bassetlaw District Council and Nottinghamshire County Council (see Location Plan [REP4-004]).
- 2.1.2 The land within the Order Limits comprises three sites referred to as West Burton 1, 2 and 3 (together known as the Sites) and the land required for the grid connection is referred to as the Cable Route Corridor. These are described below. The works forming part of the Scheme that are to be located in each Site are described in Section 3 of this Planning Statement.

2.2 Site Description

2.2.1 A full description of the Sites is set out at ES Chapter 3: The Order Limits [APP-041] The three Sites identified for built development, namely, solar panels, substations and energy storage for the Scheme are located within a 15km radius of the grid connection at the former West Burton Power Station. Combined they total 769.08ha including means of access but excluding Cable Route Corridors. The three Sites are as follows:

West Burton 1

- 2.2.2 West Burton 1 totals 91.32ha in area and is located to the east of Broxholme with the village of Bransby to the northwest. It lies within the parish of Broxholme. The developable area containing solar panels, substation and associated infrastructure totals 73.51ha. The remaining area is set aside for landscape and ecological mitigation.
- 2.2.3 The Site at West Burton 1 consists almost entirely of agricultural fields used for arable crops. The topography is relatively flat and is predominantly well screened from its immediate surroundings by tall hedges around the boundaries. The fields are generally large and typically have dividing hedgerows. There are only isolated trees outside of field margins. There are a number of existing farm access tracks and field accesses within the Site. Part of the Site adjoins the bank of a watercourse that drains into the River Till. There is a single 132kV overhead line (OHL) that crosses the southern section of the Site in a northwest to southeast orientation. The site is traversed by Main Street, a public highway linking Broxholme village and A1500 Tillbridge Lane. A section of public footpath Brox/196/1 runs though the west of the Site.
- 2.2.4 There are no Listed Buildings or Scheduled Monuments within the Site and it is not within a Conservation Area. There are no Statutory or Non-Statutory ecological designations or Ancient Woodland on the Site. The Site does not include nationally designated landscape or West Lindsey Area of Great Landscape Value (AGLV).

West Burton 2



- 2.2.5 West Burton 2 sits to the west of West Burton 1 and is located to the north of the village of Saxilby. It lies within the parish of Saxilby with Ingleby and covers an area of 306.98ha. The developable area containing solar panels, substation, and associated infrastructure totals 149.62ha. The remaining area is set aside for landscape and ecological mitigation.
- 2.2.6 The Site at West Burton 2 consists almost entirely of agricultural fields used for arable crops. The topography is relatively flat and is predominantly well screened from its immediate surroundings by tall hedges around the boundaries. The fields are generally large and typically have dividing hedgerows. There are only isolated trees outside of field margins. There are a number of existing farm access tracks and field accesses within the Site. Part of the Site adjoins the bank of the River Till. Overhead lines cross part of the landholding. The B1241 Saxilby Road/Sturton Road runs north/south through West Burton 2. In the south-eastern corner of the holding, Broxholme Lane cuts across the land in an east/west direction.
- 2.2.7 There are no PRoWs located within West Burton 2 but there is an 'Other route with Public Access' (ORPA) which runs alongside part of the western boundary.
- 2.2.8 There are no Listed Buildings, Conservation Areas or Scheduled Monuments within the Site. Heritage Assets within the vicinity of the site are shown on the Statutory and Non-Statutory Features of Historic Environment Plan [REP4-018] and are detailed within ES Appendix 13.5 Heritage Assessment [APP-117 to APP-119].
- 2.2.9 There are no statutory or non-statutory ecological designations or ancient woodland within the Site. The Site does not include nationally designated landscape or West Lindsey Area of Great Landscape Value (AGLV).

West Burton 3

- 2.2.10 West Burton 3 sits to the north west of West Burton 2 and is located between the villages of Brampton and Marton within the parishes of Marton, Brampton and Stow. It covers an area of 370.78ha. The developable area containing solar panels, substation and associated infrastructure totals 284.31ha. The remaining area is set aside for landscape and ecological mitigation.
- 2.2.11 The Site at West Burton 3 consists almost entirely of agricultural fields used for arable crops. The topography is relatively flat and is predominantly well screened from its immediate surroundings by tall hedges around the boundaries. The fields are generally large and typically have dividing hedgerows. There are only isolated trees outside of field margins. There are a number of existing farm access tracks and field accesses within the Site and a redundant farmhouse which will remain and is not proposed to be redeveloped. The A1500 Stow Park Road/Till Bridge Lane runs along the northern boundary of West Burton 3. Cowdale Lane runs along the southern boundary. The trainline between Lincoln and Sheffield runs north-south between land parcels comprising the West Burton 3 Site.





2.2.12 There are no statutory or non-statutory designated ecological sites or Ancient Woodland within the Site. The Site does not include nationally designated landscape or West Lindsey Area of Great Landscape Value (AGLV).

Cable Route Corridor

- 2.2.13 The Sites are to be connected to each other and to the grid connection point by some 21.3km of high voltage cable circuits. The cables run from West Burton 1 and 2 into West Burton 3 where the 400kV substation will be located. From there a 400kV cable runs to the Point of Connection (POC) at West Burton Power Station.
- 2.2.14 The Cable Route Corridor crosses predominantly agricultural land whilst also adopting a route of least resistance in order to avoid unnecessary disruption or severance of land or ecological features. The cable will need to cross a number of obstacles via the use of horizontal directional drilling. The main drilling sites will be located where the cable needs to cross the River Till and the River Trent. Smaller drilling sections may be required for crossing other features such as roads and ditches. The cable route avoids villages such as Sturton Le Steeple and Marton.

Additional Areas within the Order Limits

2.2.15 The Order Limits contain the full land area required to develop, operate, maintain and decommission the Scheme. As such, these also include all access points and visibility splays, as well as any additional land required for the transportation of 'abnormal indivisible loads'.

2.3 Site Surroundings

West Burton 1

- 2.3.1 The surrounding area is predominantly arable farmland, interspersed with a significant number of woodland blocks. Immediately to the east of the Site is North Carlton Covert, a small block of woodland immediately adjacent to the Site's eastern boundary. The settlements at Broxholme and Bransby lie closest to the Site. To the west lie the hamlets of Bransby and Ingleby and to the east lies the village of North Carlton. With the exception of the villages/hamlets mentioned above, the area is relatively sparsely populated with isolated residential properties and farmsteads dotted throughout the surrounding countryside.
- 2.3.2 The land is generally flat, defined by the floodplains of the River Trent and River Till with the limestone escarpment known as "The Ridge" located to the east. The River Till runs in a north/south direction up to the northern boundary of West Burton 1.
- 2.3.3 There are numerous PRoWs that run within 2km of the Site and to the Site boundaries. Public Footpath Brox/198/1 is located to the southwest corner of the Site and runs from Broxholme Lane to Carlton Lane. Public Footpath Brox/197/1 lies directly to the west of the Site connecting Broxholme Lane to the Site boundary. Public Footpaths Brox/196/1 and Scmp/196/1 lie to the west and northwest of the Site, connecting Broxholme Lane with the outskirts of Thorpe in the Fallows. A Public Bridleway, TLFe/31/1, is also located to the northwest as well as a Public Bridleway,



NCar/225/1, located to the east. The A1500 also runs to the north of West Burton 1, separated by a number of fields.

- 2.3.4 The closest scheduled monument is Broxholme medieval settlement and cultivation remains (List Entry Number: 1016797), located immediately adjacent to the southwest of the Site. The closest listed buildings in proximity to the Site are located to the southwest within Broxholme village. These are shown on the Historic Environment Features Plan [REP4-018] and are detailed within ES Appendix 13.5 Heritage Assessment [APP-117 to APP-119].
- 2.3.5 There are no statutory or non-statutory ecological designations within 2km of the Site and no ancient woodland within immediate proximity. West Lindsey AGLV (The Ridge) is located approximately 2.3km east of the Site.

West Burton 2

- 2.3.6 The surrounding area is predominantly arable farmland, interspersed with farms and villages, alongside the larger settlements of Saxilby and Sturton by Stow. The landform is relatively flat with a gentle slope to the east towards the River Till. The nearest settlement is the small village of Broxholme located immediately to the southwest of the Scheme. Around 2.5km to the northwest of the Site lies the settlement of Sturton by Stow and the larger village of Saxilby is located approximately 2.5km to the southwest of the Site. To the west lie the hamlets of Bransby and Ingleby and to the east lies the village of North Carlton. With the exception of these villages/hamlets, the area is relatively sparsely populated with isolated residential properties and farmsteads dotted throughout the surrounding countryside. The landform within the surrounding area is relatively flat with a gentle slope to the east towards the River Till.
- 2.3.7 In closest proximity are the Public Footpaths Brox/198/1 and Brox/197/1, located about 700m east of the Site. Public Footpaths Saxi/203/1, Saxi/207/1 and Saxi/208/1, are all located approximately 700m to the south of the Site.
- 2.3.8 The closest listed building to the site is Grade II Listed Ingleby Chase (Listed Number: 1147263), located close to the Site's northern boundary. Within a 2km proximity there are further Grade I and II Listed Buildings and Scheduled Monuments. These are shown on the Historic Environment Features Plan [REP4-018] and are detailed within ES Appendix 13.5 Heritage Assessment [APP-117 to APP-119].
- 2.3.9 There are no statutory or non-statutory ecological designations within 2km of the Site and no Ancient Woodland within close proximity. West Lindsey AGLV1 (The Ridge) is located approximately 3.6km east of the Site.

West Burton 3

2.3.10 The surrounding area is predominantly arable farmland. The Lincoln Golf Club is located to the southwest of the Site, surrounding the small hamlet of Brampton. A small number of residential properties on the eastern edge of the settlement are located adjacent to the southwestern corner of the Site. Located within the middle



of the Site and straddling the railway line are Stow Park Farm and Marton Moor Farm, two large farmsteads with associated outbuildings and sheds that occupy the arable farmland to the south of the A1500.

- 2.3.11 To the immediate northwest of the Site is the settlement of Marton which occupies the hillside leading down from the arable plateau to the lower lying landform alongside the River Trent. A small number of residential properties on Adams Way and Spafford Close are located alongside the north-western corner of the Site. To the west of the Site, the landform quickly drops away to the A156 and the River Trent. Embankments alongside the Trent help elevate the Trent above the surrounding lowland arable farmland. The eastern extents of the Site occupy the flatter arable plateau that is made up of gently rolling arable fields. With the exception of the villages/hamlets mentioned above, the area is relatively sparsely populated with isolated residential properties and farmsteads dotted throughout the surrounding countryside.
- 2.3.12 To the north, just outside of Marton, lies Public Footpath Mton/69/1, and to the east Public Footpaths Stow/71/2, Stow/71/4, Stow/74/2, Stur/75/1 and Stur/75/2 which connect Stow with Sturton by Stow and the surrounding landscape. To the south, there are no PRoWs other than at the southwest corner of the Site where Public Footpaths Tork/957/1, Tork/779/1 and Tork/96/1 are located between Brampton and Torksey. To the west lies Public Footpaths Bram/99/1, Mton/66/4, Mton/66/1 and the long-distance trail of the Trent Valley Way.
- 2.3.13 The Listed Buildings in closest proximity to the Site are Grade II Signal Box at Stow Park (List Entry Number: 1146606) and Stow Park Station (List Entry Number: 1064058) located 50m north of the Site. Grade II Manor Farmhouse (List Entry Number: 1064084), Priory Cottage (List Entry Number: 1064082), Richards-Havecross Cottages (List Entry Number: 1064081), the Beeches (List Entry Number: 1064080), and The Hermitage (List Entry Number: 1064080) are all approximately 50m southwest of the Site.
- 2.3.14 The Scheduled Monument Medieval Bishop's Palace and Deer Park, Stow Park (List Entry Number: 1019229), is located immediately adjacent to the Site. There are also a number of Scheduled Monuments within 2km of the Site. These are shown on the Historic Environment Features Plan [REP4-018] and are detailed within ES Appendix 13.5 Heritage Assessment [APP-117 to APP-119].
- 2.3.15 There are no statutory or non-statutory ecological designations within 2km of the Site. The nearest Ancient Woodland is 1.2km north of the Site at Gate Burton. Located approximately 350m to the northeast of the Site is West Lindsey AGLV3 (Laughton Wood). This area extends across the countryside to the north of the A1500 across Gate Burton and Knaith.

2.4 Relevant Planning History

2.4.1 The relevant planning history of the land within the Order Limits is limited due to the predominantly agricultural use of the land. Planning history searches of the



Bassetlaw and West Lindsey district councils' web portals were undertaken for the Sites and Cable Route Corridor and are contained within Appendix A: Planning Application History Search West Burton Sites and Appendix B: Planning Application History Search Cable Route Corridor respectively. These appendices are complementary to, compiled within and are to be read alongside this Planning Statement.

- 2.4.2 There are no significant implications arising from the location of the Scheme, upon any of the identified planning permissions.
- 2.4.3 There are various planning applications relating to poultry units at West Burton 2. An initial application (reference: 140380) was submitted and approved in March 2020 for the erection of a single poultry unit for meat production. The Site is accessed via a private highway off Sturton Road which tracks north towards Ingleby Hall Livery. Conditions 3 and 6 of the Permission required the applicant to improve vehicular access to the public highway (being Sturton Road) and to incorporate two passing places for HGVs along the private road respectively. This private highway falls within the Order Limits.
- 2.4.4 Through two subsequent applications (references: 141299 & 141816) requesting the confirmation of compliance with relevant conditions attached to 140380 relating to commencement, the submitted details across the two applications were approved.
- 2.4.5 An application (reference: 143040) was submitted for a further poultry unit for meat production. Having brought the private highway up to the highways standard, as requested by conditions attached to 140380, there were no further conditions relating to highways improvement of the private highway. An application relating to compliance with a drainage condition (reference: 143849) was submitted and approved.
- 2.4.6 A Scoping Opinion Request (reference: 145441) was submitted for an additional poultry unit for meat production in September 2022 and was subsequently revised by a further Scoping Opinion Request (reference: 145936) for the construction of two poultry units for meat production.



3 The Scheme

3.1 Introduction

- 3.1.1 This section describes the Scheme and its main components. It describes the components of the development and describes the activities that would take place during the construction, operational and decommissioning phases of the Scheme.
- 3.1.2 A full description of the proposed Scheme is provided in Chapter 4 of the ES [APP-042].

3.2 Components of the Scheme

- 3.2.1 All of the works that are part of the Scheme are listed in Schedule 1 of the draft DCO [EN010132/EX6EX7/WB3.1_GH]. A summary of the work packages is set out below. The extent of each Work Number is shown on the Works Plans [REP5-035].
 - Work No. 1: Solar Photovoltaic Generating Stations at West Burton 1 (Work No. 1A), West Burton 2 (Work No. 1B), West Burton 3 (Work No. 1C);
 - Work No. 2: Energy Storage Facility at the West Burton 3 Site;
 - Work No. 3: On-site substations at each Site (West Burton 1, Work No. 3A), (West Burton 2 Work No. 3B), (West Burton 3 Work No. 3C);
 - Work No. 4: Works at West Burton Power Station to facilitate the grid connection;
 - Work No. 5: Grid connection cable works connecting the three Sites (Work No.1A 1C), to the main on-site substation at West Burton 3 (Work No. 3C) and subsequently to the Point of Connection (POC) at West Burton Power Station (Work No. 4) including the provision of access tracks, construction laydown areas (construction compounds), jointing bays and fibre optic communications chambers;
 - Work No. 6: Works associated with each of the Sites including fencing, gates, boundary treatment and other means of enclosure; security and monitoring measures including CCTV columns, lighting columns and lighting, cameras, weather stations, communication infrastructure, and perimeter fencing; landscaping and biodiversity mitigation and enhancement measures including planting; improvement, maintenance and use of existing private tracks; laying down of internal access tracks, ramps, means of access and footpaths; temporary footpath diversions; earthworks; sustainable drainage system ponds, runoff outfalls, general drainage and irrigation infrastructure and improvements or extensions to existing drainage and irrigation systems; electricity and telecommunications connections; and secondary temporary construction compounds;
 - Work No. 7: Temporary construction and decommissioning laydown areas within each of the Sites and works associated with these including areas of hardstanding; car parking; site and welfare offices and workshops; security



infrastructure, including cameras, perimeter fencing and lighting; area to store materials and equipment; site drainage and waste management infrastructure (including sewerage); and electricity, water, waste water and telecommunications connections.

- Work No 8: Works to facilitate both temporary construction access and permanent access to the Sites and Cable Route Corridors;
- Work No. 9: Works to create and maintain Habitat Management Areas;
- Work No.10: Works to maintain and enhance a Habitat Management Area; and
- Work No. 11: Works to provide a permissive footpath from the track off Sykes Lane along the Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub including fencing, gates, boundary treatment and other means of enclosure and landscaping and biodiversity mitigation and enhancement measures.
- 3.2.2 The Scheme also includes further associated development in connection with Work Nos. 1 to 11 including fencing, gates, boundary treatment and other means of enclosure; bunds, embankment, trenching and swales; irrigation systems; drainage systems; services and utilities connections; works to alter the course of non-navigable rivers, streams or watercourses; ramps, bridges and means of access; security and monitoring measures; improvement, maintenance and use of existing private tracks; footpath diversions and enhancement; landscaping and related works; habitat creation and enhancement; site establishment and preparation works; earthworks and excavations; works for the protection of buildings and land; tunnelling, boring and drilling works; and other works to mitigate any adverse effects on the construction, maintenance, operation or decommissioning of the Scheme.

3.3 Construction Period Activities

- 3.3.1 The Scheme's temporal timescales (construction, operation and decommissioning) are as follows:
- 3.3.2 The Scheme currently has a grid connection date of 2028 although there is the potential that an earlier connection could be achieved. It is currently anticipated that construction works will commence, at the earliest, in Q4 2024 and will run to Q4 2026. As such, the construction programme for the entire Scheme is anticipated to be 24 months with the potential likelihood of overlapping construction works on the different Sites. This is anticipated to be as follows:



Table 3.1 Indicative Construction programme

Site/Month	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	21	22	23	24
West Burton1																								
West Burton																								
West Burton																								
BESS																								
Grid Connect																								

Shared Cable Route Corridor

- 3.3.3 As noted at ES Chapter 2: EIA Methodology [APP-040], part of the Gate Burton Energy Park cable route and Cottam Solar Project cable route will fall within the cable corridor for the Scheme, in the vicinity of Cottam Power Station. The cumulative environmental effects of the simultaneous or sequential construction of these cables have been assessed in the ES. This is in order to seek to minimise potential environmental effects and identify the benefits of combined construction activities. To accommodate the potential sequential installation of all three projects' ducts and cables, a five-year construction duration is adopted for this, and assessed in the ES. This will be over the period Q4 2024 to Q4 2029. This period has been chosen given that the grid connection date for West Burton is 2028, Gate Burton Energy Park 2028 and Cottam is 2029 and it allows for these works to take place within that period. The installation of each projects' ducts and cables will take place sequentially over a 5-year period. Over this period, it is assumed that haul roads, laydown areas / compounds and bridges remain in situ for the 5-year period. This would represent a worst-case scenario from an assessment perspective given the potential for ongoing construction activities over this period.
- 3.3.4 Main construction laydown areas (sometimes referred to as 'construction compounds') will be located within each Solar Farm Site. Construction laydown areas will also be established at locations along the Cable Route Corridor. The Solar Farm Site laydown areas will consist of compounds of approximately 13,000m2 and will contain offices, mobile welfare units, canteens, storage and waste skips, parking areas and space for storage, download and turning area. There will also be secondary temporary laydown areas progressively established across the Solar Farm Site in each working area. These will be located across the Solar Farm Site and the purpose of each one will be to service the local works. This includes storage for materials, fuel, equipment etc. needed for such works as well as welfare facilities, office space etc. required to avoid unnecessary internal movement of personnel over long distances.



- 3.3.5 The secondary laydown areas will typically be set up ahead of the installation of the PV Arrays, electrical components and cabling, and will be decommissioned as the relevant works in their locality progress and become completed.
- 3.3.6 Construction activities are likely to be carried out Monday to Friday 07:00-18:00 and between 08:00 and 13:30 on Saturdays. However, some activities may be required outside of these times (such as the delivery of abnormal loads, night-time working for cable construction works in public highways or horizontal direction drilling activities). Where possible, construction deliveries will be coordinated to avoid HGV movements during the traditional AM peak hour (08:00-09:00) and PM peak hour (17:00-18:00).

Construction Environmental Management Plan

- 3.3.7 A Construction Environmental Management Plan (CEMP) will be submitted to and approved by the relevant planning authority, and this will be secured by the Requirements in the DCO. The CEMP for each phase will be substantially in accordance with the Outline Construction Environmental Management Plan [EN010132/EX6/WB7.1_DREP6-021] submitted as part of the DCO application. This will ensure the potential construction impacts are minimised.
- 3.3.8 The CEMP will outline the allocated responsibilities, procedures and requirements for the Sites' environmental management. It includes relevant Site-specific method statements, operating practices, and arrangements for monitoring and liaison with local authorities and stakeholders.
- 3.3.9 The Applicant would ensure through the terms of the construction contract that the main contractors undertaking the construction of the Scheme would comply with the CEMP, allocate environmental management responsibilities to a Site manager and ensure that all sub-contractors' activities are effectively managed in accordance with the CEMP.
- 3.3.10 If the Scheme, the Cottam Solar Project and Gate Burton Energy Park Projects are to continue to progress in rough parallel, the Applicant will seek to plan and coordinate any construction activities, via the CEMP and Construction Traffic Management Plan, to reduce environmental impacts, if possible and where practicable. At present, both the Cottam Solar Project and Gate Burton Energy Park projects have been accepted for examination and are therefore ahead of the Scheme.

Operation

- 3.3.11 The Scheme will commence operation at the end of Q4 2026. The Scheme must be decommissioned no later than 60 years from the date of final commissioning and decommissioning is therefore estimated to be no earlier than 2066 and no later than 2086.
- 3.3.12 Once the Scheme is operational, traffic generated by it will be limited to that associated with occasional maintenance work.



- 3.3.13 Movement within the Sites will be by way of quad bike or small, farm utility vehicles. This will be secured via the Outline Operational Environmental Management Plan [REP5-020]. Personnel will visit the Sites from time to time to check the apparatus. No on-site staff will be required to operate the Scheme but there will be limited staff facilities located in the control rooms associated with the substations. Some permanent equipment for monitoring the Sites will be located in the Relay and Control Room. Whilst this would typically be accessed remotely, it would be available for occasional physical access during routine visits.
- 3.3.14 Noise impact is largely limited to the construction phase of the development. There would be a small amount of noise generated by the vehicle movements across the site coupled with the installation of equipment. There will be some noise transmitted from the transformers, substations, tracking panels and Energy Storage but these levels are predicted to be low and are addressed in full in ES Chapter 15: Noise and Vibration [APP-053].

Decommissioning

- 3.3.15 As the Scheme must be decommissioned no later than 60 years from the date of final commissioning, decommissioning is therefore estimated to be no earlier than 2086 and no later than 2086. Decommissioning is expected to take between 12 and 24 months. A 24-month decommissioning period has been assumed for the purposes of a worst-case assessment in this ES, unless specifically stated otherwise. A requirement to decommission the Scheme is secured via a Requirement in the draft DCO.
- The Decommissioning Plan for each Site or phase of decommissioning will be in accordance with the Outline Decommissioning Statement [EN010132/EX6/WB7.2_B].REP6-023]. This will ensure the potential decommissioning impacts are minimised.
- 3.3.17 The solar modules and related built infrastructure, ancillary infrastructure, substations and energy storage will be removed and recycled or disposed of in accordance with good practice and market conditions at that time.
- 3.3.18 The underground ducting within the Cable Route Corridor will be decommissioned but left in-situ to avoid unnecessary intrusion. It is possible to remove the cable itself by extracting it at the joint bays from within the ducting so that the cable can be recycled.

Waste

3.3.19 Waste will be generated during all phases of the development. Solid waste materials generated during construction and decommissioning will be segregated and stored on site prior to transport to an approved, licensed third party landfill and recycling facility. Waste arisings are assessed in ES Chapter 20: Waste [APP-058].

Site Reinstatement



- 3.3.20 Upon decommissioning, the above-ground physical infrastructure at the Solar Farm Sites will be removed and the Solar Farm Sites returned to the landowner. This will include the areas of agricultural land where the agricultural resource has been maintained (and potentially improved) during operation, and the established habitats. Post-decommissioning, the landowner may return the Solar Farm Site to arable use, although it is assumed that established habitats such as hedgerows and woodland would be retained given their potential benefits to agricultural land and the wider farming estate.
- 3.3.21 The 33kV, 132kV and 400 kV cables may be left in situ, depending on the least environmental damaging approach at the time. If these are removed, this would be achieved by pulling the cables out of the ducts, limiting the locations where the surface would need to be disturbed. This same principle will apply to the low voltage cabling throughout the Order Limits. Any cabling removed will be taken to an appropriate facility for recycling.
- 3.3.22 Foundations and other below ground infrastructure will be cut to 1m below the surface to enable future ploughing. Any piles would be removed. Areas of planting and habitats will be maintained by the Applicant until the point of handover to the landowner.
- 3.3.23 Permissive paths would be removed during decommissioning, with the precise timing to be determined by the contractor(s) and communicated to the relevant local authority in accordance with the approved Decommissioning Environmental Management Plan.
- 3.3.24 Some soil profiling may be required, and the land will be contoured in accordance with the approved Decommissioning Environmental Management Plan.
- 3.3.25 If necessary, the soil will be tilled to mitigate for any compaction. Areas where grass does not exist because of the footprint of the previous infrastructure (e.g., the BESS and on-site substations) shall be reseeded with suitable native species, in liaison with the landowner and in accordance with the approved Decommissioning Environmental Management Plan, in order to integrate the newly restored soil into the future land-use.
- 3.3.26 Further detail is set out in the Outline Decommissioning Statement [EN010132/EX6/WB7.2_B].REP6-023]. A Decommissioning Environmental Management Plan (DEMP), to include timescales and transportation methods, will be secured by a Requirement in the DCO and approved by the relevant planning authority.



4 Need and Benefits

4.1 Introduction

- 4.1.1 This section presents a high-level summary of the need for the Scheme. It uses non-technical language and outlines the practical reasons as to why large-scale solar developments, and the Scheme, are needed. The policy drivers relating to the need for the Scheme are described in Sections 5 and 6.2 of this Planning Statement. Section 4.6 lists some of the other benefits of the Scheme and 4.7 describes proposals for a community liaison group. Although it does not form part of the application, Section 4.8 outlines proposals for a community benefit fund.
- 4.1.2 The principal need for the Scheme is centred on the significant contribution it will make to the three important national energy policy aims of:
 - Decarbonisation achieving Net Zero carbon emissions by 2050, requiring deployment of zero-carbon electricity generation at scale. The Scheme will generate large-scale low carbon electricity which could be operational by 2029.
 - Security of supply geographically and technologically diverse supplies. The Scheme will contribute to security of supply due to its large scale; predictable output; ability to complement other renewables; and the efficient opportunity to integrate Energy Storage.
 - Affordability The Scheme will provide large-scale generation at low cost which will provide value for money for end-use consumers.
- 4.1.3 This need is also in the context that the above objectives will need to be delivered during a period where there will be an increasing level of demand for electricity.
- 4.1.4 The Statement of Need **[APP-320]** accompanying the DCO application sets out a detailed compelling case for why the Scheme is urgently required and at the scale proposed. Section 6.2 of this Planning Statement discusses the need for the Scheme in the context of relevant planning and energy policy.

4.2 Meeting an Increasing Demand for Electricity

- 4.2.1 As explained in Section 6 of the Statement of Need **[APP-320]** demand for electricity across England, Wales and Scotland is expected grow in the years ahead for the following reasons:
 - The switching of sources of final-use power for heating and transport from carbon-intensive sources to electricity will increase demand;
 - Carbon-intensive sources of energy are displaced by electrification of other industry sectors, or production of non-carbon energy vectors by use of electricity;
 - The least-cost energy efficiency measures, such as introduction of low-voltage LEDs for lighting, have now been implemented across business and domestic sectors; and



- Economic restructuring away from manufacturing to a service-based economy
 has largely occurred, however the growth of new high technology and highly
 skilled manufacturing, both contributing to national economic growth and
 prosperity, is likely to place additional pressures on the electricity sector.
- 4.2.2 The above is consistent with the observations provided by National Grid Electricity System Operator (NGESO) in their Future Energy Scenarios 2020 and 2021. The government's Energy White Paper: Powering our Net Zero Future published in 2020 also identifies that meeting a possible doubling of electricity demand by 2050 "would require a four-fold increase in clean electricity generation with the decarbonisation of electricity increasingly underpinning the delivery of our net zero target".
- 4.2.3 To enable decarbonisation and achieve net zero by 2050, as required by legislation and policy, paragraph 4.3.7 of the Statement of Need [APP-320] identifies that the power generation sector must urgently both increase in capacity and reduce in carbon intensity on an unprecedented scale.

4.3 Need for Decarbonisation

- 4.3.1 The UK is legally bound through the Climate Change Act (2008) (CCA 2008) to reduce carbon emissions. The CCA 2008 is underpinned by further legislation and policy measures which have developed in the last 13 years. This has been based on an increased need and urgency for decarbonisation to meet the UK's obligations under the Paris Agreement (2015).
- 4.3.2 In October 2018, following the adoption by the UN Framework Convention on Climate Change of the Paris Agreement, the Intergovernmental Panel on Climate Change (IPCC) published a Special Report on the impacts of global warming of 1.5°C above pre-industrial levels. This report concluded that human-induced warming had already reached approximately 1°C above pre-industrial levels, and that without a significant and rapid decline in emissions across all sectors, global warming would not be likely to be contained, and more urgent international action is required.
- 4.3.3 The targets for carbon emissions reduction have tightened more so in the last three years, including a legally binding commitment for the UK to reach net zero carbon emissions by 2050. The Government's Energy White Paper: Powering our Net Zero Future published in 2020 identifies the Government's aim for a fully decarbonised, reliable and low-cost power system by 2050 and that the future energy generation mix for this system is *"likely to be composed predominantly of wind and solar"*.
- 4.3.4 The Scheme would make an important contribution to the delivery of renewable generation technology that is required to decarbonise the energy system and meet the UK's commitments to reduce greenhouse gas emissions and reach net zero carbon emissions by 2050. The Statement of Need [APP-320] sets out the need for decarbonisation at paragraph 5.3.2 and Section 8, and how the Scheme would contribute to this in detail at paragraph 4.7.12 and Section 12.



4.4 The Need to Provide Security of Supply

- 4.4.1 An increasing demand for electricity and an increasing reliance on generation from renewable sources brings with it new challenges in terms of security of supply, i.e., 'keeping the lights on'.
- 4.4.2 Section 8 of the Statement of Need explains the contribution that the Scheme will make to providing security of supply. Firstly, it will supply a significant capacity of zero-carbon generation that is connected to the NETS, thereby contributing to meeting the overall demand for electricity.
- 4.4.3 The Statement of Need explains that although individual renewable assets are variable generators, the generation dependability of a portfolio which consists of different renewable technologies is more stable. In addition, the generation profiles of a diverse range of low-carbon generators would combine to meet seasonal average demand levels without requiring significant and unproductive capital investment or seasonal excess generation.
- 4.4.4 The UK benefits from substantial renewable energy resources, including 40% of Europe's wind resource and areas of developable land which receive high levels of solar irradiation. Wind and solar are also mutually compatible technologies as the weather and climatic conditions in which they generate most of their electricity generally occur at different times. Solar farms generate more electricity in the summer months when it is lighter, and days are longer. Wind farms generate more electricity when it is windy, which is more frequent in the winter months.
- 4.4.5 Even allowing for seasonal variations in the demand for electricity, the Statement of Need explains that models show that solar generation can efficiently make up the shortfall of required generation capacity from wind in the summer months without delivering significant over-generation in winter periods, as would be the case should wind power seek to make up the seasonal shortfall.
- In addition, the Scheme includes an Energy Storage Facility. NPS EN-1 (November 2023) paragraphs 3.3.25 3.3.27 recognise the key role that storage has to play in achieving net zero and providing flexibility to the energy system to help reduce electricity costs and increase reliability. The co-location of solar and storage assets helps provide efficiencies in relation to the use of land and available grid connection capability because essential infrastructure can be shared between the two technologies. The Scheme's grid connection agreement with National Grid provides 20MW of import power capacity which explains the inclusion of 20MW (as opposed to a greater capacity) of electricity storage capability as part of the Scheme. Although import capacity is not available in equal measure with export capacity at this location, the grid connection point has been used to connect renewable generation and storage to the NETS to the maximum extent possible at this point in time. This is in line with NPS EN-3 (November 2023) paragraphs 2.10.25 and 2.10.26 which states: "To maximise existing grid infrastructure, minimise disruption to existing local



community infrastructure or biodiversity and reduce overall costs the applicants may choose a site based on nearby available grid export capacity."

- 4.4.7 *"Where this is the case, applicants should consider the cumulative impacts of situating a solar farm in proximity to other energy generating stations and infrastructure".*
- 4.4.8 The Chairman of Ofgem stated in 2018:

"As the energy landscape transforms, our energy grids have got to adapt with it. In the past if new generators wanted connections or demand grew, the solution was simply to build more infrastructure. We will still need some new grid capacity in future, but we must make better use of what we have got already. If we don't then the costs to consumers of managing these changes will go up and there will continue to be delays in getting renewables and other technologies such as storage connected to the grids." (Burgess, A. Getting more out of our grid capacity. https://www.ofgem.gov.uk/news-blog/our-blog/getting-more-out-our-grid-capacity, August 2018).

The national need for solar and energy storage far exceeds the current pipelines for projects of both technologies. Therefore, not making use of the connection for 20MW of storage at West Burton means 20MW would need to be connected somewhere else, which might be significantly more expensive than using the 20MW capacity already available at West Burton. The Scheme therefore maximises the existing grid infrastructure and plays an essential role in contributing to the three pillars of energy policy: decarbonisation, security of supply, and affordability. The Scheme's proposed solar generation and energy storage are ideally suited to support the maintenance of a safe, secure and economic electricity system. Further detail on the Energy Storage element of the Scheme is set out at Section 11.5 and 11.6 of the Statement of Need [APP-320].

4.5 The Need for Large Scale Solar to Deliver Low-Cost Energy

- 4.5.1 The cost of solar generation is already very competitive against the cost of other forms of conventional and low-carbon generation, both in Great Britain and more widely. The Statement of Need [APP-320] also identifies at paragraph 10.5.4 that single large-scale solar schemes deliver more quickly and at a lower unit cost than multiple independent schemes which make up the same total capacity, bringing forward carbon reduction and economic benefits in line with government policy.
- 4.5.2 In terms of affordability, internationally and nationally, there is an ongoing trend of solar generation assets becoming bigger and cheaper, with each subsequent project demonstrating that solar generation at the size and scale proposed works in real life. Increased scale and size provide increased decarbonisation benefits and commercial benefits to consumers as set out at Section 10.4 of the Statement of Need [APP-320].
- 4.5.3 In summary, solar generation such as the Scheme can be provided at a large scale for a relatively low cost which, in relation to other electricity generation infrastructure developments, provides value for money for end-use consumers.



4.6 Other Benefits of the Scheme

- 4.6.1 In addition to meeting the urgent national need for secure and affordable low carbon energy infrastructure, the Scheme will deliver other benefits, many of which will be delivered as a result of the Scheme's careful design. These include:
 - A significant Net Gain for biodiversity, with 86.80% gains provided in habitat, 54.71% gains in hedgerow and 33.25% gains in river units, in line with local and national planning policies. Post development, the Sites will comprise the following proposed landscaping habitats: enhancement of existing hedgerows and ditches, native hedgerow with trees, native shrub planting, woodland planting, native scattered trees, long term meadow creation (partially panelled), flower rich pollinator mix, tall herb mix, tussock mix, set aside, diverse meadow mix, proposed wildlife ponds, and enhancement of existing ponds. See Biodiversity Net Gain Report [APP-088] for the detailed assessment.
 - A new permissive footpath to run from Track off Sykes Lane along the Codder Lane Belt and then south and west to rejoin Sykes Lane opposite Hardwick Scrub (Work No 11). The design and implementation of the permissive path is set out in the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025] and secured by a Requirement in the DCO.
 - The temporary employment generated by the Scheme's construction is assessed to be approximately 296 direct FTE jobs per annum as set out within Section 18.7 of ES Chapter 18: Socio Economics, Tourism and Recreation [APP-056].
 - During its operational lifetime, the Scheme is anticipated to generate a modest quantum of labour, related to ongoing operational management and site management. It is projected that the Scheme will require a gross 12 FTE direct employees per annum.
 - A Skills, Supply Chain and Employment Plan will be prepared prior to the commencement of construction. This will set out measures that the Applicant will implement to advertise and promote employment and training opportunities associated with the Scheme in construction and operation locally. It will be secured through a requirement included in the DCO for the Scheme. The Outline Skills, Supply Chain and Employment Plan [EN010132/EX6/WB7.10_BREP6-027] forms the basis for this.

4.7 Community Liaison Group

- 4.7.1 A community liaison group (CLG) will be established. This will enable local community representatives to have a formal channel for monitoring and influencing the construction, operational and decommissioning aspects of the Scheme.
- 4.7.2 The CLG is intended to provide an opportunity for regular and formal dialogue between the Applicant and the local community's representatives in relation to the



- construction and operational aspects of the Scheme. It is envisaged that local community representatives forming the CLG will be principally from the villages and communities neighbouring the Order Limits.
- 4.7.3 CLG meetings will enable members of the group to raise and formally record any issues that may arise in relation to the Scheme. It will also provide a regular forum for the Applicant to update interested parties about the construction and operation of the Scheme. The details of the CLG will be set out in the Construction Environmental Management Plan and are outlined within the Outline Construction Environmental Management Plan [EN010132/EX6/WB7.1_D].REP6-021]. The delivery of the CLG will be secured via a Requirement of the DCO.

4.8 Community Fund

4.8.1 The Applicant has also committed to providing a Community Benefit Fund (CBF). The CBF does not form part of the DCO application, and this funding is not required to mitigate the impacts of the Scheme. Therefore, the SoS cannot, and must not, apply any positive weight to the CBF when balancing the positives and negatives of the Scheme. The CBF is therefore not taken into account in consideration of the planning balance within this Planning Statement. It will, however, be available to fund local community projects.

Planning Statement: Revision C

April May 2024



5 Legislative and Policy Context

5.1 Introduction

5.1.1 This section outlines the legislative framework and the planning policy context for the Scheme. Section 5.2 sets out the relationship of the Scheme with the PA 2008. Sections 5.3 and 5.4 introduce national and local planning policy and other documents that the Applicant expects to be important and relevant to the decision and that are considered in this Planning Statement. Section 5.5 introduces other national policy documents which the SoS may consider to be important and relevant to their decision.

5.2 Legislative Context

- 5.2.1 The PA 2008 provides the legislative basis and defines the application process under which consent for a NSIP is sought. The PA 2008 sets out that projects meeting certain defined criteria are classified as NSIPs. It requires developers of NSIPs to obtain a DCO to permit the construction, operation and maintenance of their project.
- 5.2.2 The Scheme is defined as an NSIP under Sections 14(1)(a), 15(1) and 15(2) of the PA 2008 by virtue of the facts listed below:
 - The Scheme comprises the construction of a generating station (Section 14(1)(a) of the PA 2008);
 - It would be located in England (Section 15(2)(a) of the PA 2008);
 - It would not generate electricity from wind (Section 15(2)(aa) of the PA 2008);
 - It would not be an offshore generating station (Section 15(2)(b) of the PA 2008);
 and
 - Its capacity would be more than 50MW (Section 15(2)(c) of the PA 2008).
- 5.2.3 Section 115 of the PA 2008 provides that development consent may be granted for "development for which development consent is required" or for "associated development". In the case of the Scheme the development which constitutes "development for which development consent is required" is described as Work No.1 in Schedule 1 of the Draft DCO [EN010132/EX6EX7/WB3.1_GH]. This constitutes the NSIP for which development consent is required, being a ground mounted solar photovoltaic generating station with a gross electrical output capacity of over 50 megawatts,
- Works Nos. 2 to 11, including Work No. 2 (Energy Storage Facility), are associated development. Further details as to why the Applicant considers that Work Nos. 2 to 11 constitute associated development are set out in the Draft Explanatory Memorandum [EN010132/EX6/WB3.2_DREP6-013].

The Environment Act 2021



- 5.2.5 The Environment Act 2021 gained Royal Assent on 9 November 2020. It provides targets, plans and policies for improving the natural environment although the relevant policies are not yet in force. These include:
 - Establishing the Office for Environmental Protection, which states that its purpose is to protect and improve the environment by holding government and public authorities to account.
 - Increase local powers to tackle sources of air pollution.
 - Protect nature and improve biodiversity, including a requirement for 10% biodiversity net gain for developments consented under the Town and Country Planning Act 1990 and the Planning Act 2008.On 21 February 2023, the Government published a response to the consultation on biodiversity net gain (BNG) regulations and implementation where it was confirmed that the Government will keep its current position, with the requirement to be in place no later than November 2025.
 - Extend producer responsibility, ensure a consistent approach to recycling, introduce deposit return schemes, and introduce charges for specified single use plastic items.
 - Secure long-term, resilient water and wastewater services, including through powers to direct water companies to work together to meet current and future demand.

5.3 Policy Context

- 5.3.1 National Policy Statements (NPS) set out the policy basis for NSIP developments. These are sector specific, covering: energy; transport; and water, wastewater and waste. There are six Energy NPSs, each covering one of the following matters: overarching energy policy; fossil fuels; renewable energy; oil and gas supply and storage; electricity networks; and nuclear power.
- 5.3.2 The Energy NPSs are specific in terms of which energy generation technologies they cover. As previously explained in Section 1.3 above, there is currently no NPS in effect that specifically includes solar development. At the point of finalising this Planning Statement, the application for a DCO is therefore required to be decided in accordance with Section 105 of the PA 2008. This states that in deciding an application for a DCO where an NPS does not exist for the type of development applied for, the SoS must have regard to the following:
 - any local impact report (Section 105(2)(a) of the PA 2008);
 - any matters prescribed in relation to development of the description to which the application relates (Section 105(2)(b) of the PA 2008); and
 - any other matters which the SoS thinks are both important and relevant to their decision (Section 105(2)(c) of the PA 2008).



- 5.3.3 Each of the Host Authorities will have the opportunity to prepare a Local Impact Report (LIR) following submission of the DCO application.
- 5.3.4 The prescribed matters referred to in Section 105(2)(b) of the PA 2008 are set out in the Infrastructure Planning (Decisions) Regulations 2010 (as amended) (the Decisions Regulations). The Regulations that are of relevance to the Scheme are:
 - Regulation 3 Having regard to the desirability of preserving listed buildings and schedule monuments and their settings as well as preserving or enhancing the character or appearance of conservation areas where the development would affect these; and
 - Regulation 7 Having regard to the United Nations Environmental Programme Convention on Biological Diversity of 1992.
- 5.3.5 Consideration of the impact of the Scheme on listed buildings, conservation areas and scheduled monuments and their settings is assessed by ES Chapter 13: Cultural Heritage [APP-051] and discussed in Section 6.6 of this Planning Statement and takes account of the desirability of their preservation, as per Regulation 3 of the Decisions Regulations. The impact of the Scheme on biological diversity is assessed by ES Chapter 9: Ecology and Biodiversity [APP-047] and is discussed in Section 6.9 of this Planning Statement, taking account of Regulation 7 of the Decisions Regulations.
- 5.3.6 With regard to Section 105(2)(c) of the PA 2008, it is likely that the SoS will consider national and local planning policies amongst the other matters that are important and relevant to their decision. The national and local policy context is discussed in Sections 5.4 to 5.7 of this Planning Statement.

5.4 National Planning Policy

5.4.1 This section sets out the national planning policy documents that are considered in this Planning Statement. These comprise the Energy NPSs and the National Planning Policy Framework.

Energy National Policy Statements

- 5.4.2 Whilst none of the Energy NPSs in effect at the time of writing this Planning Statement specifically relate to solar development, this Planning Statement considers the conformity of the Scheme with the NPSs listed below, to the extent that they are likely to be important and relevant to the SoS's decision.
 - 1. Overarching National Policy Statement for Energy (EN-1) (NPS EN-1 2011);
 - 2. National Policy Statement for Renewable Energy (EN-3) (NPS EN-3 2011); and
 - 3. National Policy Statement for Electricity Networks Infrastructure (EN-5) (NPS EN-5 2011).
- 5.4.3 The Energy NPSs were designated on 19 July 2011. They set out matters, principles and impacts that should form the basis of the SoS's decision on DCO applications for Energy NSIPs.



- NPS EN-1 (2011) sets out general principles and impacts to be taken into account for all types of energy NSIP development covered by the Energy NPSs. It forms the primary basis for determining if development consent should be granted for development in the energy sector. NPS EN-1 (2011) states at paragraph 3.3.15 that in order to meet our obligations for 2050, there is an urgent need to bring forward low carbon energy NSIPs as soon as possible. NPS EN-1 (2011) goes on to add at paragraph 3.4.5 that new renewable energy projects are needed urgently to meet the demand for electricity generation in the United Kingdom (UK), and to reduce greenhouse gas emissions from electricity generation to meet the Government's decarbonisation targets.
- NSIP solar developments have the potential to make a direct contribution to meeting the objectives of NPS EN-1 (2011). As set out at paragraph 2.1.1 of NPS EN-1 (2011), these are to help meet the Government's objectives to deliver carbon emission reductions, energy security and affordability. Therefore, NPS EN-1 (2011) should be considered of primary importance and relevance to the Scheme and the SoS's decision.
- 5.4.6 NPS EN-1 (2011) sets out at paragraph 4.1.2 that the SoS should start with a presumption in favour of approving DCO applications for energy NSIPs. It states that the presumption applies unless any more specific and relevant policies set out in the relevant NPSs clearly indicate that consent should be refused or in other limited circumstances which include:
 - lead to the UK being in breach of its international obligations;
 - be in breach of any statutory duty that applies to the IPC;
 - be unlawful;
 - result in adverse impacts from the development outweighing the benefits; or
 - be contrary to regulations about how its decisions are to be taken.
- 5.4.7 NPS EN-3 (2011) sets out additional policies for renewable energy infrastructure that should be read in addition to the overarching policies set out in NPS EN-1 (2011). However, it does not include solar energy projects within its scope and explains that at the time of designation in 2011, types of onshore renewable energy generation not specifically covered within the document were excluded as they were not technically viable at a scale of more than 50MW at the time it was written. Consequently, there are no technology specific policies in the adopted NPS EN-3 (2011) that are relevant to the Scheme. However, solar technology has now advanced to an extent that it is now viable at a nationally significant (>50MW) scale.
- 5.4.8 NPS EN-5 (2011) principally concerns high voltage long distance transmission and distribution infrastructure. However, it also sets out at paragraph 1.8.2 that development which "constitutes associated development for which consent is sought along with an NSIP such as a generating station..." is also covered by the NPS. NPS EN-5 (2011) is considered important and relevant due to the inclusion within the



Scheme of inverters, transformers, switchgear, cabling, and substations that form part of the Scheme.

5.4.9 The Energy NPSs were prepared specifically to address the particular balance of impacts and benefits likely to emerge from energy projects that are of such a scale that their contribution to meeting the government's energy objectives is of national significance. As such, the Applicant considers NPS EN-1 (2011) and NPS EN-5 (2011) to be important and relevant to the determination of the application, and to form the primary decision-making framework for the Scheme. However, the weight attributed to them should be reduced following the designation of November 2023 Energy National Policy Statements which cover solar photovoltaic NSIPs.

Energy National Policy Statements (November 2023)

- 5.4.10 The Government has updated the Energy NPSs. It did this to reflect its policies and strategic approach for the energy system that is set out in the Energy White Paper (December 2020), and to ensure that the planning policy framework enables the delivery of the infrastructure required for the country's transition to net zero carbon emissions. As part of the Energy NPS review process, the Government published a suite of Draft Energy NPSs for consultation on 6 September 2021.
- 5.4.11 An updated version of the Energy NPSs were published on 22 November 2023 following the government's response to the March 2023 consultation on the draft statements. The November 2023 NPSs were designated on 17 January 2024. These include National Policy Statement for Renewable Energy (EN-3) (NPS EN-3, November 2023), which includes specific policies for solar photovoltaic generation NSIPs. The designation of NPS EN-3 (November 2023) has brought solar NSIP developments into the coverage of the Energy NPSs. However, the transitional arrangements mean that the SoS will still be required to decide the application for the Scheme in accordance with the matters set out under S105 of the PA 2008.
- 5.4.12 The Applicant considers the following November 2023 Energy NPSs to be important and relevant matters in the SoS's determination of the application:
 - 1. Overarching National Policy Statement for Energy (EN-1) (NPS EN-1 November 2023),
 - 2. National Policy Statement for Renewable Energy (EN-3) (NPS EN-3 November 2023); and
 - 3. National Policy Statement for Electricity Networks Infrastructure (EN-5) (NPS EN-5 November 2023).
- 5.4.13 Further, the Applicant considers that the above November 2023 Energy NPSs should be given significant weight in the planning balance and when applying the consideration of matters which are important and relevant pursuant to section 105 of the PA 2008, for the following three main reasons:
- 5.4.14 Firstly, they set out policy for Energy NSIPs that reflects the Government's current energy strategy and energy policies. They provide the planning policies that are



needed to facilitate the delivery of the energy infrastructure that is required for the Government's objectives for the energy system to be met.

- 5.4.15 Secondly, NPS EN-3 (November 2023) sets out a policy context that is directly relevant to solar NSIPs such as the Scheme. This means that NPS EN-3 (November 2023) and NPS EN-1 (November 2023) will be the only statutory planning policy documents that are directly relevant to the Scheme (or any solar NSIP). The 2011 NPSs do not include policies specifically relating to solar development, and the National Planning Policy Framework (NPPF) and local Development Plan Documents concern themselves with developments that are of local or regional (and not national) significance.
- Thirdly, given the above, the November 2023 Energy NPSs have been designated on 17 January 2024 before the DCO application is decided. The transitional arrangements set out by paragraph 1.6.2 of NPS EN-1 (November 2023) explain that for any application accepted for examination before designation of the November 2023 NPSs, the NPSs, which were enacted in 2011, should have effect. However, paragraph 1.6.3 of NPS EN-1 (November 2023) sets out that: "any emerging draft NPSs (or those designated but not yet having effect) are potentially capable of being important and relevant considerations in the decision-making process. The extent to which they are relevant is a matter for the relevant Secretary of State to consider within the framework of the Planning Act 2008 and with regard to the specific circumstances of each development consent order application."
- 5.4.17 The Applicant expects that the specific circumstances of this DCO application are such that NPS EN-1 (November 2023) and NPS EN-3 (November 2023) will be important and relevant matters and will be given significant weight in the Examining Authority's recommendation and the SoS's decision. Supplementary statements to this Planning Statement may be needed once the November 2023 NPSs are designated.
- In terms of content, NPS EN-1 (November 2023) sets out general principles and impacts to be taken into account for all types of energy NSIPs covered by the Energy NPSs. Following its designation on 17 January 2024, it forms the primary basis for determining if development consent should be granted and is underpinned by the principle that the development of large-scale renewable energy generation infrastructure is urgently needed for the Government's targets and commitments for the energy system to be met. It sets out at paragraph 3.3.21 that, along with wind; solar electricity generation will help to reduce costs and provide a clean and secure source of electricity supply, and that a secure, reliable, affordable, net zero consistent system in 2050 is likely to be composed predominantly of wind and solar. Paragraph 4.1.2 sets out a presumption in favour of granting consent to applications for energy NSIPs unless any more specific and relevant policies set out in the relevant NPSs clearly indicate that consent should be refused or in other limited circumstances.



- 5.4.19 NPS EN-1 (November 2023) paragraph 4.1.5 states that in considering any proposed development and when weighing the adverse impacts against its benefits, the Secretary of State should take into account the following:
 - "its potential benefits including its contribution to meeting the need for energy infrastructure, job creation, reduction of geographical disparities, environmental enhancements, and any long-term or wider benefits
 - its potential adverse impacts, including on the environment, and including any long-term and cumulative adverse impacts, as well as any measures to avoid, reduce, mitigate or compensate for any adverse impacts, following the mitigation hierarchy."
- 5.4.20 NPS EN-1 (November 2023) paragraph 4.1.6 goes on to state that in this context, the Secretary of State should take into account environmental, social and economic benefits and adverse impacts, at national, regional and local levels. Where the Secretary of State considers that there would still be residual adverse effects after the implementation of mitigation measures, those residual effects should be weighed against the benefits of the proposed development.
- 5.4.21 Section 4.2 of NPS EN-1 (November 2023) sets out the principles for environmental assessment of NSIPs. Paragraph 4.3.9 states that the relevance or otherwise to the decision-making process of the existence (or alleged existence) of alternatives to the proposed development is in the first instance a matter of law. It clarifies that from a policy perspective the NPS does not contain any general requirement to consider alternatives, or to establish whether the proposed project represents the best option.
- Other matters covered by NPS EN-1 (November 2023) include health impacts (Section 4.4) and environmental and biodiversity net gain (Section 4.6). Paragraph 4.6.6 states: "Energy NSIP proposals, whether onshore or offshore should seek opportunities to contribute to and enhance the natural environment by providing net gains for biodiversity and the wider environment where possible." Climate change adaptation and resilience is dealt with at section 4.10 of NPS EN-1 (November 2023) and grid connection at section 4.11. In relation to climate change adaptation, paragraph 4.10.9 states the Environmental Statement should set out how the proposal will take account of the projected impacts of climate change in accordance with the EIA Regulations.
- 5.4.23 NPS EN-3 (November 2023) sets out additional policies for renewable energy infrastructure, including policies specific to the development of solar NSIPs at paragraphs Section 2.10. These include matters that applicants should consider in selecting a site, how assessments should be undertaken and how mitigation should be provided. Section 2.10 sets out the types of impact considered of importance for solar projects. These comprise of biodiversity and nature conservation, landscape, visual and residential amenity, glint and glare, cultural heritage, construction including traffic and transport noise and vibration. NPS EN-3 (November 2023)





should be read in conjunction with the overarching policies set out in NPS EN-1 (November 2023).

5.4.24 Like NPS EN-5 (2011), NPS EN-5 (November 2023) deals with transmission and distribution infrastructure that covers long distances and is at a high voltage. It sets out at paragraph 1.6.2 that it also covers relevant associated development to generation NSIPs such as substations. It is likely that NPS EN-5 (November 2023) will be considered important and relevant in respect of the electrical infrastructure that forms part of the Scheme.

5.5 National Planning Policy Framework

- 5.5.1 This Planning Statement considers the conformity of the Scheme with the NPPF to the extent that it is likely to be important and relevant in the SoS's decision.
- The NPPF was revised in September and December 2023. References in this Planning Statement are to the December 2023 version of the NPPF. and sets out the Government's planning policies for England. It was written to guide the development of local planning policy documents and is a material consideration in the determination of planning applications under the Town and Country Planning Act 1990 (TCPA 1990). As such, its policies were designed with development that is of a scale so as to be of local or regional significance in mind. NPPF Paragraph 5 makes it clear that the document does not contain specific policies for NSIPs and that applications in relation to NSIPs are to be determined in accordance with the decision-making framework set out in the PA 2008 and relevant NPSs, as well as any other matters that are considered both important and relevant.
- 5.5.3 Given the above, the NPPF is considered to be important and relevant where policies are applicable to the Scheme but is to be given less weight in the SoS's decision-making process than the relevant policies in the Energy NPSs (2011 and November 2023).

5.6 Summary

5.6.1 Overall, all three suites of national policy documents are likely to be important and relevant to the SoS decision. The Applicant expects that the Energy NPSs will be attributed most weight when the application is determined under Section 105 of the Planning Act, and that the NPPF although less relevant, will also be important, particularly where it is represents a change in policy approach since the adopted Energy NPSs were drafted. The Applicant also considers that considerable weight should be attached to the November 2023 Energy NPSs, since these represent the only national policy that reflects an up-to-date energy policy position.

5.7 National Infrastructure Planning Guidance

- 5.7.1 There are a range of guidance documents published by Government that relate to the Planning Act 2008 process. Those considered of most relevance to the Scheme include:
 - Guidance on procedural requirements for major infrastructure projects (2020).



- Planning Act 2008: changes to Development Consent Orders (updated 2015).
- Planning Act 2008: guidance on the pre-application process for major infrastructure projects (updated 2015).
- Planning Act 2008: examination of applications for development consent (updated 2015).
- Planning Act 2008: procedures for the compulsory acquisition of land (2013).
- Planning Act 2008: associated development applications for major infrastructure projects (2013).
- Planning Act 2008: application form guidance (2013).

5.8 Planning Practice Guidance (PPG)

5.8.1 The Planning Practice Guidance supports the policies set out within the National Planning Policy Framework discussed at Section 5.5 above. The guidance covers a range of topics including climate change, renewable and low carbon energy, environmental impact assessment, flood risk, historic environment, light pollution, minerals, natural environment, noise, transport and waste.

5.9 Local Planning Policy

- 5.9.1 This Planning Statement considers the conformity of the Scheme with the following Development Plan Documents (DPDs) to the extent that they are likely to be important and relevant in the SoS's decision.
- 5.9.2 Host Authority Planning Policies are drawn from the following documents:
 - Central Lincolnshire Local Plan (CLLP) (Adopted April 2023);
 - Bassetlaw District Council Core Strategy & Development Management Policies DPD (BDCSDMP) (Adopted 2011);
 - Emerging Draft Bassetlaw Local Plan 2020-2037 (DBLP) (Publication Version) August 2021, Addendum January 2022 and Second Addendum May 2022;
 - Nottinghamshire Minerals Local Plan (NMLP) (Adopted March 2021); and
 - Lincolnshire Minerals and Waste Local Plan (LMWLP) (Core Strategy & Development Management Policies (June 2016) and Site Locations (December 2017).
- 5.9.3 Neighbourhood Plans covering part of the Order Limits comprise:
 - Sturton by Stow Parish Council and Stow Parish Council. Sturton by Stow and Stow Neighbourhood Plan 2019 2036 (Final Approved Version) (Adopted July 2022). Gainsborough: West Lindsey District Council.
 - Saxilby with Ingleby Parish Council. Saxilby with Ingleby Neighbourhood Plan 2016 2036 (Adopted February 2017). West Lindsey District Council.



- Sturton le Steeple Parish Council. Sturton Ward Neighbourhood Plan 2021 2037 (Adopted November 2021). Bassetlaw District Council.
- Treswell and Cottam Parish Council. Treswell and Cottam Neighbourhood Plan Referendum Version (Adopted February 2019). Bassetlaw District Council. It is acknowledged that this Neighbourhood Plan is undergoing review. As of early 2022, the Neighbourhood Plan is in a Pre-Submission Draft form. The Draft Plan has not been submitted as of March 2023.
- 5.9.4 Neighbourhood Plans within 2km of the Order Limits comprise:
 - Brattleby Parish Council. Brattleby Neighbourhood Plan 2016 2036 (Adopted November 2017). West Lindsey District Council.
 - Lea Parish Council. Lea Neighbourhood Plan 2017 2036 (Final Approved Version) (Adopted January 2018). West Lindsey District Council.
 - Gainsborough Town Council. Gainsborough Town Neighbourhood Plan 2020
 2036 (Adopted June 2021). West Lindsey District Council.
 - Rampton & Woodbeck Parish Council. Rampton & Woodbeck Neighbourhood Plan 2019 – 2037 (Adopted May 2021). Bassetlaw District Council.
- 5.9.5 Appendix D, Local Planning Policy Accordance Table, appended to this Planning Statement sets out the relevant adopted and draft local planning policies in full and sets out the accordance of the Scheme against the policies.
- 5.9.6 As with the NPPF, DPDs are prepared to guide decision making on planning applications submitted to Local Planning Authorities, rather than DCO applications for energy NSIPs which are to be decided by the SoS. DPDs and other local policies may be important and relevant to the SoS's decisions, particularly where the document contains a policy that identifies an allocated site, a safeguarded land use, or an environmental designation that may affect the assessment of the likely impact of the Scheme.

5.10 Supplementary Planning Documents and other local strategies

- 5.10.1 Other relevant Supplementary Planning Documents and strategies are as follows:
 - Greater Lincolnshire Enterprise Partnership Strategic Economic Plan (2016 Refresh);
 - Corporate Plan 2019 2023 West Lindsey District Council;
 - Lincolnshire Joint Health and Wellbeing Strategy (June 2018);
 - Lincolnshire Joint Strategic Needs Assessment 2021;
 - Lincolnshire Biodiversity Action Plan; 2011 2020 (3rd edition);
 - Lincolnshire Local Transport Plan 5;
 - Gainsborough Transport Strategy May 2022-2036; and
 - Joint Lincolnshire Flood Risk and Drainage Management Strategy 2019-2050.



5.11 Other Policy and Legislation

5.11.1 This section sets out legislation and policy, other than planning legislation and policy, that the Applicant considers is likely to be important and relevant to the SoS's decision.

Climate Change Act 2008

- 5.11.2 The Government, through the Climate Change Act 2008 (CCA2008), made the United Kingdom the first country in the world to set legally binding carbon budgets, aiming to cut emissions (versus 1990 baselines) by 34% by 2020 and at least 80% by 2050, "through investment in energy efficiency and clean energy technologies such as renewables, nuclear and carbon capture and storage" [11, Five Point Plan].
- 5.11.3 CCA2008 is underpinned by further legislation and policy measures. Many of these have been consolidated in the UK Low Carbon Transition Plan (2009) [11], and UK Clean Growth Strategy (2017)

Energy White Paper: Powering our Net Zero Future (2020)

- 5.11.4 The Energy White Paper published in December 2020 is one of the more recent Government policies setting out how the UK will reach net zero emissions by 2050.
- 5.11.5 The Paper explains that it is likely that overall demand for electricity will double by 2050 due to the electrification of other sectors such as transport and heating. On page 42, it states that meeting this demand by 2050 would require "a four-fold increase in clean electricity generation with the decarbonisation of electricity increasingly underpinning the delivery of our net zero target".
- 5.11.6 It identifies the Government's aim for a fully decarbonised, reliable and low-cost power system by 2050 and that market conditions will determine the best solutions for very low emissions and reliable supply, at a low cost to consumers.
- 5.11.7 The Paper explains that the Government is not targeting a particular generation mix but commits the Government to maintaining the market conditions which stimulate the cost reductions that have been seen in the renewable energy market over the last five years. It does, however, state that it is possible to determine key characteristics of the future generation mix at this stage identifying on page 43 that a "low-cost, net zero consistent system is likely to be composed predominantly of wind and solar". It highlights that this will need to be complemented by technologies which provide power, or reduce demand, to manage intermittency. It states that currently this includes "nuclear, gas with carbon capture and storage and flexibility provided by batteries, demand side response, interconnectors and short-term dispatchable generation providing peaking capacity, which can be flexed as required", thereby also highlighting the role of battery storage in the energy mix.
- 5.11.8 This Paper highlights the government's commitment to solar to achieve net zero targets and the need to provide this urgently.



National Infrastructure Strategy (2020)

- 5.11.9 The National Infrastructure Strategy (NIS) published in November 2020 sets out plans to transform the UK's infrastructure. The Strategy is the Government's response to recommendations made by the National Infrastructure Commission (NIC), which was set up to provide impartial, expert advice to the government on long-term infrastructure priorities. In July 2018, the NIC published a National Infrastructure Assessment which provided the foundation for many of the measures included within the NIS.
- 5.11.10 One of the aims of the NIS is to achieve net zero carbon emissions by 2050. The Government acknowledges in the NIS that to deliver net zero, the share of generation from renewables needs to dramatically increase. It identifies that this can be achieved by the provision of greater generation capacity from onshore wind and solar. As recommended by the NIC, the NIS sets out plans to include solar PV in the next auction round (2022) for Contracts for Difference (CfD), which is the Government's main mechanism for supporting low-carbon electricity generation. This incentivises investment in renewable energy by providing developers of projects with high upfront costs and long lifetimes with direct protection from volatile wholesale prices, and they protect consumers from paying increased support costs when electricity prices are high.
- 5.11.11 The NIS demonstrates the Government's commitment, including a financial commitment, to supporting solar generation now.

A Green Future: Our 25 Year Plan to Improve the Environment (2018)

- 5.11.12 The 25 Year Environment Plan published in 2018 sets out the Government's 25-year plan to improve the environment within a generation.
- 5.11.13 It sets out 10 goals which include the achievement of: clean air; clean and plentiful water; thriving plants and wildlife; reduced risk of harm from environmental hazards like flooding and drought; the more sustainable and efficient use of resources from nature; enhanced beauty, heritage and engagement with the natural environment; mitigation and adaption to climate change; minimisation of waste; management of exposure to chemicals; and enhanced biosecurity.
- 5.11.14 Six key areas of policy are set out in the plan and include:
 - Using and managing land sustainably (including embedding an 'environmental net gain' principle for developing and measuring natural capital and reducing flood risk).
 - Recovering nature and enhancing the beauty of landscapes (including developing a Nature Recovery Network and reviewing National Parks and AONBs).
 - Connecting people (including children) with the environment to improve health and wellbeing (including encouraging children to be close to nature, both in and out of school and greening out cities).



- Increasing resource efficiency and reducing pollution and waste (including achieving zero avoidable plastic waste by end of 2042).
- Securing clean, productive and biologically diverse seas and oceans (including a post-Brexit new sustainable fisheries policy).
- Protecting and improving the global environment (including providing 'international leadership and leading by example' and 'leaving a lighter footprint on the global environment).
- 5.11.15 This plan highlights the Government's support for the reduction in the UK's carbon footprint; protection and enhancement of the natural environment; and ensuring land is managed with environmental gains which is of relevance to the Scheme.

Climate Change Committee. The Sixth Carbon Budget: The UK's path to Net Zero. 2020

- 5.11.16 The UK Government has set five-yearly carbon budgets which currently run until 2037. On announcing the adoption of the Committee on Climate Change's recommendations for the sixth Carbon Budget in April 2021, the UK set the world's most ambitious national climate change target into law.
- 5.11.17 The UK has met its first and second carbon budgets and is currently on track to outperform the third (2018 to 2022) partly attributable to effective policy, but also attributed to changes in the applicable Emissions Trading Scheme(s) and the impact of COVID-19 on emissions.

UN Climate Change Conference COP26. COP26: The Glasgow Climate Pact. 2021

- 5.11.18 COP 26 agreed various outcomes relating to climate change mitigation: setting out the steps and commitments that Parties will take to accelerate efforts to reduce emissions "to keep 1.5 degrees in reach". Key achievements at COP26 under the theme of mitigation include:
 - Over 90% of world GDP and around 90% of global emissions are now covered by net zero commitments and 153 countries have put forward new or updated emissions NDCs, which collectively cover around 80% of the world's greenhouse gas emissions. Net Zero is a global endeavour and the world is getting on board;
 - The importance of action now to address the urgency of climate change and drive emissions down before 2030 was cemented in an agreement from all parties to revisit and strengthen their current emissions targets to 2030, in 2022;
 - The role of clean electricity in delivering climate action, and the importance of driving down emissions from fossil fuel generators as well as increase capacity of renewable generators, was acknowledged in the negotiated agreement by 190 countries at COP26 to "phase down coal power". Further commitments to cease international coal finance and direct public support of unabated fossil fuel energy, by the end of 2021 and 2022 respectively, will free funds to be redirected for deployment in renewable energy; and



Accounting for over 10% of global greenhouse gas emissions, and around half
the world's consumption of oil, road transport is a critical sector to decarbonise
with pace. Agreement was reached by countries, cities, companies, investors
and vehicle manufacturers to target all new car and van sales to be zero
emission by 2040 globally and 2035 in leading market, and ultimately to phase
our fossil fuelled vehicles. Electrification of transport is inevitable, underway
and accelerating. Low carbon electricity supply must keep growing to provide
the energy to enable the rapid displacement of oil.

British Energy Security Strategy 2022

- 5.11.19 The British Energy Security Strategy sets out the immediate need to manage the financial implications of soaring commodity prices in the near term on households and businesses which are already feeling economic pain as the post-Covid cost of living has risen: "The first step is to improve energy efficiency, reducing the amount of energy that households and businesses need." [50, p5].
- 5.11.20 In the near-term, the strategy sets out a high-level action plan to upgrade the energy efficiency of at least 700,000 homes in the UK by 2025, and to ensure that by 2050 all UK buildings will be energy efficient with low-carbon heating. Further, the strategy sets out an intent to phase out the sale of new and replacement gas boilers by 2035. [50, p12].

5.11.21 The Strategy aims to:

- Cut planning consent process time by over half through, among other measures, strengthening the Renewable National Policy Statements (EN-3) to reflect the importance of energy security and net zero;
- Increase the pace of deployment of Offshore Wind by 25% to deliver up to 50GW by 2030, including up to 5GW of innovative floating wind. Wind will contribute over half the UK's renewable generation capacity by 2030. [50, p16];
- Consider all options including Onshore Wind through the improvement of national electricity network infrastructure and support of a number of new English projects with strong local backing, so prioritising "putting local communities in control" of local onshore solutions. Repowering of existing onshore wind sites is also under consideration. [50, p18];
- Support a 5-fold increase in deployment of solar technology by 2035, recognising the abundant source of solar energy in the UK and an 85% reduction in cost over the last ten years of solar power. For ground-mounted solar, the strategy indicates a future consultation on planning rules to strengthen policy in favour of development on non-protected land, while ensuring communities continue to have a say and environmental protections remain in place. [50, p19];
- Increase UK plans for deployment of civil nuclear to up to 24GW by 2050 three times more than operational capacity in 2022 and representing up to



25% of our projected electricity demand. This includes the intention to take one project (Sizewell C) to FID during the current Parliament, and two projects to FID in the next Parliament, including Small Modular Reactors, subject to value for money and relevant approvals. [50, p21]. The selection process for further UK projects is anticipated to be initiated in 2023 [50, p22]; and

• Double the UK ambition for hydrogen production to up to 10GW by 2030, with at least half of this from electrolytic hydrogen [50, p22], facilitated by bringing forwards up to 1GW of electrolytic hydrogen into construction or operational status by 2025.

5.12 Summary of the Main Planning Policy Requirements

- 5.12.1 Following the above review, the main policy requirements which the SoS must be satisfied have been met in consideration of the Scheme can be summarised as follows:
 - Contribution towards climate change adaptation and meeting the renewable energy need as set out in NPS EN-1 (November 2023) and other legislation including the Climate Change Act 2008as well as Government energy policy including the Energy White Paper: Powering our Net Zero Future (2020), National Infrastructure Strategy (2020) and British Energy Security Strategy 2022.
 - Biodiversity impacts as set out as prescribed matters in PA 2008, NPS EN-1 (2011), NPS EN-1 (November 2023), NPS EN-3 (2011) and NPS EN-3 (November 2023). To include Biodiversity net gain as set out in NPS EN-3 (November 2023).
 - Cultural heritage impacts as set out as prescribed matters in PA 2008, NPS EN-1 (2011), NPS EN-1 (November 2023), NPS EN-3 (2011) and NPS EN-3 (November 2023).
 - Landscape and visual impacts as set out in NPS EN-3 (2011) and NPS EN-3 (November 2023);
 - Residential amenity impacts as set out in NPS EN-3 (2011) and NPS EN-3 (November 2023);
 - Glint and Glare impacts as set out in NPS EN-3 (2011) and NPS EN-3 (November 2023);
 - Design, layout and grid connection as set out in NPS EN-1 (2011), NPS EN-1 (November 2023) and NPS EN-3 (November 2023).
 - Noise and vibration impacts from construction and traffic as set out in NPS EN-3 (2011) and NPS EN-3 (November 2023);
 - Transport impacts as set out in NPS EN-3 (2011) and NPS EN-3 (November 2023);
 - Flood Risk impacts as set out in NPS EN-3 (2011) and NPS EN-3 (November 2023);





- Consideration of alternatives in so far as this is relevant, as set out in NPS EN-1 (2011) and NPS EN-1 (November 2023).
- Impacts on best and most versatile agricultural land as set out in NPS EN-1 (2011), NPS EN-1 (November 2023) and NPS EN-3 (November 2023).
- Consideration of any other matters which the SoS thinks are both important and relevant to their decision (Section 105(2)(c) of the PA 2008). For the purposes of this application, this is considered to include socio-economic and human health impacts, major accidents and disasters, waste management and ground conditions.



6 Planning Appraisal

6.1 Introduction

- 6.1.1 This section presents an appraisal of the Scheme's compliance with the main policy requirements that are applicable to it. These requirements emerged from a review of policy documents set out in Section 5 of this Planning Statement and are listed at paragraph 5.12. In addition, Appendix C, National Policy Statement Accordance Table and Appendix D, Local Policy Accordance Table, set out an analysis of the Scheme's compliance with national and local policies, respectively. The issues covered in this section are as follows:
 - Meeting the renewable energy need (section 6.2)
 - Alternative sites and site selection (section 6.3)
 - Good design (section 6.4)
 - Landscape and visual impact (section 6.5)
 - Heritage (section 6.6)
 - Agricultural land (section 6.7)
 - Mineral safeguarding (section 6.8)
 - Biodiversity (section 6.9)
 - Water and drainage (section 6.10)
 - Noise (section 6.11)
 - Glint and Glare (section 6.12)
 - Transport and access (section 6.13)
 - Waste (section 6.14)
 - Socio-economics tourism and recreation (section 6.15)
 - Human Health (section 6.16)
 - Major accidents and disasters (section 6.17)
 - Air Quality (section 6.18)
 - Ground Conditions (section 6.19)
- 6.1.2 Section 6.2 to 6.19 take account of potential effects from the construction, operation and decommissioning phases of the Scheme. They also take account of the fact that the Scheme will be decommissioned at the end of its operational life.

Overarching Local Planning Policy Requirements

6.1.3 There are a number of overarching planning policies that are relevant to the development of the Scheme. CLLP Policy S14 supports proposals for renewable



technology where the benefits outweigh the harm caused and it is demonstrated that any harm will be mitigated as far as is reasonably possible.

- 6.1.4 CLLP Policy S14 expressly sets out that there is a presumption in favour of ground based photovoltaics "including large scale proposals" unless there is clear and demonstrable significant harm arising. It additionally requires the following matters to be taken into consideration. Where these matters are addressed is also set out:
 - **Flood Risk**; Water and drainage (Section 6.10)
 - **Testing compliance with the minerals and waste policies**; Mineral safeguarding (section 6.8)
 - The land is allocated for another purpose in this Local Plan and the proposed use is not compatible; Alternative sites and site selection (section 6.3) (As a result of the Site Selection Process allocations have been avoided).
 - **Opportunities for delivering biodiversity net gain**: Ecology and Biodiversity (section 6.9 paragraphs 6.9.3 6.9.10)
- 6.1.5 As set out above, the Planning Statement clearly demonstrates that the considerations set out within CLLP Policy S14 have been addressed.
- 6.1.6 BCSDMP Policy DM10 is supportive of proposals that seek to utilise renewable and low carbon energy. Similarly, proposals must demonstrate that they comply with a number of criteria. These are set out below together with details of where it is demonstrated within this Planning Statement that the Scheme complies with these considerations:
 - Compatible with policies to safeguard the built environment and natural environment, including heritage assets and their setting, landscape character and features of recognised importance for biodiversity; Good design (section 6.4), Ecology and Biodiversity (section 6.9), Heritage (section 6.6), Landscape and visual impact (section 6.5).
 - Will not lead to the loss or damage to high-grade agricultural land (Grades 1 &2); Agricultural land (section 6.7)
 - **Are compatible with tourism and recreational facilities**; Socio-economics, tourism and recreation (section 6.15 paragraphs 6.15.16-6.15.18)
 - Will not result in unacceptable impacts in terms of visual appearance; noise; shadow-flicker; watercourse engineering and hydrological impacts; pollution; or traffic generation; Landscape and visual impact (section 6.5), Noise (section 6.11) Glint and Glare (section 6.12), Water and drainage (section 6.10), Waste (section 6.14), Ground Conditions (section 6.19) Transport and access (section 6.13).
- 6.1.7 In addition, BDCSDMP Policy DM10 states that proposals "should not result in an unacceptable cumulative impact in relation to the factors above." Cumulative impacts



- of the Scheme have been considered within the ES and have been addressed within the planning statement where relevant to the above.
- 6.1.8 Policy BDCSDMP Policy DM10 also requires that "large-scale renewable and low carbon energy proposals must provide full details of arrangements for decommissioning and reinstatement of the site if/when it ceases to operate".
- 6.1.9 CLLP policy S14 states that "Permitted proposals will be subject to a condition that will require the submission of an End of Life Removal Scheme within one year of the facility becoming non-operational".
- 6.1.10 The following sections 6.2 to 6.18 of this Planning Statement set out how decommissioning has been considered in relation to the various topic areas covered, where relevant. The application is accompanied by an Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] which sets the framework for a detailed decommissioning strategy to be prepared to ensure that the site will be responsibly decommissioned in a safe and environmentally appropriate manner. Paragraph 6.7.1 of Agricultural land (section 6.7) explains how soil quality will be protected in order to ensure that the above policy requirement is met.

6.2 Meeting the Renewable Energy Need

- 6.2.1 Section 3.4 of NPS EN-1 (2011) sets out that the large-scale deployment of renewable electricity generation is required in order meet the UK's carbon emissions targets and tackle climate change. At paragraph 3.4.5 it states:
 - "Paragraph 3.4.1... sets out the UK commitments to sourcing 15% of energy from renewable sources by 2020. To hit this target, and to largely decarbonise the power sector by 2030, it is necessary to bring forward new renewable electricity generating projects as soon as possible. The need for new renewable electricity generation projects is therefore urgent."
- 6.2.2 Parts 2 and 3 of both NPS EN-1 (2011) and NPS EN-1 (November 2023) discuss the need for energy NSIPs. These sections explain the context and drivers for identified energy infrastructure need. The November 2023 NPSs present a more up-to-date position than the 2011 NPSs, but both set out the same principles, which mainly comprise:
 - 1. The need to secure adequate energy supply to accommodate projected increased national energy use;
 - 2. The need to replace the electricity generation capacity that will be decommissioned;
 - 3. The need to reduce greenhouse gas emissions to meet decarbonisation commitments by 2050;
 - 4. The need for more electricity capacity and resilience; and



- 5. The need to diversify energy supply and reduce reliance on imports of fossil fuels.
- 6.2.3 Whilst solar is not specifically identified in NPS EN-3 (2011), as at the time of publication it was not proven at scale, NPS EN-3 (2011) does affirm the importance, set out in NPS EN-1 (2011), of the development of large-scale renewable energy infrastructure. At paragraph 1.1.1 it states:

"Electricity generation from renewable sources of energy is an important element in the Government's development of a low-carbon economy. There are ambitious renewable energy targets in place and a significant increase in generation from large-scale renewable energy infrastructure is necessary to meet the 15% renewable energy target."

- Paragraphs 3.3.5 and 3.3.15 of NPS EN-1 (2011) put a time frame of "the next 10 to 15 years" for the provision of new low carbon developments. Given the publication date of NPS EN-1 in 2011 this would require delivery by 2026.
- 6.2.5 Paragraph 3.2.3 of NPS EN-1 (2011) states that "the need for such infrastructure will often be urgent" and therefore "substantial weight" should be given to considerations of need. The weight attributed to the need for new energy capacity should be proportionate to the proposed extent of actual contribution to satisfying the need for a particular type of infrastructure.
- It is noted that policy and legislation has moved on since the energy NPSs were published. One of the aims of the recently published National Infrastructure Strategy 2020 is to achieve net zero carbon emissions by 2050 by dramatically increasing the share of generation from renewables. This is to be achieved by the provision of greater generation capacity from onshore wind and solar. Further, the Energy White Paper: Powering our net zero future, published in December 2020, identifies that "a low-cost, net zero consistent system is likely to be composed predominantly of wind and solar" and that the increase in electricity demand through decarbonisation of other sectors means "a four-fold increase in clean electricity generation with the decarbonisation of electricity increasingly underpinning the delivery of our net zero target" is needed. The urgency of renewable energy infrastructure to address the drivers set out in NPS EN-1 (2011) has therefore accelerated.
- 6.2.7 NPS EN-1 (November 2023) confirms and gives further weight to the position that is summarised in the above paragraphs, setting out the Government's up-to-date objectives and commitments for the energy system, and providing planning policy for NSIPs that is intended to facilitate the delivery of these objectives and meeting the Government's commitments.
- 6.2.8 Paragraph 2.3.3 of NPS EN-1 (November 2023) sets out the Government's three objectives of the energy system. These are to:
 - 1. Ensure security and reliability of energy supply;
 - 2. Provide affordable energy to consumers; and,



- 3. Cut greenhouse gas emissions, delivering carbon budgets and achieving net zero by 2050.
- 6.2.9 The same paragraph sets out that "This will require a step change in the decarbonisation of our energy system", and paragraphs 2.3.4 to 2.3.5 of NPS EN-1 (November 2023) go on to set out that a significant amount of energy infrastructure, including large-scale projects, will need to be delivered and the volume and proportion of energy supplied from low carbon sources will need to be "dramatically" increased. Paragraph 2.3.6 of the NPS EN-1 (November 2023) encapsulates the challenges facing the energy system:

"We need to transform the energy system, tackling emissions while continuing to ensure secure and reliable supply, and affordable bills for households and businesses."

6.2.10 Paragraph 3.3.20 of NPS EN-1 (November 2023) sets out that, along with wind, the Government expects solar to form the majority of generation capacity in a net zero, secure and cost-efficient energy system:

"Wind and solar are the lowest cost ways of generating electricity, helping reduce costs and providing a clean and secure source of electricity supply (as they are not reliant on fuel for generation). Our analysis shows that a secure, reliable, affordable, net zero consistent system in 2050 is likely to be composed predominantly of wind and solar".

- 6.2.11 Whilst NPS EN-1 (November 2023) paragraph 3.3.12 acknowledges the role that smaller scale developments play in helping to achieve the Government's objectives and commitments for the energy system, it explains that this alone will not be enough and that "the government does not believe they will replace the need for new large-scale electricity infrastructure to meet our energy objectives". The paragraph goes on to set out that large-scale centralised electricity generating facilities have numerous economic and other benefits, including the more efficient bulk transfer of power, which enables surplus generation capacity in one area to be used to cover shortfalls elsewhere.
- 6.2.12 Paragraphs 3.1.3 and 3.1.4 of the NPS EN-1 (2011) also state that all applications for nationally significant energy infrastructure should be assessed on the basis that the need for such infrastructure has been demonstrated and that substantial weight should be given to the contribution that proposals would make towards meeting the identified energy infrastructure need. Paragraphs 3.2.6 to 3.2.8 of NPS EN-1 (November 2023) reiterate this:

"The Secretary of State should assess all applications for development consent for the types of infrastructure covered by the NPS on the basis that the government has demonstrated that there is a need for those types of infrastructure which is urgent, as described for each of them in this Part"

"In addition, the Secretary of State has determined that substantial weight should be given to this need when considering applications for development consent under the Planning Act 2008".



"The Secretary of State is not required to consider separately the specific contribution of any individual project to satisfy the need established in this NPS".

- 6.2.13 In summary, NPS EN-1 (2011) and NPS EN-1 (November 2023) set out that the delivery of a large amount of renewable generation capacity is required for delivery of the Government's energy objectives and legally binding net zero commitments and that substantial weight should be given to the contribution that proposals would make towards meeting the identified energy infrastructure need.
- 6.2.14 Section 4 of the Planning Statement and the Statement of Need [APP-320] explain how the Scheme will meet the urgent national need for secure and affordable low carbon energy infrastructure. Section 12 of the Statement of Need [APP-320] explains that the Scheme is capable of delivering large amounts of low-carbon electricity to national networks and along with other solar schemes, is of critical importance on the path to net zero. It will also enable all consumers to benefit from the effect of low-marginal cost solar generation on reducing market prices. Furthermore, it explains that maximising the capacity of generation in the proposed area, is to the benefit of all GB consumers, and the solar industry generally.
- 6.2.15 The Scheme will also deliver significant amounts of low carbon power in a timescale that is short in the context of the delivery of other forms of energy generation infrastructure as solar farms are relatively quick to construct. The Scheme is expected to take 24 months to construct whereas offshore wind projects take at least 36 months on average. Hinckley Point C nuclear project commenced in 2017 with a commercial operation date currently estimated as 2026 (see Table 5.2 of Statement of Need [APP-320]. In addition, the impacts of a constructed solar farm are reversible, with the removal of solar arrays and associated infrastructure after decommissioning being relatively simple and straightforward compared with other energy infrastructure including low carbon schemes such as off shore wind.
- 6.2.16 To support the strong emphasis of Government policy on the delivery of a large amount of renewable generation capacity to meet the Government's energy objectives and commitments, NPS EN-1 (2011) paragraph 4.1.2 states that the level of urgency is such that the starting point for deciding a DCO application for an energy NSIP must be a presumption in favour of granting consent:
 - "The Energy White Paper emphasises the importance of the Government's net zero commitment and efforts to fight climate change. Given the level and urgency of need for infrastructure of the types covered by the energy NPSs set out in Part 3 of this NPS, the SoS will start with a presumption in favour of granting consent to applications for energy NSIPs."
- 6.2.17 The presumption in favour of granting consent is carried through to NPS EN-1 (November 2023) which sets out at paragraph 4.1.3 that given the level of need for energy infrastructure, if the development proposal is in accordance with the November 2023 NPS and any relevant technology specific NPS, then the IPC should operate on the basis that consent should be given, except to the extent that any of the exceptions set out in the Planning Act apply. These exceptions are set out at



paragraph 1.1.4 of NPS EN-1 (November 2023) and include if the development would:

- lead to the UK being in breach of its international obligations;
- be in breach of any statutory duty that applies to the IPC;
- be unlawful;
- result in adverse impacts from the development outweighing the benefits; or
- be contrary to regulations about how its decisions are to be taken.
- 6.2.18 Paragraphs 4.1.3 and 4.1.4 of NPS EN-1 (2011) and paragraph 4.1.5 and 4.1.6 of NPS EN-1 (November 2023) set out that potential adverse impacts, including any long-term and cumulative adverse impacts, as well as any measures to avoid, reduce, mitigate or compensate for any adverse impacts must be taken into account in considering the proposed development. These must be weighed against its potential benefits including its contribution to meeting the need for energy infrastructure, job creation, ecological enhancements, and any long-term or wider benefits.
- 6.2.19 This Planning Statement demonstrates in the following parts of Section 6 that the Scheme has taken into consideration the potential adverse impacts of the Scheme and where there are adverse impacts, the significant public benefits of the Scheme outweigh these. The Scheme is therefore in accordance with the relevant NPSs and none of the caveats within paragraph 1.1.4 of NPS EN-1 (November 2023) are relevant in the case of the Scheme. The presumption in favour of granting consent is therefore in place.
- 6.2.20 From this urgent starting point of a presumption in favour of granting consent for energy NSIPs, NPS EN-1 (2011) paragraph 3.2.3 and NPS EN-1 (November 2023) paragraph 3.1.2, go on to acknowledge that: "...it will not be possible to develop the necessary amounts of such infrastructure without some significant residual adverse impacts. These effects will be minimised by the application of policy set out in Parts 4 and 5 of this NPS." This statement is in recognition of the fact that it is rarely possible to deliver NSIPs without some significant effects due to their scale. Paragraph 4.1.5 of NPS EN-1 (November 2023) recognises that significant effects from renewable technologies can potentially affect the environment. Of relevance to the Scheme are potential effects on biodiversity, landscape and noise which have been assessed within the relevant chapters of the Environmental Statement [APP-038 to APP-044, APP-046 to APP-060, REP1-012, REP1-073, REP1-074, REP3-010]. In addition, its recognition that a few positive specific effects associated with the technologies may occur, including on biodiversity from solar farms, where land is no longer managed intensively. The biodiversity net gain report [APP-088] sets out the significant gains anticipated to result from the Scheme.
- 6.2.21 Other policies in relation to the delivery of renewable energy such as paragraph 163 of the NPPF, expect the determination of planning applications to "not require



applicants to demonstrate the overall need for renewable or low carbon energy" and "approve the application if its impacts are (or can be made) acceptable". This statement does not state that there should be no significant environmental effects, but that those effects should be 'acceptable'. The NPPF requirement for impacts to be 'acceptable' should be considered in the context of an NSIP scale of project whereby significant environmental effects are likely to be unavoidable.

- The Scheme will deliver significant carbon savings, compared to other types of electricity generation, and is expected to have a major beneficial significant effect on the climate. ES Chapter 7, Climate change [REP1-012] states at paragraph 7.7.61 that the Scheme is expected to have a total energy generation figure of around 31,425,614 MWh over the estimated 60-year assessed lifetime. Table 7.26 provides a comparison of energy intensities of various forms of energy generation compared to the West Burton Solar Project. Based on the total energy generation of the Scheme and the worst-case assumption for total lifespan project GHG emissions, the intensity of the Scheme is estimated to be 7.66gCO2e/kWh. This compares favourably with fossil fuel electricity generation. Each kilowatt hour of electricity generated by the Scheme will emit at least 370gCO2e less than if it was generated by a gas-fired CCGT generating facility (See Table 7.26 Comparison of energy intensities of various forms of energy generation in ES Chapter 7, Climate change [REP1-012]).
- 6.2.23 It is also comparable with other low carbon energy generation. It is considered that the only other viable electricity generation that could be delivered on the land would be for onshore wind which would have a comparable GHG intensity in the range of 7 20gCO2e/kWh.
- Paragraph 7.8.62 of ES Chapter 7, Climate change [REP1-012] explains that a further calculation has been done to understand at what point the GHG reductions from National Grid through the use of renewable energy at the Scheme would offset the calculated worst-case emissions generated from the products (e.g., solar panels) and the construction phase of the Scheme. It also accounts for annual emissions generated by the Scheme from water use, replacement products and energy consumption on site. This shows that it is expected that the savings from the Scheme would result in offsetting the construction emissions within 3 years of operation. Over the 60-year lifespan, there would be a reduction of 3,981,049 up to 5,688,549 tCO2e from the Scheme compared to a scenario where the development does not go ahead. This will make a significant contribution towards cutting greenhouse gas emissions, delivering carbon budgets and achieving net zero by 2050 in line with the objectives set out at Paragraph 2.3.3 of NPS EN-1 (November 2023).

Summary

- 6.2.25 In summary, the Scheme would:
 - **Deliver a large amount of renewable generation** capacity (31,425,614 MWh over the estimated 60-year assessed lifetime) to deliver the



Government's energy objectives and legally binding net zero commitments in line with the requirements of paragraph 1.1.1 of NPS EN-3 (2011), paragraph 3.3.20 of NPS EN-1 (November 2023), section 3.4 of NPS EN-1 (2011) and the National Infrastructure Strategy 2020;

- **Deliver a reduction of** 3,981,049 up to 5,688,549 tCO2e over the lifetime of the Scheme compared to if it did not go ahead which would make a significant contribution towards reducing carbon emissions as required by paragraph 1.1.1 of NPS EN-1 (2011), paragraph 2.3.3 of NPS EN-1 (November 2023), the National Infrastructure Strategy 2020 and the Energy White Paper: Powering our net zero future;
- **Deliver in a timescale that is short** in the context of the delivery of other forms of energy generation in line with the urgent need to decarbonise expressed in paragraphs 3.3.5, 3.3.15 and 3.4.5 of NPS EN-1 (2011), Paragraph 2.3.3 of NPS EN-1 (November 2023) and the National Infrastructure Strategy 2020;
- **Enable all consumers to benefit** from the effect of low-marginal cost solar generation on reducing market prices, in line with the aim to provide affordable energy for consumers set out at Paragraph 2.3.2, Paragraph 2.3.6 and 3.3.20 of NPS EN-1 (November 2023);
- **Help ensure security and reliability of energy supply** in line with Paragraph 2.3.3 and 2.3.6 of the NPS EN-1 (November 2023).

6.3 Alternative Sites and Site Selection

- 6.3.1 The Applicant selected the land within the Order Limits because it is suitable for the Scheme. Its location and characteristics mean that it is suited to the generation of a large amount of solar electricity and the export of that electricity to the NETS, whilst avoiding impacts on nationally or internationally designated sites and minimising impacts on other sensitive receptors. ES Appendix 5.1: Site Selection Assessment [AS-004] sets out the site selection process in detail. Chapter 8: landscape and Visual Impact [APP-046], Chapter 9: Ecology and Biodiversity [APP-047] and Chapter 13: Cultural Heritage [APP-051] assess the impacts on nationally or internationally designated sites and impacts on other sensitive receptors.
- 6.3.2 Section 4.4 of NPS EN-1 (2011) and paragraphs 4.3.15 to 4.3.17 of NPS EN-1 (November 2023) set out the circumstances where NPS planning policy requires the consideration of alternatives. At paragraph 4.4.1 NPS EN-1(2011) states:
 - "From a policy perspective this NPS does not contain any general requirement to consider alternatives or to establish whether the proposed project represents the best option."
- 6.3.3 Paragraph 4.3.9 of the NPS EN-1 (November 2023) states:
 - "This NPS does not contain any general requirements to consider alternatives or to establish whether the proposed project represents the best option from a policy perspective".



- 6.3.4 Paragraphs 4.4.2 of NPS EN-1 (2011) and 4.3.15 of NPS EN-1 (November 2023) set out the circumstances where the NPS 2011/ NPS (November 2023) imposes a policy requirement to consider alternatives. Paragraph 4.4.2 of NPS EN-1 (2011) states: "applicants are obliged to include in their ES, as a matter of fact, information about the main alternatives they have studied. This should include an indication of the main reasons for the applicant's choice, taking into account the environmental, social and economic effects and including, where relevant, technical and commercial feasibility."
- 6.3.5 ES Chapter 5: Alternatives and Design Evolution [APP-043] sets out the main alternatives considered, which include alternative sites, alternative technologies, alternative site layouts, and alternative cable routing.
- 6.3.6 Paragraphs 4.4.2 of NPS EN-1 (2011) and 4.3.15 and 4.3.16 of NPS EN-1 (November 2023) also set out some circumstances where there are specific legislative requirements to consider alternatives. These are in relation to the issues listed below:
 - 1. Where a scheme would lead to significant harm to biodiversity and geological conservation interests that cannot be avoided (NPS EN-1 2011 section 5.3 and NPS EN-1 (November 2023) section 5.4).
 - 2. Where a scheme would be located within, or partially within, Flood Zone 2 or Flood Zone 3 (NPS EN-1 2011 section 5.7 and NPS EN-1 November 2023 section 5.8). In this case the Sequential Test should be passed for development within Flood Zone 2 and the Sequential and Exception Tests should be passed for development within Flood Zone 3. The purpose of the Sequential test is to guide development to areas at lowest risk of flooding, by requiring applicants to demonstrate that there are no alternative lower risk sites available where the development could take place (see Sequential Test within Flood Risk Assessment and Drainage Strategy Report [APP-089]. With regard to applying the Sequential Test, paragraph 5.7.13 of NPS EN-1 (2011) and paragraph 5.8.23 of NPS EN-1 (November 2023) set out that consideration of alternative sites should take account of the approach to alternatives described in section 4.4 of NPS EN-1 (2011) and section 4.2 of NPS EN-1 (November 2023).
 - 3. Where a development would be located within either a National Park, the Broads or an AONB (NPS EN-1 2011 section 5.9 and NPS EN-1 November 2023 section 5.10).
- 6.3.7 There are no relevant adopted or emerging local plan policies that require the consideration of alternative sites other than those which seek a sequential approach to the location of development within Flood Zones. These are, Policy ST52 of the DBLP and Policy 1 paragraph 3 of the Treswell and Cottam Neighbourhood Plan.
- 6.3.8 The Order limits are not located within a National Park, the Broads or an AONB and ES Chapter 9: Ecology and Biodiversity [APP-047] concludes that there are anticipated to be no residual adverse effects upon designated ecological sites. However significant residual adverse effects are anticipated on harvest mice,



skylark, grey partridge, over wintering birds within the Sites and on hedgerows, trees, ditches and watercourses within the Cable Route Corridor at a site and local level respectively, whilst significant beneficial residual effects are anticipated for other species ranging from a site to district level (See section 6.9). Therefore, assessment of alternatives is required to address point '1' above.

- 6.3.9 This has been undertaken by the Site Selection Assessment (**[AS-004]** which is summarised below and ES Chapter 5: Alternatives and Design Evolution **[APP-043]** demonstrate that there are no suitable alternative sites for the Scheme. The design and layout of the Scheme has evolved to minimise ecological impacts as far as possible. ES Chapter 5 sets out the design changes that have taken place in response to minimising ecological impacts at Tables 5.6 5.9.
- 6.3.10 In respect of point 2 above, whilst the majority of the Order Limits are located within Flood Zone 1 (as directed by paragraphs 5.7.13 and 5.8.15 of EN-1 2011 and paragraphs 5.8.13 and 5.8.21 to 5.8.23 of EN-1 November 2023) 28.95% of the Sites are located within Flood Zone 3. These include small parts of West Burton 1 along the north-west and eastern boundaries, part of West Burton 2 along the eastern boundary with the River Till and part of the central section and western edge of West Burton 3. The majority of the Cable Route Corridor is in Flood Zone 1. The southern extent of the cable within the vicinity of the river Trent and the central extent in the vicinity of the River Till is situated within Flood Zones 2 and 3. Overall, the conclusions of the flood risk assessments (FRA and Drainage Strategy, ES Appendix 10.1 [APP-089]) are that the Scheme is at low risk of fluvial flooding. Within Flood Zone 3 areas, the proposed solar panels will be raised above surrounding ground levels with associated power infrastructure appropriately waterproofed and inherent mitigation measures included. The Flood Risk Assessment Sequential and Exception Test, ES Appendix 10.6 [APP-094] demonstrates how the Scheme satisfies the requirements and purpose of the Sequential Test.
- 6.3.11 In addition, consideration of alternative brownfield sites, or alternative sites that comprise agricultural land that is not classed as best and most versatile, also forms part of the justification that is required by national and local planning policy for the inclusion of some best and most versatile agricultural land within Order limits as set out at paragraphs 5.10.8 and 5.10.15 of NPS EN-1 (2011), 2.10.28 to 2.10.31 of NPS EN-3 (November 2023), CLLP Policy S67, BDCSDMP Policy DM10 and emerging DBLP ST51. This is discussed in Section 6.7 of this Planning Statement.
- 6.3.12 In considering inclusion of some areas of best and most versatile agricultural land within the Order limits, paragraph 4.4.3 of NPS EN-1 (2011), and paragraph 4.2.21 of NPS EN-1(November 2023) sets out the principles that should guide the Secretary of State (SoS) when considering the weight that should be given to alternatives. These include (among others) the principles described below.
 - 1. The consideration of alternatives in order to comply with policy requirements should be carried out in a proportionate manner.



- 2. Only alternatives that can meet the objectives of the proposed development need be considered,
- 3. The Secretary of State should be guided by whether there is a realistic prospect of the alternative delivering the same infrastructure capacity (including energy security, climate change, and other environmental benefits) in the same timescale as the proposed development.
- 4. Alternative proposals which mean the necessary development could not proceed, for example because the alternative proposals are not commercially viable or alternative proposals for sites would not be physically suitable, can be excluded on the grounds that they are not important and relevant to the IPC's decision.
- 6.3.13 Alternative proposals which are vague or inchoate are not important and relevant to the Secretary of State's decision.
- 6.3.14 Practically, points '2' and '3' mean that smaller scale solar farms should not be considered as reasonable alternatives to the Scheme, since they would not meet the objective of the Scheme to supply the maximum amount of renewable electricity to the NETS, and they would not deliver the same energy, climate change or environmental benefits as the Scheme.
- 6.3.15 In addition, paragraph 4.3.24 of NPS EN-1 (November 2023) sets out that:

 "The Secretary of State should not refuse an application for development on one site simply because fewer adverse impacts would result from developing similar infrastructure on another suitable site."
- 6.3.16 In considering alternatives and identifying and selecting the Site, the Applicant has been guided by principles described above and also by the technical and environmental requirements of a large-scale solar farm development project.

Site Selection Process

- 6.3.17 The following paragraphs describe the reasons why the Applicant identified and selected the Site following a process to identify land which is suitable from a technical, environmental and planning perspective. The sections below refer to the matters set out in Section 2.10 of NPS EN-3 (November 2023), "Solar Photovoltaic Generation: factors influencing site selection and design" and relevant sections of NPS EN-1 (November 2023).
- The selection of the Scheme's location has followed a systematic five-stage process. This process and confirmation of its suitability when considered against potential alternative sites is set out in detail in ES Appendix 5.1: Site Selection Assessment [AS-004]. The assessment is high level and primarily desk based. This approach is considered reasonable and proportionate and complies with the NPS EN-1 (2011) requirement set out at paragraph 4.4.3 that "the consideration of alternatives in order to comply with policy requirements should be carried out in a proportionate manner."
- 6.3.19 In summary, the stages undertaken were:



- Stage 1 Identification of the Area of Search. This was focussed on identification of a viable grid connection at West Burton Power Station and National Grid's preference at that stage (between August and early September 2020) for a connection here rather than High Marnham because fewer upgrade works were required resulting in quicker and less costly delivery of the connection. The search area was enlarged incrementally to a 15 km radius around the point of connection which is considered by the Applicant to be a viable cable connection distance for a solar project of this scale.
- Stage 2 Exclusion of Planning, Environmental and Spatial Constraints. These constraints included designated international and national ecological and geological sites, nationally designated landscapes, proximity to sensitive human receptors and all Grade 1, 2 and 3 agricultural land according to publicly available data from the Natural England Agricultural Land Classification (ALC). As the Natural England maps do not differentiate between grades 3a and 3b all land in Grades 1, 2 and 3 was excluded and the focus was on trying to identify suitable sites within areas of Grade 4, 5 or unclassified land outside of other identified planning and environmental constraints.
- Stage 3 Identifying Potential Solar Development Areas. This stage applied key operational criteria for large scale solar development, such as site size and land assembly and site topography to further refine the unconstrained areas identified at Stage 2. The use of previously developed (brownfield) land, commercial roof tops and alternative locations proposed through the statutory consultation stage were also considered and discounted for the reasons set out at paragraphs 2.1.23-2.1.31 of the Site Selection Assessment [AS-004].
- Stage 4 Evaluation of Potential Solar Development Areas (PDAs). One Potential Solar Development Area (PDA) identified in Stage 3 was evaluated against planning, environmental and other operational assessment indicators derived from national and local planning and environmental policy objectives and the operational requirements of the Scheme (see Annexes B and C of the Site Selection Assessment [AS-004]. Ultimately, following the evaluation stage, this PDA on Grade 4/5 agricultural land/unclassified land proved unsuitable for development due to significant constraints being identified. (See Annex E of Appendix 5.1: Site Selection Assessment [AS-004].
- Stage 5 Widening the Search to consider Grade 3 agricultural land. After discounting of the PDA on Grade 4/5 agricultural land/unclassified land, the site search focused on the areas of Grade 3 agricultural land within the search area. Other NSIP projects located on Grade 3 land within the Search Area were discounted from further assessment because they are not available to accommodate the Scheme. Land agents used their professional knowledge to provide details of potentially willing landowners with large-scale landholdings within the area. These were assessed against the same detailed range of



planning, environmental and operational considerations used to assess the Stage 4 PDA.

- 6.3.20 Annex E: Table 1 and Table 2 of Appendix 5.1: Site Selection Assessment **[AS-004]** show the results of the assessment.
- 6.3.21 This resulted in the choice of the West Burton original draft site area which was later reduced and refined into the Scheme. This location performed better than 3 of the other locations and equal to one (Site 4) within the RAG assessment. Site 4 is immediately adjacent to High Marnham Power Station where a grid connection was not preferred by National Grid at the time of Site Selection, but which would be the most sensible and cost effective POC for Site 4 in the future. In addition, a detailed ALC assessment has not been undertaken for Site 4 so it may contain a higher proportion of BMV land than the Scheme.
- 6.3.22 Appendix 5.1: Site Selection Assessment **[AS-004]** concludes that there are no obviously more suitable locations within the area of search than the proposed Sites for the Scheme. The Scheme's location is therefore assessed to be suitable for the scale of solar development proposed and the basis on which the Applicant has selected the Sites accords with the approach to the consideration of alternatives set out by paragraph 4.4.3 of NPS EN-1 (2011) (see 6.3.9 above).

The Selected Site

6.3.23 The land for the Scheme is considered suitable and is selected for a large-scale solar site for the reasons set out below:

Irradiance and Topography

6.3.24 The land is located within Lincolnshire, an optimal region within the UK to locate a large-scale solar farm. This is due to good irradiation levels and suitable topography, which is predominantly made up of and characterised by large flat open land. This is consistent with the factors influencing site selection for solar generation NSIPs that are set out in section 2.10 of NPS EN-3 (November 2023). In particular, paragraph 2.10.19 sets out that solar irradiance and topography are key considerations for identifying a potentially suitable site, since these directly affect the amount of electricity that can be generated on a site. The Site is suitable for a solar farm development in this regard, being located within an area of high irradiance and topography of less than 3% gradient as identified within the Site Selection Assessment [AS-004].

Grid Connection

6.3.25 Paragraph 2.10.25 of NPS EN-3 (November 2023) sets out:

"To maximise existing grid infrastructure, minimise disruption to existing local community infrastructure or biodiversity and reduce overall costs applicants may choose a site based on nearby available grid export capacity".

6.3.26 The decommissioning of large coal fired power stations within the region has led to the availability of significant grid capacity at available and accessible connection



- points. There is available capacity for the Scheme to connect to the NETS at West Burton Power Station that can be completed within a reasonable timeframe and cost (See **Section 8.4** and Chapter 9 of Statement of Need **[APP-320]** for more detail).
- 6.3.27 Paragraph 4.10.2 of NPS EN-1 (2011) states that: "The applicant will liaise with National Grid who own and manage the transmission network in England and Wales or the relevant regional DNO or TSO to secure a grid connection."
- 6.3.28 The applicant has secured a grid connection offer from National Grid for 480MW.

<u>Accessibility</u>

In identifying the Site, the Applicant took account of the requirement for it to be accessible for the purposes of its construction and operation. Paragraph 2.10.36 of NPS EN-3 (November 2023) states that "Given that potential solar farm sites are largely in rural areas, access for the delivery of solar arrays and associated infrastructure during construction can be a significant consideration for solar farm siting." The Scheme has good transport access for construction and operational maintenance, with good links to the strategic road network (the A15, A46, M180) via the A1500, and A156. See Section 14.7 of ES Chapter 14: Transport and Traffic [APP-052] and ES Addendum Chapter 14: Transport and Access [REP1-074] for detail of construction traffic impacts.

Capacity of the site

- 6.3.30 Paragraph 2.10.61 of NPS EN-3 (November 2023) sets out that "the type, spacing and aspect of panel arrays will depend on the physical characteristics of the site such as site elevation". The land is of a suitable size and has excellent topographical characteristics which meet the requirements of the Scheme to generate approximately 480MW of electricity and accommodate associated battery energy storage system. The Scheme would make a substantial contribution to the supply of the low carbon energy that is required in order for the Government's objectives and commitments for the energy system to minimised as explained in Chapter 12 of Statement of Need [APP-320].
- 6.3.31 Paragraphs 5.9.9 and 5.10.7 of NPS EN-1 (2011) and NPS EN-1 (November 2023), respectively, set out that National Parks and AONBs have the highest status of protection in relation to landscape and scenic beauty, and paragraphs 5.9.10 (NPS EN-1 2011) and 5.10.32 (NPS EN-1 November 2023) set out that the granting of development consent within a National Park or AONB would require exceptional circumstances to be demonstrated.
- 6.3.32 The Scheme is not located within a National Park or AONB and the above principles set out in NPS EN-1 (2011) do not apply. In addition, by avoiding locally designated landscapes, the Scheme ensures that it does not have any direct impact on landscapes that have been formally identified as of being of particular local value. Although not located within any designated landscape, ES Chapter 8: Landscape and Visual Impact, [APP-046] assesses the likely significant effects of the Scheme on the landscape.



Biodiversity and Geology Designations

- 6.3.33 The Solar Farm Site is not located within any nationally, internationally or locally designated biodiversity or geological sites.
- 6.3.34 A number of locally designated sites have been identified within 5km of the Sites (See Section 6.9 paragraph 6.9.21 for further detail). These sites will be protected by the Outline Ecological Protection and Mitigation Strategy (Outline EPMS) [APP-326] during the construction phase and enhanced in the long term wherever possible through the provisions of the Outline Landscape and Ecological Management Plan (Outline LEMP) [EN010132/EX6/WB7.3 E]. REP6-025].
- 6.3.35 Similarly, protected sites such as Sites of Special Scientific Interest which were noted within 5km of the Sites for their wetland habitats will be protected from potential pollution events or disturbance during construction through the measures set out in the Outline EPMS.
- 6.3.36 By avoiding and protecting designated biodiversity and geology sites as part of the Applicant's site selection and design, the Scheme is consistent with paragraphs 5.3.7 and 5.3.8 of NPS EN-1 (2011) and paragraphs 5.4.42 and 5.4.48 of NPS EN-1 (November 2023). These set out that DCO decisions should give appropriate weight to designated biodiversity and geology sites of international, national and local importance, and that significant harm to biodiversity conservation interests should be avoided. The Scheme also complies with local planning CLLP policies S14, S59 and S60, BLP Policy DM9 and DBLP policies S60 and S61, by avoiding impacts on internationally, nationally and locally designated nature conservation sites.

Flood Zones

- 6.3.37 The Site is predominantly within Environment Agency Flood Zone 1 and overall, the conclusions of the flood risk assessments are that the Scheme is at low risk of fluvial flooding (see Section 6.10 for further detail). Within Flood Zone 3 areas, the proposed solar panels will be raised above surrounding ground levels with associated power infrastructure appropriately waterproofed and inherent mitigation measures included.
- 6.3.38 The selection of the Site largely in Flood Zone 1 is therefore consistent with the objective of NPS EN-1 (2011) paragraph 5.7.3 to "...direct development away from areas at highest risk" and the NPS EN-1 (November 2023) paragraph 5.8.6 objective to "steer new development to areas with the lowest risk of flooding". The areas of the sites that are located within flood zones 2 and 3 are located at the periphery of the Sites or cross parts of fields that cannot be excluded from the Scheme without excluding whole fields, which would result in isolated and unviable parcels of land from a farming perspective. These areas are therefore retained within the Scheme and the mitigation measures set out above will ensure that panels and electrical infrastructure can be adequately waterproofed to withstand the effect of flooding.
- 6.3.39 The panels within these areas will contribute to the Scheme's significant public benefit through the delivery of renewable energy and as the solar panels will be



mounted on raised frames above surrounding ground level it will allow water to flow freely underneath and there will be no loss of floodplain volume and no increase in the risk of flooding elsewhere as a result of the proposed development. Section 6.0 of Appendix 10.1 Flood Risk and Drainage Strategy [APP-089] demonstrates that the Scheme satisfies the requirements and purpose of the Sequential Test and Exceptions Test. This is discussed in further detail within section 6.10 of the Planning Statement. The inclusion of small areas of Flood Zone 2 and 3 within the proposed development is therefore justified.

Heritage Designations

6.3.40 There are no listed buildings, scheduled monuments, Historic Parks and Gardens or Conservation Areas within the Site. Broxholme medieval settlement and cultivation remains (List Entry Number: 1016797) is located adjacent to West Burton 1 and the Scheduled Monument Medieval Bishop's Palace and Deer Park, Stow Park (List Entry Number: 1019229) is located immediately adjacent to the West Burton 3 Site. There are a number of listed buildings within 2km of the Sites. The sites were therefore chosen to avoid direct physical impact on designated heritage assets. Detailed assessment of the Scheme's impact upon designated and non-designated heritage assets is set out at Section 6.6.

Land Use Planning Allocations and Designations

- 6.3.41 There are no land use planning allocations or designations within the Site aside from mineral safeguarding. This will not be affected as the Scheme will be decommissioned at the end of its operational life and it will be possible to revert to its current land use, which would not prohibit mineral extraction in the future.
- 6.3.42 The Site is therefore compliant with paragraph 5.10.22 of NPS EN-1 (2011) and paragraph 5.11.19 of NPS EN-1 (November 2023), which states that "Applicants should safeguard any mineral resources on the proposed site as far as possible, taking into account the long-term potential of the land use after any future decommissioning has taken place".
- 6.3.43 The impact of this (as well as the Grid Connection Route) is considered by ES Chapter 12: Minerals **[APP-050]** and discussed at Section 6.8 of this Planning Statement.
- OBLP Policy ST51 requires energy proposals to demonstrate regard to Bassetlaw Council's Energy Opportunities Diagram and Renewable and Low Carbon Energy Study (or subsequent replacement) when identifying options for achieving CO2 emission reductions. The policy identifies an Area of Best Fit for Renewable Energy Development on a site at the former High Marnham power station site for development that generates, shares, transmits and/or stores zero carbon and/or low carbon renewable energy. ES Appendix 5.1: Site Selection Assessment [AS-004] explains the requirements for the Scheme in terms of land area, which far exceeds the land available within the Area of Best Fit for Renewable Energy Development, meaning this would not be suitable on its own. It also explains how other adjacent land around High Marnham Power Station was considered and ultimately



discounted because National Grid advised at that time (between August and early September 2020) that although there was capacity available at High Marnham, their preference was for a connection at the West Burton POC because fewer upgrade works to National Grid's transmissions assets would be required at the West Burton POC and it would therefore be more straightforward, quicker to deliver and more economical.

- Outside the Area of Best Fit, such developments are not precluded elsewhere within the district but will be expected to demonstrate an operational and/or economic need for the development in that location. This has been demonstrated within Meeting the renewable energy need (section 6.2) and further detail is provided within the Statement of Need [APP-320].
- 6.3.46 At Cottam Power Station, DBLP Policy ST6 identifies a Priority Regeneration Area as a broad location for mixed use regeneration rather than renewable energy generation. Neither the location of the Sites or the Cable Route Corridor are within, or adjacent the Priority Regeneration Area and will not prejudice the comprehensive redevelopment of this site as identified by the masterplan framework.
- 6.3.47 The compliance of the Scheme with the aforementioned local policy requirements and criteria is considered by the relevant parts of Section 6 of this Planning Statement.
- 6.3.48 By avoiding conflicts with Development Plan allocations and their purposes (see Annex D and E of the Site Selection Assessment [AS-004] for details), the Site and Scheme accord with the principles of NPS EN-1 (2011) paragraph 5.10.13 and NPS EN-1 (November 2023) paragraph 4.1.13, which require the Secretary of State to take account of any such conflicts in their decision.

Agricultural land classification and land type

- 6.3.49 BMV agricultural land is classified as being within grade 1, grade 2 or grade 3a. Paragraphs 2.10.28 to 2.10.31 of NPS EN-3 (November 2023) set out that applicants for solar NSIPs should take account of Agricultural Land Classification (ALC). They state that applicants should seek to locate their development on previously developed land, brownfield land, contaminated land, industrial land or lower grade agricultural land (classified as grade 3b, 4 or 5), where possible. Paragraph 5.10.8 of NPS EN-1 (2011) sets out that applicants should preferably use land in areas of poorer quality, except where this would be inconsistent with other sustainability considerations. Local planning policies CLLP S67, BDCSDMP DM10 and DBLP ST51 seek to protect the best and most versatile agricultural land.
- 6.3.50 Appendix 5.1: Site Selection Assessment **[AS-004]** explains that the Applicant undertook a sequential approach to the consideration of potential sites which first considered and discounted Grade 4 and 5 agricultural land and unclassified land before considering Grade 3 agricultural land. The Scheme maximises the utilisation of low grade, non-best and most versatile (BMV) agricultural land with 73.76% of the Sites being classified as non BMV land (See ES Appendix 19.1 **[APP-137]** for details).



6.3.51 The Applicant's application is therefore consistent with the terms of NPS EN-3 (November 2023) paragraph 2.10.31 which explains that solar farm developments are not prohibited on 'best and most versatile' agricultural land and that "It recognized that at this scale, it is likely that applicants' developments will use some agricultural land". It does go on to explain that "Applicants should explain their choice of site, noting the preference for development to be on suitable brownfield, industrial and low and medium grade agricultural land". The Applicant has complied with this policy with the explanation of the choice of site set out within Appendix 5.1: Site Selection Assessment [AS-004] and within ES Chapter 5, Alternatives and Design Evolution, [APP-043].

Proximity to dwellings

- 6.3.52 In identifying the Sites, the Applicant identified that it is remote from nearby villages and that the relatively flat landform and existing woodland and hedgerow limits views into the Site. As such, no significant residual adverse effects are anticipated upon residential receptors as explained at paragraphs 6.5.21 6.5.22. NPS EN-3 (November 2023) paragraph 2.10.27 states that "utility-scale solar farms are large sites that may have a significant zone of visual influence. The two main impact issues that determine distances to sensitive receptors are therefore likely to be visual amenity and glint and glare".
- 6.3.53 Local Plan policies CCLP S14, S33, S53, BDCSDMP DM4 and DBLP Policy 48 and ST51 seek to protect residential and visual amenity. In addition, Saxilby with Ingleby Neighbourhood Plan Policy 7, Sturton Ward Neighbourhood Plan Policy 2a, Sturton by Stow and Stow Neighbourhood Plan Policy 7 and Policy 11 and Treswell and Cottam Neighbourhood Plan Policy 1 also seek to protect residential and visual amenity. The Applicant has taken account of the visual impact on residential receptors in the design of the Scheme, including by providing stand-offs from residential curtilages (50m) to above ground solar farm infrastructure to limit visual impact and impacts of glint and glare on residential receptors as detailed within the Design and Access Statement [APP-314 and APP-315] and set out within the Concept Design Parameters [REP5-094].
- 6.3.54 Further detail on impacts on amenity is set out at section 6.4 (paragraphs 6.4.26-6.4.31. Detail on landscape and visual impact and glint and glare are set out at sections 6.7 and 6.12 of this Planning Statement and the impacts are assessed within ES Chapter 8: Landscape and Visual Impact [APP-046] and Chapter 16: Glint and Glare [APP-054].

Land Availability

6.3.55 NPS EN-3 (November 2023) paragraph 2.10.28 notes that solar is a highly flexible technology and as such can be deployed on a wide variety of land types. However, in order to deliver the substantial benefits of a large-scale solar farm, sufficient land must be available from a willing landowner or owners. Identification of a site in a limited number of landownerships can assist in the delivery of a scheme in accordance with national and local policies. The Scheme is within four land



ownerships, and this small number of landowners is advantageous as it minimises legal complexity and cost. It also provides enhanced ability to develop and deliver joined up mitigation and enhancements across the Scheme, including a coherent biodiversity scheme across the Site and permissive paths. It also provides the ability to direct development to the least agriculturally productive parts of the landholdings, and it minimises the need for compulsory acquisition.

<u>Summary</u>

- 6.3.56 Chapter 5: Alternatives and Design Evolution [APP-043] sets out the main alternatives to the Scheme which have been considered, which include alternative sites, alternative technologies, alternative site layouts, and alternative cable routing. No suitable alternatives have been identified. The Scheme therefore accords with the requirements of paragraph 4.4.2 of NPS EN-1 (2011).
- 6.3.57 Paragraphs 4.4.2 of NPS EN-1 (2011) and 4.3.15 and 5.2.18 of NPS EN-1 (November 2023) have required the consideration of alternatives due to a significant residual adverse effect being anticipated on harvest mice, skylark and grey partridge at a site and local level respectively. The Site Selection Assessment [AS-004] and ES Chapter 5: Alternatives and Design Evolution [APP-043] demonstrate that there are no suitable alternative sites for the Scheme.
- As parts of the Sites fall within Flood Zones 2 and 3, in accordance with paragraphs 4.4.2 of NPS EN-1 (2011) and 4.2.12 the sequential test and exceptions test have been applied and passed as demonstrated in Appendix 10.1 Flood Risk and Drainage Strategy [APP-089].
- 6.3.59 Consideration has also been given to the use of brownfield sites and alternative sites that comprise agricultural land that is not classed as best and most versatile as set out in the Site Selection Assessment [AS-004] and ES Chapter 5: Alternatives and Design Evolution [APP-043]. No better alternative sites on brownfield land or on lower grade agricultural land than the Scheme were identified. The Scheme therefore accords with paragraphs 5.10.8 and 5.10.15 of NPS EN-1 (2011), 3.10.13, 3.10.14, 3.10.15 and 2.10.31 of NPS EN-3 (November 2023) and with CLLP Policy S67, BDCSDMP Policy DM10 and emerging DBLP ST51.

6.4 Good Design

- This has taken account of the context and features of the land within the Order limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts, and provision of environmental and other enhancements, where practicable.
- 6.4.2 Section 4.5 of NPS EN-1 (2011) sets out the principles for good design that should be applied to all energy infrastructure. It states at paragraph 4.5.1 that good design



should "produce sustainable infrastructure sensitive to place, efficient in the use of natural resources and energy used in their construction and operation, matched by an appearance that demonstrates good aesthetic as far as possible".

- 6.4.3 Paragraph 4.5.1 does, however, acknowledge that "the nature of much energy infrastructure development will often limit the extent to which it can contribute to the enhancement of the quality of the area". NPS EN-3 (2011) expects renewable energy NSIPs to demonstrate "good design in respect of landscape and visual amenity, and in the design of the project to mitigate impacts such as noise and effects on ecology" (paragraph 2.4.2) and NPS EN-5 (2011) also identifies that proposals for electricity networks infrastructure should demonstrate good design in their approach to mitigating potential adverse impacts (paragraph 2.5.1).
- 6.4.4 NPS EN-1 (November 2023) sets out at section 4.7 that applicants should consider how 'good design' can be applied at the early stages of a project. It also recommends that applicants embed opportunities for nature inclusive design into their scheme and emphasises that wider impacts such as landscape and environmental impacts will be important factors in the design process.
- 6.4.5 NPS EN-3 (November 2023) paragraphs 2.10.98 to 2.10.101 set out that developers should consider the criteria for good design set out in section 4.7 of NPS EN-1 (2011), particularly in terms of layout, future maintenance and retention of boundary vegetation. It also sets out that solar farms should be designed sensitively to minimise environmental effects, including on landscape (paragraphs 2.10.94 to 2.10.95) and heritage assets (paragraphs 2.10.112 to 2.10.113).
- 6.4.6 In terms of local planning policy, the following policies, which are reproduced in full at Appendix D, set out requirements for good design;
 - CCLP Policy S14 (including main mods) and BCSDMP Policy DM4 require that impacts are acceptable on the amenity of sensitive neighbouring uses (including local residents);
 - CCLP Policy S53 states that development must achieve high quality sustainable
 design that contributes positively to local character, landscape and townscape,
 and supports diversity, equality and access for all, protect views, contribute to
 sense of place, incorporate and retain as far as possible existing natural
 features and minimise the need for resources both in construction and
 operation; and,
 - DBLP Policy ST35 requires all development to be of a high-quality design including positively preserving, enhancing and integrating landscape and townscape features, and natural and heritage assets and mitigating flood risk and water run-off.
- 6.4.7 The above policies are applicable to locally and regionally significant developments and primarily address developments that create buildings and streets. Therefore, not all policy criteria can easily be applied to the Scheme, but the objectives of these policies which are considered to be relevant include:



- 1. High quality of design (see below);
- 2. Make effective and efficient use of land (see section 6.2);
- 3. Respect the local context and complement the landform, layout, building orientation, scale, height, massing, type, materials, details and landscaping of the surrounding areas (see below and section 6.5);
- 4. Not result in the visual or physical coalescence with any neighbouring settlement (see section 6.5);
- 5. Positively preserve, enhance and integrate landscape and townscape features, and natural and heritage assets (see section 6.5);
- 6. Incorporate and retain as far as possible existing natural and historic features such as hedgerows, trees, ponds, boundary walls, field patterns, buildings or structures (See below and section 6.9);
- 7. Protect any important local views into, out of or through the site (see section 6.5);
- 8. Incorporate and/or link [the Scheme] to a well-defined infrastructure network of well managed and maintained public and open spaces (see section 6.5 green infrastructure);
- incorporate high quality landscape design and maximise opportunities for greening, particularly where a development site adjoins the countryside (see below);
- 10. Sustainable design and construction, and utilise modern construction methods and durable materials, where practicable (see below);
- 11. Minimise energy consumption by maximising opportunities for passive solar energy and integrating renewable and low carbon technologies where practicable (see below);
- 12. mitigate flood risk and water run-off (see section 6.10);
- 13. create well connected places that prioritise the needs of pedestrians and cyclists (see below);
- 14. protect residential amenity (see below); and,
- 15. provide opportunities to promote healthy living and wellbeing (see section 6.16).
- In accordance with NPS EN-1 (2011) section 4.6, the Scheme is the result of an iterative design development process which commenced at an early stage, and the design and layout addresses the key opportunities and challenges of the Sites and the context and setting within which they are located. The design team has worked collaboratively to provide an integrated and responsive design which has been informed by stakeholder engagement. Through the design process, the Applicant has taken account of the context and features of the land within the Order limits



- and its surroundings to develop a good design that meets the requirements and objectives of the policies described above.
- 6.4.9 The design choices that will achieve these objectives and deliver good design are described below. The design evolution and basis of design decisions taken are described in ES Chapter 5: Alternatives and Design Evolution [APP-043] and the Design and Access Statement [APP-314 and APP-315]. These inform the following paragraphs.

The scheme makes efficient use of energy and natural resources

- 6.4.10 As set out in the Design and Access Statement, Objective 1 of the Scheme is to efficiently generate a large amount of affordable renewable energy to support policy objectives and national targets for reducing carbon emissions to net zero by 2050.
- 6.4.11 To help achieve this, each of the Sites have been designed to have a generating capacity of over 50MW, with the Scheme having a total generating capacity of up to 480MW of renewable solar energy for up to 60 years for distribution by the National Grid. This will make a significant contribution towards meeting national energy demand, replacing approximately 24% of the former generation capacity of the coal powered West Burton A Power Station.
- 6.4.12 Whilst it is currently envisaged that the Scheme will utilise tracker solar panels, the DCO Application seeks consent for the Applicant to be able to utilise either tracker or fixed panels in order to be able to utilise the most up to date and efficient technology available at the time of construction. Since solar generation technology is progressing at a fast pace, the Scheme retains the ability to choose the precise technology close to the point of construction of the Scheme within the parameters defined by the DCO. This will enable the optimum production of renewable energy. Tracker panels have a maximum height of 4.5 metres, whereas fixed panels are up to 3.5 metres. The tracker panels have been assessed within the ES as the worst-case scenario.
- 6.4.13 The panels would generate a large amount of energy and would offer good potential for biodiversity enhancements below and between the solar arrays.
- 6.4.14 The design also seeks to minimise shading of PV Arrays, which can affect their generation output. The Scheme seeks to minimise generation loss due to shading by including stand-offs between arrays and trees (which create shade).
- 6.4.15 The design of the Scheme includes Battery Energy Storage (BESS). Energy can be stored from production on site, or from surplus energy on the Grid, so that it can be released back onto the Grid at times of peak demand. This will help to support policy objectives for delivery of renewable energy by reducing demand for non-renewable energy at peak times, and by providing grid balancing services to help increase the resilience of the electricity distribution network.
- 6.4.16 As set out in the Outline Construction Environmental Management Plan (CEMP) [EN010132/EX6/WB7.1_DREP6-021], the construction phase of the Scheme has



committed to adopting Considerate Constructors' Scheme (CCS) measures to assist in reducing greenhouse gases. It also commits to designing, constructing, and implementing the Scheme in such a way as to minimise the creation of waste and maximise the use of alternative materials with lower embodied carbon such as locally sourced products and materials with a higher recycled content.

- 6.4.17 The Outline Construction Traffic Management Plan (CTMP), [REP4-038EN010132/EX7/WB6.3.14.2 E] also commits the construction of the Scheme to encouraging the use of lower carbon modes of transport for staff accessing the Order limits. The Scheme is sensitive to its place, location and landscape character.
- 6.4.18 The Design and Access Statement [APP-314 and APP-315] explains that a key objective (Objective 5) is for the Scheme to be sensitive to the surrounding landscape, limiting the impact on views for key landscape receptors, residential properties, and recreational routes.
- Topography has influenced the choice of Sites as explained within the Site Selection Assessment [AS-004] helping to ensure the Scheme will be sensitively sited in the landscape. The design of the Scheme has further achieved this by responding carefully to the landscape character when considering the layout of the Scheme. The layout has been designed to avoid impacts on valuable landscape features through the incorporation of minimum offsets from ancient woodland, woodland, hedgerows, PRoW and watercourses. Existing field boundaries will also be retained and enhanced, which will help preserve these features for their own sake and will also preserve the existing pattern and scale of the landscape. The planting design shown by the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025] has been sensitively designed in this respect and responds to this varied character by allowing views to remain open, where tall screening would not be appropriate.
- 6.4.20 Siting of key infrastructure such as substations and battery storage has been carefully considered to ensure that these structures do not occupy prominent positions in the landscape as explained within the Design and Access Statement [APP-314 and APP-315]. This helps to ensure that policies in respect of landscape and visual amenity set out at paragraph 4.9 above are satisfied.
- 6.4.21 In order to minimise the impact on the landscape and avoid the introduction of new tall, linear features in the landscape, the main Cable Route will comprise below ground cables. Proposed fencing has also been designed to minimise its visual prominence. This has been achieved by avoiding heavy duty materials where possible, instead using wooden posts and wire.
- 6.4.22 The above measures demonstrate that the Scheme has been designed to make efficient use of energy and natural resources as required by Section 4.5 of NPS EN-1 (2011) and local plan policies CLLP Policy S53.

The Scheme mitigates effects on ecology and enhances biodiversity by providing a nature inclusive design.



- Enhancement of local biodiversity is a key objective of the Scheme as outlined within the Design and Access Statement [APP-314 and APP-315] under Objective 3. The choice of Sites for the Scheme sought to avoid statutorily designated ecological sites as explained in the Site Selection Assessment [AS-004]. The design of the Scheme incorporates the measures listed below, which will protect species and habitats within and near to the Order limits and help to deliver an anticipated biodiversity net gain of 86.80% for habitats (delivered through the creation of other neutral grasslands within the sites), a net gain of 54.71% for hedgerows, and a net gain of 33.25% for river units as detailed within the Biodiversity Net Gain Assessment [APP-088] (more details of the habitats to be created and / or enhanced and their management are provided within the Outline LEMP and will be confirmed at detailed design stage [EN010132/EX6/WB7.3_EREP6-025]:
 - All ancient woodland, mature/veteran trees, roadside verges, and ponds will be retained, with protection buffers around these habitats. This accords with NPS EN-3 (November 2023) paragraphs 2.10.98 to 2.10.101 and CLLP Policy S53, DBLP ST53, Sturton Ward NP Policy 2a and Sturton by Stow NP Policy 11.
 - Proposed new hedgerows with trees will provide additional linking habitat and reinforce the existing green network. A total length of 7.1km of new hedgerow is proposed within the Site. This accords with the requirement in section 4.7 of the NPS EN-1 (November 2023) to embed opportunities for nature inclusive design and CLLP Policy S53, DBLP ST53, Sturton Ward NP Policy 2b, Sturton by Stow NP Policy 11.
 - Planting of copses and shelterbelts to provide 'stepping stones' between larger areas of woodland. These have been included at all sites, with extensive shelterbelts at West Burton 3. A total area of 13.7ha of woodland is proposed. This accords with the requirements of section 4.7 of the NPS EN-1 (November 2023) to embed opportunities for nature inclusive design into the Scheme, as well as the requirements of CLLP Policy S53, DBLP ST53, Sturton Ward NP Policy 2b and Sturton by Stow NP Policy 11.
 - Bands of scattered trees with lower canopy shrub planting are proposed throughout the Site. This planting typology has been specified along water courses and to provide additional vegetative layering within the landscape. An area of 11ha of scrub habitat will be established across the Site, with wide strips at West Burton 2 and West Burton 3. This accords with the requirements of section 4.7 of the NPS EN-1 (November 2023) to embed opportunities for nature inclusive design into the Scheme, as well as the requirements of, CLLP Policy S53, DBLP ST53 and Sturton by Stow NP Policy 11.
 - Buffer areas have been incorporated to ensure an appropriately sized offset from development between the various valued habitats typically located at field boundaries (hedgerows, watercourses and woodland etc.). Buffer zones are located between the retained field boundary habitats and the perimeter security fence in the case of the 'outermost' fields within a Site, and between



field boundary habitats and the panels in other fields. ES Chapter 9: Ecology and Biodiversity [APP-047] and Appendix 9.11: Schedule of Protective Ecological Buffers [APP-087] contain detail on the layout of these buffers. This accords with NPS EN-1 (2011) Paragraph 4.5.1 through helping to mitigate effects on ecology.

- Flower rich pollinator strips to provide a floristically rich habitat will be created for pollinating insects. This would also benefit species such as farmland birds, amphibians and reptiles. Areas have also been created adjacent to residential properties on West Burton 2 and 3, as well as alongside the Public Right of Way at West Burton 3. A total area of 46.5ha of herb rich pollinator mix will be provided. This accords with the requirements of section 4.7 of the NPS EN-1 (November 2023) to embed opportunities for nature inclusive design into the Scheme, as well as the requirements of SCLLP Policy LP26, CLLP Policy S53, DBLP ST53 and Sturton by Stow NP Policy 11.
- Provision of tussock grassland margins for a range of birds, providing a food source both during breeding and wintering, as well as nesting habitat for species such as corn bunting, reed bunting, yellowhammer and whitethroats. The Landscape Plans [REP1-026 to REP1-031] show this habitat being created extensively across the Site forming an important connected corridor for wildlife with a total area of 53.14ha. This accords with the requirements section 4.7 of the NPS EN-1 (November 2023) to embed opportunities for nature inclusive design into the Scheme, as well as the requirements of CLLP Policy S53, DBLP ST53 and Sturton by Stow NP Policy 11.
- Diverse meadow creation beneath solar panels. It has been shown that diverse grassland can be created within a solar array, where managed appropriately. This can have a significant benefit to biodiversity but can also benefit surrounding agricultural land through offering an increase in pollinator species. The total area of this habitat creation measures 66.61ha. This accords with the requirements of section 4.7 of the NPS EN-1 (November 2023) to embed opportunities for nature inclusive design into the Scheme, as well as the requirements of CLLP Policy S53, DBLP ST53 and Sturton by Stow NP Policy 11.
- The large remainder of the Sites' panelled areas will be converted to a diverse grassland over a longer time span. This approach has been chosen after consultation with a seed supplier, as it is likely that there would not be enough seed available in the UK to plant the entire Site with an appropriate mix immediately. The total area proposed for this longer-term meadow creation is 416.1ha. This accords with the requirements of section 4.7 of the NPS EN-1 (November 2023) to embed opportunities for nature inclusive design into the Scheme, as well as the requirements of CLLP Policy S53, DBLP ST53 and Sturton by Stow NP Policy 11.



- Approximately 97ha at West Burton 2 will be managed as mitigation for ground nesting birds such as skylark, yellow wagtail and lapwing an also as an enhancement for these species and other red/amber listed birds such as curlew and meadow pipit. This will include spring sown cereal crops with skylark plots within and permanent grassland with shallow wetland scrapes next to the River Till. This accords with the requirements of section 4.7 of the NPS EN-1 (November 2023) to embed opportunities for nature inclusive design into the Scheme and accords with NPS EN-1 (2011) Paragraph 4.5.1 through helping to mitigate effects on ecology. It also accords with CLLP Policy S53, DBLP ST53 and Sturton by Stow NP Policy 11.
- An area of approximately 0.8ha in the south west of West Burton 2 will be planted with a native grass and wildflower mix with new native hedging, a coppice area and new wetland habitat. The initial design of this habitat management area has taken into account the objectives of the Saxibly Nature Project in light of its proximity to nearby sites that form part of the Saxibly Nature Project. This accords with the requirements of section 4.7 of the NPS EN-1 (November 2023) to embed opportunities for nature inclusive design into the Scheme, as well as the requirements of with CLLP Policy S53, DBLP ST53 and Sturton by Stow NP Policy 11.
- Adjacent to rivers and ditches, a tall herb community will be established through seeding. This marginal habitat is important for species such as water vole as well as aquatic invertebrates. A total area of 9ha of this habitat will be created. This accords with the requirements of section 4.7 the NPS EN-1 (November 2023) to embed opportunities for nature inclusive design into the Scheme, as well as the requirements of CLLP Policy S53, DBLP ST53 and Sturton by Stow NP Policy 11.
- Ponds will be created within field margin buffer zones outside the footprint of the array. A total of 5 new ponds are proposed; two ponds at West Burton 1 and three ponds at West Burton 3. This accords with the requirements of section 4.7 of the NPS EN-1 (November 2023) to embed opportunities for nature inclusive design into the Scheme, as well as the requirements of and CLLP Policy S53, DBLP ST53 and Sturton by Stow NP Policy 11.
- Bird box installation based on a one box for every 100m of hedgerow (based on an estimate of 52km of hedgerow). This gives a total of 520 boxes which have been split between various target species depending on what has been recorded within the surveys. This accords with the requirements of NPS EN-1 (November 2023) section 4.7 requirement to embed opportunities for nature inclusive design.
- Bat boxes installation based on one box for every 200m stretch of hedgerow.
 This accords with the requirements of section 4.7 of the NPS EN-1 (November 2023) to embed opportunities for nature inclusive design into the Scheme.



- Two hibernacula/log piles per pond will be created (in total 22 adjacent to 11 ponds). This accords with the requirements of section 4.6 of the NPS EN-1 (November 2023) to embed opportunities for nature inclusive design into the Scheme.
- 6.4.24 Further details of the above can be found within the Outline LEMP [EN010132/EX6/WB7.3_E].REP6-025]. The above represents a substantial enhancement to biodiversity. Through a pre-commencement requirement, the Draft DCO will necessitate the submission and approval of a detailed Landscape and Ecological Management Plan (LEMP) to deliver the provisions as set out in the Outline LEMP [EN010132/EX6/WB7.3_E].REP6-025].
- 6.4.25 The above measures demonstrate that the Scheme satisfies the requirements of NPS EN-1 (2011), NPS EN-3 (2011) and local planning policies in terms of good design in respect of ecology and biodiversity, and in the design of the Scheme to mitigate effects on ecology. The measures help to ensure that environmental effects are minimised where possible, and also demonstrate that the Scheme embeds opportunities for nature inclusive design.

The Scheme protects the amenity of residents, including visual amenity

- 6.4.26 The measures below ensure that the Scheme demonstrates good design in terms of siting relative to existing landscape character, landform and vegetation in accordance with NPS EN-1 (2011) paragraph 4.5.3 and satisfies the requirements of NPS EN-3 (2011), CCLP Policy S14, BCSDMP Policy DM4, Sturton Ward NP Policy 2a, Sturton by Stow and Stow NP Policy 7 and Policy 11, Treswell and Cottam NP Policy 1 in respect of protecting residential and visual amenity:
- 6.4.27 The Site Selection Assessment [AS-004] details how the Sites were chosen to avoid urban areas and other residential receptors as far as practicable. The design development process that followed included a great deal of effort made to minimise the impact of the Scheme on residential receptors, particularly in respect of visual amenity. A careful approach has been taken to the proposed arrangement of PV Arrays close to residential properties. This has included bespoke visits to residential properties to understand how best to implement mitigation, and includes the retention of key view corridors free from PV Arrays and other equipment from residential properties. The form and extent of offsets from residential properties has been tailored to respond to the existing character of such views to minimise the potential for adverse change. Detail of design changes made to the Scheme to help minimise impacts upon residential amenity can be found in Tables 5.6 5.9 of ES Chapter 5: Alternatives and Design Evolution [APP-043].
- 6.4.28 Existing vegetation will also be utilised and augmented, as shown by the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025] to reduce the visual impact of the Scheme on views from residential properties. Extensive new planting is also proposed to screen the Scheme where necessary. This includes an area of elephant grass (Miscanthus giganteus) to be planted to act as screening in the south west of West Burton 1. The planting proposals are used carefully in combination with retention



of key view corridors to reduce the change in the nature and distance of views from residential properties as a result of screening planting. This includes advanced mitigation planting in locations where it would be beneficial to undertake planting early, in order to maximise growth prior to the Scheme's operation.

- 6.4.29 Within the Cable Route Corridor, all development will be below ground and the impacts on views within the area will be limited only to construction and decommissioning stages. Through the adoption of an outline Construction Environmental Management Plan [EN010132/EX6/WB7.1_DREP6-021] and a Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023], the impacts of these stages have been limited wherever possible and pose only temporary impacts upon views.
- An assessment of noise and vibration impacts has been undertaken and is reported in ES Chapter 15: Noise and Vibration [APP-053]. The chapter assesses the significance of potential noise and vibration effects during the construction, operational and decommissioning phases, and concludes that, with appropriate mitigation, there would be no significant noise or vibration effects in terms of the EIA Regulations. Embedded noise mitigation measures comprising acoustic louvres around inverters are proposed in identified locations and secured through the Concept Design Parameters. In addition, Best Practicable Means (BPM) to minimise noise during the construction and decommissioning phases are included within the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023].
- 6.4.31 Furthermore, a Statutory Nuisance Statement **[APP-317]** has been prepared which has considered matters of general site condition, waste, air quality, artificial light, glint and glare, noise and vibration, and concludes that the Scheme is not envisaged to give rise to significant effects that would result in a statutory nuisance.

The Scheme protects heritage assets

The measures below demonstrate that the Scheme satisfies the requirements of NPS EN-1 (2011), NPS EN-3 (2011) and local planning policies in terms of good design in respect of protecting heritage assets, and in the design of the Scheme to mitigate effects on heritage. The proposed measures help to ensure that environmental effects are minimised. Objective 5 of the **Design and Access Statement**, [APP-314 and APP-315] sets out that the Scheme will be sensitive to heritage assets and their setting. The choice of Sites for the location of the Scheme sought to avoid designated heritage assets as far as possible, as set out within the **Site Selection Assessment** [AS-004]. This responds positively to any natural and man-made features within the landscape and townscape which positively contribute to the character of the area, such as historic buildings and monuments and ensure accordance with CLLP 53 and DBLP ST35, which require the same. It also complies with Sturton by Stow and Stow NP Policy 7 and Rampton & Woodbeck NP Policy 6 which each seek to protect heritage assets.



- Great care has been taken in the design of the layout of the Scheme in proximity to heritage assets. The Order Limits and extent of PV arrays and ancillary infrastructure has been refined in the vicinity of above and below ground heritage assets to minimise potential for direct impact on heritage assets. As evidenced in Paragraphs 4.1.1 to 4.1.2 of the **Stow Park Cultural Heritage Position Statement [REP5-027]**, the Applicant explored a range of different additional mitigation options with Historic England. However, Historic England did not agree that any of the suggested beneficial options would provide any mitigation that would reduce the level of harm caused by the Scheme. As such, these options weren't explored further or considered as part of the design of the Scheme. The resultant identified impacts upon the setting of designated heritage assets such as the medieval bishop's palace and deer park, Stow Park (NHLE 1019229) will be less than substantial, and reversible as discussed within Section 6 of the Planning Statement. These residual impacts will need to weighed in the planning balance.
- 6.4.34 The **Design and Access Statement [APP-314 and APP-315]** explains that avoidance of national cultural heritage designations and areas of significant archaeology, limits to restricted loading and non-penetrative ground foundations and consideration of the context of cultural heritage assets were priorities within the design process. This accords with paragraph 2.4.2 of NPS EN-3 (2011) and paragraphs 2.10.112 to 2.10.113 of NPS EN-3 (November 2023), Sturton by Stow and Stow NP Policy 7 and Rampton & Woodbeck NP Policy 6 which all seek to protect heritage assets.
- 6.4.35 **ES Chapter 13 [APP-051]** addresses archaeological assessment and sets out the embedded mitigation measures that have been identified and adopted as part of the evolution of the project design at 13.1.62. This accords with NPS EN-1 (November 2023) paragraph 5.9.9. The measures include the removal of panels from certain archaeologically sensitive areas and the use of concrete feet and above ground cabling ducts to avoid impacts to archaeologically sensitive areas. Where assets have been identified as requiring preservation in situ, with standard mitigation in place in the form of placing the panels on concrete feet, impacts would be avoided.
- 6.4.36 Existing woodland and hedgerows have been used wherever possible to provide screening. The **Works Plans [REP5-035]** define the extents of the Scheme permitted for the locating of PV panels, including where there are offsets from defined features. Key heritage assets have been identified and the impacts upon them assessed in the **ES Chapter 13: Cultural Heritage [APP-051]**. The extents of where new planting areas are proposed are set out in the **Outline LEMP [EN010132/EX6/WB7.3_EREP6-025]** and secured by a DCO requirement.
- 6.4.37 Areas of archaeological interest within the Order Limits have been carefully avoided through the removal of PV panel structures or overlaid by PV panel structures with non-intrusive foundations. The **Works Plans [REP5-035]** define the extents of the Scheme permitted for the locating of PV panels, including where there are offsets from defined features. Areas of archaeological interest requiring non-intrusive



foundations have been identified and mapped in ES Chapter 13: Cultural Heritage [APP-051].

The Scheme enhances connectivity

- 6.4.38 The measures below will ensure that the Scheme complies with the good design requirements in terms of access and connectivity set out at section 4.7 of NPS EN-1 (November 2023) and with local planning policies CLLP S53, and Saxilby with Ingleby NP Policy 16: Objective 7 of the Design and Access Statement [APP-314 and APP-315] sets out that the design of the Scheme seeks to ensure Public Rights of Way are safeguarded from unnecessary diversions or closures, with all efforts made to ensure they can be protected, integrated into the Scheme design, and where feasible, enhanced by planting and greater connectivity through the introduction of permissive paths. The public highway should also be protected, and as such the design of the Scheme should ensure that access to the Scheme does not negatively impact on the safety and desirability of the use of the public highway for all users.
- 6.4.39 Existing ProWs will be retained in all instances with no closures or diversions required for the duration of the Scheme. It is not anticipated that any temporary PRoW diversions will be required for the Sites. However, in the unlikely case that a temporary diversion is required for health and safety reasons, they will be limited to the minimum duration required to ensure continued connectivity. The Access and Rights of Way Plan [REP4-012] shows the PRoW and highway network within the Order limits. The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] details how construction impacts on PRoW are to be managed and is secured by a DCO requirement. See also the Outline Public Rights of Way Management Plan [REP5-018] which sets out how diversions will be managed if they are absolutely necessary.
- The safe use of PRoWs and highways will be managed through design mitigation and onsite construction traffic management including dedicated crossing point and bankspersons for highway accesses where required. The safe use of PRoWs and highways has been assessed in the ES Chapter 14: Transport and Access [APP-052] and ES Chapter 14 Addendum: Transport and Access [REP1-074]. Any required mitigation or management measures are set out in the Construction Traffic Management

 Plan

 [REP4-038REP4-038EN010132/EX7/WB6.3.14.2_E] and are secured by a DCO requirement.
- The new permissive footpath to run from the track off Sykes Lane along the Codder Lane Belt and then south and west to rejoin Sykes Lane opposite Hardwick Scrub will enhance connectivity within the local area. The Works Plans define the extents of the proposed permissive path, with details of planting set out in the Outline Landscape and Ecological Mitigation Plan [EN010132/EX6/WB7.3_EREP6-025] and secured by a DCO requirement.

Conclusion



6.4.42 The outcome of the above is that the Scheme delivers good design, meeting the requirements of the 2011 NPSs and November 2023 NPSs in the context of efficiently delivering large scale renewable energy infrastructure whilst providing a new network of environmental features which deliver a range of ecosystem services, incorporating biodiversity, heritage, landscape and access. The Scheme design also achieves the design objectives of local planning policies.

6.5 Landscape and Visual Assessment

- As detailed in ES Chapter 8: Landscape and Visual Impact [APP-046] the landscape and visual impacts of the Scheme have been assessed in accordance with NPS EN-1 (2011) paragraphs 5.9.5 to 5.9.7, NPS EN-1 (November 2023) paragraphs 5.10.16 to 5.10.22, NPS EN-3 (November 2023) paragraphs 2.10.97 to 2.10.101 and NPS EN-5 (2011) and NPS EN-5 (November 2023). The assessment includes reference to the relevant landscape character assessments and any significant effects. In making the assessment a range of factors have been considered, including visibility, views, visual amenity, light pollution, local amenity, tranquillity and nature conservation.
- 6.5.2 ES Appendix 8.5 Landscape Policy Commentary [APP-076] sets out in detail the compliance of the Scheme in terms of landscape and visual effects with relevant national and local planning policies and these are also included within Appendix C and D to this Planning Statement.
- 6.5.3 The following paragraphs set out the landscape and visual effects of the Scheme during operation, followed by construction and decommissioning and also consider cumulative effects.

Landscape and Visual effects during operation

- NPS EN-1 (2011) (paragraphs 5.9.8) and NPS EN-1 (November 2023) (paragraphs 5.10.1 to 5.10.6), acknowledge the fact that landscape effects depend on the existing character of the local landscape, its current quality, how highly it is valued and its capacity to accommodate change. All of these factors need to be considered in judging the impact of a project on landscape. They state that virtually all nationally significant energy infrastructure projects will have effects on the landscape and the aim should be to minimise harm to the landscape, providing reasonable mitigation where possible and appropriate. Local planning policies need to be considered in light of this as they have not been developed to take account of the likely level of impact of large-scale infrastructure associated with NSIPs, nor the nationally significant level of benefit arising from such projects.
- 6.5.5 NPS EN-1 (2011) paragraph 5.9.15 and NPS EN-1 (November 2023) paragraph 5.10.35 state that outside of designated landscapes, the decision maker should "...judge whether any adverse impact on the landscape would be so damaging that it is not offset by the benefits (including need) of the project." Paragraphs 5.9.16 and 5.10.36 of NPS EN-1 (2011) and NPS EN-1 (November 2023), respectively, set out that in considering the above, the decision maker should take account of whether any



adverse impact is temporary and/or is capable of being reversed in a reasonable timescale.

Landscape Assessment

- In accordance with NPS EN-1 (2011) (paragraphs 5.9.8) and NPS EN-1 (November 2023) (paragraphs 5.10.1 to 5.10.6), the existing character of the local landscape, its current quality, how highly it is valued and its capacity to accommodate change have been considered in judging the impact of the Scheme on the landscape. National Parks and AONBs are given a high status of protection by NPS EN-1 (2011) paragraph 5.9.9 and NPPF paragraph 182. There are no National Parks or AONBs within the Order limits or within the study area which has been used to assess landscape and visual effects, as explained at Section 8.5 (Individual Contributors to Landscape Character: Establishing Value) of ES Chapter 8: Landscape and Visual Impact [APP-046]. The Scheme is therefore policy compliant in terms of avoiding impacts on National Parks and AONBs.
- There are a number of local planning policies that include reference to both landscape and visual impacts. The key ones include SCLLP policy LP17 which seeks to protect and enhance the intrinsic value of the landscape and townscape, including the setting of settlements. It requires proposals to have particular regard to maintaining and responding positively to any natural and man-made features within the landscape and townscape which positively contribute to the character of the area. The design of the Scheme has taken detailed account of the landscape and landform in which it sits and has given careful consideration to its impact on views from sensitive receptors. These have been factored into the design development at all stages as explained within the Design and Access Statement [APP-314 and APP-315] and ES Chapter 5: Alternatives and Design Evolution [APP-043].
- Relevant Sturton by Stow and Stow NP policies include Policy 9: Protected Views, Sturton Ward NP policy 2a: Protecting the landscape character, significant green gaps and key views, Tresswell and Cottam NP policy 1.
- 6.5.9 There are no areas of local landscape value within the Order limits.
- 6.5.10 Although the study area for the assessment is outside and not close to nationally or locally designated landscapes, NPS EN-1 (2011) at paragraph 5.9.14 and NPS EN-1 (November 2023) at paragraphs 5.10.16 and 5.10.17 expect the consideration of local planning policies which have been based on landscape character assessment.
- 6.5.11 ES Chapter 8: Landscape and Visual Impact [APP-046] identifies the published national, regional, county and district landscape character areas that the Scheme and the applicable study area interacts with. ES Figure 8.5 [APP-157] illustrates the local level landscape character areas that have been identified. ES Chapter 8: Landscape and Visual Impact [APP-046] assesses the impact of the operational phase of the Scheme on regional landscape character areas. The assessment at section 8.7 of Chapter 8 concludes that the operation of the Scheme would result in no likely significant adverse effects at year 1 of operation on the identified Regional



Landscape Character Areas. There will be a significant beneficial effect at year 15 upon particular character areas including RLCT 4a Unwooded Vales, LLCA 3 The Till Vale and LLCA 2 Trent Valley.

- 6.5.12 The effects at a fine-grained scale have also been taken into consideration and draw upon individual contributors to landscape character. In relation to Land Use, Topography and Watercourses, Public Rights of Way and Access, Communications and Infrastructure, Scheduled Monuments, Listed Buildings, Conservation Areas and Registered Parks and Gardens and Nationally and Locally Designated Landscape, there would be no likely significant effects for the operation of the Scheme at both year 1 and year 15.
- 6.5.13 NPS EN-1 (2011) paragraph 5.9.17 and NPS EN-1 (November 2023) paragraph 5.10.37 set out that the decision maker should, "...consider whether the project has been designed carefully, taking account of environmental effects on the landscape and siting, operational and other relevant constraints, to minimise harm to the landscape, including by reasonable mitigation."
- 6.5.14 As set out in Section 6.4 of this Planning Statement and described within Section 8.6 of ES Chapter 8: Landscape and Visual Impact [APP-046], the Scheme has been the subject of an iterative design process, informed by analysis of landscape and visual constraints, iterative impact assessments and mitigation proposals. The landscape mitigation measures and residual landscape effects at year 15 are set out at Section 8.11 of ES Chapter 8: Landscape and Visual Impact [APP-046]. The mitigation strategy and design development are based on the Landscape Design Parameters set out at Table 8.21, page 140 of Chapter 8. This has helped ensure that primary landscape mitigation is co-ordinated with other relevant disciplines, such as ecology, to determine the key parameters and agree offsets to improve the value of the landscape and reflect appropriate local and regional aims and objectives for ecology and biodiversity.
- 6.5.15 The principles described below have been incorporated to ensure the landscape impacts are minimised and significant adverse effects for landscape and visual amenity to the wider area are avoided, where possible. These principles are secured by the Works Plans [REP5-035] which define where different Works are permitted to be located, and the Concept Design Parameters and Principles [REP5-094], which set out parameters and principles with which the Scheme is required to comply. These include:
 - 1. Careful siting of the built elements of the Scheme such as substations and battery storage to avoid areas of the Sites where they would be more visually prominent in the landscape and could benefit from existing screening where possible. For example, the preferred location for the siting of the substation at West Burton 1 was initially a broad area in the northwest quarter of the Site due to an absence of significant constraints. The southwest corner of field M4 M3 (refer to field numbering plans at Appendix 1 of Design and Access Statement: Figures 6.1-6.3 [APP-314 and APP-315] was initially chosen for its



easy access from Broxholme Lane. Ahead of PEIR, the location of the substation was moved by approximately 200m to the northwest corner of field M5, which sits centrally in the Site, and immediately adjacent to the originally identified area. This was as a result of preliminary landscape assessment, which identified the corner of field M5 as a preferable location due to the immediate impact on views for users of Broxholme Lane had the substation remained in field M4. This move did not cause any conflict with any other identified constraints.

- 2. Refinement of the Order Limits and the extent of built structures in order to provide stand-offs, and to retain key views from residential properties, heritage assets, roads and footpaths. For example, at West Burton 2, fields N11 and N17 were excluded from the siting of solar panels as agreed with Saxilby and Ingleby Parish Council, due to landscape impact on Ingleby Road and the preservation of views between Ingleby mediaeval village and the church in Saxilby. At West Burton 3, the locating of panels in field Q1 was truncated at the public right of way to avoid enclosing the footpath, and provide additional offsetting from residences in Marton.
- 3. Conserving existing landscape features and vegetation such as woodland, trees and hedgerows by excluding them from, and providing offsets to, any structure to be installed or constructed as part of the Scheme. For example, at West Burton 2, offsets to hedgerows and trees were refined following completion of ecological surveys, which included the surveying and offsetting from identified badger setts.
- 4. Creating new green infrastructure within the Order limits through the implementation of new woodland, hedgerows and native grassland to improve the landscape structure, screening of the proposed development, and creating a new permissive route to provide linkages within and across the Site for the life of the Scheme. This is shown by the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025].
- 5. Sensitive design in relation to form, colour and materials. This includes ensuring that the Cable Route will be underground, thereby avoiding the introduction of new tall linear features in the landscape which would increase the extent of the Scheme's visibility. Proposed perimeter fencing has also been carefully selected to minimise its visual prominence and would comprise deer wire mesh and wooden post fencing with a maximum height of 2.5m.
- 6. Sensitive design of lighting to avoid and minimise the potential for adverse landscape and visual effects. Sensitive lighting principles employed by the Scheme are summarised in ES Chapter 4: Scheme Description [APP-042]. The principles set out being downward pointing, inward facing and the minimum required for safe operations.
- 6.5.16 The approach outlined above is in direct accordance with NPS EN-1 (2011) paragraph 5.9.22 and NPS EN-1 (November 2023) paragraph 5.10.27, which state:



"Adverse landscape and visual effects may be minimised through appropriate siting of infrastructure within its development site and wider setting. The careful consideration of colours and materials will support the delivery of a well-designed scheme, as well as sympathetic landscaping and management of its immediate surroundings."

6.5.17 The landscape effects during operation of the Scheme comply with NPS EN-1 (2011) paragraph 5.9.15 and NPS EN-1 (November 2023) paragraph 5.10.35, as ES Chapter 8: Landscape and Visual Impact [APP-046] concludes that the design of the Scheme has been successful in ensuring that there are no identified significant adverse effects on the landscape at year 1 of operation of the Scheme. The benefits (including need) of the Scheme, therefore outweigh the less than significant impacts. Paragraphs 5.9.16 of NPS EN-1 (2011) and 5.10.36 of NPS EN-1 (November 2023) set out that in making a decision, the decision maker should take account of whether any adverse impact on the landscape is capable of being reversed in a reasonable timescale. The ES concludes that the minor, less than significant adverse landscape impacts of the Scheme will be largely reversed on decommissioning. The Scheme is also considered to comply with Sturton Ward NP policy 2 and Tresswell and Cottam NP policy 1 as it has been shown not to result in significant harm during the operational phase and the less than significant harm has been minimised and mitigated.

Visual Assessment

- 6.5.18 In terms of visual effects, NPS EN-1 (2011) paragraph 5.9.18 and NPS EN-1 (November 2023) paragraph 5.10.13 state that "All proposed energy infrastructure is likely to have visual effects for many receptors around proposed sites." They go on to state that the decision maker "will have to judge whether the visual effects on sensitive receptors, such as local residents, and other receptors, such as visitors to the local area, outweigh the benefits of the project."
- 6.5.19 Although introducing new energy generation infrastructure into the landscape will inevitably have some visual effects, in accordance with NPS EN-1 (2011) paragraph 5.9.17 and NPS EN-1 (November 2023) paragraph 5.10.37, the Scheme has been carefully designed to minimise visual effects as far as possible. ES Chapter 8: Landscape and Visual Impact [APP-046] and Appendix 8.3 Assessment of Potential Visual Effects [APP-074] presents an assessment of the impact of the Scheme on sensitive visual receptors. The cumulative visual effects of Cottam Solar project, Gate Burton Energy Park and Tillbridge Solar have also been assessed.
- 6.5.20 Visual effects on viewpoints, ProW, transport routes and residential receptors have all been reduced through a range of mitigation measures summarised within Tables 8.68 8.71 of ES Chapter 8: Landscape and Visual Impact [APP-046].
- 6.5.21 These tailored mitigation measures result in likely significant adverse residual effects for the operation (Year 15) stage of the Scheme to 6 viewpoint receptors (which include 4 viewpoints along Sturton Road, one along Broxholme Lane and one on Sykes Lane), 2 transport receptors (Sturton Road and track off Sykes Lane) and 2



ProW receptors (PR007 and PR038). There are no significant effects on residential receptors.

- 6.5.22 The Scheme would not impede upon any of the identified protected views as shown on Policy Maps 9.1 and 9.2 of Saxilby with Ingleby NP Policy 9: Protected Views (See Appendix 8.5 [APP-076].
- 6.5.23 The limiting of significant adverse residual visual effects at year 15 of operation to the above 10 receptors is the direct result of the careful and detailed iterative design process. Through this process, the Scheme has been carefully sited in the landscape and refined through design development to respond to the existing character of views.
- As recognised by NPS EN-1 (2011) paragraph 5.9.18 and NPS EN-1 (November 2023) paragraphs 5.10.13 to 5.10.15 state that all proposed energy infrastructure is likely to have visual effects for many receptors around proposed sites. In this case, the effects are limited to 10 receptors and are not considered to outweigh the benefits of the project, set out at Section 4 of the Planning Statement. The Scheme is therefore considered to comply with the policy requirements set out in the above paragraphs.
- 6.5.25 The significant harm to the 10 visual receptors will in this case be demonstrably outweighed by the overriding benefits of the Scheme as set out at section 4 of the Planning Statement, allowing the Scheme to be approved as an exceptional case. The visual harm has been minimised and mitigated as required and shown within Tables 8.68 8.71 of ES Chapter 8: Landscape and Visual Impact [APP-046].

Landscape and visual effects during construction and decommissioning

- 6.5.26 Landscape and visual impacts will be mitigated during construction and decommissioning through delivery of the Outline LEMP [EN010132/EX6/WB7.3_E].REP6-025]. General measures to reduce construction and decommissioning phase impacts are also set out in the Outline Construction Environmental Management Plan [EN010132/EX6/WB7.1_DREP6-021] and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023].
- 6.5.27 ES Chapter 8: Landscape and Visual Impact [APP-046] assesses the temporary impacts of the Scheme on the landscape and on visual amenity of sensitive receptors during the construction and decommissioning periods. During construction, no likely significant adverse landscape or visual effects are identified within ES Chapter 8: Landscape and Visual Impact [APP-046]. At decommissioning, planting will have been established and no significant effects on landscape or visual receptors are predicted.
- 6.5.28 The Scheme therefore accords with NPS EN-1 (2011) paragraph 5.9.15 and NPS EN-1 (November 2023) paragraph 5.10.34 as ES Chapter 8: Landscape and Visual Impact [APP-046] concludes that the design of the Scheme has been successful in ensuring that there are no identified significant landscape or visual effects during construction and decommissioning of the Scheme. The benefits (including need) of



the Scheme, outweigh the less than significant impacts. The Scheme is also considered to comply with Sturton by Stow and Stow NP Policy 9, Sturton Ward NP policy 2a, Tresswell and Cottam NP policy 1 as it has been shown not to result in significant harm during the construction and decommissioning phases and the less than significant harm has been minimised and mitigated.

Green Infrastructure Provision

- 6.5.29 Paragraph 5.10.2 of NPS EN-1 (2011) explains that the Government's policy is to ensure there is adequate provision of high-quality open space (including green infrastructure) and sports and recreation facilities to meet the needs of local communities. Green infrastructure in particular will play an increasingly important role in mitigating or adapting to the impacts of climate change.
- 6.5.30 BCSDMP policy DM9 seeks to protect and enhance green infrastructure, biodiversity and geodiversity, landscape and open space. Saxilby with Ingleby NP has a number of relevant policies including Policy 12: Green Infrastructure and Policy 16: Existing and New Non Vehicular Routes. Relevant Sturton by Stow and Stow NP policies include Policy 11: Green Infrastructure.
- Enhancement of local biodiversity is a key objective of the Scheme as outlined within the Design and Access Statement [APP-314 and APP-315] under Objective 3. The existing network of green infrastructure within and surrounding the Sites will be maintained and enhanced and the Biodiversity Net Gain Report [APP-088] demonstrates that the Scheme will result in an overall significant net gain for biodiversity, including an anticipated net gain of 86.80% for habitats (delivered through the creation of other neutral grasslands within the sites), a net gain of 54.71% for hedgerows, and a net gain of 33.25% for river units. Further detail of the habitats to be created and / or enhanced and their management are provided within the Outline LEMP and will be confirmed at detailed design stage [EN010132/EX6/WB7.3 EREP6-025].
- 6.5.32 A summary of the significant measures incorporated within the Scheme to enhance green infrastructure is included above. Further details can be found within the Outline LEMP [EN010132/EX6/WB7.3_E].REP6-025]. These measures will ensure a nature inclusive design and represent a substantial enhancement to the green infrastructure network. The Draft DCO necessitates the submission and approval of a detailed Landscape and ecology Management Plan (LEMP) as a precommencement requirement to deliver the provisions as set out in the Outline LEMP [EN010132/EX6/WB7.3_E].REP6-025].
- 6.5.33 The new permissive footpath to run from the track off Sykes Lane along the Codder Lane Belt and then south and west to rejoin Sykes Lane opposite Hardwick Scrub accords with Saxilby with Ingleby NP Policy 16 which states that new development shall take every opportunity to provide new, or enhance existing, non-vehicular routes including connections with the existing network.



6.5.34 The Scheme therefore complies with Paragraph 5.10.2 of NPS EN-1 (2011) as it provides significant green infrastructure as an integral part of the Scheme. It also complies with BCSDMP policy DM9 and relevant neighbourhood plan policies relating to the retention and enhancement of green infrastructure.

Conclusion

- In accordance with NPS EN-1 (2011) paragraph 5.9.22 and NPS EN-1 (November 2023) paragraph 5.10.27, the design of the Scheme has taken account of the landscape and landform in which it sits and has given careful consideration to its impact (including cumulative impacts) on sensitive receptors. These have factored into the design development at all stages, and the design has directly and effectively responded to potential impacts identified in relation to landscape and visual impact.
- In considering the acceptability of the landscape and visual impacts of the Scheme it is noted that NPS EN-1 (2011) paragraphs 5.9.8 and 5.9.18, and NPS EN-1 (November 2023) paragraphs 5.10.4 to 5.10.6 and 5.10.13 to 5.10.15 acknowledge that NSIP scale energy generation infrastructure is likely to have landscape and visual effects.
- 6.5.37 Taking account of the above, and in accordance with NPS EN-1 (2011) paragraphs 5.9.15, and NPS EN-1 (November 2023) paragraphs 5.10.35, it is considered that the 10 significant adverse residual visual effects of the Scheme at year 15 are clearly and comprehensively outweighed by the benefits of the Scheme set out at Section 4 of the Planning Statement, in terms of delivering renewable energy infrastructure which is urgently needed to create a secure and affordable energy system and to help combat climate change. Furthermore, no significant residual adverse landscape effects are anticipated to arise from the Scheme, whilst a number of significant beneficial landscape effects are anticipated to arise.
- In terms of local policy, the harm has been minimised and mitigated as far as possible as set out in ES Chapter 8: Landscape and Visual Impact [APP-046]. The Scheme is also considered to comply with BCSDMP policies DM4, DM8, DM9 and DM10 and with emerging policies within the DBLP: ST37, ST39, ST40, ST41, ST42, ST43 ST48, ST50 and ST51 and policies S53, S57, S59, S62 and S66 of the CLLP and relevant neighbourhood plan policies in so far as they relate to landscape and visual impacts. The Scheme delivers significant green infrastructure enhancement and with relevant neighbourhood plan policies.

6.6 Heritage

6.6.1 ES Chapter 13, Cultural Heritage [APP-051] and its supporting appendices [APP-105 to APP-125] provide an assessment of the likely effects of the Scheme upon heritage assets, including a description of the significance of the heritage assets. It also considers the contribution of their setting to their significance and the results of archaeological desk-based and field investigations. The assessment is informed by consideration of representative visualisations, where appropriate. This accords with



NPS EN1 (2011) paragraphs 5.8.8 to 5.8.10 and NPS EN-1 (November 2023) paragraph 5.9.9 to 5.9.12.

Designated heritage assets

- There are no designated heritage assets within the Order limits. The combined 5km study area surrounding the West Burton 1, 2, and 3 Sites contains 17 Scheduled Monuments. Broxholme medieval settlement and cultivation remains (NHLE 1016797), the Deserted village of North Ingleby (NHLE 1003570), and the medieval bishop's palace and deer park, Stow Park (NHLE 1019229) each directly abut the Order limits.
- 6.6.3 There are 25 Grade I and Grade II* Listed Buildings within the 5km study area surrounding the Sites. None of these Listed Buildings are located within the West Burton 1, 2, or 3 Sites. There are no Registered Parks and Gardens within 5km of the Sites. There are 54 Grade II Listed Buildings within 2km of the Sites. Details of all the above heritage assets are contained within Section 3 of the Heritage Assessment, Appendix 13.5 [APP-117 to APP-119].
- NPS-EN1 (2011) paragraph 5.8.14 states that: "There should be a presumption in favour of the conservation of designated heritage assets and the more significant the designated heritage asset, the greater the presumption in favour of its conservation should be. Once lost heritage assets cannot be replaced and their loss has a cultural, environmental, economic and social impact. Significance can be harmed or lost through alteration or destruction of the heritage asset or development within its setting. Loss affecting any designated heritage asset resulting from its alteration or development in its setting should require clear and convincing justification".
- 6.6.5 Paragraph 5.9.27 of NPS EN-1 (November 2023) states that: "When considering the impact of a proposed development on the significance of a designated heritage asset, the SoS should give great weight to the asset's conservation. The more important the asset, the greater the weight should be. This is irrespective of whether any potential harm amounts to substantial harm, total loss, or less than substantial harm to its significance."
- NPPF paragraph 205 requires the decision maker to: "When considering the impact of a proposed development on the significance of a designated heritage asset, great weight should be given to the asset's conservation (and the more important the asset, the greater the weight should be)". In the context of the Scheme, noting NPPF paragraph 206, the greatest weight is therefore to be given to the Grade I and Grade II* listed buildings within the 5km Study Area surrounding the Sites, with lesser weight given to the Grade II listed buildings. In terms of the level of impact of the Scheme on the assets affected, Section 13.11 of ES Chapter 13, Cultural Heritage [APP-051] sets out the residual effects following mitigation at the construction, operational and decommissioning stages of the Scheme.
- Table 13.32: Residual effects following mitigation: Construction Phase of ES Chapter 13 [APP-051] shows that during the construction phase, there will be a slight adverse effect (not significant in EIA terms) on the setting of four scheduled monuments and



a moderate adverse effect (significant effect) on the setting of one: the medieval bishop's palace and deer park, Stow Park (1019229). There will be slight adverse effects on the setting of 7 listed buildings.

- Table 13.33: Residual effects following mitigation: Operational Phase of ES Chapter 13 [APP-051] shows that during the operational phase there will be slight adverse effects (not significant in EIA terms) on the setting of one scheduled monument and a large adverse effect (significant effect) on one: the medieval bishop's palace and deer park, Stow Park (NHLE 1019229). There will be a slight adverse effect on the setting of one grade II Listed Building the Church of All Saints, Broxholme (1064095) which is not significant in EIA terms.
- 6.6.9 Table 13.34: Residual effects following mitigation: Decommissioning Phase of ES Chapter 13 [APP-051] shows a slight adverse effect (not significant in EIA terms) on the setting of one Scheduled Monument and a moderate adverse effect (significant effect) on one; the medieval bishop's palace and deer park, Stow Park (NHLE 1019229).
- In terms of the potential for direct physical impacts on designated heritage assets, ES Chapter 13, Cultural Heritage [APP-051] concludes that the only potential direct physical impact anticipated to arise from the Scheme is the potential for damage during construction to *No 21 and Attached Barn to Rear* Grade II Listed Building (NHLE1146594) which is located on the corner of Stow Park Road and High Street in Marton. This is due to the fact that HGVs delivering abnormal loads will need to mount the pavement adjacent to the Listed Building, but Abnormal Loads Specialists 'Wynns' have confirmed that the transport of abnormal loads will be a closely managed process travelling at crawl speed and monitored by the police, and therefore the likelihood of this impact occurring is negligible. Wynns have prepared a report detailing the required movements. This is shown in Appendix F of the Transport Assessment [REP4-036].
- 6.6.11 In summary, residual significant effects in EIA terms are anticipated on the medieval bishop's palace and deer park, Stow Park (NHLE 1019229) during all three phases of development; and on the Deserted village of North Ingleby (1003570) and Broxholme medieval settlement and cultivation remains (1016797) up until year 15 of operation only.

Harm policy test

6.6.12 Further detail on this topic is set out in Stow Park Cultural Heritage Position Statement [REP5-027]. Paragraph 5.9.10 of NPS -EN1 (Paragraph 200 of NPPF) states:

"As part of the ES the applicant should provide a description of the significance of the heritage assets affected by the proposed development, including any contribution made by their setting. The level of detail should be proportionate to the importance of the heritage assets and no more than is sufficient to understand the potential impact of the proposal on their significance. As a minimum, the applicant should have consulted the



relevant Historic Environment Record (or, where the development is in English or Welsh waters, Historic England or Cadw) and assessed the heritage assets themselves using expertise where necessary according to the proposed development's impact."

6.6.13 Paragraph 5.9.22 of NPS -EN1 states:

"In determining applications, the Secretary of State should seek to identify and assess the particular significance of any heritage asset that may be affected by the proposed development, including by development affecting the setting of a heritage asset (including assets whose setting may be affected by the proposed development), taking account of:

- relevant information provided with the application and, where applicable, relevant information submitted during the examination of the application
- any designation records, including those on the National Heritage List for England, or included on Cof Cymru for Wales.
- historic landscape character records
- the relevant Historic Environment Record(s), and similar sources of information
- representations made by interested parties during the examination process
- expert advice, where appropriate, and when the need to understand the significance of the heritage asset demands it "

6.6.14 Paragraph 2.10.118 of NPS -EN3 states:

"As the significance of a heritage asset derives not only from its physical presence but also from its setting, careful consideration should be given to the impact of large-scale solar farms which depending on their scale, design, and prominence, may cause substantial harm to the significance of the asset."

- Paragraphs 3.2.49 to 3.2.59 of ES Appendix 13.5 Heritage Statement [APP-117 to APP-119] assess the significance of the Scheduled Monument and detail the contribution made by its setting, namely the land which was formerly located within the deer park and is not scheduled. Table 5.1 of the Statement of Common Ground (SOCG) with Historic England [EN010132/EX6/WB8.3.3_AREP6-042] provides additional detail on the Applicant's conclusions in identifying how the significance of the monument has been derived, and the potential for harm caused by the Scheme to that significance.
- 6.6.16 The Scheme would not cause any direct physical harm to the significance of the Scheduled Monument as there is no proposed intervention to the fabric of any of the sections of the Scheduled Monument that would result in its permanent loss either wholly or in part. Any harm would be only that caused to the significance of the monument that is derived from its setting. This would occur through the placement of panels within land that was formerly occupied by the medieval deer park.



6.6.17 As detailed in Historic England Advice Note 12¹ pages 5, 15 and 16, a key element of a heritage assessment is to identify how the significance of a heritage asset is derived.

"The context for any analysis of the significance of a heritage asset will be a thorough familiarity with the asset itself, developed through site visits, and appropriate inspection of the fabric, its features, materials and ornament, and also its setting if needed." (Historic England Advice Note 12 page 6)

6.6.18 This is reiterated on Page 4 of Historic England Planning Note 3 (Second Edition): "Setting is not itself a heritage asset, nor a heritage designation, although land comprising

a setting may itself be designated (see below Designed settings). Its importance lies in what it contributes to the significance of the heritage asset or to the ability to appreciate that significance."

- As evidenced at ISH5 (Summary of the Applicant's Oral Submissions [REP5-037]) and in the SoCG with Historic England [EN0101032/EX6/WB8.3.3_AREP6-042], the Applicant understands the Scheduled Monument derives its significance from its archaeological and historic interest as the sole surviving element of a former enclosed medieval space, which is largely understood through desk-based research, particularly aerial imagery and historical documentation. Section 2 of the Cultural Heritage Position Statement [REP5-027] confirms how this view has been formed, based on the Reasons for Designation set out in Official List Entry for the Scheduled Monument. The agrarian landscape, the former MOD petroleum site and the railway, which bisects the Scheduled Monument, have a detrimental effect on the ability to appreciate any remaining elements of the former medieval landscape and are consequently considered to have a detrimental effect on the overall contribution made by setting to the significance of the Scheduled Monument.
- 6.6.20 Paragraph 5.9.14 of NPS -EN1 (Paragraph 208 of NPPF) states:

 "Careful consideration in preparing the scheme will be required on whether the impacts on the historic environment will be direct or indirect, temporary, or permanent."
- The nature of the Scheme was considered by the Applicant as part of the assessment on Cultural Heritage that is provided in ES Chapter 13 Cultural Heritage [APP-051]. A key aspect of the Scheme is its reversible nature which means that landscape features, such as those that are associated with the former deer park, will not be permanently impacted by the Scheme. As such any harm caused to the significance of the Scheduled Monument that is derived from its setting would be reversed following decommissioning of the Scheme.

¹ Historic England (2019) *Statements of Heritage Significance: Analysing Significance in Heritage Assets: Historic England Advice Note 12.* (Online, last accessed 27.03.2024) https://historicengland.org.uk/images-books/publications/statements-heritage-significance-advice-note-12/heag279-statements-heritage-significance/



6.6.22 Paragraph 5.9.25 of NPS -EN1 (Paragraph 203 of NPPF) states:

"The Secretary of State should consider the desirability of sustaining and, where appropriate, enhancing the significance of heritage assets, the contribution of their settings and the positive contribution that their conservation can make to sustainable communities, including to their quality of life, their economic vitality, and to the public's enjoyment of these assets."

Whilst there is currently no public access to the Scheduled Monument (i.e. public rights of way) providing an ability to experience the designated heritage asset or its understanding, this should not inhibit the provision to enhance the potential for communal benefit. How it will be experienced will vary over time and circumstance (as stated in Historic England's GPA3 *The Setting of Heritage Assets*). As evidenced in paragraphs 4.1.1 to 4.1.2 of the Cultural Heritage Position Statement [REP5-027], the Applicant explored a range of different mitigation options that had the potential to enhance the public enjoyment of the asset (for example suggestion IV). Historic England believed the benefits from community engagement would not offset any harm, and so these options were not explored further or considered as part of the design of the Scheme.

6.6.24 Paragraph 5.9.27 of NPS EN1 states:

"When considering the impact of a proposed development on the significance of a designated heritage asset, the Secretary of State should give great weight to the asset's conservation. The more important the asset, the greater the weight should be. This is irrespective of whether any potential harm amounts to substantial harm, total loss, or less than substantial harm to its significance."

6.6.25 Paragraph 5.9.28 of NPS EN1 states:

"The Secretary of State should give considerable importance and weight to the desirability of preserving all heritage assets. Any harm or loss of significance of a designated heritage asset (from its alteration or destruction, or from development within its setting) should require clear and convincing justification."

6.6.26 Paragraph 5.9.30 of NPS -EN1 (Paragraph 206 of NPPF) states:

"Substantial harm to or loss of significance of assets of the highest significance, including Scheduled Monuments; Protected Wreck Sites; Registered Battlefields; grade I and II* Listed Buildings; grade I and II* Registered Parks and Gardens; and World Heritage Sites, should be wholly exceptional."

6.6.27 Paragraph 5.9.31 of NPS -EN1 (Paragraph 207 of NPPF) states:

"Where the proposed development will lead to substantial harm to (or total loss of significance of) a designated heritage asset the Secretary of State should refuse consent unless it can be demonstrated that the substantial harm to, or loss of, significance is necessary to achieve substantial public benefits that outweigh that harm or loss, or all the following apply:

• the nature of the heritage asset prevents all reasonable uses of the site



- no viable use of the heritage asset itself can be found in the medium term through appropriate marketing that will enable its conservation
- conservation by grant-funding or some form of not for profit, charitable or public ownership is demonstrably not possible
- the harm or loss is outweighed by the benefit of bringing the site back into use "
- 6.6.28 It is acknowledged by the Applicant that substantial harm to or loss of significance of assets of the highest significance, which includes Scheduled Monuments such as The medieval bishop's palace and deer park, Stow Park (NHLE 1019229), should be wholly exceptional and consent should be refused for that element of the Scheme unless it can be demonstrated that the substantial harm to or loss of significance is necessary to achieve substantial public benefits.
- Through thorough assessment, the Applicant does not consider that the Scheme would cause substantial harm to The medieval bishop's palace and deer park, Stow Park (NHLE 1019229). As detailed in Paragraph 3.1.1 of the Cultural Heritage Position Statement [REP5-027], the Scheme would not cause any direct impact to the fabric of the Scheduled Monument, and there would be no adverse effects to its heritage values that would result in its permanent loss either wholly or in part and consequently the legibility of the deer park would be unaltered. Any effects resulting in a level of harm to the significance of the monument would be derived from changes to its setting through the placement of panels within land that was formerly occupied by the medieval deer park.
- 6.6.30 The Applicant believes, as evidenced by the Official List Entry for the Scheduled Monument (See Section 2, Paragraphs 2.1.3 to 2.1.7 of the Cultural Heritage Position Statement [REP5-027]), that the significance of the Scheduled Monument is primarily derived from its historical and archaeological interest, vested in the Scheduled earthwork features and potential below ground remains, together with that appreciated through desk-based research, particularly aerial imagery and historical documentation. Setting contributes to the understanding of these heritage interests, albeit denuded by the current composition of the landscape in which the Scheduled Monument is located. The post-medieval and modern agrarian land uses does preclude the ability to experience or appreciate the former medieval landscape of the Scheduled Monument. Furthermore, this same post medieval and modern activity has resulted in an adverse effect on elements within its setting as evidenced in the list entry which deliberately excludes post-medieval and modern features and highlights the adverse direct impacts that agricultural activity has had on the Scheduled Monument (see Paragraph 2.1.7 of the Cultural Heritage Position Statement [REP5-027]).
- 6.6.31 The reversible nature of the Scheme means that any less than substantial harm to significance as a result of changes in the setting of the Scheduled Monument would be temporary and reversed entirely following decommissioning of the Scheme.
- 6.6.32 Paragraph 5.9.32 of NPS -EN1 (Paragraph 208 of NPPF) states:



"Where the proposed development will lead to less than substantial harm to the significance of the designated heritage asset, this harm should be weighed against the public benefits of the proposal, including, where appropriate securing its optimum viable use."

- 6.6.33 The Applicant's assessment has found that the Scheme would cause less than substantial harm (at the upper end) to the Scheduled Monument. The introduction of solar panels would not cause direct physical harm to the three isolated elements of Scheduled Monument that form the surviving vestiges of the deer park. Any harm would therefore be solely to the significance the Scheduled Monument derives from its setting (i.e. would be indirect). The layout of the Scheme means that the legibility of the landscape would be unaltered. This is an important factor in the consideration of the temporary nature of the Scheme and any harm to the significance as a result of changes in the setting of the Scheduled Monument which would be reversed entirely following decommissioning of the Scheme.
- 6.6.34 As detailed of Common in the Statement Ground [EN0101032/EX6/WB8.3.3_AREP6-042], the Applicant notes that "Historic England considers that the impact of the Scheme on land within the former deer park as defined by The medieval bishop's palace and deer park, Stow Park Scheduled Monument (NHLE 1019229) would cause substantial harm (in NPS/NPPF terms) / significant environmental impact (major harmful; in EIA terms) to the significance of the Monument through loss of its character as a bounded architectural space." Consequently HE "object to installation of any part of the development within the former deer park (as defined by the lines of the scheduled Park Pale and its former course)."2 The Applicant respectfully disagrees that the Scheme represents a loss to the character of the bounded architectural space of the former deer park. The internal space of the Deer Park does not have any designation (i.e. form a Scheduled Monument, Registered Park and Garden, or Conservation Area). The Applicant believes that this is largely due to the absence of any landscape features that can be attributed or associated with the deer park and that would add to our understanding of how it functioned. Additionally, the sense of a space imparked is not clearly appreciable with the current land use, as that both within and beyond the former boundaries being indistinguishable in its agricultural use. Consequently, the surviving vestiges of the deer park are not experienced collectively within the modern landscape, and it is difficult to reconstruct, understand and appreciate an imparked high status medieval space without the aid of aerial imagery or historical documentation. Instead, the experience is of an agrarian landscape, and the post enclosure field system is the dominant experience.
- 6.6.35 If the Secretary of State is minded to agree that the Scheme will cause less than substantial harm, the Applicant notes that the Secretary of State must give "great weight to the asset's conservation" and "considerable importance and weight to the

² Please see **Draft**-Statement of Common Ground with Historic England [EN0101032/EX6/WB8.3.3_AREP6-042] Table 5.1



desirability of preserving" the asset with any harm or loss of significance require clear and convincing justification. Having applied the abovementioned weight, the policy tests confirm that where the public benefits of the Scheme can be demonstrated to outweigh harm to the significance (as a result of changes to the setting) of the three elements constituting the Scheduled Monument (NHLE 1019229), consent should be approved. Sections 4 and 7 of this document evidence the benefits of the Scheme, which are considered to outweigh any less than substantial harm to the Scheduled Monument.

6.6.36 Paragraph 5.9.36 of NPS -EN1 states:

"When considering applications for development affecting the setting of a designated heritage asset, the Secretary of State should give appropriate weight to the desirability of preserving the setting such assets and treat favourably applications that preserve those elements of the setting that make a positive contribution to, or better reveal the significance of, the asset. When considering applications that do not do this, the Secretary of State should give great weight to any negative effects, when weighing them against the wider benefits of the application. The greater the negative impact on the significance of the designated heritage asset, the greater the benefits that will be needed to justify approval."

- As evidenced in Paragraphs 4.1.1 to 4.1.2 of the Stow Park Cultural Heritage Position Statement [REP5-027], the Applicant explored a range of different mitigation options that had the potential to better reveal the significance of the Asset (for example suggestion III). Historic England believed the benefits from community engagement would not offset any harm, and so these options weren't explored further or considered as part of the design of the Scheme.
- 6.6.38 In terms of local planning policy, Policy S57 and DBLP Policy 43 set out that the level of harm of any development proposal on a designated heritage asset and its setting should be considered against the public benefits of the proposed development. These policies set out similar principles to national policy, in that considerable importance and weight should be given to preserving all designated heritage assets.
- The assessment of the effects of the Scheme on designated heritage assets is summarised above at paragraph 6.6.11 and in Stow Park Cultural Heritage Position Statement [REP5-027]. With regards to the medieval bishop's palace and deer park, Stow Park (NHLE 1019229), the level of harm is assessed within ES Chapter 13: Cultural Heritage [APP-051] to be moderate adverse on the setting of the asset at construction and decommissioning and large adverse during operation, in terms of the EIA Regulations. No direct physical impact is anticipated as a result of the Scheme. The Heritage Assessment, Appendix 13.5 [APP-117 to APP-119] concludes that this level of harm would equate to the upper end of the "less than substantial" scale given the anticipated up to 60 year operational life of the Scheme and the reversible nature of the impact.
- 6.6.40 At the end of its operational life, the Scheme will be decommissioned in accordance with the principles set out in the Outline Decommissioning Statement



[EN010132/EX6/WB7.2_BREP6-023], which will be secured through a DCO requirement. There will be no permanent loss of the significance of designated assets as a result of the Scheme, allowing future generations to retain an understanding of their settings.

6.6.41 The significant public benefits of the Scheme in terms of renewable energy generation and climate change benefits set out at Section 4.0 and discussed from a policy perspective at Section 6.2 of this Planning Statement clearly and demonstrably outweigh the reversible, less than substantial residual harm to the medieval bishop's palace and deer park, Stow Park (NHLE 1019229). The Scheme, therefore, is considered to be compliant with the policy set out in NPS EN-1 (2011) at paragraph 5.8.12; NPS EN-1 (November 2023) at paragraph 5.9.28; CLLP Policy S57; and DBLP Policy 43 in relation to its impact on designated heritage assets.

Non-Designated Heritage Assets

- NPS EN-1 (2011) paragraph 5.8.6 and paragraph 209 of the NPPF state that the decision maker should also consider the impacts on non-designated heritage assets. Paragraph 5.8.12, NPS EN-1 (2011) sets out that for any heritage asset the particular significance of the asset and the value that it holds for this, and future generations should be taken into account. NPS EN-1 (November 2023) sets out at paragraph 5.9.28 that "In weighing applications that directly or indirectly affect non-designated heritage assets, a balanced judgement will be required having regard to the scale of any harm or loss and the significance of the heritage asset."
- 6.6.43 In terms of CLLP Policy S57 and DBLP Policy 43 set out similar principles to national policy, in that harm or loss to non-designated heritage assets should be minimised through design and the benefits of a scheme should be required to outweigh any harm to non-designated heritage assets that would result.
- 6.6.44 The assessment of the effects of the Scheme on non-designated heritage assets is reported in ES Chapter 13: Cultural Heritage [APP-051] and summarised below:
- Table 13.32: Residual effects following mitigation: Construction Phase of ES Chapter 13: Cultural Heritage [APP-051] shows that during the construction phase, there could be potentially significant residual adverse effects on six non-designated archaeological remains.
- Table 13.33: Residual effects following mitigation: Operational Phase of ES Chapter 13: Cultural Heritage [APP-051] shows that during the operational phase there are anticipated to be significant neutral to moderate beneficial effects on ten non-designated archaeological remains and significant neutral to large beneficial effects on twelve non designated heritage assets. There are anticipated to be significant moderate adverse effects on four non designated historic landscapes.
- 6.6.47 Table 13.34: Residual effects following mitigation: Decommissioning Phase of ES Chapter 13 Cultural Heritage [APP-051] shows no significant effects on any non-designated heritage assets at decommissioning stage.



As none of the non-designated assets are of equal significance to designated assets, then the substantial harm test does not apply. The significant public benefits of the Scheme set out at Section 4.0 of the Planning Statement clearly and demonstrably outweigh the reversible, low level, less than substantial harm to non-designated heritage assets, that would result. The Scheme, therefore, satisfies the requirements of NPS EN-1 (2011), NPS EN-1 (November 2023) and the NPPF in relation to its impact on non- designated heritage assets. The design of the Scheme has been carefully and sensitively developed to minimise harm to non-designated heritage assets and their settings, most notably through the embedded mitigation which avoids areas of archaeological sensitivity entirely or preserves them in situ by mounting the solar panels on concrete feet. The scheme therefore also complies with CLLP Policy S57 and DBLP Policy 43 through minimising harm or loss to non-designated heritage assets through design and through the benefits of the Scheme outweighing the harm to non-designated heritage assets.

<u>Summary</u>

- An assessment of the likely effects of the Scheme upon heritage assets, including a description of the significance of the heritage assets has been undertaken within ES Chapter 13, Cultural Heritage [APP-051] and its supporting appendices [APP-105 to APP-125] as required by NPS EN-1 (2011) paragraphs 5.8.8 to 5.8.10 and NPS EN-1 (November 2023) paragraph 5.9.9 to 5.9.12.
- 6.6.50 Significant residual impacts upon the setting of designated heritage assets are limited to the medieval bishop's palace and deer park, Stow Park (NHLE 1019229). These are concluded, within the Heritage Assessment, Appendix 13.5 [APP-117 to APP-119] to be less than substantial, albeit the impacts are anticipated to be towards the upper end of this scale. In accordance with NPS EN-1 (2011) paragraph 5.8.12, SCLLP Policy LP25, CLLP Policy S57 and DBLP Policy 43, this less than substantial harm should be weighed against the public benefits of the proposal. In this case, the significant public benefits of the Scheme in terms of renewable energy generation and climate change benefits set out at Section 4.0 and discussed at Section 6.2 of this Planning Statement clearly and demonstrably outweigh the reversible, less than substantial harm to the designated heritage asset.
- 6.6.51 The significant public benefits of the Scheme set out at Section 4.0 of the Planning Statement also clearly and demonstrably outweigh the reversible, low level, less than substantial residual harm to ten non-designated heritage assets, that would result, and the Scheme therefore accords with NPS EN-1 (2011) paragraph 5.8.6, NPS EN-1 (November 2023) paragraph 5.9.33, paragraph 209 of the NPPF and SCLLP Policy LP25, CLLP Policy S57 and DBLP Policy 43.

6.7 Agriculture

6.7.1 Agricultural land can be classified as grade 1, 2, 3a, 3b, 4 and 5 in accordance with its quality and productivity. This is known as its agricultural land classification (ALC)



- grade. Agricultural land classified in grades 1, 2 and 3a of the ALC is defined as 'best and most versatile' agricultural land (BMV land).
- 6.7.2 National and local planning policy is consistent in seeking to minimise impact on BMV land. It also seeks to guide development away from BMV land where possible, except where its use is justified by other sustainability considerations. National and local policy also requires the use of BMV land to be justified.
- 6.7.3 NPS EN-1 (2011) paragraph 5.10.8 states: "Applicants should seek to minimise impacts on the best and most versatile agricultural land (defined as land in grades 1, 2 and 3a of the Agricultural Land Classification) and preferably use land in areas of poorer quality (grades 3b, 4 and 5) except where this would be inconsistent with other sustainability considerations." This is also reflected in paragraph 5.11.12 of the NPS EN-1 (November 2023).
- 6.7.4 NPS EN-1 (2011) paragraph 5.10.15 and NPS EN-1 (November 2023) paragraph 5.11.34 state that the decision maker: "should ensure that applicants do not site their scheme on the best and most versatile agricultural land without justification" and that little weight should be given to the loss of poorer quality agricultural land (in grades 3b, 4 and 5).
- 6.7.5 NPS EN-3 (November 2023) provides clarification and guidance on how policies relating to BMV agricultural land should be interpreted for solar NSIP schemes. It clarifies at paragraphs 2.10.15 and 2.10.16 that the development of solar arrays on BMV agricultural land is not prohibited and that given the scale of NSIP solar projects, the use of some agricultural land is likely. The compliance with policy is considered in light of this important clarification of the policy context.
- 6.7.6 At paragraphs 2.10.30 and 2.10.31 of NPS EN-3 (November 2023) it is clarified that whilst the development of ground mounted solar arrays is not prohibited on sites of agricultural land classified 1, 2 and 3a, or designated for their natural beauty, or recognised for ecological or archaeological importance, the impacts of such are expected to be considered and are discussed under paragraphs 2.10.73 to 2.10.92 and 2.10.107 to 2.10.11. It is recognised that at this scale, it is likely that applicants' developments may use some agricultural land, however, applicants should explain their choice of site, noting the preference for development to be on brownfield and non-agricultural land.
- 6.7.7 Local planning policies CLLP S67 (including main mods) and DBLP ST51 require justification for the use of best and most versatile agricultural land.
- 6.7.8 CLLP Policy S67 additionally requires the need for the development to be clearly established and the benefits and/or sustainability considerations to outweigh the need to protect such land. The clear need for the Scheme and its benefits are set out at Section 4.0 and discussed from a policy perspective at Section 6.2 of the Planning Statement. These significant public benefits are sufficient to outweigh the need to protect the 26.24% BMV land (made up of within the Sites in accordance with CLLP Policy S67.



6.7.9 BCSDMP DM10 states:

"The Council will be supportive of proposals that seek to utilise renewable and low carbon energy to minimise CO2 emissions. Proposals for renewable and low carbon energy infrastructure will also need to demonstrate that they:

ii. will not lead to the loss of or damage to high-grade agricultural land (Grades 1 & 2);"

- 6.7.10 It should be noted that the element of the Scheme that crosses into Bassetlaw district is only the Grid Connection, buried cable and Electrical Switchgear housing at the Grid Connection Point. Solar panels would not be sited within Bassetlaw District. Paragraph 19.3.9 of ES Chapter 19: Soils and Agriculture [APP-057] anticipates that there will be limited impact from the Cable Route Corridor works on soils, agricultural land and farming activity. This is because the duration of cable laying works will be brief. The cable laying work will be similar to that for the existing routine practice of installing agricultural field drains, typically renewed after 60 years of operation.
- 6.7.11 The following paragraphs consider the compliance of the Scheme with the policy objectives listed below, which are derived from the policy context described above:
 - 1. Sequential assessment of ALC and use of lower quality land in preference to BMV agricultural land.
 - 2. Minimisation of the impact on BMV agricultural land.
 - 3. Justification for the use of BMV land.
- In considering the Scheme, the Applicant has had regard to agricultural land quality. Detailed Agricultural Land Classification surveys (ALC) have been undertaken to identify the grade of the land within the Sites and are reported in ES Chapter 19: Soils and Agriculture [APP-057] and associated Appendix 19.2 Agricultural Land Classification Reports [REP3-016] ALC assessment has not been undertaken for the Cable Corridor Route. This is because the development proposed is a buried cable, with the interruption of the existing agricultural use limited to the brief cable laying operation.

Sequential Assessment of ALC

- 6.7.13 The Site Selection Assessment **[AS-004]** details the five-stage process that the Applicant undertook to select the location of the Scheme. This process is summarised at Section 6.3 above.
- 6.7.14 There was no obviously preferable site that would enable construction of a solar farm of a comparable scale to the Scheme on non-agricultural land, or land that is of a lower ALC grade than the vast majority of the land within the Sites. The land within the Sites therefore passes a sequential assessment based upon agricultural land quality.
- 6.7.15 The sequential approach taken in the Site Selection Assessment **[AS-004]** has demonstrated that there are insufficient areas of available non-BMV land without



constraints, on which to accommodate the whole Scheme. The Scheme therefore complies with this element of the policy and with NPS EN-1 (2011) paragraph 5.10.8 and NPS EN-1 (November 2023) paragraphs 5.11.12 to 5.11.14 and 5.11.18.

Minimisation of the impact on BMV agricultural land

- 6.7.16 The Applicant has taken account of ALC ratings and agricultural land productivity throughout the development of the Scheme design and sought to minimise the amount of BMV agricultural land included within the Sites. At the start of the Scheme this included discussion with the landowners in order to focus the Scheme on land known from decades of experience to be least agriculturally productive and most difficult to farm effectively. This has minimised the impact of the Scheme on the viability of the wider landholding.
- 6.7.17 ES Chapter 5: Alternatives and Design Evolution, [APP-043] and the Design and Access Statement [APP-314 and APP-315] detail how the Sites were refined following detailed ALC assessment. This included the complete removal of the West Burton 4 site from the Scheme following the results of further soil sampling (including in-field carbonates testing) to supplement the initial Agricultural Land Classification reports published at PEIR, which showed the site was 100% BMV land (see Table 5.9 of ES Chapter 5: Alternatives and Design Evolution [APP-043] for further detail.
- Other aspects of the Scheme further act to reduce and minimise the impact on BMV land. Firstly, the Scheme is reversible by its nature and will be decommissioned after the end of its operational life. Upon decommissioning, the above-ground physical infrastructure at the Sites will be removed and the Sites returned to the landowners. This will include the areas of agricultural land where the agricultural resource has been maintained (and potentially improved) during operation, and the established habitats. Post-decommissioning, the landowners may return the Sites to arable use, although it is assumed that established habitats such as hedgerows and woodland would be retained.
- 6.7.19 When considering the impact of the Scheme on BMV agricultural land, it is necessary to distinguish between the agricultural land as a long-term resource, agricultural production, and arable management. The Scheme would not affect the long-term agricultural resource. It would also not affect the continuation of agricultural production if the land was to continue to be grazed. It is only the arable management of part of the Sites which would cease during the life of the Scheme.
- 6.7.20 The Scheme effectively minimises impacts on agricultural land in line with local and national policy by: keeping the inclusion of BMV agricultural land to a low level; retaining the ability to reinstate arable agriculture after decommissioning; and facilitating a continued agricultural use through biodiversity management grazing throughout the operational life of the Scheme.
- 6.7.21 Table 19.11 of ES Chapter 19: Soils and Agriculture **[APP-057]** concludes that without any mitigation required, there will be no significant adverse effects on soils and



agriculture as a result of the Scheme during the construction, operation and decommissioning phases. This includes impacts on the loss of agricultural land resource, loss and degradation of the soil resource and loss of land to the farm businesses and disruption to agricultural occupants outside the Sites. The Chapter also concludes that there will be no significant cumulative impacts.

- 6.7.22 The Scheme is therefore considered to successfully minimise impacts upon BMV land in accordance with NPS EN-1 (2011) paragraph 5.10.8 and NPS EN-1 (November 2023) paragraph 5.11.12 to 5.11.14 and 5.11.18. Nearly three quarters of the land within the Sites is non-BMV land and it has been demonstrated that there are insufficient areas of available non-BMV land without constraints, on which to accommodate the whole Scheme.
- 6.7.23 Furthermore, although more onerous than the requirements of NPS EN-1 (2011) paragraph 5.10.15 and NPS EN-1 (November 2023) paragraph 5.11.34 which seek to ensure that applicants do not site their scheme on BMV land 'without justification', the Scheme is considered to generally comply with the requirements of BCSDMP DM10 which seeks to avoid any loss or damage to BMV land. The Scheme has been demonstrated not to lead to any permanent loss of the 26.24% BMV land included within the Scheme, the temporary loss is reversible and will not lead to damage to this land as explained above.

Justification for the inclusion of some BMV land within Order Limits

- In terms of the specific areas of the 26.24% BMV land that are included within the Scheme, these are justified in accordance with NPS EN-1 (2011) paragraph 5.10.15 and NPS EN-1 (November 2023) paragraph 5.11.34 by particular factors related to their location and context within the Scheme, the wider landholding, and in relation to adjacent and surrounding land. Table 5.9: Stage 4 Design Updates up to DCO Submission (August-November 2022) of ES Chapter 5: Alternatives and Design Evolution, [APP-043] sets out the changes made to the Scheme following detailed ALC assessment and provides the detailed justification for retaining limited areas of BMV land and an explanation as to why others were removed. The reasons why some areas are retained is generally because they form parts of larger fields of lower grade land and it would not be practical to remove these from the Scheme from a Site layout perspective, or to continue to farm them as small, isolated land parcels surrounded by the Scheme.
- 6.7.25 The inclusion of the 26.24% BMV land is further justified by the following:
 - the urgent need for the delivery of a large amount of renewable energy;
 - the lack of identifiable alternative sites within the 15km Search Area around the West Burton Point of Connection;
 - the non-permanent, reversible impact of the Scheme on agricultural land meaning the permanent agricultural resource is not lost;



- the possible retention of an element of agricultural use throughout the life of the Scheme; and,
- the Applicant's careful design to limit the amount of BMV land included within Order limits.
- Provision of a Soil Management Plan (see measures outlined in the Outline Soil
 Management Plan [REP3-016] to ensure the preservation of the soil resource
 at the Sites avoiding both the loss of soil material from the site and the loss
 of soil functional capacity at the Sites. This will ensure that the land will be at
 least equal quality to that which existed prior to the development taking place.
- 6.7.26 The Scheme is therefore considered to comply with NPS EN-1 (2011) paragraph 5.10.15 and NPS EN-1 (November 2023) paragraph 5.11.34. It also demonstrates that once the Scheme has ceased its useful life, the land will be restored to its former use, and will be of at least equal quality to that which existed prior to the development taking place .

Viability of the agricultural holding

- 6.7.27 The Applicant has worked closely with the landowners in developing and finalising the boundary of the Order limits. The aim has been to develop on largely lower quality land within land holdings, to enable the retention of large areas of productive farmland and to avoid the creation of pockets of agricultural land that would be remote from the rest of the agricultural land holdings.
- An assessment of agricultural circumstances has been undertaken and is contained at ES Appendix 19.1: Agricultural Land Quality, Soil Resources and Farming Circumstances [APP-137]. Four farm businesses manage land within the Sites. All are owner occupiers of the land occupied and all own and occupy additional agricultural land outside of the Scheme. Figure 19.4 within ES Chapter 19: Soils and Agriculture [APP-057] shows the extent of land within the Sites for each of these farm businesses. The assessment shows the following:
 - Farm Business A occupies approximately 1200ha. The farmer's opinion is that the solar farm will improve the viability of the farm through diversification. A small reduction in the total arable area will not undermine that enterprise and the farm may respond by taking on additional land should suitable sites become available.
 - Farm Business B is the owner occupier of approximately 810ha of land. The
 farmer considers that the diversification of having land under solar would be
 economically beneficial to the farm. They also anticipate that the extended
 fallow period will improve soil health, in particular the structural stability of
 topsoil that at present, is unable to be managed for direct drilling of crops.
 - Farm Business C is an arable enterprise with approximately 562ha spread across three farm units. Land within the solar farm would comprise approximately 210.8ha out of a total area of 562ha owned by the farm. Farm



Business C is also the landowner of approximately 132.3ha of land that is part of a separate application for a solar energy development. With all farming operations undertaken by contractors, Farm Business C will not have any surplus capacity of farm labour or machinery should one or both solar farm applications progress. Any surplus grain storage could be let to other farm businesses or grain merchants. For Farm Business C the solar farm will be the addition of a diversified enterprise that is not subject to the same fluctuations of income as the arable enterprise.

- Farm Business D manages approximately 274ha of owner occupied land. The farmer wants the diversification to solar to progress so that the farm business has a more stable income that will enable retirement and be able to support any successor taking over the farm.
- 6.7.29 ES Chapter 19: Soils and Agriculture [APP-057] notes that the start of construction work will mark the start of the temporary curtailment of arable management at the Sites for each of the four Farm Businesses. While construction work is taking place the land will not be available for grazing livestock either. It assesses that the sensitivity of the arable enterprises of the four farm businesses to a reduction in cropped area will be medium. With the arable enterprises constrained rather than terminated, the magnitude of change is low. The resulting short term, reversable and local effect of construction disturbance on the farm businesses occupying land within the Sites will be a Minor Impact that is not considered significant in EIA terms.
- During operation, grass below and between the solar panels will need to be managed. This management can include grazing by livestock where appropriate. All four farm businesses will receive income from the Scheme's occupation of their land, a new diversified enterprise. This diversified enterprise will provide a new income stream independent of variations in profitability of arable production. The sensitivity of the farm businesses to this creation of a new farm enterprise will be medium (beneficial) with the magnitude of change being medium. For the operational phase there will be a reversable moderate impact which is a medium term significant beneficial effect in EIA terms.
- 6.7.31 Decommissioning of the Scheme will allow a return to arable management of the land. However, there is no obligation for land to return to arable production just as at present there is no obligation to maintain arable management. There is assessed to be a short term, reversable and local effect of decommissioning on the return of agricultural land to the farm businesses. This will be a Minor Impact, beneficial and not significant in EIA terms.
- 6.7.32 By avoiding as far as possible, best and most versatile agricultural land, enabling continuation of grazing by livestock where appropriate during the operational phase, provision of a new income stream for the farm businesses which is independent of variations in profitability of arable production and enabling a return to arable management of the land upon decommissioning, the impacts of the proposal upon ongoing agricultural operations have been minimised. This approach



accords with NPS EN-1 (2011) paragraph 5.10.8 and NPS EN-1 (November 2023) paragraph 5.11.12 to 5.11.14 and 5.11.20. It also accords with the requirements of local planning policies CLLP S67, BCSDMP DM10 and DBLP ST51.

Summary

- 6.7.33 The Scheme successfully minimises impacts upon BMV land in accordance with NPS EN-1 (2011) paragraph 5.10.8 and NPS EN-1 (November 2023) paragraph 5.11.12 to 5.11.14 and 5.11.20, CLLP S67, BCSDMP DM10 and DBLP ST51. Specifically:
 - 73.76% of the land within the Sites is non-BMV land and it has been demonstrated that there are insufficient areas of available non-BMV land without constraints, on which to accommodate the whole Scheme;
 - A sequential approach to the locating of the Scheme which has sought to direct development towards non BMV land has been demonstrated;
 - The inclusion of a limited amount of BMV land within the Scheme has been justified by the nature of the Scheme and its design in accordance with NPS EN-1 (2011) paragraph 5.10.15 and NPS EN-1 (November 2023) paragraph 5.11.34;
 - The Site Selection Assessment [AS-004] demonstrates that the use of any other land in this area for a comparably sized scheme would likely result in a similar impact on agricultural land;
 - The impacts of the proposal upon ongoing agricultural operations have been minimised by enabling continuation of grazing by livestock and provision of a new income stream for the farm businesses which is independent of variations in profitability of arable production; and
 - Once the Scheme has ceased operation and been decommissioned, the land will be restored to its former use, and will be of at least equal quality to that which existed prior to the Scheme taking place.
- 6.7.34 The significant public benefits of the Scheme set out at section 4.0 of the Planning Statement outweigh the reversible loss of 26.24% BMV agricultural land for the duration of the Scheme, particularly noting that NPS EN-3 (November 2023) paragraph 2.10.29 states that land type should not be the predominating factor in determining the suitability of a site for solar development.

6.8 Mineral Safeguarding

- 6.8.1 The Applicant has considered the impact of the Scheme on safeguarded minerals and has concluded that no sterilisation of minerals within the Sites or the Cable Route Corridor would result, as no impediment to mineral extraction would remain after the Scheme has been decommissioned.
- 6.8.2 With regard to mineral safeguarding, paragraph 5.10.9 of NPS EN-1 (2011) and paragraph 5.11.19 of NPS EN-1 (November 2023) state that applicants should



safeguard any mineral resources on the proposed site as far as possible, taking into account the long-term potential of the land use after any future decommissioning has taken place. Paragraph 5.10.22 of NPS EN-1 (2011) and paragraph 5.11.28 of the NPS EN-1 (November 2023) further add that the decision maker should ensure that appropriate mitigation measures have been put in place to safeguard mineral resources in the event that a proposed development has an impact on a Mineral Safeguarding Area (MSA). Paragraph 218 of the NPPF states that Local Planning Authorities "should not normally permit development proposals in Mineral Safeguarding Areas if it might constrain potential future use for mineral working".

- 6.8.3 Lincolnshire Minerals and Waste Local Plan Core Strategy (LMWCS) Policy M2 states that the County Council will ensure a steady and adequate supply of sand and gravel for aggregate purposes by making provision for sand and gravel extraction. LMWCS Policy M11 and Nottinghamshire Minerals Local Plan (March 2021) (NMLP) Policy SP7 require that sand and gravel, blown sand and limestone resources that are considered to be of current or future economic importance within the Minerals Safeguarding Areas will be protected from sterilisation from other development. Applications for non-minerals development in a minerals safeguarding area must be accompanied by a Minerals Assessment. Planning permission will be granted for development within a Minerals Safeguarding Area provided that it would not sterilise mineral resources.
- 6.8.4 LMWCS M4 identifies Areas of Search for sand and gravel where planning permission will be granted for sand and gravel extraction for aggregate purposes where the site is required to meet a proven need that cannot be met from the existing permitted reserves; or a specific shortfall in the landbank of the relevant Production Area. New sites must also form an extension to an existing Active Mining Site; or replace an existing Active Mining Site that is nearing exhaustion.
- 6.8.5 ES Chapter 12: Minerals **[APP-050]** assesses the impacts of the Scheme upon mineral resources in accordance with LMWCS Policy M11 and NMLP Policy SP7. Neither the Lincolnshire Minerals and Waste Local Plan nor the Nottinghamshire Minerals Local Plan make any allocations for future mineral extraction in the vicinity of the Sites. The Scheme and immediate surroundings are not currently subject to mineral working. There is no apparent evidence to suggest there has been any mineral working in the recent past within the area covered by the Scheme.
- 6.8.6 The majority of the land within the Sites is located outside any Mineral Safeguarding Areas or the Area of Search. A 12 ha area of land in the south western corner of West Burton 3 is within a Lincolnshire Mineral Safeguarding Area and 180ha along the western side is within the Area of Search. The majority of the cable route within Nottinghamshire is identified in the Nottinghamshire Minerals Local Plan as being within a sand and gravel Mineral Safeguarding Area. This safeguarding area continues across the River Trent for approximately 0.4 km into Lincolnshire extending in an easterly direction beyond Brampton. These areas are shown on the Minerals Resource Plans [APP-285 to APP-288].



- 6.8.7 ES Chapter 12: Minerals **[APP-050]** concludes that in terms of the Mineral Safeguarding Areas the Scheme affects relatively small areas of deposits which are of much greater geographic extent. It is therefore very unlikely that these small areas would be required for mineral extraction during the life of the Scheme.
- In view of the current policies of the Mineral Planning Authority, the current sand and gravel landbank and the extensive areas covered by the Area of Search, ES Chapter 12: Minerals [APP-050] concludes that it is highly unlikely that the sand and gravel reserve within the Area of Search will need to be worked within the lifetime of the Scheme.
- 6.8.9 The proposed cabling connecting the individual Sites to each other and the grid are unlikely to sterilise any significant volume of safeguarded mineral. The proposed Cable Route Corridor particularly in the Trent Valley, however, does have the potential to result in operational issues for future mineral operations and might restrict the efficient exploitation of the resource. This impact has been mitigated wherever possible by cable routes following existing infrastructure corridors or edges of significant landscape features. This is in accordance with paragraph 5.10.22 of NPS EN-1 (2011) and paragraph 5.11.28 of the NPS EN-1 (November 2023) which require that appropriate mitigation measures have been put in place to safeguard mineral resources.
- 6.8.10 The Scheme will be decommissioned at the end of its operational life, all above ground structures will be removed and the Sites restored. The Scheme does not require deep excavations or foundations and thus disturbance is limited to the surface layers rather than underlying deposits thus any underlying mineral deposit would not be permanently sterilised and would be available to exploit if required at a future date. The Scheme therefore complies with LMWCS Policy M11 and Nottinghamshire Minerals Local Plan (March 2021) (NMLP) Policy SP7 in ensuring that mineral deposits within the Minerals Safeguarding Areas will be protected from sterilisation.
- 6.8.11 The whole of the Scheme is within a Petroleum Exploration and Development License (PEDL) area where oil and gas extraction is licensed under the Petroleum Act 1998. A PEDL allows the pursuit a range of oil and gas exploration activities, subject to necessary drilling/development consents and planning permission.
- 6.8.12 Oil and gas deposits are found at much greater depths than other minerals and therefore surface development has less potential impact in terms of exploiting the resource. Neither Lincolnshire nor Nottinghamshire have identified Mineral Safeguarding Areas for hydrocarbons. Existing oil fields are identified and safeguarded with mineral consultation zone around each. The Scheme does not affect an existing oil field or come within a mineral consultation zone. It is not considered that the proposed Scheme would have any implications for existing or proposed exploration and eventual exploitation of oil and gas resources.

<u>Summary</u>



6.8.13 The Scheme has therefore been demonstrated not to sterilise mineral resources and is considered to be in accordance with paragraph 5.10.9 of NPS EN-1 (2011) and paragraph 5.11.19 of NPS EN-1 (November 2023) and Lincolnshire Minerals and Waste Local Plan Core Strategy Policy M2 and M11 and Nottinghamshire Minerals Local Plan (March 2021) Policy SP7.

6.9 Ecology and Biodiversity

- 6.9.1 NPS EN-1 (2011) paragraph 5.3.3 and NPS EN-1 (November 2023) paragraphs 5.4.17 and 5.4.18 state that the ES should clearly set out any effects on internationally, nationally and locally designated sites of ecological or geological conservation importance, on protected species and on habitats and other species identified as being of principal importance for the conservation of biodiversity. NPS EN-1 (2011) paragraphs 5.3.8, 5.3.9, 5.3.10, 5.3.11 and 5.3.13, and NPS EN-1 (November 2023) paragraphs 5.4.48, 5.4.4, 5.4.5, 5.4.7, 5.4.12 and 5.4.13 expect the Secretary of State to attach appropriate weight to these ecological receptors noting the most important are those identified through international conventions and European legislation and to consider those that are also proposed for designation.
- As a general principle, paragraph 5.3.7 of NPS EN-1 (2011), expects development to "avoid significant harm to biodiversity and geological conservation interests, including through mitigation and consideration of reasonable alternatives; where significant harm cannot be avoided, then appropriate compensation measures should be sought". NPS EN-1 (November 2023) paragraphs 5.4.42 and 5.4.43 set out the same principle, but with the additional point that if significant harm to biodiversity resulting from a development cannot be avoided, mitigated or compensated for then the Secretary of State will give "significant weight to any residual harm". The NPPF at paragraph 186(a) goes further and directs the decision maker to refuse consent if significant harm to biodiversity resulting from a development cannot be avoided, adequately mitigated or as a last resort compensated for. NPS EN-3 (2011) paragraph 2.4.2 further adds that renewable energy NSIPs should demonstrate 'good design' by mitigating effects on ecology.
- 6.9.3 Paragraphs 2.10.128 to 2.10. 130 of NPS EN-3 (November 2023) sets out that where there are proposed biodiversity enhancements incorporated within solar farm developments, these should aim to achieve biodiversity net gain in line with the ambition set out in the 25 Year Environment Plan and should take account of the factors set out in section 5.4 of NPS EN-1 (November 2023). These include embedding opportunities for nature inclusive design in the design process. Paragraph 5.4.5 of NPS EN-1 (November 2023) sets out that the Secretary of State should have regard to the aims and goals of the Government's 25 Year Environment Plan when making their decision. It also acknowledges that the benefits of significant low carbon infrastructure in themselves may include benefits for biodiversity and that those benefits may outweigh other harm to biodiversity interests.
- 6.9.4 CLLP Policy S14 and CLLP S60 require the protection of habitats and species, minimisation of impacts upon biodiversity and seek to deliver a net gain in



biodiversity. CLLP Policy S60 and DBLP Policy ST40 set out a hierarchy of sites which will apply in the consideration of development proposals with the highest level of protection to be afforded to internationally protected sites. Development likely to have an adverse effect on locally designated sites, their features or their function as part of the ecological network, will only be supported where the benefits of the development clearly outweigh the loss, and the coherence of the local ecological network is maintained. CLLP Policy S56 also requires potential environmental impacts on biodiversity to be taken into consideration.

- 6.9.5 CLLP Policy S66 states that development proposals should be prepared based on the overriding principle that the existing tree and woodland cover is maintained, improved and expanded. DBLP Policy 41 also seeks to protect trees and hedgerows. BDCSDMP Policy DM9 states that development proposals will be expected to take opportunities to restore or enhance habitats and species' populations and to demonstrate that they will not adversely affect or result in the loss of features of recognised importance. Such habitats and species include Protected trees and hedgerows; ancient woodlands; Sites of Special Scientific Interest (SSSI); Regionally Important Geodiversity Sites; Local Wildlife Sites (Sites of Importance for Nature Conservation (SINC)); Local and UK Biodiversity Action Plan Habitats; and Protected Species.
- 6.9.6 CLLP Policy S61 states that proposals for major and large-scale development should seek to deliver wider environmental net gains where feasible. All major development proposals must deliver at least a 10% measurable biodiversity net gain attributable to the development. The net gain for biodiversity should be calculated using Natural England's Biodiversity Metric. DBLP Policy ST40 also requires all new development to make provision for at least 10% net biodiversity gain on site.
- 6.9.7 Sturton by Stow and Stow NP Policy 12 states that all developments, projects and activities will be supported which (among other matters) identify, protect, maintain and expand as appropriate networks of ecological interest and provide for appropriate management and identify measures to avoid and/or reduce any potentially adverse impacts on the natural environment to acceptable levels (commensurate with the status of specific sites where applicable). Other matters include mitigating against any necessary impacts through appropriate habitat creation, restoration or enhancement on site or elsewhere and seeking opportunity to conserve, augment and reinstate the stock of trees, hedges, woodlands, wetlands and countryside as wildlife habitat. Saxilby with Ingleby NP Policy 11 supports development which protects and enhances existing features in the natural environment. Development will be expected to retain well-established landscape features such as mature trees, species-rich hedgerows and ponds. Sturton Ward NP Policy 2b states that proposals that would result in the net loss of biodiversity will not be supported. If significant ecological impacts are identified, appropriate mitigation or compensation measures will be required.



6.9.8 ES Chapter 9: Ecology and Biodiversity [APP-047] provides an assessment of the Scheme's impact on ecological receptors and is supported by extensive survey work to confirm the ecological habitats and species likely to be affected by the Scheme. The location of Statutory and Non-Statutory Sites can be found on the Statutory and Non-Statutory Sites/ Features of Nature Conservation Plan [REP4-016]. In accordance with NPS EN-1 (2011) paragraph 5.3.3 and NPS EN-1 (November 2023) paragraphs 5.4.17 and 5.4.18, sites of geological conservation importance have been considered but are not located within the Order limits and have therefore not been identified as receptors requiring assessment.

Internationally designated ecological sites

- 6.9.9 Paragraphs 4.3.1 and 5.3.9 of NPS EN-1 (2011) and paragraph 5.4.8 of the NPS EN-1 (November 2023) set out that the Conservation of Habitats and Species Regulations 2017 (as amended) require the decision maker to consider whether the Scheme may have a significant effect on a European site, or on any site to which the same protection is applied as a matter of policy, either alone or in combination with other plans or projects. In terms of cumulative impacts, ES Chapter 9: Ecology and Biodiversity [APP-047] has assessed that there are no cumulative effects in relation to internationally designated sites.
- 6.9.10 No Special Protection Areas (SPA), Special Areas of Conservation (SAC) or RAMSAR designations are located within 10km of the Scheme.
- 6.9.11 Thorne and Hatfield Moors SPA is located approximately 20.7km north-west of the point of connection at the West Burton Power Station, which is the closest point within the Scheme to the SPA. The closest Site to the SPA is West Burton 3 which is 26.4km south-east of it. The site is designated for its breeding populations of nightjar and is considered to be of international importance.
- 6.9.12 With regard to Hatfield Moor SPA and Thorne Moor SPA, ES Chapter 9: Ecology and Biodiversity [APP-047] concludes that owing to the physical separation between the Scheme and the SPA sites or even potentially functionally linked land, combined with the absence of suitable habitat or survey/desk study records, it is considered that impacts upon the SPAs are unlikely to result from the Scheme. No mitigation measures are considered necessary and no residual effects likely.
- 6.9.13 The Humber Estuary SPA is located approximately 36km from the closest point within the Scheme. ES Chapter 9: Ecology and Biodiversity [APP-047] assesses that it is not considered at risk of adverse effects due to this substantial distance. This conclusion is supported by advice received from Natural England on the subject during application preparation (See Table 9.1 Consultation Summary, ES Chapter 9: Ecology and Biodiversity [APP-047].
- 6.9.14 The Scheme therefore accords with NPS EN-1 (2011) Paragraph 5.3.3 and NPS EN-1 (November 2023) paragraphs 5.4.17 and 5.4.18 in clearly setting out any effects on internationally designated sites. In accordance with paragraph 5.3.7 of NPS EN-1 (2011), the Scheme avoids significant harm to these sites as demonstrated above.



The Scheme also accords with local planning policies CLLP S56, S60 and DBLP ST40 by avoiding adverse impacts on internationally designated nature conservation sites. The Scheme complies with Sturton by Stow and Stow NP Policy 12 and Sturton Ward NP Policy 2b in so far as they relate to the protection of ecological sites.

Nationally designated ecological sites

- 6.9.15 Paragraph 5.3.11 of NPS EN-1 (2011) states that development consent should not normally be granted "where a proposed development on land within or outside an SSSI is likely to have an adverse effect on an SSSI (either individually or in combination with other developments)" with an exception made "where the benefits (including need) of the development at this site, clearly outweigh both the impacts that it is likely to have on the features of the site that make it of special scientific interest and any broader impacts on the national network of SSSIs". This principle is also set out in paragraph 5.4.8 of NPS EN-1 (November 2023) and paragraph 186 of the NPPF.
- No Sites of Special Scientific Interest (SSSI) are located within the Order Limits or within the vicinity of West Burton 1 and West Burton 3. Doddington Clay Woods SSSI is located 4.7km south of West Burton 2. Five SSSIs are located within 5km of the Cable Route Study Area (CRSA). These are Ashton's Meadow SSSI, Lea Marsh SSSI, Clarborough Tunnel SSSI, Chesterfield Canal SSSI and Treswell Wood SSSI. For the purposes of ecological surveys, the Cable Route Study Area comprises a 100m wide swathe of land for the most part, with larger or narrower areas where other constraints or uncertainties were present at the time of adopting the study area. Field surveys within this area took place before the Cable Route Corridor red line was finalised, however the Cable Route Corridor red line is wholly contained within the CRSA.
- 6.9.17 ES Chapter 9: Ecology and Biodiversity [APP-047] has assessed the impact of the Scheme on these 6 sites and identifies that they are all situated between 2.6km and 4.8km away from the Scheme and so are considered to be at a significantly reduced risk from indirect fragmentation or degradation impacts from the construction phase. Direct impacts are not considered likely. In the absence of mitigation, potential impacts upon these sites could arise from minor indirect fragmentation, or reduction in habitat quality from pollution into watercourses or the likely linked hydrological network.
- 6.9.18 Embedded mitigation measures are set out within the Outline EPMS [APP-326] at Section 9.5 and cover the avoidance of accidental dust deposition or pollution events, along with ECoW presence/monitoring and restrictions on working in adverse weather. These measures will be secured by DCO Requirement. ES Chapter 9: Ecology and Biodiversity [APP-047] concludes that with the implementation of these measures, construction and operational phase effects on nationally designated sites are anticipated to be neutral. This conclusion is supported by advice received from Natural England on the subject during application preparation. (See Table 9.1 Consultation Summary, ES Chapter 9: Ecology and Biodiversity [APP-047]. For decommissioning effects see paragraphs 6.9.52 to 6.9.55 below.





- 6.9.19 In terms of cumulative impacts, ES Chapter 9: Ecology and Biodiversity **[APP-047]** has assessed that there are no cumulative effects in relation to nationally designated sites.
- 6.9.20 The Scheme therefore accords with NPS EN-1 (2011) Paragraph 5.3.3 and NPS EN-1 (November 2023) paragraphs 5.4.17 and 5.4.18 in clearly setting out any effects on nationally designated sites. In accordance with paragraph 5.3.7 of NPS EN-1 (2011), the Scheme avoids significant harm to these sites as demonstrated above. The Scheme also accords with local planning policies CLLP S56, S60, DBLP ST40 and BDCSDMP Policy DM9 by avoiding impacts on nationally designated nature conservation sites. The Scheme complies with and Sturton Ward NP Policy 2b and Sturton by Stow and Stow NP Policy 12 in so far as it relates to the protection of ecological sites.

Locally designated sites

- 6.9.21 Paragraph 5.3.13 of NPS EN-1 (2011) and paragraph 5.4.12 of NPS EN-1 (November 2023) state that decision-makers should give due consideration to sites of regional and local biodiversity and geological interest, including Regionally Important Geological Sites, Local Nature Reserves and Local Sites. However, the NPS qualify this statement and state that given the need for new infrastructure, these designations should not be used in themselves to refuse development consent. NPS EN-1 (November 2023) paragraphs 5.4.12 and 5.4.52 also adds that development "will still be expected to comply with the biodiversity and geological conservation requirements set out in this NPS".
- 6.9.22 No locally designated wildlife sites are located within 2km of West Burton 1. Three non-statutorily designated sites of County importance were identified within 2km of West Burton 2. Seven non-statutorily designated sites of County importance were identified within 2km of West Burton 3, several of which are the same as those returned for West Burton 2. Twenty-three Local Wildlife Sites (LWS) of County importance are located within 2km of the CRSA, many of which are the same as listed for the Sites.
- 6.9.23 ES Chapter 9: Ecology and Biodiversity [APP-047] has assessed the impact of the Scheme on all of these sites. Six of the sites: Coates Wetlands LWS, North Leys Road Ditch LWS, Mr. Rose's Hay Meadow LWS, West Burton Meadows LWS, Trent Port Wetlands LWS and Torksey Grassland LWS are located within 100m of the Sites or within the CRSA.
- 6.9.24 Due to the physical separation of these sites from the Order Limits, the potential for direct damage to these habitats is considered to be low. However, their proximity means they are potentially the most susceptible of all the listed designated sites to short to medium-term impacts from possible discharge or deposition of sediments, dust and contaminants. Once the cable is installed, the cable route will remain undisturbed for the life of the Scheme. Therefore, impacts upon these sites are not anticipated during operation.



- 6.9.25 Embedded mitigation measures are set out within the Outline EPMS [APP-326] at Section 9.5 and cover the avoidance of accidental dust deposition or pollution events, along with ECoW presence/monitoring and restrictions on working in adverse weather. These measures will be secured by DCO Requirement. The Outline LEMP [EN010132/EX6/WB7.3_EREP6-025] sets out how habitats will be reinstated following the completion of the cable installation works such that there will be no long-term adverse effects on the habitats within the Cable Route Corridor, and also any functional linkage to the LWSs. Additionally, and with particular reference to North Leys Road Ditch LWS, an Outline Construction Traffic Management Plan (CTMP) [REP4-038EN010132/EX7/WB6.3.14.2 E] has been produced to detail how vehicles, plant and materials will be transported to the construction zone and the measures required to avoid over-run and damage of the verges/ditch banks associated with the LWS. ES Chapter 9: Ecology and Biodiversity [APP-047] concludes that with the implementation of these measures, construction and operational phase effects on these locally designated sites are anticipated to be neutral. For decommissioning effects see paragraphs 6.9.52 to 6.9.55 below.
- 6.9.26 The remaining 17 locally designated sites are all situated between 370m and 4.8km away from the Scheme and so are considered to be at a significantly reduced risk from indirect fragmentation or degradation impacts from the construction phase. Direct impacts are not considered likely. The implementation of embedded mitigation measures set out within the Outline EPMS [APP-326] to control dust and pollution events are anticipated to ensure that any construction and operational phase effects will be neutral. For decommissioning effects see paragraphs 6.9.52 to 6.9.55 below.
- 6.9.27 In terms of cumulative impacts, ES Chapter 9: Ecology and Biodiversity [APP-047] has concluded that the only locally designated sites which are considered at risk of cumulative effects are those in proximity to the part of the Cable Route Corridor within the Shared Cable Corridor. These are; Coates Wetland LWS, Trent Port Wetland LWS (which occur close to the proposed River Trent crossing point) and Cow Pasture Lane Drains LWS. It is proposed that these sites are protected through the use of Horizontal Directional Drilling. In which case, a simultaneous or sequential cable installation programme should not cause any cumulative impacts. Horizontal Directional Drilling will be secured via the Ecological Protection and Mitigation Strategy (EPMS).
- 6.9.28 The Scheme therefore accords with NPS EN-1 (2011) Paragraph 5.3.3 and NPS EN-1 (November 2023) paragraphs 5.4.17 and 5.4.18 in clearly setting out any effects on locally designated sites. In accordance with paragraph 5.3.7 of NPS EN-1 (2011), the Scheme avoids significant harm to these sites as demonstrated above. The Scheme also accords with local planning policies CLLP S56, S60, DBLP ST40 and BCSDMP Policy DM9 by avoiding adverse impacts on locally designated nature conservation sites. The Scheme complies with Sturton Ward Neighbourhood Plan Policy 2b and Sturton by Stow and Stow NP Policy 12 in so far as they relate to the protection of ecological sites.



Protected species and habitats of importance

- 6.9.29 Many individual wildlife species receive statutory protection under a range of legislative provisions. Other species and habitats are also identified as being of principal importance for the conservation of biodiversity. Paragraph 5.3.8 of NPS EN-1 (2011) states that "appropriate weight should be attached to… protected species; habitats and other species of principal importance for the conservation of biodiversity". Paragraph 5.4.48 of NPS EN-1 (November 2023) sets out the same principle and local planning policies including CLLP S60 also seeks to protect these habitats and species.
- Numerous badger setts have been recorded within 2km of each of the Sites. Four main badger setts were identified within the Sites with a further two subsidiary setts and four outlying setts. All setts within the Sites were located at field boundaries. A total of 8 badger setts were recorded during the surveys of the CRSA. These will each be protected within the EPMS that will follow on from the Outline Ecological Protection and Mitigation Strategy [APP-326] through the adoption of a development free buffer zone between 10 and 30m in radius depending on their status. Habitat connectivity for badgers will be maintained and foraging will be enhanced through reversion of the land from arable to grassland. Perimeter fencing will remain permeable to movement by badgers. ES Chapter 9: Ecology and Biodiversity [APP-047] concludes neutral effects on badgers during the construction phase and assuming full implementation of the LEMP, a beneficial residual effect, significant at the Site level.
- As set out within ES Chapter 9: Ecology and Biodiversity [APP-047] a reasonably 6.9.31 diverse assemblage of bat species has been recorded using the Sites, while numerous trees located at field boundaries have roost potential, as do a number of buildings adjacent to the Sites. The arable fields themselves are of low value to bats owing to the uniformity of habitat and low productivity for night flying invertebrate prey. All hedgerows and ditches will be buffered from development as will be set out in the EPMS, and these will be managed for habitat diversity in the LEMP. Buffer widths will vary according to the potential value of the trees within the hedgerows to bats as possible roosts. Substantial planting of new trees and hedgerows will also be undertaken and new linear habitat linkages between isolated trees and nearby woodland will be created. Any trees subject to development impacts will be subject to further inspection and survey and all steps necessary to avoid impacts will be taken. Habitats post-construction on Site are likely to be improved for bat foraging, roosting and dispersal. As a result of the implementation of the measures set out in the LEMP (which are secured via a DCO Requirement), ES Chapter 9: Ecology and Biodiversity [APP-047] concludes a beneficial effect, which is significant at a District level is likely to occur.
- 6.9.32 Several of the larger and more permanently wetted ditches and watercourses on Site support otters and water voles. These watercourses have been buffered by at least 8m, up to 30m in places, from development and enhanced through targeted



ditch management. Impacts on water voles and otters are considered unlikely, with the potential for improvements post-construction. ES Chapter 9: Ecology and Biodiversity [APP-047] concludes that due to the cessation of arable practices which result in runoff of pesticides and other inputs, in combination with the favourable management of wider buffer zones, a beneficial effect significant at a local level should be possible in the operational phase in the medium to long term provided the LEMP is followed in full.

- 6.9.33 The Outline EPMS **[APP-326]** sets out precautionary measures for working. Taking these into account, ES Chapter 9: Ecology and Biodiversity **[APP-047]** concludes the effects on polecats and hedgehogs in the construction phase should be neutral. For the operational phase, provided the LEMP is followed, there are likely to be beneficial effects on these species potentially significant at a District Level.
- 6.9.34 Brown hare and harvest mouse are species associated with the open arable habitats which may be impacted by the proposals. ES Chapter 9: Ecology and Biodiversity [APP-047] concludes an adverse significant impact on harvest mice in the construction phase. However, these local impacts are expected to reduce to Site level in the operational phase due to the partial replacement of lost suitable habitat and cessation of intensive arable practices. Brown hare will continue to have unimpeded access to the array fields and as noted in ES Chapter 9: Ecology and Biodiversity [APP-047] have been seen to benefit from solar arrays at other sites, often increasing in numbers post development.
- 6.9.35 Great crested newt were recorded within one pond within the site and a further four adjacent. The arable habitats to be impacted by the Scheme are of low habitat suitability for great crested newts and all ponds within the Site will be retained and buffered by at least 50m of habitat free of development activities. The development will not adversely impact the movement of amphibians through the landscape. Grass snake and common lizard have been only ever noted on Site once each. The Outline Ecological Protection and Mitigation Strategy [APP-326] sets out protection, supervision and precautionary methods of working for these species required during construction works. With the implementation of these measures, no significant effects are anticipated for these species during construction. Longer term beneficial effects for reptiles, significant at the local level, are expected to result from habitat enhancement measures and favourable management of buffer zones provided the LEMP is implemented in full.
- 6.9.36 Farmland birds such as skylark, yellow wagtail, grey partridge, yellowhammer and lapwing were all recorded either nesting or foraging on the Site, with several other species of conservation concern associated predominantly with the field boundary habitats. Ground nesting species which choose to nest within open arable fields, such as skylark and yellow wagtail, stand to be displaced to a degree by the development. ES Chapter 9: Ecology and Biodiversity [APP-047] concludes for all species, that nest avoidance procedures during the construction phase will ensure that direct impacts on birds and their nests will be minimised to neutral levels.



During operational phase the mitigation proposed is considered to result in neutral and beneficial effects for yellow wagtail and lapwing respectively. Moderate adverse significant effects on skylark and grey partridge at a local level and over wintering birds at a site level are expected to remain.

- 6.9.37 Construction phase impacts on birds during the nesting season will be avoided through a combination of habitat inspections by an Ecological Clerk of Works, sensitive timing of works and the imposition of exclusion buffers around known and potential nest sites. A substantial habitat enhancement package has been produced to focus on areas of the Site which are free of development, whereby ecologically led management will produce a mosaic of grassland and other habitat types of greater foraging and nesting productivity than baseline levels for many of the species recorded.
- 6.9.38 The Scheme has been designed so that negative impacts upon important habitats (comprising woodland, grassland, hedgerow and ponds) are avoided or reduced, and that the habitats are enhanced during the operational life of the Scheme where reasonably practicable. Table 9.3 of ES Chapter 9: Ecology and Biodiversity [APP-**047**] contains a summary of residual effects on habitats and species after proposed mitigation measures have been implemented. Two residual adverse construction impacts are anticipated in respect of the Cable Route Corridor on hedgerows and trees where the loss of 60-142m of largely species-poor hedging due to temporary cabling operations is likely to constitute an adverse residual effect significant at a Site level in the medium term given that it would take approximately 3-5 years for the full re-establishment of re-planted hedgerows, and on ditches and watercourses where in the medium term the temporary disturbance or damage to 61-107m of ditches is likely to constitute an adverse residual effect significant at a Site level given that it would take approximately 1-3 years for the full re-establishment of reseeded/remediated ditches. For decommissioning effects see paragraphs 6.9.52 to 6.9.55 below.
- 6.9.39 In summary, the Scheme will minimise impacts on protected species and habitats in line with national and local planning policy and will provide high quality ecological habitat during the operation of the Scheme. The significant impacts identified on harvest mice (at a site level), skylark and grey partridge (at a local level), over wintering birds (at a site level) and hedgerows and trees, ditches and watercourses affected by cable route construction (at a site level) will be mitigated as far as possible through appropriate habitat provision and management. In accordance with Paragraph 5.3.8 of NPS EN-1 (2011) and paragraph 5.4.48 of NPS EN-1 (November 2023) appropriate weight should be attached to protected species; habitats and other species of principal importance for the conservation of biodiversity in decision-making. In this case, as noted by NPS EN-1 (2011) paragraph 3.2.3 and NPS EN-1 (November 2023) paragraphs 3.1.1 and 3.1.2, it will not be possible to develop the necessary large-scale solar infrastructure without some significant residual adverse impacts. These identified impacts are limited for a Scheme of this scale and have been minimised through mitigation. They are



considered to be justified by the substantial public benefits of the Scheme outlined at Section 4.

6.9.40 The Scheme is also generally in accordance with CLLP policies S14, S60, BCSDMP Policy DM9. Sturton Ward LP policy 2b and Sturton by Stow and Stow NP Policy 12 in so far as they relate to the protection of species. These local policies should be considered in the context of the nationally significant benefits that the Scheme will bring, and the likely increased level of effect that is associated with, and acceptable for, a scheme of this scale in comparison with a smaller scheme that would deliver only locally or regionally significant benefits and for which the local policies are designed to deal with.

Ancient woodlands and veteran trees

- 6.9.41 Paragraph 5.3.14 of NPS EN-1 (2011) and paragraph 5.4.53 of NPS EN-1 (November 2023) seek to protect ancient woodland and veteran trees. The latter states that the "SoS should not grant development consent for any development that would result in the loss or deterioration of any irreplaceable habitats, including ancient woodland, and ancient veteran trees unless there are wholly exceptional reasons and a suitable compensation strategy exists".
- 6.9.42 The NPPF at paragraph 186(c) also seeks to protect ancient woodland and veteran trees directing the decision maker to refuse consent for development resulting in their loss or deterioration unless there are exceptional reasons and a suitable compensation strategy exists. There are no designated ancient woodlands within the Sites and no Tree Preservation Orders. No veteran trees will be adversely impacted by the Scheme. The Scheme will also retain existing hedgerow field boundaries and will enhance hedgerows where possible. In order to mitigate against the loss of hedgerows, HDD will be conducted to minimise disruption. Whilst some loss of vegetation will be required, this loss is vastly outweighed by the additional planting and mitigation measures proposed.
- All existing hedgerows will be buffered from development as set out in the Outline Ecological Protection and Mitigation Strategy [APP-326], and these will be managed for habitat diversity as set out in the Outline Landscape and Ecological Mitigation Plan [EN010132/EX6/WB7.3_E].REP6-025]. Buffer widths will vary according to the potential value of the trees within the hedgerows to bats as possible roosts. Substantial planting of new trees and hedgerows will also be undertaken and new linear habitat linkages between isolated trees and nearby woodland will be created. Any trees subject to development impacts will be subject to further inspection and survey and all steps necessary to avoid impacts will be taken.
- 6.9.44 The Scheme would not therefore negatively impact on any areas of ancient woodland and veteran trees and so is in accordance with Paragraph 5.3.14 of NPS EN-1 (2011), paragraph 5.4.54 of NPS EN-1 (November 2023), paragraph 186(c) of the NPPF, CLLP policy S66, BCSDMP policy DM9 and DBLP Policy 41.

Biodiversity net gain



- 6.9.45 Delivering biodiversity net gain as part of development proposals is supported through recent policy and legislation. As discussed in Section 5 of this Planning Statement, the Environment Act 2021, which has been delayed several times, proposes that NSIPs should deliver 10% biodiversity net gain. The Government's 25-year plan to improve the environment published in 2018 also requires environmental net gain to be part of development.
- 6.9.46 NPS EN-1 (2011) does not state that delivering net gain is necessary, however, paragraph 5.3.4 requires applicants to demonstrate how the project has taken advantage of opportunities to conserve and enhance biodiversity and geological conservation interests. The recently updated NPPF at paragraph 186(d) expects "opportunities to improve biodiversity in and around developments should be integrated as part of their design, especially where this can secure measurable net gains for biodiversity or enhance public access to nature where this is appropriate".
- 6.9.47 NPS EN-1 (November 2023) paragraph 4.6.6 states that: "Energy NSIP proposals, whether onshore or offshore, should seek opportunities to contribute to and enhance the natural environment by providing net gains for biodiversity, or the wider environment where possible."
- 6.9.48 DBLP ST40 seeks 10% net gains in biodiversity from new developments. Other local policies seek non-specific biodiversity enhancement (CLLP S56, S60) and Sturton and Stow and Stow neighbourhood Plan Policy 7 and Policy 12, and Sturton Ward NP Policy 2b.
- 6.9.49 The Biodiversity Net Gain Report [APP-088] sets out the results of the Scheme's BNG assessment. It concludes the Scheme will result in an overall significant net gain for biodiversity, including a net gain of 86.8% for habitats (delivered through the creation of other neutral grasslands within the sites), a net gain of 54.71% for hedgerows, and a net gain of 33.25% for river units.
- 6.9.50 The Net Gains for biodiversity will be in linear, wetland and area-based habitat terms, and will be secured for the long term through the LEMP. A package of habitat and species-specific ecological enhancements will also be carried out. The predominant habitat management to be carried out within the operational Scheme will be grassland cutting, with an emphasis on the generation of a mosaic of grassland types being more diverse than the baseline habitat condition. The LEMP's habitat creation and management priorities have been in part driven by the Biodiversity Opportunities Mapping produced by Greater Lincolnshire Nature Partnership and local policies promoting the connection of Green Infrastructure and Nature Recovery Networks, such as those associated with the River Till.
- 6.9.51 The principles of biodiversity net gain (BNG) have played a fundamental part of the design development of the Scheme as set out at paragraph 6.4.23 above with significant areas identified solely for habitat creation and enhancement. In addition, the land below and between the PV arrays will be managed to enhance biodiversity.

Decommissioning



- 6.9.52 Decommissioning effects are considered in Section 9.9 of ES Chapter 9: Ecology and Biodiversity [APP-047].
- 6.9.53 This sets out that the restoration of the land back to open arable farmland is considered likely to be beneficial for some species of farmland bird, as well as for plant species associated with arable margins, but much of the biodiversity value which it is anticipated will develop in the preceding (approximately) forty years of the Scheme would be lost along with habitat for a variety of other species. ES Chapter 9: Ecology and Biodiversity [APP-047] concludes that in order to revert back to arable food production, it may be necessary to enhance the nutrient content of the soil if it has been depleted, which would likely be achieved through treatment with fertilisers, although it is believed that this is highly unlikely and an increase in soil fertility is likely to arise (see Chapter 19 of the ES, Soils and Agriculture [APP-057]). An increase in the use of pesticides and herbicides would also be expected. The decision on the farming type to be used will be made by the landowner prior to decommissioning.
- 6.9.54 Depending on the ecological value of the habitats that develop over the lifespan of the Scheme, it is realistic that certain areas of the Site may be retained due to their value for wildlife on decommissioning.
- 6.9.55 No more than twelve months prior to decommissioning commencing, the Site will be visited by an appropriately qualified ecologist to identify any ecological constraints arising from decommissioning activities. Further surveys, mitigation and/or compensatory measures may then be required in line with prevailing guidance. As a minimum, an extended Phase 1 Habitat survey (or equivalent) is considered likely to be required to identify the potential presence of protected species and important habitats.

Summary

6.9.56 Through careful and sensitive design, the Scheme will limit and mitigate any significant harm to locally, nationally and internationally designated ecology sites, important or protected habitats and species, ancient woodlands and veteran trees. No residual adverse effects are anticipated upon designated ecological sites. Significant residual adverse effects are anticipated on harvest mice and over wintering birds and on skylark and grey partridge at a site and local level respectively, whilst significant beneficial residual effects are anticipated for other species ranging from a site to district level. Temporary adverse effects on hedgerows and trees and ditches and watercourses at a site level are also anticipated during cable route construction although these features will subsequently be reestablished through replanting and seeding. Adverse effects will be mitigated as far as possible through appropriate habitat provision and management. In accordance with Paragraph 5.3.8 of NPS EN-1 (2011) and paragraph 5.4.48 of NPS EN-1 (November 2023) appropriate weight should be attached to protected species; habitats and other species of principal importance for the conservation of biodiversity in decision-making. In this case, as noted by NPS EN-1 (2011) paragraph



3.2.3 and NPS EN-1 (November 2023) paragraph 3.1.1, it will not be possible to develop the necessary large-scale solar infrastructure without some significant residual adverse impacts. The three identified adverse residual effects are limited for a scheme of this scale and have been minimised through mitigation. They are considered to be justified by the substantial public benefits of the Scheme outlined at Section 4 and should also be considered alongside the benefits to other species and biodiversity resulting from the Scheme as set out in this section.

6.9.57 The Scheme will provide biodiversity net gain in accordance with NPS EN-1 (2011), NPS EN-3 (2011), NPS EN-1 (November 2023) and NPS EN-3 (November 2023). It vastly exceeds the requirement set out in the Environment Act 2021, the NPPF and local planning policies CLLP S61 and DBLP Policy ST40 seeking 10% net gains in biodiversity. It also accords with CLLP S56 and S60, and Sturton and Stow and Stow NP Policy 7 and Policy 12 and Sturton Ward NP Policy 2b in so far as they seek biodiversity enhancements.

6.10 Water and Drainage

- 6.10.1 NPS EN-1 (2011), at section 5.7, and NPS EN-1 (November 2023), at section 5.8, set out the generic impacts and considerations associated with flood risk. Paragraph 5.7.4 of NPS EN-1 (2011) and paragraphs 5.8.13 and 5.8.14 of NPS EN-1 (November 2023) require all proposals for energy projects located in Flood Zones 2 and 3 to be accompanied by a flood risk assessment (FRA). This includes the requirement for a Flood Risk Assessment (FRA) to be submitted with the Application and guidance on what this should contain (NPS EN-1 2011 paragraph 5.7.5). NPS EN-1 (November 2023) paragraph 5.8.15 also sets out updated requirements for the contents of an FRA. The NPPF stipulates the requirement for an FRA in certain cases at paragraph 166 (noting that this is in connection with 'strategic policies').
- 6.10.2 Local planning policy CLLP S14 is supportive of renewable energy schemes provided impacts on matters including flood risk are satisfied. DBLP Policy ST35 requires all development to mitigate flood risk and surface water run-off. DBLP Policy ST52 requires a flood risk assessment to be undertaken and where relevant, proposals must demonstrate that they pass the Sequential Test and if necessary, the Exception Test. Sturton by Stow and Stow NP Policy 1 and 13 require development to be located so that flood risk is mitigated. Treswell and Cottam NP Policy 1 states that developments shall be located within areas at least risk of flooding. Proposals that are located within either flood zones 2 or 3 must apply the sequential test. Sturton Ward NP policy 4 requires that all development proposals consider and, when necessary, address the effect of the proposed development on flood risk both onsite and offsite, commensurate with the scale and impact of the development.
- 6.10.3 Flood Risk Assessments have been carried out for each of the Sites and for the Cable Route Corridor in accordance with the above policy requirements. The FRAs are included at Appendices 10.1 to 10.5 [APP-089 to APP-093] of the ES. A summary of the methodology and findings of the FRA are also presented in ES Chapter 10, Hydrology, Flood Risk and Drainage [APP-048] and ES Addendum Chapter 10:



Hydrology, Flood Risk and Drainage [REP1-073] The requirements set out in national policy for consultation with the Environment Agency have also been met by the Applicant and are detailed in the Chapter.

Sequential and Exception Test

- 6.10.4 NPS EN-1 (2011) paragraph 5.7.13 and NPS EN-1 (November 2023) paragraphs 5.8.22 and 5.8.23 explain that preference should be given to locating development in Flood Zone 1 but acknowledges that if there is no reasonably available site then projects can be located in Flood Zone 2, or if no suitable land is available in Flood Zone 2 a scheme can be located in Flood Zone 3, subject to the Exception Test.
- 6.10.5 NPS EN-1 (2011) paragraphs 5.7.9, 5.7.12 and 5.7.13 and NPS EN-1 (November 2023) paragraphs 5.8.36 and 5.8.22 to 5.8.23; NPPF paragraph 162 states that "The sequential approach should be used in areas known to be at risk now or in the future from any form of flooding."
- 6.10.6 NPPF paragraph 165 states that "inappropriate development in areas at risk of flooding should be avoided and that development should be directed away from areas at highest risk. Where development is necessary in areas of flood risk, the development should be made safe for its lifetime without increasing flood risk elsewhere."
- 6.10.7 The Sites are predominantly within Environment Agency Flood Zone 1. However, the north-west corner and eastern edge of West Burton 1 are in flood zone 3 and flood zone 2 extends around the eastern and southern edges. The eastern part of West Burton 2 is located in flood zone 3 and a strip of land along the western side is in flood zone 2. A central band and a thin strip along the western edge of West Burton 3 are located in flood zones 2 and 3. These areas are located at the periphery of the Sites or cross parts of fields. The Flood Zone 3 area within the Sites equates to 28.95% of the total site area. In accordance with paragraph 5.7.23 of NPS EN-1 (2011) and paragraph 5.8.29 of NPS EN-1 (November 2023), the Applicant has, therefore, applied the sequential test.
- 6.10.8 ES Appendix 10.6: Flood Risk Assessment Sequential and Exception Test [APP-094] sets out how the Scheme satisfies the requirements and purpose of the Sequential Test in accordance with NPS EN-1 (2011) paragraph 5.7.13 and NPS EN-1 (November 2023) paragraphs 5.8.22 and 5.8.23 It demonstrates that there are no available sequentially preferable sites to the Scheme that are of sufficient size to accommodate the Scheme. Given the above it is considered that the proposals pass the Sequential Test.
- 6.10.9 Paragraph 5.7.16 of NPS EN-1 (2011), paragraph 5.8.11 of NPS EN-1 (November 2023) and paragraph 170 of the NPPF set out similar but slightly differing criteria that are required for the Exception Test to be passed, which is a requirement given that a proportion of the site is located within Flood Zone 3.
- 6.10.10 NPS EN-1 (2011) sets out (at paragraph 5.7.16) the following three elements of the Exception Test:



- 1. it must be demonstrated that the project provides wider sustainability benefits to the community that outweigh flood risk;
- 2. the project should be on developable, previously developed land or, if it is not on previously developed land, that there are no reasonable alternative sites on developable previously developed land subject to any exceptions set out in the technology specific NPSs; and
- 3. a FRA must demonstrate that the project will be safe, without increasing flood risk elsewhere subject to the exception below and, where possible, will reduce flood risk overall.
- 6.10.11 NPS EN-1 (November 2023) represents the most recently drafted emerging policy. It requires the following criteria to be met (paragraph 5.8.11):
 - 1. the project provides wider sustainability benefits to the community that outweigh flood risk, and
 - 2. the project reduces flood risk overall, where possible.
- 6.10.12 Footnotes 116 of NPS EN-1 (2011) and 217 of NPS EN-1 (November 2023) note that sustainability benefits to the community would include the benefits (including need), for the infrastructure.
- 6.10.13 The NPPF represents the most recent designated policy, and sets out that for the Exception Test to be passed, it should be demonstrated that the following criteria are be met (paragraph 170):
 - 1. the development would provide wider sustainability benefits to the community that outweigh the flood risk; and
 - 2. the development will be safe for its lifetime taking account of the vulnerability of its users, without increasing flood risk elsewhere, and, where possible, will reduce flood risk overall.
- 6.10.14 The majority of the Order limits lie within Flood Zone 1 and 2. The Exception Test is therefore applied because parts of the main Cable Route (which will contain a below ground cable only) and parts of the Sites lie within Flood Zone 3.
- 6.10.15 The Scheme passes the Exception Test, as set out by NPS EN-1 (2011), NPS EN-1 (November 2023) and the NPPF by virtue of the following:
 - 1. It demonstrably provides wider sustainability benefits to the community which outweigh the low flood risk to and from the Scheme. These are in the form of the delivery of a large amount of renewable energy generation capacity that is urgently needed to help meet national energy and climate change objectives and commitments, as explained by the Statement of Need [APP-320]. The significant public benefits of the Scheme are set out at Section 4.0 of the Planning Statement. This also sets out the other sustainability benefits of the Scheme including biodiversity net gain, and improved connectivity across the Order limits via a new permissive path.



- 2. The Scheme cannot be developed on previously developed land as the Site Selection Assessment **[AS-004]** has demonstrated that there are no reasonable alternative sites on developable previously developed land.
- 3. The FRA and Drainage Strategy Appendix 10.1 [APP-089] concludes that the Scheme remains safe for its lifetime and does not increase flood risk elsewhere. It explains that the solar panels will be mounted on raised frames above surrounding ground level allowing flood water to flow freely underneath. Therefore, there will be no loss of floodplain volume as a result of the Scheme. The Scheme is free draining through perimeter gaps around all panels, allowing for infiltration as existing within the grassland/vegetation surrounding and beneath the panels. There will be minimal increase in impermeable area meaning the proposals will not increase surface water flood risk elsewhere.
- 4. The battery storage and substation infrastructure have been sequentially located outside of the flood zone extents.
- 5. Where conversion units have been proposed within the Sites, it has been recommended that the structures are sequentially located outside of the 1% Annual Exceedance Probability (AEP) + Climate Change (CC) extent and/or the 0.1% Annual Probability Surface Water proxy extent. Where this is not possible, the conversion units will be raised 600 mm above the design flood level and designed to be flood resilient in line with best practice guidance. Associated infrastructure will also be designed to be flood resilient.
- 6. The FRA and Drainage Strategy Appendix 10.1 [APP-089] states at paragraph 6.2.5: "this Flood Risk Assessment demonstrates that the Site will not increase flood risk elsewhere and the ground beneath the panels will remain entirely permeable, draining as existing. The development may reduce existing greenfield rates by replacing intensive agricultural surfaces with a landcover comprising a mixture of wildflowers and grassland."
- 7. The FRA and Drainage Strategy Appendix 10.1 **[APP-089**] also states at paragraph 5.3.8: "Based on the above, the proposed development is likely to provide betterment over the existing surface water runoff regime".
- 6.10.16 The above demonstrates that all the requirements of both the sequential test and Exception Test are satisfied in accordance with the requirements of NPS EN-1 (2011), NPS EN-1 (November 2023), the NPPF and DBLP Policy ST52.

Managing flood risk to and from the Scheme

6.10.17 NPS EN-1 (2011) paragraphs 5.7.24 and 5.7.25 state that "Essential energy infrastructure which has to be located in flood risk areas should be designed to remain operational when floods occur" and that "the receipt of and response to warnings of floods is an essential element in the management of the residual risk of flooding". NPS EN-1 (2011) paragraph 5.7.9, and NPS EN-1 (November 2023) paragraph 5.8.36 set out the matters that the SoS should be satisfied of in decision making. These include



that projects should be appropriately flood resilient and safe during their lifetime. NPS EN-5 (2011) paragraph 2.4.1 and NPS EN-5 (November 2023) paragraph 2.6.1 also expects electricity infrastructure such as substations to be resilient to flooding.

- 6.10.18 Paragraph 165 of the NPPF expects development to not increase flood risk elsewhere and stipulates various requirements for development to meet in flood risk zones. These requirements include locating the most vulnerable development in areas of lowest flood risk, unless there are overriding reasons to prefer a different location; ensuring development is appropriately flood resistant and resilient; any residual risk can be safely managed, and safe access and escape can be provided. BLP policies DM12, ST52 and ST53 require developments to be safe from flooding and to not worsen flood risk elsewhere.
- 6.10.19 The FRA and Drainage Strategy, **ES Appendix 10.1 [APP-089]** considers flood risk to and from the Sites and contains the following summary of flood risk for the Scheme at paragraph 3.1.3 Table 3:

Site	Summary of Flood Risk
Cable Route	The risk to the Site from all sources of flooding is Negligible to Low .
West Burton 1	The risk to the Site from all sources of flooding is Negligible to Low .
West Burton 2	The risk to the Site from all sources of flooding is Negligible to Low .
West Burton 3	The risk to the Site from all sources of flooding is Negligible to Low .

- 6.10.20 Based on the assessed flood risk the following embedded design has been implemented:
 - 8m easements have been established around all watercourses, including Main Rivers and Ordinary Watercourses and 9 m from IDB assets.
 - Substations and energy storage compounds) have been sequentially located within Zone 1, an area with a "Low probability of flooding" and therefore in land assessed as having a less than 1 in 1,000 annual probability of river or sea flooding (<0.1%).
 - Where possible conversion units have been located in parts of the Site that are
 within Flood Zone 1. Where this hasn't been possible the conversion units will
 be raised 600mm above the 0.1% AEP flood level or where this is not possible
 as high as practicable.
 - Non-flood sensitive infrastructure forming the wider Scheme (PV arrays and cabling) has been sequentially located outside the 1 in 100 plus climate change annual probability extent (1% +CC) or where this is not possible restricted to areas which experience less than 1 m depth of flooding during the same event.
 - Flexibility for either tracker or fixed panels have been built into the EIA. Foundations are most likely to be galvanised steel poles driven into the



ground. These will either be piles rammed into a pre-drilled hole, or a pillar attaching to a steel ground screw.

- For both fixed and tracker panels all sensitive and electrical equipment on the solar panel will be elevated by the legs so that it is no less than 0.6 m above the surrounding peak flood level.
- Tracker panel units will be mounted on raised frames (usually raised a minimum of 0.4m when on maximum rotation angle) and will therefore, be raised above surrounding ground levels and fitted with a tracking system. During times of flooding, solar panels may be stowed by the tracking system algorithm onto a horizontal plane, to the minimum post height of 2.3 m above ground level. This ensures that all sensitive and electrical equipment on the solar panel is raised to a minimum of 2.3 m above ground level in the horizontal position.
- The design of the Scheme has ensured that the flood defences protecting the Scheme can be inspected and maintained by the operator of the Scheme to ensure their functionality throughout the lifetime of the Scheme.
- 6.10.21 **ES Chapter 10, Hydrology, Flood Risk and Drainage [APP-048]** and ES Addendum Chapter 10: Hydrology, Flood Risk and Drainage **[REP1-073]** states that from a flood risk perspective, the potential significant effects include mud and debris blockages and temporary increases in impermeable areas during the construction phase and the increase in permanent impermeable area and increase in discharge to local watercourses and blockages of drainage networks during the operational phase.
- 6.10.22 Proposed mitigation is set out in the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] and final versions of these documents prior to construction and decommissioning respectively will include detailed mitigation measures to prevent adverse impacts occurring to controlled waters and simple SuDS measures to mitigate the surface water flood risk.
- 6.10.23 Inclusion of permeable surfacing for the Site access, linear infiltration trenches around any proposed infrastructure (substations and batteries) and wildflower planting at the leeward edge of solar panels should in general provide sufficient treatment as well as the attenuation required to maintain existing runoff rates.
- 6.10.24 ES Chapter 10, Hydrology, Flood Risk and Drainage [APP-048] and ES Addendum Chapter 10: Hydrology, Flood Risk and Drainage [REP1-073] and FRA and Drainage Strategy, ES Appendix 10.1 [APP-089] assess that the Scheme is acceptable with the mitigation measures identified which would ensure there would be no significant flood risk effects.

The above measures ensure that the Scheme minimises flood risk to essential energy infrastructure as required by NPS EN-1 (2011) paragraphs 5.7.24 and 5.7.25 and ensures that it will be flood resilient in accordance with NPS EN-1 (2011) paragraph 5.7.9, NPS EN-1 (November 2023) paragraph 5.8.36, and NPS EN-5 (2011)



paragraph 2.4. As no staff will be based on the site, provision of safe access and escape as required by NPPF paragraph 173 is not relevant to the Scheme. The Scheme also complies with CLLP policy S14, DBLP Policy ST35 and ST52, Sturton Ward NP policy 4 and Sturton by Stow and Stow NP Policy 1 and 13 as it has been demonstrated, with the implementation of mitigation measures, to be safe from flooding and to not worsen flood risk elsewhere.

Drainage

- 6.10.25 NPS EN-1 (2011) paragraph 5.7.19 explains the range of sustainable approaches to surface water drainage management and paragraph 5.7.21 requires "surface water drainage arrangements for any project to be such that the volumes and peak flow rates of surface water leaving the site are no greater than the rates prior to the proposed project, unless specific off-site arrangements are made and result in the same net effect". Paragraph 5.7.22 also states that it "may be necessary to provide surface water storage and infiltration to limit and reduce both the peak rate of discharge from the site and the total volume discharged from the site. There may be circumstances where it is appropriate for infiltration facilities or attenuation storage to be provided outside the project site, if necessary, through the use of a planning obligation".
- 6.10.26 NPPF paragraphs 175 states that SuDS should be incorporated into major developments which should also take account of Lead Local Flood Authority (LLFA) advice; have appropriate proposed minimum operational standards; provide multifunctional benefits; and be able to be maintained to an acceptable standard for the operational life of the development. CLLP policy S21, DBLP policy ST6 and Sturton Ward NP policy 2b and policy 4 also require incorporation of SuDS.
- 6.10.27 A Drainage Strategy for the Sites is contained at FRA and Drainage Strategy Appendix 10.1 **[APP-089].** It proposes an onsite drainage strategy in line with NPS EN-1 (2011), NPPF policy and local planning policy. In summary this includes:
 - The Scheme will be free draining through perimeter gaps around all panels, meaning the proposals will not increase surface water risk elsewhere.
 - In order to mitigate against potential erosion from rainwater dripping onto the ground from the solar panels, the existing intensively managed agricultural land will be replaced by planted wildflower and grassland below the solar panels which will help prevent erosion.
 - The panels forming the solar array will not be tightly compacted which will allow water to drip onto the ground below from several locations rather than as concentrated runoff which will reduce the potential for erosion to occur.
 - The access track will be designed to be permeable, thereby allowing surface water runoff to percolate into the ground below. This is in accordance with the SuDS principles set out at NPS EN-1 (2011) paragraph 5.7.19.
 - Electrical infrastructure associated with the panels will be sited on concrete pads. surrounded by gravel filled filter trenches, constructed to limit the lateral



flow of water and replace the loss of natural infiltration caused by the concrete bases themselves. Surface water would be stored within the gravel sub-base prior to infiltrating into the ground as per the existing situation. This is in accordance with the SUDS principles set out at NPS EN-1 (2011) paragraph 5.7.19.

- 6.10.28 Based on the above, the Scheme is concluded likely to provide betterment over the existing surface water runoff regime.
- 6.10.29 In addition, the Drainage Strategy recommends that the movement of large vehicles is limited to proposed access tracks in order to reduce the potential for soil compaction to occur. Vehicles should be fitted with low pressure tyres to further reduce the impact on the underlying soil.
- 6.10.30 The aforementioned techniques employ SuDS principles in accordance with NPS EN1 (2011) paragraph 5.7.19, NPPF paragraph 175 and local planning policies and will
 discourage soil erosion within the site, whilst maintaining the existing overland flow
 paths. The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] sets out the basic
 principles ensuring soil compaction by large construction vehicles is minimised and
 will be secured through the detailed CEMP.
- 6.10.31 In summary, the design of the Scheme accords with NPS EN-1 (2011), NPS EN-1 (November 2023), the NPPF and local planning policies with regards to drainage because it achieves the required runoff rates using sustainable drainage methods and does not increase flood risk elsewhere.

Water quality and resources

- 6.10.32 NPS EN-1 (2011), paragraphs 5.15.2 and 5.15.3, and NPS EN-1 (November 2023) paragraphs 5.16.3 and 5.16.7 require applicants to undertake an assessment of the likely effects of energy NSIPs on the water environment with specific focus on the impacts upon "water quality, water resources and physical characteristics of the water environment" as well as "any impacts of the proposed project on water bodies or protected areas under the Water Framework Directive and source protection zones (SPZs) around potable groundwater abstractions". Paragraph 5.15.5 of NPS EN-1 (2011) and paragraph 5.16.12 of NPS EN-1 (November 2023) direct the SoS to give more weight to adverse effects of projects on achieving Water Framework Directive (WFD) objectives and paragraph 5.16.14 expects projects to have had regard to River Basin Management Plans (RBMP).
- 6.10.33 Paragraph 180(e) of the NPPF expects developments to not result in unacceptable levels of water pollution and wherever possible improve water quality, taking into consideration river basin management plans.
- 6.10.34 ES Chapter 10, Hydrology, Flood Risk and Drainage [APP-048] and ES Addendum Chapter 10: Hydrology, Flood Risk and Drainage [REP1-073] presents the existing status of the water environment and the likely effects of the Scheme upon it. It concludes that with appropriate mitigation there are likely to be no significant



adverse effects on the water environment following the implementation of the Scheme.

By protecting water quality, water resources and the physical water environment, and by remaining consistent with WFD objectives, the Scheme is compliant with NPS EN-1 (2011), paragraphs 5.15.2, 5.15.3, and 5.15.5, NPS EN-1 (November 2023) paragraphs 5.16.3, 5.16.7, and 5.16.14, Paragraph 180(e) of the NPPF.

Construction and decommissioning

- 6.10.35 The sections above deal primarily with the operational phase. The following paragraphs outline how water and drainage matters will be managed during construction and decommissioning and comply with NPS EN-1 (2011) paragraph 5.7.10 and NPS EN-1 (November 2023) paragraph 5.8.37. These set out that drainage during the construction stage of projects should comply with national standards published by Ministers under Paragraph 5(1) of Schedule 3 to the Flood and Water Management Act 2010.
- 6.10.36 The proposed drainage strategy within the FRA and Drainage Strategy Appendix 10.1 **[APP-089]** has been prepared in accordance with the Flood and Water Management Act 2010. It sets out measures at Section 5.3 which will be employed to ensure that greenfield runoff rates are maintained during the construction phase of the Scheme.
- 6.10.37 In addition, the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and the Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] set out mitigation and management measures to be employed during the construction and decommissioning phases of the Scheme to manage and mitigate impacts related to flood risk, drainage, groundwater and surface water during the construction and decommissioning phases of the Scheme. These include that the Applicant will comply with relevant Guidance for Pollution Prevention (GPP) and a Water Management Plan (WMP) which would be prepared in support of the detailed CEMP. The WMP will include details of pre-, during and post-construction water quality monitoring. This will be based on a combination of visual observations and reviews of the Environment Agency's automatic water quality monitoring network. The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and the Outline Decommissioning Statement [EN010132/EX6/WB7.2 BREP6-023] also include commitments that the detailed CEMP and detailed DEMP will include methods for the safe storage of materials, plans to deal with accidental pollution and spills, pollution control measures incorporated into construction and decommissioning phase drainage, and flood risk.
- Taking account of the construction and decommissioning stage mitigation set out in the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and the Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023], ES Chapter 10, Hydrology, Flood Risk and Drainage [APP-048] and ES Addendum Chapter 10: Hydrology, Flood Risk and Drainage [REP1-073] does not identify any significant residual effects on the water environment or flood risk during construction or decommissioning of the Scheme. The Scheme is therefore policy compliant in this



regard, including with NPS EN-1 (2011) paragraph 5.7.10 and NPS EN-1 (November 2023) paragraph 5.8.37.

Summary

- 6.10.39 The Scheme has been demonstrated to meet the requirements of relevant national and local planning policies pertaining to water and drainage because:
 - Flood Risk Assessments have been carried out for each of the Sites and for the Cable Route Corridor in accordance with Paragraph 5.7.4 of NPS EN-1 (2011), paragraph 5.8.13 of NPS EN-1 (November 2023) and DBLP Policy ST52;
 - ES Appendix 10.6: Flood Risk Assessment Sequential Test [APP-094] sets out how the Scheme satisfies the requirements and purpose of the Sequential Test in accordance with NPS EN-1 (2011) paragraph 5.7.13, NPS EN-1 (November 2023) paragraphs 5.8.22 and 5.8.23 and DBLP Policy ST52;
 - The Scheme passes the Exceptions Test as it has been demonstrated to provide wider sustainability benefits in the form of significant renewable energy generation and meeting climate change objectives which outweigh the low flood risk to and from the Scheme, it will not increase flood risk elsewhere, has been shown likely to provide betterment over the existing surface water runoff regime and it has been demonstrated that there are no suitable alternative brownfield sites available on which to locate it. It therefore complies with Paragraph 5.7.16 of NPS EN-1 (2011), paragraph 5.8.11 of NPS EN-1 (November 2023), paragraph 170 of the NPPF and DBLP Policy ST52;
 - It has been demonstrated to be safe from flooding and to not worsen flood risk elsewhere as required by CLLP policy S14, DBLP Policy ST35 and ST52, Sturton Ward NP policy 4 and Sturton by Stow and Stow NP Policy 1 and 13;
 - It minimises flood risk to essential energy infrastructure as required by NPS EN-1 (2011) paragraphs 5.7.24 and 5.7.25;
 - It ensures that it will be flood resilient in accordance with NPS EN-1 (2011) paragraph 5.7.9, NPS EN-1 (November 2023) paragraph 5.8.11, and NPS EN-5 (2011) paragraph 2.4;
 - No staff will be based on the site, so provision of safe access and escape as required by NPPF paragraph 167 is not relevant to the Scheme; and,
 - It employs SuDS principles in accordance with NPS EN-1 (2011) paragraph 5.7.19, NPPF paragraph 173 and local planning policies CLLP policy S21, DBLP policy ST6 and Sturton Ward NP policy 2b and policy 4.

6.11 Noise and Vibration

6.11.1 Elements of the Scheme, primarily the BESS and inverters will generate noise. ES Chapter 15, Noise and Vibration [APP-053] provides a noise and vibration assessment. The layout of the Scheme has been carefully designed to mitigate and minimise noise impacts on sensitive receptors, such as residential properties, and



acoustic barriers are embedded into the design of the Scheme. The noise impacts of the Scheme have been assessed and no significant impacts have been identified, as set out in detail in the ES Chapter 15, Section 7. In accordance with planning policy, as set out below, the assessment has concluded that no significant impacts on health and quality of life from noise will result from the Scheme, and that the minor impacts will be mitigated and minimised.

- 6.11.2 NPS EN-1 (2011) paragraph 5.11.4, and NPS EN-1 (November 2023) paragraph 5.12.6 require a noise assessment to be prepared where noise and vibration impacts are likely to arise and sets out the methodology for this assessment. NPS EN-3 (November 2023) paragraphs 2.10.120 to 2.10.126 set out that the noise and vibration impact of construction traffic should also be considered. NPS EN-1 (2011) paragraph 5.11.6, and NPS EN-1 (November 2023) paragraph 5.12.9 add that for operational noise with respect to human receptors should be assessed using the principles of the relevant British Standards and other guidance.
- 6.11.3 NPS EN-1 (2011) paragraph 5.11.9, and NPS EN-1 (November 2023) paragraph 5.12.17 state that the decision maker should not grant development consent unless it is satisfied that the proposals will meet the following aims:
 - 1. avoid significant adverse impacts on health and quality of life from noise;
 - 2. mitigate and minimise other adverse impacts on health and quality of life from noise; and
 - 3. where possible, contribute to improvements to health and quality of life through the effective management and control of noise.
- 6.11.4 Part (e) of NPPF paragraph 180 outlines that planning decisions should prevent "new and existing development from contributing to, being put at unacceptable risk from, or being adversely affected by, unacceptable levels of…noise pollution". At paragraph 185 part (a) it also states that decisions should "mitigate and reduce to a minimum potential adverse impact resulting from noise from new development and avoid noise giving rise to significant adverse impacts on health and the quality of life".
- 6.11.5 BCSDMP Policy DM4 also seeks to prevent new development from causing unacceptable impacts on residential amenity of nearby residents.
- 6.11.6 ES Chapter 15, Noise and Vibration [APP-053] has assessed the noise and vibration impacts of the Scheme through a combination of consultation, background noise survey and computer modelling.
- 6.11.7 The assessment has included consideration of:
 - Noise and vibration from construction activities on sensitive receptors;
 - Noise and vibration from construction traffic on sensitive receptors;
 - Operational noise on sensitive receptors.



- 6.11.8 To inform the assessment of operational noise, background noise monitoring was carried out at a large number of locations representing the nearest sensitive existing receptors surrounding the potential development areas.
- 6.11.9 The noise emissions of plant associated with the Scheme, including the solar PV arrays, energy storage and electrical substations have been predicted at the nearest sensitive receptors.
- 6.11.10 Advice has been sought from the relevant Local Planning Authorities on the appropriateness of the methodology adopted to assess operational noise, but, as of the date of submitting this DCO application, a response has not been received. It is considered that the approach described above represents a reflection of industry best practice in such circumstances where existing background levels surrounding a development of this kind are very low.
- 6.11.11 When the predicted noise levels are compared against the existing background noise levels at most of the sensitive receptors, the assessment results in significant adverse effects at the receptors, depending on the context. However, the existing measured background noise levels at these receptors, particularly during the night-time period are considered to be very low. For very low existing background noise levels, the guidance that would usually be considered for a development of this nature BS 4142³ contains a clause that states that alternative guidance WHO/BS 8233 ⁴ and IEMA⁵ guidance should be considered and used to inform the assessment.
- 6.11.12 The alternative guidance sets noise limits which should not be exceeded internally at each nearby sensitive receptor due to noise emissions from the proposed Scheme. When assessed against these criteria and including recommended mitigation measures, noise emissions during the operational phase do not result in significant impacts at any sensitive receptors.
- 6.11.13 A further assessment of operational noise has been utilised to assess the impact of noise emissions from the proposed Scheme which considers the likely change in noise level due to the contribution of noise emissions from the development at each receptor. When the predicted contribution of noise from the proposed development is combined with the existing noise climate at each receptor, the change in noise level is considered to be below the threshold of 'unlikely to be perceptible' and therefore insignificant.

³ Operational Noise from the solar farms – BS 4142:2014+A1:2019 Method for rating and assessing industrial and commercial sound, British Standards Institute (2014, with amendments), Bsi, London

⁴ World Health Organization (WHO) Guidelines for Community Noise (1999) and Operational Noise from solar farms (Alternative) – BS 8233:2014 Guidance on sound insulation and noise reduction for buildings, British Standards Institute (2014), Bsi, London

⁵ Operational Noise from solar farms (Alternative) – IEMA 'Guidelines for Environmental Noise Impact Assessment' (2014);



- 6.11.14 NPS EN-1 (2011) and NPS EN-3 (2011) also expect energy NSIPs to demonstrate good design with regard to mitigating noise impacts. Specifically, NPS EN-1 (2011) paragraph 5.11.8 expects projects to "demonstrate good design through selection of the quietest cost-effective plant available; containment of noise within buildings wherever possible; optimisation of plant layout to minimise noise emissions; and, where possible, the use of landscaping, bunds or noise barriers to reduce noise transmission." NPS EN-1 (November 2023) at paragraph 5.12.15 contains the same policy.
- 6.11.15 NPS EN-1 (2011) paragraph 5.11.12 and NPS EN-1 (November 2023) paragraph 5.12.14 suggest that mitigation measures may include solutions related to engineering, layout and administration (i.e restricting activities or setting noise limits).
- 6.11.16 Embedded noise mitigation measures comprising acoustic louvres around inverters are proposed in identified locations and secured through the Concept Design Parameters [REP5-094] for the Scheme.
- 6.11.17 In summary, the Scheme accords with NPS EN-1 (2011) and NPS EN-1 (November 2023), specifically the three policy aims of paragraph 5.11.9 (and 5.12.17 in NPS EN-1 November 2023); the NPPF and DBLP policy DM4 by avoiding significant noise and vibration impacts on health and quality of life; minimising adverse impacts of noise and vibration through appropriate mitigation; and providing additional mitigation through the design and selection of operational plant to effectively manage and control operational noise.

Construction and decommissioning

- 6.11.18 ES Chapter 15, Noise and Vibration [APP-053] includes an assessment of construction noise and vibration generated by the Scheme in terms of traffic and the use of plant and heavy ground works such as piling. For decommissioning the assessment assumes the same effects as construction.
- 6.11.19 Vibration effects during construction activities are below the assessment criteria for the sensitive receptors and no significant effects are assessed.
- 6.11.20 Noise and vibration during peak periods of construction traffic is assessed as, at most, minor effect and no significant effects are assessed.
- 6.11.21 Noise and vibration levels during construction have been predicted at the nearest sensitive receptor locations. The predictions for construction noise along the cable route are marginally above the threshold criteria when undertaken at the closest point at which they take place at two of the assessed receptors. However, given that the construction activities for the cable route are transient, it is considered unlikely that a major impact would be experienced for any prolonged duration due to the temporary nature of construction operations. As such, the effect of construction noise on sensitive receptors is not significant. All other construction activities are predicted to be below the threshold criteria.



- 6.11.22 Mitigation is required in order for effects to be not significant and has been included as embedded mitigation as set out above. No additional mitigation has been specified within ES Chapter 15, Noise and Vibration [APP-053].
- 6.11.23 In terms of the residual effects of the Scheme, the construction noise levels at all receptors are predicted to be within the 65 dB(A) noise level limit. The construction noise is also temporary, and the assessment assumes that all construction activities will be happening simultaneously across the Site as this is considered worst-case. Construction activity on Site would likely be experienced by limited receptors at any given time as work progresses across the Proposed Development. The residual effects are not assessed as significant.
- 6.11.24 Best Practicable Means (BPM) to minimise noise during the construction and decommissioning phases are included within the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023].
- 6.11.25 Furthermore, a Statutory Nuisance Statement [APP-317] has been prepared which has considered matters of general site condition, waste, air quality, artificial light, glint and glare, noise and vibration, and concludes that the Scheme is not envisaged to give rise to significant effects that would result in a statutory nuisance.
- 6.11.26 The construction and decommissioning phases of the Scheme comply with the first two objectives of NPS EN-1 (2011) paragraph 5.11.9 and NPS EN-1 (November 2023) paragraph 5.12.17.

6.12 Glint and Glare

- 6.12.1 NPS EN-3 (November 2023) paragraph 2.10.103 states that in some instances, it may be necessary to seek a glint and glare assessment as part of the application. This may need to account for 'tracking' panels if they are proposed as these may cause differential diurnal and/or seasonal impacts. Paragraph 2.10.158 sets out that solar PV panels are designed to absorb, not reflect, irradiation. However, the Secretary of State should assess the potential impact of glint and glare on nearby homes and motorists.
- 6.12.2 NPS EN-3 (November 2023) paragraph 2.10.159 also states: "Whilst there is some evidence that glint and glare from solar farms can be experienced by pilots and air traffic controllers in certain conditions, there is no evidence that glint and glare from solar farms results in significant impairment on aircraft safety. Therefore, unless a significant impairment can be demonstrated, the Secretary of State is unlikely to give any more than limited weight to claims of aviation interference because of glint and glare from solar farms".
- 6.12.3 CLLP policy S14 also requires consideration to be given to impacts upon aviation and defence navigation and rail safety. CLLP policy LP26, CLLP policy S14, BDCSDMP policy DM4 and DBLP Policy 48 also seeks to protect residential and visual amenity.



- 6.12.4 BDCSDMP Policy DM10 is relevant to all types of renewable energy proposal and requires shadow flicker to be considered. This is considered to be relevant to wind farm proposals where the blades can cause flicker as they rotate but is not considered to be relevant to the Scheme as the panels do not cause flicker. Glint and glare matters associated with the tilting of tracker panels are however considered within ES Chapter 16: Glint and Glare [APP-054].
- 6.12.5 ES Chapter 16: Glint and Glare [APP-054] considers the glint and glare impacts of the Scheme. The Scheme is located in a rural area and the review of available imagery shows no presence of other solar farms of a similar size or large reflective surfaces (such as bodies of water).
- 6.12.6 The most reflective and visible components of solar development is the upper surface of the solar panel. Although the Glint and Glare chapter concludes that while the panels' frames and structures can also be a source of glare, it is unlikely that they will be visible, and their totally reflective surface is much smaller when compared to the total panel area.
- 6.12.7 Other components such as the substation or energy storage are not a source of solar reflections due to their lack of reflective materials, and the cables that export the electricity generated by the Scheme are buried underground and therefore do not require to be considered in the Glint and Glare Assessment.
- 6.12.8 Taking all factors into account, the glint and glare effects can occur from any solar panels installed at the Scheme's Sites, although as not all panels will be deployed during the construction or decommissioning phase, the length and intensity of any solar reflections will be less than or equal to the operational phase. The Assessment therefore only considered the Operational Effects, which represents the worst-case scenario.
- 6.12.9 Following the findings of the initial impact assessment, a series of embedded mitigations measures have been incorporated to reduce the impacts of the Scheme to acceptable levels. These embedded mitigation options involve screening in the form of vegetation, or instant screening for ground base receptors if necessary. For a tracking system an additional mitigation option, if required, is the change in backtracking angle which can be modified to project solar reflections away from receptors.
- 6.12.10 ES Chapter 16: Glint and Glare [APP-054] considered varying sensitivity receptors ranging from 'low sensitivity' on local roads (because traffic volumes are predicted to be low), to 'medium sensitivity' on regional, national, and major roads (with higher levels of traffic), dwellings, railways, and aviation-related receptors.
- 6.12.11 For dwelling and road receptors where a Moderate Adverse impact is predicted, the developer has proposed screening in the form of vegetation (and opaque fencing, if necessary, as an interim measure, while vegetation grows to a sufficient height to be effective). For railway receptors, where no significant impact is predicted towards a train driver, the developer has proposed screening along the boundary of the



proposed development to further obstruct views of the reflecting panel area in the even any trackside vegetation is removed. The provision of opaque fencing is included in the Outline Operational Environment Management Plan [REP5-020] which is secured by a requirement in the draft DCO. These mitigation measures will be effective irrespective of the type of panel mounting system used. Once the mitigation is in place, the impacts will not be significant in EIA terms. The screening is shown on the Landscape and Ecology Mitigation and Enhancement Plans [REP1-026 to REP1-031].

6.12.12 ES Chapter 16: Glint and Glare [APP-054] also considered the cumulative effects of the glint and glare arising from other nearby schemes including Cottam Solar Project and Gate Burton Energy Park. It concluded that shared receptors are either unlikely to concurrently have visibility of multiple areas or, if visibility is possible, no significant impact is predicted due to the presence of existing and proposed screening.

<u>Summary</u>

6.12.13 The glint and glare impacts of the Scheme have been shown not to be significant in EIA terms. There will be no unacceptable impact on the operation of aircraft movement or operational radar and no adverse impacts upon railway safety or residential amenity as a result of glint and glare once the proposed mitigation is in place. The Scheme is therefore considered to meet the requirements of NPS EN-3 (November 2023) paragraph 2.10.103, BDCSDMP DM4, DM10, DCLLP S14 and DBLP Policy 48.

6.13 Transport and Access

- 6.13.1 Paragraph 5.13.2 of NPS EN-1 (2011) and Paragraph 5.14.4 of NPS EN-1 (November 2023) states that "the consideration and mitigation of transport impacts is an essential part of Government's wider policy objectives for sustainable development". Paragraph 5.13.3 of NPS EN-1 (2011) states that "if a project is likely to have significant transport implications, the applicant's ES should include a transport assessment". Paragraph 5.13.6 of NPS EN-1 (2011) states that "A new energy NSIP may give rise to substantial impacts on the surrounding transport infrastructure and [the Secretary of State] should therefore ensure that the applicant has sought to mitigate these impacts, including during the construction phase of the development".
- 6.13.2 Section 5.13 of NPS EN-1 (2011) and section 5.14 of NPS EN-1 (November 2023) discuss the requirements for considering the potential transport and traffic related impacts and mitigation of NSIPs. NPS EN-3 (November 2023) Paragraph 2.10.125 states an assessment of whether the access roads are suitable for the transportation of components to the site should be undertaken. This will include whether they are sufficiently wide for the proposed vehicles, or bridges sufficiently strong for the heavier components to be transported to the site. Paragraph 2.10.126 states that where a cumulative impact is likely, then a cumulative transport



- assessment should form part of the ES. This is also an expectation of local planning policy as set out at paragraphs 6.12.8 and 6.12.9 below.
- 6.13.3 With regard to access, NPS EN-1 (2011) and NPS EN-1 (November 2023) expect developments to include "proposed measures to improve access by public transport, walking and cycling, to reduce the need for parking associated with the proposal and to mitigate transport impacts" (paragraphs 5.13.4 and 5.14.7, respectively).
- 6.13.4 NPS EN-1 (2011) paragraph 5.13.10 and NPS EN-1 (November 2023) paragraph 5.14.16 also require applicants to consider the use of water-borne or rail transport over road transport at all stages of the project, where cost-effective.
- 6.13.5 Paragraph 2.10.35 of NPS EN-3 (November 2023) sets out that solar NSIP developments should consider the suitability of potential access routes, since solar farms are often located in rural areas. The NPPF, at paragraph 108, also expects consideration and mitigation of transport impacts of development including the environmental impacts and impacts on transport networks. At paragraph 115, the NPPF also expects development to only be "prevented or refused on highways grounds if there would be an unacceptable impact on highway safety, or the residual cumulative impacts on the road network would be severe".
- 6.13.6 Paragraph 2.10.39 of NPS EN-3 (November 2023) sets out that access routes to a solar farm site should be able to accommodate traffic required for its construction and that the effects of traffic should be assessed.
- 6.13.7 Paragraphs 2.10.40 to 2.10.45 of NPS EN-3 (November 2023) encourages applicants to design the layout and appearance of their site to enable continued recreational use of public rights of way (PRoW), during operation and (where possible) during construction. It also notes that sites may provide the opportunity to facilitate enhancements to the local footpath network and the adoption of new public rights of way through site layout and design of access. Paragraph 2.10.45 of NPS EN-3 (November 2023) sets out that an Outline PRoW Management Plan should be provided.
- 6.13.8 Policy ST51 of the DBLP states that, "Development that generates, shares, transmits and/or stores renewable and low carbon energy, including community energy schemes, will be supported subject to the provision of details of expected power generation based upon yield or local self-consumption of electricity and by demonstrating the satisfactory resolution of all relevant wider impacts...". The impacts include, "existing highway capacity and highway safety".
- As required by Paragraph 5.13.3 of NPS EN-1 (2011), the Applicant has considered the likely traffic generation from the Scheme and undertaken a transport assessment which is contained at Appendix 14.1 [REP4-036]. Consultation has been undertaken with Lincolnshire County Council highway officers to seek agreement of the assessment approach and mitigation measures. Further detail can be found at Section 14.2, Consultation of ES Chapter 14: Transport and Access [APP-052]

Construction Phase Impacts



- 6.13.10 In terms of highway capacity, the Scheme is located within a rural area with good access to the strategic road network. West Burton 1, 2 and 3 are all located to the south of the A1500 Till Bridge Lane, near Sturton by Stow.
- 6.13.11 There will be a total of eight access points across West Burton 1, 2, 3. All will be used for both the construction and operational phases. The access locations are detailed in Table 14.11 and shown in Figure 14.4 of ES Chapter 14: Transport and Access [APP-052]. Six of the access points will be existing and improved existing field accesses. The other two will be new accesses. Seven of the access points will be used for construction and operation. The eighth will be used for operation only.
- 6.13.12 Table 14.13 and Table 14:14 of ES Chapter 14: Transport and Access [APP-052] set out the anticipated traffic flows for the 520 construction period working days. They describe that 372 daily staff car, shuttle bus and LGV trips and 46 HGV trips are expected during a peak construction phase day for the three sites. Construction worker shifts will be scheduled so that workers are not traveling during the network peak hours of 08:00-09:00 and 17:00-18:00.
- 6.13.13 In relation to the construction of the cable within the Cable Route Corridor, it is anticipated that this will be built out in phases and each of the 19 accesses for the Cable Corridor Route will be used for approximately 90 days during the construction phase. It is likely that four or five accesses will be in use concurrently. It is forecast that each access will generate up to eight arrivals and eight departures per day for the delivery of material and equipment. Around half of these will be HGV trips and half LGV trips. There will also be around 10 construction workers per access, arriving by car and shuttle bus.
- 6.13.14 ES Chapter 14: Transport and Access [APP-052] has assessed the impacts of these movements in terms of severance; driver delay; pedestrian delay; pedestrian and cyclist amenity; fear and intimidation; accidents and safety; and hazardous loads during construction of the Scheme. Table 14.24 of ES Chapter 14: Transport and Access [APP-052] shows that there are not expected to be any significant residual effects in relation to Transport and Access as a result of the construction of the Scheme. The ES also concludes that cumulative effects are not expected to change compared to the residual effects, that are set out within Table 14.24.
- 6.13.15 Although no more than negligible or minor effects are expected, mitigation measures proposed include a Public Right of Way Management Plan, Traffic Management Measures, including signage and a Stage 1 Road Safety Audit at all access junctions to recommend additional safety measures at the access points. A Construction Traffic Management Plan (CTMP) is also proposed to be secured by a Requirement of the DCO in order to manage HGV and staff access to the Order The outline CTMP is provided at ES Appendix 038].EN010132/EX7/WB6.3.14.2 E]. This includes measures to construction vehicle access and routing to the Order limits. Section 5 of the CTMP sets out construction traffic routing for each of the Sites and the Cable Route Corridor to avoid passing through the villages. It also sets out routes for abnormal



loads in order to ensure that the access roads are suitable for the transportation of components to the site as required by NPS EN-3 (November 2023) Paragraphs 2.10.139 and 2.10.140.

- 6.13.16 The outline CTMP [REP4-038EN010132/EX7/WB6.3.14.2 E] also includes a construction worker travel plan which sets out proposed measures and controls for staff vehicles, including proposals to discourage and limit access to the Order limits by car. This includes a shuttle bus service to the Order limits from local worker accommodation. Staff accessing the Order limits by car will be encouraged to car share to help minimise additional vehicles on local roads. This is in accordance with paragraph 5.13.4 of NPS EN-1 (2011) and paragraphs 5.14.7 and 5.14.8 of NPS EN-1 (November 2023) which require provision of a travel plan where appropriate.
- 6.13.17 NPS EN-1 (2011) paragraph 5.13.10 and NPS EN-1 (November 2023) paragraph 5.14.16 also require applicants to consider the use of water-borne or rail transport over road transport at all stages of the project, where cost-effective. In response to this specific policy, Chapter 14: Transport and Access of the ES [APP-052] notes that use of the river will be considered where appropriate. However, in all cases, the 'final leg' of deliveries during the construction phase will be undertaken by the roads set out in the study area. This is the same for rail transport. For example, larger equipment, such as transformers, will be transported to Immingham Docks. The final leg of the delivery will be via the strategic and local highway network. This is set out in the outline CTMP [REP4-038EN010132/EX7/WB6.3.14.2 E].
- 6.13.18 The Scheme therefore complies with Paragraph 5.13.6 of NPS EN-1 (2011) in mitigating construction phase impacts and with CLLP Policy ST51 of the DBLP in ensuring highway safety and no adverse highway impact in relation to the suitability of local access roads and highway capacity.
- 6.13.19 Impacts upon PRoW, pedestrians and cyclists during the construction phase are assessed in both ES Chapter 14: Transport and Access [APP-052] and the Transport Assessment [REP4-036]. The assessment concludes that the construction of the Scheme is not expected to result in any significant effects.
- 6.13.20 Sixteen Public Rights of Way are located within the Order limits. The majority of these, cross the Cable Corridor rather than being located within the Sites. They are listed within Table 2.1 of the Outline Public Rights of Way Management Plan [REP5-018]. These, together with other Public Rights of Way in the vicinity of the Sites are set out at Table 14.5: Public Rights of Way of ES Chapter 14: Transport and Access [APP-052] and are shown within the Transport Figures [APP-052].
- An Outline Public Rights of Way Management Plan [REP5-018] has been submitted with the DCO application in compliance with paragraph 2.10.45 of the NPS EN-3 (November 2023). This sets out that all PRoW within the sites are to remain open for the duration of construction and diversions are not proposed given the low number of surveyed users. It also sets out measures including a widened access track to ensure vehicles can pass PRoW users safely, and the provision of banksmen to hold



vehicles if a PRoW user is present and advise PRoW users of the potential for construction vehicles to be present.

- 6.13.22 In terms of the cable route, when the cable is installed, there will be some instances where PRoW need to be closed to users for a short period. This will not occur at all PRoWs, as directional drilling will be used in some places. The Outline Public Rights of Way Management Plan [REP5-018] explains that where there is a requirement to temporarily close the PRoW, works will be undertaken over-night so far as is practicable to do so, when there are unlikely to be any PRoW users. It is anticipated that the installation of cables over short sections where the PRoW is located can be undertaken in a single overnight period. The PRoW will remain open, and managed, during the daytime period so far as it is practicable to do so. The Scheme would therefore comply with Paragraphs 2.10.40 to 2.10.44 of NPS EN-3 (November 2023) through enabling continued recreational use of public rights of way (PRoW), during construction. It would also accord with Paragraph 5.13.6 of NPS EN-1 (2011) in mitigating construction phase impacts and with CLLP Policy ST51 of the DBLP in ensuring highway safety and no adverse highway impact in relation to PROW.
- As set out above, there are not expected to be any significant residual effects or cumulative effects in relation to Transport and Access as a result of the construction of the Scheme. The Scheme is considered to comply with the requirements of NPS EN-1 (2011), NPS EN-1 (November 2023) and NPS-3 (2011) in ensuring that the transport implications of the Scheme during construction are properly assessed and through demonstrating mitigation of construction phase impacts, access road suitability and highway safety. It also complies with CLLP Policy ST51 of the DBLP in ensuring highway safety and no adverse highway impact and Paragraphs 2.10.40 to 2.10.44 of NPS EN-3 (November 2023), in terms of keeping PROW open for recreational users.

Operational Phase Impacts

- During operation, it is anticipated that there will be around five visits to each Site per month for maintenance purposes. These would typically be made by light van or 4x4 type vehicles with HGVs rarely expected to access the Order limits. There will be no transport operational effects associated with the installed grid connection cables (within the Cable Route Corridor) as they will be located underground. Access may be required for maintenance, but this is only likely once or twice a year. In light of this, effects on accidents and safety, severance, driver delay, pedestrian delay and amenity and hazardous loads during the operational phase of the Scheme are considered within ES Chapter 14: Transport and Access [APP-052] to be negligible and not significant. No cumulative impacts are anticipated.
- 6.13.25 The Scheme therefore complies with Paragraph 5.13.6 of NPS EN-1 (2011) in mitigating operational phase impacts and with CLLP Policy ST51 of the DBLP in ensuring highway safety and no adverse highway impact in relation to the suitability of local access roads and highway capacity.



- Ouring the operational phase of the Scheme, the Outline Public Rights of Way Management Plan [REP5-018] sets out that existing PRoW will remain open. In addition, a new permissive path from the track off Sykes Lane to Sturton Road along the Codder Lane Belt will be provided within the Order limits. Along with enhancement of existing PRoW as set out within the Outline Landscape and Ecological ManagementPlan (OLEMP) [EN010132/EX6/WB7.3_EREP6-025] this will help to enhance connectivity within the Sites and pedestrian and cycle access. This is in accordance with paragraphs 2.10.40 to 2.10.45 of NPS EN-3 (November 2023) which notes that sites may provide the opportunity to facilitate enhancements to the local footpath network and the adoption of new public rights of way through site layout and design.
- 6.13.27 The avoidance and minimisation of effects on PRoW as set out by the Outline Public Rights of Way Management Plan [REP5-018] and the enhancement of connectivity through the Order limits by the incorporation of a new permissive path is in accordance with NPS EN-3 (November 2023) paragraphs 2.10.40 to 2.10.45. This also shows that the Scheme is in accordance with CLLP Policy ST51 by providing well designed, safe and convenient access, enhancing pedestrian permeability and avoiding barriers to movement through closure or diversion of PRoW during the operational phase.
- 6.13.28 There are not expected to be any significant residual effects in relation to Transport and Access as a result of the operation of the Scheme which is considered to comply with the requirements of NPS EN-1 (2011), NPS EN-1 (November 2023) and NPS-EN3 (2011) in ensuring that the transport implications of the Scheme during operation are properly assessed and through demonstrating mitigation of construction phase impacts, access road suitability and highway safety. It also complies with CLLP Policy ST51 of the DBLP in ensuring highway safety and no adverse highway impact and paragraphs 2.10.40 to 2.10.45 of NPS EN-3 (November 2023) in terms of keeping PROW open during operation, maximising pedestrian permeability and avoiding barriers to movement.

Decommissioning Phase Impacts

- 6.13.29 ES Chapter 14: Transport and Access [APP-052] has assessed the impacts of vehicle movements in terms of severance; driver delay; pedestrian delay; pedestrian and cyclist amenity; fear and intimidation; accidents and safety; and hazardous loads during decommissioning of the Scheme. The number of vehicles associated with the decommissioning phase are not anticipated to exceed the number set out for the construction phase. The effects will be short term and temporary and are assessed to be equivalent to, and no worse than, the negligible or minor construction phase impacts. They are not therefore significant in EIA terms.
- 6.13.30 Mitigation measures will be the same as for the construction phase and will include a Public Right of Way Management Plan, Traffic Management Measures, including signage, Stage 1 Road Safety Audit at all access junctions to recommend additional safety measures at the access points.



- 6.13.31 A Decommissioning Plan will be submitted to the local planning authority for approval prior to decommissioning. This will be secured by a requirement of the DCO and will be based on the measures set out in the Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023].
- 6.13.32 The Scheme therefore complies with Paragraph 5.13.6 of NPS EN-1 (2011) in mitigating decommissioning phase impacts and with CLLP Policies 13 and LP19 and Policy ST51 of the DBLP in ensuring highway safety and no adverse highway impact in relation to the suitability of local access roads and highway capacity.
- An Outline Public Rights of Way Management Plan [REP5-018] has been submitted with the DCO application in compliance with paragraph 2.10.45 of the NPS EN-3 (November 2023). This sets out that all PRoW within the sites will be managed in a similar way to the construction phase to ensure user safety and continued access to routes. It would also accord with Paragraph 5.13.6 of NPS EN-1 (2011) in mitigating decommissioning phase impacts and with CLLP Policy ST51 of the DBLP in ensuring highway safety and no adverse highway impact in relation to PROW.

<u>Summary</u>

- In summary, traffic generated by the Scheme during construction, operation and decommissioning is not expected to result in any significant adverse environmental effects upon strategic and local highway network users; including pedestrians, cyclists and users of public transport. It is also not expected to have a significant impact on the strategic or local highway networks in terms of capacity and highway safety. The Scheme is considered to comply with the requirements of NPS EN-1 (2011), NPS EN-1 (November 2023) and NPS-3 (2011), CLLP ST51 of the DBLP in ensuring that the transport implications of the Scheme are properly assessed, (including cumulative impacts), and through demonstrating mitigation of impacts during all phases, access road suitability and highway safety.
- 6.13.35 PRoWs within the sites will not be closed or diverted during construction but will be appropriately managed. The provision of a new permissive path will also provide a benefit to local recreational users by increasing public access across the Order limits. The Scheme is therefore in accordance with the transport and access policies of NPS EN-1 (2011), NPS EN-1 (November 2023) and would comply with Paragraphs 2.10.40 to 2.10.44 of NPS EN-3 (November 2023) seeks to maximise pedestrian permeability and avoid barriers to movement.

6.14 Waste

- 6.14.1 The Applicant has considered the waste streams arising from the Scheme, and the implications for existing waste facilities, in the context of planning policy.
- 6.14.2 The Environmental Protection Act 1990 provides the structure and authority for waste management and control of emissions into the environment. Part II of the Act relates to Waste on Land and places a Duty of Care on anyone who produces, stores, transports or disposes of waste to take all reasonable steps to ensure that waste is



managed properly. This Duty of Care will be applied throughout the lifetime of the Scheme.

- 6.14.3 The Waste Framework Directive provides the framework for the management of waste across the EU. The Waste (England and Wales) Regulations 2011 (as amended) transposed the Waste Framework Directive into domestic law in England and Wales. The framework requires waste prevention programmes and waste management plans to apply the waste hierarchy, with prevention being the most preferred method, through reduction, recycling, recovery, to disposal as the least preferred method. The waste hierarchy is to be applied throughout the lifetime of the Scheme, predominantly at the construction and decommissioning phases.
- 6.14.4 The Waste Electrical and Electronic Equipment ('WEEE') Recycling Government Guidance Note (January 2014) provides specific advice about compliance with the WEEE Regulations 2013. The WEEE Regulations 2013 apply to all Electrical and Electronic Equipment ('EEE') placed on the market in the UK covered by the scope of the Regulations. Obligations are imposed on producers, distributors and consumers of EEE. The Applicant will comply with the WEEE Regulations as relevant to the Scheme and will have regard to the DEFRA document titled "Guidance on Best Available Treatment Recovery and Recycling Techniques (BATRRT) and treatment of Waste Electrical and Electronic Equipment (WEEE), "or other document relevant at the time, when formulating its decommissioning strategy.
- 6.14.5 The Environment Act 2021 is to operate as the UK's new framework of environmental protection. Given that the UK has left the EU, new laws that relate to nature protection, water quality, clean air, as well as additional environmental protections, needed to be established. The Environment Act allows the UK to enshrine some environmental protection into law. It offers new powers to set new binding targets, including for (among other things) waste reduction. Part 3 is related to waste and resource efficiency, and will include obligations for managing waste, enforcement and regulation. The Applicant intends to accord with the regulations when enshrined into law as far as relevant to the Scheme.
- 6.14.6 The Overarching National Policy Statement for Energy (NPS EN-1 2011) sets out in Section 5.14 'Waste Management' the strategy for reducing the amount of waste where possible and trying to use it as a resource wherever possible. Paragraph 5.14.6 states that, "The applicant should set out the arrangements that are proposed for managing any waste produced and prepare a Site Waste Management Plan. The arrangements described and Management Plan should include information on the proposed waste recovery and disposal system for all waste generated by the development, and an assessment of the impact of the waste arising from development on the capacity of waste management facilities to deal with other waste arising in the area for at least five years of operation."
- 6.14.7 It goes on to further state that applicants should seek to minimise the volume of waste produced and the volume of waste sent to disposal unless it can be demonstrated that this is the best overall environmental outcome. Construction



best practices should be utilised in relation to storing of materials in an adequate and protected place on site to prevent waste.

- An Outline Construction Environmental Management Plan (outline CEMP) 6.14.8 [EN010132/EX6/WB7.1_DREP6-021] and Outline Decommissioning Statement (outline DS) [EN010132/EX6/WB7.2_BREP6-023] have been submitted with this Development Consent Order Application. Section 2.10 of the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] relates to waste and recycling and identifies measures to control and manage waste on-site. This includes (among other things) separation of the main waste streams onsite, prior to transport to an approved, licensed third party waste facility. Part of Table 3.1 of the Outline DS [EN010132/EX6/WB7.2_BREP6-023] also relates to waste. Both will be secured through a DCO Requirement. Furthermore, a detailed Construction Resource Management Plan (CRMP), Construction Environmental Management Plan (CEMP), Decommissioning Environmental Management Plan (DEMP) and Decommissioning Resource Management Plan (DRMP) will be prepared for the construction and decommissioning phases, and will be approved by the relevant Planning Authority prior to works commencing in that phase. In this context, it is considered that the Scheme accords with the requirements of the 2011 NPS for Energy in respect of Waste Management.
- 6.14.9 The same approach to waste management set out in the NPS EN-1 (2011) is reflected in the NPS EN-1 (November 2023), which also encourages applicants to refer to the Waste Prevention Programme for England. Paragraph 5.15.9 of NPS EN-1 (2011) states that "The IPC should consider the extent to which the applicant has proposed an effective system for managing hazardous and non-hazardous waste arising from construction, operation and decommissioning of the proposed development. It should be satisfied that: any such waste will be properly managed, both on-site and off-site; the waste from the proposed facility can be dealt with appropriately by the waste infrastructure which is, or is likely to be, available; and adequate steps have been taken to minimise the volume of waste arisings, and of the volume of waste arising sent to disposal, except where that is the best overall environmental outcome".
- 6.14.10 In light of the Outline CEMP and Outline DS being secured through a DCO Requirement, and a commitment to a CRMP, CEMP, DEMP and DRMP being prepared and approved prior to commencement of the construction and decommissioning phases, it is considered that the Scheme accords with the requirements of NPS EN-1 (November 2023) in respect of Waste Management.
- 6.14.11 The Lincolnshire County Council Minerals and Waste Local Plan (June 2016) sets out the vision, objectives, spatial strategy and development management policies for minerals and waste development in Lincolnshire up to 2031. The policies in the Local Plan solely focus on the provision of waste facilities, and therefore are not considered explicitly relevant in the context of the Scheme.
- 6.14.12 Lincolnshire County Council is in the process of reviewing the Minerals and Waste Local Plan. This follows a review of the existing policy framework undertaken in



2020. The final report identifying the conclusions of the review was approved by the County Council on 19 February 2021. The review highlighted issues with a number of policies of the Local Plan and concluded that the most appropriate course of action would be to update the Local Plan in its entirety. The latest timetable for the review is set out in the Lincolnshire Minerals and Waste Development Scheme (February 2021). A consultation on the issues and options for updating the Local Plan took place from 28 June 2022 to 12 August 2022, although no draft policies are included within the consultation document. On this basis, the emerging Local Plan Review is not considered of relevance in the context of the Scheme.

- 6.14.13 A number of evidence base documents support the emerging review. This includes (among other documents) the Lincolnshire Waste Needs Assessment 2021 Report 3 (June 2021) which estimates Lincolnshire's future management requirements for Construction, Demolition and Excavation Waste. As set out in ES Chapter 20: Waste [APP-058], it is considered that there will be no significant effects on waste handling facilities in Lincolnshire, and therefore the Scheme is not likely to be in conflict with its emerging policies in respect of Waste Management.
- 6.14.14 The 'Saved' Policies of the Nottinghamshire and Nottingham Waste Local Plan (2002) and Waste Core Strategy (2013) provide the policy context in respect of waste management. The Local Plan 'Saved' policies are partly replaced by the Core Strategy. The 'Saved' Policies relate to proposals for waste management and associated facilities and are therefore not considered explicitly relevant in the context of the Scheme. The Waste Core Strategy sets out the approach to delivering sustainable waste management in Nottinghamshire and Nottingham until 2031. The strategy also sets out strategic policy and criteria on the location and types of facilities that are needed. The Core Strategy is not considered explicitly relevant in the context of the Scheme.
- 6.14.15 The County Council is working on preparing a new Local Plan which will replace both the Local Plan 'Saved' Policies and Core Strategy once adopted. A Draft Plan was published for consultation on 7 February 2022. It included draft policies against which proposals for new waste development will be assessed once adopted. The emerging policies of the Draft Plan are not therefore considered relevant in the context of the Scheme.
- 6.14.16 The BCSDMP does not contain any policies related to waste management and therefore it is not considered explicitly relevant in the context of the Scheme. There are no draft policies related to waste management within the DBLP, and therefore it is not considered explicitly relevant in the context of the Scheme.
- 6.14.17 As previously noted, the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and Outline DS [EN010132/EX6/WB7.2_BREP6-023] are to be secured through a DCO Requirement, and a commitment is included to prepare and approve a CRMP, CEMP, DEMP and DRMP prior to commencement of the construction and decommissioning phases. These commitments, in combination, will ensure that construction waste is minimised.



- 6.14.18 CLLP Policy S10 supports proposals which, in principle, demonstrate their compatibility with, or the furthering of, a circular economy in the local area. It is considered that the Scheme accords with this emerging requirement on the basis that the Outline CEMP and Outline DS are to be secured through a DCO Requirement, and a commitment is included to prepare and approve a CRMP, CEMP, DEMP and DRMP prior to commencement of the construction and decommissioning phases.
- 6.14.19 ES Chapter 20: Waste **[APP-058]** assesses the waste impacts of the Scheme. When considered both in isolation and cumulatively with the identified projects within the assumed proximity (i.e., within the Local Impact Area which is the area covered by Lincolnshire County Council, Nottingham City Council and Nottinghamshire County Council), the environmental effects from waste generated by the Scheme and cumulative projects are considered to be as follows:
 - The overall effects of waste handling facilities in the Local Impact Area are not likely to be significant at any stage of the assessed time frame;
 - No waste handling facilities in Lincolnshire are likely to see significant effects at any stage of the assessed timeframe;
 - No waste handling facilities in Nottinghamshire are likely to see significant effects during the construction or operational lifetime of the Scheme;
 - Waste recycling and recovery handling facilities in Nottinghamshire are not likely to see significant effects during the construction or operational lifetime of the Scheme;
 - Waste handling facilities for landfill waste handling in Nottinghamshire are likely to see a significant effect during the decommissioning of the Scheme and cumulative decommissioning phase as a result of the lack of landfill capacity from the year 2030.
- 6.14.20 It is considered that the anticipated impacts from the Scheme can be sufficiently mitigated through adherence to the measures set out in the Outline CEMP and OEMP and the Outline DS. These, along with their full counterparts to be provided post-consent (i.e., the CRMP, CEMP, DEMP and DRMP) will ensure that the Scheme is developed with good practices towards use of materials and water, and management of waste in keeping with the principles of the Waste Hierarchy.
- 6.14.21 The scheme is therefore considered to be in accordance with the Environmental Protection Act 1990, the Environment Act 2021, the Waste Framework Directive, the Waste Electrical and Electronic Equipment Regulations 2013, NPS EN-1 (2011), NPS EN-1 (November 2023), CLLP Policy S10.

6.15 Socio-economics, Tourism and Recreation

6.15.1 Section 5.12 of NPS EN-1 (2011) and section 5.13 of NPS EN-1 (November 2023) set out the requirements for the assessment of local and regional socio-economic impacts of energy NSIPs. NPS EN-1 (2011) Paragraph 5.12.13 states that the



assessment should consider all relevant socio-economic impacts, which may include: the creation of jobs and training opportunities; the provision of additional local services and improvements to local infrastructure; effects on tourism and the impact of a changing influx of workers during the different construction, operation and decommissioning phases of the energy infrastructure.

- 6.15.2 NPS EN-1 (2011) Paragraph 4.13.4 notes that provision of energy infrastructure may have indirect health impacts, for example if it in some way affects access to key public services, transport or the use of open space for recreation and physical activity.
- 6.15.3 The NPPF (paragraphs 85, 86, 96, 101 and 104) supports sustainable economic growth; the achievement of healthy, inclusive and safe places; and the protection of existing land uses and community infrastructure including rights of way.
- 6.15.4 The relevant Local planning policies are set out at paragraph 18.3.23 of ES Chapter 18, Socio Economics Tourism and Recreation [APP-056]. These cover a range of topics and include CLLP Policy S5 which support non-residential development in the countryside if the rural location of the enterprise is justifiable to maintain or enhance the rural economy, or the location is justified by means of proximity to existing established businesses or natural features.
- In addition, BCSDMP Policy DM1 seeks to support economic development in the countryside, e.g., tourist attractions; equine enterprises; rural business. BCSDMP Policy DM10 seeks to ensure that renewable energy proposals are compatible with tourism and recreational facilities. DBLP Policy ST 11 seeks to achieve rural economic growth and Policy ST 12 seeks to develop the visitor economy. Policy ST47 promotes sport and recreation.
- 6.15.6 ES Chapter 18, Socio Economics Tourism and Recreation [APP-056], provides an assessment of socio-economic effects including upon employment, the local economy, development land, public rights of way and local amenities and land use, in accordance with NPS EN-1 (2011) paragraph 5.12.3 and NPS EN-1 (November 2023) paragraph 5.13.4. The socio-economic effects of the Scheme are set out in the following sections.

Effects on employment and the local economy

6.15.7 ES Chapter 18, Socio Economics Tourism and Recreation [APP-056] presents the impacts on employment and its effects on the local economy of the Scheme during construction, operation and decommissioning. It identifies that the Scheme will have significant beneficial effects in terms of access to employment and education during the construction phase of the Scheme. It identifies that the Scheme will support 222 net direct jobs per annum during the construction period. Of these, 142 jobs per annum will be expected to be taken-up by residents within the combined areas of Bassetlaw District and West Lindsey District. As set out in section 4.6 of this Planning Statement, a local skills and employment plan will be prepared prior to the commencement of construction. This will set out measures that the Applicant will



implement in order to advertise and promote employment opportunities associated with the Scheme in construction and operation locally.

- The gross value added (GVA) to the economy of these workers is expected to be £13.3 million, of which £7.7 million will be of benefit to the local economy within the combined areas of Bassetlaw District and West Lindsey District. The operation and maintenance of the Scheme is anticipated to generate a net uplift to Gross Value Added of £500,000 per annum with £300,000 of this to the benefit to the local economy within the combined areas of Bassetlaw District and West Lindsey District. The decommissioning of the Scheme is likely to generate approximately 80% of the GVA per annum as the construction phase (adjusted for inflation).
- 6.15.9 During operation, ES Chapter 18 [APP-056] sets out that the Scheme would directly generate a gross 12 FTE employees per annum, 8 of which will be within the combined areas of Bassetlaw and West Lindsey. This number of workers for operation and maintenance has been provided by the Applicant based on industry experience and professional judgement. There are approximately 13 agricultural sector jobs and 5 tourism and recreation jobs that will remain lost during the Scheme's operational lifetime. There will be a net loss of 2 FTE jobs as a result of the Scheme within the combined areas of Bassetlaw District and West Lindsey District. However, the actual number of jobs generated by the Scheme may be greater as part-time staff will be created to perform maintenance and engineering works from time to time to ensure the Scheme is operational over a long period. Furthermore, an estimated 10 FTE jobs per annum will be generated in the combined areas of Bassetlaw District and West Lindsey District as a result of indirect or induced employment, such as through supply chains. In addition, as set out in section 4.6 of this Planning Statement, the Applicant will make a skills and education contribution to assist and encourage local people to access apprenticeships and training.
- 6.15.10 In addition, there will be significant medium term temporary major beneficial effects upon local accommodation sector employment and upon the accommodation stock during the construction period. Accommodation sector employment will also benefit during the decommissioning phase.
- 6.15.11 Overall, it is considered that the direct and indirect employment creation resulting from the Scheme and gross value added (GVA) to the economy, is in accordance with the NPPF (paragraphs 85, 86, 97, 109 and 118) which supports sustainable economic growth. It also accords with the aims of Policy ST 11 in terms of delivering jobs, economic prosperity and rural economic growth within the District respectively. The clear benefits arising from the Scheme in terms of employment generation through the construction, operation and decommissioning of the Scheme, outweigh the impacts of the loss of these energy sector jobs at the end of the Scheme's life.

Effects on Tourism and Recreation

Tourism Attractions



6.15.12 Impacts on tourism attractions have been assessed in Chapter 18 [APP-056] during construction, operation and decommissioning. The Scheme's estimated two-year construction period is likely to have a degree of impact on tourism attractions in the immediate locality and combined districts of Bassetlaw and West Lindsey. The potential changes to landscape views, both temporarily from construction equipment and longer-term from the installation of the Scheme infrastructure, and the impacts from construction traffic impacting the desirability and accessibility of tourism and recreation routes and attractions, could negatively impact the prosperity of the local tourism economy. The construction phase effects upon local tourist attractions are assessed to be significant for this temporary period.

Public Rights of Way and Long Distance Recreational Routes

- As discussed in section 6.12, Public Rights of Way cross the Order limits. ES Chapter 18 [APP-056] sets out that the Scheme's construction is likely to have direct impacts on a number of Public Rights of Way and long distance recreational routes as a result of temporary use as construction accesses, any required diversions and closures, and secondary temporary impacts as a result of movement of construction goods and employee vehicles. Embedded mitigation to limit the impacts is set out in the Outline Public Rights of Way Management Plan [REP5-018], OCEMP [EN010132/EX6/WB7.1_DREP6-021], and CTMP [REP4-038EN010132/EX7/WB6.3.14.2 E]. The residual impacts upon long distance recreational routes are assessed as significant for this temporary period.
- 6.15.14 The creation of the permissive path from Sykes Lane up to the Codder Lane Belt is assessed to have a localised moderate-minor beneficial effect on recreational walking and cycling, and this resultantly on health and wellbeing. Access to existing PRoW routes will be retained with only short term (e.g overnight closures) and active management of routes as set out within the Outline Public Rights of Way Management Plan [REP5-018].

Recreation facilities and attractions

6.15.15 Waterways and bodies of water used for recreation are not anticipated to be impacted directly by the Scheme due to their physical separation from construction works on the Sites, and the use of horizontal directional drilling for crossing major waterways, as demonstrated in the **Crossing Schedule [REP4-056].** Recreational use of the River Trent and other local waterways is not considered to be significantly impacted. Neither are significant impacts expected for the formal recreational facilities for activities such as golf, cricket, and flying which are located within 5km of the Sites or other attractions.

<u>Summary</u>

6.15.16 There are significant beneficial socio-economic effects of the Scheme as a result of the employment and education opportunities created during construction and decommissioning. In addition, there will be benefits to the use of the accommodation stock during construction plus the creation of new permissive



paths during operation. The assessment of tourism impacts identifies that there is a significant adverse effect to local tourism attractions, however this effect is limited to a very small number of locations, and only for the duration of the Scheme's construction. There is also a significant adverse effect on long distance recreational routes, which again is only for the duration of construction. Cumulative effects have been assessed and do not raise any additional issues. Therefore these impacts present only a limited, short term conflict with Policy DM10 which seeks to ensure that renewable energy proposals are compatible with tourism and recreational facilities. The significant public and other benefits of the Scheme set out at Section 4.0, are considered to outweigh this short term conflict. There is no conflict with SCLLP Policy LP7 which seeks to deliver a Sustainable Visitor Economy. The Scheme accords with NPS EN-1 (2011), NPS EN-1 (November 2023), and the NPPF which support sustainable economic growth, existing and future land uses and community infrastructure including rights of way.

6.16 Effects on Human Health

- 6.16.1 Section 4.13 of NPS EN-1 (2011) and section 4.3 of NPS EN-1 (November 2023) describe the potential health impacts of energy NSIPs. Paragraph 4.13.2 states that the ES should assess these effects for each element of the project, identifying any adverse health impacts, and identifying measures to avoid, reduce or compensate for these impacts as appropriate.
- 6.16.2 NPS EN-1 (November 2023) paragraph 4.4.7 states that:

"Generally, those aspects of energy infrastructure which are most likely to have a significantly detrimental impact on health are subject to separate regulation (for example for air pollution) which will constitute effective mitigation of them, so that it is unlikely that health concerns will either by themselves constitute a reason to refuse consent or require specific mitigation under the Planning Act 2008."

- 6.16.3 Paragraph 4.4.8 goes on to state that:
 - "However, not all potential sources of health impacts will be mitigated in this way and the Secretary of State may want to take account of health concerns when setting requirements relating to a range of impacts such as noise."
- 6.16.4 DBLP Policy ST44 promotes healthy active lifestyles.
- 6.16.5 In accordance with NPS EN-1 (2011) section 4.13 and NPS EN-1 (November 2023) section 4.3, the Applicant has undertaken a Human Health Assessment which is set out in ES Chapter 21: Other Environmental Matters [APP-059]. This has assessed the principal health benefits and disbenefits to residents of the local community of the Scheme. Table 21.5.4 contains a summary of significant likely effects.
- 6.16.6 No significant effects on health are anticipated in respect to flooding, ground conditions, noise and vibration, glint and glare, air quality, waste and major accidents and disasters. The chapter identifies that during the construction and decommissioning periods the Scheme is expected to lead to a positive health impact



on access to work and training opportunities as a result of the local employment created. Short term adverse effects on long distance recreation routes are anticipated as a result of construction noise, traffic and views.

- 6.16.7 The residual cumulative effects on access to primary healthcare and on disability and long-term health are assessed within Chapter 21 as significant as a result of incombination and cumulative construction impacts on the use and desirability of long-distance recreation routes.
- 6.16.8 The cumulative uplift in employment and skills training and education opportunity are anticipated to have significant beneficial effects on human health and wellbeing as a result of improved measures of indices of multiple deprivation.
- 6.16.9 The creation of the new semi-accessible habitat management area and a permissive path from Sykes Lane up to the Codder Lane Belt is assessed to have a localised moderate-minor beneficial effect on recreational walking and cycling, and this resultantly on health and wellbeing.

<u>Summary</u>

- 6.16.10 There are positive effects on human health as a result of the employment and skills training and education opportunity as well as through the significant employment created during construction and decommissioning. The Scheme therefore accords with NPS EN-1 (2011), NPS EN-1 (November 2023), and the NPPF which support sustainable economic growth and the protection of health, existing and future land uses and community infrastructure including rights of way.
- 6.16.11 Mitigation measures are set out at paragraph 21.5.32 21.5.33 of ES Chapter 21: Other Environmental Matters [APP-059] as required by the policy. The Scheme complies with DBLP Policy ST44 and ST47 in terms of aiding healthy active lifestyles and helping to promote recreation opportunities through the provision of the new permissive path from Sykes Lane up to the Codder Lane Belt. It also enables access to existing PRoW routes to be retained with only short term (e.g overnight closures) and active management of routes as set out within the Outline Public Rights of Way Management Plan [REP5-018].

6.17 Major Accidents and disasters

- 6.17.1 The Infrastructure Planning (Environmental Impact Assessment) Regulations 2017 require assessment of the potential effects of the Scheme on the environment as a result of the vulnerability of the Scheme to risks of major accidents or disasters which are relevant to the Scheme.
- 6.17.2 As the energy NPSs were published in 2011, they pre-date the existing EIA Regulations. The NPPF does refer, at paragraph 101, to the fact that: "Planning policies and decisions should promote public safety and take into account wider security and defence requirements by: a) anticipating and addressing possible malicious threats and natural hazards... This includes appropriate and proportionate steps that can be taken to reduce vulnerability, increase resilience and ensure public safety and security".



- 6.17.3 CLLP Policy 14 in relation to renewable energy development sets out a number of tests to determine whether a proposal is acceptable. These include impacts on highway safety and impacts on aviation and defence navigation system/communications.
- 6.17.4 ES Chapter 21: Other Environmental Matters [APP-059] considers a number of potential accidents and disasters, including floods, fire, road accidents, rail accidents, aircraft disasters, flood defence failure, utilities failure, mining/extractive industry, and plant disease. The residual effects of major accidents and disasters are assessed as not being significant.
- 6.17.5 Minimising the risk of major accidents during construction, operation and decommissioning will be addressed through appropriate measures set out in the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023]. The detailed preparation and implementation of these plans are secured via requirements to the DCO.
- 6.17.6 An Outline Battery Storage Safety Management Plan (BSSMP) [EN010132/EX6/WB7.9_BREP6-019] has been prepared for the Scheme. The proposed design for fire mitigation includes the BESS containers being fitted with thermal monitoring, battery cooling systems, remote and local emergency stops, fire detection, and fire suppression equipment.
- 6.17.7 In terms of highway safety, the assessment of road accidents and safety presented in ES Chapter 14: Transport and Access [APP-052] Table 14.19 concludes that the effects of construction traffic to the Sites and to the Cable Corridor on accidents and highway safety are negligible. The effects of transportation of hazardous loads are deemed to be negligible and temporary.
- 6.17.8 With regard to aviation safety, the assessment of effects set out in ES Chapter 16: Glint and Glare [APP-054] states that no significant effects are predicted in respect of aviation receptors during the operational lifetime of the Scheme. As such, there are no significant effects relating to major accidents and disasters with regard to aviation accidents.

Summary

6.17.9 In summary the Scheme is unlikely to pose a significant risk to the health and safety of the public from major accidents and disasters and therefore is in accordance with NPPF paragraph 101, CLLP 14 with regard to public safety.

6.18 Air Quality

6.18.1 Paragraphs 5.2.6 and 5.2.7 of NPS EN-1 (2011) and 5.2.8 and 5.2.9 of NPS EN-1 (November 2023) state that where a project is likely to have adverse effects on air quality the applicant should undertake an assessment of the impacts of the proposed project as part of the Environmental Statement (ES). In accordance with these policies, the air quality impacts of the Scheme have been assessed within ES Chapter 17: Air Quality [APP-055].



- 6.18.2 With regards to the decision-making process, NPS EN-1 (2011) paragraph 5.2.9 states that air quality considerations should be given substantial weight where a project would lead to a deterioration in air quality in an area, or lead to a new area where air quality breaches any national air quality limits.
- 6.18.3 Paragraph 5.2.9 of NPS EN-1 (2011) goes on to state that where substantial changes in air quality levels are expected, even if this does not lead to any breaches of national air quality limits, air quality considerations will also be important. Any relevant statutory air quality limits must be taken account of in all cases. Additionally, paragraph 5.2.10 of NPS EN-1 (2011) and NPS EN-1 (November 2023) says that where a project is likely to lead to a breach of such limits, appropriate mitigation measures should be secured.
- In all cases, the Secretary of State must take account of any relevant statutory air quality limits. Where a project is likely to lead to a breach of such limits the applicant should work with the relevant authorities to secure appropriate mitigation measures to allow the proposal to proceed.
- 6.18.5 NPPF Paragraph 192 states that planning policies and decisions should sustain and contribute towards compliance with relevant limit values or national objectives for pollutants, taking into account the presence of Air Quality Management Areas and Clean Air Zones, and the cumulative impacts from individual sites in local areas.
- 6.18.6 NPPF Paragraph 194 states: "The focus of planning policies and decisions should be on whether proposed development is an acceptable use of land, rather than the control of processes or emissions (where these are subject to separate pollution control regimes). Planning decisions should assume that these regimes will operate effectively".
- 6.18.7 CLLP Policy S14 requires that the impacts of the development are acceptable on the amenity of sensitive neighbouring uses (including local residents) by virtue of matters such as noise, dust, odour, shadow flicker, air quality and traffic. CLLP Policy S53 also requires proposals not to result in adverse impacts upon air quality from odour, fumes, smoke, dust and other sources.
- In accordance with Paragraphs 5.2.6 and 5.2.7 of NPS EN-1 (2011) effects of the Scheme on air quality including odour, fumes, smoke, dust and other sources at nearby sensitive receptors during construction, operation and decommissioning phases have been considered within ES Chapter 17: Air Quality [APP-055]. The assessment predicts the levels of air quality pollutants and assesses them to determine whether there are any likely significant effects taking account of relevant policy, guidelines and best practice.
- 6.18.9 Section 17.7 of ES Chapter 17: Air Quality **[APP-055]** identifies and evaluates the likely significant effects of the Scheme and identifies that these are likely to be construction and decommissioning phase dust and particulate matter and during the operational phase the effects of a fire incident on surrounding residents and the public. Following the implementation of the appropriate site-specific mitigation measures set out at Section 17.8 of the ES, the significance of the effects from dust



and PM10 emissions associated with the construction works is considered to be negligible on all receptors which is not significant in EIA terms. This assessment is based on Institute of Air Quality Management Guidance. All effects are considered to be temporary, direct, adverse and short term.

- Site specific mitigation measures relating to construction and decommissioning phase dust and particulate matter are incorporated within the Outline Construction Environmental Management Plan [EN010132/EX6/WB7.1_DREP6-021] and the outline decommissioning statement [EN010132/EX6/WB7.2_BREP6-023]. This will be secured through a DCO Requirement. This demonstrates that the importance of air quality considerations in respect of dust have been recognised in developing the Scheme, as required by paragraph 5.2.9 of NPS EN-1 (2011) and appropriate mitigation measures have been secured in accordance with paragraph 5.2.10 of NPS EN-1 (2011) and NPS EN-1 (November 2023). It also demonstrates that adverse impacts upon air quality during the construction and decommissioning phases have been considered and addressed as required by CLLP Policy S53. Furthermore, the Scheme accords with CLLP Policy S14 in so far as demonstrating that the dust impacts of the Scheme are acceptable on the amenity of sensitive neighbouring uses (including local residents).
- 6.18.11 In respect of the effects of a fire incident during the operational phase, an 'Air Quality Assessment on Emission Impact from the Battery Energy Storage Systems (BESS) Fire' has been undertaken and is included at Appendix 17.4 of the ES. This recommends various measures to be undertaken in the case of a fire, including informing any potential affected residents and advising the public about health effects of smoke, related symptoms, and ways to reduce exposure.
- 6.18.12 Following the implementation of these measures during an occurrence of a fire incident, ES Chapter 17: Air Quality [APP-055] determines the air quality effects to be negligible, which is not significant in EIA terms. An Outline Battery Safety Storage Safety Management Plan [EN010132/EX6/WB7.9_BREP6-019] has been produced incorporating these measures and has been submitted with the DCO application. This will be secured through a DCO Requirement.
- 6.18.13 This demonstrates that the importance of air quality considerations in respect of dust have been recognised in developing the Scheme, as required by paragraph 5.2.9 of NPS EN-1 (2011) and appropriate mitigation measures have been secured in accordance with paragraph 5.2.10 of NPS EN-1 (2011) and NPS EN-1 (November 2023). It also demonstrates that adverse impacts upon air quality from potential fire incidents during the operational phase have been considered and addressed as required by CLLP Policy S53. Furthermore, the Scheme accords with CLLP Policy S14 in so far as demonstrating that the air quality impacts of the Scheme are acceptable on the amenity of sensitive neighbouring uses (including local residents).
- 6.18.14 In terms of potential cumulative effects, the potential for cumulative traffic air quality effects have been considered within Section 17.9 of ES Chapter 17: Air Quality [APP-055]. This chapter anticipates that the cumulative vehicle numbers would not



exceed the 'Indicative criteria for requiring an air quality assessment' detailed within IAQM Guidance on 'Land-use planning & development control: Planning for air quality', January 2017 and, therefore, air quality modelling for cumulative traffic assessment is not required.

6.18.15 Following the implementation of the site-appropriate mitigation measures detailed at Section 17.8 of ES Chapter 17: Air Quality [APP-055], there will be no effects from the Scheme that could combine with effects from other sites and other developments to lead to cumulative effects. In accordance with paragraph 5.2.10 of NPS EN-1 (2011) and paragraph 5.2.12 of NPS EN-1 (November 2023) appropriate mitigation measures have been secured in relation to construction and decommissioning phase dust and potential operational phase fire incidents and no further mitigation is required to ensure the Scheme will not result in any substantial changes in air quality levels as a result of cumulative effects.

<u>Summary</u>

- 6.18.16 ES Chapter 17: Air Quality **[APP-055]** has assessed the effects of the Scheme upon air quality during construction, operation and decommissioning. The Scheme therefore complies with the requirements of paragraphs 5.2.6 and 5.2.7 of NPS EN-1 (2011) and 5.2.8 and 5.2.9 of NPS EN-1 (November 2023).
- 6.18.17 Appropriate mitigation measures have been secured in accordance with NPS EN-1 (2011) paragraphs 5.2.9 and 5.2.10 and the conclusions of ES Chapter 17: Air Quality [APP-055] are that the air quality effects are anticipated to be negligible, which is not significant in EIA terms. No cumulative effects are anticipated. As required by CLLP Policy S14 and S53, the ES shows that the impacts are acceptable on the amenity of sensitive neighbouring uses (including local residents) and the Scheme will not result in adverse impacts upon air quality from odour, fumes, smoke, dust and other sources.

6.19 Ground Conditions

- NPS EN-1 (2011) paragraph 4.10.7 and NPS EN-1(November 2023) paragraphs 4.12.14 and 4. 12.15 state the Secretary of State should be satisfied that development consent can be granted taking full account of environmental impacts. The effects of existing sources of pollution in and around the site should not be such that the cumulative effects of pollution when the proposed development is added would make the development unacceptable, particularly in relation to statutory environmental quality limits. NPS EN-1 (2011) Paragraph 4.10.8 states that the Secretary of State should not refuse consent on the basis of pollution impacts unless it has good reason to believe that any relevant necessary operational pollution control permits or licences or other consents will not subsequently be granted.
- 6.19.2 Local planning policy CLLP S56 states that site layout and drainage should take account of ground conditions. BDCSDMP Policy DM10 seeks to prevent unacceptable pollution impacts.



- 6.19.3 ES Chapter 11: Ground conditions and contamination [APP-049] provides an overview and description of the baseline conditions for the Sites and the associated Cable Route Corridor, with regards to their current and historical uses, geology, hydrogeology, hydrology and mining. Full details are included within the Preliminary Geo-Environmental Risk Assessments [APP-095 to APP-104] prepared for each area.
- 6.19.4 The baseline data has been used to develop a Conceptual Site Model which assesses whether the presence of contamination could potentially lead to significant harm via migration along a pathway to affect a receptor. This model then forms the basis of a qualitative risk assessment.
- 6.19.5 Five key receptors with plausible contamination linkages have been assessed for the Scheme across the solar array Sites and Cable Route Corridor:
 - Workers Direct contact/ingestion and inhalation of dust, vapours and asbestos fibres;
 - Adjacent site users or residents Direct contact/ingestion and inhalation of dust, vapours and asbestos fibres;
 - Controlled waters Leaching of contamination into groundwater and vertical/lateral migration through permeable deposits below the Site;
 - Future site users Direct contact/ingestion and inhalation of dust, vapours and asbestos fibres; and
 - Built Environment Direct contact between and accumulation of gas in enclosed spaces and sub-floor voids.
- 6.19.6 Based on the nature of the Cable Route Corridor comprising linear infrastructure, the works involving the ground are temporary, with the land returned to former use. Following the cable being laid. As such, the receptors involved in this work are limited to construction and decommissioning groundworkers, controlled water and the built environment.
- 6.19.7 The history of the Sites and Cable Route Corridor largely comprises agricultural land with discrete areas of development including farmyards, railways lines and West Burton Power Station in the north. The assessment shows that with the embedded mitigation outlined in the ES and the implementation of well-established good industry practices in construction for maintaining contaminated land, the potential impact of the construction and decommissioning of the Scheme are not significant in EIA terms.
- 6.19.8 The Scheme includes embedded mitigation for ground conditions and contamination in the form of a Construction Environmental Management Plan (CEMP) and Decommissioning Strategy, which will include procedures for the identification and mitigation of contaminant risks associated with the construction. An Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] form part of the application. Maintenance works will require similar mitigation measures. These



measures would ensure that the effects and risk of contamination from the Scheme, will not be significant and would enable the Secretary of State to be satisfied that development consent can be granted taking full account of environmental impacts in accordance with NPS EN-1 (2011) paragraph 4.10.7 (2011) and NPS EN-1 (November 2023) paragraphs 4.12.14 and 4.12.15. They also demonstrate that the site layout takes account of ground conditions in accordance with CLLP S56, and there will be no unacceptable pollution impacts as required by BDCSDMP Policy DM10.

- 6.19.9 ES Chapter 11: Ground conditions and contamination **[APP-049]** considers incombination effects including climate change and its potential to modify ground conditions, in which the key variable is the future change in rainfall levels. Paragraph 11.10.8 of ES Chapter 11 explains that given the likely absence of contaminated soil or groundwater, there is unlikely to be migration of contaminants which could be exacerbated by climate change.
- 6.19.10 The potential cumulative impacts of the Scheme are also considered, and Section 11.11 of ES Chapter 11: Ground conditions and contamination [APP-049] concludes that given modern methods of construction and the low sensitivity end use, the cumulative effects of the proposal in combination with other proposals (Cottam Solar Project, Gate Burton Energy Park, Tillbridge Solar) are anticipated to be negligible with the implementation of embedded mitigation measures such as the CEMP. This demonstrates that the cumulative effects of pollution would be acceptable in relation to statutory environmental quality limits as required by NPS EN-1 (2011) paragraph 4.10.7 and NPS EN-1 (November 2023) paragraphs 4.12.14 and 4.12.15. They would also ensure that the site is acceptable in terms of ground conditions in line with policy CLLP S56and there will be no unacceptable pollution impacts as required by BDCSDMP Policy DM10.

Summary

6.19.11 In summary, ES Chapter 11: Ground conditions and contamination [APP-049] demonstrates that the site is acceptable for the proposed use in respect of ground conditions and contamination. No potential significant effects, including cumulative effects, have been identified after the implementation of embedded, well-established good industry practices in construction for managing contaminated land, which will be incorporated into the CEMP and Decommissioning Strategy and utilised in all phases of the Scheme. Consent for the Scheme can, therefore, be granted in line with NPS EN-1 (2011) paragraph 4.10.7 and NPS EN-1 (November 2023) paragraphs 4.12.14 and 4.12.15. The Scheme complies with local planning policies CLLP S56 and BDCSDMP DM10 as it is shown to have taken account of ground conditions with no unacceptable pollution impacts anticipated.



7 Conclusions and Planning Balance

- 7.0.17.1.1 The Scheme will be determined pursuant to section 105 of the PA 2008 as set out at Section 1.3 above. Applications determined under this section require the SoS to have regard to: (a) any local impact report; (b) matters prescribed in relation to development of the description to which the application relates; and (c) any other matters which the SoS considers to be both important and relevant. This Planning Statement provides evidence of the Scheme's compliance with the relevant prescribed matters and relevant planning policy and other matters the Applicant considers are likely to be important and relevant, to inform the SoS's decision as to whether to grant a DCO for the Scheme.
- 7.0.27.1.2 There are no specific references to solar NSIPs in NPS EN-1 (2011), although the Energy NPSs of November 2023 were designated on 17 January 2024, and new applications for solar NSIPs are required to be determined in accordance with the designated November 2023 versions of NPS EN-1 and NPS EN-3. It is expected that the SoS will consider the November 2023 NPSs as important and relevant matters in their decision. The November 2023 NPSs have been prepared in light of up-to-date government policy and commitments relating to energy and decarbonisation.
- 7.0.37.1.3 Although solar NSIPs are not specifically identified in the 2011 Energy NPSs the Applicant considers that significant weight should be given to the Scheme's compliance with the policies of the 2011 Energy NPSs and the November 2023 Energy NPSs, with greater weight attributed to the November 2023 NPSs following their designation, and with less weight given to the NPPF and local planning policy, owing to their focus on guiding development at regional and local levels.
- 7.0.47.1.4 The 2011 Energy NPSs, November 2023 Energy NPSs, and other national energy policy set out the Government's aims to provide secure and affordable energy supplies whilst decarbonising the energy system. This is in order to enable the UK to achieve its legally binding commitment to reduce carbon emissions and achieve net zero carbon emissions by 2050; as well as provide a resilient and low-cost energy network for the future. The Government recognises that the need to deliver these aims and commitments is immediate and therefore renewable energy NSIPs, including large scale solar projects, need to be delivered urgently.
- 7.0.57.1.5 The Scheme will deliver these policy aims, providing a significant amount of low carbon electricity over its lifetime; and providing resilience, security and affordability of supplies due to its large scale and proposed integration of BESS. It will therefore be a critical part of the national portfolio of renewable energy generation that is required to decarbonise the country's energy supply quickly whilst providing security and affordability of national energy supply. It is clear that there is a compelling case for the need for the Scheme and that it will deliver national economic and social benefits in line with the Government's wider objectives of delivering sustainable development.



- 7.0.67.1.6 The Scheme will also deliver other more localised economic, social and environmental benefits. These relate to biodiversity net gain, employment creation and permissive path creation. With the exception of employment, these have been as a result of the choice of location for the Scheme and the iterative design process which the Applicant has undertaken.
- 7.0.77.1.7 The gross value added (GVA) to the economy of workers employed in the construction of the Scheme is expected to be £13.3 million, of which £7.7 million will be of benefit to the local economy within the combined areas of Bassetlaw District and West Lindsey District. The operation and maintenance of the Scheme is anticipated to generate a net uplift to Gross Value Added of £500,000 per annum with £300,000 of this to the benefit to the local economy within the combined areas of Bassetlaw District and West Lindsey District. With regard to biodiversity, the Scheme is expected to deliver an exemplary project with an anticipated biodiversity net gain of 86.8% provided in habitat, 54.71% gains in hedgerow and 33.25% gains in river units, in exceedance of local and national planning policies. A new permissive path will be created from the track off Sykes Lane along the Codder Lane Belt and then south and west to rejoin Sykes Lane opposite Hardwick Scrub, which will enhance local connectivity and recreational opportunities.
- 7.0.87.1.8 NPS EN-1 (2011) paragraph 4.1.2 sets a presumption in favour of granting permission for energy NSIP projects. This is carried through to NPS EN-1 (November 2023) at paragraphs 4.1.2, 4.1.3 and 4.1.4. NPS EN-1 (2011) paragraph 3.2.3, and NPS EN-1 (November 2023) paragraphs 3.1.1 and 3.1.2, acknowledge that it will not be possible to develop the necessary amounts of such infrastructure without some significant residual adverse impacts.
- 7.0.97.1.9 The analysis of planning policy compliance demonstrates that the need for the Scheme is supported by planning policy and other national energy and environmental policy and that the Scheme addresses relevant national and local planning policies through its design, avoiding and minimising adverse impacts where possible.
- 7.0.107.1.10 With the mitigation proposed, the ES demonstrates that the Scheme will not have any significant adverse effects in relation to hydrology, flood risk and drainage; ground conditions and contamination; minerals; transport and access; noise and vibration; glint and glare; air quality; and major accidents and disasters. In relation to climate change and soils and agriculture, it will have a major beneficial effect. It is, however, acknowledged that the Scheme will result in residual significant adverse effects in EIA terms upon landscape and views; ecology (at a site and local level); heritage; socio-economics, tourism and recreation; waste and human health (albeit temporary construction traffic impacts and short-term temporary closure of the Trent Valley Way, and National Byways long distance recreational routes). All of these effects (apart from effects upon non-designated archaeological remains and the rise in energy sector employment opportunities at Scheme decommissioning)



will only occur while the Scheme is under construction, operational or being decommissioned and are therefore limited to the lifetime of the Scheme.

The significant residual impacts upon one designated heritage asset (The medieval 7.0.117.1.11 bishop's palace and deer park, Stow Park (NHLE 1019229) are concluded within the Heritage Assessment, Appendix 13.5 [APP-117 to APP-119] to be less than substantial, albeit the impacts are anticipated to be towards the upper end of this scale. In accordance with NPS EN-1 (2011) paragraph 5.8.12, NPS EN-1 (2023) paragraphs 5.9.27-31, CLLP Policy S57 and DBLP Policy 43, this less than substantial harm should be weighed against the public benefits of the proposal, acknowledging that the Secretary of State should give great weight to the asset's conservation and give considerable importance and weight to the desirability of preserving all heritage assets with any harm or loss of significance require clear and convincing justification. In this case, the significant public benefits of the Scheme in terms of renewable energy generation which is urgently needed to create a secure and affordable energy system and to help combat climate change, as set out at Section 4.0 of this Planning Statement, clearly and demonstrably outweigh the reversible, less than substantial harm to the affected designated heritage asset.

7.0.12 With regard to landscape and visual amenity the Applicant has carefully designed the Scheme to ensure landscape and visual impacts are minimised through sensitive siting of the largest Scheme components in the most well screened areas of the Site and a green infrastructure led landscape and ecological design. In accordance with NPS EN-1 (2011) paragraphs 5.9.15, and NPS EN-1 (November 2023) paragraphs 5.10.35, it is considered that the significant adverse residual visual effects of the Scheme to ten non-designated heritage assets are clearly and comprehensively outweighed by the benefits of the Scheme set out at Section 4 of the Planning Statement, in terms of delivering renewable energy infrastructure. The fact that the adverse effects are localised; will be reversed following decommissioning at the end of the Scheme's operational life; and that NPS EN-1 (2011) and NPS EN-1 (November 2023) acknowledge that adverse effects are likely, given the scale of energy NSIPs, also weigh in favour of the Scheme.

7.0.13 The Scheme successfully minimises impacts upon BMV land in accordance with NPS EN-1 (2011) paragraph 5.10.8 and NPS EN-1 (November 2023) paragraphs 5.11.12 to 5.11.14 and 5.11.18 with the majority of the site (73.76%) not being located on BMV agricultural land. In respect of the inclusion of some BMV agricultural land within the Order Limits, the Applicant has explained that this is justified by other sustainability considerations, including the need to maximise the amount of low carbon electricity generated by the Scheme and the particular opportunities and constraints offered by some of the areas of BMV land. The impacts on BMV land have been minimised by the nature of the Scheme and its design, including the management of soil resource during the life of the Scheme.

7.0.14 As described in Section 6 of this Planning Statement, whilst it has not been possible to avoid all impacts, these have been minimised, where possible, through careful





and sensitive design and detailed mitigation strategies secured through this DCO Application. When considered against the NPS and NPPF, the Scheme accords with relevant policies, and with regard to specific policy tests, the national and local benefits of the Scheme are considered on balance to outweigh its adverse impacts. The Scheme is also considered to be broadly consistent with relevant local planning policy. Therefore, it is considered that development consent for the Scheme should be granted.

West Burton Solar Project

Planning Statement Appendix A: Planning Application History Search West Burton Sites Revision CD

Prepared by: Lanpro Services

April May 2024

PINS reference: EN010132

Document reference: **EX6**EX7/WB7.5_**CD**

APFP Regulation 5(2)(q)





Contents

<u>1</u>	INTRODUCTION	3
<u>2</u>	PLANNING APPLICATIONS	3
3	PLANNING APPLICATIONS (ADJACENT)	4



Issue Sheet

Report Prepared for: West Burton Solar Project Ltd. Planning Statement

Appendix A: Planning Application History Search West Burton Sites

Prepared by:

Name: Gabriel Cooper

Title: Graduate Planner

Approved by:

Name: Ian Douglass

Title: Director of Planning

Date: March 2023

Revision: 01

Revision	Date	Prepared by:	Approved by:
А	9 January 2024	GV	JC
В	28 February 2024	SF	JC
С	30 April 2024	SF	JC
D	8 May 2024	<u>SF</u>	<u>JC</u>



1 Introduction

1.1.1 The planning application history information contained within this document has been compiled using the application search function on the West Lindsey District Council and Bassetlaw District Council websites.

2 Planning Applications

Table 1: West Burton 1

Reference & Date	Description	Decision
	No Relevant onsite history	

Table 2: West Burton 2

Reference & Date	Description	Decision
145936 - 28/11/2022	Request for a scoping opinion to erect 2no. broiler poultry units with feed silos, ancillary biomass boiler heating system, roof mounted PV panels & hardstanding.	12/01/2023 – EIA Required
145441 - 19/08/2022	Request for a screening opinion to erect 1no. broiler poultry unit.	08/09/2022 - EIA Required
143894 - 25/10/2021	Request for confirmation of compliance with condition 2 of planning permission 143040 granted 2 August 2021.	25/04/2022 – Condition Discharged
143040 - 12/05/2021	Planning application for erection of 1no. poultry unit for meat production, including feed silos, hardstanding and ancillary buildings	02/08/2021 - Granted time limit plus conditions
141816 - 14/10/2020	Request for confirmation of compliance with condition 5 of planning permission 140380 granted 8 April 2020.	01/12/2020 - Granted without conditions
141299 - 01/07/2020	Request for confirmation of compliance with conditions 2,3 and 6 of planning permission 140380 granted 8 April 2020	20/08/2020 - Granted without conditions
140380 - 12/12/2019	Planning application for erection of 1no. poultry unit for meat production, ancillary buildings, hardstanding and access.	08/04/2020 - Granted time limit plus conditions
129832 - 03/04/2013	Request for confirmation of compliance with conditions 2 and 3 of planning permission 129433 granted	08/05/2013 - Condition discharged
129757 - 04/03/2013	Overhead Lines - Rebuild an existing 11000 kilovolt line from Saxilby to Ingleby	16/05/2013 - Issued



129433 - 10/12/2012	Planning application for a proposed	08/02/2013 -
	change of use of agricultural buildings to	Granted time limit
	premises for horse livery, including tack	plus conditions
	rooms, office, stables and indoor exercise	
	area, outdoor manege and horsewalker.	
	area, outdoor manege and norsewarker.	

Table 3: West Burton 3

Reference & Date	Description	Decision
142498 - 18/02/2021	Application for prior notification of agricultural or forestry development - proposed road.	25/03/2021 - Prior Approval Not Required
130238 - 15/07/2013	Planning application to vary condition 7 of planning permission 129135 granted 3 January 2013 - relocation of the dwelling siting	17/09/2013 - Granted time limit plus conditions
130230 - 12/07/2013	Request for confirmation of compliance with conditions 1 - 10 of planning permission 129135 granted 3 January 2013	03/01/2013 - Condition discharged
W103/722/82/CM - 17/12/82	Planning permission to drill an exploratory borehole for coal	Conditional Consent

3 Planning Applications (Adjacent)

Table 4: West Burton 1

Reference & Date	Description	Decision
138615 - 14/11/2018	Listed building consent for a new first floor window to the West elevation.	23/01/2019 - Granted time limit plus conditions
138614 - 14/11/2018	Planning application for a new first floor window to the West elevation.	23/01/2019 - Granted time limit plus conditions
129135 - 10/09/2012	Planning application for demolition of existing derelict farmhouse and farm buildings and construction of new farm buildings and farmhouse. Upgrade existing farm road and access.	03/01/2013 - Granted time limit plus conditions
120969 - 28/08/2007	Planning Application for provision of manege with associated access and car parking, change of use of agricultural building to allow for stabling and the	21/02/2008 - Withdrawn by Applicant



	provision of 2no. holiday let properties with associated access road.	
147501- 26/10/2023	Planning application for the retention of a field access.	26/01/2024 - Granted time limit plus conditions
147528 - 07/12/2023	Planning application for the retention of containers used for storage purposes.	19/01/2024 - Refused
147512 – 20/11/2023	Planning application for the change of use of land for the retention of a static caravan for the use as AirBnB accommodation.	30/01/2024 - Refused

Table 5: West Burton 2

Reference & Date	Description	Decision
147094 – 25/07/2023	Application for prior notification to erect agricultural storage building.	16/08/2023 – Prior Approval Not Required
146171 - 23/01/2023	Request for confirmation of compliance with condition 2(method statement) of planning permission 141263 granted 15 October 2020.	31/01/2023 Conditions discharged
143488 - 30/07/2021	Planning application for alterations and extension to existing dwelling and the creation of a linked family annexe.	13/10/2021 - Granted time limit plus conditions
143040 - 12/05/2021	Planning application for erection of 1no. poultry unit for meat production, including feed silos, hardstanding and ancillary buildings	02/08/2021 - Granted time limit plus conditions
142044 - 23/11/2020	Planning application to change the use of agricultural land to a wildlife area and for the construction of wetland/pond being an application to vary condition 2 of planning permission 140167 granted 18 December 2019 - relocation of proposed wetland/pond	29/01/2021 - Granted time limit plus conditions
141464 - 05/08/2020	Prior approval for proposed change of use from light industrial to dwelling house.	24/09/2020 - Refused
141299 - 01/07/2020	Request for confirmation of compliance with conditions 2,3 and 6 of planning permission 140380 granted 8 April 2020	20/08/2020 - Granted without conditions



141263 - 24/06/2020	Planning application for erection of 1no. poultry rearing unit with ancillary feed silos, hardstanding and access.	15/10/2020 - Granted time limit plus conditions
140380 - 12/12/2019	Planning application for erection of 1no. poultry unit for meat production, ancillary buildings, hardstanding and access.	08/04/2020 - Granted time limit plus conditions
139239 - 15/04/2019	Application for non-material amendment to planning permission 138856 granted 15 March 2019 - Change of roof design to a flat roof with skylight	25/04/2019 - Granted without conditions
138856 - 06/01/2019	Planning application to replace conservatory with a larger conservatory to the side elevation.	15/03/2019 - Granted time limit plus conditions
137919 - 05/06/2018	Request for a screening opinion to erect 4no. poultry units	25/06/2018 - EIA Required
137920 - 05/06/2018	Request for a scoping opinion to erect 4no. poultry units	11/07/2018 – Issued
129600 - 04/01/2013	Request for confirmation of compliance with conditions 2 and 3 of planning permission 129395 granted 9 January 2013	14/02/2013 - Condition discharged
129395 - 22/11/2012	Planning application for extension to existing residential care home and alteration to existing link entrance replace existing timber door with timber window.	09/01/2013 - Granted time limit plus conditions
122755 - 28/07/2008	Planning Application to erect grain store/vehicle storage building.	13/10/2008 - Granted time limit plus conditions
M06/P/1217 - 21/11/2006	Planning application to convert existing garage to sitting room and erect extension to form new garage	10/01/2007 - Granted time limit plus conditions

Table 6: West Burton 3

Reference & Date	Description	Decision
147720 - 02/01/2024	Listed building consent for installation of	15/03/2024 -
	solar panels to roof of the cart garages.	Granted time limit
		plus conditions
147478 - 19/10/2023	Planning application for extension and	19/12/2023 -
	alterations, with change of use of land to	Granted time limit
	domestic curtilage.	plus conditions
146684 - 17/05/2023	Application for lawful development	04/07/2023 -
	certificate for existing outbuildings B, C	Refused.



	and D for commercial workshops and	
	storage space for mechanical fixings.	
143519 - 05/08/2021	Planning application for alterations and	22/09/2021 -
	extension to existing dwelling.	Granted time limit
		plus conditions
142901 - 21/04/2021	Planning application for extension to	06/07/2021 -
	existing dwelling to form first floor	Granted time limit
	accommodation including dormer	plus conditions
	windows and the erection of a side and	
4.44.674 4.6.400.40000	rear extension.	0.4/4.4/0.000
141671 - 16/09/2020	Application for advertisement consent for	04/11/2020 -
	1no. coated aluminum business sign to the	Granted time limit
4.40762 42.402.42020	side of dwelling	plus conditions
140762 - 12/03/2020	Planning application for the erection of a	28/05/2020 – Granted time limit
	barn stable being the removal of condition 5 of planning permission 132960 granted	plus conditions
	18 August2015 - removal for use from	pius conditions
	recreational hobby purposes to	
	commercial business	
139537 - 03/06/2019	Application for lawful development	25/07/2019 -
	certificate for single storey side extension	Granted without
		conditions
138374 - 20/09/2018	Planning application for proposed first	02/11/2018 -
	floor extension and internal alterations	Granted time limit
		plus conditions
136555 - 14/07/2017	Notification under Electricity Act 1989	26/07/2017 -
	Overhead Lines Exemption Regulations	No observation/
	2009 - reference ENQ5354212 - erect 2no.	objections
	wooden poles.	
134344 - 25/04/2016	Planning application for proposed lounge	15/06/2016 -
	extension	Granted time limit
424222 4040442046	Dispusion and institute for the disputalistic of	plus conditions
134323 - 19/04/2016	Planning application for the installation of	01/07/2016 - Granted time limit
	pressurised oil pipeline	plus conditions
128629 - 23/04/2012	Application for prior approval of proposed	19/06/2012 - No
120025 2510712012	works to railway bridge, Stow Park, Sturton	observation/
	by Stow	objections
128387 - 05/03/2012	Request for confirmation of compliance	20/03/2012 -
	with condition 2 of planning permission	Condition
	128116 granted 06 February 2012	discharged
126121 - 30/06/2010	Agricultural Determination to construct a	28/07/2010 - Prior
	polytunnel no. 2.	Approval required
126120 - 30/06/2010	Agricultural Determination to construct a	28/07/2010 - Prior
	polytunnel no. 1.	Approval Not
		Required



Planning application for change of use of	26/11/2010 -
land to accommodate natural burial.	Withdrawn by
	Applicant
0	28/05/2010 -
	Granted time limit
of 125459	plus conditions
Planning application for the construction of	25/03/2010 -
a workshop-garage	Refused
Planning application for proposed	17/01/2008 -
temporary dwelling.	Granted time limit
	plus conditions
Deemed Consent Hazardous substances	22/12/2006 -
	Deemed Approved
Planning Application to erect single storey	04/10/2006 -
brick extension with slate roof comprising	Granted time limit
of kitchen, bathroom and garage	plus conditions
Outline planning application to erect one	13/10/2006 -
bungalow	Refused and
	Appeal Dismissed
Planning Application to erect single storey	16/05/2006 -
side extension	Granted time limit
	plus conditions
Planning Application to erect conservatory	21/03/2006 -
	Granted Time
	Limit Cond only
Planning Application to erect precast	01/11/2005 -
concrete garage	Granted Time
	Limit Cond only
Agricultural Determination to erect farm	23/03/2005 -
workshop/stock shed	Planning
	Permission
	Required
	land to accommodate natural burial. Planning application for the construction of a workshop-double garage - resubmission of 125459 Planning application for the construction of a workshop-garage Planning application for proposed temporary dwelling. Deemed Consent Hazardous substances Planning Application to erect single storey brick extension with slate roof comprising of kitchen, bathroom and garage Outline planning application to erect one bungalow Planning Application to erect single storey side extension Planning Application to erect conservatory Planning Application to erect precast concrete garage Agricultural Determination to erect farm

West Burton Solar Project

Planning Statement Appendix B:

Planning Application History Search Cable Route Corridor Revision CD

Prepared by: Lanpro Services

April May 2024

PINS reference: EN010132

Document reference: **EX6EX7**/WB7.5_**CD**

APFP Regulation 5(2)(q)







Contents

<u>1</u>	INTRODUCTION	3
<u>2</u>	PLANNING APPLICATIONS	3
3	PLANNING APPLICATIONS (ADJACENT)	5



Issue Sheet

Report Prepared for: West Burton Solar Project Ltd. Planning Statement

Appendix B: Planning Application History Search Cable Route Corridor

Prepared by:

Name: Gabriel Cooper

Title: Graduate Planner

Approved by:

Name: Ian Douglass

Title: Director of Planning

Date: March 2023

Revision: 01

Revision	Date	Prepared by:	Approved by:
А	9 January 2024	GV	JC
В	28 February 2024	SF	JC
С	30 April 2024	SF	JC
D	8 May 2024	<u>SF</u>	<u>JC</u>



1 Introduction

1.1.1 The planning application history information contained within this document has been compiled using the application search function on the West Lindsey District Council and Bassetlaw District Council websites.

2 Planning Applications

Table 1: Cable Route Corridor

Reference & Date	Description	Decision
147710 – 06/12/2023	Request for a scoping opinion for proposed solar farm development.	25/01/2024 – EIA Required
145882 - 18/11/2022	Planning application to erect 2no. agricultural storage buildings.	18/01/2023 - Granted time limit plus conditions
22/00831/SCR – 17/06/2022	Screening Opinion - Demolition of Power Station	08/07/2022 – EIA Required
22/00047/CDM – 10/01/2022	Variation of the trigger date of conditions 67 and 68 to 31 December 2024 to afford sufficient time for additional surveys, to secure all necessary approvals under non-planning regimes and implementation works to take place prior to extraction recommencing	25/01/2022 – Decided, No Objection.
20/00605/CDM – 26/05/2020	Planning Application for non Compliance with Condition 68 of Planning Consent 1/16/00354/CDM to Defer the Submission of a Revised Restoration Scheme for a Period of Two Years to 15th April 2022.	15/07/2020 – Decided, No Objection.
19/01236/HAZ – 18/09/2019	Hazardous Substances Consent for the Storage of Various Hazardous Substances - Propane and Oxygen Bottle Store to be Relocated	29/10/2019 - Grant
16/01262/HAZ – 05/09/2016	Hazardous Substances Consent for the Storage of Various Hazardous Substances	22/10/2016 – Grant
16/00354/CDM – 07/04/2016	Vary Conditions 8 and 11 of Planning Permission 46/11/00002/R to enable the Quarry Access Road to be Constructed in Two Stages. The Initial Stage Incorporates the Construction of 500m Section of Bound Surface adjacent to	23/04/2016 – Decided, No Objection



	Gainsborough Road which shall be used for the Removal of the Remaining Mineral in the Permitted Reserve.	
15/00829/SCR – 22/06/2015	Screening Opinion - Erect a Single Wind Turbine Generator of up to 500 kw With a Hub Height of up to 75m and Diameter of up to 54m (Total Height to Blade Tip 102m), Substation and Transformer Cubicle, Electrical Infrastructure, Access Track, Crane Hardstanding, Temporary Meteorological Mast and Temporary Construction Compound	06/07/2015 – EIA Required
12/09/00001 – 03/03/2009	ERECTION OF 12 WIND TURBINES, WITH A MAXIMUM HEIGHT OF 145mtrs, 1 X 100mtr HIGH METEOROLOGICAL MAST, SUBSTATION, TEMPORARY CONSTRUCTION COMPOUND, ACCESS TRACKS AND ASSOCIATED SITE INFRASTRUCTURE	29/07/2010 – Refused and Appeal Dismissed
52/00/00001 - 05/01/2000	CONSTRUCTION OF ASH PROCESSING PLANT	17/08/2000 – Granted
52/99/00002 - 07/05/1999	DEMOLISH EXISTING PORTABLE BUILDING AND CONSTRUCT NEW STUDY CENTRE	03/06/1999 – Grant
52/99/00001 - 02/02/1999	EXTEND WEST BURTON COAL-FIRED POWER STATION	19/08/1999 – Grant
52/97/00002 - 15/09/1997	ERECT NEW GATEHOUSE AND WEIGHBRIDGE	26/11/1997 – Grant
52/96/00001 - 10/10/1996	ERECT OFFICE ACCOMMODATION	19/11/1996 – Granted
52/05/00002 - 19/08/1995	CONSTRUCT A BIOMASS FUEL MATERIAL HANDLING FACILITY, COMPRISING OF STORAGE BUILDING AND CONVERYING SYSTEM TO EXISTING COAL CONVERYORS	07/11/2005 – Grant
52/95/00001 - 11/01/1995	ERECT TWO SILOS AND ASSOCIATED PLANT FOR PROCESSING OF ASH RESIDUE	03/04/1995 – Grant
52/94/00003 - 07/12/1994	ERECT SINGLE STOREY OFFICES	31/01/1995 – Grant



52/94/00002 - 01/11/1994	EXTENSION TO EXISTING SUBSTATION AND CABLE SEALING END COMPOUND	29/12/1994 – Grant
52/93/00001 – 12/07/1993	CHANGE SUB-STATION ROOF FROM FLAT TO SHALLOW PITCH PROFILED STEEL SHEET ROOF	25/08/1993 – Grant
52/91/00002 – 29/08/1991	DIVERT OVERHEAD POWER LINES	07/11/1991 – Granted
52/89/00001 – 04/12/1989	CLAY PIGEON SHOOTING CLUB	04/01/1990 – Decided, Permission not Required

3 Planning Applications (Adjacent)

Table 2: Cable Route Corridor

Reference & Date	Description	Decision
23/01260/FUL – 16/10/2023	Erect a temporary office building for a period of 5 years.	08/12/2023 – Granted time limit plus conditions
23/01202/COU – 02/10/2023	Change of use of land for caravan storage	05/03/2024 - Withdrawn
144739 - 06/04/2022	Planning application to rebuild existing structure with insulated sip panels and insulated roof.	13/10/2022 - Granted time limit plus conditions
132960 - 29/04/2015	Planning application for the erection of a barn stable	18/08/2015 - Granted time limit plus conditions
13/00423/FUL – 15/05/2013	Erection Of Three 145m (Tip Height) Wind Turbines With Associated Infrastructure	04/12/2014 - Finally Disposed Of
128933 - 09/07/2012	Outline planning application with all matters reserved for erection of two bungalows.	28/08/2012 – Refused and Appeal Dismissed
46/04/00003 – 16/05/2004	ERECT BUILDING AS COVER TO SILAGE CLAMP AND STRAW STORE	17/06/2004 – Granted.
35/03/00008 – 02/07/2003	ERECT AGRICULTURAL COLD STORE AND LOADING BUILDING	11/09/2003 - Granted
35/02/00007 – 26/02/2002	CHANGE OF USE FROM AGRICULTURAL LAND TO LEISURE/SPORTS USE	20/08/2002 – Granted



35/00/00006 – 08/05/2000	CONSTRUCT AGRICULTURAL BUILDING FOR GRADING AND STORAGE OF PRODUCE	03/07/2000 – Granted
35/94/00007 – 02/06/1994	ERECT ONION/POTATO STORE	21/07/1994 - Granted
12/93/00001/F – 22/02/1993	ERECTION OF POTATO STORE AND GRADING AREA	01/04/1993 – Decided. Application not required.
W62/300/80 - 21/04/80	Erect building for storage purposes.	Unconditional Consent
51/80/00001 - 02/01/1980	INSTALLATION OF OIL PIPELINE	19/02/1980 – Decided, No Objection.
35/08/00002/P	Wind Energy Development	Unknown
23/00485/DEM	Prior Notification of Proposed Site Clearance of West Burton A Power Station	26/10/2023 – Granted

West Burton Solar Project

Planning Statement Appendix C: National Policy Accordance Table Revision CD

Prepared by: Lanpro Services

April May 2024

PINS reference: EN010132

Document reference: **EX6EX7**/WB7.5_**CD**

APFP Regulation 5(2)(q)





Contents

<u>1</u>	NATIONAL POLICY ACCORDANCE TABLE	1
1.1	Table 1: National Policy Statement EN-1 (2011)	1
1.2	TABLE 2: NATIONAL POLICY STATEMENT EN-3 (2011)	92
1.3	TABLE 3: NATIONAL POLICY STATEMENT EN-5 (2011)	94
1.4	TABLE 4: NATIONAL POLICY STATEMENT EN-1 (NOVEMBER 2023)	100
1.5	TABLE 5: NATIONAL POLICY STATEMENT EN-3 (NOVEMBER 2023)	160
1.6	TABLE 6: NATIONAL PLANNING POLICY FRAMEWORK	185
1.7	TABLE 7: NSIP ACTION PLAN (FEBRUARY 2023)	186
1.8	TABLE 8: POWERING UP BRITAIN ENERGY SECURITY PLAN (MARCH 2023)	187



Issue Sheet

Report Prepared for: West Burton Solar Project Ltd. Planning Statement

Appendix C National Policy Accordance Table

Prepared by:

Name: Gabriel Cooper

Title: Graduate Planner

Approved by:

Name: Ian Douglass

Title: Director of Planning

Date: March 2023

Revision: 01

Revision	Date	Prepared by:	Approved by:
Α	9 January 2024	GV	JC
В	28 February 2024	GV	JC
С	30 April 2024	BR	JC
D	8 May 2024	<u>SF</u>	<u>JC</u>



1 National Policy Accordance Table

1.1 Table 1: National Policy Statement EN-1 (2011)

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
emissions by at least 80% by 2050, compared to 1990 levels. Analysis done on possible 2050 pathways shows that moving to a secure, low carbon energy system is challenging, but achievable. It requires major investment in new technologies to		The Scheme will quickly deliver significant amounts of low carbon power. Solar is also relatively quick to construct compared to other technologies which have longer construction timeframes or have potentially not yet been proven at scale.
	transport, prioritisation of sustainable bioenergy and cleaner power generation.	The Applicant, as a private sector organisation, has developed proposals for the Scheme, which will be a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to decarbonise, with solar technology supported by recent government policy. Its proposed National Electricity Transmission System (NETS) connection means that it would play its part in helping National Grid ESO (NGESO) manage the national electricity system to ensure security of supply and bring cost benefits to electricity consumers, both of which are identified in government policy as being required for resilient energy supplies in the future.
Paragraph 2.2.2	The Government is working to ensure their efforts produce the major, rapid change the UK needs. Within a market-based system and with severe constraints on public	Section 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] considers the Scheme in



	is clear. It should be on developing a clear, long-term policy framework which facilitates investment in the necessary new infrastructure (by the private sector) and in energy efficiency.	the context of policy setting out the need for renewable energy development.
		The Statement of Need [APP-320] presents a detailed compelling case for why the Scheme is urgently required and at the scale proposed.
		The Applicant, as a private sector organisation, has developed proposals for the Scheme, which will be a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to decarbonise, with solar technology supported by recent government policy. Its proposed National Electricity Transmission System (NETS) connection means that it would play its part in helping National Grid ESO (NGESO) manage the national electricity system to ensure security of supply and bring cost benefits to electricity consumers, both of which are identified in government policy as being required for resilient energy supplies in the future. The Scheme will quickly deliver significant amounts of low
		carbon power. Solar is also relatively quick to construct compared to other technologies which have longer construction timeframes or have potentially not yet been proven at scale.
Paragraph 2.2.6	The UK needs to wean itself off such a high carbon energy mix: to reduce greenhouse gas emissions, and to improve the security, availability and affordability of energy through diversification. Under some of the illustrative 2050 pathways, electricity generation would need to be virtually emission-free, given that we would expect some	Section 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] considers the Scheme in



emissions from industrial and agricultural processes, transport and waste to persist. By 2050, we can expect that fossil fuels will be scarcer, but will still be in demand, and that prices will therefore be far higher. Further, the UK's own oil and gas resources will be depleting and, worldwide, the costs and risks of extracting oil in particular will increase.

the context of policy setting out the need for renewable energy development.

As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to decarbonise with solar technology supported by recent government policy.

Chapter 7 Climate change of the ES [REP1-012] presents a lifecycle greenhouse gas (GHG) impact assessment which considers the impact of GHG emissions arising over the lifetime of the Scheme on the climate. This concludes that the assessed 40-year operational period of the Scheme will produce 21,956,988 MWh of electricity with an average operational greenhouse gas intensity of 7.72 grams of carbon dioxide equivalent per kWh (gCO2e/kWh). This demonstrates its very low carbon attributes compared to other non-renewable forms of electricity generation, providing an overall major beneficial impact in relation to the UK meeting its carbon reduction targets and therefore represents a major beneficial effect on the climate.

The Scheme will make a contribution to the delivery of, secure, affordable, low carbon electricity generation infrastructure, in direct accordance with the Government's objectives.



Paragraph 2.2.20	 It is critical that the UK continues to have secure and reliable supplies of electricity as we make the transition to a low carbon economy. To manage the risks to achieving security of supply we need: sufficient electricity capacity (including a greater proportion of low carbon generation) to meet demand at all times. Electricity cannot be stored so demand for it must be simultaneously and continuously met by its supply. This requires a safety margin of spare capacity to accommodate unforeseen fluctuations in supply or demand; reliable associated supply chains (for example fuel for power stations) to meet demand as it arises; a diverse mix of technologies and fuels, so that we do not rely on any one technology or fuel. Diversity can be achieved through the use of different technologies and multiple supply routes (for example, primary fuels imported from a wide range of countries); and there should be effective price signals, so that market participants have sufficient incentives to react in a timely way to minimise imbalances between supply and demand. 	The Statement of Need [APP-320] presents a detailed compelling case for why the Scheme is urgently required and at the scale proposed. This is also summarised in Section 4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD]. The Applicant, as a private sector organisation, has developed proposals for the Scheme, which will be a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to decarbonise, with solar technology supported by recent government policy. Its proposed National Electricity Transmission System (NETS) connection means that it would play its part in helping National Grid ESO (NGESO) manage the national electricity system to ensure security of supply and bring cost benefits to electricity consumers, both of which are identified in government policy as being required for resilient energy supplies in the future.
Paragraph 2.2.22	Looking further ahead, the 2050 pathways show that the need to electrify large parts of the industrial and domestic heat and transport sectors could double demand for electricity over the next forty years. It makes sense to switch to electricity where practical, as electricity can be used for a wide range of activities (often with better efficiency than other fuels) and can, to a large extent, be scaled up to meet demand. To meet emissions targets, the electricity being consumed will need to be almost exclusively from low carbon sources. Contrast this with the first quarter of 2011, when around 75% of our electricity was supplied by burning gas and coal.	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the meaningful and timely contributions offered by the Scheme to UK decarbonisation and security of supply, while helping lower bills for consumers throughout its operational life, will be critical on the path to Net Zero. Without the Scheme, a significant and vital opportunity to develop a large-scale low-carbon generation scheme will have been



		passed over, increasing materially the risk that future Carbon Budgets and Net Zero 2050 will not be achieved. The Scheme will make a sizeable contribution to meeting the Government's decarbonisation commitments, in direct accordance with this policy.
Paragraph 2.2.23	The UK must therefore reduce over time its dependence on fossil fuels, particularly unabated combustion. The Government plans to do this by improving energy efficiency and pursuing its objectives for renewables, nuclear power and carbon capture and storage. However, some fossil fuels will still be needed during the transition to a low carbon economy.	The Scheme will be part of the solution to moving the UK's energy mix away from fossil fuels towards renewable energy and it will make a sizeable contribution to meeting the Government's decarbonisation commitments, in direct accordance with this policy.
Paragraph 3.1.1	The UK needs all the types of energy infrastructure covered by this NPS in order to achieve energy security at the same time as dramatically reducing greenhouse gas emissions.	Although solar is not included within the scope of the current NPS, this is because at that time they were published (2011) it was not proven at scale. However, 10 years on, large scale solar generation is economically and technically viable. As a result, solar NSIP developments are included in the Draft Energy NPSs that the Government published for consultation on 6 September 2021. The 2011 NPSs and the 2021 Draft NPSs are therefore considered to be important and relevant to the Scheme and are likely to form the primary decision-making framework.
Paragraph 3.1.2	It is for industry to propose new energy infrastructure projects within the strategic framework set by Government. The Government does not consider it appropriate for planning policy to set targets for or limits on different technologies.	As explained in the Statement of Need [APP-320], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable and low carbon electricity generation system which is sufficient to meet future demand.



Paragraph 3.1.3	The IPC should therefore assess all applications for development consent for the types of infrastructure covered by the energy NPSs on the basis that the Government has demonstrated that there is a need for those types of infrastructure and that the scale and urgency of that need is as described for each of them in this Part.	Whilst solar generation schemes are not directly covered within the 2011 Energy NPSs, this is because at the time they were prepared solar technology was not considered viable at NSIP scale. The Applicant considers that NPS EN-1 should be an important and relevant matter in consideration of the Scheme and further notes that the scope of Draft NPS EN-3 includes solar generation schemes. The Applicant therefore considers that the SoS should assess the application on the basis that the need for the Scheme has been demonstrated. In any case, the Statement of Need [APP-320] sets out a compelling case for the need for the Scheme.
Paragraph 3.1.4	The IPC should give substantial weight to the contribution which projects would make towards satisfying this need when considering applications for development consent under the Planning Act 2008.	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to decarbonise with solar technology supported by recent government policy. The meaningful and timely contributions offered by the Scheme to UK decarbonisation and security of supply, while helping lower bills for consumers throughout its operational life, will be critical on the path to Net Zero. Without the Scheme, a significant and vital opportunity to develop a large-scale low-carbon generation scheme will have been passed over, increasing materially the risk that future Carbon Budgets and Net Zero 2050 will not be



		achieved. This contribution is considered to be an important and relevant matter which should be given substantial weight in the SoS decision, as per this policy.
Paragraph 3.2.2	As we move towards 2050 the ways in which we use energy will be transformed. We need to become less dependent on some forms of energy, as new and innovative low carbon technologies and energy efficiency measures are taken up. We also shall become more dependent on others – for example, demand for electricity will increase if we electrify large parts of transport, heating and industry	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable and low carbon electricity generation system which is sufficient to meet future demand. The Government expects solar technology to play a major role in delivery of these objectives.
		Furthermore, as explained in the Statement of Need [APP-320], demand for electricity is predicted to increase by 2050 by many expert projections. For example, paragraph 2.2.22 of NPS EN-1 predicts that 'the need to electrify large parts of the industrial and domestic heat and transport sectors could double demand for electricity over the next forty years. The Scheme responds to this urgent and increasing demand for electricity.
Paragraph 3.2.3	This Part of the NPS explains why the Government considers that, without significant amounts of new large-scale energy infrastructure, the objectives of its energy and climate change policy cannot be fulfilled. However, as noted in Section 1.7, it will not be possible to develop the necessary amounts of such infrastructure without some significant residual adverse impacts. This Part also shows why the Government considers that the need for such infrastructure will often be urgent. The IPC should	As explained in the Statement of Need [APP-320], the meaningful and timely contributions offered by the Scheme to UK decarbonisation and security of supply, while helping lower bills for consumers throughout its operational life, will be critical on the path to Net Zero. Without the Scheme, a significant and vital opportunity to



	therefore give substantial weight to considerations of need. The weight which is attributed to considerations of need in any given case should be proportionate to the anticipated extent of a project's actual contribution to satisfying the need for a particular type of infrastructure.	develop a large-scale low-carbon generation scheme will have been passed over, increasing materially the risk that future Carbon Budgets and Net Zero 2050 will not be achieved.
		An EIA has been undertaken to assess the environmental impacts of the Scheme and an ES prepared to report the findings [APP-039 to APP-061]. Overall, with appropriate mitigation implemented, this identifies the residual significant adverse effects of the Scheme. These effects are considered to be outweighed by the nationally significant benefits that the Scheme will provide by providing much needed large scale renewable energy generation.
Paragraph 3.3.1	Electricity meets a significant proportion of our overall energy needs and our reliance on it is likely to increase as we move towards our 2050 goals. The key reasons why the Government believes there is an urgent need for new electricity NSIPs are set out below.	As explained in the Statement of Need [APP-320], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to help meet the UK's urgent need to develop a secure, affordable and low carbon electricity generation system which is sufficient to meet future demand. The Government expects solar technology to play a major role in delivery of these objectives.
Paragraph 3.3.2	The Government needs to ensure sufficient electricity generating capacity is available to meet maximum peak demand, with a safety margin or spare capacity to accommodate unexpectedly high demand and to mitigate risks such as unexpected plant closures and extreme weather events. This is why there is currently around 85 GW of total generation capacity in the UK, whilst the average demand across a year is only for around half of this	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable and low carbon electricity generation system which is sufficient to



		meet future demand. The Government expects solar technology to play a major role in delivery of these objectives. The Statement of Need [APP-320] also explains why large-scale solar developments, such as the Scheme, are needed as part of an efficient, net-zero electricity generation mix, and how solar and wind generation are complementary technologies within the electricity generation system.
Paragraph 3.3.3	The larger the difference between available capacity and demand (i.e., the larger the safety margin), the more resilient the system will be in dealing with unexpected events, and consequently the lower the risk of a supply interruption. This helps to protect businesses and consumers, including vulnerable households, from rising and volatile prices and, eventually, from physical interruptions to supplies that might impact on essential services.	As explained in the Statement of Need [APP-320], and summarised in Sections 3 and 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable and low carbon electricity generation system which is sufficient to meet future demand. The Government expects solar technology to play a major role in delivery of these objectives. The Statement of Need [APP-320] also explains why large-scale solar developments, such as the Scheme, are needed as part of an efficient, net-zero electricity generation mix, and how solar and wind generation are complementary technologies within the electricity generation system. The Scheme includes a Battery Energy Storage System (BESS) to control the release of energy to the NETS, enabling it to be released when it is most needed.

Large scale solar farms, and the Scheme in particular, directly respond to the urgent need to deliver a large



Paragraph 3.3.4	 There are benefits of having a diverse mix of all types of power generation. It means we are not dependent on any one type of generation or one source of fuel or power and so helps to ensure security of supply. In addition, as set out briefly below, the different types of electricity generation have different characteristics which can complement each other: fossil fuel generation can be brought online quickly when there is high demand and shut down when demand is low, thus complementing generation from nuclear and the intermittent generation from renewables. However, until such time as fossil fuel generation can effectively operate with Carbon Capture and Storage (CCS), such power stations will not be low carbon (see Section 3.6). renewables offer a low carbon and proven (for example, onshore and offshore wind) fuel source, but many renewable technologies provide intermittent generation (see Section 3.4); and nuclear power is a proven technology that is able to provide continuous low carbon generation, which will help to reduce the UK's dependence on imports of fossil fuels (see Section 3.5). Whilst capable of responding to peaks and troughs in demand or supply, it is not as cost efficient to use nuclear power stations in this way when compared to fossil fuel generation. 	As explained in the Statement of Need [APP-320], and summarised in Sections 3 and 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable and low carbon electricity generation system which is sufficient to meet future demand. The Government expects solar technology to play a major role in delivery of these objectives. The Statement of Need [APP-320] also explains why large-scale solar developments, such as the Scheme, are needed as part of an efficient, net-zero electricity generation mix, and how solar and wind generation are complementary technologies within the electricity generation system.
Paragraph 3.3.5	The UK is choosing to largely decarbonise its power sector by adopting low carbon sources quickly. There are likely to be advantages to the UK of maintaining a diverse range of energy sources so that we are not overly reliant on any one technology (avoiding dependency on a particular fuel or technology type). This is why Government would like industry to bring forward many new low carbon developments (renewables, nuclear and fossil fuel generation with CCS) within the next 10 to 15 years to meet the twin challenge of energy security and climate change as we move towards 2050	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme will deliver significant amounts of low-carbon power. Solar is also relatively quick to construct compared to other technologies which have longer construction timeframes or have potentially not yet been proven at scale.



		amount of renewable generation capacity quickly. This is an important and relevant consideration for the Secretary of State in his decision-making process. Without the Scheme, a significant and vital opportunity to develop a large-scale low-carbon generation scheme will have been passed over, increasing materially the risk that future Carbon Budgets and Net Zero 2050 will not be achieved. Many forms of low-carbon generation will be required to meet the UK Climate objectives. A diverse mix of generation is required to minimise integration costs for those times when variable technologies are not generating electricity, but this does not mean that low-carbon generation developments should be curtailed to promote diversity. Indeed, by increasing the installed capacity of diverse renewable generation technologies across a broad geography, intermittency impacts are lower than they would be from a single-source supply deployed across a tighter geography.
		The Scheme will be part of the solution to moving the UK's energy mix away from fossil fuels towards renewable energy and it will make a sizeable contribution to meeting the Government's decarbonisation commitments, in direct accordance with this policy.
Paragraph 3.3.6	Within the strategic framework established by the Government it is for industry to propose the specific types of developments that they assess to be viable. This is the nature of a market-based energy system. The IPC should therefore act in accordance	As explained in the Statement of Need [APP-320], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable



	with the policy set out at in Section 3.1 when assessing proposals for new energy NSIPs.	and low carbon electricity generation system which is sufficient to meet future demand. The Government expects solar technology to play a major role in delivery of these objectives.
Paragraph 3.3.14	Government analysis of the different pathways to 2050 shows that it will be vital to make energy efficiency improvements per head of population if we are to meet the target of reducing emissions by at least 80% by 2050 (see paragraph 3.3.26 below). However, even with major improvements in overall energy efficiency, we expect that demand for electricity is likely to increase, as significant sectors of energy demand (such as industry, heating and transport) switch from being powered by fossil fuels to using electricity. As a result of this electrification of demand, total electricity consumption (measured in terawatt hours over a year) could double by 2050. Depending on the choice of how electricity is supplied, the total capacity of electricity generation (measured in GW) may need to more than double to be robust to all weather conditions. In some outer most circumstances, for example if there was very strong electrification of energy demand and a high level of dependence on intermittent electricity generation, then the capacity of electricity generation could need to triple. The Government therefore anticipates a substantial amount of new generation will be needed.	As explained in the Statement of Need [APP-320], and summarised in Sections 3 and 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable and low carbon electricity generation system which is sufficient to meet future demand. The Government expects solar technology to play a major role in delivery of these objectives. Furthermore, as explained in the Statement of Need [APP-320], demand for electricity is predicted to increase by 2050 by many expert projections. For example, paragraph 2.2.22 of NPS EN-1 predicts that 'the need to electrify large parts of the industrial and domestic heat and transport sectors could double demand for electricity over the next forty years'. The Scheme responds to this urgent and increasing demand for electricity.
Paragraph 3.3.15	In order to secure energy supplies that enable us to meet our obligations for 2050, there is an urgent need for new (and particularly low carbon) energy NSIPs to be brought forward as soon as possible, and certainly in the next 10 to 15 years, given the crucial role of electricity as the UK decarbonises its energy sector.	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme will deliver significant amounts of low-carbon power. Solar is also relatively quick to construct compared to other



		technologies which have longer construction timeframes or have potentially not yet been proven at scale. Large scale solar farms, and the Scheme in particular, directly respond to the urgent need to deliver a large amount of renewable generation capacity quickly. This is an important and relevant consideration for the Secretary of State in his decision-making process. Without the Scheme, a significant and vital opportunity to develop a large-scale low-carbon generation scheme will have been passed over, increasing materially the risk that future Carbon Budgets and Net Zero 2050 will not be achieved.
Paragraph 3.3.23	To minimise risks to energy security and resilience, the Government therefore believes it is prudent to plan for a minimum need of 59 GW of new electricity capacity by 2025.	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme will deliver significant amounts of low-carbon power. Solar is also relatively quick to construct compared to other technologies which have longer construction timeframes or have potentially not yet been proven at scale.
Paragraph 3.4.1	The UK has committed to sourcing 15% of its total energy (across the sectors of transport, electricity and heat) from renewable sources by 2020 and new projects need to continue to come forward urgently to ensure that we meet this target. Projections suggest that by 2020 about 30% or more of our electricity generation – both centralised and small-scale – could come from renewable sources, compared to 6.7% in 2009. The Committee on Climate Change in Phase 1 of its advice to Government in September 2010 agreed that the UK 2020 target was appropriate, and should not be increased. Phase 2 was published in May 2011 and provided	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Government has adopted more ambitious targets and commitments to decarbonise energy generation since NPS EN-1 was published. The meaningful and timely contributions offered by the Scheme to UK decarbonisation and security of supply, while helping lower bills for consumers throughout its operational life,



	recommendations on the post 2020 ambition for renewables in the UK, and possible pathways to maximise their contribution to the 2050 carbon reduction targets.	will be critical on the path to Net Zero. Without the Scheme, a significant and vital opportunity to develop a large-scale low-carbon generation scheme will have been passed over, increasing materially the risk that future Carbon Budgets and Net Zero 2050 will not be achieved.
Paragraph 3.4.2	Large scale deployment of renewables will help the UK to tackle climate change, reducing the UK's emissions of carbon dioxide by over 750 million tonnes by 2030. It will also deliver up to half a million jobs by 2020 in the renewables sector. Renewable electricity generation is currently supported in the UK through the Renewables Obligation (RO), which is a market-based support mechanism to encourage investment. Renewables have potential to improve security of supply by reducing reliance on the use of coal, oil and gas supplies to keep the lights on and power our businesses. Meeting the 15% renewables target could reduce fossil fuel demand by around 10% and gas imports by 20-30%. We are committed to meeting 2020 targets and have further ambitions for renewables post-2020. The Committee on Climate Change's May 2011 report included advice on moving to 30% renewable energy capacity by 2030 and a central scenario of 40% renewable electricity.	As explained in the Statement of Need [APP-320], and summarised in Sections 3 and 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the meaningful and timely contributions offered by the Scheme to UK decarbonisation and security of supply, while helping lower bills for consumers throughout its operational life, will be critical on the path to Net Zero. Without the Scheme, a significant and vital opportunity to develop a large-scale low-carbon generation scheme will have been passed over, increasing materially the risk that future Carbon Budgets and Net Zero 2050 will not be achieved.
Paragraph 3.4.3	The UK has substantial renewable energy resources, for example the British Isles have 40% of Europe's wind and some of the highest tidal reaches in the world. Unlike other technologies, the cost of renewables is in the construction and maintenance alone as the resource itself is usually free, so it helps protect consumers against the volatile but generally increasing cost of fossil fuels. Future large-scale renewable energy generation is likely to come from the following sources: - Onshore Wind – onshore wind is the most well-established and currently the most economically viable source of renewable electricity available for future large-scale deployment in the UK;	This was drafted and came into force in 2011 and relates to the technology available at the time. Since then, technology has developed and solar is now viable at NSIP scale. This is acknowledged by the Government in Draft NPS EN-1 which was published for consultation on 6 September 2021. This sets out an up-to-date position on the Government's expectations for the technologies that will be needed to meet its objectives and commitments for the energy system. This includes solar development as a



- Offshore Wind offshore wind is expected to provide the largest single contribution towards the 2020 renewable energy generation targets;
- Biomass biomass is a significant source of renewable and low carbon energy. It involves the combustion of fuel, such as wood, which is renewable because, through replanting and regrowth, the biomass can be replaced in a matter of decades and this cycle can be continuously repeated. Whilst energy is required to grow, harvest and transport it, biomass is considered to be low carbon, providing that the biomass has been cultivated, processed and transported with due consideration of sustainability. Its combustion also displaces emissions of carbon dioxide ordinarily released using fossil fuels;
- Energy from Waste (EfW) the principal purpose of the combustion of waste, or similar processes (for example pyrolysis or gasification) is to reduce the amount of waste going to landfill in accordance with the Waste Hierarchy and to recover energy from that waste as electricity or heat. Only waste that cannot be re-used or recycled with less environmental impact and would otherwise go to landfill should be used for energy recovery. The energy produced from the biomass fraction of waste is renewable and is in some circumstances eligible for Renewables Obligation Certificates, although the arrangements vary from plant to plant; and
- Wave and Tidal the UK has the potential for wave and tidal energy and there are now full-scale prototypes working towards array scale and pre-commercial deployment. However, many of the technologies for making use of the wave resource and tidal currents are still developing.

Proven technology exists for tidal range generation, but proposed projects are still some way off from commencement of construction. Paragraph 1.4.5 explains how this NPS relates to wave and tidal generation.

key element, as demonstrated by the following extracts from Draft NPS EN-1 (emphasis added):

3.3.44: "Known technologies that are included within the scope of this NPS are: Offshore Wind (including floating wind), Solar PV, Wave, Tidal Range, Tidal Stream, Pumped Hydro, Energy from Waste (including ACTs) with or without CCS, Biomass with or without CCS, Natural Gas with or without CCS, low carbon hydrogen, large-scale nuclear, Small Modular Reactors, Advanced Modular Reactors, and fusion power plants. The need for all these types of infrastructure is established by this NPS and is urgent."

3.3.21: "Our analysis shows that a secure, reliable, affordable, net zero consistent system in 2050 is likely to be composed predominantly of wind and solar."

It is therefore considered that this policy should be read as if solar generation is included in the list of future sources of large-scale renewable energy generation.



Paragraph 3.4.5	Paragraph 3.4.1 above sets out the UK commitments to sourcing 15% of energy from renewable sources by 2020. To hit this target, and to largely decarbonise the power sector by 2030, it is necessary to bring forward new renewable electricity generating projects as soon as possible. The need for new renewable electricity generation projects is therefore urgent.	As explained in the Statement of Need [APP-320], the Scheme will deliver significant amounts of low-carbon power. Solar is also relatively quick to construct compared to other technologies which have longer construction timeframes or have potentially not yet been proven at scale.
		Large scale solar farms, and the Scheme in particular, directly respond to the urgent need to deliver a large amount of renewable generation capacity quickly. This is an important and relevant consideration for the Secretary of State in his decision-making process Without the Scheme, a significant and vital opportunity to develop a large-scale low-carbon generation scheme will have been passed over, increasing materially the risk that future Carbon Budgets and Net Zero 2050 will not be achieved.
Paragraph 4.1.2	Given the level and urgency of need for infrastructure of the types covered by the energy NPSs set out in Part 3 of this NPS, the IPC should start with a presumption in favour of granting consent to applications for energy NSIPs. That presumption applies unless any more specific and relevant policies set out in the relevant NPSs clearly indicate that consent should be refused. The presumption is also subject to the provisions of the Planning Act 2008 referred to at paragraph 1.1.2 of this NPS.	The Applicant notes this policy and considers that the Scheme should be treated as if presumption in favour of granting consent applies, as more recent planning and energy policies set out that solar generation is expected to comprise an import part of an energy mix required to meet objectives and commitments for the energy system and climate change. For example, Draft NPS EN-1 states (emphasis added):
		3.3.44: "Known technologies that are included within the scope of this NPS are: Offshore Wind (including floating wind), Solar PV, Wave, Tidal Range, Tidal Stream, Pumped Hydro, Energy from Waste (including ACTs) with or without



		CCS, Biomass with or without CCS, Natural Gas with or without CCS, low carbon hydrogen, large-scale nuclear, Small Modular Reactors, Advanced Modular Reactors, and fusion power plants. The need for all these types of infrastructure is established by this NPS and is urgent." 3.3.21: "Our analysis shows that a secure, reliable, affordable, net zero consistent system in 2050 is likely to be composed predominantly of wind and solar."
Paragraph 4.1.3	 In considering any proposed development, and in particular when weighing its adverse impacts against its benefits, the IPC should take into account: its potential benefits including its contribution to meeting the need for energy infrastructure, job creation and any long-term or wider benefits; and its potential adverse impacts, including any long-term and cumulative adverse impacts, as well as any measures to avoid, reduce or compensate for any adverse impacts. 	The ES [APP-039 to APP-061] details extensively those potential benefits as well as the potential adverse impacts associated with the Scheme.
Paragraph 4.1.7	The IPC should only impose requirements in relation to a development consent that are necessary, relevant to planning, relevant to the development to be consented, enforceable, precise, and reasonable in all other respects. The IPC should take into account the guidance in Circular 11/95, as revised, on "The Use of Conditions in Planning Permissions" or any successor to it.	The Applicant notes that any Requirements imposed on the DCO, should consent be granted, necessary, relevant to planning, relevant to the development to be consented, enforceable, precise, and reasonable in all other respects.
Paragraph 4.1.8	The IPC may take into account any development consent obligations that an applicant agrees with local authorities. These must be relevant to planning, necessary to make the proposed development acceptable in planning terms, directly related to the proposed development, fairly and reasonably related in scale and kind to the proposed development, and reasonable in all other respects.	The Applicant includes draft Requirements in Schedule 1 of the Draft DCO [EN010132/EX6EX7/WB3.1_GH]. These have been provided to the Host Authorities for comment prior to submission of the DCO application



Paragraph 4.3.1	Prior to granting a development consent order, the IPC must, under the Habitats and Species Regulations, (which implement the relevant parts of the Habitats Directive and
	the Birds Directive in England and Wales) consider whether the project may have a
	significant effect on a European site, or on any site to which the same protection is
	applied as a matter of policy, either alone or in combination with other plans or
	projects. Further information on the requirements of the Habitats and Species
	Regulations can be found in a Government Circular. Applicants should also refer to
	Section 5.3 of this NPS on biodiversity and geological conservation. The applicant
	should seek the advice of Natural England and/or the Countryside Council for Wales,
	and provide the IPC with such information as it may reasonably require to determine
	whether an Appropriate Assessment is required. In the event that an Appropriate
	Assessment is required, the applicant must provide the IPC with such information as
	may reasonably be required to enable it to conduct the Appropriate Assessment. This
	should include information on any mitigation measures that are proposed to minimise

In accordance with this policy, the Applicant has consulted Natural England with regard to the Appropriate Assessment. Information that will enable the Secretary of State to carry out the Appropriate Assessment to be undertaken by the Secretary of State can be found in the Information to Support a Habitats Regulations Assessment report [REP3-024].

Paragraph 4.4.1

or avoid likely effects.

As in any planning case, the relevance or otherwise to the decision making process of the existence (or alleged existence) of alternatives to the proposed development is in the first instance a matter of law, detailed guidance on which falls outside the scope of this NPS. From a policy perspective this NPS does not contain any general requirement to consider alternatives or to establish whether the proposed project represents the best option.

Section 6.3 of the Planning Statement

[EN010132/EX6EX7/WB7.5_CD] sets out a consideration of the Scheme in the context of relevant policy that is applicable to alternatives. This notes that there is no general requirement from a policy perspective to consider alternatives or to establish whether the Scheme represents the 'best option'. The Planning Statement sets out how the Scheme accords with policies and legislation where consideration of alternatives may be relevant and explains how the Scheme has taken account of the locational criteria for solar farms that is set out in relevant policies.



		In addition, Chapter 5: Alternatives and Design Evolution of the ES [APP-043] sets out information in relation to alternatives that is required by the Infrastructure Planning (Environmental Impact Assessment) Regulations 2017. This discusses the following alternative options for the Scheme: • Alternative sites; • Alternative Technologies; • Alternative Layouts for Solar Panel Areas; • Alternative Substation Locations; and • Alternative Cable Routes.
Paragraph 4.4.2	 applicants are obliged to include in their ES, as a matter of fact, information about the main alternatives they have studied. This should include an indication of the main reasons for the applicant's choice, taking into account the environmental, social and economic effects and including, where relevant, technical and commercial feasibility; in some circumstances there are specific legislative requirements, notably under the Habitats Directive, for the IPC to consider alternatives. These should also be identified in the ES by the applicant; and in some circumstances, the relevant energy NPSs may impose a policy requirement to consider alternatives (as this NPS does in Sections 5.3, 5.7 and 	Section 6.3 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] sets out a consideration of the Scheme in the context of relevant policy that is applicable to alternatives. This notes that there is no general requirement from a policy perspective to consider alternatives or to establish whether the Scheme represents the 'best option'. The Planning Statement sets out how the Scheme accords with policies and legislation where consideration of alternatives may be relevant and explains how the Scheme has taken account of the locational criteria for solar farms that is set out in relevant policies.
	5.9).	In addition, Chapter 5: Alternatives and Design Evolution of the ES [APP-043] sets out information in relation to alternatives that is required by Schedule 4, paragraph (2)



of the Infrastructure Planning (Environmental Impact Assessment) Regulations 2017 (the EIA Regulations), which states: "A description of the reasonable alternatives (for example in terms of development design, technology, location, size and scale) studied by the developer, which are relevant to the proposed project and its specific characteristics, and an indication of the main reasons for selecting the chosen option, including a comparison of the environmental effects".

Paragraph 4.4.3

Where there is a policy or legal requirement to consider alternatives, the applicant should describe the alternatives considered in compliance with these requirements. Given the level and urgency of need for new energy infrastructure, the IPC should, subject to any relevant legal requirements (e.g., under the Habitats Directive) which indicate otherwise, be guided by the following principles when deciding what weight should be given to alternatives:

- the consideration of alternatives in order to comply with policy requirements should be carried out in a proportionate manner;
- the IPC should be guided in considering alternative proposals by whether
 there is a realistic prospect of the alternative delivering the same
 infrastructure capacity (including energy security and climate change benefits)
 in the same timescale as the proposed development;
- where (as in the case of renewables) legislation imposes a specific quantitative target for particular technologies or (as in the case of nuclear) there is reason to suppose that the number of sites suitable for deployment of a technology on the scale and within the period of time envisaged by the relevant NPSs is constrained, the IPC should not reject an application for development on one site simply because fewer adverse impacts would result from developing similar infrastructure on another suitable site, and it should have regard as

Section 6.3 of the Planning Statement

[EN010132/EX6EX7/WB7.5_CD] sets out a consideration of the Scheme in the context of relevant policy that is applicable to alternatives. This notes that there is no general requirement from a policy perspective to consider alternatives or to establish whether the Scheme represents the 'best option'. The Planning Statement sets out how the Scheme accords with policies and legislation where consideration of alternatives may be relevant and explains how the Scheme has taken account of the locational criteria for solar farms that is set out in relevant policies. In doing so it notes the requirements of this policy, including that consideration of alternatives should be proportionate, take account of an alternative's ability to deliver the same infrastructure capacity as the Scheme, and that Development Consent should not be rejected on one site simply because fewer adverse impacts would result from developing similar infrastructure on another suitable site.



- appropriate to the possibility that all suitable sites for energy infrastructure of the type proposed may be needed for future proposals;
- alternatives not among the main alternatives studied by the applicant (as
 reflected in the ES) should only be considered to the extent that the IPC thinks
 they are both important and relevant to its decision;
- as the IPC must decide an application in accordance with the relevant NPS (subject to the exceptions set out in the Planning Act 2008), if the IPC concludes that a decision to grant consent to a hypothetical alternative proposal would not be in accordance with the policies set out in the relevant NPS, the existence of that alternative is unlikely to be important and relevant to the IPC's decision;
- alternative proposals which mean the necessary development could not proceed, for example because the alternative proposals are not commercially viable or alternative proposals for sites would not be physically suitable, can be excluded on the grounds that they are not important and relevant to the IPC's decision;
- alternative proposals which are vague or inchoate can be excluded on the grounds that they are not important and relevant to the IPC's decision; and
- it is intended that potential alternatives to a proposed development should, wherever possible, be identified before an application is made to the IPC in respect of it (so as to allow appropriate consultation and the development of a suitable evidence base in relation to any alternatives which are particularly relevant). Therefore, where an alternative is first put forward by a third party after an application has been made, the IPC may place the onus on the person proposing the alternative to provide the evidence for its suitability as such and the IPC should not necessarily expect the applicant to have assessed it.

In addition, Chapter 5: Alternatives and Design Evolution of the ES [APP-043] sets out information in relation to alternatives that is required by Schedule 4, paragraph (2) of the Infrastructure Planning (Environmental Impact Assessment) Regulations 2017 (the EIA Regulations), which states: "A description of the reasonable alternatives (for example in terms of development design, technology, location, size and scale) studied by the developer, which are relevant to the proposed project and its specific characteristics, and an indication of the main reasons for selecting the chosen option, including a comparison of the environmental effects".



Paragraph 4.5.1

The visual appearance of a building is sometimes considered to be the most important factor in good design. But high quality and inclusive design goes far beyond aesthetic considerations. The functionality of an object — be it a building or other type of infrastructure — including fitness for purpose and sustainability, is equally important. Applying "good design" to energy projects should produce sustainable infrastructure sensitive to place, efficient in the use of natural resources and energy used in their construction and operation, matched by an appearance that demonstrates good aesthetic as far as possible. It is acknowledged, however that the nature of much energy infrastructure development will often limit the extent to which it can contribute to the enhancement of the quality of the area.

As detailed in Section 6.4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme has been subject to a detailed and sensitive iterative design process. This has taken account of the context and features of the land within the Order limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts, and provision of environmental and other enhancements, where practicable. The design process and basis of design decisions taken are described in the Chapter 5: Alternatives and Design Evolution of the ES [APP-043] and the Design and Access Statement [APP-314 to APP-315].

Paragraph 4.5.3

In the light of the above, and given the importance which the Planning Act 2008 places on good design and sustainability, the IPC needs to be satisfied that energy infrastructure developments are sustainable and, having regard to regulatory and other constraints, are as attractive, durable and adaptable (including taking account of natural hazards such as flooding) as they can be. In so doing, the IPC should satisfy itself that the applicant has taken into account both functionality (including fitness for purpose and sustainability) and aesthetics (including its contribution to the quality of the area in which it would be located) as far as possible. Whilst the applicant may not have any or very limited choice in the physical appearance of some energy infrastructure, there may be opportunities for the applicant to demonstrate good design in terms of siting relative to existing landscape character, landform and vegetation. Furthermore, the design and sensitive use of materials in any associated

As detailed in Section 6.4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme has been subject to a detailed and sensitive iterative design process. This has taken account of the context and features of the land within the Order limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts, and provision of environmental and other enhancements, where practicable. The design process and



	development such as electricity substations will assist in ensuring that such development contributes to the quality of the area.	basis of design decisions taken are described in the Chapter 5: Alternatives and Design Evolution of the ES [APP-043] and the Design and Access Statement [APP-314 to APP-315].
Paragraph 4.5.4	For the IPC to consider the proposal for a project, applicants should be able to demonstrate in their application documents how the design process was conducted and how the proposed design evolved. Where a number of different designs were considered, applicants should set out the reasons why the favoured choice has been selected. In considering applications the IPC should take into account the ultimate purpose of the infrastructure and bear in mind the operational, safety and security requirements which the design has to satisfy.	Sections 5.5 to 5.9 of Chapter 5: Alternatives and Design Evolution of the ES [APP-043] describes in detail the several stages of design evolution. This has been informed by ongoing environmental assessments, engineering and design considerations, as well as engagement with stakeholders. Alternative design options considered are also described in Chapter 5: Alternatives and Design Evolution of the ES [APP-043].
Paragraph 4.5.5	Applicants and the IPC should consider taking independent professional advice on the design aspects of a proposal. In particular, Design Council CABE can be asked to provide design review for nationally significant infrastructure projects and applicants are encouraged to use this service.	The Applicant has engaged extensively with the local planning authorities and their landscape architect advisors in the development of the design, through meetings, correspondence and a site visit.
Paragraph 4.8.3	To support planning decisions, the Government produces a set of UK Climate Projections and is developing a statutory National Adaptation Programme. In addition, the Government's Adaptation Reporting Power will ensure that reporting authorities (a defined list of public bodies and statutory undertakers, including energy utilities) assess the risks to their organisation presented by climate change. The IPC may take into account energy utilities' reports to the Secretary of State when considering adaptation measures proposed by an applicant for new energy infrastructure.	As stated in Chapter 7: Climate Change of the ES [REP1-012], UKCP18 climate projections have been used to identify potential future climate change impacts on the Scheme. The potential impacts of climate change on the Scheme, and associated mitigation measures, are outlined in Sections 7.7, 7.8 and 7.9 of Chapter 7: Climate Change of the ES [REP1-012].



Paragraph 4.8.4	In certain circumstances, measures implemented to ensure a scheme can adapt to climate change may give rise to additional impacts, for example as a result of protecting against flood risk, there may be consequential impacts on coastal change	No additional impacts have been identified as a result of the climate change mitigation measures presented in Chapter 7: Climate Change of the ES [REP1-012].
Paragraph 4.8.5	New energy infrastructure will typically be a long-term investment and will need to remain operational over many decades, in the face of a changing climate. Consequently, applicants must consider the impacts of climate change when planning the location, design, build, operation and, where appropriate, decommissioning of new energy infrastructure. The ES should set out how the proposal will take account of the projected impacts of climate change. While not required by the EIA Directive, this information will be needed by the IPC.	As outlined in Sections 7.7 and 7.8 of Chapter 7: Climate Change of the ES [REP1-012], account of the effects of climate change have been taken in the design of the Scheme, and its construction and decommissioning. This includes: • The effect of projected temperature increases on electrical equipment over the course of the Scheme's design life has been taken into account. Inverters (PV and BESS) will have a cooling system installed to control the temperature and allow the inverters to operate efficiently in warmer conditions. The PV modules and transformers have a wide range of acceptable operating temperatures, and it has been determined that increasing temperatures will not adversely affect their operation. • Any health and safety plans developed for construction and decommissioning activities will be required to account for potential climate change impacts on workers, such as flooding and heatwaves. • The design of drainage systems will ensure that there will be no significant increases in flood risk downstream during storms up to and including the



		1 in 100 (1%) annual probability design flood, with an allowance of 20% for climate change.
		A Decommissioning Environmental Management Plan (DEMP) (taking account of climate change risks at the time) will be prepared prior to decommissioning. An Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] is provided as part of the Application.
Paragraph 4.8.6	The IPC should be satisfied that applicants for new energy infrastructure have taken into account the potential impacts of climate change using the latest UK Climate Projections available at the time the ES was prepared to ensure they have identified appropriate mitigation or adaptation measures. This should cover the estimated lifetime of the new infrastructure. Should a new set of UK Climate Projections become available after the preparation of the ES, the IPC should consider whether they need to request further information from the applicant.	As stated in Chapter 7: Climate Change of the ES [REP1-012], UKCP18 climate projections have been used to identify potential future climate change impacts on the Scheme. Future climate change impacts are reviewed based on the UKCP18 projections and have been taken into account in the design of the Scheme. The effect of projected temperature increases on electrical equipment over the course of the Scheme's design life has been taken into account. Inverters (PV and BESS) will have a cooling system installed to control the temperature and allow the inverters to operate efficiently in warmer conditions. The PV modules and transformers have a wide range of acceptable operating temperatures, and it has been determined that increasing temperatures will not adversely affect their operation. The Scheme is therefore in full compliance with this policy.
Paragraph 4.8.7	Applicants should apply as a minimum, the emissions scenario that the Independent Committee on Climate Change suggests the world is currently most closely following –	The RCP 8.5 scenario has been used to generate the UKCP18 climate projections used. As per the UKCP18 user guidance, this is the closest available model to the 'high



	and the 10%, 50% and 90% estimate ranges. These results should be considered alongside relevant research which is based on the climate change projections.	emissions scenario' available within UKCP09, which were the latest available projections at the time of publication of the NPS EN-1. The UKCP18 climate projections are presented in Section 7.6 of Chapter 7: Climate Change of the ES [REP1-012].
Paragraph 4.8.8	The IPC should be satisfied that there are not features of the design of new energy infrastructure critical to its operation which may be seriously affected by more radical changes to the climate beyond that projected in the latest set of UK climate projections, taking account of the latest credible scientific evidence on, for example, sea level rise (for example by referring to additional maximum credible scenarios – i.e. from the Intergovernmental Panel on Climate Change or EA) and that necessary action can be taken to ensure the operation of the infrastructure over its estimated lifetime.	As stated in Chapter 7: Climate Change of the ES [REP1-012], UKCP18 climate projections have been used to identify potential future climate change impacts on the Scheme. Future climate change impacts are reviewed based on the UKCP18 projections. The effect of projected temperature increases on electrical equipment over the course of the Scheme's design life has been taken into account. Inverters (PV and BESS) will have a cooling system installed to control the temperature and allow the inverters to operate efficiently in warmer conditions. The PV modules and transformers have a wide range of acceptable operating temperatures, and it has been determined that increasing temperatures will not adversely affect their operation. The Scheme is therefore in full compliance with this policy.
Paragraph 4.8.9	Where energy infrastructure has safety critical elements (for example parts of new fossil fuel power stations or some electricity sub-stations), the applicant should apply the high emissions scenario (high impact, low likelihood) to those elements. Although the likelihood of this scenario is thought to be low, it is appropriate to take a more risk-averse approach with elements of infrastructure which are critical to the safety of its operation.	The RCP 8.5 scenario has been used to generate the UKCP18 climate projections used. As per the UKCP18 user guidance, this is the closest available model to the 'high emissions scenario' available within UKCP09, which were the latest available projections at the time of publication of the NPS EN-1. The UKCP18 climate projections are presented in Section 7.6 of Chapter 7: Climate Change of



		the ES [REP1-012]. The ES methodology therefore demonstrates compliance with this policy.
Paragraph 4.8.10	If any adaptation measures give rise to consequential impacts (for example on flooding, water resources or coastal change) the IPC should consider the impact of the latter in relation to the application as a whole and the impacts guidance set out in Part 5 of this NPS.	No consequential impacts have been identified as a result of climate change adaptation measures.
Paragraph 4.8.11	Any adaptation measures should be based on the latest set of UK Climate Projections, the Government's latest UK Climate Change Risk Assessment, when available and in consultation with the EA.	Chapter 7: Climate Change of the ES [REP1-012] utilises the latest UK climate projections (UKCP18) to determine the historic and future baseline conditions. Adaption measures are embedded in the design, such as the design of equipment tolerating a temperature range and the drainage design both taking account of the effects of climate change.
Paragraph 4.8.12	Adaptation measures can be required to be implemented at the time of construction where necessary and appropriate to do so. However, where they are necessary to deal with the impact of climate change, and that measure would have an adverse effect on other aspects of the project and/or surrounding environment (for example coastal processes), the IPC may consider requiring the applicant to ensure that the adaptation measure could be implemented should the need arise, rather than at the outset of the development (for example increasing height of existing, or requiring new, sea walls).	No consequential adverse impacts on other aspects of the project and/or surrounding environment have been identified as a result of climate change adaptation measures.
Paragraph 4.9.1	The connection of a proposed electricity generation plant to the electricity network is an important consideration for applicants wanting to construct or extend generation plant. In the market system, it is for the applicant to ensure that there will be necessary infrastructure and capacity within an existing or planned transmission or distribution network to accommodate the electricity generated. The applicant will liaise with National Grid who own and manage the transmission network in England	The Applicant has secured a connection to the National Grid via a new below ground grid connection cable located within the Grid Connection Route. This will connect West Burton 1 and 2 whose cables will independently enter the West Burton 3 substation at 132kV before exiting the West Burton 3 substation collectively (as three 132kV cables) at



	and Wales or the relevant regional Distribution Network Operator (DNO) to secure a grid connection. It may be the case that the applicant has not received or accepted a formal offer of a grid connection from the relevant network operator at the time of the application, although it is likely to have applied for one and discussed it with them. This is a commercial risk the applicant may wish to take for a variety of reasons, although the IPC will want to be satisfied that there is no obvious reason why a grid connection would not be possible.	400kV which are to then enter the West Burton 400kV substation spare bay, part of the NETS, at West Burton Power Station. Further details of this are included in the Grid Connection Statement [APP-316].
Paragraph 4.9.2	The Planning Act 2008 aims to create a holistic planning regime so that the cumulative effect of different elements of the same project can be considered together. The Government therefore envisages that wherever possible, applications for new generating stations and related infrastructure should be contained in a single application to the IPC or in separate applications submitted in tandem which have been prepared in an integrated way. However, this may not always be possible, nor the best course in terms of delivery of the project in a timely way, as different aspects may have different lead-in times and be undertaken by different legal entities subject to different commercial and regulatory frameworks (for example grid companies operate within OFGEM controls). So the level of information available on the different elements may vary. In some cases, applicant(s) may therefore decide to put in an application that seeks consent only for one element but contains some information on the second. Where this is the case, the applicant should explain the reasons for the separate application.	In line with this policy, the Applicant is seeking a DCO for the construction, operation (including maintenance), and decommissioning of ground mounted solar photovoltaic (PV) panel arrays, a Battery Energy Storage System (BESS) and supporting infrastructure. The DCO covers all infrastructure required to construct, operate (including maintain) and decommission the Scheme, with no further planning consent expected to be needed.
Paragraph 4.9.3	If this option is pursued, the applicant(s) accept the implicit risks involved in doing so, and must ensure they provide sufficient information to comply with the EIA Directive including the indirect, secondary and cumulative effects, which will encompass information on grid connections. The IPC must be satisfied that there are no obvious reasons why the necessary approvals for the other element are likely to be refused.	The Applicant is not following the route referred to by the policy.



	The fact that the IPC has decided to consent one project should not in any way fetter its subsequent decisions on any related projects.	
Paragraph 4.10.7	The IPC should be satisfied that development consent can be granted taking full account of environmental impacts. Working in close cooperation with EA and/or the pollution control authority, and other relevant bodies, such as the MMO, Natural England, the Countryside Council for Wales, Drainage Boards, and water and sewerage undertakers, the IPC should be satisfied, before consenting any potentially polluting developments, that: • the relevant pollution control authority is satisfied that potential releases can be adequately regulated under the pollution control framework; and	Phase 1 Preliminary Ecological Assessments (PEA) have been prepared, covering land within the Order limits, and are available in Appendices 9.2 and 9.4 of the ES [APP-078 and APP-080]. The information collected as part of the PEA suggests that there are no significant constraints with regards to contamination of soil and groundwater that would limit the development of the Order limits.
	 the effects of existing sources of pollution in and around the site are not such that the cumulative effects of pollution when the proposed development is added would make that development unacceptable, particularly in relation to statutory environmental quality limits. 	The potential risks that have been identified have all been assessed by the PEA as being very low to low, presented in Chapter 11: Ground Conditions and Contamination [APP-049].
		As stated in Chapter 11: Ground Conditions and Contamination [APP-049] provided that the requirements of relevant policy and legislation relating to land contamination and remediation are integrated within the design and appropriate mitigation measures are applied during the demolition and construction phases of each cumulative scheme, it is considered that the cumulative effect on ground conditions will be negligible.
Paragraph 4.10.8	The IPC should not refuse consent on the basis of pollution impacts unless it has good reason to believe that any relevant necessary operational pollution control permits or licences or other consents will not subsequently be granted.	Preliminary Geo-Environmental Risk Assessments [APP-095 to APP-104] have been prepared for the Scheme and do not identify any significant constraints in terms of ground conditions and contamination.



		The Scheme includes embedded mitigation for ground conditions and contamination in the form of A Construction Environmental Management Plan (CEMP) and Decommissioning Strategy, which will include procedures for the identification and mitigation of contaminant risks associated with the construction. An Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and Outline Decommissioning Strategy [EN010132/EX6/WB7.2_BREP6-023] form part of the application. Maintenance works will require similar mitigation measures.
		ES Chapter 11: Ground conditions and contamination [APP-049] concludes no potential significant effects have been identified after the implementation of embedded well-established good industry practices in construction for managing contaminated land which will be incorporated into a CEMP and Decommissioning Strategy and utilised in all phases of the Scheme. It is considered that the potential effects of contamination or risk of contamination will not be significant.
		It is anticipated that the permits outlined in the Consents and Agreements Position Statement [APP-312] will be granted. It is therefore considered that the Scheme is compliant with this policy.
Paragraph 4.13.1	Energy production has the potential to impact on the health and well-being ("health") of the population. Access to energy is clearly beneficial to society and to our health as a whole. However, the production, distribution and use of energy may have negative impacts on some people's health.	Chapter 21: Other Environmental Matters of the ES [APP-059] details what human health impacts the Scheme may have in Section 21.2 whilst also proposing mitigation measures.



Paragraph 4.13.2	As described in the relevant sections of this NPS and in the technology specific NPSs, where the proposed project has an effect on human beings, the ES should assess these effects for each element of the project, identifying any adverse health impacts, and identifying measures to avoid, reduce or compensate for these impacts as appropriate. The impacts of more than one development may affect people simultaneously, so the applicant and the IPC should consider the cumulative impact on health.	Chapter 21: Other Environmental Matters of the ES [APP-059] details what human health impacts the Scheme may have in Section 21.5 whilst also proposing mitigation measures.
Paragraph 4.13.4	New energy infrastructure may also affect the composition, size and proximity of the local population, and in doing so have indirect health impacts, for example if it in some way affects access to key public services, transport or the use of open space for recreation and physical activity.	There are several PRoW receptors within or abutting the Scheme. These are shown in Figures 8.10.1 to 8.10.4 of the ES [APP-180 to APP-183]. These PRoW are predominantly used for recreational purposes and form part of a wide network of PRoW in the surrounding area providing residents with alternative routes. As detailed in the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline OEMP [REP5-020] and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023], appropriate measures to mitigate temporary impacts on users of PRoW during the construction and decommissioning phases have been proposed. These Outline Plans are secured by Requirements in the Draft Development Consent Order Revision E [EN010132/EX6H [EX7/WB3.1_GH] submitted at Deadline 37. The need for any temporary diversions will be minimised and supported by clear signs and where possible will be planned and programmed to minimise disruption to users.



		The Scheme is not anticipated to have any indirect health impacts or include any proposals that affect access to public services. It is therefore considered that the Scheme accords with this policy.
Paragraph 4.13.5	Generally, those aspects of energy infrastructure which are most likely to have a significantly detrimental impact on health are subject to separate regulation (for example for air pollution) which will constitute effective mitigation of them, so that it is unlikely that health concerns will either constitute a reason to refused consents or require specific mitigation under the Planning Act 2008. However, the IPC will want to take account of health concerns when setting requirements relating to a range of impacts such as noise.	Mitigation measures have been embedded within the Scheme design to reduce operational effects such as noise, air quality and landscape, in turn these measures will mitigate the effects on the local community and existing facilities from a human health perspective. These are described in the ES [APP-039 to APP-061], including in Chapter 4, the Scheme Description, Chapter 15, Noise and Vibration, Chapter 17, Air Quality, and Chapter 21, Other Environmental Matters. In addition, measures to control the impacts of construction, operation and decommissioning are set out in the Outline Construction Environmental Management Plan (CEMP) [EN010132/EX6/WB7.1_DREP6-021], Outline Operational Environmental Management Plan (OEMP) [REP5-020], and Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023], respectively.
Paragraph 5.2.6	Where the project is likely to have adverse effects on air quality the applicant should undertake an assessment of the impacts of the proposed project as part of the Environmental Statement (ES).	Chapter 17: Air Quality of the ES [APP-055] includes an Air Quality Assessment, incorporating a Construction Phase Dust Risk Assessment.
Paragraph 5.2.7	The ES should describe:	Chapter 17: Air Quality of the ES [APP-055] contains a prediction of absolute air emission levels during the



	 any significant air emissions, their mitigation and any residual effects distinguishing between the project stages and taking account of any significant emissions from any road traffic generated by the project; the predicted absolute emission levels of the proposed project, after mitigation methods have been applied; 	construction, operation and decommissioning stages of the Scheme, and considers road traffic emissions. The Air Quality Assessment also includes the change in predicted NOx concentrations at relevant sensitive receptors.
	 existing air quality levels and the relative change in air quality from existing levels; and any potential eutrophication impacts. 	The chapter concludes that that there are anticipated to be negligible effects on air quality receptors as a result of the construction, operation or decommissioning of the Scheme.
		Mitigation measures to ensure that there are no off-site impacts from dust and represent good industry practice are incorporated into the Outline CEMP [EN010132/EX6/WB7.1_D].REP6-021]. Production of a final CEMP is secured by way of a requirement in the draft DCO. It is therefore considered that the Scheme is compliant with this policy.
Paragraph 5.2.9	The IPC should generally give air quality considerations substantial weight where a project would lead to a deterioration in air quality in an area, or leads to a new area where air quality breaches any national air quality limits. However, air quality considerations will also be important where substantial changes in air quality levels are expected, even if this does not lead to any breaches of national air quality limits.	Chapter 17: Air Quality of the ES [APP-055] concludes that there are anticipated to be no significant adverse effects on air quality as a result of the construction, operation or decommissioning of the Scheme. The Scheme is compliant with this policy.
Paragraph 5.2.10	In all cases the IPC must take account of any relevant statutory air quality limits. Where a project is likely to lead to a breach of such limits the developers should work with the relevant authorities to secure appropriate mitigation measures to allow the	Chapter 17: Air Quality of the ES [APP-055] concludes that there are anticipated to be no significant adverse effects on air quality as a result of the construction, operation or decommissioning of the Scheme; therefore, the Scheme is



	proposal to proceed. In the event that a project will lead to non-compliance with a statutory limit the IPC should refuse consent.	unlikely to lead to a breach of any statutory air quality limits.
Paragraph 5.2.11	The IPC should consider whether mitigation measures are needed both for operational and construction emissions over and above any which may form part of the project application. A construction management plan may help codify mitigation at this stage.	Mitigation measures pertaining to air quality are incorporated into the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021]. Production of a final CEMP is secured by way of a requirement in the draft DCO. The Scheme is compliant with this policy.
Paragraph 5.3.3	Where the development is subject to EIA the applicant should ensure that the ES clearly sets out any effects on internationally, nationally and locally designated sites of ecological or geological conservation importance, on protected species and on habitats and other species identified as being of principal importance for the conservation of biodiversity. The applicant should provide environmental information proportionate to the infrastructure where EIA is not required to help the IPC consider thoroughly the potential effects of a proposed project.	Section 9.5 of Chapter 9: Ecology and Biodiversity of the ES [APP-047] sets out all the designated sites of ecological or geological conservation importance; protected species; and habitats and other species identified as being of principal importance for the conservation of biodiversity within the study area for the Scheme. Sections 9.7 to 9.9 of Chapter 9: Ecology and Biodiversity of the ES [APP-047] clearly set out the expected effects on the above receptors during the construction, operation, and decommissioning phases of the Scheme. This concludes that there are anticipated to be no significant adverse effects on any internationally, nationally, or locally designated sites as a result of the Scheme. The scope of the ES [APP-039 to APP-061] accords with this policy.
Paragraph 5.3.4	The applicant should show how the project has taken advantage of opportunities to conserve and enhance biodiversity and geological conservation interests.	A Biodiversity Net Gain (BNG) assessment, using Defra's Metric 3.0, has been provided with the DCO application [APP-088].



		For the purposes of BNG, the Scheme will result in an overall significant net gain of biodiversity net gain of 86.80% provided in habitat, 54.71% gains in hedgerow units and 33.25% gains in river units. The Scheme has therefore taken advantage of opportunities to conserve and enhance biodiversity and accords with this policy.
Paragraph 5.3.6	In having regard to the aim of the Government's biodiversity strategy the IPC should take account of the context of the challenge of climate change: failure to address this challenge will result in significant adverse impacts to biodiversity. The policy set out in the following sections recognises the need to protect the most important biodiversity and geological conservation interests. The benefits of nationally significant low carbon energy infrastructure development may include benefits for biodiversity and geological conservation interests and these benefits may outweigh harm to these interests. The IPC may take account of any such net benefit in cases where it can be demonstrated.	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6.2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's commitments to decrease carbon emissions and reach net zero by 2050. As noted by the policy, failure to address climate change will result in significant adverse impacts to biodiversity. Without the Scheme, a significant and vital opportunity to develop a large-scale low-carbon generation scheme will have been passed over, increasing materially the risk that future Carbon Budgets and Net Zero 2050 will not be achieved. In addition, a Biodiversity Net Gain (BNG) assessment, using Defra's Metric 3.0, has been provided with the DCO application [APP-088]. For the purposes of BNG, the Scheme will result in an overall significant net gain of biodiversity net gain of 86.80% provided in habitat, 54.71% gains in hedgerow units and 33.25% gains in river units. The urgent and national need for the Scheme as outlined in the Statement of Need [APP-320], combined with the



		overall net gain for biodiversity achieved as a result of the Scheme, should be considered in the planning balance. By enhancing biodiversity within the Order limits, and by generating renewable electricity and thereby helping to address the causes of climate change, the Scheme delivers benefits in relation to both elements of this policy.
Paragraph 5.3.7	As a general principle, and subject to the specific policies below, development should aim to avoid significant harm to biodiversity and geological conservation interests, including through mitigation and consideration of reasonable alternatives (as set out in Section 4.4 above); where significant harm cannot be avoided, then appropriate compensation measures should be sought.	As outlined in Section 9.7 to 9.9 of Chapter 9: Ecology and Biodiversity of the ES [APP-047], there are anticipated to be no potential for significant adverse effects on any designated ecological sites, habitats or protected species. Embedded design mitigation measures are outlined in Section 9.6 of Chapter 9: Ecology and Biodiversity of the ES [APP-047] and are illustrated within the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline OEMP [REP5-020] and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023]. These include habitat avoidance, creation and replacement measures; mitigation relating to protected and notable species; and standard mitigation measures that comply with industry good practice and environmental legislation. Production of a final CEMP, OEMP and DEMP are secured by way of a requirement in the draft DCO. The Scheme is compliant with this policy.
Paragraph 5.3.8	In taking decisions, the IPC should ensure that appropriate weight is attached to designated sites of international, national and local importance; protected species;	Section 9.6 of Chapter 9: Ecology and Biodiversity of the ES [APP-047] sets out all the designated sites of ecological or geological conservation importance; protected species;



	habitats and other species of principal importance for the conservation of biodiversity; and to biodiversity and geological interests within the wider environment.	and habitats and other species identified as being of principal importance for the conservation of biodiversity. Section 9.7 of Chapter 9: Ecology and Biodiversity of the ES [APP-047] clearly sets out the expected effects on the above receptors during the construction, operation and decommissioning phases of the Scheme. This concludes that there are anticipated to be no significant adverse effects on any internationally, nationally or locally designated sites as a result of the Scheme. The Scheme is therefore in accordance with this policy.
Paragraph 5.3.9	The most important sites for biodiversity are those identified through international conventions and European Directives. The Habitats Regulations provide statutory protection for these sites but do not provide statutory protection for potential Special Protection Areas (pSPAs) before they have been classified as a Special Protection Area. For the purposes of considering development proposals affecting them, as a matter of policy the Government wishes pSPAs to be considered in the same way as if they had already been classified. Listed Ramsar sites should, also as a matter of policy, receive the same protection.	Section 9.5 of Chapter 9: Ecology and Biodiversity of the ES [APP-047] identifies that there are no SPA or SAC designations within 10km of the Scheme.
Paragraph 5.3.10	Many SSSIs are also designated as sites of international importance and will be protected accordingly. Those that are not, or those features of SSSIs not covered by an international designation, should be given a high degree of protection. All National Nature Reserves are notified as SSSIs.	There are no SSSIs within the Order Limits. Section 9.5 in Chapter 9: Ecology of the ES [APP-047] identifies five SSSIs within 5km of the Order Limits, namely: Doddington Clay Woods SSSI, Ashton's Meadow SSSI, Tresswell Wood SSSI, Clarborough Tunnel SSSI and Lea Marsh SSSI.
Paragraph 5.3.11	Where a proposed development on land within or outside an SSSI is likely to have an adverse effect on an SSSI (either individually or in combination with other developments), development consent should not normally be granted. Where an	Section 9.7 of Chapter 9: Ecology of the ES [APP-047] concludes that it is not anticipated that there will be any significant adverse effects on any SSSIs either alone or in



	adverse effect, after mitigation, on the site's notified special interest features is likely, an exception should only be made where the benefits (including need) of the development at this site, clearly outweigh both the impacts that it is likely to have on the features of the site that make it of special scientific interest and any broader impacts on the national network of SSSIs. The IPC should use requirements and/or planning obligations to mitigate the harmful aspects of the development and, where possible, to ensure the conservation and enhancement of the site's biodiversity or geological interest.	combination with other projects. This policy therefore does not apply to this Scheme.
Paragraph 5.3.13	Sites of regional and local biodiversity and geological interest, which include Regionally Important Geological Sites, Local Nature Reserves and Local Sites, have a fundamental role to play in meeting overall national biodiversity targets; contributing to the quality of life and the well-being of the community; and in supporting research and education. The IPC should give due consideration to such regional or local designations. However, given the need for new infrastructure, these designations should not be used in themselves to refuse development consent.	ES Appendix 9.2 Preliminary Ecological Appraisal Appendix [APP-078] identifies non-statutory sites of regional and local biodiversity and geological interest within 2km of the Order Limits. Chapter 9: Ecology of the ES [APP-047] concludes that there are no potential significant adverse effects on local or regional biodiversity sites as a result of the construction, operation or decommissioning of the Scheme.
Paragraph 5.3.14	Ancient woodland is a valuable biodiversity resource both for its diversity of species and for its longevity as woodland. Once lost it cannot be recreated. The IPC should not grant development consent for any development that would result in its loss or deterioration unless the benefits (including need) of the development, in that location outweigh the loss of the woodland habitat. Aged or 'veteran' trees found outside ancient woodland are also particularly valuable for biodiversity and their loss should be avoided. Where such trees would be affected by development proposals the applicant should set out proposals for their conservation or, where their loss is unavoidable, the reasons why.	As detailed within the Ecology Chapter [APP-047], there are no ancient woodlands within the Order Limits although there are several small stands of managed and unmanaged woodland present adjacent and in the surrounding landscape. As detailed within the Ecology Chapter [APP-047], there are no aged or veteran trees that are expected to be affected by the Scheme as those trees found within the Order Limits include trees, of which, roughly half contain occasional mature and semi-mature trees.



		The Policy is noted although is not considered to be applicable to the Scheme.
Paragraph 5.3.15	Development proposals provide many opportunities for building-in beneficial biodiversity or geological features as part of good design. When considering proposals, the IPC should maximise such opportunities in and around developments, using requirements or planning obligations where appropriate.	A Biodiversity Net Gain (BNG) report, using Defra's Metric 3.0, has been provided with the DCO application [APP-088]. For the purposes of BNG, the Scheme will result in an overall significant net gain of biodiversity net gain of 86.80% provided in habitat, 54.71% gains in hedgerow units and 33.25% gains in river units.
Paragraph 5.3.17	Other species and habitats have been identified as being of principal importance for the conservation of biodiversity in England and Wales and thereby requiring conservation action. The IPC should ensure that these species and habitats are protected from the adverse effects of development by using requirements or planning obligations. The IPC should refuse consent where harm to the habitats or species and their habitats would result, unless the benefits (including need) of the development outweigh that harm. In this context the IPC should give substantial weight to any such harm to the detriment of biodiversity features of national or regional importance which it considers may result from a proposed development.	Section 9.5 of Chapter 9: Ecology of the ES [APP-047] sets out all the designated sites of ecological or geological conservation importance; protected species; and habitats and other species identified as being of principal importance for the conservation of biodiversity. Section 9.7 of Chapter 9: Ecology of the ES [APP-047] sets out the expected effects on the above receptors during the construction, operation and decommissioning phases of the Scheme. This concludes that there are anticipated to be no significant adverse effects on any internationally, nationally or locally designated sites, or on protected or priority species and habitats as a result of the Scheme. The Scheme is therefore in accordance with this policy.
Paragraph	The applicant should include appropriate mitigation measures as an integral part of	Embedded design mitigation measures of the kind set out
5.3.18	the proposed development. In particular, the applicant should demonstrate that:	in this policy are outlined in Section 9.6 of Chapter 9: Ecology of the ES [APP-047] and are illustrated within the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline



	 during construction, they will seek to ensure that activities will be confined to the minimum areas required for the works; during construction and operation best practice will be followed to ensure that risk of disturbance or damage to species or habitats is minimised, including as a consequence of transport access arrangements; habitats will, where practicable, be restored after construction works have finished; and 	OEMP [REP5-020]) and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023]. These include habitat avoidance, creation and replacement measures; mitigation relating to protected and notable species; and standard mitigation measures that comply with industry good practice and environmental legislation. Production of a final CEMP, OEMP and DEMP are secured by way of a requirement in the draft DCO.
	 opportunities will be taken to enhance existing habitats and, where practicable, to create new habitats of value within the site landscaping proposals. 	The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] includes best practice measures to ensure that activities will be confined to the minimum areas required for the works during construction, in accordance with this part of the policy.
		Section 9.6 of Chapter 9: Ecology of the ES [APP-047] outlines mitigation measures pertaining to habitat avoidance, creation and replacement measures that comply with this part of the policy.
Paragraph 5.3.20	The IPC will need to take account of what mitigation measures may have been agreed between the applicant and Natural England (or the Countryside Council for Wales) or the Marine Management Organisation (MMO), and whether Natural England (or the Countryside Council for Wales) or the MMO has granted or refused or intends to grant or refuse, any relevant licences, including protected species mitigation licences.	Chapter 9: Ecology of the ES [APP-047] has assessed embedded mitigation. The Outline LEMP [EN010132/EX6/WB7.3_EREP6-025] supports this assessment in that the Outline LEMP is not limited to embedded mitigation and also contains additional mitigation, for example the mitigation of adverse impacts upon IEFs such as ground nesting birds.
Paragraph 5.4.2	UK airspace is important for both civilian and military aviation interests. It is essential that the safety of UK aerodromes, aircraft and airspace is not adversely affected by	Chapter 16: Glint and Glare of the ES [APP-054] concludes that the Scheme design has embedded sufficient



	new energy infrastructure. Similarly, aerodromes can have important economic and social benefits, particularly at the regional and local level. Commercial civil aviation is largely confined to designated corridors of controlled airspace and set approaches to airports. However, civilian leisure and military aircraft may often fly outside of 'controlled air space'. The approaches and flight patterns to aerodromes are not necessarily routine and can be irregular owing to a variety of factors including the performance characteristics of the aircraft concerned and the prevailing meteorological conditions.	mitigation to avoid significant adverse effects upon RAF Scampton.
Paragraph 5.4.10	Where the proposed development may have an effect on civil or military aviation and/or other defence assets an assessment of potential effects should be set out in the ES (see Section 4.2).	The ES [APP-039 to APP-061] concludes that the Scheme design has embedded sufficient mitigation to avoid significant adverse effects on civil or military aviation and/or other defence assets.
Paragraph 5.4.11	The applicant should consult the MoD, CAA, NATS and any aerodrome – licensed or otherwise – likely to be affected by the proposed development in preparing an assessment of the proposal on aviation or other defence interests.	The Applicant has consulted with all relevant aerodromes (licensed or otherwise) through the PEIR stage. Doncaster Sheffield Airport raised concerns over the West Burton 4 Site which has since been removed from the Scheme.
Paragraph 5.4.13	If any relevant changes are made to proposals during the pre-application and determination period, it is the responsibility of the applicant to ensure that the relevant aviation and defence consultees are informed as soon as reasonably possible.	Notifications of changes to the proposal have been duly passed onto the relevant aviation and defence consultees. The Applicant recognises this continued responsibility throughout the determination period.
Paragraph 5.4.17	 Where, after reasonable mitigation, operational changes, obligations and requirements have been proposed, the IPC considers that: a development would prevent a licensed aerodrome from maintaining its licence; the benefits of the proposed development are outweighed by the harm to aerodromes serving business, training or emergency service needs, taking 	The Applicant has consulted with all relevant aerodromes (licensed or otherwise) through the PEIR stage. Doncaster Sheffield Airport raised concerns over the West Burton 4 Site which has since been removed from the Scheme. Through mitigation measures captured within the ES [APP-039 to APP-061], the development is not considered to



	 into account the relevant importance and need for such aviation infrastructure; or the development would significantly impede or compromise the safe and effective use of defence assets or significantly limit military training; the development would have an impact on the safe and efficient provision of en route air traffic control services for civil aviation, in particular through an adverse effect on the infrastructure required to support communications, navigation or surveillance systems; consent should not be granted. 	significantly impede or compromise the safe and effective use of defence assets or significantly limit military training or impact on the safe and efficient provision of en route air traffic control services for civil aviation. The Scheme is therefore considered to comply with this Policy.
Paragraph 5.4.18	Where a proposed energy infrastructure development would significantly impede or compromise the safe and effective use of civil or military aviation or defence assets and or significantly limit military training, the IPC may consider the use of 'Grampian111, or other forms of condition which relate to the use of future technological solutions, to mitigate impacts. Where technological solutions have not yet been developed or proven, the IPC will need to consider the likelihood of a solution becoming available within the time limit for implementation of the development consent. In this context, where new technologies to mitigate the adverse effects of wind farms on radar are concerned, the IPC should have regard to any Government guidance which emerges from the joint Government/Industry Aviation Plan.	The Scheme is not considered to significantly impede or compromise the safe and effective use of civil or military aviation or defence assets and or significantly limit military training. As such, this policy is not considered applicable to the scheme although it is noted.
Paragraph 5.6.4	The applicant should assess the potential for insect infestation and emissions of odour, dust, steam, smoke and artificial light to have a detrimental impact on amenity, as part of the Environmental Statement.	A Dust Assessment has been carried out as part of Chapter 17: Air Quality of the ES [APP-055]. Artificial lighting will be required during construction and decommissioning in areas where natural lighting is unable to reach (sheltered/confined areas), and during core working hours within winter months. All construction lighting will be deployed in accordance with the



		recommendations set out in the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021].
		Details of operational lighting are set out by Chapter 4, Development Description, of the ES [APP-042]. This explains that no part of the Scheme will be continuously lit. Manually operated, and motion-detection lighting will be utilised for operational and security purposes around electrical infrastructure. Lighting will be directed downward and away from boundaries. No visible lighting will be utilised at the site perimeter fence, aside from the site entrance points.
		The Scheme is not expected to result in an increased risk of insect infestation and will not emit any odour. Construction and decommissioning activities will not include burning materials (as set out in the Outline CEMP [EN010132/EX6/WB7.1_D].REP6-021]. For these reasons, smoke, odour and insect infestation risk has not been assessed in the ES.
Paragraph 5.6.5	 In particular, the assessment provided by the applicant should describe: the type, quantity and timing of emissions; aspects of the development which may give rise to emissions; premises or locations that may be affected by the emissions; effects of the emission on identified premises or locations; and measures to be employed in preventing or mitigating the emissions. 	A Dust Assessment has been carried out as part of Chapter 17: Air Quality of the ES [APP-055] in line with the requirements of this policy. The assessment considers the potential dust risk across a set of pre-defined zones following IAQM guidance, up to 350m from the Order limits.



Paragraph 5.7.4	Applications for energy projects of 1 hectare or greater in Flood Zone 1 in England or Zone A in Wales and all proposals for energy projects located in Flood Zones 2 and 3 in England or Zones B and C in Wales should be accompanied by a flood risk assessment (FRA). An FRA will also be required where an energy project less than 1 hectare may be subject to sources of flooding other than rivers and the sea (for example surface water), or where the EA, Internal Drainage Board or other body have indicated that there may be drainage problems. This should identify and assess the risks of all forms of flooding to and from the project and demonstrate how these flood risks will be managed, taking climate change into account.	A Flood Risk Assessment (FRA) is provided at Appendices 10.1 – 10.6 of the ES [APP-089 to APP-094]. The FRA provides a detailed assessment of the risk of flooding to and from the Scheme (taking account of climate change) and concludes that the risk of flooding will not be increased as a result of the construction, operation or decommissioning of the Scheme. It is therefore considered that the Scheme is compliant with this policy.
Paragraph 5.7.5	 The minimum requirements for FRAs are that they should: be proportionate to the risk and appropriate to the scale, nature and location of the project; consider the risk of flooding arising from the project in addition to the risk of flooding to the project; take the impacts of climate change into account, clearly stating the 	An FRA is provided at Appendices 10.1 – 10.6 of the ES [APP-089 to APP-094]. The FRA appendices outline the objectives of the FRA, as stipulated by the NPS, and the requirements which are addressed throughout the FRA. The Scheme is therefore compliant with this policy.
	 development lifetime over which the assessment has been made; be undertaken by competent people, as early as possible in the process of preparing the proposal; consider both the potential adverse and beneficial effects of flood risk management infrastructure, including raised defences, flow channels, flood storage areas and other artificial features, together with the consequences of their failure; consider the vulnerability of those using the site, including arrangements for safe access; 	



- consider and quantify the different types of flooding (whether from natural and human sources and including joint and cumulative effects) and identify flood risk reduction measures, so that assessments are fit for the purpose of the decisions being made;
- consider the effects of a range of flooding events including extreme events on people, property, the natural and historic environment and river and coastal processes;
- include the assessment of the remaining (known as 'residual') risk after risk reduction measures have been taken into account and demonstrate that this is acceptable for the particular project;
- consider how the ability of water to soak into the ground may change with development, along with how the proposed layout of the project may affect drainage systems;
- consider if there is a need to be safe and remain operational during a worstcase flood event over the development's lifetime; and
- be supported by appropriate data.

Paragraph 5.7.7

Applicants for projects which may be affected by, or may add to, flood risk should arrange pre-application discussions with the EA, and, where relevant, other bodies such as Internal Drainage Boards, sewerage undertakers, navigation authorities, highways authorities and reservoir owners and operators. Such discussions should identify the likelihood and possible extent and nature of the flood risk, help scope the FRA, and identify the information that will be required by the IPC to reach a decision on the application when it is submitted. The IPC should advise applicants to undertake these steps where they appear necessary, but have not yet been addressed.

An FRA is provided at Appendices 10.1 – 10.6 of the ES [APP-089 to APP-094]. The preparation of the FRA, and the ES has taken account of advice from the EA and LLFAS (West Lindsey District Council and Bassetlaw District Council). The Order Limits are not shown to be located within the operational boundary of an Internal Drainage Board.



Paragraph 5.7.9

In determining an application for development consent, the IPC should be satisfied that where relevant:

- the application is supported by an appropriate FRA;
- the Sequential Test has been applied as part of site selection;
- a sequential approach has been applied at the site level to minimise risk by directing the most vulnerable uses to areas of lowest flood risk;
- the proposal is in line with any relevant national and local flood risk management strategy;
- priority has been given to the use of sustainable drainage systems (SuDs) (as required in the next paragraph on National Standards); and
- in flood risk areas the project is appropriately flood resilient and resistant, including safe access and escape routes where required, and that any residual risk can be safely managed over the lifetime of the development.

An FRA is provided at Appendices 10.1 – 10.6 of the ES [APP-089 to APP-094]. The FRA Sequential and Exception Test Report demonstrates how the development passes both the Sequential Test and Exception Test [APP-089 to APP-094].

Appendices 10.1 – 10.5 of the ES **[APP-089 to APP-094]** set out allowable discharge rates set out in which will be achieved through sustainable drainage systems SuDS Strategy contained within the ES chapter **[APP-048]**. The appendices also explain that, through the sequential process and design iterations, there are no buildings located within the floodplain. All compounds for site staff and battery storage units have been located out of Flood Zones 2 and 3, i.e., within Flood Zone 1, and it is envisaged access to the PV Panels would not be sought during flooding conditions. Access to the Scheme will therefore be safe from flooding.

During construction, the Outline CEMP

[EN010132/EX6/WB7.1_DREP6-021] sets out measures to

ensure the safety of staff during construction from flood risk. This includes the appointment of at least one designated Flood Warden who is familiar with the risks and remains vigilant to news reports, Environment Agency flood warnings, relevant weather warnings and water levels of the local waterway. The Contractor will be required to produce a Flood Risk Management Action Plan/Method Statement which will provide details of the



		response to an impending flood, including evacuation and site closedown procedures. The requirement for the Flood Risk Management Action Plan/Method Statement would be determined within the detailed CEMPs. It is therefore considered that the Scheme has met the requirements of this policy.
Paragraph 5.7.10	For construction work which has drainage implications, approval for the project's drainage system will form part of the development consent issued by the IPC. The IPC will therefore need to be satisfied that the proposed drainage system complies with any National Standards published by Ministers under Paragraph 5(1) of Schedule 3 to the Flood and Water Management Act 2010. In addition, the development consent order, or any associated planning obligations, will need to make provision for the adoption and maintenance of any SuDS, including any necessary access rights to property. The IPC should be satisfied that the most appropriate body is being given the responsibility for maintaining any SuDS, taking into account the nature and security of the infrastructure on the proposed site. The responsible body could include, for example, the applicant, the landowner, the relevant local authority, or another body, such as an Internal Drainage Board.	ES Appendices 10.1 – 10.6 Flood Risk Assessment [APP-089 to APP-094] sets out allowable discharge rates set out in which will be achieved through sustainable drainage systems SuDS Strategy contained within the ES Chapter APP-048]. This includes details of maintenance of SuDS features.
Paragraph 5.7.12	The IPC should not consent development in Flood Zone 2 in England or Zone B in Wales unless it is satisfied that the sequential test requirements have been met. It should not consent development in Flood Zone 3 or Zone C unless it is satisfied that the Sequential and Exception Test requirements have been met. The technology-specific NPSs set out some exceptions to the application of the sequential test. However, when seeking development consent on a site allocated in a development plan through the application of the Sequential Test, informed by a strategic flood risk assessment, applicants need not apply the Sequential Test, but should apply the sequential approach to locating development within the site.	As stated in ES Appendix 10.1 Flood Risk Assessment and Drainage Strategy [APP-089] the majority of the developable Sites are located out of Flood Zones 2 and 3 (including climate change allowance). Substations and conversion units have been proposed to be located outside of the 1% AEP + CC extent and/or the 0.1% Annual Probability Surface Water proxy extent. The solar panels will be mounted on raised frames above



		surrounding ground level allowing flood water to flow freely underneath.
		The proposed development is free draining through perimeter gaps around all panels, allowing for infiltration as existing within the grassland/vegetation surrounding and beneath the panels. There will be minimal increase in impermeable area meaning the proposals will not increase surface water flood risk elsewhere.
		It is considered that, with the above measures, the proposals pass the Sequential Test.
		As for the exception test, the Flood Risk Assessment demonstrates that the Site will not increase flood risk elsewhere and that the ground beneath the panels will remain entirely permeable, draining as existing. The development may also reduce existing greenfield run-off rates by replacing intensive agricultural surfaces with a landcover comprising a mixture of wildflowers and grassland.
Paragraph 5.7.13	Preference should be given to locating projects in Flood Zone 1 in England or Zone A in Wales. If there is no reasonably available site in Flood Zone 1 or Zone A, then projects can be located in Flood Zone 2 or Zone B. If there is no reasonably available site in Flood Zones 1 or 2 or Zones A & B, then nationally significant energy infrastructure projects can be located in Flood Zone 3 or Zone C subject to the Exception Test. Consideration of alternative sites should take account of the policy on alternatives set out in Section 4.4 above	As stated in ES Appendix 10.1 Flood Risk Assessment and Drainage Strategy [APP-089], the majority of the Sites are located out of Flood Zones 2 and 3 (including climate change allowance). Where development is proposed within Flood Zones 2 and 3, sensitive electrical equipment will be raised 0.6 m above the 0.1% AEP + CC flood level or where this is not possible as high as practicably possible. The Scheme therefore accords with this policy.



Paragraph 5.7.14	If, following application of the sequential test, it is not possible, consistent with wider sustainability objectives, for the project to be located in zones of lower probability of flooding than Flood Zone 3 or Zone C, the Exception Test can be applied. The test provides a method of managing flood risk while still allowing necessary development to occur.	As stated in ES Appendix 10.1 Flood Risk Assessment and Drainage Strategy [APP-089], Substations and conversion units (sensitive electrical equipment) have been proposed to be located outside of the 1% AEP + CC extent and/or the 0.1% Annual Probability Surface Water proxy extent. The solar panels will be mounted on raised frames above surrounding ground level allowing flood water to flow freely underneath.
		The proposed development is free draining through perimeter gaps around all panels, allowing for infiltration as existing within the grassland/vegetation surrounding and beneath the panels. There will be minimal increase in impermeable area meaning the proposals will not increase surface water flood risk elsewhere. It is considered that, with the above measures, the proposals pass the Sequential Test.
Paragraph 5.7.15	The Exception Test is only appropriate for use where the sequential test alone cannot deliver an acceptable site, taking into account the need for energy infrastructure to remain operational during floods. It may also be appropriate to use it whereas a result of the alternative site(s) at lower risk of flooding being subject to national designations such as landscape, heritage and nature conservation designations, for example Areas of Outstanding Natural Beauty (AONBs), Sites of Special Scientific Interest (SSSIs) and World Heritage Sites (WHS) it would not be appropriate to require the development to be located on the alternative site(s).	The majority of the Order Limits lie within Flood Zone 1 and so do not require the Exception Test to be passed, given that the proposal is considered to pass the Sequential Test. The Scheme will deliver wider sustainability benefits, being a renewable energy development that will make a substantial contribution to the country achieving net-zero carbon emissions. The Scheme could not be delivered on previously developed land in sufficient proximity to the point of connection to the NETS as demonstrated in the Site Selection



		Assessment [AS-004] , and the project will remain safe in its lifetime.
Paragraph 5.7.16	All three elements of the test will have to be passed for development to be consented. For the Exception Test to be passed:	The Scheme is considered to pass the Exception Test as outlined within the Sequential and Exception Test Report
	• it must be demonstrated that the project provides wider sustainability benefits to the community that outweigh flood risk;	[APP-094].
	 the project should be on developable, previously developed land or, if it is not on previously developed land, that there are no reasonable alternative sites on developable previously developed land subject to any exceptions set out in the technology-specific NPSs; and 	
	 an FRA must demonstrate that the project will be safe, without increasing flood risk elsewhere subject to the exception below and, where possible, will reduce flood risk overall. 	
Paragraph 5.7.17	Exceptionally, where an increase in flood risk elsewhere cannot be avoided or wholly mitigated, the IPC may grant consent if it is satisfied that the increase in present and future flood risk can be mitigated to an acceptable level and taking account of the benefits of, including the need for, nationally significant energy infrastructure as set out in Part 3 above. In any such case the IPC should make clear how, in reaching its decision, it has weighed up the increased flood risk against the benefits of the project, taking account of the nature and degree of the risk, the future impacts on climate change, and advice provided by the EA and other relevant bodies.	As detailed within the Flood Risk Assessment and Drainage Strategy [APP-089], Substations and conversion units (sensitive electrical equipment) have been proposed to be located outside of the 1% AEP + CC extent and/or the 0.1% Annual Probability Surface Water proxy extent. The solar panels will be mounted on raised frames above surrounding ground level allowing flood water to flow freely underneath.
		The proposed development is free draining through perimeter gaps around all panels, allowing for infiltration as existing within the grassland/vegetation surrounding and beneath the panels. There will be minimal increase in



		impermeable area meaning the proposals will not increase surface water flood risk elsewhere.
Paragraph 5.7.18	To satisfactorily manage flood risk, arrangements are required to manage surface water and the impact of the natural water cycle on people and property	FRAs are provided in Appendices 10.1-10.6 of the ES [APP-089 to APP-094]. This considers the effects of a range of flooding events including extreme events on people, property, the natural and historic environment and river and coastal processes.
		ES Chapter 10 [APP-048] appendices have set out allowable discharge rates which will be achieved through sustainable drainage systems SuDS Strategy contained within the ES [APP-039 to APP-061].
		Flood risks during construction and decommissioning have been set out within the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and the Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] where proposed mitigation, both embedded and additional, have been captured to mitigate potential flood risk.
Paragraph 5.7.19	In this NPS, the term Sustainable Drainage Systems (SuDS) refers to the whole range of sustainable approaches to surface water drainage management including, where appropriate: • source control measures including rainwater recycling and drainage;	ES Chapter 10 [APP-048]] appendices have set out allowable discharge rates which will be achieved through sustainable drainage systems SuDS Strategy contained within the ES [APP-039 to APP-061].
	 infiltration devices to allow water to soak into the ground, that can include individual soakaways and communal facilities; 	



	filter strips and swales, which are vegetated features that hold and drain water downhill mimicking natural drainage patterns;	
	 filter drains and porous pavements to allow rainwater and run-off to infiltrate into permeable material below ground and provide storage if needed; 	
	 basins ponds and tanks to hold excess water after rain and allow controlled discharge that avoids flooding; and 	
	 flood routes to carry and direct excess water through developments to minimise the impact of severe rainfall flooding. 	
Paragraph 5.7.20	Site layout and surface water drainage systems should cope with events that exceed the design capacity of the system, so that excess water can be safely stored on or conveyed from the site without adverse impacts.	As detailed within the Flood Risk Assessment and Drainage Strategy [APP-089], the Scheme is undeveloped agricultural land and wholly permeable, informally draining to ground and in exceedance events in excess of the infiltration capacity, into the surrounding Land Drains. Any surface water runoff in excess of the infiltration capacity of the ground may naturally drain into the surrounding land drains as per the existing scenario.
Paragraph 5.7.21	The surface water drainage arrangements for any project should be such that the volumes and peak flow rates of surface water leaving the site are no greater than the rates prior to the proposed project, unless specific off-site arrangements are made and result in the same net effect.	As detailed within the Flood Risk Assessment and Drainage Strategy [APP-089], the Scheme is not considered to result in an increase in volumes or peak flow rates of surface water across the Solar Panels. There will be minimal increase in impermeable area meaning the proposals will not increase surface water flood risk elsewhere.
		These demonstrate that the Scheme is in accordance with this policy.



Paragraph 5.7.22	It may be necessary to provide surface water storage and infiltration to limit and reduce both the peak rate of discharge from the site and the total volume discharged from the site. There may be circumstances where it is appropriate for infiltration facilities or attenuation storage to be provided outside the project site, if necessary, through the use of a planning obligation.	The proposed development is free draining through perimeter gaps around all panels, allowing for infiltration as existing within the grassland/vegetation surrounding and beneath the panels. The Scheme proposes to maintain the existing surface water run-off regime by utilising permeable surfacing for the Site access, linear infiltration trenches around any proposed infrastructure (substations and batteries) and wildflower planting at the leeward edge of solar panels. It is considered, through proposed mitigation, that the residual effect is considered to be negligible and the Scheme is therefore considered to comply with this Policy.
Paragraph 5.7.23	The sequential approach should be applied to the layout and design of the project. More vulnerable uses should be located on parts of the site at lower probability and residual risk of flooding. Applicants should seek opportunities to use open space for multiple purposes such as amenity, wildlife habitat and flood storage uses. Opportunities should be taken to lower flood risk by reducing the built footprint of previously developed sites and using SuDS.	As stated in the FRAs provided in Appendix 10.1 the ES [APP-089], the majority above ground development is located out of Flood Zones 2 and 3 (including climate change allowance). Where development is proposed within Flood Zones 2 and 3, sensitive electrical equipment will be raised 0.6 m above the 0.1% AEP + CC flood level or where this is not possible as high as practicably possible.
Paragraph 5.7.24	Essential energy infrastructure which has to be located in flood risk areas should be designed to remain operational when floods occur. In addition, any energy projects proposed in Flood Zone 3b the Functional Floodplain (where water has to flow or be stored in times of flood), or Zone C2 in Wales, should only be permitted if the development will not result in a net loss of floodplain storage, and will not impede water flows.	As stated in the FRAs provided in Appendices 10.1 – 10.6 of the ES, the majority of above ground development is located out of Flood Zones 2 and 3 (including climate change allowance). Where development is proposed within Flood Zones 2 and 3, sensitive electrical equipment will be raised 0.6 m above the 0.1% AEP + CC flood level or where this is not possible as high as practicably possible.



Paragraph	l
5.7.25	

The receipt of and response to warnings of floods is an essential element in the management of the residual risk of flooding. Flood Warning and evacuation plans should be in place for those areas at an identified risk of flooding. The applicant should take advice from the emergency services when producing an evacuation plan for a manned energy project as part of the FRA. Any emergency planning documents, flood warning and evacuation procedures that are required should be identified in the FRA.

Where development is proposed within Flood Zones 2 and 3, sensitive electrical equipment will be raised 0.6 m above the 0.1% AEP + CC flood level or where this is not possible as high as practicably possible. Areas at risk of flooding will not be occupied by operational staff.

The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] sets out measures to ensure the safety of staff during construction from flood risk. This includes the appointment of at least one designated Flood Warden who is familiar with the risks and remains vigilant to news reports, Environment Agency flood warnings, relevant weather warnings and water levels of the local waterway.

The Outline OEMP [REP5-020] sets out that Staff on site will undertake regular weather checks to forecast any heavy rain events and to prepare for flooding where necessary. Areas of the Order limits at risk of flooding are not expected to be frequently occupied by staff and access to the Solar Farm Site is primarily located in Flood Zone 1.

Paragraph 5.8.2

The historic environment includes all aspects of the environment resulting from the interaction between people and places through time, including all surviving physical remains of past human activity, whether visible, buried or submerged, landscaped and planted or managed flora. Those elements of the historic environment that hold value to this and future generations because of their historic, archaeological, architectural or artistic interest are called "heritage assets". A heritage asset may be any building, monument, site, place, area or landscape, or any combination of these. The sum of the heritage interests that a heritage asset holds is referred to as its significance.

Heritage assets as defined in this policy have been considered and where relevant assessed in Chapter 13: Cultural Heritage of the ES [APP-051]. Section 13.5 of Chapter 13: Cultural Heritage of the ES describes the significance of these assets.

The ES [APP-039 to APP-061] has therefore identified a suitable baseline from which to assess the Scheme in relation to this policy.



Paragraph 5.8.3	Some heritage assets have a level of significance that justifies official designation. Categories of designated heritage assets are: a World Heritage Site; Scheduled Monument; Protected Wreck Site; Protected Military Remains, Listed Building; Registered Park and Garden; Registered Battlefield; Conservation Area; and Registered Historic Landscape (Wales only)	Designated heritage assets are identified in Chapter 13: Cultural Heritage of the ES [APP-051]. Section 13.5 of Chapter 13: Cultural Heritage of the ES [APP-051] describes the significance of these assets.
Paragraph 5.8.4	There are heritage assets with archaeological interest that are not currently designated as scheduled monuments, but which are demonstrably of equivalent significance. These include: • those that have yet to be formally assessed for designation; • those that have been assessed as being designatable but which the Secretary of State has decided not to designate; and • those that are incapable of being designated by virtue of being outside the scope of the Ancient Monuments and Archaeological Areas Act 1979.	Non designated heritage assets are identified in Chapter 13: Cultural Heritage of the ES [APP-051]. Section 13.5 of Chapter 13: Cultural Heritage of the ES describes these assets and their significance. The assessment made in Section 13.7 concludes that these assets are beyond the Order Limits but that there are a few non-designated heritage assets which are anticipated to experience significant adverse effects from the Scheme. As none of the non-designated assets are of equal significance to designated assets, then the substantial harm test does not apply. The significant public benefits of the Scheme clearly and demonstrably outweigh the reversible, low level, less than substantial harm to non-designated heritage assets, that would result.
Paragraph 5.8.5	The absence of designation for such heritage assets does not indicate lower significance. If the evidence before the IPC indicates to it that a non designated heritage asset of the type described in 5.8.4 may be affected by the proposed development, then the heritage asset should be considered subject to the same policy considerations as those that apply to designated heritage assets.	Non designated heritage assets are assessed within Chapter 13: Cultural Heritage of the ES [APP-051]. Following embedded and additional mitigation measures being incorporated, the following receptors are still expected to experience adverse significant effects: HLI21266, HLI20787, HLI20791 and HLI20860. Holistically, the Scheme is considered to have given due consideration



		to and in mitigating for harm against non-designated heritage assets.
Paragraph 5.8.6	The IPC should also consider the impacts on other non-designated heritage assets, as identified either through the development plan making process (local listing) or through the IPC's decision-making process on the basis of clear evidence that the assets have a heritage significance that merits consideration in its decisions, even though those assets are of lesser value than designated heritage assets.	The impacts upon non-designated heritage assets are identified in Section 13.11 of Chapter 13: Cultural Heritage of the ES [APP-051]. Appendix 13.8 of Chapter 13: Cultural Heritage of the ES [APP-051] also describes these assets and their significance. Impacts on non-designated heritage assets are also presented in Appendix 13.8. The ES [APP-039 to APP-061] therefore considers impacts on non-designated heritage buildings as required by this policy.
Paragraph 5.8.8	As part of the ES (see Section 4.2) the applicant should provide a description of the significance of the heritage assets affected by the proposed development and the contribution of their setting to that significance. The level of detail should be proportionate to the importance of the heritage assets and no more than is sufficient to understand the potential impact of the proposal on the significance of the heritage asset. As a minimum the applicant should have consulted the relevant Historic Environment Record (or, where the development is in English or Welsh waters, English Heritage or Cadw) and assessed the heritage assets themselves using expertise where necessary according to the proposed development's impact.	Chapter 13: Cultural Heritage of the ES [APP-051] describes these assets (having considered the HER and through the Applicant's own assessment) and their significance, and the contribution of their setting to that significance. The level of detail is considered to be proportionate to the significance of the asset. The ES [APP-039 to APP-061] is therefore in full compliance with this policy.
Paragraph 5.8.9	Where a development site includes, or the available evidence suggests it has the potential to include, heritage assets with an archaeological interest, the applicant should carry out appropriate desk-based assessment and, where such desk-based research is insufficient to properly assess the interest, a field evaluation. Where	Archaeological evaluations were undertaken in addition to a desk-based assessment, including a geophysical survey (detailed magnetometry) of the whole Scheme and targeted trial trenching. The results of these surveys (Appendix 13.1, Appendix 13.2 and Appendix 13.3 of the ES



	proposed development will affect the setting of a heritage asset, representative visualisations may be necessary to explain the impact.	[APP-039 to APP-061]) have been incorporated in Section ES Chapter 13: Cultural Heritage [APP-051].
Paragraph 5.8.10	The applicant should ensure that the extent of the impact of the proposed development on the significance of any heritage assets affected can be adequately understood from the application and supporting documents.	Section 13.5 of Chapter 13: Cultural Heritage of the ES [APP-051] describes the heritage assets within the study area for the Scheme and their significance and the contribution of their setting to that significance. The Chapter contains a clear assessment of likely impacts and effects of the Scheme on cultural heritage. The ES [APP-039 to APP-061] is therefore in full compliance with this policy.
Paragraph 5.8.12	In considering the impact of a proposed development on any heritage assets, the IPC should take into account the particular nature of the significance of the heritage assets and the value that they hold for this and future generations. This understanding should be used to avoid or minimise conflict between conservation of that significance and proposals for development.	Section 13.5 of Chapter 13: Cultural Heritage of the ES [APP-051] describes the heritage assets within the study area for the Scheme and their significance and the contribution of their setting to that significance. The Chapter contains a clear assessment of likely impacts and effects of the Scheme on cultural heritage. The ES [APP-039 to APP-061] is therefore in full compliance with this policy.
Paragraph 5.8.13	The IPC should take into account the desirability of sustaining and, where appropriate, enhancing the significance of heritage assets, the contribution to their settings and the positive contribution they can make to sustainable communities and economic vitality. The IPC should take into account the desirability of new development making a positive contribution to the character and local distinctiveness of the historic environment. The consideration of design should include scale, height, massing, alignment, materials and use. The IPC should have regard to any relevant local	Section 13.8 of Chapter 13: Cultural Heritage of the ES [APP-051] outlines the mitigation measures embedded within the Scheme design pertaining to cultural heritage. This includes the provision of stand-offs between the Scheme and heritage assets in order to help to preserve their setting during the construction, operational and decommissioning periods.



	authority development plans or local impact report on the proposed development in respect of the factors set out in footnote 122.	Appropriate and sensitive screening has also been developed and implemented to minimise the visual intrusion of the Scheme, while avoiding obscuring or intruding upon key views and relationships between heritage assets. Following decommissioning, the solar farm will be removed, and its impact on the setting of heritage assets reversed.
Paragraph 5.8.14	There should be a presumption in favour of the conservation of designated heritage assets and the more significant the designated heritage asset, the greater the presumption in favour of its conservation should be. Once lost heritage assets cannot be replaced and their loss has a cultural, environmental, economic and social impact. Significance can be harmed or lost through alteration or destruction of the heritage asset or development within its setting. Loss affecting any designated heritage asset should require clear and convincing justification. Substantial harm to or loss of a grade II listed building park or garden should be exceptional. Substantial harm to or loss of designated assets of the highest significance, including Scheduled Monuments; registered battlefields; grade I and II* listed buildings; grade I and II* registered parks and gardens; and World Heritage Sites, should be wholly exceptional.	Section 13.7 of Chapter 13: Cultural Heritage [APP-051] sets out the impacts upon designated heritage assets, including their value. During construction, following mitigation, the medieval bishop's palace and deer park (Stow Park) and the Church of St Botolph (Saxilby with Ingleby) are to experience a moderate adverse impact, being significant in EIA terms. During operation, following mitigation, there is not expected to be any significant adverse impacts upon designated heritage assets. During decommissioning, following mitigation, the medieval bishop's palace and deer park (Stow Park) is expected to experience a moderate adverse impact, being significant in EIA terms. Section 6.6 of the Planning Statement
		[EN010132/EX6EX7/WB7.5_CD], set out the harm policy test. The significant public benefits of the Scheme clearly



		and demonstrably outweigh the reversible harm that would result.
Paragraph 5.8.15	Any harmful impact on the significance of a designated heritage asset should be weighed against the public benefit of development, recognising that the greater the harm to the significance of the heritage asset the greater the justification will be needed for any loss. Where the application will lead to substantial harm to or total loss of significance of a designated heritage asset the IPC should refuse consent unless it can be demonstrated that the substantial harm to or loss of significance is necessary in order to deliver substantial public benefits that outweigh that loss or harm.	Section 13.9 of Chapter 13: Cultural Heritage [APP-051] sets out the impacts upon designated heritage assets, including their value. Section 6.6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], set out the harm policy test. The significant public benefits of the Scheme clearly and demonstrably outweigh the reversible harm that would result.
		Section 13.8 of Chapter 13: Cultural Heritage of the ES [APP-051] outlines the mitigation measures embedded within the Scheme design pertaining to cultural heritage. This includes the provision of stand-offs between the Scheme and heritage assets in order to help to preserve their setting during the construction, operational and decommissioning periods.
	Appropriate and sensitive screening has also been developed and implemented to minimise the visual intrusion of the Scheme, while avoiding obscuring or intruding upon key views and relationships between heritage assets.	
		In addition to the proposed mitigation strategy, the Scheme will be decommissioned at the end of its operational life. There will therefore be no permanent loss of the significance of designated assets as a result of the



		Scheme allowing future generations to retain an understanding of their settings. The Statement of Need [APP-320] explains in detail the compelling case for the Scheme in relation to urgently delivering low carbon renewable energy to meet the aim of decarbonising the UK's electricity supplies by 2050; providing security of supply as well as affordability for end consumers. The less than substantial harm caused to one designated heritage asset is outweighed by this urgent national need.
Paragraph 5.8.16	Not all elements of a World Heritage Site or Conservation Area will necessarily contribute to its significance. The policies set out in paragraphs 5.8.11 to 5.8.15 above apply to those elements that do contribute to the significance. When considering proposals, the IPC should take into account the relative significance of the element affected and its contribution to the significance of the World Heritage Site or Conservation Area as a whole.	There are no World Heritage Sites affected by the Scheme. ES Chapter 13: Cultural Heritage of the ES [APP-051] includes an assessment of the Scheme upon Conservation Area within 5km of the Scheme, attached in Appendix 13.5 [APP-293]. This assessment concludes that on the whole, there will be a negligible to slight effect on the Conservation Area, which is not significant in EIA terms. The Scheme therefore does not lead to significant adverse effects to a World Heritage Site of Conservation Area and complies with this policy.
Paragraph 5.8.17	Where loss of significance of any heritage asset is justified on the merits of the new development, the IPC should consider imposing a condition on the consent or requiring the applicant to enter into an obligation that will prevent the loss occurring until it is reasonably certain that the relevant part of the development is to proceed.	The impact of the Scheme on heritage assets has been assessed and the impacts reported through section 13.7 Chapter 13: Cultural Heritage of the ES [APP-051].



Paragraph 5.8.18	When considering applications for development affecting the setting of a designated heritage asset, the IPC should treat favourably applications that preserve those elements of the setting that make a positive contribution to, or better reveal the significance of, the asset. When considering applications that do not do this, the IPC should weigh any negative effects against the wider benefits of the application. The greater the negative impact on the significance of the designated heritage asset, the greater the benefits that will be needed to justify approval.	The Scheme has been carefully designed to incorporate stand-offs between the Scheme and heritage assets in order to help to preserve important elements of their setting during the construction, operational and decommissioning periods. The need for the stand-offs to help preserve important elements of their setting has been carefully considered in the context of the need for the generation of renewable energy by the Scheme, as set out by the Statement of Need [APP-320]. Taking account of this, it is considered that the negative impacts of the Scheme on designated heritage assets are outweighed by the need and benefits of the Scheme. In addition, it is noted that the impacts of the solar farm on the setting of designated heritage assets would be reversed following decommissioning at the end of its operational life.
Paragraph 5.8.19	A documentary record of our past is not as valuable as retaining the heritage asset and therefore the ability to record evidence of the asset should not be a factor in deciding whether consent should be given.	Chapter 13: Cultural Heritage of the ES [APP-051] states that where no appropriate design mitigation can be applied to the management of the archaeological resource within the footprint of the Scheme, a programme of archaeological recording will be undertaken commensurate with the significance of the asset.
Paragraph 5.8.20	Where the loss of the whole or a material part of a heritage asset's significance is justified, the IPC should require the developer to record and advance understanding of the significance of the heritage asset before it is lost. The extent of the requirement should be proportionate to the nature and level of the asset's significance. Developers should be required to publish this evidence and deposit copies of the reports with the	Chapter 13: Cultural Heritage of the ES [APP-051] states that where no appropriate design mitigation can be applied to the management of the archaeological resource within the footprint of the Scheme, a programme of archaeological recording will be undertaken commensurate with the significance of the asset.



	relevant Historic Environment Record. They should also be required to deposit the archive generated in a local museum or other public depository willing to receive it.	
Paragraph 5.8.21	Where appropriate, the IPC should impose requirements on a consent that such work is carried out in a timely manner in accordance with a written scheme of investigation that meets the requirements of this Section and has been agreed in writing with the relevant Local Authority (where the development is in English waters, the Marine Management Organisation and English Heritage, or where it is in Welsh waters, the MMO and Cadw)) and that the completion of the exercise is properly secured.	Chapter 13: Cultural Heritage of the ES [APP-051] sets out that archaeological evaluations have been undertaken to refine and augment the desk-based data, including a geophysical survey (detailed magnetometry) of the whole scheme and targeted trial trenching. In addition, Chapter 13: Cultural Heritage of the ES [APP-051] also sets out that extensive areas of intrusive ground activities required for the Scheme will be subject to an archaeological evaluation prior to or during construction.
Paragraph 5.8.22	Where the IPC considers there to be a high probability that a development site may include as yet undiscovered heritage assets with archaeological interest, the IPC should consider requirements to ensure that appropriate procedures are in place for the identification and treatment of such assets discovered during construction.	Chapter 13: Cultural Heritage of the ES [APP-051] sets out that extensive areas of intrusive ground activities required as part of the Scheme will be subject to an archaeological evaluation prior to or during construction. This includes the following activities which could result in impacts to the archaeological resource which have not been investigated by the prior programme of trial trenching evaluation: • Electrical Cables (Works Order Nos. 1, 2 and 5) – programme of archaeological trial trenching and/or archaeological monitoring of intrusive activities; • Grid Connection Route (Work No 4.) – programme of archaeological trial trenching and/or archaeological monitoring of intrusive activities.
Paragraph 5.9.5	The applicant should carry out a landscape and visual assessment and report it in the ES. (See Section 4.2) A number of guides have been produced to assist in addressing	An assessment of the potential landscape and visual impacts associated with the construction, operation and



	landscape issues. The landscape and visual assessment should include reference to any landscape character assessment and associated studies as a means of assessing landscape impacts relevant to the proposed project. The applicant's assessment should also take account of any relevant policies based on these assessments in local development documents in England and local development plans in Wales.	decommissioning of the Scheme has been carried out and is presented in Chapter 8: Landscape and Visual Assessment of the ES [APP-046]. Section 8.5 of Chapter 8: Landscape and Visual Assessment of the ES [APP-046] outlines the relevant landscape character assessments and related studies at national, regional, county and neighbourhood levels. As demonstrated in the local policy sections of this Accordance Table below, the landscape and visual impact assessment has taken account of relevant policies in local development documents.
		Refer to Figure 8.5 [APP-157], which illustrates Landscape Character Areas at the Regional Level and the assessment undertaken within the LVIA Chapter 8 [APP-046].
		The Scheme therefore demonstrates full compliance with this policy.
Paragraph 5.9.6	The applicant's assessment should include the effects during construction of the project and the effects of the completed development and its operation on landscape components and landscape character.	An assessment of the potential landscape and visual impacts associated with the construction, operation and decommissioning of the Scheme has been carried out and is presented in Chapter 8: Landscape and Visual Assessment of the ES [APP-046].
		The Scheme therefore demonstrates full compliance with this policy.
Paragraph 5.9.7	The assessment should include the visibility and conspicuousness of the project during construction and of the presence and operation of the project and potential	The assessment contained in Chapter 8: Landscape and Visual Assessment of the ES [APP-046] includes the potential landscape and visual impacts associated with the



	impacts on views and visual amenity. This should include light pollution effects, including on local amenity, and nature conservation.	construction, operation and decommissioning of the Scheme (including light pollution effects) on local amenity and nature conservation.
		The Scheme therefore demonstrates full compliance with this policy.
Paragraph 5.9.8	Landscape effects depend on the existing character of the local landscape, its current quality, how highly it is valued and its capacity to accommodate change. All of these factors need to be considered in judging the impact of a project on landscape. Virtually all nationally significant energy infrastructure projects will have effects on the landscape. Projects need to be designed carefully, taking account of the potential impact on the landscape. Having regard to siting, operational and other relevant constraints the aim should be to minimise harm to the landscape, providing reasonable mitigation where possible and appropriate.	This is undertaken within the LVIA Chapter 8 [APP-046] in sections 8.6 and 8.8. Good design has been a key consideration from the outset. The LVIA has informed the iterative design process, including taking account of published landscape character assessment guidance and fieldwork analysis. The overall objective of the landscape design is to integrate the Scheme into its landscape setting and avoid or minimise adverse landscape and visual effects as far as practicable. The design has been developed in collaboration with the wider design team, other specialists and the Host Authorities landscape advisors to achieve a solution that achieves this objective whilst maximising opportunities to deliver net gains in biodiversity gain. Accordingly, the landscape design aims to achieve the following: • To integrate the Scheme into the existing landscape pattern as far as possible by retaining and following existing features, including vegetation, where practicable.



		 To retain vegetation as far as possible and enhance the quality and connectivity of green infrastructure through carefully designed planting that is sensitive to the character of the area. To filter and screen more prominent components of the Scheme in views from visual receptors.
		Details of the landscape measures embedded into the Scheme design, including a summary of their environmental functions, is presented in the Outline LEMP [EN010132/EX6/WB7.3_E].REP6-025].
Paragraph 5.9.9	National Parks, the Broads and AONBs have been confirmed by the Government as having the highest status of protection in relation to landscape and scenic beauty. Each of these designated areas has specific statutory purposes which help ensure their continued protection and which the IPC should have regard to in its decision. The conservation of the natural beauty of the landscape and countryside should be given substantial weight by the IPC in deciding on applications for development consent in these areas.	As outlined in Section 8.5 of Chapter 8: Landscape and Visual Assessment of the ES [APP-046], neither the study area, nor the DCO Site, is covered by any statutory landscape designations (e.g., National Parks or Areas of Outstanding Natural Beauty). No impacts on National Parks, the Broads and AONBs have been identified.
Paragraph 5.9.12	The duty to have regard to the purposes of nationally designated areas also applies when considering applications for projects outside the boundaries of these areas which may have impacts within them. The aim should be to avoid compromising the purposes of designation and such projects should be designed sensitively given the various siting, operational, and other relevant constraints. This should include projects in England which may have impacts on National Scenic Areas in Scotland.	As stipulated by this policy, Appendix 8.2.3 [APP-073] of Chapter 8: Landscape and Visual Assessment of the ES [APP-046] shows regard for nationally designated areas. No impacts on National Parks, the Broads and AONBs have been identified.
Paragraph 5.9.13	The fact that a proposed project will be visible from within a designated area should not in itself be a reason for refusing consent.	There are no National Parks (or the Broads) or AONB near to the Order limits. The Scheme would not be visible from any such site.



Paragraph 5.9.14	Outside nationally designated areas, there are local landscapes that may be highly valued locally and protected by local designation. Where a local development document in England or a local development plan in Wales has policies based on landscape character assessment, these should be paid particular attention. However, local landscape designations should not be used in themselves to refuse consent, as this may unduly restrict acceptable development.	In line with this policy, section 8.5 of Chapter 8: Landscape and Visual Assessment of the ES [APP-046] outlines the relevant landscape character assessments and related studies at national, regional, county and neighbourhood levels. Refer to Figure 8.6 [APP-159] which illustrate Landscape Receptors and illustrate Areas of Great Landscape Value. The Scheme is located outside of these areas; however, the Scheme has the potential to affect these local designations and therefore an assessment of effects on these local designations is undertaken within section 8.7 of LVIA Chapter 8 [APP-046].
Paragraph 5.9.15	The scale of such projects means that they will often be visible within many miles of the site of the proposed infrastructure. The IPC should judge whether any adverse impact on the landscape would be so damaging that it is not offset by the benefits (including need) of the project.	The assessment presented in Chapter 8: Landscape and Visual Assessment of the ES [APP-046] concludes that the Scheme would have 'not significant' effects on the landscape. Effects would be further minimised where possible through measures set out in the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline Landscape and Ecological Management Plan [EN010132/EX6/WB7.3_EREP6-025] and the Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023]. During the operational phase (Year 15) of the Scheme, Chapter 8: Landscape and Visual Assessment of the ES [APP-046] the following receptors are expected to



		experience moderate adverse residual effects post additional mitigation: Viewpoint Receptors: VP 8 and VP 24. Viewpoint Receptors: VP 18, VP 26, VP 27 and VP 28. Transport Receptors: TO09 and TO10. PRoW Receptors: PR007 and PR038.
Paragraph 5.9.16	In reaching a judgment, the IPC should consider whether any adverse impact is temporary, such as during construction, and/or whether any adverse impact on the landscape will be capable of being reversed in a timescale that the IPC considers reasonable.	Construction and decommissioning stage impacts will be for a relatively short duration, and operational effects beginning at year 1 will reduce over time as mitigation planting matures, as set out in Outline LEMP [EN010132/EX6/WB7.3_EREP6-025]. The change to the landscape character, via the introduction of solar panels and associated infrastructure is considered to be localised and is reversible following decommissioning.
Paragraph 5.9.17	The IPC should consider whether the project has been designed carefully, taking account of environmental effects on the landscape and siting, operational and other relevant constraints, to minimise harm to the landscape, including by reasonable mitigation.	Good design has been a key consideration from the outset. The LVIA has informed the iterative design process, including taking account of published landscape character assessment guidance and fieldwork analysis. The overall objective of the landscape design is to integrate the Scheme into its landscape setting and avoid or minimise adverse landscape and visual effects as far as practicable. The design has been developed in collaboration with the wider design team, other specialists and the Host Authorities landscape advisors to achieve a



		solution that achieves this objective whilst maximising opportunities to deliver net gains in biodiversity gain. Accordingly, the landscape design aims to achieve the following:
		To integrate the Scheme into the existing landscape pattern as far as possible by retaining and following existing features, including vegetation, where practicable.
		To replace vegetation lost because of construction of the Scheme through areas of new planting.
		To filter and screen more prominent components of the Scheme in views from visual receptors.
		Details of the landscape measures embedded into the Scheme design, including a summary of their environmental functions, is presented in the Outline LEMP [EN010132/EX6/WB7.3_E].REP6-025].
Paragraph 5.9.18	All proposed energy infrastructure is likely to have visual effects for many receptors around proposed sites. The IPC will have to judge whether the visual effects on sensitive receptors, such as local residents, and other receptors, such as visitors to the local area, outweigh the benefits of the project. Coastal areas are particularly vulnerable to visual intrusion because of the potential high visibility of development on the foreshore, on the skyline and affecting views along stretches of undeveloped coast.	Chapter 8: Landscape and Visual Assessment of the ES [APP-046] and Appendix 8.3, Visual Assessment of Residential Properties Methodology of the ES [APP-074] have assessed the: construction; short term operational (lasts for up to 12 months); medium term operational (lasts for 1 - 5 years); long term operational (more than 5 years) and decommissioning visual impacts of the Scheme.
		Visual effects on PRoW have been reduced by the inclusion of more open areas at strategic points in the PRoW and permissive path network, and by the provision of



		alternative routes which may be less 'enclosed'. The significant effects identified on PRoW cannot practically be further mitigated without a reduction in electrical output from the Scheme. It is not therefore considered that the localised visual effects on transient recreational receptors (PRoW users) predicted would outweigh the national benefits of the Scheme, outlined in detail in the Statement of Need [APP-320]. The Scheme therefore shows compliance with this policy.
Paragraph 5.9.21	Reducing the scale of a project can help to mitigate the visual and landscape effects of a proposed project. However, reducing the scale or otherwise amending the design of a proposed energy infrastructure project may result in a significant operational constraint and reduction in function – for example, the electricity generation output. There may, however, be exceptional circumstances, where mitigation could have a very significant benefit and warrant a small reduction in function. In these circumstances, the IPC may decide that the benefits of the mitigation to reduce the landscape and/or visual effects outweigh the marginal loss of function.	Good design has been a key consideration from the outset. The LVIA has informed the iterative design process, including taking account of published landscape character assessment guidance and fieldwork analysis. The overall objective of the landscape design is to integrate the Scheme into its landscape setting and avoid or minimise adverse landscape and visual effects as far as practicable. The design has been developed in collaboration with the wider design team, other specialists and the Host Authorities landscape advisors to achieve a solution that achieves this objective. This has included consideration of the scale of the project throughout the design development process.
Paragraph 5.9.22	Within a defined site, adverse landscape and visual effects may be minimised through appropriate siting of infrastructure within that site, design including colours and materials, and landscaping schemes, depending on the size and type of the proposed	Good design has been a key consideration from the outset. The LVIA has informed the iterative design process,



project. Materials and designs of buildings should always be given careful consideration.

including taking account of published landscape character assessment guidance and fieldwork analysis.

The overall objective of the landscape design is to integrate the Scheme into its landscape setting and avoid or minimise adverse landscape and visual effects as far as practicable. The design has been developed in collaboration with the wider design team, other specialists and the Host Authorities landscape advisors to achieve a solution that achieves this objective whilst maximising opportunities to deliver net gains in biodiversity gain. Accordingly, the landscape design aims to achieve the following:

- To integrate the Scheme into the existing landscape pattern as far as possible by retaining and following existing features, including vegetation, where practicable.
- To replace vegetation lost because of construction of the Scheme through areas of new planting.
- To filter and screen more prominent components of the Scheme in views from visual receptors.

Details of the landscape measures embedded into the Scheme design, including a summary of their environmental functions, is presented in the Outline LEMP [EN010132/EX6/WB7.3_E].REP6-025].



Paragraph 5.9.23	Depending on the topography of the surrounding terrain and areas of population it may be appropriate to undertake landscaping off site. For example, filling in gaps in existing tree and hedge lines would mitigate the impact when viewed from a more distant vista.	No offsite landscaping is required or proposed.
Paragraph 5.10.2	The Government's policy is to ensure there is adequate provision of high-quality open space (including green infrastructure) and sports and recreation facilities to meet the needs of local communities. Open spaces, sports and recreational facilities all help to underpin people's quality of life and have a vital role to play in promoting healthy living. Green infrastructure in particular will also play an increasingly important role in mitigating or adapting to the impacts of climate change.	The Scheme is not anticipated to have any direct effect on open space or recreational facilities. There are several PRoWs within or abutting the Scheme. These are shown in Appendix 14.3 Public Rights of Way Management Plan [REP5-018]. The PRoWs are detailed in Section 18.7 of Chapter 18: Socio-Economics, Tourism and Recreation of the ES [APP-056]. These PRoW are predominantly used for recreational purposes and form part of a wide network of PRoW in the surrounding area providing residents with alternative routes. They will be kept open and on their existing alignment throughout the operational phase of the Scheme. During construction, PRoW will be kept open, and on their existing alignment as far as possible, with short, convenient, temporary diversions included where this is not possible. The Scheme will also create a new permissive footpath to run from the track off Sykes Lane along the Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub. This will enhance the network of routes and accessibility within and across the Order limits.



		It is therefore considered that the Scheme accords with this policy.
Paragraph 5.10.3	Although the re-use of previously developed land for new development can make a major contribution to sustainable development by reducing the amount of countryside and undeveloped greenfield land that needs to be used, it may not be possible for many forms of energy infrastructure.	This policy accepts that previously developed land may not be possible to use for many forms of infrastructure, as in the case of this Scheme. An assessment of the potential use of previously developed land is included within the Site Selection Assessment [AS-004].
Paragraph 5.10.5	The ES (see Section 4.2) should identify existing and proposed land uses near the project, any effects of replacing an existing development or use of the site with the proposed project or preventing a development or use on a neighbouring site from continuing. Applicants should also assess any effects of precluding a new development or use proposed in the development plan.	Section 18.5 of Chapter 18: Socio-Economics Tourism and Recreation of the ES [APP-056] identifies the existing baseline land use and socio-economic conditions for the Order limits, including the existing arable agricultural use of the majority of the site, and takes account of these in its assessment.
		Section 2 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] identifies the planning history associated with the Order limits and nearby mineral and waste sites.
Paragraph 5.10.6	Applicants will need to consult the local community on their proposals to build on open space, sports or recreational buildings and land. Taking account of the consultations, applicants should consider providing new or additional open space including green infrastructure, sport or recreation facilities, to substitute for any losses as a result of their proposal. Applicants should use any upto-date local authority assessment or, if there is none, provide an independent assessment to show whether the existing open space, sports and recreational buildings and land is surplus to requirements.	The Scheme does not include any proposals to build on open space, sports or recreational buildings and land.



Paragraph 5.10.8	Applicants should seek to minimise impacts on the best and most versatile agricultural land (defined as land in grades 1, 2 and 3a of the Agricultural Land Classification) and preferably use land in areas of poorer quality (grades 3b, 4 and 5) except where this would be inconsistent with other sustainability considerations. Applicants should also identify any effects and seek to minimise impacts on soil quality taking into account any mitigation measures proposed. For developments on previously developed land, applicants should ensure that they have considered the risk posed by land contamination.	The majority of the Order Limits comprises Grade 3b agricultural land, and 26.24% BMV land is included within the Order Limits. This is justified by other sustainability considerations, as explained in Section 6.7 of this Planning Statement [EN010132/EX6EX7/WB7.5_CD].
Paragraph 5.10.9	Applicants should safeguard any mineral resources on the proposed site as far as possible, taking into account the long-term potential of the land use after any future decommissioning has taken place.	ES Chapter 3: The Order Limits [APP-041] identifies, at a broad scale, the Mineral Resource areas that are contained within the Order Limits.
		ES Chapter 12: Minerals [APP-050] examines the impacts of the Scheme on safeguarded minerals. It is concluded that the Scheme is not anticipated to have any adverse impacts upon safeguarded minerals.
Paragraph 5.10.13	Where the project conflicts with a proposal in a development plan, the IPC should take account of the stage which the development plan document in England or local development plan in Wales has reached in deciding what weight to give to the plan for the purposes of determining the planning significance of what is replaced, prevented or precluded. The closer the development plan document in England or local development plan in Wales is to being adopted by the LPA, the greater weight which can be attached to it.	As illustrated in Section 5 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme does not conflict with any proposals in a Development Plan.
Paragraph 5.10.14	The IPC should not grant consent for development on existing open space, sports and recreational buildings and land unless an assessment has been undertaken either by the local authority or independently, which has shown the open space or the buildings and land to be surplus to requirements or the IPC determines that the benefits of the project (including need), outweigh the potential loss of such facilities, taking into	The Scheme does not affect any existing open space, sports and recreational buildings or land.



	account any positive proposals made by the applicant to provide new, improved or compensatory land or facilities. The loss of playing fields should only be allowed where applicants can demonstrate that they will be replaced with facilities of equivalent or better quantity or quality in a suitable location.	
Paragraph 5.10.15	The IPC should ensure that applicants do not site their scheme on the best and most versatile agricultural land without justification. It should give little weight to the loss of poorer quality agricultural land (in grades 3b, 4 and 5), except in areas (such as uplands) where particular agricultural practices may themselves contribute to the quality and character of the environment or the local economy.	The majority of the Order Limits comprises Grade 3b agricultural land and 26.24% BMV land is included. This is justified by other sustainability considerations, as explained in Section 6.7of this Planning Statement [EN010132/EX6EX7/WB7.5_CD].
Paragraph 5.10.19	Although in the case of much energy infrastructure there may be little that can be done to mitigate the direct effects of an energy project on the existing use of the proposed site (assuming that some at least of that use can still be retained post project construction) applicants should nevertheless seek to minimise these effects and the effects on existing or planned uses near the site by the application of good design principles, including the layout of the project.	The Scheme has been subject to a detailed and sensitive iterative design process. This has taken account of the context and features of the land within the Order limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts, and provision of environmental and other enhancements, where practicable. The design process and basis of design decisions taken are described in the Chapter 5: Alternatives and Design Evolution of the ES [APP-043].
Paragraph 5.10.24	Rights of way, National Trails and other rights of access to land are important recreational facilities for example for walkers, cyclists and horse riders. The IPC should expect applicants to take appropriate mitigation measures to address adverse effects on coastal access, National Trails and other rights of way. Where this is not the case	There are several PRoWs within or abutting the Scheme. These are shown in Figures 8.10.1 to 8.10.4 of the ES [APP-180 to APP-183]. These PRoW are predominantly used for recreational purposes and form part of a wide network of



	the IPC should consider what appropriate mitigation requirements might be attached to any grant of development consent.	PRoW in the surrounding area providing residents with alternative routes.
		They will be kept open and on their existing alignment throughout the operational phase of the Scheme. During construction PRoW will be kept open, and on their existing alignment as far as possible, with short, convenient, temporary diversions included where this is not possible.
		The Scheme will also create a new permissive footpath to run from the track off Sykes Lane along the Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub. This will enhance the network of routes and accessibility within and across the Order limits. It is therefore considered that the Scheme accords with this policy.
Paragraph 5.11.1	Excessive noise can have wide-ranging impacts on the quality of human life, health (for example owing to annoyance or sleep disturbance) and use and enjoyment of areas of value such as quiet places and areas with high landscape quality. The Government's policy on noise is set out in the Noise Policy Statement for England. It promotes good health and good quality of life through effective noise management. Similar considerations apply to vibration, which can also cause damage to buildings. In this section, in line with current legislation, references to "noise" below apply equally to assessment of impacts of vibration.	Chapter 15: Noise & Vibration of the ES [APP-053] recognises and assesses the impacts of noise and vibration of the Scheme on health and quality of life. It is therefore considered that the Scheme is compliant with this policy.
Paragraph 5.11.2	Noise resulting from a proposed development can also have adverse impacts on wildlife and biodiversity. Noise effects of the proposed development on ecological receptors should be assessed by the IPC in accordance with the Biodiversity and Geological Conservation section of this NPS.	Section 9.7 of Chapter 9: Ecology and Biodiversity of the ES [APP-047] includes an assessment of the likely impacts and effects of noise on relevant ecological features. It is



		therefore considered that the Scheme is compliant with this policy.
Paragraph 5.11.3	 Factors that will determine the likely noise impact include: the inherent operational noise from the proposed development, and its characteristics; the proximity of the proposed development to noise sensitive premises (including residential properties, schools and hospitals) and noise sensitive areas (including certain parks and open spaces); the proximity of the proposed development to quiet places and other areas that are particularly valued for their acoustic environment or landscape quality; and the proximity of the proposed development to designated sites where noise may have an adverse impact on protected species or other wildlife. 	Section 15.4 of Chapter 15: Noise of the ES [APP-053] and its supporting appendices explain the noise assessment methodology which has considered the factors identified by this policy. ES Chapter 15: Noise of the ES [APP-053] describes the noise sensitive premises and areas that have been identified. These have been determined through desktop study during the scoping process and confirmed during site visits. The locations of these receptors have been considered in both the construction and operational noise assessments and are considered representative of adjacent properties. Noise from the construction, operation and
		decommissioning of the Scheme is considered throughout Chapter 11 and therefore it is considered that the Scheme is compliant with this policy. Section 9.7 of Chapter 9: Ecology and Biodiversity of the ES
		[APP-047] includes an assessment of the likely impacts and effects of noise on designated ecological sites. Chapter 8: Landscape and Visual Assessment of the ES [APP-046] considers the impact of the Scheme on
		tranquillity in its assessments.



		It is therefore considered that the methodology used in the ES [APP-039 to APP-061] complies with his policy.
Paragraph 5.11.4	 Where noise impacts are likely to arise from the proposed development, the applicant should include the following in the noise assessment: a description of the noise generating aspects of the development proposal leading to noise impacts, including the identification of any distinctive tonal, impulsive or low frequency characteristics of the noise; identification of noise sensitive premises and noise sensitive areas that may be affected; the characteristics of the existing noise environment; a prediction of how the noise environment will change with the proposed 	Chapter 15: Noise and Vibration of the ES [APP-053] presents a noise assessment in accordance with the requirements of this policy. ES Chapter 15: Noise and Vibration of the ES [APP-053] describes the noise sensitive premises and areas that have been identified. These have been determined through desktop study during the scoping process and confirmed during site visits. The locations of these receptors have been considered in both the construction and operational noise assessments and are considered representative of adjacent properties. Section 15.6 of Chapter 15: Noise and Vibration of the ES [APP-053] describes the embedded design mitigation for the Scheme with respect to noise and vibration, encompassing the construction, operation and decommissioning phases. Section 15.7 of Chapter 15: Noise and Vibration of the ES [APP-053] assesses the noise generated by the Scheme
	 a prediction of now the noise environment will change with the proposed development; in the shorter term such as during the construction period; in the longer term during the operating life of the infrastructure; at particular times of the day, evening and night as appropriate. an assessment of the effect of predicted changes in the noise environment on any noise sensitive premises and noise sensitive areas; and measures to be employed in mitigating noise. The nature and extent of the 	
	noise assessment should be proportionate to the likely noise impact.	during the construction period and operating life of the infrastructure (including tonality), including at particular times of the day and at night on the noise sensitive premises and areas outlined.



Paragraph 5.11.5	The noise impact of ancillary activities associated with the development, such as increased road and rail traffic movements, or other forms of transportation, should also be considered.	The construction noise assessment is evaluated within section 15.7 Chapter 15: Noise and Vibration of the ES [APP-053]. It is concluded that construction traffic does not result in more than moderate/ minor adverse impacts which is not significant in EIA terms. It is therefore considered that the Scheme is compliant with this policy.
Paragraph 5.11.6	Operational noise, with respect to human receptors, should be assessed using the principles of the relevant British Standards and other guidance. Further information on assessment of particular noise sources may be contained in the technology-specific NPSs. In particular, for renewables (EN-3) and electricity networks (EN-5) there is assessment guidance for specific features of those technologies. For the prediction, assessment and management of construction noise, reference should be made to any relevant British Standards and other guidance which also give examples of mitigation strategies.	Section 15.7 of Chapter 15: Noise & Vibration of the ES [APP-053] assesses operational noise with respect to human receptors. As outlined in Section 15.7 of Chapter 15: Noise & Vibration of the ES [APP-053], operational plant noise has been assessed. It is therefore considered that the Scheme is compliant with this policy.
Paragraph 5.11.7	The applicant should consult EA and Natural England (NE), or the Countryside Council for Wales (CCW), as necessary and in particular with regard to assessment of noise on protected species or other wildlife. The results of any noise surveys and predictions may inform the ecological assessment. The seasonality of potentially affected species in nearby sites may also need to be taken into account.	The Applicant has taken account of advice from the EA and Natural England throughout the preparation of the Environmental Statement [APP-039 to APP-061]. Chapter 9: Ecology and biodiversity, of the ES [APP-047] takes account of noise in its assessment of the impact of the Scheme on protected species and other wildlife.
Paragraph 5.11.8	The project should demonstrate good design through selection of the quietest cost- effective plant available; containment of noise within buildings wherever possible; optimisation of plant layout to minimise noise emissions; and, where possible, the use of landscaping, bunds or noise barriers to reduce noise transmission.	As detailed in Section 15.6 of Chapter 15: Noise & Vibration of the ES [APP-053], embedded mitigation measures for the operational phase have been considered with reference to this policy.



		The concept design of the Scheme has incorporated measures such as distancing of inverters away from sensitive receptors and locating the BESS compound in an area away from large concentrations of receptors where existing ambient noise levels are higher (such that noise emissions from the BESS are less impactful).
		The embedded design will ensure the use of acoustic barriers around inverters within 250m of residential dwellings.
		Solar PV tracker panels have a Sound Pressure Level of $50.1~dB~L_{Aeq}$ at 1m distance, the assessment of these panels is to produce a negligible effect and moderate/minor significance.
Paragraph 5.11.9	 The IPC should not grant development consent unless it is satisfied that the proposals will meet the following aims: avoid significant adverse impacts on health and quality of life from noise; mitigate and minimise other adverse impacts on health and quality of life from noise; and where possible, contribute to improvements to health and quality of life through the effective management and control of noise. 	Section 15.11 of Chapter 15: Noise & Vibration of the ES [APP-053] concludes that the magnitude of change from the construction, decommissioning or operation is negligible which results in a moderate/minor residual effect. It also sets out mitigation measures to be incorporated into the Scheme to mitigate and minimise noise impacts. No noise existing issues that the Scheme could contribute to improving have been identified. The Scheme is therefore considered to accord with this policy.
Paragraph 5.11.11	The IPC should consider whether mitigation measures are needed both for operational and construction noise over and above any which may form part of the project application. In doing so the IPC may wish to impose requirements. Any such	Given the outcome of the noise and vibration assessment for the Scheme and the proposed mitigation it is not anticipated that the Secretary of State will need to consider



	requirements should take account of the guidance set out in Circular 11/95 (see Section 4.1) or any successor to it.	additional mitigation measures above those already embedded in the design of the Scheme and those set out within the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline OEMP [REP5-020] and the Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023]. It is considered that the Scheme is compliant with this policy.
Paragraph	Mitigation measures may include one or more of the following:	Given the outcome of the noise and vibration assessment
5.11.12	 engineering: reduction of noise at point of generation and containment of noise generated; 	for the Scheme and the proposed mitigation it is not anticipated that the Secretary of State will need to consider additional mitigation measures above those already
	 lay-out: adequate distance between source and noise-sensitive receptors; incorporating good design to minimise noise transmission through screening by natural barriers, or other buildings; and 	embedded in the design of the Scheme and those set out within the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline OEMP [REP5-020] and the Outline
	 administrative: restricting activities allowed on the site; specifying acceptable noise limits; and taking into account seasonality of wildlife in nearby designated sites. 	Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023].
		It is considered that the Scheme is compliant with this policy.
Paragraph 5.12.2	Where the project is likely to have socio-economic impacts at local or regional levels, the applicant should undertake and include in their application an assessment of these impacts as part of the ES (see Section 4.2).	Section 18.7 of Chapter 18: Socio-Economics, Tourism and Recreation of the ES [APP-056] includes an assessment of socio-economic impacts at local and regional levels, including employment, the local economy, users of Public Rights of Way (PRoW), residential properties, business properties and community facilities.



Paragraph 5.12.3	This assessment should consider all relevant socio-economic impacts, which may include: • the creation of jobs and training opportunities; • the provision of additional local services and improvements to local infrastructure, including the provision of educational and visitor facilities;	Chapter 18: Socio-Economics, Tourism and Recreation of the ES [APP-056] includes an assessment of socio-economic impacts in Section 18.7 for the Scheme and in 18.10 cumulatively which fulfils the requirements of this policy.
	 effects on tourism; the impact of a changing influx of workers during the different construction, operation and decommissioning phases of the energy infrastructure. This could change the local population dynamics and could alter the demand for services and facilities in the settlements nearest to the construction work (including community facilities and physical infrastructure such as energy, water, transport and waste). There could also be effects on social cohesion depending on how populations and service provision change as a result of the development; and 	
	 cumulative effects – if development consent were to be granted to for a number of projects within a region and these were developed in a similar timeframe, there could be some short-term negative effects, for example a potential shortage of construction workers to meet the needs of other industries and major projects within the region. 	
Paragraph 5.12.4	Applicants should describe the existing socio-economic conditions in the areas surrounding the proposed development and should also refer to how the development's socio-economic impacts correlate with local planning policies.	The current socio-economic baseline conditions of the study area have been described in Section 18.5 of Chapter 18: Socio-Economics, Tourism and Recreation of the ES [APP-056].



		The Scheme's compliance with local planning policies is considered in Appendix D of the Planning Statement [EN010132/EX6EX7/WB7.5_CD].
Paragraph 5.12.8	The IPC should consider any relevant positive provisions the developer has made or is proposing to make to mitigate impacts (for example through planning obligations) and any legacy benefits that may arise as well as any options for phasing development in relation to the socio-economic impacts.	The development of farmland for solar power generation involves little disturbance of the soil and includes retention of the land resource for future use. After decommissioning, the soil resource is expected to have benefitted from a recovery of soil organic matter over the operational duration of the Scheme.
		A detailed soil resource management plan will be prepared prior to the commencement of construction, prior to operation, and prior to decommissioning, as set out by the Requirements of the draft DCO [EN010132/EX6EX7/WB3.1_GH].
		Primary mitigation measures are embedded within the Scheme, these measures are set out in the respective chapters of the ES [APP-039 to APP-061], to reduce other construction and operational effects (such as noise, air quality, transport and landscape) which in turn will mitigate the effects on the local community and existing facilities from a socio-economic and land use perspective.
		Chapter 18: Socio-Economics, Tourism and Recreation of the ES [APP-039 to APP-061] identifies that the Scheme will result in beneficial effects that are significant on the local economy as a result of employment generation during the construction and decommissioning periods. During the operational phase a support system to enable



local people to be trained in the sustainable development sector will be established.

Benefits of the Scheme to the local community (other than the generation of a substantial amount of renewable energy) are set out in Section 6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD]. These include:

- A significant biodiversity net gain of 86.80% provided in habitat, 54.71% gains in hedgerow and 33.25% gains in river units.
- A new permissive footpath to run from the track off Sykes Lane along the Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub, improving connectivity across the Order limits.
- Employment during the construction phase. It is expected that an average of 296 jobs will be created during the construction period. During the operational phase, 12 FTE staff would be employed on the site.

An Outline Skills, Supply Chain and Employment Plan [EN010132/EX6/WB7.10_BREP6-027] will be prepared prior to the commencement of construction. This will set out measures that the Applicant will implement in order to advertise and promote employment opportunities associated with the Scheme in construction and operation locally.



Paragraph 5.12.9	The IPC should consider whether mitigation measures are necessary to mitigate any adverse socio-economic impacts of the development. For example, high quality design can improve the visual and environmental experience for visitors and the local community alike.	Primary mitigation measures are embedded within the Scheme, these measures are set out in the respective chapters of the ES [APP-039 to APP-061], to reduce other construction and operational effects (such as noise, air quality, transport and landscape) which in turn will mitigate the effects on the local community and existing facilities from a socio-economic and land use perspective.
Paragraph 5.13.3	If a project is likely to have significant transport implications, the applicant's ES (see Section 4.2) should include a transport assessment, using the NATA/WebTAG139 methodology stipulated in Department for Transport guidance, or any successor to such methodology. Applicants should consult the Highways Agency and Highways Authorities as appropriate on the assessment and mitigation.	Appendix 14.1 of the ES [REP4-036] contains a transport assessment. As outlined in Chapter 14, Transport and Access, of the ES [APP-052], this is in accordance with the appropriate guidance which includes the Government's Planning Practice Guidance; Travel Plans, TAs and Transport Statements in Decision Taking (2014). The Applicant has consulted with the relevant Highways Authorities and National Highways regarding the assessment and mitigation. Comments from these stakeholders are presented in Chapter 14: Transport and Access of the ES [APP-052].
Paragraph 5.13.4	Where appropriate, the applicant should prepare a travel plan including demand management measures to mitigate transport impacts. The applicant should also provide details of proposed measures to improve access by public transport, walking and cycling, to reduce the need for parking associated with the proposal and to mitigate transport impacts.	A Construction Traffic Management Plan (CTMP) is included as Appendix 14.2 of the ES [REP4-038EN010132/EX7/WB6.3.14.2_E]. It outlines measures that will be included in the final CTMP to mitigate transport impact, manage demand, and improve and encourage construction staff to access the Order limits by public transport, cycling and reduce car transport to, and parking at, the Order Limits.



		As explained within the Outline Construction Environmental Management Plan [EN010132/EX6/WB7.1_DREP6-021], a Construction Worker Travel Plan will be implemented, to encourage construction workers to travel to the Site via sustainable travel.
Paragraph 5.13.6	A new energy NSIP may give rise to substantial impacts on the surrounding transport infrastructure and the IPC should therefore ensure that the applicant has sought to mitigate these impacts, including during the construction phase of the development. Where the proposed mitigation measures are insufficient to reduce the impact on the transport infrastructure to acceptable levels, the IPC should consider requirements to mitigate adverse impacts on transport networks arising from the development, as set out below. Applicants may also be willing to enter into planning obligations for funding infrastructure and otherwise mitigating adverse impacts.	Section 14.6 of Chapter 14: Transport and Access of the ES [APP-052] outlines the embedded design mitigation measures in relation to traffic and transport, including HGV deliveries and staff vehicles. Section 14.7 of Chapter 14: Transport and Access of the ES [APP-052] states that there are anticipated to be no significant adverse effects as a result of the construction, operation or decommissioning of the Scheme. Therefore, it is considered that the Scheme is compliant with this policy.
Paragraph 5.13.7	Provided that the applicant is willing to enter into planning obligations or requirements can be imposed to mitigate transport impacts identified in the NATA/WebTAG transport assessment, with attribution of costs calculated in accordance with the Department for Transport's guidance, then development consent should not be withheld, and appropriately limited weight should be applied to residual effects on the surrounding transport infrastructure.	Section 14.7 of Chapter 14: Transport and Access of the ES [APP-052] states that there are anticipated to be no significant adverse effects as a result of the construction, operation or decommissioning of the Scheme following the implementation of the embedded mitigation measures identified in Section 14.6 of Chapter 14: Transport and Access of the ES [APP-052]. It is therefore considered that the Scheme is compliant with this policy and development consent should not be withheld.



Paragraph 5.13.8	Where mitigation is needed, possible demand management measures must be considered and if feasible and operationally reasonable, required, before considering requirements for the provision of new inland transport infrastructure to deal with remaining transport impacts.	Provision of new transport infrastructure is not required, as Section 14.7 of Chapter 14: Transport and Access of the ES [APP-052] states that there are anticipated to be no significant adverse effects as a result of the construction, operation or decommissioning of the Scheme following the implementation of the embedded mitigation measures identified in Section 14.6 of Chapter 14: Transport and Access of the ES [APP-052].
Paragraph 5.13.9	The IPC should have regard to the cost-effectiveness of demand management measures compared to new transport infrastructure, as well as the aim to secure more sustainable patterns of transport development when considering mitigation measures.	Traffic generated by the Scheme during its operational phase will not be of a level that requires management. No new transport infrastructure is therefore proposed as part of the Scheme. During the construction and decommissioning periods, traffic impact will be managed in accordance with measures set out in the Outline CTMP provided in Appendix 14.2 of the ES [REP4-038EN010132/EX7/WB6.3.14.2 E], and the Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023].
Paragraph 5.13.10	Water-borne or rail transport is preferred over road transport at all stages of the project, where cost-effective.	Given the context of the Order limits and the requirements for construction deliveries, rail and water borne transports are not considered to be appropriate methods of transport. See Section 6.12 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD].
Paragraph 5.13.11	The IPC may attach requirements to a consent where there is likely to be substantial HGV traffic that:	Section 14.7 of Chapter 14: Transport and Access of the ES [APP-052] states that there are anticipated to be no significant adverse effects on the wider transport network



Paragraph 5.14.3	Disposal of waste should only be considered where other waste management options are not available or where it is the best overall environmental outcome.	As detailed in Section 20.8 of Chapter 20: Waste of the ES [APP-058], waste arisings will be prevented and designed
		It is therefore considered that the Scheme is compliant with this policy.
Paragraph 5.14.2	Sustainable waste management is implemented through the "waste hierarchy", which sets out the priorities that must be applied when managing waste: a) prevention; b) preparing for reuse; c) recycling; d) other recovery, including energy recovery; and e) disposal.	As detailed in Section 20.8 of Chapter 20: Waste of the ES [APP-058], waste arisings will be prevented and designed out where possible. Opportunities to re-use material resources will be sought where practicable. Where re-use and prevention are not possible, waste arisings will be managed in line with the Waste Hierarchy and detailed Construction Resource Management Plan (CRMP). Waste will also be biased towards Nottinghamshire as, compared to Lincolnshire, since it has a greater capacity to take on landfill.
	 control numbers of HGV movements to and from the site in a specified period during its construction and possibly on the routing of such movements; make sufficient provision for HGV parking, either on the site or at dedicated facilities elsewhere, to avoid 'overspill' parking on public roads, prolonged queuing on approach roads and uncontrolled onstreet HGV parking in normal operating conditions; and ensure satisfactory arrangements for reasonably foreseeable abnormal disruption, in consultation with network providers and the responsible police force. 	as a result of the construction, operation or decommissioning of the Scheme following the implementation of the mitigation measures identified in Section 14.6 of Chapter 14: Transport and Access of the ES [APP-052]. The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] sets out controls that will be applied to manage the impacts of construction of the Scheme. The Outline Plan will be secured by Requirement in the Draft Development Consent Order [EN010132/EX6EX7/WB3.1_GH] Therefore, it is considered that there is not likely to be a need to attach additional requirements to the DCO.



		out where possible. Opportunities to re-use material resources will be sought where practicable. Where re-use and prevention are not possible, waste arisings will be managed in line with the Waste Hierarchy and detailed Construction Resource Management Plan (CRMP). It is therefore considered that the Scheme is compliant with this policy.
Paragraph 5.14.4	All large infrastructure projects are likely to generate hazardous and non-hazardous waste. The EA's Environmental Permitting (EP) regime incorporates operational waste management requirements for certain activities. When an applicant applies to the EA for an Environmental Permit, the EA will require the application to demonstrate that processes are in place to meet all relevant EP requirements.	Potential sources of waste associated with the Scheme are set out by Section 20.7 of Chapter 20: Waste of the ES [APP-058]. The Consents and Agreements Position Statement [REP4-046] sets out information on the additional consents and licences that are or may be required to construct and operate the Scheme.
Paragraph 5.14.6	The applicant should set out the arrangements that are proposed for managing any waste produced and prepare a Site Waste Management Plan. The arrangements described and Management Plan should include information on the proposed waste recovery and disposal system for all waste generated by the development, and an assessment of the impact of the waste arising from development on the capacity of waste management facilities to deal with other waste arising in the area for at least five years of operation. The applicant should seek to minimise the volume of waste produced and the volume of waste sent for disposal unless it can be demonstrated that this is the best overall environmental outcome.	As detailed in Section 20.5 of Chapter 20: Waste of the ES [APP-058], it is proposed that a CRMP will be prepared to ensure recycling and reuse of materials is maximised. The CRMP will be finalised with specific measures to be implemented prior to the start of construction. It is not anticipated that there would be a significant effect on waste during the construction operation or decommissioning of the Scheme. The Scheme is therefore considered to be compliant.



Paragraph 5.14.7	The IPC should consider the extent to which the applicant has proposed an effective system for managing hazardous and non-hazardous waste arising from the construction, operation and decommissioning of the proposed development. It should be satisfied that: • any such waste will be properly managed, both on-site and off-site; • the waste from the proposed facility can be dealt with appropriately by the waste infrastructure which is, or is likely to be, available. Such waste arisings should not have an adverse effect on the capacity of existing waste management facilities to deal with other waste arisings in the area; and • adequate steps have been taken to minimise the volume of waste arisings, and of the volume of waste arisings sent to disposal, except where that is the best overall environmental outcome.	During the construction, operation and decommissioning of the Scheme, the re-use or recycling of materials will be explored before resorting to landfill options. As detailed in Section 20.6 of Chapter 20: Waste of the ES [APP-058], waste arisings will be prevented and designed out where possible. Opportunities to re-use material resources will be sought where practicable. Where re-use and prevention are not possible, waste arisings will be managed in line with the Waste Hierarchy and detailed CRMP. Section 20.7 of Chapter 20: Waste in the ES [APP-058] also sets out potential sources of waste arising from the Scheme and states that any toxic and/or hazardous waste must be treated by an authorised operator. Transportation of hazardous waste will also require an authorised carrier. Materials are to be dealt with in accordance with the CEMP and Construction Resource Management Plan (CRMP) which will be secured through a DCO Requirement. With these in place and the appropriate control measures followed, no significant effects are anticipated. It is not anticipated that there would be no significant effect on waste from the Scheme and the Scheme is therefore considered to be compliant.
Paragraph 5.15.2	Where the project is likely to have effects on the water environment, the applicant should undertake an assessment of the existing status of, and impacts of the proposed project on, water quality, water resources and physical characteristics of the water environment as part of the ES or equivalent. (See Section 4.2.)	Chapter 10, Hydrology, Flood Risk and Drainage of the ES [APP-048] presents the existing status of the water environment and the likely effects of the Scheme upon it. This concludes that with appropriate embedded mitigation



		measures, there are likely to be no significant adverse effects on water quality, water resources or physical characteristics of the water environment as a result of the Scheme.
Paragraph 5.15.3	 The ES should in particular describe: the existing quality of waters affected by the proposed project and the impacts of the proposed project on water quality, noting any relevant existing discharges, proposed new discharges and proposed changes to discharges; existing water resources affected by the proposed project and the impacts of the proposed project on water resources, noting any relevant existing abstraction rates, proposed new abstraction rates and proposed changes to abstraction rates (including any impact on or use of mains supplies and reference to Catchment Abstraction Management Strategies); existing physical characteristics of the water environment (including quantity and dynamics of flow) affected by the proposed project and any impact of physical modifications to these characteristics; and any impacts of the proposed project on water bodies or protected areas under the Water Framework Directive and source protection zones (SPZs) around potable groundwater abstractions. 	Section 10.5 of Chapter 10: Hydrology, Flood Risk and Drainage of the ES [APP-048] provides an assessment of the baseline that complies with this policy. The ES Chapter [APP-048] includes a Water Framework Directive (WFD) Assessment [REP1-040], which assesses impacts on water bodies or protected areas under the WFD and SPZs. It is therefore considered that the Scheme is compliant with this policy.
Paragraph 5.15.5	The IPC will generally need to give impacts on the water environment more weight where a project would have an adverse effect on the achievement of the environmental objectives established under the Water Framework Directive.	The Hydrology ES Chapter [APP-048] provides a WFD Assessment. This concludes that the Scheme is compliant with the objectives of the WFD: it would not cause deterioration in status of the water bodies and would not prevent the water bodies achieving Good Ecological Status.



		The Scheme also contributes to the delivery of WFD objectives.
Paragraph 5.15.6	The IPC should satisfy itself that a proposal has regard to the River Basin Management Plans and meets the requirements of the Water Framework Directive (including Article 4.7) and its daughter directives, including those on priority substances and groundwater. The specific objectives for particular river basins are set out in River Basin Management Plans. The IPC should also consider the interactions of the proposed project with other plans such as Water Resources Management Plans and Shoreline/Estuary management Plans.	Chapter 10, Hydrology, Flood Risk and Drainage of the ES [APP-048] takes into account the Anglian and Humber River Basin Management Plan (RBMP) areas. The Scheme is therefore compliant with this policy.
Paragraph 5.15.8	The IPC should consider whether mitigation measures are needed over and above any which may form part of the project application. (See Sections 4.2 and 5.1.) A construction management plan may help codify mitigation at that stage.	Mitigation measures during the construction of the Scheme will be according to Best Practical Means that are included within the Outline CEMP [EN010132/EX6/WB7.1_D].REP6-021]. Therefore, it is considered that the Scheme is compliant with this policy.



1.2 Table 2: National Policy Statement EN-3 (2011)

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
Paragraph 1.1.1	Electricity generation from renewable sources of energy is an important element in the Government's development of a low carbon economy. There are ambitious renewable energy targets in place and a significant increase in generation from large-scale renewable energy infrastructure is necessary to meet the 15% renewable energy target (see Section 3.4 of EN-1).	As explained in the Statement of Need [APP-320], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to decarbonise with solar technology supported by recent government policy. Its proposed National Electricity Transmission System (NETS) connection means that it would play its part in helping National Grid ESO (NGESO) manage the national electricity system to ensure security of supply and bring cost benefits to electricity consumers, both of which are identified in government policy as being required for resilient energy supplies in the future. The meaningful and timely contributions offered by the Scheme to UK decarbonisation and security of supply, while helping lower bills for consumers throughout its operational life, will be critical on the path to Net Zero. Without the Scheme, a significant and vital opportunity to develop a large-scale low-carbon generation scheme will have been passed over, increasing materially the risk that future Carbon Budgets and Net Zero 2050 will not be achieved.



Paragraph 2.4.2 Proposals for renewable energy infrastructure should demonstrate good design in respect of landscape and visual amenity, and in the design of the project to mitigate impacts such as noise and effects on ecology.

As detailed in Section 6.3 of the Planning Statement [EN010132/EX6EX7/WB7.5 CD], the Scheme has been subject to a detailed and sensitive iterative design process. This has taken account of the context and features of the land within the Order limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts, and provision of environmental and other enhancements, where practicable. The design process and basis of design decisions taken are described in the Chapter 5: Alternatives and Design Evolution of the ES [APP-043] and the Design and Access Statement [APP-314 to APP-315].



1.3 Table 3: National Policy Statement EN-5 (2011)

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
Paragraph 2.2.4	Where the network company does not own (or wish to own) the relevant land itself, it may reach a voluntary agreement that gives it either an easement over the land or at least a wayleave permission to use it during the tenure of the current owner or occupier. Where it does not succeed in reaching the agreement it wants, the company may, as part of its application to the IPC, seek to acquire rights compulsorily over the relevant land by means of a provision in the DCO. The applicant may also apply for the compulsory purchase of land: this is not normally sought where lines and cables are installed, but may occur where other electricity network infrastructure, such as a new substation, is required. The above issues may be relevant considerations when the electricity company is considering various potential routes.	The Scheme, by way of a DCO application, establishes how new substations will be constructed, located and designed. The DCO Works Schedule [EN010132/EX6EX7/WB3.1_GH] captures the creation of new on-Site substations (Work No.3).
Paragraph 2.2.5	There will usually be some flexibility around the location of the associated substations and applicants will give consideration to how they are placed in the local landscape taking account of such things as local topography and the possibility of screening. See Section 2.8 below and Section 5.9 in EN-1.	As detailed in Section 6.3 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme has been subject to a detailed and sensitive iterative design process. This has taken account of the context and features of the land within the Order limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts, and provision of environmental and other enhancements, where practicable. The design process and basis of design decisions taken are described in the



		Chapter 5: Alternatives and Design Evolution of the ES [APP-043] and the Design and Access Statement [APP-314 to APP-315].
Paragraph 2.3.4	If the IPC believes it needs to probe further then factors it may wish to consider include whether the project would make a significant contribution to the promotion of renewable energy, the achievement of climate change objectives, the maintenance of an appropriate level of security of electricity supply or whether it helps achieve other energy policy objectives.	As explained in the Statement of Need [APP-320], the meaningful and timely contributions offered by the Scheme to UK decarbonisation and security of supply, while helping lower bills for consumers throughout its operational life, will be critical on the path to Net Zero. Without the Scheme, a significant and vital opportunity to develop a large-scale low-carbon generation scheme will have been passed over, increasing materially the risk that future Carbon Budgets and Net Zero 2050 will not be achieved.
		The Applicant, as a private sector organisation, has developed proposals for the Scheme, which will be a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to decarbonise, with solar technology supported by recent government policy. Its proposed National Electricity Transmission System (NETS) connection means that it would play its part in helping National Grid ESO (NGESO) manage the national electricity system to ensure security of supply and bring cost benefits to electricity consumers, both of which are identified in government policy as being required for resilient energy supplies in the future.



		The Scheme will quickly deliver significant amounts of low carbon power. Solar is also relatively quick to construct compared to other technologies which have longer construction timeframes or have potentially not yet been proven at scale.
		The Scheme therefore directly responds to the Government's objective of delivering a major and rapid change to the energy system through the delivery of infrastructure by private sector developers in the market system.
Paragraph 2.3.5	The IPC should also take into account that National Grid, as the owner of the electricity transmission system in England and Wales, as well as Distribution Network Operators (DNOs), are required under section 9 of the Electricity Act 1989 to bring forward efficient and economical proposals in terms of network design, taking into account current and reasonably anticipated future generation demand. National Grid is also required to facilitate competition in the supply and generation of electricity and so has a statutory duty to provide a connection whenever or wherever one is required.	The Applicant has secured a connection to the National Grid via a new below ground Cable Route Corridor located within the Grid Connection Route. This will connect the West Burton 3 substation to an existing 400kV substation connection bay at West Burton Power Station. Further details of this are included in the Grid Connection Statement [APP-316].
		As outlined in Section 7.8 of Chapter 7: Climate Change of the ES [REP1-012], account of the effects of climate change have been taken in the design of the Scheme, and its construction and decommissioning. This includes:
		- The effect of projected temperature increases on electrical equipment over the course of the Scheme's design life has been taken into account. Inverters (PV and BESS) will have a cooling system installed to control the temperature and allow the inverters to operate efficiently in warmer conditions.



The PV modules and transformers have a wide range of acceptable operating temperatures, and it has been determined that increasing temperatures will not adversely affect their operation.

- Any health and safety plans developed for construction and decommissioning activities will be required to account for potential climate change impacts on workers, such as flooding and heatwayes.
- The design of drainage systems will ensure that there will be no significant increases in flood risk downstream during storms up to and including the 1 in 100 (1%) annual probability design flood, with an allowance of 20% for climate change. A Decommissioning Environmental Management Plan (DEMP) (taking account of climate change risks at the time) will be prepared prior to decommissioning. An Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] is provided as part of the Application.

Flood Risk Screening is provided within Appendices 10.1 to 10.6 of the ES [APP-089 to APP-094]. The Flood Risk Screening provides a detailed assessment of the risk of flooding to the Scheme, taking account of climate change, and concludes that the Scheme is resilient to flood risk.

It is therefore considered that the Scheme is compliant with this policy.



Paragraph 2.4.2	Section 4.8 of EN-1 advises that the resilience of the project to climate change should be assessed in the Environmental Statement (ES) accompanying an application. For example, future increased risk of flooding would be covered in any flood risk assessment (see Section 5.7 in EN-1).	Chapter 7: Climate Change of the ES [REP1-012] assesses the resilience of the Scheme to climate change, including increased risk of flooding, as required by this policy.
Paragraph 2.8.9	The impacts and costs of both overhead and underground options vary considerably between individual projects (both in absolute and relative terms). Therefore, each project should be assessed individually on the basis of its specific circumstances and taking account of the fact that Government has not laid down any general rule about when an overhead line should be considered unacceptable. The IPC should, however only refuse consent for overhead line proposals in favour of an underground or subsea line if it is satisfied that the benefits from the non-overhead line alternative will clearly outweigh any extra economic, social and environmental impacts and the technical difficulties are surmountable. In this context it should consider: • the landscape in which the proposed line will be set, (in particular, the impact on residential areas, and those of natural beauty or historic importance such as National Parks, AONBs and the Broads)14; • the additional cost of any undergrounding or sub-sea cabling (which experience shows is generally significantly more expensive than overhead lines, but varies considerably from project to project depending on a range of factors, including whether the line is buried directly in open agricultural land or whether more complex tunnelling and civil engineering through conurbations and major cities is required15. Repair impacts are also significantly higher than for overhead lines as are the costs associated with any later uprating.); and • the environmental and archaeological consequences (undergrounding a 400kV line may mean disturbing a swathe of ground up to 40 metres across16, which can disturb sensitive habitats, have an impact on soils and geology, and damage heritage assets, in many cases more than an overhead line would).	The Scheme as detailed within ES Chapter 4: The Development Proposal [APP-042] states that all cables between Sites and the Grid Connection Point will be undergrounded. The potential implications upon archaeological remains have been considered. Both non-intrusive surveys and desk-based assessments have been conducted to provide a baseline. In order to establish a grounded truth, 'blank' areas have undergone trial trenching as well as areas which are indicative of high archaeological interest. The assessments of archaeological value can be found appended to ES Chapter 13 [APP-051].



Paragraph 2.9.7

Audible noise effects can also arise from substation equipment such as transformers, quadrature boosters and mechanically switched capacitors. Transformers are installed at many substations, and generate low frequency hum. Whether the noise can be heard outside a substation depends on a number of factors, including transformer type and the level of noise attenuation present (either engineered intentionally or provided by other structures).

Chapter 15: Noise & Vibration of the ES [APP-053] has assessed the impacts of all aspects of the Scheme including substations in accordance with this policy.

Resultingly, the distance from the nearest residential receptors to the substation and energy storage facility and onsite transformers and inverters has been maximised.



1.4 Table 4: National Policy Statement EN-1 (November 2023)

1.4.1 Table 4 considers the Scheme in the context of policy in NPS EN-1 (November 2023) where that policy differs from policy set out in NPS EN-1 (2011). Where the policy set out by NPS EN-1 (November 2023) is to the same or similar effect as policy in NPS EN-1 (2011), it is not included in this table.

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
Paragraph 2.1.3	To produce enough energy required for the UK and ensure it can be transported to where it is needed, a significant amount of infrastructure is needed at both local and national scale. High quality infrastructure is crucial for economic growth, boosting productivity and competitiveness. Part 3 of this NPS provides further details on the need for and importance of energy to economic prosperity and social well-being.	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable and low carbon electricity generation system which is sufficient to meet future demand. This will help to meet the need for new energy structure that is crucial for economic growth, boosting productivity and competitiveness, as identified by this policy.
Paragraph 2.2.1	In June 2019, the UK became the first major economy to legislate for a 2050 net zero Greenhouse Gases ('GHG') emissions target through the Climate Change Act 2008 (2050 Target Amendment) Order 2019. In December 2020, the UK communicated its Nationally Determined Contributions to reduce GHG emissions by at least 68 per cent from 1990 levels by 2030. In April 2021, the Government legislated for the sixth carbon budget (CB6), which requires the UK to reduce GHG emissions by 78% by 2035 compared to 1990 levels.	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon reduction and climate change, including the legally binding



		emissions reduction target for 2050 and carbon budgets described by this policy.
		Chapter 7 Climate change of the ES [REP1-012] presents a lifecycle greenhouse gas (GHG) impact assessment which considers the impact of GHG emissions arising over the lifetime of the Scheme on the climate. This concludes that over its 40-year operational lifetime the Scheme will produce 21,956,988 MWh of electricity with an average operational greenhouse gas intensity of 7.72 grams of carbon dioxide equivalent per kWh (gCO2e/kWh). This demonstrates its very low carbon attributes compared to other non-renewable forms of electricity generation, providing an overall major beneficial impact in relation to the UK meeting its carbon reduction targets and therefore represents a major beneficial effect on the climate.
Paragraph 2.3.3	Our objectives for the energy system are to ensure our supply of energy always remains secure, reliable, affordable, and consistent with meeting our target to cut GHG emissions to net zero by 2050, including through delivery of our carbon budgets and Nationally Determined Contribution. This will require a step change in the decarbonisation of our energy system.	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon reduction and climate change, including the legally binding emissions reduction target for 2050 and carbon budgets described by this policy.
		The Scheme incorporates a Battery Energy Storage System (BESS) which is an important aspect of regularising the supply



		of electricity from renewable energy production such as solar PV.
Paragraph 2.3.4	Meeting these objectives necessitates a significant amount of energy infrastructure, both large nationally significant developments and small-scale developments determined at a local level. This includes the infrastructure needed to convert primary sources of energy (e.g. wind) into energy carriers (e.g. electricity or hydrogen), and to store and transport primary fuels and energy carriers into and around the country. It also includes the infrastructure needed to capture, transport and store carbon dioxide. The requirement for new energy infrastructure will present opportunities for the UK and contributes towards our ambition to support jobs in the UK's clean energy industry and local supply chains.	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon reduction and climate change, including the legally binding emissions reduction target for 2050 and carbon budgets described by this policy.
Paragraph 2.3.5	The sources of energy we use will also need to change. Since the industrial revolution, our energy system is dominated by fossil fuels. That remains the case today. Although representing a record low, fossil fuels still accounted for just over 76 per cent of energy supply in 2020. We need to dramatically increase the volume of energy supplied from low carbon sources.	As explained in the Statement of Need [APP-320] the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity, in direct accordance with this policy, to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon reduction and climate change.
Paragraph 2.3.7	Decarbonisation means we are likely to become more dependent on some forms of energy compared to others. Using electrification to reduce emissions in large parts of transport, heating and industry could lead to more than half of final energy demand being met by electricity in 2050, up from 17 per cent in 2019, representing a doubling in demand for electricity. Low carbon hydrogen is also likely to play an increasingly significant role.	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon



		reduction and climate change, including the legally binding emissions reduction target for 2050 and carbon budgets described by this policy. The Statement of Need [APP-320] also explains that solar generation is expected to be an important part of the future energy mix.
Paragraph 2.5.1	Given the vital role of energy to economic prosperity and social well-being, it is important that our supplies of energy remains secure, reliable and affordable.	As explained in the Statement of Need [APP-320] the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity, in direct accordance with this policy, to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon reduction and climate change.
Paragraph 3.1.1 3.1.2	This Part of the NPS explains why the government sees a need for significant amounts of new large-scale energy infrastructure to meet its energy objectives and why the government considers that the need for such infrastructure is urgent. However, it will not be possible to develop the necessary amounts of such infrastructure without some significant residual adverse impacts. These effects will be minimised by the application of policy set out in Parts 4 and 5 of this NPS. See also Part 2 of each technology specific NPS.	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon reduction and climate change.
		An EIA has been undertaken to assess the environmental impacts of the Scheme and an ES [APP-039 to APP-061] prepared to report the findings. Overall, with appropriate mitigation implemented, this identifies the residual significant adverse effects of the Scheme. When considered relative to the large-scale nature of the Scheme these effects are



		considered to be outweighed by the significant national benefits that the Scheme will provide by providing much needed large scale renewable energy generation. Section 6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] sets out detailed consideration the Scheme's compliance with policy, taking account of the significant effects identified in the ES [APP-039 to APP-061], and Section 7 of the Planning Statement considers the planning balance taking account of its benefits and effects.
Paragraph 3.2.1	The government's objectives for the energy system are to ensure our supply of energy always remains secure, reliable, affordable, and consistent with net zero emissions in 2050 for a wide range of future scenarios, including through delivery of our carbon budgets and Nationally Determined Contributions.	As explained in the Statement of Need [APP-320] the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity, in direct accordance with this policy, to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon reduction and climate change.
Paragraph 3.2.3	It is not the role of the planning system to deliver specific amounts or time limit any form of infrastructure covered by this NPS. It is for industry to propose new energy infrastructure projects that they assess to be viable within the strategic framework set by government. This is the nature of a market-based energy system. With the exception of new coal or large-scale oil-fired electricity generation, the government does not consider it appropriate for planning policy to set limits on different technologies but planning policy can be used to support the government's ambitions in energy policy and other policy areas.	As per this policy, the Applicant proposes the Scheme by way of DCO application.
Paragraph 3.2.6	The Secretary of State should assess all applications for development consent for the types of infrastructure covered by the energy NPS on the basis that the government	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial



	has demonstrated that there is a need for those types of infrastructure which is urgent, as described for each of them in this Part.	infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon reduction and climate change.
		The Applicant notes that, in accordance with this policy, the need infrastructure such the Scheme is acknowledged, and the application should be considered on the basis that the need has been demonstrated.
Paragraph 3.2.7	In addition, the Secretary of State has determined that substantial weight should be given to this need when considering applications for development consent under the Planning Act 2008.	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon reduction and climate change.
		As per this policy, the established urgent need for the Scheme should be given substantial weight in the decision.
		The Applicant notes that, in accordance with this policy, the need for infrastructure such as the Scheme is urgent and considers that the SoS should give substantial weight to this in their decision.



Paragraph 3.3.3	To ensure that there is sufficient electricity to meet demand, new electricity infrastructure will have to be built to replace output from retiring plants and to ensure we can meet increased demand. Our analysis suggests that even with major improvements in overall energy efficiency, and increased flexibility in the energy system, demand for electricity is likely to increase significantly over the coming years and could more than double by 2050 as large parts of transport, heating and industry decarbonise by switching from fossil fuels to low carbon electricity. The Impact Assessment for CB6 shows an illustrative range of 465-515TWh in 2035 and 610-800TWh in 2050.	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] the Scheme will help meet the demand for energy which is expected to rise substantially in the future.
Paragraph 3.3.8	The government has considered alternatives to the need for new large-scale electricity infrastructure and concluded that these would be limited to reducing total demand for electricity through efficiency measures or through greater use of low carbon hydrogen in decarbonising the economy; reducing maximum demand through demand side response; and, increasing the contribution of decentralised and smaller-scale electricity infrastructure. In addition, there are alternative ways of decarbonising heating and transportation, which are being developed alongside electrification of these sectors.	The Statement of Need [APP-320] provides detailed information on why large scale solar is needed alongside other forms of generation. As per paragraph 3.2.6, the Scheme should be considered on the basis that its need is established and this established and urgent need should be given substantial weight in the decision.
Paragraph 3.3.10	The precise level of electricity demand during the transition to net zero is uncertain and could be affected by alternative means of decarbonising these sectors, such as the use of low carbon hydrogen, and the pace of that decarbonisation. However, it is prudent to plan on a conservative basis to ensure that there is sufficient supply of electricity to meet demand across a wide range of future scenarios, including where the use of hydrogen is limited.	The Statement of Need [APP-320] provides detailed information on future energy demand and how this is identified. As per paragraph 3.2.6, the Scheme should be considered on the basis that its need is established and this established and urgent need should be given substantial weight in the decision.
Paragraph 3.3.12	Decentralised and community energy systems such as micro-generation contribute to our targets on reducing carbon emissions and increasing energy security. These technologies could also lead to some reduction in demand on the main generation and transmission system. However, the government does not believe they will	As also explained in the Statement of Need [APP-320], this policy acknowledges that large scale electricity generation



	replace the need for new large-scale electricity infrastructure to meet our energy objectives. This is because connection of large-scale, centralised electricity generating facilities via a high voltage transmission system enables the pooling of both generation and demand, which in turn offers a number of economic and other benefits, such as more efficient bulk transfer of power and enabling surplus generation capacity in one area to be used to cover shortfalls elsewhere.	facilities are needed and are complementary to decentralised and community energy systems. The Scheme would connect directly to the NETS, to enable the transfer of the electricity it generates over a wide geographical area, as per this policy.
		As per paragraph 3.2.6, the Scheme should be considered on the basis that its need is established and this established and urgent need should be given substantial weight in the decision.
Paragraph 3.3.20	Wind and solar are the lowest cost ways of generating electricity, helping reduce costs and providing a clean and secure source of electricity supply (as they are not reliant on fuel for generation). Our analysis shows that a secure, reliable, affordable, net zero consistent system in 2050 is likely to be composed predominantly of wind and solar.	As explained in the Statement of Need [APP-320], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable and low carbon electricity generation system which is sufficient to meet future demand.
		This policy sets out that the government expects solar technology to play a major role in delivery of these objectives. The Scheme is in direct accordance with this expectation.
Paragraph 3.3.25	Storage has a key role to play in achieving net zero and providing flexibility to the energy system, so that high volumes of low carbon power, heat and transport can be integrated.	This paragraph explains the need for energy storage systems to compliment generation. In accordance with this need, the Scheme includes a Battery Energy Storage System (BESS) to control the release of energy to the NETS, enabling it to be released when it is most needed.
Paragraph 3.3.26 3.3.27	Storage is needed to reduce the costs of the electricity system and increase reliability by storing surplus electricity in times of low demand to provide electricity when demand is higher. There is currently around 4GW of electricity storage operational in	This paragraph explains the need for energy storage systems to compliment generation. In accordance with this need, the Scheme includes a Battery Energy Storage System (BESS) to



	GB, around 3GW of which is pumped hydro storage and around 1GW is battery storage.	control the release of energy to the NETS, enabling it to be released when it is most needed.
	Storage can provide various services, locally and at the national level. These include maximising the usable output from intermittent low carbon generation (e.g. solar and wind), reducing the total amount of generation capacity needed on the system; providing a range of balancing services to the NETSO and Distribution Network Operators (DNOs) to help operate the system; and reducing constraints on the networks, helping to defer or avoid the need for costly network upgrades as demand increases.	
3.3.58	Given the urgent need for new electricity infrastructure and the time it takes for electricity NSIPs to move from design conception to operation, there is an urgent need for new (and particularly low carbon) electricity NSIPs to be brought forward as soon as possible, given the crucial role of electricity as the UK decarbonises its economy.	As explained in the Statement of Need [APP-350], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon reduction and climate change.
		As per paragraph 3.2.6, the Scheme should be considered on the basis that its need is established and this established and urgent need should be given substantial weight in the decision.
Paragraph 3.3.59	 All the generating technologies mentioned above are urgently needed to meet the government's energy objectives by: providing security of supply (by reducing reliance on imported oil and gas, avoiding concentration risk and not relying on one fuel or generation type) providing an affordable, reliable system (through the deployment of technologies with complementary characteristics) 	As explained in the Statement of Need [APP-320], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon reduction and climate change.



	 ensuring the system is net zero consistent (by remaining in line with our carbon budgets and maintaining the options required to deliver for a wide range of demand, decarbonisation and technology scenarios, including where there are difficulties with delivering any technology) 	As per paragraph 3.2.6, the Scheme should be considered on the basis that its need is established and this established and urgent need should be given substantial weight in the decision.
Paragraph 3.3.60 3.3.61	Known generation technologies that are included within the scope of this NPS (and would be classed as an NSIP if above the relevant capacity thresholds set out under the Planning Act 2008) include: Offshore Wind (including floating wind) Solar PV Wave Tidal Range Tidal Stream Pumped Hydro Energy from Waste (including ACTs) with or without CCS Biomass with or without CCS Natural Gas with or without CCS Low carbon hydrogen Large-scale nuclear, Small Modular Reactors, Advanced Modular Reactors, and fusion power plants Geothermal The need for all these types of infrastructure is established by this NPS and a combination of many or all of them is urgently required for both energy security and Net Zero, as set out above.	This confirms that solar PV generation facilities, such as the Scheme, are covered by the emerging suite of draft Energy NPSs. As explained in the Statement of Need [APP-320], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon reduction and climate change. As per paragraph 3.2.6, the Scheme should be considered on the basis that its need is established and this established and urgent need should be given substantial weight in the decision.



Paragraph 3.3.83	Given the urgent need for new electricity infrastructure and the time it takes for electricity NSIPs to move from design conception to operation, there is an urgent need for new (and particularly low carbon) electricity NSIPs to be brought forward as soon as possible, given the crucial role of electricity as the UK decarbonises its economy.	As explained in the Statement of Need [APP-320], solar generation is a proven technology that can be delivered quickly in comparison to other forms of generation technology. The Scheme therefore has great potential to deliver a substantial amount of low-carbon electricity in a short timescale. This paragraph further emphasises that the substantial benefits of the Scheme in making a substantial contribution to meeting the UK's urgent energy needs.
Paragraph 3.2.4	It is not the government's intention in presenting any of the figures or targets in this NPS to propose limits on any new electricity infrastructure that can be consented in accordance with the energy NPSs. A large number of consented projects can help deliver an affordable electricity system, by driving competition and reducing costs within and amongst different technology and infrastructure types. Consenting new projects also enables projects utilising more advanced technology and greater efficiency to come forward.37 The delivery of an affordable energy system does not always mean picking the least cost technologies. A diversity of supply can aid in ensuring affordability for the system overall and relative costs can change over time, particularly for new and emerging technologies. It is not the role of the planning system to compare the costs of individual developments or technology types.	As explained in the Statement of Need [APP-320], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable electricity generation system which is sufficient to meet future demand and contribute to meeting the government's objectives in respect of carbon reduction and climate change. This paragraph further emphasises the scale of the urgent need and establishes that the fact there may be other similar schemes in the planning system is not a reason to limit the number of approvals and a large number of approved schemes is beneficial in terms of enabling the market to efficiently deliver the infrastructure that is needed.
Paragraph 4.1.5	 In considering any proposed development, in particular when weighing its adverse impacts against its benefits, the Secretary of State should take into account: its potential benefits including its contribution to meeting the need for energy infrastructure, job creation, reduction of geographical disparities, environmental enhancements, and any long-term or wider benefits; 	As considered within Section 6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the benefits and adverse effects have been assessed and duly weighted. Where adverse effects have emerged through the ES [APP-039 to APP-061] mitigation measures, both embedded and



	its potential adverse impacts, including on the environment, and including any long-term and cumulative adverse impacts, as well as any measures to avoid, reduce, mitigate or compensate for any adverse impacts, following the mitigation hierarchy	additional have been incorporated into the Scheme and its design.
Paragraph 4.3.1 4.3.2 4.3.3	All proposals for projects that are subject to the Infrastructure Planning (Environmental Impact Assessment) Regulations 2017 (the EIA Regulations) must be accompanied by an Environmental Statement (ES) describing the aspects of the environment likely to be significantly affected by the project. The Regulations specifically refer to effects on population, human health, biodiversity, land, soil, water, air, climate, the landscape, material assets and cultural heritage, and the interaction between them. The Regulations require an assessment of the likely significant effects of the proposed project on the environment, covering the direct effects and any indirect, secondary, cumulative, transboundary, short, medium, and long-term, permanent and temporary, positive and negative effects at all stages of the project, and also of	The DCO Application encompasses an ES which constitutes a total of 23 Chapter which assess likely significant effects, direct and indirect effects, secondary, cumulative, transboundary, short, medium, long-term, permanent and temporary, positive and negative effects at all stages of the project and also proposes mitigation measures where necessary to avoid significant adverse effects [APP-039 to APP-061].
Paragraph 4.3.5 4.3.7	the measures envisaged for avoiding or mitigating significant adverse effects. For the purposes of this NPS and the technology specific NPSs the ES should cover the environmental, social and economic effects arising from pre-construction, construction, operation and decommissioning of the project. In the absence of any additional information on additional assessments, the principles set out in this Section will apply to all assessments.	The ES [APP-039 to APP-061] has been produced in accordance with these paragraphs.
Paragraph 4.3.11	In some instances, it may not be possible at the time of the application for development consent for all aspects of the proposal to have been settled in precise detail. Where this is the case, the applicant should explain in its application which elements of the proposal have yet to be finalised, and the reasons why this is the case.	Where it has not been possible, the DCO has produced draft and outline documents such as the Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] and the Outline Construction Environmental Management Plan [EN010132/EX6/WB7.1_DREP6-021].



Paragraph 4.3.12	Where some details are still to be finalised, the ES should, to the best of the applicant's knowledge, assess the likely worst-case environmental, social and economic effects of the proposed development to ensure that the impacts of the	Where these documents have been produced in a draft form, documents point towards the production of succeeding detailed statements and management plans. Where details are yet to be finalised (such as the location of ancillary equipment) a worst-case analysis has been conducted to reflect the maximum design parameters across
	project as it may be constructed have been properly assessed.	the Site(s).
Paragraph 4.3.22 4.3.23 4.3.24 4.3.25 4.3.26 4.3.27 4.3.28 4.3.29	Given the level and urgency of need for new energy infrastructure, the Secretary of State should, subject to any relevant legal requirements (e.g. under the Habitats Regulations) which indicate otherwise, be guided by the following principles when deciding what weight should be given to alternatives: • the consideration of alternatives in order to comply with policy requirements should be carried out in a proportionate manner • only alternatives that can meet the objectives of the proposed development need be considered The Secretary of State should be guided in considering alternative proposals by whether there is a realistic prospect of the alternative delivering the same infrastructure capacity (including energy security, climate change, and other environmental benefits) in the same timescale as the proposed development. The Secretary of State should not refuse an application for development on one site simply because fewer adverse impacts would result from developing similar infrastructure on another suitable site, and it should have regard as appropriate to the possibility that all suitable sites for energy infrastructure of the type proposed may be needed for future proposals.	Section 6.3 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] sets out the consideration of the Scheme in the context of relevant policy that is applicable to alternatives. This sets out how the Scheme accords with policies and legislation where consideration of alternatives may be relevant. In doing so it notes the requirements of this policy, including that consideration of alternatives should be proportionate, take account of an alternative's ability to deliver the same infrastructure capacity as the Scheme, and that Development Consent should not be rejected on one site simply because fewer adverse impacts would result from developing similar infrastructure on another suitable site. Consideration of Alternatives and Design evolution is also addressed within ES Chapter 5: [APP-043] and the Site Selection Report [AS-004].



	Alternatives not among the main alternatives studied by the applicant (as reflected in the ES) should only be considered to the extent that the Secretary of State thinks they are both important and relevant to the decision.	
	As the Secretary of State must assess an application in accordance with the relevant NPS (subject to the exceptions set out in the Planning Act 2008), if the Secretary of State concludes that a decision to grant consent to a hypothetical alternative proposal would not be in accordance with the policies set out in the relevant NPS, the existence of that alternative is unlikely to be important and relevant to the Secretary of State's decision .	
	Alternative proposals which mean the necessary development could not proceed, for example because the alternative proposals are not commercially viable or alternative proposals for sites would not be physically suitable, can be excluded on the grounds that they are not important and relevant to the Secretary of State's decision.	
	Alternative proposals which are vague or inchoate can be excluded on the grounds that they are not important and relevant to the Secretary of State's decision.	
	It is intended that potential alternatives to a proposed development should, wherever possible, be identified before an application is made to the Secretary of State (so as to allow appropriate consultation and the development of a suitable evidence base in relation to any alternatives which are particularly relevant). Therefore, where an alternative is first put forward by a third party after an application has been made, the Secretary of State may place the onus on the person proposing the alternative to provide the evidence for its suitability as such and the Secretary of State should not necessarily expect the applicant to have assessed it.	
Paragraph 4.4.7	Generally, those aspects of energy infrastructure which are most likely to have a significantly detrimental impact on health are subject to separate regulation (for example for air pollution) which will constitute effective mitigation of them, so that it	This is addressed across numerous ES Chapters with a human health summary contained within Section 21.5 of Chapter 21: Other Environmental Matters [APP-059].



4.4.8	is unlikely that health concerns will either by themselves constitute a reason to refuse consent or require specific mitigation under the Planning Act 2008. However, not all potential sources of health impacts will be mitigated in this way and the Secretary of State may want to take account of health concerns when setting requirements relating to a range of impacts such as noise.	Primary mitigation measures are embedded within the Scheme, as set out in the respective chapters, to reduce other operational effects (such as noise, air quality and landscape) which in turn will mitigate the effects on the local community and existing facilities from a human health perspective. It is therefore considered that the Scheme is compliant with this policy.
Paragraph 4.6.1 4.6.2	Environmental net gain is an approach to development that aims to leave the natural environment in a measurably better state than beforehand. Projects should therefore not only avoid, mitigate and compensate harms, following mitigation hierarchy, but also consider whether there are opportunities for enhancements. Biodiversity net gain is an essential component of environmental net gain. Projects should consider and seek to incorporate improvements in natural capital, ecosystem services and the benefits they deliver when planning how to deliver biodiversity net gain.	A Biodiversity Net Gain (BNG) assessment, using Defra's Metric 3.0, has been provided within the DCO application [APP-088]. For the purposes of BNG, the Scheme will result in an overall significant net gain of 86.80% provided in habitat units, 54.71% gains in hedgerow units and 33.25% gains in river units. The Scheme has therefore taken advantage of opportunities to conserve and enhance biodiversity and accords with this policy. The Scheme has therefore incorporated improvements in biodiversity and accords with this policy. See also Section 6.9 of the Planning Statement [EN010132/EXGEXT/WB7.5_CD]. For further detail on the biodiversity measures incorporated and compliance with planning policy.
Paragraph 4.6.6 4.6.7	Energy NSIP Proposals, whether onshore or offshore should seek opportunities to contribute to and enhance the natural environment by providing net gains for biodiversity, or the wider environment where possible.	A Biodiversity Net Gain (BNG) assessment, using Defra's Metric 3.0, has been provided with the DCO application [APP-088].



4.6.10	In England applicants for onshore elements of any development are encouraged to use the most current version of the Defra biodiversity metric to calculate their biodiversity baseline and present planned biodiversity net gain outcomes. This calculation data should be presented in fullas part of their application.	For the purposes of BNG, the Scheme will result in an overall significant net gain of 86.80% provided in habitat units, 54.71% gains in hedgerow units and 33.25% gains in river units.
	Biodiversity net gain should be applied in conjunction with the mitigation hierarchy and does not change or replace existing environmental obligations, although compliance with those obligations will be relevant to the question of the baseline for assessing net gain and if they deliver an additional enhancement beyond meeting the existing obligation, that enhancement will count towards net gain.	The Scheme has therefore incorporated improvements in biodiversity and accords with this policy. See also Section 6.9 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD]. for further detail on the biodiversity measures incorporated and compliance with planning policy.
Paragraph 4.6.13	In addition to delivering biodiversity net gain, developments may also deliver wider environmental gains and benefits to communities relevant to the local area, and to national policy priorities, such as • reductions in GHG emissions, • reduced flood risk, • improvements to air or water quality, • climate adaption, • landscape enhancement • increased access to natural greenspace, or • the enhancement, expansion or provision of trees and woodlands The scope of potential gains will be dependent on the type, scale, and location of specific projects. Applicants should look for a holistic approach to delivering wider environmental net gains and benefits through the use of nature-based solutions and Green Infrastructure.	The Scheme will deliver a substantial reduction in greenhouse gas emissions over its lifetime, as explained by Chapter 7, Climate Change, of the ES [REP1-012]. The Scheme is estimated to produce low carbon energy at 7.72 gCO2e/kWh. The Scheme has taken other opportunities to provide enhancements, including by providing and connecting green infrastructure (as set out by the OLEMP) [EN010132/EX6/WB7.3_E].REP6-025].



4.6.15	Applications for development consent should be accompanied by a statement demonstrating how opportunities for delivering wider environmental net gains have been considered and where appropriate, incorporated into proposals as part of good design (including any relevant operational aspects) of the project.	
4.6.16	Applicants should make use of available guidance and tools for measuring natural capital assets and ecosystem services, such as the Natural Capitals Committee's 'How to Do it: natural capital workbook', the government guidance on Enabling a Natural Capital Approach (ENCA), and other tools that aim to enable wider benefits for people and nature.	
4.6.17	Where environmental net gain considerations have featured as part of the strategic options appraisal process to select a project, the statement should reference that information to supplement the site specific details.	
Paragraph 4.7.3 4.7.4	Good design is also a means by which many policy objectives in the NPS can be met, for example the impact sections show how good design, in terms of siting and use of appropriate technologies, can help mitigate adverse impacts such as noise. Projects should look to use modern methods of construction and sustainable design practices such as use of sustainable timber and low carbon concrete. Where possible, projects should include the reuse of material. Given the benefits of good design in mitigating the adverse impacts of a project, applicants should consider how good design can be applied to a project during the early stages of the project lifecycle.	As detailed in Section 6.3 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme has been subject to a detailed and sensitive iterative design process. This has taken account of the context and features of the land within the Order limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts, and provision of environmental and other enhancements, where practicable. The design process and basis of design decisions taken are described in the Chapter 5: Alternatives and Design Evolution of the ES [APP-043] and the Design and Access Statement [APP-314 to APP-315].



Paragraph 4.7.7	Applicants must demonstrate in their application documents how the design process was conducted and how the proposed design evolved. Where a number of different	The Design and Access Statement [APP-314 to APP-315] details the iterative design process and how the Scheme in its form has been developed.
	designs were considered, applicants should set out the reasons why the favoured choice has been selected.	The Site Selection Assessment [AS-004] details the stages and methodology used in the assessment for determining the site
4.7.12	In considering applications, the Secretary of State should take into account the ultimate purpose of the infrastructure and bear in mind the operational, safety and security requirements which the design has to satisfy. Many of the wider impacts of a development, such as landscape and environmental impacts, will be important factors in the design process.	selection process. The wider impacts of the Scheme have been captured, detailed, assessed, and mitigated for (where possible) through the production of the ES [APP-039 to APP-061].
4.7.13	The Secretary of State should consider such impacts under the relevant policies in this NPS. Assessment of impacts must be for the stated design life of the scheme rather than a shorter time period.	
Paragraph		Consideration has been given to incorporating nature-based
4.10.5	In certain circumstances, measures implemented to ensure a scheme can adapt to climate change may give rise to additional impacts, for example as a result of	climate change adaption into the Scheme, and proposals for SuDS have been included.
	protecting against flood risk, there may be consequential impacts on coastal change.	As outlined within the Operational Environmental
	In preparing measures to support climate change adaptation applicants should take reasonable steps to maximise the use of nature-based solutions alongside other conventional techniques.	Management Plan [REP5-020] and the Landscape Ecological Mitigation Plan [EN010132/EX6/WB7.3_EREP6-025], nature based solutions have been adopted as a form of mitigation/
4.10.7	In addition to avoiding further GHG emissions when compared with some more traditional adaptation approaches, nature-based solutions can also result in biodiversity benefits and net gain, as well as increasing absorption of carbon dioxide from the atmosphere.	enhancement to the Scheme in relation to potential impacts.
Paragraph		As outlined in Section 7.8 of Chapter 7: Climate Change of the
4.10.8	New energy infrastructure will typically need to remain operational over many decades, in the face of a changing climate. Consequently, applicants must consider	ES [REP1-012], account of the effects of climate change have



the direct (e.g. site flooding, limited water availability, storms, heatwave and wildfire threats to infrastructure and operations) and indirect (e.g. access roads or other critical dependencies impacted by flooding, storms, heatwaves or wildfires) impacts of climate change when planning the location, design, build, operation and, where appropriate, decommissioning of new energy infrastructure.

The ES should set out how the proposal will take account of the projected impacts of climate change, using government guidance and industry standard benchmarks such as the Climate Change Allowances for Flood Risk Assessments, Climate Impacts Tool, and British Standards for climate change adaptation, in accordance with the EIA Regulations.

been taken in the design of the Scheme, and its construction and decommissioning. This includes:

- The effect of projected temperature increases on electrical equipment over the course of the Scheme's design life has been taken into account. Inverters (PV and BESS) will have a cooling system installed to control the temperature and allow the inverters to operate efficiently in warmer conditions. The PV modules and transformers have a wide range of acceptable operating temperatures, and it has been determined that increasing temperatures will not adversely affect their operation.
- Any health and safety plans developed for construction and decommissioning activities will be required to account for potential climate change impacts on workers, such as flooding and heatwaves.
- The design of drainage systems will ensure that there will be no significant increases in flood risk downstream during storms up to and including the 1 in 100 (1%) annual probability design flood, with an allowance of 20% for climate change. A Decommissioning Environmental Management Plan (DEMP) (taking account of climate change risks at the time) will be prepared prior to decommissioning. An Outline Decommissioning Statement

 [EN010132/EX6/WB7.2_BREP6-023] is provided as part of the Application.

4.10.9



Paragraph		As outlined in Section 7.8 of Chapter 7: Climate Change of the
4.10.10	Applicants should assess the impacts on and from their proposed energy project across a range of climate change scenarios, in line with appropriate expert advice and guidance available at the time.	ES [REP1-012], account of the effects of climate change have been taken in the design of the Scheme, and its construction and decommissioning. This includes:
4.10.11	Applicants should demonstrate that proposals have a high level of climate resilience built-in from the outset and should also demonstrate how proposals can be adapted over their predicted lifetimes to remain resilient to a credible maximum climate change scenario. These results should be considered alongside relevant research which is based on the climate change projections.	- The effect of projected temperature increases on electrical equipment over the course of the Scheme's design life has been taken into account. Inverters (PV and BESS) will have a cooling system installed to control the temperature and allow the inverters to operate efficiently in warmer conditions. The PV modules and transformers have a wide range of acceptable operating temperatures, and it has been determined that increasing temperatures will not adversely affect their operation.
		- Any health and safety plans developed for construction and decommissioning activities will be required to account for potential climate change impacts on workers, such as flooding and heatwaves.
		- The design of drainage systems will ensure that there will be no significant increases in flood risk downstream during storms up to and including the 1 in 100 (1%) annual probability design flood, with an allowance of 20% for climate change. A Decommissioning Environmental Management Plan (DEMP) (taking account of climate change risks at the time) will be prepared prior to decommissioning. An Outline Decommissioning Statement



	Flood Risk Assessments (FRA) are provided at Appendices 10.1 – 10.6 of the ES [APP-048] . The FRA provides a detailed
	assessment of the risk of flooding to and from the Scheme (taking account of climate change) and concludes that the risk of flooding will not be increased as a result of the construction, operation or decommissioning of the Scheme. It is therefore considered that the Scheme is compliant with this policy.
	Chapter 7 Climate change of the ES [REP1-012] presents a lifecycle greenhouse gas (GHG) impact assessment which considers the impact of GHG emissions arising over the lifetime of the Scheme on the climate. This concludes that over its 40-year operational lifetime the Scheme will produce 21,956,988 MWh of electricity with an average operational greenhouse gas intensity of 7.72 grams of carbon dioxide equivalent per kWh (gCO2e/kWh). This demonstrates its very low carbon attributes compared to other non-renewable forms of electricity generation, providing an overall major beneficial impact in relation to the UK meeting its carbon reduction targets and therefore represents a major beneficial effect on the climate.
The connection of a proposed electricity generation plant to the electricity network is an important consideration for applicants wanting to construct or extend generation	As captured within the Grid Connection Statement [APP-316], the Applicant has submitted a grid application to the National Grid Electricity System Operator Limited (NGESO), the system operator of NETS. An offer was then received for which the



4.11.2	In the market system and in the past, it has been for the applicant to ensure that there will be necessary infrastructure and capacity within an existing or planned transmission or distribution network to accommodate the electricity generated.	Applicant accepted. The Scheme is considered compliant with this paragraph.
4.11.3	To support the achievement of the transition to net zero, government is accelerating the co-ordination of the development of the grid network to facilitate the UK's net zero energy generation development and transmission.	
Paragraph		As captured within the Grid Connection Statement [APP-316],
4.11.5	The applicant must liaise with National Grid who own and manage the transmission network in England and Wales or the relevant regional DNO or TSO to secure a grid connection.	the Applicant has submitted a grid application to the National Grid Electricity System Operator Limited (NGESO), the system operator of NETS. An offer was then received for which the Applicant accepted. The Scheme is considered compliant with
4.11.6	Applicants may wish to take a commercial risk where they have not received or accepted a formal offer of a grid connection from the relevant network operator at the time of the application.	these paragraphs.
4.11.0		
Paragraph 4.12.1	Issues relating to discharges or emissions from a proposed project, and which lead to other direct or indirect impacts on terrestrial, freshwater, marine, onshore and offshore environments, or which include noise and vibration may be subject to separate regulation under the pollution control framework or other consenting and licensing regimes, for example local planning consent or marine licences (see para 4.5.6 for more information).	Construction and Decommissioning Phases (as well as BESS failure) of the Scheme pose the greatest risk of potential discharges of emissions to freshwater sources. These potential effects have been recognised against separate regulation where applicable within the outline Construction Environmental Management Plan [EN010132/EX6/WB7.1_DREP6-021], Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] and the Outline Battery Storage Safety Management Plan [APP-318].
Paragraph		The Scheme is not located in or near to an education or
5.2.17		healthcare facility and air quality limits are not in danger of being exceeded.



5.2.18	The Secretary of State should give air quality considerations substantial weight where a project is proposed near a sensitive receptor site, such as an education or healthcare facility, residential use or a sensitive or protected habitat.	Mitigation measures, during Construction [REP3-018], Operation [REP5-020] and Decommissioning [EN010132/EX6/WB7.2_BREP6-023], have been incorporated
5.2.19	Where a project is proposed near to a sensitive receptor site of air quality, if the applicant cannot provide justification for this location, and a suitable mitigation plan, the Secretary of State should refuse consent.	to minimise dust emissions from the Scheme which may be impactful upon the air quality
	In all cases, the Secretary of State must take account of any relevant statutory air quality limits, objectives and targets. If a project will leave to noncompliance with a statutory limit, objective or target the Secretary of State should refuse consent.	
Paragraph 5.3.4	 All proposals for energy infrastructure projects should include a GHG assessment as part of their ES (See Section 4.3). This should include: A whole life GHG assessment showing construction, operational and decommissioning GHG impacts, including impacts from change of land use. An explanation of the steps that have been taken to drive down the climate change impacts at each of those stages Measurement of embodied GHG impact from the construction stage How reduction in energy demand and consumption during operation has been prioritised in comparison with other measures How operational emissions have been reduced as much as possible through the application of best available techniques for that type of technology Calculation of operational energy consumption and associated carbon emissions Whether and how any residual GHG emissions will be (voluntarily) offset or removed using a recognised framework 	Chapter 7, Climate change, of the ES [REP1-012] presents a lifecycle greenhouse gas (GHG) impact assessment which considers the impact of GHG emissions arising over the lifetime of the Scheme on the climate. The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and the Outline OEMP [REP5-020] set out measures to control and drive down carbon emissions during construction and operation of the Scheme.



	Where there are residual emissions, the level of emissions and the impact of those on national and international efforts to limit climate change, both alone and where relevant in combination with other developments at a regional or national level, or sector level, if sectoral targets are developed	
Paragraph 5.4.17	Where the development is subject to EIA the applicant should ensure that the ES clearly sets out any effects on internationally, nationally, and locally designated sites of ecological or geological conservation importance (including those outside England), on protected species and on habitats and other species identified as being of principal importance for the conservation of biodiversity, including irreplaceable habitats.	The ES has considered the effects on designated sites of ecological and geological conservation importance. This is captured within ES Chapter 9: Ecology and Biodiversity [APP-047] and ES Chapter 11: Ground Conditions and Contamination [APP-049].
5.4.18	The applicant should provide environmental information proportionate to the infrastructure where EIA is not required to help the Secretary of State consider thoroughly the potential effects of a proposed project.	
Paragraph		The Design and Access Statement [APP-314 to APP-315]
5.4.19	The applicant should show how the project has taken advantage of opportunities to conserve and enhance biodiversity and geological conservation interests.	explains how opportunities to protect and enhance biodiversity have been incorporated into the Scheme.
5.4.20	Applicants should consider wider ecosystem services and benefits on natural capital when designing enhancement measures.	A Biodiversity Net Gain (BNG) assessment, using Defra's Metric 3.0, has been provided with the DCO application [APP-
5.4.21	As set out in Section 4.7, the design process should embed opportunities for nature inclusive design. Energy infrastructure projects have the potential to deliver significant benefits and enhancements beyond Biodiversity Net Gain, which result in wider environmental gains (see Section 4.6 on Environmental and Biodiversity Net Gain). The scope of potential gains will be dependent on the type, scale, and location of each project.	088] . For the purposes of BNG, the Scheme will result in an overall significant net gain of 86.80% provided in habitat units, 54.71% gains in hedgerow units and 33.25% gains in river units.
		The Scheme has therefore taken advantage of opportunities to conserve and enhance biodiversity. Through incorporating



		improvements in biodiversity, the Scheme accords with this policy. See also Section 6.9 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] for further detail on the biodiversity measures incorporated and compliance with planning policy.
Paragraph		Chapter 9: Ecology and Biodiversity of the ES [APP-047] has
5.4.39	The government's 25 Year Environment Plan and the Environment Act 2021 mark a step change in ambition for wildlife and the natural environment. The Secretary of State should have regard to the aims and goals of the government's Environmental Improvement Plan 2023, and in Wales the objectives of the Nature Recovery Plan, and any relevant measures and targets, including statutory targets set under the Environment Act elsewhere.	been produced with regard to the aims and goals of the 25 Year Environment Plan, as evidenced by the extensive habitat to be provided pursuant to the Outline LEMP. It is therefore considered that the Scheme is compliant with this policy.
5.4.41	The benefits of nationally significant low carbon energy infrastructure development may include benefits for biodiversity and geological conservation interests and these benefits may outweigh harm to these interests. The Secretary of State may take account of any such net benefit in cases where it can be demonstrated.	
Paragraph		As outlined in Sections 9.6 and 9.7 of Chapter 9: Ecology and
5.4.42	As a general principle, and subject to the specific policies below, development should, in line with the mitigation hierarchy, aim to avoid significant harm to biodiversity and geological conservation interests, including through consideration of reasonable alternatives (as set out in Section 4.3 above). Where significant harm cannot be avoided, impacts should be mitigated as a last resort, appropriate	Biodiversity of the ES [APP-047], the residual effects post mitigation during operation results in local significant adverse impacts upon harvest mice (at Site Level) and adverse not-significant impacts upon Skylarks and Grey Partridges (at Local Level)
	compensation measures should be sought.	These will be mitigated as far as possible through the
5.4.43	If significant harm to biodiversity resulting from a development cannot be avoided (for example through locating on an alternative site with less harmful impacts), adequately mitigated, or, as a last resort, compensated for, then the Secretary of State will give significant weight to any residual harm.	provision and management of appropriate habitat to be secured through the LEMP.



		Embedded design mitigation measures are outlined in Section 9.6 of Chapter 9: Ecology and Biodiversity of the ES [APP-047] and are illustrated within the outline LEMP [EN010132/EX6/WB7.3_EREP6-025], Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline OEMP [REP5-020] and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023]. These include habitat avoidance, creation, and replacement measures; mitigation relating to protected and notable species; and standard mitigation measures that comply with industry good practice and environmental legislation. Production of a final LEMP, CEMP, OEMP and DEMP are secured by way of a requirement in the -Draft Development
		Consent Order Revision E-[EN010132/EX6H [EX7/WB3.1_GH] submitted at Deadline 37.
		It is therefore considered that the Scheme is compliant with this policy.
Paragraph		Section 9.6 of Chapter 9: Ecology and Biodiversity of the ES
5.4.4	The highest level of biodiversity protection is afforded to sites identified through international conventions. The Habitats Regulations set out sites for which an HRA will assess the implications of a plan or project, including Special Areas of Conservation and Special Protection Areas.	[APP-047] sets out that no such sites are present within the study area.
5.4.5	As a matter of policy, the following should be given the same protection as sites covered by the Habitat's Regulations and an HRA will also be required:	
	(a) potential Special Protection Areas and possible Special Areas of Conservation;	
	(b) listed or proposed Ramsar sites; and	



	(c) sites identified, or required, as compensatory measures for adverse effects on any of the other sites covered by this paragraph.	
Paragraph 5.4.8	Development on land within or outside a SSSI, and which is likely to have an adverse effect on it (either individually or in combination with other developments), should not normally be permitted. The only exception is where the benefits (including need) of the development in the location proposed clearly outweigh both its likely impact on the features of the site that make it of special scientific interest, and any broader impacts on the national network of SSSIs.	The assessment in Section 9.6 of Chapter 9: Ecology and Biodiversity of the ES [APP-047] considers the impacts of the Scheme on designated sites and concludes that there are no potential significant adverse effects as a result of the construction or operation of the Scheme on any SSSIs. The Scheme therefore accords with this policy.
Paragraph 5.4.12	Sites of regional and local biodiversity and geological interest, which include Regionally Important Geological Sites, Local Nature Reserves and Local Wildlife Sites, are areas of substantive nature conservation value and make an important contribution to ecological networks and nature's recovery. They can also provide wider benefits including public access (where agreed), climate mitigation and helping to tackle air pollution.	The assessment in Section 9.6 of Chapter 9: Ecology and Biodiversity of the ES [APP-047] considers the likely significant impacts of the Scheme on designated sites, and concludes that there are no potential significant adverse effects as a result of the construction or operation of the Scheme on any sites of regional and local biodiversity and geological interest. The Scheme therefore accords with this policy.
5.4.13	National planning policy expects plans to identify and map Local Wildlife sites, and to include policies that not only secure their protection from harm or loss but also help to enhance them and their connection to wider ecological networks.	
Paragraph 5.4.32	Applicants should include measures to mitigate fully the direct and indirect effects of development on ancient woodland, ancient and veteran trees or other irreplaceable habitats during both construction and operational phase.	The Scheme does not incorporate or propose the loss of any Ancient Woodlands or Veteran Trees.
Paragraph 5.4.35	Applicants should include appropriate avoidance, mitigation, compensation and enhancement measures as an integral part of the proposed development. In particular, the applicant should demonstrate that:	Embedded design mitigation measures of the kind set out in this policy are outlined in Section 9.6 of Chapter 9: Ecology and Biodiversity of the ES [APP-047] and are illustrated within the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025] and Outline CEMP [EN010132/EX6/WB7.1_DREP6-021]. These



	 during construction, they will seek to ensure that activities will be confined to the minimum areas required for the works the timing of construction has been planned to avoid or limit disturbance during construction and operation best practice will be followed to ensure that risk of disturbance or damage to species or habitats is minimised, including as a consequence of transport access arrangements habitats will, where practicable, be restored after construction works have finished opportunities will be taken to enhance existing habitats rather than replace them, and where practicable, create new habitats of value within the site landscaping proposals. Where habitat creation is required as mitigation, compensation, or enhancement the location and quality will be of key importance. In this regard habitat creation should be focused on areas where the most ecological and ecosystems benefits can be realised. mitigations required as a result of legal protection of habitats or species will be complied with. 	include habitat avoidance, creation, and replacement measures; mitigation relating to protected and notable species; and standard mitigation measures that comply with industry good practice and environmental legislation. Production of a final CEMP and LEMP are secured by way of a requirement in the draft DCO. The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] includes best practice measures to ensure that activities will be confined to the minimum areas required for the works during construction, in accordance with this part of the policy. Section 9.6 of Chapter 9: Ecology and Biodiversity of the ES [APP-047] outlines mitigation measures pertaining to habitat avoidance, creation and replacement measures that comply with this part of the policy.
Paragraph 5.4.36	Applicants should produce and implement a Biodiversity Management Strategy as part of their development proposals. This could include provision for biodiversity awareness training to employees and contractors so as to avoid unnecessary adverse impacts on biodiversity during the construction and operation stages.	The management of Biodiversity throughout the life of the Scheme is covered by the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025] Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline OEMP [REP5-020] and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023]. The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] sets out that an Environmental Clerk of Works (ECOW) will provide advice about environmental and ecological issues during



		construction including for example, management of protected species, surface water management, pollution, air quality and noise.
		It is therefore considered that the Scheme is compliant with this policy.
Paragraph 5.4.44	The Secretary of State should consider what appropriate requirements should be attached to any consent and/or in any planning obligations entered into, in order to ensure that any mitigation or biodiversity net gain measures, if offered, are delivered and maintained. Any habitat creation or enhancement delivered including linkages with existing habitats for compensation or biodiversity net gain should generally be maintained for a minimum period of 30 years, or for the lifetime of the project, if longer.	The Outline Landscape and Ecology Management Plan (LEMP) [EN010132/EX6/WB7.3_EREP6-025] outlines proposed habitat creation at the site and the Biodiversity Design Strategy.
		The Scheme is covered by the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021]), Outline OEMP [REP5-020] and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023]. Production of a final CEMP, OEMP and DEMP are secured by way of a requirement in the draft DCO.
		Habitat created by the Scheme would be managed and maintained through the operational life of the Scheme, which is expected to exceed 30 years.
		It is therefore considered that the Scheme is compliant with this policy.
Paragraph 5.7.1	During the construction, operation and decommissioning of energy infrastructure there is potential for the release of a range of emissions such as odour, dust, steam, smoke, artificial light and infestation of insects. All have the potential to have a detrimental impact on amenity or cause a common law nuisance or statutory nuisance under Part III, Environmental Protection Act 1990. However, they are not	Emissions resulting in potential statutory nuisances have been captured and assessed within the Statutory Nuisance Statement [APP-317]. Mitigation measures to limit the rise of statutory nuisances have been detailed within the Outline Construction Management Plan [EN010132/EX6/WB7.1_DREP6-021], Outline Operational



	regulated by the environmental permitted regime, so mitigation of these impacts will need to be included in the Development Consent Order.	Management Plan [REP5-020] and Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023].
Paragraph 5.7.5	The applicant should assess the potential for insect infestation and emissions of odour, dust, steam, smoke and artificial light to have a detrimental impact on amenity, as part of the ES.	The potential for insect infestation and emissions of odour, dust, steam, smoke, and artificial light to have a detrimental impact on amenity have been taken forward in the ES to have a detrimental impact.
Paragraph 5.8.13	A site-specific flood risk assessment should be provided for all energy projects in Flood Zones 2 and 3 in England or Zones B and C in Wales. In Flood Zone 1 in England or Zone A in Wales, an assessment should accompany all proposals involving: • sites of 1 hectare or more • land which has been identified by the EA or NRW as having critical drainage problems • land identified (for example in a local authority strategic flood risk assessment) as being at increased flood risk in future • land that may be subject to other sources of flooding (for example surface water) • where the EA or NRW, Lead Local Flood Authority, Internal Drainage Board or other body have indicated that there may be drainage problems. This assessment should identify and assess the risks of all forms of flooding to and from the project and demonstrate how these flood risks will be managed, taking climate change into account.	Flood Risk Assessments (FRA) are provided at Appendices 10.1 – 10.6 of the ES [APP-089 to APP-094]. The FRA provides a detailed assessment of the risk of flooding to and from the Scheme (taking account of climate change) and concludes that the risk of flooding will not be increased as a result of the construction, operation or decommissioning of the Scheme. It is therefore considered that the Scheme is compliant with this policy.



Paragraph 5.8.15

The minimum requirements for Flood Risk Assessments (FRA) are that they should:

- be proportionate to the risk and appropriate to the scale, nature and location of the project
- consider the risk of flooding arising from the project in addition to the risk of flooding to the project
- take the impacts of climate change into account, across a range of climate scenarios, clearly stating the development lifetime over which the assessment has been made;
- be undertaken by competent people, as early as possible in the process of preparing the proposal
- consider both the potential adverse and beneficial effects of flood risk
 management infrastructure, including raised defences, flow channels, flood
 storage areas and other artificial features, together with the consequences of
 their failure and exceedance
- consider the vulnerability of those using the site, including arrangements for safe access and escape
- consider and quantify the different types of flooding (whether from natural and human sources and including joint and cumulative effects) and include information on flood likelihood, speed-of-onset, depth, velocity, hazard and duration
- identify and secure opportunities to reduce the causes and impacts of flooding overall, making as much use as possible of natural flood management techniques as part of an integrated approach to flood risk management

Flood Risk Assessments (FRA) are provided at Appendices 10.1 – 10.6 of the ES [APP-089 to APP-094]. The FRA provides a detailed assessment of the risk of flooding to and from the Scheme (taking account of climate change) and concludes that the risk of flooding will not be increased as a result of the construction, operation or decommissioning of the Scheme. This professional judgement has been made as a result of considering both the potential adverse and beneficial effects anticipated through the design of the Scheme. The FRAs have considered and addressed the criteria within this paragraph.

It is therefore considered that the Scheme is compliant with this policy.



- consider the effects of a range of flooding events including extreme events on people, property, the natural and historic environment and river and coastal processes
- include the assessment of the remaining (known as 'residual') risk after risk reduction measures have been taken into account and demonstrate that these risks can be safely managed, ensuring people will not be exposed to hazardous flooding
- consider how the ability of water to soak into the ground may change with development, along with how the proposed layout of the project may affect drainage systems. Information should include:
 - i. Describe the existing surface water drainage arrangements for the site
 - ii. Set out (approximately) the existing rates and volumes of surface water run-off generated by the site. Detail the proposals for restricting discharge rates
 - iii. Set out proposals for managing and discharging surface water from the site using sustainable drainage systems and accounting for the predicted impacts of climate change. If sustainable drainage systems have been rejected, present clear evidence of why their inclusion would be inappropriate
 - iv. Demonstrate how the hierarchy of drainage options has been followed.
 - v. Explain and justify why the types of SuDS and method of discharge have been selected and why they are considered appropriate.



	vi.	Explain how sustainable drainage systems have been integrated with other aspects of the development such as open space or green infrastructure, so as to ensure an efficient use of the site	
	vii.	Describe the multifunctional benefits the sustainable drainage system will provide	
	viii.	Set out which opportunities to reduce the causes and impacts of flooding have been identified and included as part of the proposed sustainable drainage system	
	ix.	Explain how run-off from the completed development will be prevented from causing an impact elsewhere	
	X.	Explain how the sustainable drainage system been designed to facilitate maintenance and, where relevant, adoption. Set out plans for ensuring an acceptable standard of operation and maintenance throughout the lifetime of the development	
	safe	I those measures that will be included to ensure the development will be and remain operational during a flooding event throughout the lopment's lifetime without increasing flood risk elsewhere;	
		ify and secure opportunities to reduce the causes and impacts of ling overall during the period of construction; and	
		upported by appropriate data and information, including historical mation on previous events.	
Paragraph			A sequential Test has been conducted where it is considered
5.8.22		ogy specific NPSs set out some exceptions to the application of the	that the Scheme passes the test requirements [APP-094].
		est. However, when seeking development consent on a site allocated in a t plan through the application of the Sequential Test, informed by a	Given that the Scheme falls within areas of Flood Zone 3, the Scheme has been subject to the Exception Test [APP-094]. It



	strategic flood risk assessment, applicants need not apply the Sequential Test, provided the proposed development is consistent with the use for which the site was allocated and there is no new flood risk information that would have affected the outcome of the test.	has been concluded that the Scheme meets the requirements of the Exception Test and is therefore in conformity with this paragraph.
5.8.23	Consideration of alternative sites should take account of the policy on alternatives set out in Section 4.3 above. All projects should apply the Sequential Test to locating development within the site.	
Paragraph 5.8.9	If, following application of the Sequential Test, it is not possible, (taking into account wider sustainable development objectives), for the project to be located in areas of lower flood risk the Exception Test can be applied as defined in https://www.gov.uk/guidance/flood-risk-and-coastal-change#table2. The test provides a method of allowing necessary development to go ahead in situations where suitable sites at lower risk of flooding are not available.	It has not been considered possible to locate the Scheme in areas of lower flood risk as explored within THE Sequential and Exception Test Report [APP-094]. As such, the Report has included an assessment of the Exception Test where it is considered the Scheme is in conformity with it's requirements.
Paragraph 5.8.10	The Exception Test is only appropriate for use where the Sequential Test alone cannot deliver an acceptable site. It would only be appropriate to move onto the Exception Test when the Sequential Test has identified reasonably available, lower risk sites appropriate for the proposed development where, accounting for wider sustainable development objectives, application of relevant policies would provide a clear reason for refusing development in any alternative locations identified. Examples could include alternative site(s) that are subject to national designations such as landscape, heritage and nature conservation designations, for example Areas of Outstanding Natural Beauty (AONBs), SSSIs and World Heritage Sites (WHS) which would not usually be considered appropriate.	The Sequential Test was not able to deliver an acceptable Site in terms of flood risk. Resultingly, the Exception Test was conducted where it was found that the Scheme is in conformity with the requirements of the Test [APP-094].
Paragraph 5.8.11	Both elements of the Exception Test will have to be satisfied for development to be consented. To pass the Exception Test it should be demonstrated that: • the project provides wider sustainability benefits to the community that outweigh flood risk; and	The Exception Test presented in [APP-094] has considered these two criteria's. The Scheme is considered to satisfy these two criteria's and as such complies with this paragraph.



	 the project will be safe for its lifetime taking account of the vulnerability of its users, without increasing flood risk elsewhere, and, where possible will reduce flood risk overall. 	
Paragraph 5.9.1	The construction, operation and decommissioning of energy infrastructure has the potential to result in adverse impacts on the historic environment above, at and below the surface of the ground.	The historic environment impacts of the Scheme and the impacts associated significance in relation to above, at and below ground level Construction, Operation and Decommissioning has been explored within section 13.7 of ES Chapter 13: Cultural Heritage [APP-051].
		The WSI [REP5-016] ensures the Scheme complies with this paragraph.
Paragraph 5.9.6	Non-designated heritage assets of archaeological interest that are demonstrably of equivalent significance to Scheduled Monuments or Protected Wreck Sites should be considered subject to the same policies for considerations designated heritage assets. The absence of designation for such heritage assets does not indicate lower significance or necessarily imply that it is not of national importance.	Non designated heritage assets with archaeological interest are identified in ES Chapter 13: Cultural Heritage of the ES [APP-051]. Section 13.5 of Chapter 13: Cultural Heritage of the ES describes these assets and their significance. Section 13.7 of ES Chapter 13: Cultural Heritage [APP-051], considers the likely significant effects of the Scheme.
Paragraph 5.9.9	The applicant should undertake an assessment of any likely significant heritage impacts of the proposed development as part of the EIA and describe these along with how the mitigation hierarchy has been applied in the ES (see Section 4.3). This should include consideration of heritage assets above, at, and below the surface of the ground. Consideration will also need to be given to the possible impacts, including cumulative, on the wider historic environment. The assessment should include reference to any historic landscape or seascape character assessment and associated studies as a means of assessing impacts relevant to the proposed project.	Section 13.7 of Chapter 13: Cultural Heritage of the ES [APP-051] contains a clear and detailed assessment of likely impacts and effects of the Scheme on cultural heritage.
Paragraph 5.9.10	As part of the ES the applicant should provide a description of the significance of the heritage assets affected by the proposed development, including any contribution made by their setting. The level of detail should be proportionate to the importance	Paragraphs 3.2.49 to 3.2.59 of ES Appendix 13.5 Heritage Statement [APP-117 to APP-119] assess the significance of the Scheduled Monument and detail the contribution made



of the heritage assets and no more than is sufficient to understand the potential impact of the proposal on their significance. As a minimum the applicant should have consulted the relevant Historic Environment Record (or, where the development is in English or Welsh waters, Historic England or Cadw) and assessed the heritage assets themselves using expertise where necessary according to the proposed development's impact.

by its setting, namely the land which was formerly located within the deer park and is not scheduled. Table 5.1 of the Statement of Common Ground (SOCG) with Historic England [EN010132/EX6/WB8.3.3_AREP6-042] provides additional detail for how the Applicant's conclusions in identifying how the significance of the monument have been derived, and the potential for harm caused by the Scheme to that significance.

The Scheme would not cause any direct physical harm to the significance of the Scheduled Monument as there is no proposed intervention to the fabric of any of the sections of the Scheduled Monument that would result in its permanent loss either wholly or in part. Any harm would be only that caused to the significance of the monument that is derived from its setting. This would occur through the placement of panels within land that was formerly occupied by the medieval deer park.

As detailed in Historic England Advice Note 12¹ pages 5, 15 and 16, a key element of a heritage assessment is to identify how the significance of a heritage asset is derived.

"The context for any analysis of the significance of a heritage asset will be a thorough familiarity with the asset itself, developed through site visits, and appropriate inspection of the

135 | Page

¹ Historic England (2019) *Statements of Heritage Significance: Analysing Significance in Heritage Assets: Historic England Advice Note 12.* (Online, last accessed 27.03.2024) https://historicengland.org.uk/images-books/publications/statements-heritage-significance-advice-note-12/heag279-statements-heritage-significance/



fabric, its features, materials and ornament, and also its setting if needed." (Historic England Advice Note 12 page 5)

This is reiterated on Page 4 of Historic England Planning Note 3 (Second Edition):

"Setting is not itself a heritage asset, nor a heritage designation, although land comprising a setting may itself be designated (see below Designed settings). Its importance lies in what it contributes to the significance of the heritage asset or to the ability to appreciate that significance."

As evidenced at ISH5 and in the SoCG with Historic England [EN010132/EX6/WB8.3.3 AREP6-042], the Applicant understands the Scheduled Monument derives its significance from its historic interest as the sole surviving element of a former enclosed medieval space, which is largely understood through desk-based research, particularly aerial imagery and historical documentation. Section 2 of the Cultural Heritage Position Statement [REP5-027] confirms how this view has been formed, based on the Reasons for Designation set out in Official List Entry for the Scheduled Monument. The agrarian landscape, the former MOD petroleum site and the railway, which bisects the Scheduled Monument, have a detrimental effect on the ability to appreciate any remaining elements of the former medieval landscape and are consequently considered to have a detrimental effect on the overall contribution made by setting to the significance of the Scheduled Monument.



Paragraph 5.9.12	The applicant should ensure that the extent of the impact of the proposed development on the significance of any heritage assets affected can be adequately understood from the application and supporting documents. Studies will be required on those heritage assets affected by noise, vibration, light and indirect impacts, the extent and detail of these studies will be proportionate to the significance of the heritage asset affected.	Section 13.5 of Chapter 13: Cultural Heritage of the ES [APP-051] describes the heritage assets within the study area for the Scheme and their significance and the contribution of their setting to that significance.
		Section 13.7 of Chapter 13: Cultural Heritage of the ES [APP-051] contains a clear and detailed assessment of likely impacts and effects of the Scheme on cultural heritage.
		A Non-Technical Summary has been deduced from each Chapter, inclusive of Chapter 13: Cultural Heritage of the ES [APP-051] which ensures that the core elements of the Chapter can be understood simply and at a glance.
Paragraph	The applicant is encouraged, where opportunities exist, to prepare proposals which	Section 13.8 of Chapter 13: Cultural Heritage of the ES [APP-
5.9.13	can make a positive contribution to the historic environment, and to consider how their scheme takes account of the significance of heritage assets affected. This can include, where possible:	051] outlines the mitigation measures embedded within the Scheme design pertaining to cultural heritage. This includes the provision of stand-offs between the Scheme and heritage
	 enhancing, through a range of measures such a sensitive design, the significance of heritage assets or setting affected 	assets in order to help to preserve their setting during the construction, operational and decommissioning periods.
	 considering measures that address those heritage assets which are at risk or which may become at risk, as a result of the scheme 	Appropriate and sensitive screening has also been developed and implemented to minimise the visual intrusion of the Scheme, while avoiding obscuring or intruding upon key
	 considering how visual or noise impacts can affect heritage assets, and whether there may be opportunities to enhance access to, or interpretation, understanding and appreciation of, the heritage assets affected by the scheme. 	views and relationships between heritage assets.
		The Scheme will also be decommissioned at the end of its operational life. Following decommissioning, any impacts on the setting of heritage assets as a result of the solar farm will have been reversed.



Paragraph 5.9.14	Careful consideration in preparing the scheme will be required on whether the impacts on the historic environment will be direct or indirect, temporary or permanent.	The nature of the Scheme was considered by the Applicant as part of the assessment on Cultural Heritage that is provided in ES Chapter 13 Cultural Heritage [APP-051]. A key aspect of the Scheme is its reversible nature which means that landscape features, such as those that are associated with the former deer park, will not be impacted by the Scheme. As such any harm caused to the significance of the Scheduled Monument that is derived from its setting would be reversed following decommissioning of the Scheme.
Paragraph 5.9.15	Applicants should look for opportunities for new development within Conservation Areas and World Heritage Sites, and within the setting of heritage assets, to enhance or better reveal their significance. Proposals that preserve those elements of the setting that make a positive contribution to the asset (or which better reveal its significance) should be treated favourably.	There are no World Heritage Sites affected by the Scheme.
Paragraph 5.9.22	In determining applications, the Secretary of State should seek to identify and assess the particular significance of any heritage asset that may be affected by the proposed development, including by development affecting the setting of a heritage asset (including assets whose setting may be affected by the proposed development), taking account of: • relevant information provided with the application and, where applicable, relevant information submitted during the examination of the application	See response to 5.9.10 above
	 any designation records, including those on the National Heritage List for England, or included on Cof Cymru for Wales. historic landscape character records 	



	 the relevant Historic Environment Record(s), and similar sources of information representations made by interested parties during the examination process expert advice, where appropriate, and when the need to understand the significance of the heritage asset demands it. 	
Paragraph 5.9.25	The Secretary of State should consider the desirability of sustaining and, where appropriate, enhancing the significance of heritage assets, the contribution of their settings and the positive contribution that their conservation can make to sustainable communities, including to their quality of life, their economic vitality, and to the public's enjoyment of these assets.	Whilst there is currently no public access to the Scheduled Monument (i.e. public rights of way) providing an ability to experience the designated heritage asset or its understanding, this should not inhibit the provision to enhance the potential for communal benefit. How it will be experienced will vary over time and circumstance (as stated in Historic England's GPA3 The Setting of Heritage Assets). As evidenced in Paragraphs 4.1.1 to 4.1.2 of the Cultural Heritage Position Statement [REP5-027], the Applicant explored a range of different mitigation options that had the potential to enhance the public enjoyment of the asset (for example suggestion IV). Historic England believed the benefits from community engagement would not offset any harm, and so these options were not explored further or considered as part of the design of the Scheme.



Paragraph 5.9.27	When considering the impact of a proposed development on the significance of a designated heritage asset, the Secretary of State should give great weight to the asset's conservation. The more important the asset, the greater the weight should be. This is irrespective of whether any potential harm amounts to substantial harm, total loss, or less than substantial harm to its significance.	
5.9.28	The Secretary of State should give considerable importance and weight to the desirability of preserving all heritage assets. Any harm or loss of significance of a designated heritage asset (from its alteration or destruction, or from development within its setting) should require clear and convincing justification.	
5.9.29	Substantial harm to or loss of significance of a grade II Listed Building park or grade II Registered Park or Garden should be exceptional.	
5.9.30	Substantial harm to or loss of significance of assets of the highest significance, including Scheduled Monuments; Protected Wreck Sites; Registered Battlefields; grade I and II* Listed Buildings; grade I and II* Registered Parks and Gardens; and World Heritage Sites, should be wholly exceptional.	
5.9.31	Where the proposed development will lead to substantial harm to (or total loss of significance of) a designated heritage asset the Secretary of State should refuse consent unless it can be demonstrated that the substantial harm to, or loss of, significance is necessary to achieve substantial public benefits that outweigh that harm or loss, or all the following apply:	
	the nature of the heritage asset prevents all reasonable uses of the site	
	no viable use of the heritage asset itself can be found in the medium	

term through appropriate marketing that will enable its conservation

or public ownership is demonstrably not possible

conservation by grant-funding or some form of not for profit, charitable

It is acknowledged by the Applicant that substantial harm to or loss of significance of assets of the highest significance, which includes Scheduled Monuments such as The medieval bishop's palace and deer park, Stow Park (NHLE 1019229), should be wholly exceptional and consent should be refused for that element of the Scheme unless it can be demonstrated that the substantial harm to or loss of significance is necessary to achieve substantial public benefits.

Through thorough assessment, the Applicant does not consider that the Scheme would cause substantial harm to The medieval bishop's palace and deer park, Stow Park (NHLE 1019229). As detailed in Paragraph 3.1.1 of the Cultural Heritage Position Statement [REP5-027] the Scheme would not cause any direct impact to the fabric of the Scheduled Monument, and there would be no adverse effects to its heritage values that would result in its permanent loss either wholly or in part and consequently the legibility of the deer park would be unaltered. Any effects resulting in a level of harm to the significance of the monument would be derived from changes to its setting through the placement of panels within land that was formerly occupied by the medieval deer park.

The Applicant believes, as evidenced by the Official List Entry for the Scheduled Monument (See Section 2 Paragraphs 2.1.3 to 2.1.7 of the Cultural Heritage Position Statement [REP5-027],), that the significance of the Scheduled Monument is



• the harm or loss is outweighed by the benefit of bringing the site back into use "

primarily derived from its historical and archaeological interest, vested in the Scheduled earthwork features and potential below ground remains, together with that appreciated through desk-based research, particularly aerial imagery and historical documentation. Setting contributes to the understanding of these heritage interests, albeit denuded by the current composition of the landscape in which the Scheduled Monument is located. The post-medieval and modern agrarian land uses does preclude the ability to experience or appreciate the former medieval landscape of the Scheduled Monument. Furthermore, this same post medieval and modern activity has resulted in an adverse effect on elements within its setting as evidenced in the list entry which deliberately excludes post-medieval and modern features and highlights the adverse direct impacts that agricultural activity has had on the Scheduled Monument (see Paragraph 2.1.7 of the Cultural Heritage Position Statement [REP5-027]).

The reversible nature of the Scheme means that any harm to significance as a result of changes in the setting of the Scheduled Monument would be temporary and reversed entirely following decommissioning of the Scheme.



Paragraph 5.9.32	Where the proposed development will lead to less than substantial harm to the significance of the designated heritage asset, this harm should be weighed against the public benefits of the proposal, including, where appropriate securing its optimum viable use.	The Applicant's assessment has found that the Scheme would cause less than substantial harm (at the upper end) to the Scheduled Monument. The introduction of solar panels would not cause direct physical harm to the three isolated elements of Scheduled Monument that form the surviving vestiges of the deer park. Any harm would therefore be solely to the significance the Scheduled Monument derives from its setting (i.e. would be indirect). The layout of the Scheme means that the legibility of the landscape would be unaltered. This is an important factor in the consideration of the temporary nature of the Scheme and any harm to the significance as a result of changes in the setting of the Scheduled Monument which would be reversed entirely following decommissioning of the Scheme. As detailed in the Statement of Common Ground [EN01032/EX6/WB8.3.3_AREP6-042], the Applicant notes that "Historic England considers that the impact of the Scheme on land within the former deer park as defined by The medieval bishop's palace and deer park, Stow Park Scheduled Monument (NHLE 1019229) would cause substantial harm (in NPS/NPPF terms) / significant environmental impact (major harmful; in EIA terms) to the significance of the Monument through loss of its character as a bounded architectural space."



Consequently HE "object to installation of any part of the development within the former deer park (as defined by the lines of the scheduled Park Pale and its former course)."2 The Applicant respectfully disagrees that the Scheme represents a loss to the character of the bounded architectural space of the former deer park. The internal space of the Deer Park does not have any designation (i.e. form a Scheduled Monument, Registered Park and Garden, or Conservation Area). The Applicant believes that this is largely due to the absence of any landscape features that can be attributed or associated with the deer park and that would add to our understanding of how the it functioned. Additionally, the sense of a space imparked is not clearly appreciable with the current land use, as that both within and beyond the former boundaries being indistinguishable in its agricultural use. Consequently, the surviving vestiges of the deer park are not experienced collectively within the modern landscape, and it is difficult to reconstruct, understand and appreciate an imparked high status medieval space without the aid of aerial imagery or historical documentation. Instead, the experience is of an agrarian landscape, and the post enclosure field system is the dominant experience. If the Secretary of State is minded to agree that the Scheme will cause less than substantial harm, the Applicant notes that the Secretary of State must give "great weight to the asset's conservation" and "considerable importance and weight to the

² Please see **Draft-Statement of Common Ground with Historic England [EN010132/EX6/WB8.3.3_AREP6-042]** Table 5.1



		desirability of preserving" the asset with any harm or loss of significance require clear and convincing justification. Having applied the abovementioned weight, the policy tests confirm that where the public benefits of the Scheme can be demonstrated to outweigh harm to the significance (as a result of changes to the setting) of the three elements constituting the Scheduled Monument (NHLE 1019229), consent should be approved. Sections 4 and 5 of the Cultural heritage Position Statement [REP5-027] evidence the benefits of the proposed development, which are considered to outweigh any potential harm to the Scheduled Monument.
Paragraph 5.9.33	In weighing applications that directly or indirectly affect non-designated heritage assets, a balanced judgement will be required having regard to the scale of any harm or loss and the significance of the heritage asset.	Non designated heritage assets are identified in Chapter 13: Cultural Heritage of the ES [APP-051]. Section 13.7 of Chapter 13 Cultural Heritage of the ES describes these assets and their significance. It identifies some significant effects upon non-designated heritage assets as a result of the Scheme. There are a number of Non-Designated Historic Landscape being:
		HLI21266, HLI20787, HLI20791, HLI20860 which all experience moderate adverse residual effects. The Statement of Need [APP-320] presents a detailed and compelling case for why the Scheme is urgently required and at the scale proposed. The scale of the Scheme inherently poses some adverse harm upon some non-designated heritage assets, with a balanced judgement in mind, it is considered that this harm is demonstrably outweighed by the benefits of the Scheme.



Paragraph 5.9.35	Where there is evidence of deliberate neglect of, or damage to, a heritage asset, the Secretary of State should not take its deteriorated state into account in any decision.	There are no heritage assets identified in the study area where evidence was found of deliberate neglect of, or damage to, the asset.
Paragraph 5.9.36	When considering applications for development affecting the setting of a designated heritage asset, the Secretary of State should give appropriate weight to the desirability of preserving the setting such assets and treat favourably applications that preserve those elements of the setting that make a positive contribution to, or better reveal the significance of, the asset. When considering applications that do not do this, the Secretary of State should give significant weight to any negative effects, when weighing them against the wider benefits of the application. The greater the negative impact on the significance of the designated heritage asset, the greater the benefits that will be needed to justify approval.	As evidenced in Paragraphs 4.1.1 to 4.1.2 of the Cultural Heritage Position Statement [REP5-027], the Applicant explored a range of different mitigation options that had the potential to better reveal the significance of the Asset (for example suggestion III). Historic England believed the benefits from community engagement would not offset any harm, and so these options weren't explored further or considered as part of the design of the Scheme.
Paragraph 5.10.22	The assessment should also address the landscape and visual effects of how noise and light pollution, and other emissions (see Section 5.2 and Section 5.7), from construction and operational activities on residential amenity and on sensitive locations, receptors and views, how these will be minimised.	Artificial lighting will be required during construction and decommissioning in areas where natural lighting is unable to reach (sheltered/confined areas), and during core working hours within winter months. All construction lighting will be deployed in accordance with the recommendations set out in the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021].
		Details of operational lighting are set out by Chapter 4, Scheme Description, of the ES [APP-042] This explains that no part of the Scheme will be continuously lit. Manually operated, and motion-detection lighting will be utilised for operational and security purposes around electrical infrastructure. Lighting will be directed downward and away from boundaries. No visible lighting will be utilised at the site perimeter fence, aside from the site entrance points.



		The impact of lighting is taken into account in the visual assessment for residential receptors set out in Chapter 8, Landscape and Visual Impact of the ES [APP-046].
		The impact of noise from the Scheme on residential receptors is assessed in Chapter 15, Noise and Vibration, of the ES [APP-053].
Paragraph	Landscape effects arise not only from the sensitivity of the landscape but also the	Due consideration has been given to the effects of the
5.10.4	nature and magnitude of change proposed by the development, whose specific siting and design make the assessment a case-by-case judgement.	Scheme upon the landscape. As detailed in Section 6.4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the
5.10.5	Virtually all nationally significant energy infrastructure projects will have adverse effects on the landscape, but there may also be beneficial landscape character impacts arising from mitigation.	Scheme has been subject to a detailed and sensitive iterative design process. This has taken account of the context and features of the land within the Order limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts, and provision of environmental and other enhancements, where practicable. The design process and basis of design decisions taken are described in the Chapter 5: Alternatives and Design Evolution of the ES [APP-043] and the Design and Access Statement [APP-314 to APP-315].
5.10.6	Projects need to be designed carefully, taking account of the potential impact on the landscape. Having regard to siting, operational and other relevant constraints the aim should be to minimise harm to the landscape, providing reasonable mitigation where possible and appropriate.	
Paragraph 5.10.24	Applicants should consider how landscapes can be enhanced using landscape management plans, as this will help to enhance environmental assets where they contribute to landscape and townscape quality.	Good design has been a key consideration from the outset. The LVIA has informed the iterative design process, including taking account of published landscape character assessment guidance and fieldwork analysis.



		The overall objective of the landscape design is to integrate the Scheme into its landscape setting and avoid or minimise adverse landscape and visual effects as far as practicable. The design has been developed in collaboration with the wider design team, other specialists and the Host Authorities landscape advisors to achieve a solution that achieves this objective whilst maximising opportunities to deliver net gains in biodiversity gain. Accordingly, the landscape design aims to achieve the following:
		To integrate the Scheme into the existing landscape pattern as far as possible by retaining and following existing features, including vegetation, where practicable.
		To replace vegetation lost because of construction of the Scheme through areas of new planting.
		To filter and screen more prominent components of the Scheme in views from visual receptors
		Details of the landscape measures embedded into the Scheme design, including a summary of their environmental functions, is presented in the Outline LEMP [EN010132/EX6/WB7.3_E]-REP6-025].
Paragraph 5.10.12	Outside nationally designated areas, there are local landscapes that may be highly valued locally. Where a local development document in England or a local development plan in Wales has policies based on landscape or waterscape character assessment, these should be paid particular attention. However, locally valued landscapes should not be used in themselves to refuse consent, as this may unduly restrict acceptable development.	As detailed within ES Chapter 3: The Order Limits [APP-041], the Sites are within close proximity to two Areas of Great Landscape Value. Consideration for these designations has been captured within ES Chapter 8: Landscape and Visual Impact [APP-046] with mitigation measures incorporated where required.



Paragraph		The majority of the Order Limits comprises Grade 3b
5.11.12	Applicants should seek to minimise impacts on the best and most versatile agricultural land (defined as land in grades 1, 2 and 3a of the Agricultural Land Classification) and preferably use land in areas of poorer quality (grades 3b, 4 and 5).	agricultural land. 26.24% BMV land is included within the Order Limits. This is justified by other sustainability considerations, as explained in Section 6.7 of this Planning Statement [EN010132/EX6EX7/WB7.5_CD].
5.11.13	Applicants should also identify any effects and seek to minimise impacts on soil health and protect and improve soil quality taking into account any mitigation measures proposed.	
5.11.18	For developments on previously developed land, applicants should ensure that they have considered the risk posed by land contamination, and where contamination is present, applicants should consider opportunities for remediation where possible. It is important to do this as early as possible as part of engagement with the relevant bodies before the official pre-application stage.	
5.11.14	Applicants are encouraged to develop and implement a Soil Management Plan which could help minimise potential land contamination. The sustainable reuse of soils needs to be carefully considered in line with good practice guidance where large quantities of soils are surplus to requirements or are affected by contamination.	
Paragraph 5.11.19	Applicants should safeguard any mineral resources on the proposed site as far as possible, taking into account the long-term potential of the land use after any future decommissioning has taken place.	As detailed within ES Chapter 3: The Order Limits [APP-041], the Sites fall within a Petroleum Exploration Development Licence (PEDL) Block with sections of West Burton 3 being within a Sand and Gravel Minerals Safeguarding Area. Given the unintrusive nature of the Scheme, it is proposed that no adverse harm will occur upon these Safeguarding Areas.
Paragraph 5.11.28	Where a proposed development has an impact upon a Mineral Safeguarding Area (MSA), the Secretary of State should ensure that appropriate mitigation measures have been put in place to safeguard mineral resources.	The Scheme is not anticipated to result in any impacts upon Mineral Safeguarding Areas.
Paragraph		The Scheme has sought to minimise disruption upon Public Rights of Way (PRoW) and Open Access land. Where this has



5.11.30	Public Rights of way, National Trails and other rights of access to land are important recreational facilities for example for walkers, cyclists and horse riders. The Secretary of State should expect applicants to take appropriate mitigation measures to address adverse effects on coastal access, National Trails, other rights of way and open access land and, where appropriate, to consider what opportunities there may be to improve or create new access. In considering revisions to an existing right of way,	not been possible, ensuring minimal disruption, providing ample notice and ensuring continued passage has been of paramount importance. Where PRoW issues have arisen, the Public Rights of Way Management Plan [REP5-018] has captured appropriate mitigation measures.
	consideration should be given to the use, character, attractiveness and convenience of the right of way.	The Scheme, through Works No. 11, seeks to create a new permissive footpath. This is detailed within the Draft Development Consent Order Revision C-{EN010132/EX6H
5.11.31	The Secretary of State should consider whether the mitigation measures put forward by an applicant are acceptable and whether requirements or other provisions in respect of these measures should be included in any grant of development consent.	[EX7/WB3.1_GH] submitted at Deadline 37.
Paragraph 5.12.5	Factors that will determine the likely noise impact of a proposed development include:	Noise generation arising from the Scheme have been identified within ES Chapter 15: Noise [APP-053] section 15.7.
	 the inherent operational noise from the proposed development, and its characteristics 	Through embedded mitigation measures (detailed within section 15.6), no significant residual effects are predicted during construction, operation and decommissioning of the
	 the proximity of the proposed development to noise sensitive premises (including residential properties, schools and hospitals) and noise sensitive areas (including certain parks and open spaces) 	Scheme.
	the proximity of the proposed development to quiet places and other areas that are particularly valued for their soundscape or landscape quality	
	 the proximity of the proposed development to sites where noise may have an adverse impact on protected species or other wildlife, including migratory species 	
	the potential presence of unexploded ordnance on the seabed	



Paragraph 5.12.6

Where noise impacts are likely to arise from the proposed development, the applicant should include the following in the noise assessment:

- a description of the noise generating aspects of the development proposal leading to noise impacts, including the identification of any distinctive tonal characteristic, if the noise is impulsive, whether the noise contains particularly high or low frequency content or temporal characteristics of the noise
- identification of noise sensitive receptors and noise sensitive areas that may be affected
- the characteristics of the existing noise environment
- a prediction of how the noise environment will change with the proposed development
 - o in the shorter term, such as during the construction period
 - o in the longer term, during the operating life of the infrastructure
 - at particular times of the day, evening and night (and weekends) as appropriate, and at different times of year
- an assessment of the effect of predicted changes in the noise environment on any noise-sensitive receptors, including an assessment of any likely impact on health and quality of life / and well-being where appropriate, particularly among those disadvantaged by other factors who are often disproportionately affected by noise-sensitive areas
- if likely to cause disturbance, an assessment of the effect of underwater or subterranean noise

Chapter 15: Noise & Vibration of the ES [APP-053] presents a noise assessment in accordance with the requirements of this policy.

Appendix 15.5 of Chapter 15: Noise & Vibration of the ES **[APP-053]** describes the noise sensitive premises and areas that have been identified. These have been determined through desktop study during the scoping process and confirmed during site visits. The locations of these receptors have been considered in both the construction and operational noise assessments and are considered representative of adjacent properties.

Section 15.5 of Chapter 15: Noise and Vibration of the ES **[APP-053]** outlines the characteristics of the existing noise environment for the Scheme and surrounding areas.

Section 15.6 of Chapter 15: Noise and Vibration of the ES **[APP-053]** describes the embedded design mitigation for the Scheme with respect to noise and vibration, encompassing the construction, operation and decommissioning phases.

Section 15.7 of Chapter 15: Noise and Vibration of the ES **[APP-053]** assesses the noise generated by the Scheme during the construction period and operating life of the infrastructure (including tonality), including at particular times of the day and at night, on the noise sensitive premises and areas outlined in Table 15.3 of Chapter 15: Noise & Vibration of the ES **[APP-053]**.



	all reasonable steps taken to mitigate potential adverse effects on health and quality of life	
Paragraph 5.12.10	Some noise impacts will be controlled through environmental permits and parallel tracking is encouraged where noise impacts determined by an environmental permit interface with planning issues (i.e., physical design and location of development). The applicant should consult EA and/or the SNCB, and other relevant bodies, such the MMP or NRW, as necessary, and in particular regarding assessment of noise on protected species or other wildlife. The results of any noise surveys and predictions may inform the ecological assessment. The seasonality of potentially affected species in nearby sites may also need to be taken into account.	Chapter 9: Ecology and Biodiversity of the ES [APP-047] assesses the likely significant effects of the Scheme on protected species and other wildlife. The assessment takes account of noise impact and concludes that no significant effects arise. It is not expected that a protected species Environmental Permit will be needed. The Applicant has taken account of advice from the EA and Natural England in preparing the Environmental Statement [APP-039 to APP-061].
		Chapter 9, Ecology and Biodiversity, of the ES [APP-047] takes account of noise in its assessment of the impact of the Scheme on protected species and other wildlife.
5.12.12	Applicants should submit a detailed impact assessment and mitigation plan as part of any development plan, including the use of noise mitigation and noise abatement technologies during construction and operation.	Chapter 15: Noise & Vibration of the ES [APP-053] presents a noise assessment in accordance with the requirements of this policy.
		Appendix 15.5 of Chapter 15: Noise & Vibration of the ES [APP-053] describes the noise sensitive premises and areas that have been identified. These have been determined through desktop study during the scoping process and confirmed during site visits. The locations of these receptors have been considered in both the construction and operational noise assessments and are considered representative of adjacent properties.



		Section 15.5 of Chapter 15: Noise and Vibration of the ES [APP-053] outlines the characteristics of the existing noise environment for the Scheme and surrounding areas. Section 15.6 of Chapter 15: Noise and Vibration of the ES [APP-053] describes the embedded design mitigation for the
		Scheme with respect to noise and vibration, encompassing the construction, operation and decommissioning phases.
		Section 15.7 of Chapter 15: Noise and Vibration of the ES [APP-053] assesses the noise generated by the Scheme during the construction period and operating life of the infrastructure (including tonality), including at particular times of the day and at night, on the noise sensitive premises and areas outlined in Table 15.3 of Chapter 15: Noise & Vibration of the ES [APP-053].
Paragraph 5.13.4	This applicants assessment should consider all relevant socio-economic impacts, which may include:	Chapter 18: Socio-Economics, Tourism and Recreation of the ES [APP-056] includes an assessment of socio-economic
	 the creation of jobs and training opportunities. Applicants may wish to provide information on the sustainability of the jobs created, including where they will help to develop the skills needed for the UK's transition to Net Zero 	impacts and addresses all pointed mentioned.
	 the contribution to the development of low-carbon industries at the local and regional level as well as nationally 	
	 the provision of additional local services and improvements to local infrastructure, including the provision of educational and visitor facilities 	
	any indirect beneficial impacts for the region hosting the infrastructure, in particular in relation to use of local support services and supply chains	



	 effects (positive and negative) on tourism and other users of the area impacted the impact of a changing influx of workers during the different construction, operation and decommissioning phases of the energy infrastructure. This could change the local population dynamics and could alter the demand for services and facilities in the settlements nearest to the construction work (including community facilities and physical infrastructure such as energy, water, transport and waste). There could also be effects on social cohesion depending on how populations and service provision change as a result of the development 	
	 cumulative effects - if development consent were to be granted to for a number of projects within a region and these were developed in a similar timeframe, there could be some short-term negative effects, for example a potential shortage of construction workers to meet the needs of other industries and major projects within the region 	
Paragraph 5.13.5	Applicants should describe the existing socio-economic conditions in the areas surrounding the proposed development and should also refer to how the development's socio-economic impacts correlate with local planning policies.	Chapter 18: Socio-Economics, Tourism and Recreation of the ES [APP-056] includes a baseline assessment of the conditions of the area.
Paragraph 5.13.6	Socio-economic impacts may be linked to other impacts, for example visual impacts considered in Section 5.10 but may also have an impact on tourism and local businesses. Applicants are encouraged, where possible, to demonstrate that local suppliers are considered in any supply chain.	Chapter 18, Socio-economics, Recreation and Tourism, of the ES [APP-056] considers the socio-economic impact of the Scheme. It also sets out that in procurement of the contractor to complete the construction works, strong consideration will be given to their strategy for engaging the local supply chain and using local materials where possible and practical. The permanent jobs created to support the Scheme are a reflection of the requirements to maintain the infrastructure.



		An Outline Skills, Supply Chain and Employment Plan [EN010132/EX6/WB7.10_BREP6-027] will be prepared prior to the commencement of construction. This will set out measures that the Applicant will implement in order to
		- advertise and promote employment opportunities associated with the Scheme in construction and operation locally.
		- advertise those elements of the supply chain required for the construction and operation of the authorised development and which provide opportunities for Local Companies
Paragraph 5.13.7	Applicants should consider developing accommodation strategies where appropriate, especially during construction and decommissioning phases, that would include the need to provide temporary accommodation for construction workers if required.	Chapter 18, Socio-economics, Recreation and Tourism, of the ES [APP-056] considers temporary accommodation and the requirements for the housing of construction workers.
Paragraph		Section 4.6 of the Planning Statement
5.13.11	The Secretary of State should consider any relevant positive provisions the applicant has made or is proposing to make to mitigate impacts (for example through planning obligations) and any legacy benefits that may arise as well as any options for phasing development in relation to the socio-economic impacts.	[EN010132/EX6EX7/WB7.5_CD] describes some of the other the benefits of the Scheme, in addition to the energy and climate change benefits. Benefits of the Scheme to the local community (other than the generation of a substantial
5.13.12	The Secretary of State may wish to include a requirement that specifies the approval by the local authority of an employment and skills plan detailing arrangements to promote local employment and skills development opportunities, including apprenticeships, education, engagement with local schools and colleges and training	amount of renewable energy) are set out in Section 4.6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD]. These include: - A significant biodiversity net gain of 86.80% for habitats
	programmes to be enacted.	(delivered through the creation of other neutral grasslands within the sites), a net gain of 54.71% for



		hedgerows, and a net gain of 33.25% for river units as shown within the Biodiversity Net Gain Assessment [APP-088].
		- A new permissive footpath to run from the track off Sykes Lane along the Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub, improving connectivity across the Order limits.
		- Employment during the construction phase. It is expected that an average of 296 jobs will be created during the construction period. During the operational phase, 12 FTE staff would be employed on the site.
		- An Outline Skills, Supply Chain and Employment Plan [EN010132/EX6/WB7.10_BREP6-027] will be prepared prior to the commencement of construction. This will set out measures that the Applicant will implement in order to advertise and promote employment opportunities associated with the Scheme in construction and operation locally.
Paragraph		An Outline Construction Traffic Management Plan (CTMP) is
5.14.7	The applicant should prepare a travel plan including demand management and monitoring measures to mitigate transport impacts. The applicant should also provide details of proposed measures to improve access by active, public and shared transport to:	included as Appendix 14.2 of the ES [REP4- 038EN010132/EX7/WB6.3.14.2 E]. It outlines measures that will be included in the final CTMP to mitigate transport impact, manage demand, and improve and encourage construction staff to access the Order limits by public
	 reduce the need for parking associated with the proposal; 	transport, cycling and reduce car transport to, and parking at,
	contribute to decarbonisation of the transport network; and	the Order Limits.



	improve user travel options by offering genuine modal choice.	
5.14.8	The assessment should also consider any possible disruption to services and infrastructure (such as road, rail and airports).	
Paragraph 5.14.21	The Secretary of State should only consider refusing development on highways grounds if there would be an unacceptable impact on highway safety, residual cumulative impacts on the road network would be severe, or it does not show how consideration has been given to the provision of adequate active public or shared transport access and provision.	Section 14.7 of Chapter 14: Transport and Access of the ES [APP-052] states that there are anticipated to be no significant adverse effects on vehicle travellers, Non-Motorised Users (NMUs) or public transport users as a result of the construction, operation or decommissioning of the Scheme.
		The Scheme is also expected to have a negligible impact on accidents and safety for the remainder of the highway network.
		The Construction Traffic Management Plan (CTMP) has been prepared to minimise the impact of construction vehicle movement [REP4-038EN010132/EX7/WB6.3.14.2_E].
		As part of the Outline Decommissioning Plan [EN010132/EX6/WB7.2_BREP6-023], a Decommissioning Traffic Management Plan will be agreed to mitigate the impacts of decommissioning activities at the relevant time to reflect the conditions at the time.
		Therefore, it is considered that the Scheme is compliant with this policy.
Paragraph 5.15.1	Government policy on hazardous and non-hazardous waste is intended to protect human health and the environment by producing less waste and by using it as a resource wherever possible. Where this is not possible and disposal is required as a last resort, waste management regulation ensures that waste is disposed of in a way that is least damaging to the environment and to human health.	Section 20.7 of ES Chapter 20: Waste [APP-058] quantifies the estimated volume of waste that is to be produced during construction, operation, and decommissioning. The Scheme will seek to minimise and design out waste streams where possible. Opportunities to re-use materials, such as BESS



		equipment for renewable wind storage, will be explored prior to the recycling of equipment.
Paragraph 5.15.3	Disposal of waste should only be considered where other waste management options are not available or where it is the best overall environmental outcome.	The Scheme seeks to reuse wasted equipment where a second life use is possible. Where it is not, materials will be recycled as far as practical. These measures are captured within the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and the Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023].
Paragraph 5.16.5	Where possible, applicants are encouraged to manage surface water during construction by treating surface water runoff from exposed topsoil prior to discharging and to limit the discharge of suspended solids e.g., from car parks or other areas of hard standing, during operation.	The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] sets out measures to manage surface water runoff during the construction period, including limiting the discharge of suspended solids. This includes: - appropriate pollution control measures as agreed with the sewerage undertaker or the Environment Agency as appropriate;
		- following the relevant sections of BS 6031: Code of Practice for Earthworks for the general control of site drainage;
		 where practical, undertaking earthworks during the drier months of the year;
		- topsoil/subsoil will be stored a minimum of 20m from watercourses on flat lying land. Where this is not practicable, and it is to be stockpiled for longer than a two-week period, the material will either be covered with geotextile mats, seeded to promote vegetation



		growth, or runoff prevented from draining to a watercourse without prior treatment; and runoff storage areas for the settlement of excessive fine particulates in runoff will be provided.
Paragraph 5.16.6	Applicants are encouraged to consider protective measures to control the risk of pollution to groundwater beyond those outlined in River Basin Management Plans and Groundwater Protection Zones - this could include, for example, the use of protective barriers.	The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] details the measures that would be undertaken during construction to mitigate the temporary effects on the water environment. This includes good practice methods which would also focus on managing the risk of pollution to surface waters and the groundwater environment. It is therefore considered that the Scheme is compliant with this policy.
Paragraph 5.16.7	 The ES should in particular describe: the existing quality of waters affected by the proposed project and the impacts of the proposed project on water quality, noting any relevant existing discharges, proposed new discharges and proposed changes to discharges existing water resources affected by the proposed project and the impacts of the proposed project on water resources, noting any relevant existing abstraction rates, proposed new abstraction rates and proposed changes to abstraction rates (including any impact on or use of mains supplies and reference to Catchment Abstraction Management Strategies) and also demonstrate how proposals minimise the use of water resources and water consumption in the first instance existing physical characteristics of the water environment (including quantity and dynamics of flow) affected by the proposed project and any impact of physical modifications to these characteristics 	Section 10.5 of Chapter 10: Hydrology, Flood Risk and Drainage of the ES [APP-048] sets out the baseline conditions of water receptors for all the sites and surrounding areas with regards to water quality, including the existing quality and physical characteristics of waters nearby and potentially affected by the Scheme.



- any impacts of the proposed project on water bodies or protected areas (including shellfish protected areas) under the Water Environment (Water Framework Directive) (England and Wales) Regulations 2017 and source protection zones (SPZs) around potable groundwater abstractions
- how climate change could impact any of the above in the future
- any cumulative effects



1.5 Table 5: National Policy Statement EN-3 (November 2023)

1.5.1 Table 5 considers the Scheme in the context of policy in NPS EN-3 (November 2023) where that policy differs from policy set out in NPS EN-3 (2011). Where the policy set out by NPS EN-3 (November 2023) is to the same or similar effect as policy in NPS EN-3 (2011), it is not included in this table.

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
Paragraph 2.4.11	Solar photovoltaic PV sites may also be proposed in low lying exposed sites. For these proposals, applicants should consider, in particular, how plant will be resilient to: • increased risk of flooding; and • impact of higher temperatures	As outlined in Section 7.8 of Chapter 7: Climate Change of the ES [REP1-012], account of the effects of climate change have been taken in the design of the Scheme, and its construction and decommissioning. This includes: The effect of projected temperature increases on electrical equipment over the course of the Scheme's design life has been taken into account. Inverters (PV and BESS) will have a cooling system installed to control the temperature and allow the inverters to operate efficiently in warmer conditions. The PV modules and transformers have a wide range of acceptable operating temperatures, and it has been determined that increasing temperatures will not adversely affect their operation. Any health and safety plans developed for construction and decommissioning activities will be required to



		account for potential climate change impacts on workers, such as flooding and heatwaves. The design of drainage systems will ensure that there will be no significant increases in flood risk downstream during storms up to and including the 1 in 100 (1%) annual probability design flood, with an allowance of 20% for climate change. A Decommissioning Environmental Management Plan (DEMP) (taking account of climate change risks at the time) will be prepared prior to decommissioning. An Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] is provided as part of the Application which secures the production of the DEMP.
Section 2.5 Consideration of good design for	Section 4.6 of EN-1 sets out the criteria for good design that should be applied to all energy infrastructure. 2.5.2 Proposals for renewable energy infrastructure should demonstrate good design, particularly in respect of landscape and visual amenity, opportunities for co-	Good design has been a key consideration from the outset. The LVIA has informed the iterative design process, including taking account of published landscape character assessment guidance and fieldwork analysis.
energy infrastructure	existence/co-location with other marine and terrestrial uses, and in the design of the project to mitigate impacts such as noise and effects on ecology and heritage.	The overall objective of the landscape design is to integrate the Scheme into its landscape setting and avoid or minimise adverse landscape and visual effects as far as practicable. The design has been developed in collaboration with the wider design team, other specialists and the Host Authorities landscape advisors to achieve a solution that achieves this objective whilst maximising opportunities to deliver net gains in biodiversity gain. Accordingly, the landscape design aims to achieve the following:



		To integrate the Scheme into the existing landscape pattern as far as possible by retaining and following existing features, including vegetation, where practicable.
		To replace vegetation lost because of construction of the Scheme through areas of new planting.
		To filter and screen more prominent components of the Scheme in views from visual receptors.
		Details of the landscape measures embedded into the Scheme design, including a summary of their environmental functions, is presented in the Outline LEMP [EN010132/EX6/WB7.3_E].REP6-025].
Section 2.6 Flexibility in the project	2.6.1 Where details are still to be finalised applicants should explain in the application which elements of the proposal have yet to be finalised, and the reason why this is the case.	The Applicant is seeking flexibility in the Scheme as set out in Section 4.3 in Environmental Statement Chapter 4 [APP-042]. The flexibility is to address uncertainties in the Scheme
details	2.6.2 Where flexibility is sought in the consent as a result, applicants should, to the best of their knowledge, assess the likely worst-case environmental, social and	design and to allow for the most up to date technology possible to be utilised at the time of construction.
	economic effects of the proposed development to ensure that the impacts of the project as it may be constructed have been properly assessed.10	The flexibility is sought by using the 'Rochdale Envelope' approach which allows assessment of the maximum
	2.6.3 Full guidance on how applicants and the Secretary of State should manage flexibility is set out in Section 4.3 of EN-1	parameters for the Scheme, while ensuring all potentially significant effects (positive or adverse) are considered. The maximum design scenarios are identified from the range of potential options for each design parameter for the Scheme.
		The maximum design scenario assessed is therefore the scenario which would give rise to the greatest potential impact. The maximum design scenarios are set out in the



		Concept Design Parameters and Principles [REP5-094] which is secured by a Requirement in the draft DCO.
Paragraph 2.10.9	The government has committed to sustained growth in solar capacity to ensure that we are on a pathway that allows us to meet net zero emissions by 2050. As such solar is a key part of the government's strategy for low-cost decarbonisation of the energy sector.	As explained in the Statement of Need [APP-320], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to develop a secure, affordable and low carbon electricity generation system which is sufficient to meet future demand. The government expects solar technology to play a
2.10.13	Solar farms are one of the most established renewable electricity technologies in the UK and the cheapest form of electricity generation.	major role in delivery of these objectives.
2.10.14	Solar farms can be built quickly and, coupled with consistent reductions in the cost of materials and improvements in the efficiency of panels, large-scale solar is now viable in some cases to deploy subsidy-free.	This paragraph further emphasises that large scale solar development, in particular, is needed to meet the government's objectives. The Scheme directly accords with this.
Paragraph 2.10.19	Irradiance will be a key consideration for the applicant in identifying a potential site as the amount of electricity generated on site is directly affected by irradiance levels. Irradiance of a site will in turn be affected by surrounding topography, with an uncovered or exposed site of good elevation and favourable south-facing aspect more likely to increase year-round irradiance levels. This in turn affects the carbon emission savings and the commercial viability of the site.	As shown in Figure 7.4 of the Statement of Need [APP-320], the Scheme is proposed to be located in one of the higher solar irradiation areas of the UK. This increases the benefit it will bring to the UK, in relation to the bulk generation of low-carbon electricity per MW installed. The Statement of Need [APP-320] also concludes that the site is of a size and has topography which meets the requirements of the Scheme to generate significant amounts of electricity and store it.
Paragraph 2.10.20	In order to maximise irradiance, applicants may choose a site and design its layout with variable and diverse panel aspects, and panel arrays may also follow the movement of the sun in order to further maximise the solar resource.	The Scheme, as described in ES Chapter 4: Scheme Description [APP-042], is likely to utilise tracker solar panels although as an optionality, fixed panels are included within the Application.
Paragraph		The Scheme proposes to connect to an existing National Grid substation (West Burton Power Station) which is located on



2.10.25	To maximise existing grid infrastructure, minimise disruption to existing local community infrastructure or biodiversity and reduce overall costs applicants may choose a site based on nearby available grid export capacity. Where this is the case, applicants should consider the cumulative impacts of situating a solar farm in proximity to other energy generating stations and infrastructure.	one of the major connections. As explained at Section 7.5 of the Statement of Need [APP-320], by connecting at West Burton Power Station, the Scheme is making use of an existing connection point and existing transmission infrastructure in a way which does not present the risk of overload or congestion on the NETS during any period of foreseen operation, and provides a regional source of locally generated bulk low carbon supplies of electricity to consumers in Nottinghamshire and the wider midlands area. Section 9.3 of the Statement of Need [APP-320] discusses this point further and provides additional evidence which underpins West Burton Power Station National Grid Substation as an excellent point of connection for The Scheme. The site Selection Assessment [AS-004] sets out the detailed site selection process undertaken by the Applicant.
Paragraph 2.10.28 2.10.29	Solar is a highly flexible technology and as such can be deployed on a wide variety of land types. While land type should not be a predominating factor in determining the suitability	The majority of the Order Limits comprises Grade 3b agricultural land. 26.24% BMV land is included within the Order Limits. This is justified by other sustainability considerations, as explained in Section 6.7 of this Planning Statement [EN010132/EX6EX7/WB7.5_CD].
2.10.29	of the site location applicants should, where possible, utilise previously developed land, brownfield land, contaminated land, and industrial land. Where the proposed use of any agricultural land has been shown to be necessary, poorer quality land should be preferred to higher quality land (avoiding the use of "Best and Most Versatile" agricultural land where possible. Best and Most Versatile agricultural land is defined as land in grades 1, 2 and 3a of the Agricultural Land Classification.	As stipulated by this policy, land type should not be a predominating factor in determining the suitability of the site location.
Paragraph 2.10.33	The Agricultural Land Classification (ALC) is the only approved system for grading agricultural quality in England and Wales and, if necessary, field surveys should be used to establish the ALC grades in accordance with the current, or any successor to	The ALC survey is provided in Appendix 19.1 of the ES [APP-137]. Some of the Grid Connection Route is outside of the ALC survey area, however, some of this is not agricultural land,



	it, grading criteria and identify the soil types to inform soil management at the construction, operation and decommissioning phases in line with Defra Construction Code.	and the installation of the grid connection cable will be short term and will not be an impediment to the reestablishment of its existing agricultural use following laying of the grid connection cable.
Paragraph 2.10.30 2.10.31	Whilst the development of ground mounted solar arrays is not prohibited on Best and Most Versatile agricultural land, or sites designated for their natural beauty, or recognised for ecological or archaeological importance, the impacts of such are expected to be considered and are discussed under paragraphs 2.10.66 – 2.10.83 and 2.10.98 – 2.10.100. It is recognised that at this scale, it is likely that applicants' developments may use some agricultural land, however applicants should explain their choice of site, noting the preference for development to be on suitable brownfield, industrial and low and medium grade agricultural land.	The majority of the Order Limits comprises Grade 3b agricultural land. 26.24% BMV land is included within the Order Limits. This is justified by other sustainability considerations, as explained in Section 6.7 of this Planning Statement [EN010132/EX6EX7/WB7.5_CD]. The Site Selection Assessment [AS-004] explains the choice of site as required by this paragraph.
Paragraph 2.10.35	Applicants will need to consider the suitability of the access routes to the proposed site for both the construction and operation of the solar farm with the former likely to raise more issues. Given that potential solar farm sites are largely in rural areas, access for the delivery of solar arrays and associated infrastructure during construction can be a significant consideration for solar farm siting.	A Transport Assessment, Appendix 14.1 of the ES [REP4-036] has been prepared to assess the suitability and impact of the necessary access to the Order limits. This concludes that the Scheme with respect to transport and access is considered to be in accordance with relevant national and local policy and that it avoids any adverse impacts on highway safety or any severe residual cumulative impacts on the road network.
Paragraph 2.10.37	Developers will usually need to construct on-site access routes for operation and maintenance activities, such as footpaths, earthworks, or landscaping.	A Transport Assessment, Appendix 14.1 of the ES [REP4-036] has been prepared to assess the suitability and impact of the necessary access to the Order limits. This concludes that the Scheme with respect to transport and access is considered to be in accordance with relevant national and local policy and



2.10.38	In addition, sometimes access routes will need to be constructed to connect solar farms to the public road network.	that it avoids any adverse impacts on highway safety or any severe residual cumulative impacts on the road network.
2.10.39	Applications should include the full extent of the access routes for operation and maintenance and their effects.	
Paragraph 2.10.40 2.10.41 2.10.42 2.10.43	Proposed developments may affect the provision of public rights of way networks. Public rights of way may need to be temporarily closed or diverted to enable construction, however, applicants should keep, as far as is practicable and safe, all public rights of way that cross the proposed development site open during construction and to protect users where a public right of way borders or crosses the site. Applicants are encouraged to design the layout and appearance of the site to ensure continued recreational use of public rights of way, where possible during construction, and in particular during operation of the site. Applicants are encouraged where possible to minimise the visual impacts of the development for those using existing public rights of way, considering the impact this may have on any other visual amenities in the surrounding landscape. Applicants should consider and maximise opportunities to facilitate enhancements to the public rights of way and the inclusion, through site layout and design of access, of new opportunities for the public to access and cross proposed solar development sites (whether via the adoption of new public rights of way or the creation of permissive paths) taking into account where appropriate the views of	There are several PRoW within or abutting the Scheme. These are shown in Appendix 14.3 of the ES [REP5-018]. These PRoW are predominantly used for recreational purposes and form part of a wide network of PRoW in the surrounding area providing residents with alternative routes. As detailed in the Public Rights of Way Management Plan provided as Appendix 14.3 to the ES [REP5-018], PRoWs will be kept open throughout all phases of the Scheme, with appropriate safety measures in place. Where diversions cannot be avoided during the construction period, these would be convenient, clearly signed and for as short a duration as is necessary, usually overnight. The creation of a new permissive footpath to run from the track off Sykes Lane along the Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub.
Paragraph 2.10.45	Applicants should set out detail on how public rights of way would be managed to ensure they are safe to use is set out in an outline Public Rights of Way Management Plan.	A Public Rights of Way Management Plan is provided as Appendix 14.3 to the ES [REP5-018].



Paragraph 2.10.63	It is likely that underground and overhead cabling will be required to connect the electrical assets of the site, such as from the substation to the panel arrays or storage facilities. In the case of underground cabling, applicants are expected to provide a method	Details of cables, cable trenches and construction methodology are provided in Chapter 4, Scheme Description, of the ES [APP-042] and a section of a cable trench is shown by Figure 4.2 of the ES: Scheme Description [APP-042].
2.10.64	statement describing cable trench design, installation methodology, as well as details of the operation and maintenance regime.	
Paragraph 2.10.46	Security of the site is a key consideration for developers. Applicants may wish to consider not only the availability of natural defences such as steep gradients, hedging and rivers, but also perimeter security measures such as fencing, electronic security, CCTV and lighting, with the measures proposed on a site-specific basis.	Security measures, including fencing and CCTV are described Chapter 4, Scheme Description, of the ES [APP-042] and are taken into account in the assessment presented in the ES [APP-039 to APP-061].
2.10.47	Applicants should assess the visual impact of these security measures, as well as the impacts on local residents, including for example issues relating to intrusion from CCTV and light pollution in the vicinity of the site.	
Paragraph 2.10.69	Applicants should set out what would be decommissioned and removed from the site at the end of the operational life of the generating station, considering instances where it may be less harmful for the ecology of the site to keep or retain certain types of infrastructure, for example underground cabling, and where there may be socio-economic benefits in retaining site infrastructure after the operational life, such as retaining pathways through the site or a site substation.	Details of the decommissioning phase, including which elements will be decommissioned and which will be retained are provided in Chapter 4, Scheme Description, of the ES [APP-042]. This sets out that the Solar PV Array Works Area and related components, Ancillary Infrastructure, West Burton Substation and the BESS Compound will be removed and recycled or disposed of in accordance with good practice and market conditions at that time. The underground cable within the Grid Connection Route would be removed to a depth of 1m, otherwise would remain in situ. The West Burton Power Substation would remain operational.
Paragraph 2.10.118	As the significance of a heritage asset derives not only from its physical presence but also from its setting, careful consideration should be given to the impact of large-	[APP-117 to APP-119] assess the significance of the Scheduled Monument and detail the contribution made by its setting, namely the land which was formerly located within



scale solar farms which depending on their scale, design, and prominence, may cause substantial harm to the significance of the asset.

the deer park and is not scheduled. Table 5.1 of the Statement of Common Ground (SOCG) with Historic England [EN010132/EX6/WB8.3.3_AREP6-042] provides additional detail for how the Applicant's conclusions in identifying how the significance of the monument have been derived, and the potential for harm caused by the Scheme to that significance.

The Scheme would not cause any direct physical harm to the significance of the Scheduled Monument as there is no proposed intervention to the fabric of any of the sections of the Scheduled Monument that would result in its permanent loss either wholly or in part. Any harm would be only that caused to the significance of the monument that is derived from its setting. This would occur through the placement of panels within land that was formerly occupied by the medieval deer park.

As detailed in Historic England Advice Note 123 pages 5, 15 and 16, a key element of a heritage assessment is to identify how the significance of a heritage asset is derived.

"The context for any analysis of the significance of a heritage asset will be a thorough familiarity with the asset itself, developed through site visits, and appropriate inspection of the fabric, its features, materials and ornament, and also its setting if needed." (Historic England Advice Note 12 page 5)

168 | Page

³ Historic England (2019) *Statements of Heritage Significance: Analysing Significance in Heritage Assets: Historic England Advice Note 12.* (Online, last accessed 27.03.2024) https://historicengland.org.uk/images-books/publications/statements-heritage-significance-advice-note-12/heag279-statements-heritage-significance/



	This is reiterated on Page 4 of Historic England Planning Note 3 (Second Edition):
	"Setting is not itself a heritage asset, nor a heritage designation, although land comprising a setting may itself be designated (see below Designed settings). Its importance lies in what it contributes to the significance of the heritage asset or to the ability to appreciate that significance."
	As evidenced at ISH5 and in the SoCG with Historic England [EN010132/EX6/WB8.3.3_AREP6-042], the Applicant understands the Scheduled Monument derives its significance from its historic interest as the sole surviving element of a former enclosed medieval space, which is largely understood through desk-based research, particularly aerial imagery and historical documentation. Section 2 of the Cultural Heritage Position Statement [REP5-027] confirms how this view has been formed, based on the Reasons for Designation set out in Official List Entry for the Scheduled Monument. The agrarian landscape, the former MOD petroleum site and the railway, which bisects the Scheduled Monument, have a detrimental effect on the ability to appreciate any remaining elements of the former medieval landscape and are consequently considered to have a detrimental effect on the overall contribution made by setting to the significance of the Scheduled Monument.
Paragraph	The Scheme will be decommissioned at the end of its operational life in accordance with a decommissioning



2.10.147	Where the consent for a solar farm is to be time-limited, the DCO should impose a requirement setting that time-limit from the date the solar farm starts to generate electricity.	environmental management plan, as secured by requirements of the Draft DCO [EN010132/EX6EX7/WB3.1_GH]. The Applicant has
2.10.148	Such a requirement should also secure the decommissioning of the generating station after the expiration of its permitted operation to ensure that inoperative plant is removed after its operational life.	committed to decommissioning the Scheme no later than 60 years from the date of final commissioning.
2.10.149	An upper limit of 40 years is typical, although applicants may seek consent without a time period or for differing time-periods for operation.	
Paragraph		The Scheme will be decommissioned at the end of its
2.10.150	The time limited nature of the solar farm, where a time-limit is sought as a condition of consent, is likely to be an important consideration for the Secretary of State.	operational life in accordance with a decommissioning environmental management plan, as secured by
2.10.151	The Secretary of State should consider the period of time the applicant is seeking to operate the generating station as well as the extent to which the site will return to its original state when assessing impacts such as landscape and visual effects and potential effects on the settings of heritage assets and nationally designated	requirements of the Draft DCO [EN010132/EX6EX7/WB3.1_GH]. The Applicant has committed to decommissioning the Scheme no later than 60 years from the date of final commissioning.
	landscapes.	The assessments in the ES [APP-039 to APP-061] have taken account of this. Chapter 4: Scheme Description, of the ES [APP-042] describes how the Order limits would be left on completion of decommissioning.
Paragraph	In many cases, not all aspects of the proposal may have been settled in precise detail	Chapter 2: EIA Process and Methodology [APP-040] and
2.10.70	at the point of application. Such aspects may include:	Chapter 4: Scheme Description of the ES [APP-042] explain that the parameters for the project are defined by the Outline
	the type, number and dimensions of the panels;	Design Principles, which have informed the assessments in
	layout and spacing;	the ES [APP-039 to APP-061].
	the type of inverter or transformer; and	The Works Plans [REP5-035] and Design Principles of the Design and Access Statement [APP-314 to APP-315] define parameters for the Scheme. The approach to flexibility is



2.10.71	whether storage will be installed (with the option to install further panels as a substitute).	explained in Chapter 4, Scheme Description, of the ES [APP-042]].
	Applicants should set out a range of options based on different panel numbers, types and layout, with and without storage.	
Paragraph		Section 9.5 of Chapter 9: Ecology and Biodiversity of the ES
2.10.76	The applicant's ecological assessments should identify any ecological risk from developing on the proposed site.	[APP-047] sets out all the protected species, habitats and other species identified as being of principal importance for
2.10.77	Issues that need assessment may include habitats, ground nesting birds, wintering and migratory birds, bats, dormice, reptiles, great crested newts, water voles and badgers.	the conservation of biodiversity within the study area for the Scheme. Sections 9.7 and 9.9 of Chapter 9: Ecology and Biodiversity of
2.10.78	The applicant should use an advising ecologist during the design process to ensure that adverse impacts are avoided, minimised or mitigated in line with the mitigation hierarchy and biodiversity enhancements are maximised.	the ES [APP-047] clearly set out the expected effects on the above receptors during the construction, operation and decommissioning phases of the Scheme. This concludes that there are anticipated to be no significant adverse effects on
2.10.79	The assessment may be informed by a 'desk study' of existing ecological records, an evaluation of the likely impacts of the solar farm upon ecological features and should specify mitigation to avoid or minimise these impacts, and any further surveys required.	any of these protected species as a result of the Scheme.
Paragraph		Earthworks required for the Scheme are described in Chapter
2.10.80	Applicants should consider earthworks associated with construction compounds, access roads and cable trenching.	4, Scheme Description, of the ES [APP-042] and are taken into account by the assessments in the ES [APP-039 to APP-061].
2.10.81	Where soil stripping occurs topsoil and subsoil should be stripped, stored, and replaced separately to minimise soil damage and to provide optimal conditions for site restoration. Further details on minimising impacts on soil and soil handling	There are minimal earthworks identified for the Scheme. The Cable Route Corridor will require the redistribution and management of soil.
	above at paragraphs 2.10.18 and 2.10.19.	A detailed soil resource management plan will be prepared prior to the commencement of construction, prior to operation, and prior to decommissioning, as set out by the



		Requirements of the draft DCO [EN010132/EX6EX7/WB3.1_GH].
Paragraph 2.10.82	Applicants should consider how security and lighting installations may impact on the local ecology. Where pole mounted CCTV facilities are proposed the location of these facilities should be carefully considered in order to minimise impact. If lighting is necessary, it should be minimised and directed away from areas of likely habitat.	Lighting and CCTV required for the Scheme are described in Chapter 4, Scheme Description, of the ES [APP-042] and are taken into account by the assessments in the ES [APP-039 to APP-061].
Paragraph 2.10.83	Applicants should consider how site boundaries are managed. If any hedges/scrub are to be removed, further surveys may be necessary to account for impacts. Buffer strips between perimeter fencing and hedges may be proposed, and the construction and design of any fencing should account for enabling mammal, reptile and other fauna access into the site if required to do so in the ecological report.	The ES [APP-039 to APP-061] takes account of all works to boundaries, and any works to hedgerows. Buffers to woodland and hedgerow are included, and proposals for fencing incorporate features to enable the movement of mammals, reptiles and other fauna.
Paragraph		An FRA is included in Appendix 10.1 of the Environmental Statement [APP-089].
2.10.84	Where a Flood Risk Assessment has been carried out this must be submitted alongside the applicant's ES. This will need to consider the impact of drainage. As solar PV panels will drain to the existing ground, the impact will not in general be significant.	The solar panels will be mounted on raised frames above surrounding ground level allowing flood water to flow freely underneath. Therefore, there will be no loss of floodplain volume as a result of the proposed development. The proposed development is free draining through perimeter gaps around all panels, allowing for infiltration as
2.10.85	Where access tracks need to be provided, permeable tracks should be used, and localised Sustainable Drainage Systems (SuDS), such as swales and infiltration trenches, should be used to control any run-off where recommended.	
2.10.86	Given the temporary nature of solar PV farms, sites should be configured or selected to avoid the need to impact on existing drainage systems and watercourses.	existing within the grassland/vegetation surrounding and beneath the panels. There will be minimal increase in
	impermeable area meaning the proposals will not increase surface water flood risk elsewhere.	
2.10.87	Where culverting for access is unavoidable, applicants should demonstrated that no reasonable alternatives exist and where necessary it will only be in place temporarily for the construction period.	



2.10.88		
Paragraph 2.10.90	For projects in England, applicants should consider enhancement, management, and monitoring of biodiversity in line with the ambition as set out in the Environmental Improvement Plan and any relevant measures and targets, including statutory	The Scheme has taken advantage of opportunities to conserve and enhance biodiversity and accords with this paragraph.
	targets set under the Environment Act or elsewhere.	A Biodiversity Net Gain (BNG) assessment, using Defra's Metric 3.0, has been provided with the DCO application [APP-088]. For the purposes of BNG, the Scheme will result in an overall significant net gain of 86.80% provided in habitat, 54.71% gains in hedgerow and 33.25% gains in river units.
Paragraph 2.10.92	Applicants should consider whether they need to provide geotechnical and hydrological information (such as identifying the presence of peat at each site) including the risk of landslide connected to any development work.	The Applicant does not consider that the nature of the Order limits or the Scheme is such that this information is required.
Paragraph		A Biodiversity Net Gain (BNG) assessment, using Defra's
2.10.128	In England, proposed enhancements should take account of the above factors and as set out in Sections 4.6 and 5.5 of EN-1 aim to achieve environmental and biodiversity net gain in line with the ambition set out in the Environmental Improvement Plan and any relevant measures and targets, including statutory	Metric 3.0, has been provided with the DCO application [APP-088]. For the purposes of BNG, the Scheme will result in an overall significant net gain of 86.80% provided in habitat, 54.71% gains in hedgerow and 33.25% gains in river units.
	targets set under the Environment Act or elsewhere.	The Scheme has taken advantage of opportunities to
2.10.129	This might include maintaining or extending existing habitats and potentially creating new important habitats, for example by installing: cultivated strips/plots for rare arable plants, rough grassland margins, bumble bee plant mixes, and wild bird seed mixes.	conserve and enhance biodiversity and accords with this policy.
2.10.130	Applicants are advised to develop an ecological monitoring programme to monitor impacts upon the flora of the site and upon any particular ecological receptors (such as bats and wintering birds). Results of the monitoring will then inform any changes	



	needed to the land management of the site, including, if appropriate, any livestock grazing regime.	
Paragraph 2.10.154 2.10.155	Water management is a critical component of site design for ground mount solar plants. Where previous management of the site has involved intensive agricultural practice, solar sites can deliver significant ecosystem services value in the form of drainage, flood attenuation, natural wetland habitat, and water quality management. The Secretary of State will consider the worst-case effects in its consideration of the application and consent.	Appendix 10.1 of the ES [APP-089] sets out how water and drainage will be managed as part of the Scheme. The Scheme provides 33.25% uplift in water river units and is reflective of drainage, flood attenuation, natural wetland habitat, and water quality management enhancements.
Paragraph 2.10.97	Applicants should carry out a landscape and visual assessment and report it in the ES. Visualisations may be required to demonstrate the effects of a proposed solar farm on the setting of heritage assets and any nearby residential areas or viewpoints.	An assessment of the potential landscape and visual impacts associated with the construction, operation and decommissioning of the Scheme has been carried out and is presented in Chapter 8: Landscape and Visual Impact of the ES [APP-046] Visualisations have been produced for both summer and winter photography and visualisations have been produced for Winter views at year 1 of operation to represent a worst-case scenario and summer at year 15 post construction to represent the effects of mature mitigation. The visualisations are verifiable and provide a variety of representative views where significant effects are considered likely. Accurate Visual Representations (AVR's) have been produced at AVR level 1 and 3. AVR Level 1shows the location, size and the degree of visibility of the proposals alongside a verifiable photograph with the Scheme represented by a wireframe. Level 3 visualisations show the same as level 1 AVR's but include the



		visualisation produced is provided in the LVIA Chapter 8 and
Paragraph		figure sheets are shown in Appendix 8.3 [APP-074] of the ES. Good design has been a key consideration from the outset.
2.10.98	Applicants should follow the criteria for good design set out in Section 4.6 of EN-1 when developing projects and will be expected to direct considerable effort towards minimising the landscape and visual impact of solar PV arrays especially within	The LVIA has informed the iterative design process, including taking account of published landscape character assessment guidance and fieldwork analysis.
	nationally designated landscapes	The overall objective of the landscape design is to integrate
2.10.99	Whilst there is an acknowledged need to ensure solar PV installations are adequately secured, required security measures such as fencing should consider the need to minimise the impact on the landscape and visual impact (see paragraphs 2.10.31 – 2.10.33 above).	the Scheme into its landscape setting and avoid or minimise adverse landscape and visual effects as far as practicable. The design has been developed in collaboration with the wider design team, other specialists and the Host Authorities landscape advisors to achieve a solution that achieves this objective whilst maximising opportunities to deliver net gains in biodiversity gain. Accordingly, the landscape design aims to achieve the following:
		To integrate the Scheme into the existing landscape pattern as far as possible by retaining and following existing features, including vegetation, where practicable.
		To replace vegetation lost because of construction of the Scheme through areas of new planting.
		To filter and screen more prominent components of the Scheme in views from visual receptors.
		Details of the landscape measures embedded into the Scheme design, including a summary of their environmental functions, is presented in the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025].



		Refer also to the LVIA Chapter 8 and Landscape and Ecology Mitigation & Enhancement Measures which are shown in Figures 8.15 – 8-17 of the ES.
Paragraph 2.10.100 2.10.101	The applicant should consider as part of the design, layout, construction and future maintenance plans how to protect and retain, wherever possible, the growth of vegetation on site boundaries, as well as the growth of existing hedges, established vegetation, including mature trees within boundaries. The impact of the proposed development on established trees and hedges should be	Refer to the LVIA Chapter 8 and Landscape and Ecology Mitigation and Enhancement Measures which are shown in Figures 8.18.1 [REP1-026] to Figure 8.18.3 [REP1-030] of the ES. The Landscape and Ecology Mitigation and Enhancement Measures illustrate the use of extensive landscape mitigation
	informed by a tree survey and arboricultural/ hedge assessment as appropriate.	to screen the Scheme from sensitive views. Site fencing has been proposed in proximity to existing hedgerows to allow the hedgerows to grow into the fencing to screen it where possible. This approach is secured through the Outline Landscape and Ecological Management Plan (LEMP) [EN010132/EX6/WB7.3_EREP6-025] with the management of existing and proposed hedgerows prescribed in this document.
Paragraph	Applicants should consider the potential to mitigate landscape and visual impacts	Refer to the LVIA Chapter 8 and Landscape and Ecology
2.10.131	through, for example, screening with native hedges, trees and woodlands.	Mitigation and Enhancement Measures which are shown in Figures 8.18.1 [REP1-026] to Figure 8.18.3 [REP1-030] of the
2.10.132	Applicants should aim to minimise the use and height of security fencing. Where possible applicants should utilise existing features, such as hedges or landscaping to assist in site security or screen security fencing.	ES.
2.10.132	Applicants should minimise the use of security lighting. Any lighting should utilise a	The Landscape Mitigation Plans illustrate the use of extensive landscape mitigation to screen the Scheme from sensitive
2.10.133	passive infra-red (PIR) technology and should be designed and installed in a manner which minimises impact.	views. Site fencing has been proposed in proximity to existing hedgerows to allow the hedgerows to grow into the fencing to screen it where possible. This approach is secured through the Outline Landscape and Ecological Management Plan (LEMP) [EN010132/EX6/WB7.3_EREP6-025] with the



		management of existing and proposed hedgerows prescribed in this document.
		Details of operational lighting are set out by Chapter 4, Scheme Description, of the ES [APP-042]. This explains that no part of the Scheme will be continuously lit. Manually operated, and motion-detection lighting will be utilised for operational and security purposes around electrical infrastructure. Lighting will be directed downward and away from boundaries. No visible lighting will be utilised at the site perimeter fence, aside from the site entrance points.
Paragraph 2.10.157	The Secretary of State will consider the landscape and visual impact of any proposed solar PV farm, taking account of any sensitive visual receptors, and the effect of the development on landscape character, together with the possible cumulative effect with any existing or proposed development. Nationally designated landscapes (National Parks, The Broads and Areas of Outstanding Beauty) are afforded extra protection due to their statutory purpose. Development in these areas needs to satisfy policy as set out in EN-1 Section 5.10.	The Scheme complies with this requirement through the provision of an LVIA chapter within the PEIR and ES. The impacts on landscape and visual amenity have influenced the iterative design of the Scheme. The proposals have considered the need to mitigate landscape and visual impacts. Details of the identified mitigation required are included within the LVIA Chapter 8 [APP-046] - see Sections 8.6 and 8.8, and the Outline Landscape and Ecological Management Plan [EN010132/EX6/WB7.3_E].REP6-025].
Paragraph		A glint and glare assessment has been undertaken for the
2.10.103	Applicants should map receptors to qualitatively identify potential glint and glare issues and determine if a glint and glare assessment is necessary as part of the	Scheme and is presented in Appendix 16.1 of the ES [APP-132].
	application.	The glint and glare assessment concludes that with the
2.10.105	The extent of reflectivity analysis required to assess potential impacts will depend on the specific project site and design. This may need to account for 'tracking' panels if they are proposed as these may cause differential diurnal and/or seasonal impacts.	inclusion of mitigation in the form of hedgerow planting and maintenance in the locations outlined, there are acceptable levels of adverse impact which are predicted for one dwelling at West Burton 3 if a fixed mounting system is implemented



2.10.106	When a glint and glare assessment is undertaken, the potential for solar PV panels, frames and supports to have a combined reflective quality may need to be assessed, although the glint and glare of the frames and supports is likely to be significantly less than the panels.	and two dwellings at West Burton 3 if a tracking mounting system is implemented.
Paragraph 2.10.134	Applicants should consider using, and in some cases the Secretary of State may require, solar panels s to comprise of (or be covered) with anti-glare/non-reflective coating with a specified angle of maximum reflection attenuation for the lifetime of the permission.	Chapter 4, Scheme Description, of the ES [APP-042]. sets out that the solar PV panels will consist of a series of photovoltaic cells beneath a layer of toughened and anti-reflective glass.
Paragraph 2.10.158	Solar PV panels are designed to absorb, not reflect, irradiation. However, the Secretary of State should assess the potential impact of glint and glare on nearby homes, motorists, public rights of way, and aviation infrastructure (including aircraft departure and arrival flight paths).	A glint and glare assessment has been undertaken for the Scheme and is presented in Appendix 16.1 of the ES [APP-132].
		The glint and glare assessment concludes that with the inclusion of mitigation in the form of hedgerow planting and maintenance in the locations outlined, there are acceptable levels of adverse impact which are predicted for one dwelling at West Burton 3 if a fixed mounting system is implemented and two dwellings at West Burton 3 if a tracking mounting system is implemented.
Paragraph 2.10.159	Whilst there is some evidence that glint and glare from solar farms can be experienced by pilots and air traffic controllers in certain conditions, there is no evidence that glint and glare from solar farms results in significant impairment on aircraft safety. Therefore, unless a significant impairment can be demonstrated, the Secretary of State is unlikely to give any more than limited weight to claims of aviation interference because of glint and glare from solar farms.	As stated in the glint and glare appendix [APP-132], impacts on aviation were assessed in detail by the glint and glare assessment. Consistent with the statement in this policy, this concluded that there would be no impacts on aviation receptors.
Paragraph		Heritage assets as defined in this policy have been considered
2.10.107	The impacts of solar PV developments on the historic environment will require expert assessment in most cases and may have effect both above and below ground.	and where relevant assessed in Chapter 13: Cultural Heritage of the ES [APP-051]. Section 13.5 of Chapter 13: Cultural



	Above ground impacts may include the effects on the setting of Listed Buildings and other designated heritage assets as well as on Historic Landscape Character.	Heritage of the ES [APP-051] describes the significance of these assets.
2.10.108	Below ground impacts, although generally limited, may include direct impacts on archaeological deposits through ground disturbance associated with trenching, cabling, foundations, fencing, temporary haul routes etc.	Archaeological evaluations were undertaken to in addition to a desk-based assessment, including a geophysical survey (detailed magnetometry) of the whole scheme and targeted
2.10.109	Equally solar PV developments may have a positive effect, for example	trial trenching.
2.10.110	archaeological assets may be protected by a solar PV farm as the site is removed from regular ploughing and shoes or low-level piling is stipulated.	The ES [APP-039 to APP-061] has therefore identified a suitable baseline from which to assess the Scheme in relation to this policy.
Paragraph		The assessment set out in Chapter 13: Cultural Heritage of the
2.10.112	Applicant assessments should be informed by information from Historic Environment Record (HERs) or the local authority.	ES [APP-051] has been informed by the HER.
2.10.113	Where a site on which development is proposed includes or has the potential to include heritage assets with archaeological interest, the applicant should submit an appropriate desk-based assessment and, where necessary, a field evaluation. These are expected to be carried out, using expertise where necessary and in consultation with the local planning authority, and should identify archaeological study areas and propose appropriate schemes of investigation, and design measures, to ensure the protection of relevant heritage assets.	
Paragraph		Archaeological evaluations were undertaken to in addition to
2.10.114	In some instances, field studies may include investigative work (and may include trial trenching beyond the boundary of the proposed site) to assess the impacts of any ground disturbance, such as proposed cabling, substation foundations or mounting supports for solar panels on archaeological assets.	a desk-based assessment, including a geophysical survey (detailed magnetometry) of the whole scheme and targeted trial trenching. The scope and specification of each field investigation have been set out in Written Scheme of
	The extent of investigative work should be proportionate to the sensitivity of, and extent of proposed ground disturbance in, the associated study area	Investigations (WSI) [REP5-016].



2.10.115		The results of these surveys (Appendix 13.1 and Appendix 13.2 of the ES [APP-105 to APP-114]) have been incorporated in Section 13.6 of Chapter 13: Cultural Heritage of the ES [APP-051].
Paragraph	Applicants should take account of the results of historic environment assessments in their design, proposal.	Section 13.8 of Chapter 13: Cultural Heritage of the ES [APP-051] outlines the mitigation measures embedded within the Scheme design pertaining to cultural heritage. This includes the provision of stand-offs between the Scheme and heritage assets in order to help to preserve their setting during the construction, operational and decommissioning periods.
2.10.116	Applicants should consider what steps can be taken to ensure heritage assets are conserved in a manner appropriate to their significance, including the impact of proposals on views important to their setting.	
2.10.118	As the significance of a heritage asset derives not only from its physical presence, but also from its setting, careful consideration should be given to the impact of large-scale solar farms which depending on their scale, design and prominence, may cause substantial harm to the significance of the asset.	Appropriate and sensitive screening has also been developed and implemented to minimise the visual intrusion of the Scheme, while avoiding obscuring or intruding upon key views and relationships between heritage assets.
	Applicants may need to include visualisations to demonstrate the effects of a proposed solar farm on the setting of heritage assets.	
2.10.119		
Paragraph 2.10.137	The ability of the applicants to microsite specific elements of the proposed development during the construction phase should be an important consideration by the Secretary of State when assessing the risk of damage to archaeology.	The final layout of the components of the Scheme is required to be within the Works Areas identified by the Works Plans [REP5-035] and within the Design and Access Statement [APP-314 to APP-315]. These enable micrositing. The approach to flexibility is explained in Chapter 4, Scheme Description, of the
2.10.138	Where requested by the applicant, the Secretary of State should consider granting consents which allow for the micro siting within a specified tolerance of elements of the permitted infrastructure so that precise locations can be amended during the	ES [APP-042].



	construction phase if unforeseen circumstances, such as the discovery of previously unknown archaeology, arise.	
Paragraph 2.10.160	Solar farms are generally consented on the basis that they will be time-limited in operation. The Secretary of State should therefore consider the length of time for which consent is sought when considering the impacts of any indirect effect on the historic environment, such as effects on the setting of designated heritage assets.	The design life of the Scheme is approximately 40 years; however, if equipment is still operating successfully and safely, the developer may choose to operate beyond the Scheme's design life. This is a common occurrence for generating stations; many stations operate beyond the design life if they are well maintained. The Applicant has committed to decommission the Scheme no later than 60 years from the date of final commissioning, and it will be decommissioned in accordance with the Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] once it has ceased to operate permanently.
Paragraph		A CTMP is provided in Appendix 14.2 of the ES [REP4-
2.10.120	Modern solar farms are large sites that are mainly comprised of small structures that can be transported separately and constructed on-site, with developers designating a compound on-site for the delivery and assemblage of the necessary components.	038EN010132/EX7/WB6.3.14.2 E]. This sets out the proposals to manage construction traffic and staff vehicles during the construction of the Scheme. It identifies the management of freight traffic i.e., HGVs to and from the designated
2.10.121	Many solar farms will be sited in areas served by a minor road network. Public perception of the construction phase of solar farm will derive mainly from the effects of traffic movements, which is likely to involve smaller vehicles than typical onshore energy infrastructure but may be more voluminous. It is important that all sections of roads and bridges on the proposed delivery route can accommodate the weight and volume of the loads.	construction compounds, as well as staff vehicles. The CTMP has been informed by extensive consultation with Lincolnshire and Nottinghamshire County Councils Highways and National Highways.
Paragraph		A CTMP is provided in Appendix 14.2 of the ES [REP4-
2.10.123	Applicants should assess the various potential routes to the site for delivery of materials and components where the source of the materials is known at the time of the application and select the route that is the most appropriate.	038EN010132/EX7/WB6.3.14.2_E]. This sets out the proposals to manage construction traffic and staff vehicles during the construction of the Scheme. It identifies the management of freight traffic i.e., HGVs to and from the designated



2.10.124	Where the exact location of the source of construction materials, such as crushed stone or concrete is not be known at the time of the application applicants should assess the worst-case impact of additional vehicles on the likely potential routes.	construction compounds, as well as staff vehicles. The CTMP has been informed by extensive consultation with Lincolnshire and Nottinghamshire County Councils Highways and National Highways.
Paragraph 2.10.125	Applicants should ensure all sections of roads and bridges on the proposed delivery route can accommodate the weight and volume of the loads and width of vehicles. Although unlikely, where modifications to roads and/or bridges are required, these should be identified, and potential effects addressed in the ES.	As stated in the Transport Assessment provided in Appendix 14.1 of the ES [REP4-036], other than to provide the two new access points for the West Burton Substation extension and to accommodate the installation of the Grid Connection Route, there is not expected to be a requirement for any offsite road modifications as a result of construction works. A CTMP is provided in Appendix 14.2 of the ES [REP4-038].EN010132/EX7/WB6.3.14.2 E]. This sets out the proposals to manage construction traffic and staff vehicles during the construction of the Scheme. It identifies the management of freight traffic i.e., HGVs to and from the designated construction compounds, as well as staff vehicles. The CTMP has been informed by extensive consultation with Lincolnshire and Nottinghamshire County Councils Highways and National Highways.
Paragraph 2.10.126	Where a cumulative impact is likely because multiple energy infrastructure developments are proposing to use a common port and/or access route and pass through the same towns and villages, applicants should include a cumulative transport assessment as part of the ES. This should consider the impacts of abnormal traffic movements relating to the project in question in combination with those from any other relevant development. Consultation with the relevant local highways authorities is likely to be necessary.	Cumulative schemes for consideration have been agreed in consultation with NCC, LCC and National Highways and have been considered in the ES. These are detailed in Section 14.9 of Chapter 14: Transport and Access of the ES [APP-052]. Chapter 14: Transport and Access of the ES [APP-052] concludes that no cumulative impacts upon the highway network are envisaged based on the assessment in the ES.



		The cumulative effects are therefore expected to remain negligible.
Paragraph		As stated in the Transport Assessment provided in Appendix
2.10.139	In some cases, the local highway authority may request that the Secretary of State impose controls on the number of vehicle movements to and from the solar farm site in a specified period during its construction and, possibly, on the routeing of such movements particularly by heavy vehicles.	14.1 of the ES [REP4-036], as agreed with LCC, NCC, Highways, construction HGVs will travel to/ from the Solar Farm Site via agreed routes to avoid passing along any Protected Lanes. Local off-site highway improvements (e.g., verge clearance, hedge cutting and/ or carriageway widening) will be carried
2.10.140	Where the Secretary of State agrees that this is necessary, requirements could be imposed on development consent.	out at the required locations to provide the desired 6.0m carriageway width for HGVs along routes where possible. A vehicle routing plan showing the agreed routing strategy for HGVs is contained within the CTMP [REP4-038 EN010132/EX7/WB6.3.14.2 E] of the ES [APP-039 to APP-061].
Paragraph	Where cumulative effects on the local road network or residential amenity are	Cumulative schemes for consideration have been agreed in
2.10.141	predicted from multiple solar farm developments, it may be appropriate for applicants for various projects to work together to ensure that the number of abnormal loads and deliveries are minimised, and the timings of deliveries are managed and coordinated to ensure that disruption to residents and other highway	consultation with NCC, LCC and National Highways and have been considered in the ES. These are detailed in Section 14.9 of Chapter 14: Transport and Access of the ES [APP-052].
	users is reasonably minimised.	Chapter 14: Transport of the ES [APP-052] concludes that no cumulative impacts upon the highway network are envisaged
2.10.142	It may also be appropriate for the highway authority to set limits for and coordinate these deliveries through active management of the delivery schedules through the abnormal load approval process.	based on the assessment in the ES. The cumulative effects are therefore expected to remain negligible.
Paragraph		A CTMP is provided in Appendix 14.2 of the ES [REP4-
2.10.143	Once consent for a scheme has been granted, applicants should liaise with the relevant local highway authority (or other coordinating body) regarding the start of construction and the broad timing of deliveries. Applicants may need to agree a	038 <u>EN010132/EX7/WB6.3.14.2_E</u>]. This sets out the proposals to manage construction traffic and staff vehicles during the construction of the Scheme. It identifies the management of freight traffic i.e., HGVs to and from the designated



2.10.144

planning obligation to secure appropriate measures, including restoration of roads and verges.

Further, it may be appropriate for any non-permanent highway improvements carried out for the development (such as temporary road widening) to be made available for use by other subsequent solar farm developments.

construction compounds, as well as staff vehicles. The CTMP has been informed by extensive consultation with Lincolnshire and Nottinghamshire County Councils Highways and National Highways. It includes proposals for the Scheme's Transport Coordinator to liaise as appropriate with local transport and traffic groups, local planning authorities, local highway authorities and Highways England.



1.6 Table 6: National Planning Policy Framework

1.6.1 Table 6 considers the Scheme in the context of policy in the National Planning Policy Framework (2023). The relevant paragraphs and compliance with policy is considered below.

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
Paragraph 5	The Framework does not contain specific policies for nationally significant infrastructure projects. These are determined in accordance with the decision-making framework in the Planning Act 2008 (as amended) and relevant national policy statements for major infrastructure, as well as any other matters that are relevant (which may include the National Planning Policy Framework). National policy statements form part of the overall framework of national planning policy, and may be a material consideration in preparing plans and making decisions on planning applications.	The NPPF is considered to be important and relevant where policies are applicable to the Scheme but is to be given less weight in the SoS's decision making process than the relevant policies in the adopted Energy NPSs and Draft Energy NPSs.
Paragraph 7	The purpose of the planning system is to contribute to the achievement of sustainable development, including the provision of homes, commercial development, and supporting infrastructure in a sustainable manner. At a very high level, the objective of sustainable development can be summarised as meeting the needs of the present without compromising the ability of future generations to meet their own needs4. At a similarly high level, members of the United Nations – including the United Kingdom – have agreed to pursue the 17 Global Goals for Sustainable Development in the period to 2030. These address social progress, economic well-being and environmental protection5.	Scheme will deliver wider sustainability benefits, being a renewable energy development that will make a substantial contribution to the country achieving net-zero carbon emissions. Other sustainability considerations are explained in Section 6.7 of this Planning Statement [EN010132/EX6EX7/WB7.5_CD].



1.7 Table 7: NSIP Action Plan (February 2023)

1.7.1 Table 7 considers the Scheme in the context of the NSIP Action Plan (February 2023).

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
Paragraph 3.4.10	Nationally Significant Infrastructure: action plan for reforms to the planning process published in February 2023, sets out an extensive plan to reform the planning process for nationally significant infrastructure projects (NSIPs). The actions are grouped under give broad reform areas: setting a clear strategic direction, bringing forward operational reforms to support faster consenting, realising better outcomes for the environment, recognising the role of local authorities and strengthening community engagement with NSIPs and improving system-wide capacity and capability.	It is considered that the proposed changes to the NSIP Action Plan do not alter the assessment of the Scheme presented in the Planning Statement [EN010132/EX6EX7/WB7.5_CD]. Given the early stages of the reform with the Government's aim to publish consultation responses by Spring 2024, the NSIP Action Plan (February 2023) should be given limited weight at this stage.



1.8 Table 8: Powering Up Britain Energy Security Plan (March 2023)

1.8.1 Table 8 considers the Scheme in the context of the policy paper Powering Up Britain Security Plan (March 2023). The relevant paragraphs and compliance with policy is considered below.

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
Page 37	The UK has huge deployment potential for solar power, and we are aiming for 70 gigawatts of ground and rooftop capacity together by 2035. This amounts to a fivefold increase on current installed capacity. We need to maximise deployment of both types of solar to achieve our overall target.	Section 6.2 of WB7.5_A Planning Statement [EN010132/EX6EX7/WB7.5_CD] sets out how the Scheme will meet the compelling need for renewable energy in accordance with relevant national planning policies. In summary, the Scheme would:
		 Deliver a large amount of renewable generation capacity (21,956,988 MWh over the estimated 60-year assessed lifetime) to deliver the Government's energy objectives and legally binding net zero commitments in line with the requirements of paragraph 1.1.1 of NPS EN-3, paragraph 3.3.21 of draft NPS EN-1, section 3.4 of NPS EN-1 and the National Infrastructure Strategy 2020 (para. 6.2.32);
		Deliver a reduction of 3,981,049 tCO2e over the lifetime of the Scheme compared to if it did not go ahead which would make a significant contribution towards reducing carbon emissions as required by paragraph 1.1.1 of NPS EN-1, paragraph 2.3.2 of Draft NPS EN-1, the National Infrastructure Strategy 2020



		and the Energy White Paper: "Powering our net zero future" (para. 6.2.35);
		 Deliver in a timescale that is short in the context of the delivery of other forms of energy generation in line with the urgent need to decarbonise set out in paragraphs 3.3.5, 3.3.15 and 3.4.5 of NPS EN-1, Paragraph 2.3.2 of Draft NPS EN-1 and the National Infrastructure Strategy 2020 (paras. 6.2.1, 6.2.4 and 6.2.8);
		 Enable all consumers to benefit from the effect of low-marginal cost solar generation by reducing market prices, in line with the aim to provide affordable energy for consumers set out at Paragraph 2.3.2, Paragraph 2.3.5 and 3.3.21 of Draft NPS EN-1 (para 6.2.8, 6.2.9, and 6.2.10); and
		 Help ensure security and reliability of energy supply in line with Paragraph 2.3.2 and 2.3.5 of the Draft NPS EN-1 (para 6.2.8 and 6.2.9).
Page 37	Deploying rooftop solar remains a key priority for the Government, and it continues to be one of the most popular and easily deployed renewable energy sources; over a million homes now have solar panels installed. Solar can benefit households and businesses by allowing them to reduce electricity bills significantly and receive	WB6.2.5 ES Chapter 5 Alternatives and Design Evolution [APP-043] and its accompanying appendix WB6.3.5.1 ES Appendix 5.1 Site Selection Assessment [AS-004] explain how the site was chosen in light of that need.
	payment for excess electricity generated. Warehouses, distribution centres and industrial buildings with high electricity demand can offer significant potential for solar deployment, which can rapidly pay for itself by means of energy bill savings. The Government is looking to facilitate and promote extensive deployment of rooftop	Specifically, paragraph 2.1.10 of WB6.3.5.1 ES Appendix 5.1 Site Selection Assessment [AS-004] explains the reasons why a site of the size proposed is required to meet the 600MW grid connection offer. The methodology used for the site selection process is considered reasonable and



	solar on industrial and commercial property in order to make maximum usage of available surfaces for business as well as environmental and climate benefits.	proportionate and complies with the requirements of NPS EN-1 4.4.3 as explained at Section 2.1 [AS-004]. The consideration of alternatives has been undertaken within WB6.2.5 ES Chapter 5 Alternatives and Design Evolution [APP-043] and its accompanying appendix WB6.3.5.1 ES Appendix 5.1 Site Selection Assessment [AS-004]. Specifically, paragraphs 2.1.23 to 2.1.32 detail the consideration of brownfield land and roof tops and sets out why these were discounted as unsuitable. The methodology used for the site selection process is considered reasonable and proportionate and complies with the requirements of NPS EN-1 4.4.3.
Page 37 and 38	Ground-mounted solar is one of the cheapest forms of electricity generation and is readily deployable at scale. The Government seeks large scale ground-mount solar deployment across the UK, looking for development mainly on brownfield, industrial and low and medium grade agricultural land. Solar and farming can be complementary, supporting each other financially, environmentally and through shared use of land. We consider that meeting energy security and climate change goals is urgent and of critical importance to the country, and that these goals can be achieved together with maintaining food security for the UK. We encourage deployment of solar technology that delivers environmental benefits, with consideration for ongoing food production or environmental improvement. The Government will therefore not be making changes to categories of agricultural land in ways that might constrain solar deployment.	The clarification makes it clear that there is no intention to change the definitions of BMV land. It also states that it expects solar developments to take place on low/medium grade agricultural land. 73.76% of the Site, utilises 'low' grade, non-best and most versatile (BMV) agricultural land and is considered to be in a location supported by the Powering Up Britain Plan. Only 26.24% of the Site is located on best and most versatile land with clear justification for why these areas remain within the scheme set out in Section 5.7 of ES Chapter 5: Alternatives and Design Evolution [APP-043]. Given the reversible nature of the Scheme, BMV land will not be permanently lost and the Applicant therefore considers that the Scheme accords with this policy.

West Burton Solar Project

Planning Statement Appendix D:

Local Planning Policy
Accordance Table
Revision ED

Prepared by: Lanpro Services

April May 2024

PINS reference: EN010132

Document reference: EX6EX7/WB7.5 CD

APFP Regulation 5(2)(q)





Contents

<u>1</u>	LOCAL PLANNING POLICY ACCORDANCE TABLE	<u>3</u>
1.1	CENTRAL LINCOLNSHIRE LOCAL PLAN (CLLP) ADOPTED APRIL 2023.	3
1.2	BASSETLAW CORE STRATEGY & DEVELOPMENT MANAGEMENT POLICIES (BCSDMP) (ADOPTE 2011)	⊕ 42
1.3	EMERGING DRAFT BASSETLAW LOCAL PLAN 2020-2038 (PUBLICATION VERSION COMPOSITE (JULY 2022)	
1.4	NOTTINGHAMSHIRE MINERALS LOCAL PLAN (NMLP)	71
1.5	LINCOLNSHIRE MINERALS AND WASTE LOCAL PLAN (LMNLP) (CORE STRATEGY & DEVELOPM	I ENT
	MANAGEMENT POLICIES (JUNE 2016)	73
1.6	LINCOLNSHIRE MINERALS AND WASTE LOCAL PLAN SITE LOCATIONS (DECEMBER 2017)	7 8
2	NEIGHBOURHOOD PLANS	-80
2.1	——————————————————————————————————————)
2.1	2019 – 2036 Final Approved Version May 2017. West Lindsey District Council.	- 2/114
2.2	——STURTON BY STOW PARISH COUNCIL AND STOW PARISH COUNCIL (2022). STURTON BY STON	
2.2	AND STOW NEIGHBOURHOOD PLAN 2019 – 2036 FINAL APPROVED VERSION MARCH 2022.	
	GAINSBOROUGH: WEST LINDSEY DISTRICT COUNCIL.	86
2.3	——STURTON WARD NEIGHBOURHOOD PLAN (2021), STURTON WARD NEIGHBOURHOOD PLAN	
	Review (2021-2037). West Lindsey District Council.	94
2.4	TRESSWELL AND COTTAM NEIGHBOURHOOD PLAN (2019). (REFERENDUM VERSION). 2018 –	
_,	2033 (ADOPTED FEBRUARY 2019). BASSETLAW DISTRICT COUNCIL.	100
1	LOCAL PLANNING POLICY ACCORDANCE TABLE	4
1.1	CENTRAL LINCOLNSHIRE LOCAL PLAN (CLLP) ADOPTED APRIL 2023.	4
1.2	BASSETLAW CORE STRATEGY & DEVELOPMENT MANAGEMENT POLICIES (BCSDMP) (ADOPTE	<u>:D</u>
	<u>2011)</u>	_43
<u>1.3</u>	EMERGING DRAFT BASSETLAW LOCAL PLAN 2020-2038 (PUBLICATION VERSION COMPOSITE)
	(JULY 2022)	52
1.4	Nottinghamshire Minerals Local Plan (NMLP)	73
<u>1.5</u>	LINCOLNSHIRE MINERALS AND WASTE LOCAL PLAN (LMNLP) (CORE STRATEGY & DEVELOPM	<u>ENT</u>
	Management Policies (June 2016)	75
1.6	LINCOLNSHIRE MINERALS AND WASTE LOCAL PLAN SITE LOCATIONS (DECEMBER 2017)	80
2	NEIGHBOURHOOD PLANS	82
2.1	Saxilby with Ingleby Parish Council (2022). Saxilby with Ingleby Neighbourhood F	LAN
	2019 – 2036 Final Approved Version May 2017. West Lindsey District Council.	82
2.2	STURTON BY STOW PARISH COUNCIL AND STOW PARISH COUNCIL (2022). STURTON BY STOW	<u>N</u>
	AND STOW NEIGHBOURHOOD PLAN 2019 – 2036 FINAL APPROVED VERSION MARCH 2022.	
	GAINSBOROUGH: WEST LINDSEY DISTRICT COUNCIL.	88
2.3	STURTON WARD NEIGHBOURHOOD PLAN (2021). STURTON WARD NEIGHBOURHOOD PLAN	
	REVIEW (2021-2037). WEST LINDSEY DISTRICT COUNCIL.	96



2.4 TRESSWELL AND COTTAM NEIGHBOURHOOD PLAN (2019). (REFERENDUM VERSION). 2018 – 2033 (ADOPTED FEBRUARY 2019). BASSETLAW DISTRICT COUNCIL. 102



Issue Sheet

Report Prepared for: West Burton Solar Project Ltd. Planning Statement

Appendix D Local Planning Policy Accordance Table

Prepared by:

Name: Gabriel Cooper

Title: Graduate Planner

Approved by:

Name: Ian Douglass

Title: Director of Planning

Date: March 2023

Revision: 01

Revision	Date	Prepared by:	Approved by:
А	9 January 2024	GV	JC
В	28 February 2024	GV	JC
С	30 April 2024	SF	JC
D	8 May 2024	<u>SF</u>	<u>JC</u>



1 Local Planning Policy Accordance Table

1.1 Central Lincolnshire Local Plan (CLLP) Adopted April 2023.

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
Policy S1	The spatial strategy will focus on delivering sustainable growth for Central Lincolnshire that meets the needs for homes and jobs, regenerates places and communities, and supports necessary improvements to facilities, services and infrastructure. Development should create strong, sustainable, cohesive and inclusive communities, making the most effective use of previously developed land and enabling a larger number of people to access jobs, services and facilities locally. Development should provide the scale and mix of housing types and a range of new job opportunities that will meet the identified needs of Central Lincolnshire in order to secure balanced communities. Decisions on investment in services and facilities, and on the location and scale of development, will be assisted by the Central Lincolnshire Settlement Hierarchy. 8. Countryside Unless allowed by: a) policy in any of the levels 1-7 above; or b) any other policy in the Local Plan (such as Policies S4, S5, S34, or S43) or a relevant policy in a neighbourhood plan, development will be regarded as being in the countryside and as such restricted to:	Due to the scale of the land required to deliver the substantial renewable energy generation capacity that the Scheme will provide, and the need to be in sufficient proximity of the connection point to the National Electricity Transmission System (NETS), the Scheme could not be located within an urban area or settlement boundary as explained within the Site Selection Assessment [AS-004]. In terms of the specific S1 policy requirements: a) A solar development of this scale is not allowed by policy in any of the levels 1-7 above; b) The Scheme is allowed by Policy S14 as demonstrated at 6.1.4 of the Planning Statement and Policy S14 below. It meets the criteria for renewable energy development set out therein. The Scheme meets other requirements of Policy LP2, which are relevant to the type of development proposed, as follows: The application is for renewable energy generation as specifically allowed within Policy S1. The Scheme therefore complies with Policy S1.



	 that which is demonstrably essential to the effective operation of agriculture, horticulture, forestry, outdoor recreation, transport or utility services; delivery of infrastructure; renewable energy generation; and to minerals or waste development in accordance with separate Minerals and Waste Local Development Documents. 	As explained in the Statement of Need [APP-320] and summarised in Sections 3 and 4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme is a substantial infrastructure asset, capable of delivering large amounts of low-carbon electricity to help meet the UK's urgent need to decarbonise with solar technology supported by recent government policy. As well as more recently to provide security of supply as well as affordability for end consumers.
		The contribution the Scheme would make to meeting the established urgent need for renewable energy generation infrastructure warrants its location in a rural area.
Policy S5	Part E: Non-residential development in the countryside Proposals for non-residential development will be supported provided that: a) The rural location of the enterprise is justifiable to maintain or enhance the rural economy or the location is justified by means of proximity to existing established businesses or natural features; b) The location of the enterprise is suitable in terms of accessibility; c) The location of the enterprise would not result in conflict with neighbouring uses; and d) The development is of a size and scale commensurate with the proposed use and with the rural character of the location. Part F: Agricultural diversification	 a) Due to the scale of the land required to deliver the substantial renewable energy generation capacity that the Scheme will provide, and the need to be in sufficient proximity of the connection point to the National Electricity Transmission System (NETS), the Scheme could not be located within an urban area or settlement boundary as explained within the Site Selection Assessment [AS-004]. The rural location is therefore justified. b) There are no significant adverse impacts on the local highway network as demonstrated by ES Chapter 14: Transport and Access [APP-052]. The location is therefore suitable. c) The Scheme is acceptable in terms of neighbour amenity as detailed at Section 6.4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD]. The location has been carefully chosen to minimise impacts on surrounding land uses (see Site Selection Assessment [AS-004].



Proposals involving farm-based diversification to non-agricultural activities or operations will be permitted, provided that the proposal will support farm enterprises and providing that the development is:

- a) In an appropriate location for the proposed use;
- b) Of a scale appropriate to its location; and
- c) Of a scale appropriate to the business need.

Part G: Agricultural, forestry, horticultural or other rural land-based development

Proposals which will help farms modernise and/or adapt to funding changes or climate change will be supported in principle and any such proposals will be considered against relevant design, landscape and natural environment policies in this plan. Where permission is required, development proposals for buildings required for agriculture or other rural land-based development purposes will be supported where:

- a) It is demonstrated that there is a functional need for the building which cannot be met by an existing, or recently disposed of, building;
- b) the building is of a scale that is proportionate to the proposed functional need;
- c) the building is designed specifically to meet the functional need identified;
- d) the site is well related to existing buildings in terms of both physical and functional location, design and does not introduce isolated structures away from existing buildings; and
- e) significant earthworks are not required, and there will be no harm to natural drainage and will not result in pollution of soils, water or air.

The scale is required in order to provide the 480MW of electricity generation allowed via the grid connection offer from National Grid and in order to provide the significant benefits in terms of renewable energy generation as set out at Section 4 of the Planning Statement [EN010132/EX6EX7/WB7.5 CD].

In terms of agricultural diversification, the Scheme allows the landowner to diversify the uses within the land holding and use some of the lower grade agricultural land for solar generation. This helps to support the agricultural side of the business. See ES Chapter 19: Soils and Agriculture [APP-057].

Additionally, solar is recognised as having a significant role to play in the immediate decarbonisation and greenification of the UKs energy network, the Scheme is therefore considered to adapt agricultural land to combat climate change concerns.

The Scheme is therefore considered to comply with Policy S5.



Policy S11	All development should, where practical and viable, take opportunities to reduce the development's embodied carbon content, through the careful choice, use and sourcing of materials. Major development proposals: All major development proposals should explicitly set out what opportunities to lower a building's embodied carbon content have been considered, and which opportunities, if any, are to be taken forward. In the period to 31 December 2024, there will be no requirement (unless mandated by Government) to use any specific lower embodied carbon materials in development proposals, provided the applicant has at least demonstrated consideration of options and opportunities available. From 1 January 2025, there will be a requirement for a development proposal to demonstrate how the design and building materials to be used have been informed by a consideration of embodied carbon, and that reasonable opportunities to minimise embodied carbon have been taken. Further guidance is anticipated to be issued by the local planning authorities on this matter prior to 1 January 2025.	The Scheme makes a significant contribution towards limiting climate change and ES Chapter 7: Climate change [REP1-012] concludes it will have a significant beneficial effect in terms of climate change. Large scale solar farms, and the Scheme in particular, directly respond to the urgent need to deliver a large amount of renewable generation capacity quickly. The Scheme therefore represents a significant contribution to the zero-carbon hierarchy on a national scale. The construction of the Scheme has considered the impacts of the resource use and climate change. Mitigation includes the use of lower carbon construction methods; the recycling of waste and the reuse of materials is maximised wherever possible. Measures are detailed in the Outline CEMP [EN010132/EX6/WB7.1_D].REP6-021]. The Scheme therefore demonstrates compliance with this aspect of the policy. In addition, large scale solar farms, and the Scheme in particular, directly respond to the urgent need to deliver a large amount of
	authorities of this matter prior to 1 january 2023.	renewable generation capacity quickly as set out in the Statement of Need [APP-320]. The Scheme therefore represents a significant contribution to the zero-carbon hierarchy on a national scale.
Policy S14	The Central Lincolnshire Joint Strategic Planning Committee is committed to supporting the transition to a net zero carbon future and will seek to maximise appropriately located renewable energy generated in Central Lincolnshire (such energy likely being wind and solar based). Proposals for renewable energy schemes, including ancillary development, will be supported where the direct, indirect, individual	In terms of the specific requirements of this policy, robust evidence is provided within the application submission. Direct, indirect, and cumulative impacts of the Scheme have been assessed within the ES [APP-039 to APP-061] and help demonstrate the following: Scale siting and design has been given careful consideration and is acceptable as set out at Section 6.4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD].



and cumulative impacts on the following considerations are, or will be made, acceptable. To determine whether it is acceptable, the following tests will have to be met:

- i) The impacts are acceptable having considered the scale, siting and design, and the consequent impacts on landscape character; visual amenity; biodiversity; geodiversity; flood risk; townscape; heritage assets, their settings and the historic landscape; and highway safety and railway safety; and
- ii) The impacts are acceptable on aviation and defence navigation system/communications; and
- iii) The impacts are acceptable on the amenity of sensitive neighbouring uses (including local residents) by virtue of matters such as noise, dust, odour, shadow flicker, air quality and traffic;

Testing compliance with part (i) above will be via applicable policies elsewhere in a development plan document for the area (i.e. this Local Plan; a Neighbourhood Plan, if one exists; any applicable policies in a Minerals or Waste Local Plan); and any further guidance set out in a Supplementary Planning Document.

In order to test compliance with part (ii) above will require, for relevant proposals, the submission by the applicant of robust evidence of the potential impact on any aviation and defence navigation system/communication, and within such evidence must be documented areas of agreement or disagreement reached with appropriate bodies and organisations responsible for such infrastructure.

In order to test compliance with part (iii) above will require, for relevant proposals, the submission by the applicant of a robust assessment of

Impacts upon landscape Character and Visual amenity are assessed within ES Chapter 8: Landscape and Visual Impact [APP-046]. Section 6.5 of the Planning Statement concludes the 'acceptability' of the Scheme's limited landscape and visual impacts need to be weighed against the nationally significant benefits of the Scheme and acknowledge that with NSIP scale generation schemes, some landscape and visual impacts are acceptable. In this context it is considered that the landscape and visual effects that would result are not unacceptable, and that the Scheme is therefore generally compliant with S14.

Assessment of Ecological impacts is set out in ES Chapter 9: Ecology and Biodiversity [APP-047]. Section 6.9 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] concludes there are significant impacts identified on harvest mice (at a site level). The Scheme is therefore generally in accordance with S14.

These local policies must be considered in the context of the nationally significant benefits that the Scheme will bring, and the likely increased level of effect that is associated with, and acceptable for, a scheme of this scale in comparison with a smaller scheme that would deliver only locally or regionally significant benefits and for which the local policies are designed to deal with.

Flood Risk impacts are assessed in ES Chapter 10: **[APP-048]** and are concluded not significant.

ES Chapter 13: Cultural Heritage **[APP-051]** assesses the heritage impacts of the Scheme. Section 6.6 of the Planning Statement sets out the harm assessment. Arising from this, it is considered that the



the potential impact on such users, and the mitigation measures proposed to minimise any identified harm.

For all matters in (i)-(iii), the applicable local planning authority may commission its own independent assessment of the proposals, to ensure it is satisfied what the degree of harm may be and whether reasonable mitigation opportunities are being taken.

Where significant adverse effects are concluded by the local planning authority following consideration of the above assessment(s), such effects will be weighed against the wider environmental, economic, social and community benefits provided by the proposal. In this regard, and as part of the planning balance, significant additional weight in favour of the proposal will arise for any proposal which is community-led for the benefit of that community. In areas that have been designated for their national importance, as identified in the National Planning Policy Framework, renewable energy infrastructure will only be permitted where it can be demonstrated that it would be appropriate in scale, located in areas that do not contribute positively to the objectives of the designation, is sympathetically designed and includes any necessary mitigation measures.

Additional matters for solar based energy proposals

Proposals for solar thermal or photovoltaics panels and associated infrastructure to be installed on existing property will be under a presumption in favour of permission unless there is clear and demonstrable significant harm arising.

Proposals for ground-based photovoltaics and associated infrastructure, including commercial large-scale proposals, will be under a presumption in favour unless:

significant public benefits of the Scheme clearly and demonstrably outweigh the reversible harm.

The Scheme will be adequately served by highways infrastructure and there will be no significant impacts upon highway safety as demonstrated by ES Chapter 14: Transport and Access [APP-052].

- The impacts are acceptable on aviation and defence navigation system/communications as set out within Section 6.12 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD].
- The impacts are acceptable on the amenity of sensitive neighbouring uses (including local residents) by virtue of matters such as noise, dust, odour, shadow flicker, air quality and traffic as demonstrated in Good design (section 6.4), Landscape and Visual Impact (section 6.5), Noise (section 6.11), Glint and Glare (section 6.12), Air Quality (section 6.17) of the Planning Statement [EN010132/EX6EX7/WB7.5_CD].

Additional Matters for solar based energy proposals

Policy S15s presumption in favour of permission unless there is clear and demonstrable significant harm arising from solar developments is noted.

In this case, as set out above, the level of harm arising from
the Scheme is limited and must be considered in the context
of the nationally significant benefits that the Scheme will bring,
and the likely increased level of effect that is associated with,
and acceptable for, a scheme of this scale in comparison with



	 there is clear and demonstrable significant harm arising; or the proposal is (following a site-specific soil assessment) to take place on Best and Most Versatile (BMV) agricultural land and does not meet the requirements of Policy S67; or the land is allocated for another purpose in this Local Plan or other statutory based document (such as a nature recovery strategy or a Local Transport Plan), and the proposal is not compatible with such other allocation. Proposals for ground-based photovoltaics should be accompanied by evidence demonstrating how opportunities for delivering biodiversity net gain will be maximised in the scheme taking account of soil, natural features, existing habitats, and planting proposals accompanying the 	 a smaller scheme that would deliver only locally or regionally significant benefits as recognised by NPS EN-1 (2011) paragraph 3.2.3 and NPS EN-1 (November 2023) paragraph 3.1.1. The Scheme is largely located on Grade 3b agricultural land with 26.24% of the Sites comprising BMV agricultural land. This is justified by other sustainability considerations, as explained in Section 6.7 of this Planning Statement [EN010132/EX6EX7/WB7.5_CD]. The land is not allocated for another purpose within the Plan.
	Decommissioning renewable energy infrastructure Permitted proposals will be subject to a condition that will require the submission of an End-of-Life Removal Scheme within one year of the facility becoming non-operational, and the implementation of such a scheme within one year of the scheme being approved. Such a scheme should demonstrate how any biodiversity net gain that has arisen on the site will be protected or enhanced further, and how the materials to be removed would, to a practical degree, be re-used or recycled.	Decommissioning impacts have been assessed within the topic chapters of the ES [APP-039 to APP-061]. An outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] has been submitted setting out key principles for the safe and sustainable decommissioning of the Site. Provision of the detailed Decommissioning Strategy will be secured through the DCO. Based upon the above, the Scheme is considered to be generally compliant with Policy S14.
Policy S14 Main Modifications	Amend criterion (i) as follows: "i. The impacts are acceptable having considered the scale, siting and design, and the consequent impacts on landscape character; visual amenity; biodiversity; geodiversity; flood risk; townscape; heritage	ES Chapter 13: Cultural Heritage [APP-051] assesses the heritage impacts of the Scheme. Section 6.6 of the Planning Statement sets out the harm assessment. The significant public benefits of the Scheme clearly and demonstrably outweigh the reversible harm.



	assets , and their settings and the historic landscape ; and highway safety and rail safety ; and"	Rail safety with regards to the Scheme has been considered within the ES. Issues pertaining to Glint and Glare have been duly assessed through ES Chapter 16 [APP-054].
	In the section with the sub heading 'Additional matters for solar based energy proposals', amend the text as follows:	through ES Chapter To [APP-034].
	"Proposals for ground based photovoltaics and associated infrastructure, including commercial large scale proposals, will be under a presumption in favour unless:	
	• [no change]	
	 the proposal is (following a site specific soil assessment) to take place on Best and Most Versatile (BMV) agricultural land and does not meet the requirements of Policy S67, unless such land is peat based and the proposal is part of a wider scheme to protect or enhance the carbon sink of such land; or 	
	• [no change]"	
Policy S16	The Joint Committee is committed to supporting the transition to net zero carbon future and, in doing so, recognises and supports, in principle, the need for significant investment in new and upgraded energy infrastructure. Where planning permission is needed from a Central Lincolnshire authority, support will be given to proposals which are necessary for, or form part of, the transition to a net zero carbon sub-region, which could include: energy storage facilities (such as battery storage or thermal storage); and upgraded or new electricity facilities (such as transmission facilities sub stations are other electricity infrastructure)	The construction of the Scheme has considered the impacts of the resource use and climate change. Mitigation includes the use of lower carbon construction methods; the recycling of waste and the reuse of materials is maximised wherever possible. Measures are detailed in the Outline CEMP [EN010132/EX6/WB7.1_D].REP6-021]. The Scheme therefore demonstrates compliance with this aspect of the policy. In addition, large scale solar farms, and the Scheme in particular, directly respond to the urgent need to deliver a large amount of
	facilities, sub-stations or other electricity infrastructure. However, any such proposals should take all reasonable opportunities to mitigate any harm arising from such proposals, and take care to	renewable generation capacity quickly. The Scheme therefore represents a significant contribution to the zero-carbon hierarchy on a national scale.



	select not only appropriate locations for such facilities, but also design solutions (see Policy S53) which minimises harm arising.	As detailed in Section 3 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme has been subject to a detailed and sensitive iterative design process. This has taken account of the context and features of the land within the Order limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts, and provision of environmental and other enhancements, where practicable. The design process and basis of design decisions taken are described in the Chapter 5: Alternatives and Design Evolution of the ES [APP-043]. The Scheme is considered to comply with Policy S16.
Policy S17	Carbon Sequestration The demonstration of meaningful carbon sequestration through nature-based solutions within a proposal will be a material consideration in the decision-making process. Material weight in favour of a proposal will be given where the net situation is demonstrated to be a significant gain in nature-based carbon sequestration as a consequence of the proposal. Where a proposal will cause harm to an existing nature-based carbon sequestration process, weight against such a proposal will be given as a consequence of the harm, with the degree of weight dependent on the scale of net loss.	In tandem with ensuring a biodiversity net gain, planting schemes and the gaping of hedgerows is considered to contain carbon. This is due to the embodied carbon that is expected to be captured as the plantations mature. The Scheme is considered to comply with Policy S17.
Policy S21	Protecting the Water Environment Development proposals that are likely to impact on surface or ground water should consider the requirements of the Water Framework Directive.	As detailed in ES Chapter 10: Hydrology, Flood Risk and Drainage [APP-048], the main risks relating to water and drainage are silt laden runoff, spillages, leaks and pollutants during the construction / decommissioning stage and diffuse pollution contained in urban runoff during the operation phase from a water quality / resource



Development proposals should demonstrate:

- g) that water is available to support the development proposed;
- h) that adequate mains foul water treatment and disposal already exists or can be provided in time to serve the development. Non mains foul sewage disposal solutions should only be considered where it can be shown to the satisfaction of the local planning authority that connection to a public sewer is not feasible;
- i) that they meet the Building Regulation water efficiency standard of 110 litres per occupier per day or the highest water efficiency standard that applies at the time of the planning application (see also Policy S12);
- j) that water reuse and recycling and rainwater harvesting measures have been incorporated wherever possible in order to reduce demand on mains water supply as part of an integrated approach to water management (see also Policy S11);
- k) that they have followed the surface water hierarchy for all proposals:
- i. surface water runoff is collected for use;
- ii. discharge into the ground via infiltration;
- iii. discharge to a watercourse or other surface water body;
- iv. discharge to a surface water sewer, highway drain or other drainage system, discharging to a watercourse or other surface water body;
- v. discharge to a combined sewer;
- l) that no surface water connections are made to the foul system;
- m) that surface water connections to the combined or surface water system are only made in exceptional circumstances where it can be demonstrated that there are no feasible alternatives (this applies to

perspective. Mitigation measures are to be included within a detailed CEMP and DEMP. There is considered to be a low risk of pollution from the Scheme where mitigation measures and schemes of work are complied with.

The mitigation measures are considered to comply with the details of this policy.



new developments and redevelopments) and where there is no detriment to existing users;

- n) that no combined sewer overflows are created in areas served by combined sewers, and that foul and surface water flows are separated;
- o) that development contributes positively to the water environment and its ecology where possible and does not adversely affect surface and ground water quality in line with the requirements of the Water Framework Directive:
- p) that development with the potential to pose a risk to groundwater resources is not located in sensitive locations to meet the requirements of the Water Framework Directive;
- q) how Sustainable Drainage Systems (SuDS)/ Integrated Water Management to deliver improvements to water quality, the water environment and to improve amenity and biodiversity net gain wherever possible have been incorporated into the proposal unless they can be shown to be impractical;
- r) that relevant site investigations, risk assessments and necessary mitigation measures for source protection zones around boreholes, wells, springs and water courses have been agreed with the relevant bodies (e.g. the Environment Agency and relevant water companies);
- s) that suitable access is safeguarded for the maintenance of watercourses, water resources, flood defences and drainage infrastructure; and
- t) that adequate provision is made to safeguard the future maintenance of water bodies to which surface water and foul water treated on the site of the development is discharged, preferably by an appropriate



	authority (e.g. Environment Agency, Internal Drainage Board, Water Company, the Canal and River Trust or local Council). In order to allow access for the maintenance of watercourses, development proposals that include or abut a watercourse should ensure no building, structure or immovable landscaping feature is included that will impede access within 8m of a watercourse, or within 16m of a tidal watercourse. Conditions may be included where relevant to ensure this access is maintained in perpetuity and may seek to ensure responsibility for maintenance of the watercourse including land ownership details up to and of the watercourse is clear and included in maintenance arrangements for future occupants.	
Policy S28	In principle, employment related development proposals should be consistent with meeting the following overall spatial strategy for employment. The strategy is to strengthen the Central Lincolnshire economy offering a wide range of employment opportunities focused mainly in and around the Lincoln urban area and the towns of Gainsborough and Sleaford, with proportionate employment provision further down the Settlement Hierarchy (see Policy S1). Aligned to the Greater Lincolnshire Local Industrial Strategy, and as a key component of the Midlands Engine, there will be significant growth in a number of sectors, most notably agri-food, manufacturing, business services and the visitor economy, including accommodation and food services. Land has been made available in appropriate locations in this plan to meet the strategic needs identified in Central Lincolnshire. Strategic Employment Sites (SES), and existing Important Established Employment Areas (IEEA) will be protected for their importance to the	 The Scheme would have a positive impact on employment in the renewable energy sector. This includes the following: Employment during the construction phase. It is expected that 296 net FTE jobs will be created during the construction period. During the operational phase, 12 FTE staff would be employed for operation and maintenance of the site. Diversification of local employment from a predominantly agricultural and tourism base. Due to the scale of the land required to deliver the substantial renewable energy generation capacity that the Scheme will provide, it is considered reasonable that the Scheme could not be located within an SES, IEEA or SUEs. Chapter 18: Socio-Economics, Tourism and Recreation of the ES [APP-056] includes an assessment of socio-economic impacts of the Scheme, including employment.



	economy. Employment development will mainly be directed to these SES and IEEA and at Sustainable Urban Extensions (SUEs) as part of mixed-use communities being created. Elsewhere, policies will seek to protect Local Employment Areas (LEA) to help ensure there are jobs and services available to meet the local needs of the community and to allow enterprises to flourish at suitable sites across Central Lincolnshire. Outside of existing employment areas and allocated sites, economic development will typically be limited to small-scale proposals which satisfy the requirements of Policy S33 or Policy S34.		
Policy S34	In locations outside of the settlements named in the Settlement Hierarchy in Policy S1, proposals for employment generating development will be limited to the expansion of an existing employment use and development proposals that support the growth of the agrifood sector or other land-based rural businesses and buildings in accordance with relevant parts of Policy S5, and only where the following criteria are satisfied: a) It would be consistent in scale with its rural location, without unacceptable environmental and/or visual impacts; and	a)	The rural location is justified due to the scale of the land required to deliver the substantial renewable energy generation capacity that the Scheme will provide, and the need to be in sufficient proximity of the connection point to the National Electricity Transmission System (NETS). The Scheme could not be located within an urban area or settlement boundary as explained within the Site Selection Assessment [AS-004]. The visual effects of the Scheme have been considered within ES Chapter 8: Landscape and Visual [APP-046].
	 b) It would not adversely affect existing local community services and facilities; and c) It is designed to be compatible with the landscape in which it would be situated; and d) It would not cause undue harm to the open nature of the countryside, or any site protected for its natural or heritage qualities, including designated and non-designated sites; and 	b)	The nature of the Scheme is not considered to result in adverse impacts upon existing community services. As assessed within ES Chapter 8: Landscape and Visual [APP-046], the Scheme is considered to be generally compatible with the landscape for which it is set.



	e) It will not impact unacceptably on the local and/or strategic highway network; and f) In the case of a conversion, the building is not in such a state of dereliction or disrepair that significant reconstruction would be required.	 d) The assessments of potential visual and landscape effects are captured within appendices 8.2 and 8.3 [APP-073] & [APP-074]. e) Highway impacts of the Scheme have been assessed within ES Chapter 14 Transport and Access [APP-052]. The Scheme is not considered to unacceptably impact upon the local/strategic highway network.
Policy S42	Development proposals which result in the loss of facilities or attractions that support the visitor economy, including hotels and guesthouses, will not be permitted unless: e) there are overriding sustainability and regeneration benefits from the	The Scheme directly responds to the urgent need to deliver a large amount of renewable generation capacity quickly. The Scheme therefore represents a significant contribution to the zero-carbon hierarchy on a national scale which presents a sustainability benefit.
	proposal; or f) the existing use is demonstrated to be unviable and with no reasonable prospect of becoming viable; or	Chapter 18: Socio-Economics, Tourism and Recreation of the ES [APP-056] includes an assessment of socio-economic impacts of the Scheme, including impacts upon tourism.
	g) the facility has been appropriately marketed for a continuous period of 12 months or more without successful conclusion on terms that reflect the lawful use and condition of the premises – this evidence will be considered in the context of the local market conditions and state of the wider national economy.	There will be no loss of facilities or attractions that support the visitor economy as a result of the Scheme. The Scheme is therefore considered to comply with this Policy.
Policy S45	Planning permission will only be granted if it can be demonstrated that there is, or will be, sufficient infrastructure capacity to support and meet all the necessary requirements arising from the proposed development. Development proposals must consider all of the infrastructure implications of a scheme; not just those on the site or its immediate vicinity. Conditions or planning obligations, as part of a package or combination of infrastructure delivery measures, are likely to be	ES Chapter 14: Transport and Access [APP-052] assesses the Baseline Conditions through Section 14.5. The Chapter, through Section 14.7, summarises that the effects during construction and decommissioning are anticipated to be similar and negligible to minor in severity. The operational impact upon transport and access is also considered to be negligible to minor in severity. Mitigation measures have been incorporated within Appendix 14.2 Construction Traffic Management Plan [REP4-



	required for many proposals to ensure that new development meets this principle. Consideration must be given to the likely timing of infrastructure provision. As such, development may need to be phased. Conditions or a planning obligation may be used to secure this phasing arrangement.	O38EN010132/EX7/WB6.3.14.2_E], Public Rights of Way Management Plan [REP5-018], Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline OEMP [REP5-020] and the Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023]. The Scheme will be adequately served by highways infrastructure and there will be no significant impacts upon highway safety as demonstrated by ES Chapter 14: Transport and Access [APP-052].
	New Development should be supported by, and have good access to infrastructure.	demonstrated by E3 Chapter 14. Transport and Access [AFF-032].
	Development Contributions	No offsite developer contributions towards infrastructure are
	Developers will be expected to contribute towards the delivery of relevant infrastructure, either through direct provision or contribution towards the provision of local and strategic infrastructure to meet the needs arising from the development either alone or cumulatively with other developments.	necessary as a result of the development and no S106 agreement is proposed.
Policy S47	Development proposals which contribute towards an efficient and safe transport network that offers a range of transport choices for the	A) The nature and scale of the proposed use, means that a rural location is necessary for the Scheme.
	movement of people and goods will be supported. All developments should demonstrate, where appropriate, that they have had regard to the following criteria:	B) A Construction Traffic Management Plan has been produced [REP4-038]. EN010132/EX7/WB6.3.14.2 E]. Workers will be encouraged to car share where possible while a shuttle bus
	a) Located where travel can be minimised and the use of sustainable transport modes maximised;	service will consolidate trips to the Site.
	b) Minimise additional travel demand through the use of measures such as travel planning, safe and convenient public transport, car clubs, walking and cycling links and integration with existing infrastructure;	C) As secured by the Construction Environmental Management Plan [EN010132/EX6/WB7.1_DREP6-021], vehicles will be switched off when possible and will not be unnecessarily revved. Construction vehicles will also conform to current EU
	c) Making allowance for low and ultra-low emission vehicle refuelling infrastructure.	emission standards.



Policy S48	Development proposals should facilitate active travel by incorporating measures suitable for the scheme from the design stage. Plans and evidence accompanying applications will demonstrate how the ability to travel by foot or cycle will be actively encouraged by the delivery of well designed, safe and convenient access for all both into and through the site. Priority should be given to the needs of pedestrians, cyclists, people with impaired mobility and users of public transport by providing a network of high quality pedestrian and cycle routes and green corridors, linking to existing routes and public rights of way where opportunities exist, that give easy access and permeability to adjacent areas. Proposals will: a) protect, maintain and improve existing infrastructure, including closing gaps or deficiencies in the network; b) provide high quality attractive routes that are safe, direct, legible and pleasant and are integrated into the wider network; c) ensure the provision of appropriate information, including signposting and way-finding to encourage the safe use of the network; d) encourage the use of supporting facilities, especially along principle cycle routes; e) make provision for secure cycle parking facilities in new developments and in areas with high visitor numbers across Central Lincolnshire; and f) consider the needs of all users through inclusive design.	The Public Rights of Way Management Plan [REP5-018] highlights how existing footpaths and Public Rights of Way will be protected and limited from disruption as far as possible. Where diversions will be required for construction and decommissioning works, disruption will be kept to a minimum. As captured within the Draft Development Consent Order [EN010132/EX6EX7/WB3.1_GH], Work No. 11 is to create a permissive footpath from the track off Sykes Lane along the Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub Appropriate signposting across the Scheme is to be provided for public safety, knowledge and wayfinding.
Policy S48 Main Modifications	Amend bullet point a) of Policy S48 to read:	Given the land take of the Scheme, the rural nature of the Scheme does not lend to connecting communities and facilities.



	"protect, maintain and improve existing infrastructure, including closing gaps or deficiencies in the network and connecting communities and facilities ;"	The Scheme does however propose to enhance existing PRoWs as per the Public Rights of Way Management Plan [REP5-018]. Additionally, through Work No. 11, the Scheme proposes the creation of a permissive footpath from the track off Sykes Lane along the Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub. The Scheme is therefore considered to contribute to rural improvements to connecting communities.
Policy S49	Parking Provision Non-Residential Development All other types of development should incorporate a level of car parking that is suitable for the proposed development taking into account its location, its size and its proposed use, including the expected number of employees, customers or visitors. Infrastructure relating to electric vehicle charging points should be provided in accordance with Policy NS18.	During Construction, when it is proposed that there will be 296 FTE Staff on Site, the provision of parking compounds has been detailed within the Outline CTMP [REP4-038].EN010132/EX7/WB6.3.14.2_E]. It is considered that a suitable allocation, in the form of temporary compounds, has been provisioned in relation to meeting parking needs. The Scheme therefore complies with this policy.
Policy S53	All development, including extensions and alterations to existing buildings, must achieve high quality sustainable design that contributes positively to local character, landscape and townscape, and supports diversity, equality and access for all. Good design will be at the centre of every development proposal, and this will be required to be demonstrated through evidence supporting planning applications to a degree proportionate to the proposal. All development proposals will be assessed against, and will be expected to meet the following relevant design and amenity criteria. All development proposals will: 1. Context	As detailed in Section 3 and Section 6.4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] and by the Design and Access Statement [APP-314 to APP-315] , the Scheme has been subject to a detailed and sensitive iterative design process, resulting in a high standard of design as required by this policy. This has taken account of the context and features of the land within the Order limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts, and provision of environmental and other enhancements, where practicable. The



- a) Be based on a sound understanding of the context, integrating into the surroundings and responding to local history, culture and heritage;
- b) Relate well to the site, its local and wider context and existing characteristics including the retention of existing natural and historic features wherever possible and including appropriate landscape and boundary treatments to ensure that the development can be satisfactorily assimilated into the surrounding area;
- c) Protect any important local views into, out of or through the site;
- 2. Identity
- a) Contribute positively to the sense of place, reflecting and enhancing existing character and distinctiveness;
- 4. Movement
- b) Maximise pedestrian and cycle permeability and avoid barriers to movement through careful consideration of street layouts and access routes both within the site and in the wider context contributing to the delivery of walkable and cyclable neighbourhoods in accordance with Policy S48;
- 5. Nature
- a) Incorporate and retain as far as possible existing natural features including hedgerows, trees, and waterbodies particularly where these features offer a valuable habitat to support biodiversity, aligned with policies in the Natural Environment chapter of the Local Plan;
- b) Incorporate appropriate landscape and boundary treatments to ensure that the development can be satisfactorily assimilated into the surrounding area, maximising opportunities to deliver diverse ecosystems and biodiverse habitats, strengthening wildlife corridors and green infrastructure networks, and helping to achieve wider goals

design process and basis of design decisions taken are described in the Chapter 5: Alternatives and Design Evolution of the ES [APP-043].

As explained by the Design and Access Statement [APP-314 to APP-315], the design of the Scheme and its components will be sensitive to its surroundings. Maximum height parameters seek to deliver a scheme that integrates with its surroundings, whilst delivering the technical requirements that enable the efficient generation of a large amount of electricity. For example: the maximum heights of solar arrays have been designed to deliver the technical requirements whilst enabling effective screening by hedgerows; BESS units are not proposed to be double stacked in order to minimise height; and where possible, fencing will comprise deer fence or other wire mesh security fencing on timber poles that is in-keeping with the character of the Order limits. The extent and layout of the Scheme is also sensitive to landscape character and locating the largest structures in the less tranquil and most well screened areas of the Order limits.

As explained by the Design and Access Statement [APP-314 to APP-315], the design of the Scheme has been sensitive to the visual amenity of residential properties and the setting of heritage assets, incorporating stand-offs between these and PV Arrays where to mitigate potential impacts.

The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline OEMP [REP5-020] and Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] set out measures for the efficient use of resources including, where possible, the reuse and recycling of materials.



for biodiversity net gain, climate change mitigation and adaptation and water management;

9. Resources

- a) Minimise the need for resources both in construction and operation of buildings and be easily adaptable to avoid unnecessary waste in accordance with Policies S10 and S11;
- b) Use high quality materials which are not only suitable for the context but that are durable and resilient to impacts of climate change in accordance with the requirements of Policy S20;

The ES Chapter, Waste, [APP-058] details how waste arisings will be prevented and designed out where possible. Opportunities to re-use material resources will be sought where practicable. Where re-use and prevention are not possible, waste arisings will be managed in line with the Waste Hierarchy and detailed Construction Resource Management Plan (CRMP) (see Outline CEMP [EN010132/EX6/WB7.1_DREP6-021]).

Careful consideration will be given to the selection of materials, including, for example, the use of deer fence or other wire mesh security fencing on timber poles that is in-keeping with the character of the Order limits.

The Scheme will enhance the PRoW network within Order limits with an additional permissive path which will help to enhance the identity of the local area whilst maximising pedestrian and cycle permeability. A minimum width has been incorporated into the Scheme design for PRoW and permissive paths, as well as the corridor in which they will be provided (between Scheme infrastructure). In all cases the PRoW and new permissive paths will be of typical width, 1.5–3.0m, with at least 5m spacing either side of the centreline of the PRoW and therefore delivering a minimum 10m space. This will avoid the perception of being channelled into narrow passages between PV Arrays. The details of these are explored within the Public Rights of Way Plan [REP4-010].

Planting proposals set out in the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025] will use native species. This also sets out maintenance arrangements for planting.



		The construction of the Scheme has considered the impacts of the resource use and climate change. Mitigation includes the use of lower carbon construction methods, segregation of materials for recycling and the reuse of materials wherever possible. Measures are detailed in the Outline CEMP [EN010132/EX6/WB7.1_D].REP6-021]. The Scheme therefore demonstrates compliance with this aspect of the policy.
		Chapter 7 Climate Change of the ES [REP1-012] presents a lifecycle greenhouse gas (GHG) impact assessment which considers the impact of GHG emissions arising over the lifetime of the Scheme on the climate. This concludes that over its up to 60 year operational lifetime the Scheme will produce 31,425,614 MWh of electricity and deliver a reduction of 3,981,049 tCO2e over the lifetime of the Scheme compared to if it did not go ahead. This demonstrates its very low carbon attributes compared to other non-renewable forms of electricity generation, providing an overall major beneficial impact in relation to the UK meeting its carbon reduction targets and therefore represents a major beneficial effect on the climate.
		On the basis of the above, the Scheme is considered to comply with Policy S53.
Policy S54	The potential for achieving positive mental and physical health outcomes will be taken into account when considering all development proposals. Where any potential adverse health impacts are identified, the applicant will be expected to demonstrate how these will be addressed and mitigated. The Central Lincolnshire authorities will expect development proposals to promote, support and enhance	Where potential adverse health impacts have been identified, ES Chapter 22: Summary of Mitigation [APP-060] consolidates how the Scheme will address and mitigate these impacts. The promotion, support and enhancement of both physical and mental wellbeing has been a guiding principle of the Scheme. This has resulted in a number of enhancements to existing PRoWs and Footpaths as captured within the Public Rights of Way Management



	 physical and mental health and wellbeing, and thus contribute to reducing health inequalities. This will be achieved by: d) Ensuring quality green infrastructure provides adequate access to nature for its benefits to mental and physical health and wellbeing and potential to overcome health inequalities. 	Plan [REP5-018]. The Scheme also proposes, through Work No.11, the creation of a permissive footpath to run from the track off Sykes Lane along the Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub. Furthermore, a summary document has also been prepared which draws together the information on human health [REP4-077]. This document sets out how the issue of mental health and wellbeing has been assessed and considered in the Environmental Statement.
Policy S56	Development proposals must take into account the potential environmental impacts on people, biodiversity, buildings, land, air and water arising from the development itself and any former use of the site, including, in particular, adverse effects arising from pollution.	Geo-Environmental Risk Assessments [APP-095 to APP-104] have been prepared for the Scheme and demonstrate that there are no significant constraints to development as a result of ground conditions and contamination.
	Where development is proposed on a site which is known to be or has the potential to be affected by contamination, a preliminary risk assessment should be undertaken by the developer and submitted to the relevant Central Lincolnshire Authority as the first stage in assessing the risk of contamination.	The Scheme includes embedded mitigation for ground conditions and contamination in the form of a Construction Environmental Management Plan (CEMP) and Decommissioning Strategy, which will include procedures for the identification and mitigation of contaminant risks associated with the construction. An
	Proposals will only be permitted if:	Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and Outline
	 it can be demonstrated that the site is suitable for its proposed use; layout and drainage have taken adequate account of ground 	Decommissioning Strategy [EN010132/EX6/WB7.2_BREP6-023] form part of the application. Maintenance works will require similar mitigation measures.
	conditions, contamination and gas risks arising from previous uses and any proposed sustainable land remediation and	ES Chapter 11: Ground conditions and contamination [APP-049] concludes no potential significant effects have been identified after
	there are no significant impacts on future users, neighbouring users, groundwater or surface water.	the implementation of embedded well-established good industry practices in construction for managing contaminated land which will be incorporated into a CEMP and Decommissioning Strategy and utilised in all phases of the Scheme. It is considered that the



		potential effects of contamination or risk of contamination will not be significant.
Policy S57	Development proposals should protect, conserve and seek opportunities to enhance the historic environment of Central Lincolnshire.	ES Chapter 13: Cultural Heritage [APP-051] assesses the heritage impacts of the Scheme. Section 6.6 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD].
	In instances where a development proposal would affect the significance of a heritage asset (whether designated or non-designated), including any contribution made by its setting, the applicant will be required to undertake and provide the following, in a manner	Chapter 13: Cultural Heritage of the ES [APP-051] confirms that there would be moderate adverse residual effects upon one listed building over the construction phase.
	proportionate to the asset's significance: a) describe and assess the significance of the asset, including its setting, to determine its architectural, historical or archaeological interest; and b) identify the impact of the proposed works on the significance and	Clear and convincing justification for the works is provided within Section 4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Statement of Need [APP-320] and design evolution of the Scheme is explained within the Design and Access Statement [APP-314 to APP-315].
	c) provide a clear justification for the works, especially if these would harm the significance of the asset, including its setting, so that the harm can be weighed against public benefits. outlines the mitigation mean design pertaining to cultural stand-offs between the Sche	Section 13.8 of Chapter 13: Cultural Heritage of the ES [APP-051] outlines the mitigation measures embedded within the Scheme design pertaining to cultural heritage. This includes the provision of stand-offs between the Scheme and heritage assets in order to help to preserve their setting during the construction, operational and
	d) protect the significance of heritage assets (including where relevant their setting) by protecting and enhancing architectural and historic character, historical associations, landscape and townscape features and through consideration of scale, design, architectural detailing, materials, siting, layout, mass, use, and views and vistas both from and towards the asset;	decommissioning periods. By providing the embedded mitigatio and stand-offs the Scheme respects and responds to the local context of heritage assets, in accordance with this policy. The Scheme does not involve any internal or external alterations extensions to a listed building or listed structure, nor does it involved the structure.



e) promote opportunities to better reveal significance of heritage assets, where possible;

f) take into account the desirability of sustaining and enhancing nondesignated heritage assets and their setting.

Proposals to alter or to change the use of a heritage asset, will be supported provided:

- g) the proposed use is compatible with the significance of the heritage asset, including its fabric, character, appearance, setting and, for listed buildings, interior; and
- h) such a change of use will demonstrably assist in the maintenance or enhancement of the heritage asset; and
- i) features essential to the special interest of the individual heritage asset are not harmed to facilitate the change of use.

Development proposals that will result in substantial harm to, or the total loss of, a designated heritage asset will only be granted permission where it is necessary to achieve substantial public benefits that outweigh the harm or loss, and the following criteria can be satisfied:

- j) the nature of the heritage asset prevents all reasonable uses of the site; and
- k) no viable use of the heritage asset itself can be found in the medium term through appropriate marketing that will enable its conservation; and
- l) conservation by grant-funding or some form of not for profit, charitable or public ownership is demonstrably not possible; and
- m) the harm or loss is outweighed by the benefit of bringing the site back into use

Section 13.5 of the ES Chapter 13: Cultural Heritage [APP-051] includes an assessment of the impact of the Scheme upon conservation areas within 5km of the Order Limits.

Archaeological evaluations were undertaken in addition to a desk-based assessment, including a geophysical survey of the whole scheme and targeted trial trenching. The scope and specification of each field investigation have been set out in Written Scheme of Investigations (WSI). The results of these surveys (Appendix 13.2 the ES [APP-109 to APP-114]) have been incorporated in Section 13.5 of Chapter 13, Cultural Heritage, of the ES [APP-051].

The Scheme is considered to comply with the requirements of Policy S57.



Where a development proposal would result in less than substantial harm to a designated heritage asset, permission will only be granted where the public benefits, including, where appropriate, securing its optimum viable use, outweigh the harm.

Where a non-designated heritage asset is affected by development proposals, there will be a presumption in favour of its retention, though regard will be had to the scale of any harm or loss and the significance of the heritage asset. Any special features which contribute to an asset's significance should be retained and reinstated, where possible.

Listed Buildings

Permission to change the use of a Listed Building or to alter or extend such a building will be granted where the local planning authority is satisfied that the proposal is in the interest of the building's conservation and does not involve activities or alterations prejudicial to the special architectural or historic interest of the Listed Building or its setting.

Development proposals that affect the setting of a Listed Building will, in principle, be supported where they make a positive contribution to, or better reveal the significance of the Listed Building.

Conservation Areas

Significant weight will be given to the protection and enhancement of Conservation Areas.

Development within, affecting the setting of, or affecting views into or out of, a Conservation Area should conserve, or where appropriate enhance, features that contribute positively to the area's special character, appearance and setting, including as identified in any adopted Conservation Area appraisal. Proposals should:



- n) retain buildings/groups of buildings, existing street patterns, historic building lines and ground surfaces and architectural details that contribute to the character and appearance of the area;
- o) where relevant and practical, remove features which have a negative impact on the character and appearance of the Conservation Area;
- p) retain and reinforce local distinctiveness with reference to height, massing, scale, form, materials and plot widths of the existing built environment:
- q) assess, and mitigate against, any negative impact the proposal might have on the townscape, roofscape, skyline and landscape; and
- r) aim to protect trees, or where losses are proposed, demonstrate how such losses are appropriately mitigated against.

Archaeology

Development affecting archaeological remains, whether known or potential, designated or undesignated, should take every practical and reasonable step to protect and, where possible, enhance their significance.

Planning applications for such development should be accompanied by an appropriate and proportionate assessment to understand the potential for and significance of remains, and the impact of development upon them.

If initial assessment does not provide sufficient information, developers will be required to undertake field evaluation in advance of determination of the application. This may include a range of techniques for both intrusive and non-intrusive evaluation, as appropriate to the site.



	Wherever possible and appropriate, mitigation strategies should ensure the preservation of archaeological remains in-situ. Where this is either not possible or not desirable, provision must be made for preservation by record according to an agreed written scheme of investigation submitted by the developer and approved by the planning authority. Any work undertaken as part of the planning process must be	
Policy S57 Main Modifications	appropriately archived in a way agreed with the local planning authority. Amend the fourth paragraph of Policy S57 to read: "Proposals to alter or to change the use of a heritage asset, or proposals that would affect the setting of a heritage asset, will be supported provided:"	This amendment sees the removal of the consideration to the setting of a heritage asset. Notwithstanding this, ES Chapter 13 Cultural Heritage [APP-051] has captured the Scheme's impact upon the setting of heritage assets.
Policy S57 Main Modifications	Amend the first paragraph under the "Conservation Areas" heading to read: "Significant weight will be given to the protection and enhancement of Conservation Areas (as defined on the Policies Map)."	This amendment does not impact upon the policy assessment above.
Policy S58	All development proposals should contribute to the realisation of the following key principles: Lincoln a) Protect the dominance and approach views of Lincoln Cathedral, Lincoln Castle and uphill Lincoln on the skyline; b) Protect Lincoln's distinct built heritage and townscape character as set out in the Lincoln Townscape Character Assessment;	The Scheme is not considered to impact upon the views upon Lincoln's historic skyline.
	c) Respect Lincoln's unique character and setting and relationship with surrounding villages by maintaining and enhancing a strategic green infrastructure network around and into the City, including Green	



	Wedges (see Policy S63) to protect the City's green character and to maintain the setting and integrity of surrounding villages;	
	d) Proposals within, adjoining or affecting the setting of the 11 Conservation Areas and 3 historic parks and gardens within the built up area of Lincoln, should preserve and enhance their special character, setting, appearance and respect their special historic and architectural context;	
	e) Support the development of art, cultural and leisure assets and facilities, such as the Collection, the Theatre Royal, the Engine Shed, Arboretum and Whisby Nature Park, and improve access to such assets and facilities; and	
	f) Do not detract from the open character of Lincoln's Brayford Pool and waterways, protecting and enhancing them as a major focal points in and through the City.	
Policy S59	Development proposals should ensure that existing and new green and blue infrastructure is considered and integrated into the scheme design from the outset. Where new green infrastructure is proposed, the design and layout should take opportunities to: a) incorporate a range of types and sizes of green and blue spaces, green routes and environmental features that are appropriate to the development and the wider green and blue infrastructure network to	Green infrastructures, in the form of Public Rights of Way have undergone analysis and are to be enhanced during the construction phase as secured within the CEMP [EN010132/EX6/WB7.1_DREP6-021] in order to deliver lasting improvements to the green infrastructure, and indirectly to the social infrastructure that is Public Rights of Way. No loss or harm to the green and blue infrastructure network is anticipated.
	maximise the delivery of multi-functionality; b) deliver biodiversity net gain and support ecosystem services;	The Scheme will protect and enhance biodiversity. A Biodiversity Net Gain (BNG) assessment, using Defra's Metric 3.0, has been provided
	c) respond to landscape/townscape and historic character;	with the DCO application [EN010132/EX6EX7/WB3.1_GH]. For the purposes of BNG, the Scheme will result in an overall significant net
	d) support climate change adaptation and resilience including through use of appropriate habitats and species; and	gain. Measures to enhance the biodiversity value of the Order limits and enhance the quality and connectivity of habitats are set out by the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025].



	e) encourage healthy and active lifestyles. Development proposals must protect the linear features of the green and blue infrastructure network that provide connectivity between green infrastructure assets, including public rights of way, bridleways, cycleways and waterways, and take opportunities to improve and expand such features. Development will be expected to make a contribution proportionate to their scale towards the establishment, enhancement and on-going management of green and/or blue infrastructure by contributing to the development of the strategic green infrastructure network within Central Lincolnshire, in accordance with the Developer Contributions SPD.	An assessment of the potential landscape and visual impacts associated with the construction, operation and decommissioning of the Scheme has been carried out and is presented in Chapter 8: Landscape and Visual Impact of the ES [APP-046]. Section 8.7 of Chapter 8: Landscape and Visual Impacts of the ES [APP-046] outlines and identifies the likely significant effects of the Scheme before addressing mitigation measures in section 8.8. Where possible, the Scheme has been designed to minimise its impacts on the landscape, townscape and historic character of surrounding areas. The construction of the Scheme has considered the impacts of the resource use and climate change. Mitigation includes the use of lower carbon construction methods, segregation of materials for recycling and the reuse of materials wherever possible. Measures are detailed in the Outline CEMP [EN010132/EX6/WB7.1_D]_REP6-021]. The Scheme therefore demonstrates compliance with this aspect of the policy. Through the Public Rights of Way Management Plan [REP5-018], it is proposed that the PRoWs which travers the Scheme are to be improved. This is considered to constitute a contribution which is proportionate to the Scheme. The enhancements to the PRoWs encourage healthy and active lifestyles. The Scheme is considered to comply with the requirements of Policy S59.
Policy S60	All development should: a) protect, manage, enhance and extend the ecological network of habitats, species and sites of international, national and local	Chapter 9: Ecology and Biodiversity of the ES [APP-047] sets out all the designated sites of ecological or geological conservation importance, including internationally, nationally, and locally



- importance (statutory and non-statutory), including sites that meet the criteria for selection as a Local Site;
- b) minimise impacts on biodiversity and features of geodiversity value;
- c) deliver measurable and proportionate net gains in biodiversity in accordance with Policy S61; and
- d) protect and enhance the aquatic environment within or adjoining the site, including water quality and habitat.

Part One: Designated Sites

The following hierarchy of sites will apply in the consideration of development proposals:

International Sites

The highest level of protection will be afforded to internationally protected sites. Development proposals that will have an adverse impact on the integrity of such areas, will not be supported other than in exceptional circumstances, in accordance with the NPPF.

Development proposals that are likely to result in a significant adverse effect, either alone or in combination with other proposals, on any internationally designated site, must satisfy the requirements of the Habitats Regulations (or any superseding similar UK legislation). Development requiring Appropriate Assessment will only be allowed where it can be determined, taking into account mitigation, that the proposal would not result in significant adverse effects on the site's integrity.

2. National Sites (NNRs and SSSIs)

designated sites; protected species; and habitats and other species identified as being of principal importance for the conservation of biodiversity. It assesses the impact of the Scheme upon designated sites, protected species and habitats.

The Scheme delivers a significant net gain in biodiversity of 86.80% gains provided in habitat, 54.71% gains in hedgerow and 33.25% gains in river units as detailed within the Biodiversity Net Gain Report **[APP-088]**.

Measures have been taken to ensure that where waste/ polluted water may be produced (for example: fire fighting a BESS failure) water is captured to ensure it cannot enter the aquatic environment. Measures to protect the aquatic environment have been captured within the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], OEMP [REP5-020], Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] and the outline BSSMP [APP-318].

Section 6.9 of the Planning Statement

[EN010132/EX6EX7/WB7.5_CD] sets out the Scheme's compliance policy S60.



Development proposals should avoid impact on these nationally protected sites. Development proposals within or outside a national site, likely to have an adverse effect, either individually or in combination with other developments, will not normally be supported unless the benefits of the development, at this site, clearly outweigh both the adverse impacts on the features of the site and any adverse impacts on the wider network of nationally protected sites.

3. Irreplaceable Habitats

Planning permission will be refused for development resulting in the loss, deterioration or fragmentation of irreplaceable habitats, including ancient woodland and aged or veteran trees, unless there are wholly exceptional reasons, and a suitable compensation strategy will be delivered.

4. Local Sites (LNR, LWS and LGS)

Development likely to have an adverse effect on locally designated sites, their features or their function as part of the ecological network, will only be supported where the benefits of the development clearly outweigh the loss, and the coherence of the local ecological network is maintained. Where significant harm cannot be avoided, the mitigation hierarchy should be followed.

Where adverse impacts are likely, development will only be supported where the need for and benefits of the development clearly outweigh these impacts. In such cases, appropriate mitigation or compensatory measures will be required.

Part Three: Mitigation of Potential Adverse Impacts

Development should avoid adverse impact on existing biodiversity and geodiversity features as a first principle, in line with the mitigation



	hierarchy. Where adverse impacts are unavoidable, they must be adequately and proportionately mitigated. If full mitigation cannot be provided, compensation will be required as a last resort where there is no alternative.	
	Development will only be supported where the proposed measures for mitigation and/or compensation along with details of net gain are acceptable to the Local Planning Authority in terms of design and location, and are secured for the lifetime of the development with appropriate funding mechanisms that are capable of being secured by condition and/or legal agreement.	
	If significant harm to biodiversity resulting from development cannot be avoided, adequately mitigated, or, as a last resort, compensated for, then planning permission will be refused.	
Policy S60 Main Modifications	Remove references to the policies map in the headings under Part One of the policy as follows:	These modifications do not impact upon the Scheme's compliance with the Policy.
	"2. National Sites (NNRs and SSSIs as shown on the Policies Map)"	
	And	
	"4. Local Sites (LNR, LWS and LGS as shown on the Policies Map) "	
Policy S61	Following application of the mitigation hierarchy, all development proposals should ensure opportunities are taken to retain, protect and enhance biodiversity and geodiversity features proportionate to their scale, through site layout, design of new buildings and proposals for existing buildings with consideration to the construction phase and ongoing site management.	The Scheme has taken all opportunities to ensure biodiversity features and geodiversity features are retained, protected and enhanced as far as possible. These measures are captured within Chapter 9: Ecology of the ES [APP-047] and Chapter 11: Ground Conditions and Contamination of the ES [APP-049].
	Development proposals should create new habitats, and links between habitats, in line with Central Lincolnshire Biodiversity Opportunity and Green Infrastructure Mapping evidence, the biodiversity opportunity	Chapter 9: Ecology of the ES [APP-047] also sets out all the designated sites of ecological or geological conservation importance, including internationally, nationally, and locally designated sites;



area principles set out in Appendix 4 to this Plan and the Local Nature Recovery Strategy (once completed), to maintain and enhance a network of wildlife sites and corridors, to minimise habitat fragmentation and provide opportunities for species to respond and adapt to climate change.

Proposals for major and large-scale development should seek to deliver wider environmental net gains where feasible.

All qualifying development proposals must deliver at least a 10% measurable biodiversity net gain attributable to the development. The net gain for biodiversity should be calculated using Natural England's Biodiversity Metric.

Biodiversity net gain should be provided on-site wherever possible. Offsite measures will only be considered where it can be demonstrated that, after following the mitigation hierarchy, all reasonable opportunities to achieve measurable net gains on-site have been exhausted or where greater gains can be delivered off-site where the improvements can be demonstrated to be deliverable and are consistent with the Local Nature Recovery Strategy.

All development proposals, unless specifically exempted by Government, must provide clear and robust evidence for biodiversity net gains and losses in the form of a biodiversity gain plan, which should ideally be submitted with the planning application (or, if not, the submission and approval of a biodiversity gain plan before development commences will form a condition of any planning application approval), setting out:

a) information about the steps to be taken to minimise the adverse effect of the development on the biodiversity of the onsite habitat and any other habitat;

protected species; and habitats and other species identified as being of principal importance for the conservation of biodiversity. It assesses the impact of the Scheme upon designated sites, protected species and habitats.

The Scheme delivers a significant net gain in biodiversity of 86.80% gains provided in habitat, 54.71% gains in hedgerow and 33.25% gains in river units as detailed within the Biodiversity Net Gain Assessment [APP-088].

The management of new and improved on onsite and offsite habitats will be planned for the operational lifetime of the project, being up to 60 years. The Scheme is therefore in compliance with the 30-year obligation.

Section 6.9 of the Planning Statement

[EN010132/EX6EX7/WB7.5_CD] sets out the Scheme's compliance.



	b) the pre-development biodiversity value of the onsite habitat;	
	 the post-development biodiversity value of the onsite habitat following implementation of the proposed ecological enhancements/interventions; 	
	d) the ongoing management strategy for any proposals;	
	e) any registered off-site gain allocated to the development and the biodiversity value of that gain in relation to the development; and	
	f) exceptionally any biodiversity credits purchased for the development through a recognised and deliverable offsetting scheme.	
	Demonstrating the value of the habitat (pre- and post-development) with appropriate and robust evidence will be the responsibility of the applicant. Proposals which do not demonstrate that the post-development biodiversity value will exceed the pre-development value of the onsite habitat by a 10% net gain will be refused.	
	Ongoing management of any new or improved onsite and offsite habitats, together with monitoring and reporting, will need to be planned and funded for 30 years after completion of a development.	
Policy S61 Main Modifications	Add new fourth paragraph to Policy S61 and amend the current fifth and sixth paragraphs as follows:	These policy amendments do not propose an impact upon the Scheme given the significant uplift and proposed enhancements to
	Biodiversity Net Gain The following part of the policy applies unless, and until, subsequently superseded, in whole or part, by national regulations or Government policy associated with the delivery of mandatory biodiversity net gain arising from the Environment Act 2021. Where conflict between the policy below and the provisions of Government regulations or national policy arises, then the latter should prevail.	BNG. The Scheme delivers a significant net gain in biodiversity of 86.80% gains provided in habitat, 54.71% gains in hedgerow and 33.25% gains in river units as detailed within the Biodiversity Net Gain Assessment [APP-088].



	Biodiversity net gain should be provided on-site wherever possible. Off- site measures will only be considered where it can be demonstrated that, after following the mitigation hierarchy, all reasonable opportunities to achieve measurable net gains on-site have been exhausted or Biodiversity offsetting schemes should only be used in exceptional circumstances, where net gain cannot be achieved within the site boundary or where greater gains can be delivered off-site where the improvements can be demonstrated to be deliverable and are consistent with the Local Nature Recovery Strategy. All development proposals, unless specifically exempted by	
	Government, must provide clear and robust evidence for biodiversity net gains and losses in the form of a biodiversity gain plan, which should ideally be submitted with the planning application (or, if not, the submission and approval of a biodiversity gain plan before development commences will form a condition of any planning application approval), setting out:	
Policy S62	Areas of Great Landscape Value Areas of Great Landscape Value (AGLV) are locally designated landscape areas recognised for their intrinsic character and beauty and their natural, historic and cultural importance. A high level of protection will be afforded to AGLV reflecting their locally important high scenic quality, special landscape features and sensitivity. Development proposals within, or within the setting of, AGLV shall: e) conserve and enhance the qualities, character and distinctiveness of locally important landscapes; and	The Scheme is not located within an Area of Great Landscape Value. An assessment of the potential landscape and visual impacts associated with the construction, operation and decommissioning of the Scheme has been carried out and is presented in Chapter 8: Landscape and Visual Impact of the ES [APP-046]. Section 8.7 of Chapter 8: Landscape and Visual Impacts of the ES [APP-046] outlines and identifies the likely significant effects of the Scheme before addressing mitigation measures in section 8.8. Areas of Great Landscape Value where their setting is applicable have been considered in this Chapter. Mitigation measures have been proposed to minimise, protect and conserve their qualities and features. The



	f) protect, and where possible enhance, specific landscape, wildlife and historic features which contribute to local character and landscape quality; and	Chapter concludes that the Scheme will not have adverse impacts upon AGLVs and is therefore in compliance with the Policy.
	g) maintain landscape quality and minimise adverse visual impacts through high quality building and landscape design; and	
	h) demonstrate how proposals have responded positively to the landscape character in relation to siting, design, scale and massing and where appropriate have retained or enhanced important views, and natural, historic and cultural features of the landscape; and	
	i) where appropriate, restore positive landscape character and quality.	
	Where a proposal may result in adverse impacts, it may exceptionally be supported if the overriding benefits of the development demonstrably outweigh the harm – in such circumstances the harm should be minimised and mitigated through design and landscaping.	
Policy S66	Development proposals should be prepared based on the overriding principle that:	As stated in Chapter 9: Ecology and Biodiversity of the ES [APP-047], The Scheme will not result in the loss of ancient woodland or veteran
	the existing tree and woodland cover is maintained, improved and expanded; and	trees. The Scheme will also retain existing hedgerow field boundaries and will enhance hedgerows where possible. In order to mitigate
	opportunities for expanding woodland are actively considered, and implemented where practical and appropriate to do so.	against the loss of hedgerows, HDD will be conducted to minimise disruption. Whilst some loss of vegetation will be required, this loss is vastly outweighed by the additional planting and mitigation
	Existing Trees and Woodland	measures imposed.
	Planning permission will only be granted if the proposal provides evidence that it has been subject to adequate consideration of the impact of the development on any existing trees and woodland found on-site (and off-site, if there are any trees near the site, with 'near' defined as the distance comprising 12 times the stem diameter of the off-site tree). If any trees exist on or near the development site,	As detailed in Section 3 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme has been subject to a detailed and sensitive iterative design process. As a result, the impact upon existing trees and woodlands has been given due consideration and has been influential in shaping the design process.



'adequate consideration' is likely to mean the completion of a British Standard 5837 Tree Survey and, if applicable, an Arboricultural Method Statement.

Where the proposal will result in the loss or deterioration of:

- a) ancient woodland; and/or
- b) the loss of aged or veteran trees found outside ancient woodland, permission will be refused, unless and on an exceptional basis the need

for, and benefits of, the development in that location clearly outweigh the loss.

Where the proposal will result in the loss or deterioration of a tree protected by a Tree Preservation Order or a tree within a Conservation Area, then permission will be refused unless:

- c) there is no net loss of amenity value which arises as a result of the development; or
- d) the need for, and benefits of, the development in that location clearly outweigh the loss.

Where the proposal will result in the loss of any other tree or woodland not covered by the above, then the Council will expect the proposal to retain those trees that make a significant contribution to the landscape or biodiversity value of the area, provided this can be done without compromising the achievement of good design for the site.

Mitigating for loss of Trees and Woodland

Where it is appropriate for higher value tree(s) (category A or B trees (BS5837)) and/or woodland to be lost as part of a development proposal, then appropriate mitigation, via compensatory tree planting,

Where there is to be any hedgerow or tree loss, these losses are outweighed by the substantial public benefits of the Scheme set out at Section 4 of the Planning Statement

[EN010132/EX6EX7/WB7.5_CD] and within the Statement of Need [APP-320].

Undeveloped buffers will be included to protect all hedgerows and ponds during construction and operation. Within some of these buffers, particularly around the natural regeneration of woodland will create additional scrub and woodland habitat. Other areas will be managed as grassland. Tree Root Protection fencing will be erected around retained trees, in line with British Standard BS 5837: Trees in relation to design, demolition and construction.

Appropriate and sensitive screening has also been developed and implemented to minimise the visual intrusion of the Scheme, while avoiding obscuring or intruding upon key views and relationships between heritage assets. This has resulted in the proposed gapping up of existing hedgerows to minimise visual intrusion.

The Scheme delivers a significant net gain in biodiversity of 86.80% gains provided in habitat, 54.71% gains in hedgerow and 33.25% gains in river units as detailed within the Biodiversity Net Gain Assessment [APP-088]. This uplift is considered to be reflective of substantial planting of trees, hedgerows etc.

The Scheme is therefore considered to comply with the requirements of Policy S66.



will be required. Such tree planting should be on-site wherever possible and should:

- e) take all opportunities to meet the six Tree Planting Principles (see supporting text); and
- f) unless demonstrably impractical or inappropriate, provide the following specific quantity of compensatory trees:

Management and Maintenance

In instances where new trees and/or woodlands are proposed, it may be necessary for the council to require appropriate developer contributions to be provided, to ensure provision is made for appropriate management and maintenance of the new trees and/or woodland.

Hedgerows

Proposals for new development will be expected to retain existing hedgerows where appropriate and integrate them fully into the design having regard to their management requirements.

Proposals for new development will not be supported that would result in the loss of hedges of high landscape, heritage, amenity or biodiversity value unless the need for, and benefits of, the development clearly outweigh the loss and this loss can be clearly demonstrated to be unavoidable.

Development requiring the loss of a hedgerow protected under The Hedgerow Regulations will only be supported where it would allow for a substantially improved overall approach to the design and landscaping of the development that would outweigh the loss of the hedgerow. Where any hedges are lost, suitable replacement planting or restoration



	of existing hedges, will be required within the site or the locality, including appropriate provision for maintenance and management.		
Policy S67	Proposals should protect the best and most versatile agricultural land so as to protect opportunities for food production and the continuance of the agricultural economy. With the exception of allocated sites, significant development resulting in the loss of the best and most versatile agricultural land will only be supported if:	[APP Grad Versa consi	emonstrated by ES Chapter: Soils and Agriculture Appendices 2-303 to APP-307], the Scheme is predominantly located on le 3b land. 26.24% of the land within the Sites is Best and Most atile (BMV) land. This is justified by other sustainability iderations, as explained in Section 6.7 of the Planning Statement 1010132/EX6EX7/WB7.5_CD].
	a) The need for the proposed development has been clearly established and there is insufficient lower grade land available at that settlement (unless development of such lower grade land would be inconsistent with other sustainability considerations); and	a)	The need for the development is clearly established as set out within Section 4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] and the Statement of Need [APP-320].
	b) The benefits and/or sustainability considerations outweigh the need to protect such land, when taking into account the economic and other benefits of the best and most versatile agricultural land; and	b)	There is clear justification for including the BMV land as explained in Section 6.7 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD].
	 c) The impacts of the proposal upon ongoing agricultural operations have been minimised through the use of appropriate design solutions; and d) Where feasible, once any development which is supported has ceased its useful life the land will be restored to its former use (this condition will be secured by planning condition where appropriate). Where proposals are for sites of 1 hectare or larger, which would result in the loss of best and most versatile agricultural land, an agricultural land classification report should be submitted, setting out the justification for such a loss and how criterion b has been met. 	c) d)	The impacts of the proposal upon ongoing agricultural operations have been minimised through the use of appropriate design solutions. This is also explained within Section 6.7 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD]. The land will be restored to its former use upon decommissioning. Section 6.7 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] explains how soil quality will be protected in order to ensure that the above policy requirement is met



		Agricultural Land Classification Reports have been undertaken and are provided at ES Appendix [APP-303 to APP-305]. These reports provide a justification for such a loss of BMV.
		The Scheme is therefore considered to comply with the requirements of this policy.
Policy S67 Main Modifications	Amend Policy S67 to read: Proposals should protect the best and most versatile agricultural land so as to protect opportunities for food production and the continuance of the agricultural economy. With the exception of allocated sites, significant development resulting in the loss of the best and most versatile agricultural land will only be supported if:	The amendment to this policy increases its specificity to the Scheme given its significant nature. The address of the policy stipulations explored above in Policy S67 remain pertinent and it is considered that this modification does not impact upon the Scheme's compliance with the Policy.



1.2 Bassetlaw Core Strategy & Development Management Policies (BCSDMP) (Adopted 2011)

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
POLICY DM1	Economic Development in the Countryside	With regard to the specific requirements of this policy:
	A. General Principles	i. Due to the scale of the land required to deliver the substantial
	Proposals for standalone economic development (e.g., tourist attractions; equine enterprises; rural business) in rural areas will be supported where they can demonstrate that:	renewable energy generation capacity that the Scheme will deliver, and the need to be in sufficient proximity of the connection point to the National Electricity Transmission System (NETS), the Scheme could not be located within
	i) any necessary built facilities will be provided by the re-use of	development boundaries or reuse existing buildings.
	existing buildings or, where the re-use of existing buildings is not feasible, new buildings are located and designed to minimise their impact upon the character and appearance of the countryside;	 The Scheme location is justified as set out within the Site Selection Assessment [AS-004]. This is within West Lindsey District except for part of the Cable Corridor.
	ii) the development requires the specific location proposed and there are no other suitable sites in, or close to, settlements	iii. The viability of the Scheme is demonstrated by the Statement of Need [APP-320].
	covered by policies CS2-CS8 or on brownfield land;	iv. Section 6.4 of the Planning Statement demonstrates that the
	iii) they are viable as a long-term business;	scale, design and form of the proposal, in terms of both buildings and operation, will be appropriate for its location
	iv) the scale, design and form of the proposal, in terms of both	and setting and be compatible with surrounding land uses;
	buildings and operation, will be appropriate for its location and setting and be compatible with surrounding land uses;	v. The proposal does not contain a retail use; and
	v) where the proposal includes a retail use, it is demonstrated that this will not have an adverse impact on the vitality or viability of local centres; rural service centres; and shops and services in surrounding villages; and	vi. The Scheme will be adequately served by highways infrastructure and there will be no significant impacts upon highway safety as demonstrated by ES Chapter 14: Transport and Access [APP-052].



	vi) they will not create significant or exacerbate existing environmental or highway safety problems. B. Farm Diversification Proposals to diversify the range of activities operating on a farm will be supported where it can be demonstrated that they meet the above criteria, and that the diversification proposal is required to support the continued viability of the existing farming enterprise.	Although the Scheme does assist in diversifying the landowners' range of activities within their land holdings and will assist in the continued viability of the farms as discussed within ES Chapter 19: Soils and Agriculture [APP-057], the requirement for justification re ensuring the continued viability of the existing farming enterprises is in this case considered to be overridden by the strong need case for the Scheme set out at Section 4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] and within the Statement of Need [APP-320]. The Scheme is considered to generally comply with the requirements of the above policy.
POLICY DM4	A. Major Development Principles All major development proposals will need to demonstrate that they: i. make clear functional and physical links with the existing settlement and surrounding area and have not been designed as 'standalone' additions. Where physical links cannot be made (e.g. for reasons of third party land ownership) provision must be made such that they can be provided in future should the opportunity arise; ii. complement and enhance the character of the built, historic and natural environment; iii. are of a scale appropriate to the existing settlement and surrounding area and in line with the levels of proposed growth for that settlement as set out in policies CS1-CS9; and iv. provide a qualitative improvement to the existing range of houses, services, facilities, open space and economic development opportunities.	Whilst the Scheme, given its nature, does not strictly complement and enhance the character of the built and historic environment. The Scheme's impact has been minimised within Bassetlaw given that the Cable Route Corridor is to be laid underground. The impacts are acceptable on the amenity of sensitive neighbouring uses (including local residents) by virtue of matters such as noise, dust, odour, shadow flicker, air quality and traffic as demonstrated in Good design (section 6.4), Landscape and Visual Impact (section 6.5), Noise (section 6.11), Glint and Glare (section 6.12), Air Quality (section 6.17) Waste (Section 6.14) of the Planning Statement [EN010132/EX6EX7/WB7.5_CD].



	New development should ensure that it does not have a detrimental effect on the residential amenity of nearby residents; provides a decent standard of private amenity space; allows adequate space for waste and recycling storage and collection; and is not to the detriment of highway safety.	
POLICY DM7	A. Future Development Proposals Particular support will be given to economic development proposals that are able to: ii. guarantee employment programmes for local residents that provide opportunities for training and development and will contribute to raised workforce skills levels within the District; and/or iii. deliver, or contribute to, opportunities for the growth of indigenous businesses; and/or iv. bring significant, good quality inward investment opportunities to the District; and/or	In terms of guaranteeing employment for local residents, the Skills and Supply Chain Plan [APP-319] has been produced to analyse the current economic baseline, the economic effects, opportunities for economic improvement and monitoring and feedback for seeking to secure employment and skills for local people. The Plan seeks to develop ways in which to raise workforce skills. Within ES Chapter 18: Socioeconomics, Tourism and Recreation [APP-056], the effects of the Scheme on economic development have been assessed. It is assumed that an element of inward investment will be present as FTE employees work during the Construction, Operation and Decommissioning of the Scheme. The Scheme is considered to comply with this policy.
POLICY DM8	B. Development Affecting Heritage Assets and Non-Designated Assets There will be a presumption against development, alteration, advertising or demolition that will be detrimental to the significance of a heritage asset. Proposed development affecting heritage assets, including alterations and extensions that are of an inappropriate scale, design or material, or which lead to the loss of important spaces, including infilling, will not be supported.	The only part of the Scheme located within Bassetlaw is the Cable Route Corridor which is to be located underground. ES Chapter 13: Cultural Heritage [APP-051] does not conclude any significant impacts upon heritage assets within Bassetlaw District. ES Chapter 13: Cultural Heritage [APP-051] concludes in Section 13.11 that the Scheme's construction, operation and decommissioning will not result in any significant adverse effects for Non-Designated Heritage Assets. The Scheme is considered to comply with the requirements of this policy.



	The setting of an asset is an important aspect of its special architectural or historic interest and proposals that fail to preserve or enhance the setting of a heritage asset will not be supported.	
POLICY DM9	B. Biodiversity and Geodiversity	B. Biodiversity and Geodiversity
	Development proposals will be expected to take opportunities to restore or enhance habitats and species' populations and to demonstrate that they will not adversely affect or result in the loss of features of recognised importance, including:	Assessment of Ecological impacts on all the habitats and species listed i-vii in Policy DM9 is set out in ES Chapter 9: Ecology and Biodiversity [APP-047]. Section 6.9 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] concludes there are significant
	i. Protected trees and hedgerows;	impacts identified on harvest mice (at a site level). This harm will be
	ii. Ancient woodlands;	mitigated as far as possible through appropriate habitat provision and management and the impacts are justified by the substantial
	iii. Sites of Special Scientific Interest (SSSI);	public benefits of the Scheme outlined at Section 4 of the Planning
	iv. Regionally Important Geodiversity Sites;	Statement [EN010132/EX6EX7/WB7.5_CD]. The Scheme is therefore
	v. Local Wildlife Sites (Sites of Importance for Nature Conservation (SINC)); vi. Local and UK Biodiversity Action Plan Habitats (including Open Mosaic Habitats on Previously Developed Land); and	generally in accordance with Policy DM9. The Scheme delivers a significant net gain in biodiversity of 86.80% gains provided in habitat, 54.71% gains in hedgerow and 33.25% gains in river units as detailed within the Biodiversity Net Gain
	vii. Protected Species. Development that will result in the loss of such features may be supported where replacement provision is made that is considered to be of equal or greater value than that which will be lost, and which is likely to result in a net gain in biodiversity. Where new development may have an adverse impact on such features, alternative scheme designs that minimise impact must be presented to the Council for consideration before the use of mitigation measures is considered. Where sufficient mitigation measures cannot be delivered, compensation measures must be provided as a last resort.	Assessment [APP-088]. This local policy must be considered in the context of the nationally significant benefits that the Scheme will bring, and the likely increased level of effect that is associated with, and acceptable for a scheme of this scale in comparison with a smaller scheme that would deliver only locally or regionally significant benefits and for which the local policies are designed to deal with. Additionally, C. Landscape Character



C. Landscape Character

New development proposals in and adjoining the countryside will be expected to be designed so as to be sensitive to their landscape setting. They will be expected to enhance the distinctive qualities of the landscape character policy zone in which they would be situated, as identified in the Bassetlaw Landscape Character Assessment. Proposals will be expected to respond to the local recommendations made in the Assessment by conserving, restoring, reinforcing, or creating landscape forms and features accordingly.

Impacts upon landscape Character are assessed within ES Chapter 8: Landscape and Visual Impact [APP-046]. The Cable Route Corridor is to traverse the landscape underground and will therefore is not anticipated to harm the landscape setting.

The Scheme complies with policy DM9 as it protects and enhances green infrastructure assets through retention of existing vegetation and the introduction of a significant quantum of new green infrastructure leading to the creation of new habitats and the long-term management of such features, as secured through the LEMP. This includes the retention of existing protected features and designations outlined in Section B of the policy. The Scheme will also conserve, restore, and create landscape forms and features which aim to reinforce and enhance landscape character,

The Scheme is considered generally compliant with DM9.

POLICY DM10

A. Carbon Reduction

The Council will be supportive of proposals that seek to utilise renewable and low carbon energy to minimise CO2 emissions. Proposals for renewable and low carbon energy infrastructure will also need to demonstrate that they:

- are compatible with policies to safeguard the built and natural environment, including heritage assets and their setting, landscape character and features of recognised importance for biodiversity;
- ii) will not lead to the loss of or damage to high-grade agricultural land (Grades 1 & 2);
- iii) are compatible with tourism and recreational facilities;

Large scale solar farms, and the Scheme in particular, directly respond to the urgent need to deliver a large amount of renewable generation capacity quickly. The Scheme therefore represents a significant contribution to the zero-carbon hierarchy on a national scale. In terms of the specific policy requirements:

i) It is generally compatible (taking into consideration the nationally significant benefits that the Scheme will bring, and the likely increased level of effect that is associated with, and acceptable for a scheme of this scale in comparison with a smaller scheme) with policies to safeguard the built and natural environment, including heritage assets and their setting, landscape character and features of recognised importance for biodiversity as demonstrated by Section 6.4,



- iv) will not result in unacceptable impacts in terms of visual appearance; noise; shadow-flicker; watercourse engineering and hydrological impacts; pollution; or traffic generation; and
- v) will not result in an unacceptable cumulative impact in relation to the factors above.

Large-scale renewable and low carbon energy proposals must provide full details of arrangements for decommissioning and reinstatement of the site if/when it ceases to operate.

6.5, 6.7 and 6.9 of the Planning Statement **[EN010132/EX6EX7/WB7.5_CD]**;

- ii) The Scheme will result in a the Order Limits being developed across 26.24% BMV land with the remainder of the Scheme being built across Non-BMV land [EN010132/EX6EX7/WB7.5 CD];
- iii) The Scheme is considered generally compatible with tourism and recreation facilities. ES Chapter 18: Socio-Economics, Tourism and Recreation [APP-056] considers the impacts of Construction, Operation and Decommissioning activities and employment on the consumption of temporary accommodation facilities. These impacts are considered of minor adverse impact on the tourism economy and is therefore in compliance with the Policy.
- iv) The Scheme is not considered to result in unacceptable impacts in terms of any aspects captured within the policy other than for traffic generation where, during construction and decommissioning, it is anticipated to produce negligible adverse impacts upon the network with regard to the construction of the Cable Route Corridor in Bassetlaw. See ES Chapter 18: Socioeconomics, Tourism and Recreation [APP-056].
- v) The Scheme is not considered to result in unacceptable adverse impacts.

An Outline Decommissioning Statement

[EN010132/EX6/WB7.2_BREP6-023] has been produced in order to provide a base from which the decommissioning and reinstatement of the site will be conducted. Given the timescale and nature of the



		proposal, full details will be provided in time within a Decommissioning Statement, following the Outline Statement.
		The Scheme is considered to generally accord with the requirements of this policy.
POLICY DM11	All applications will be expected to demonstrate that the necessary infrastructure (social, physical and green) will be in place in advance of, or can be provided in tandem with, new development and, where appropriate, that arrangements are in place for its subsequent maintenance. Arrangements for the provision or improvement of infrastructure required by the proposed development and/or to mitigate the impact of	The Scheme, through Work No.11, will result in the creation of a permissive footpath from the track off Sykes Lane along the Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub [EN010132/EX6EX7/WB3.1_GH]. This Works package of the Scheme is considered to facilitate betterment with regards to physical health outcomes. This Work No. is to be located within West Lindsey District Council but forms part of the Scheme's whole.
	that development will, in line with national guidance and legislation, be secured by Community Infrastructure Levy (CIL) charge, planning obligation or, where appropriate, via conditions attached to a planning permission.	In addition, other green infrastructures, in the form of Public Rights of Way have undergone analysis and are to be enhanced during the construction phase as secured within the CEMP [EN010132/EX6/WB7.1_DREP6-021] in order to deliver lasting
	Obligations may include, but not be limited to: v. Green infrastructure:	improvements to the green infrastructure, and indirectly to the social infrastructure that is Public Rights of Way.
	 Open Space (e.g., Play Areas; Sports Fields/Youth and Adult Areas; amenity open space); 	Good design has been a key consideration from the outset. The LVIA has informed the iterative design process, including taking account of published landscape character assessment guidance and
	 Natural Heritage (e.g., mitigation measures; habitat restoration; habitat protection; habitat creation; landscaping; site 	fieldwork analysis.
	management; or site interpretation); viii. Flood mitigation measures (e.g., flood warning measures; reopening of culverts);	The overall objective of the landscape design is to integrate the Scheme into its landscape setting and avoid or minimise adverse landscape and visual effects as far as practicable. The design has been developed in collaboration with the wider design team, other
		specialists and the Host Authorities landscape advisors to achieve a



		solution that achieves this objective whilst maximising opportunities to deliver net gains in biodiversity gain (green infrastructure). Habitat protection and habitat creation has been one of the key focusses of ES Chapter 9: Ecology and Biodiversity [APP-047]. It has been concluded that the Scheme delivers a significant net gain in biodiversity of 86.80% gains provided in habitat, 54.71% gains in hedgerow and 33.25% gains in river units as detailed within the Biodiversity Net Gain Assessment [APP-088].
		During construction, the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] sets out measures to ensure the safety of staff and the Site during construction from flood risk. This includes the appointment of at least one designated Flood Warden who is familiar with the risks and remains vigilant to news reports, Environment Agency flood warnings, relevant weather warnings and water levels of the local waterway. Flood risks during operation will be managed through the instillation of mitigation measures as explored within ES Chapter 10: Hydrology, Flood Risk and Drainage [APP-048] and the Flood Risk Assessment and Drainage Strategy Report [APP-089].
POLICY DM13	A. Development proposals will be expected to: i. Minimise the need to travel by private car; ii. Provide linkages, or develop new, footways, cycle paths and bridleways giving access, to key local facilities (especially town centres); and iii. Provide appropriate facilities to support access to high-quality public transport.	The Construction Environmental Management Plan [EN010132/EX6/WB7.1_DREP6-021] and the Construction Traffic Management Plan [REP4-038EN010132/EX7/WB6.3.14.2_E] proposes that car sharing for construction personnel is encouraged and that a minibus service to the Scheme is utilised to consolidate trips thereby minimising private car trips. Footpaths and Bridleways within the Scheme's Order Limits are subject to some disruption during the Construction and



Optimisation of the highway network and highway capacity improvements should only be considered once the above criteria have been addressed.

C. Parking Standards

Non-residential parking should be provided in line with the 6Cs Highway Design Guide adopted by Nottinghamshire County Council on 1 April 2009.

Decommissioning phases of the Scheme. Due regard has been taken in order to minimise disruption through mitigation measures. Enhancement measures have been put in place and are explored within the Public Rights of Way Management Plan [REP5-018]. In addition, and in support of the aims of the Public Rights of Way Improvement Plan, there is to be the creation of a new permissive footpath from the track off Sykes Lane along the Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub.

During Construction, when it is proposed that there will be 296 FTE Staff on Site, the provision of parking compounds has been detailed within the Outline CTMP [REP4-038].EN010132/EX7/WB6.3.14.2_E]. It is considered that a suitable allocation, in the form of temporary compounds, has been provisioned in relation to meeting parking needs.



1.3 Emerging Draft Bassetlaw Local Plan 2020-2038 (Publication Version Composite) (July 2022)

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
POLICY ST1	 Managed sustainable development and growth, appropriate to the size of each settlement to meet the evidenced need for new homes and jobs, regenerate the District's town centres, and support necessary improvements to infrastructure, services and facilities will be achieved by: promoting the efficient and effective use of land and the re-use of previously developed land in sustainable locations, unless there are overriding amenity, biodiversity or heritage matters that preclude such use; and by seeking to minimise the use of the most versatile Grade 1-3 agricultural land, where practicable; emphasising the need to develop in sustainable locations in close proximity to transport hubs and key public transport nodes, and encourage higher density development in those locations; ensuring that sufficient physical, social and green/blue infrastructure is delivered to meet identified needs in a timely manner. 	The rural location is justified due to the scale of the land required to deliver the substantial renewable energy generation capacity that the Scheme will provide, and the need to be in sufficient proximity of the connection point to the National Electricity Transmission System (NETS). The Scheme could not be located within an urban area or settlement boundary as explained within the Site Selection Assessment [AS-004]. The design process has taken into account and sought to minimise the best and most versatile (BMV) land take. This has resulted in 26.24% of the Scheme being BMV.
POLICY ST6	1. Land at the former Cottam Power Station site is identified as a broad location for mixed use regeneration. As such, the site will be safeguarded from development which would jeopardise the comprehensive remediation, reclamation and redevelopment of the whole site.	The scheme does not fall within the Policy area and will not prejudice this regeneration area coming forward.
	2. The proposed development at the Cottam Power Station should deliver a scheme in accordance with a comprehensive	



masterplan framework, design code and agreed site infrastructure delivery and phasing plan and open book viability assessment. All must be agreed with the Local Planning Authority.

- 3. Proposals for the development of this Priority Regeneration Area will permitted where they form part of the comprehensive redevelopment of the site as identified by the masterplan framework and;
 - a) enable the phased reclamation of the site in line with an agreed programme of works and phasing plan;
 - b) comprise a scheme of an appropriate scale, layout, form and materials which respects the significance and setting of affected heritage assets, including the Fleet Plantation Scheduled Monument, supported by a heritage statement to include the results of an archaeological assessment;
 - c) protect and enhance the biodiversity value of the Cottam Wetlands Local Wildlife Site, its buffer zone evidenced by an Ecological Impact Assessment; and, promote linkages to the wider green/blue infrastructure network;
 - d) protect and where appropriate enhance the water quality of the River Trent, including through consideration of integrated water management;
 - e) deliver a flood management scheme which incorporates an appropriate Sustainable Drainage System (SuDS), including green/blue infrastructure measures, informed by a Flood Risk Assessment (FRA), a hydrology assessment and, a Surface Water Management Masterplan and Strategy, in accordance with Policy ST52. Whole life management and



maintenance arrangements must be agreed through the
planning application process;

- demonstrate that the full impact(s) of the proposed regeneration of the site, individually and cumulatively with other development and site allocations in this Plan can be mitigated; and ensure opportunities to reduce transport movements by private vehicles are minimised, and, opportunities to access the site via bus, cycling and walking are maximised, evidenced through a through a comprehensive Transport Assessment and Travel Plan. All proposals must be agreed with the Local Highways Authority;
- g) ensure the continued operation of the Cottam
 Development Centre, by providing, through good design
 and mitigation where necessary, an appropriate standard
 of amenity for future occupiers and residents;
- h) ensure wayleave access arrangements to on site third party infrastructure assets and to the River Trent are maintained and long term management and maintenance arrangements with relevant bodies is in place before development starts, and that these arrangements are reflected in the design of the site;
- ensure the requirements for non-minerals development in Minerals Safeguarding Areas in the Nottinghamshire Minerals Local Plan37 have been met;
- j) protect the Pulverised Fuel Ash North and South Lagoons, and slurry lagoon from inappropriate development, and



	ensure their appropriate restoration and after care in line with relevant permissions; k) give consideration to utilising the River Trent and existing railway line for the transportation of construction and waste materials to and from the site during	
	redevelopment.	
POLICY ST11	Proposals for the growth of businesses in the rural area and outside established employment sites/allocations will be supported where all of the following are met: a) there is a proven need for the development in terms of a business opportunity or operational requirements; b) in the case of existing sites, the proposed development cannot physically and reasonably be accommodated within the existing curtilage; c) the scale of development is appropriate in the proposed location; d) where appropriate the proposal makes efficient use of previously developed land and re-use of existing buildings; e) the development will have no adverse impact on the character of the location, the surrounding townscape or landscape, the form and character of the settlement or upon biodiversity and heritage assets; f) safe access can be achieved by vehicles, and where appropriate sustainable transport and public transport, and that there will be no unacceptable impact on the safe operation of the highway network; and,	 This Policy is to be addressed in turn, given that all aspects of the policy are to be met. a. The Statement of Need [APP-320] explains in detail the compelling case for the Scheme in relation to urgently delivering low carbon renewable energy to meet the aim of decarbonising the UK's electricity supplies by 2035; providing security of supply as well as affordability for end consumers. Given the scale of the Scheme, it is considered unavoidable that the Scheme is located anywhere other than in the Countryside. The Site Selection Assessment explains the site requirements and the choice of site [AS-004]. b. The Scheme is not an existing Site therefore this is not relevant. c. Given the rurality of the Scheme, the scale of the Scheme is considered appropriate. d. Not considered appropriate. e. ES Chapter 8: Landscape and Visual Impact [APP-046] has considered the impacts of the Scheme in relation to the surrounding townscape/ landscape.



	 g) the development generates no adverse impact on residential amenity in accordance with Policy 48. 2. Proposals that support the diversification of the rural economy, including educational facilities to provide training for rural and heritage professions will be supported subject to the provisions of this policy. 	f. Heritage matters have been explored within ES Chapter 13: Cultural Heritage [APP-051]. The impacts of the Scheme are concluded within Section 13.11.
		g. It has been concluded that the Scheme delivers a significant net gain in biodiversity of 86.80% gains provided in habitat, 54.71% gains in hedgerow and 33.25% gains in river units as detailed within the Biodiversity Net Gain Assessment [APP- 088].
		h. It is concluded within the Construction Traffic Management Plan [REP4-038EN010132/EX7/WB6.3.14.2 E] that during construction, safe access to the Scheme will achieved. During operation, it is considered that there will be no unacceptable impact on the safe operation of the highway. Decommissioning impacts are considered to be similar to Construction.
		 Residential receptors have been considered throughout the ES [APP-039 to APP-061].
		The proposal is considered to support the diversification of the rural economy. Through the Skills, Supply Chain and Employment Plan [APP-319] educational training is proposed as part of the Scheme.
		The Scheme generally complies with this policy.
POLICY ST35	1. All development must be of a high-quality design that:	The design rationale for the Scheme is set out within the Design and
	a) has a clear function, character and identity based upon a robust	Access Statement [APP-314 to APP-315]. Section 6.4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD] sets out the Scheme's
	understanding of local context, constraints and distinctiveness, while reflecting the principles of relevant national and local design guidance	compliance with relevant design policies. In terms of the specific policy requirements:



- c) where appropriate, positively preserves, enhances and integrates landscape and townscape features, and natural and heritage assets;
- j) incorporates and/or links to a well-defined green/blue infrastructure network of well-managed and maintained public and open space;
- n) is sustainable in design and construction, and utilises modern construction methods and durable materials, where practicable;
- p) mitigates flood risk and water run-off utilising the drainage hierarchy in accordance with Policy ST52, and integrates water management appropriate to place;
- a) As far as is relevant to the type of development proposed, the design is demonstrated to have a clear function and character based upon a robust understanding of local context, constraints, and distinctiveness. This has been informed by the Landscape and Visual Impact Assessment work and ecological survey work undertaken. See Planning Statement section 6.4

 [EN010132/EX6EX7/WB7.5_CD].
- c) The Scheme is considered to positively preserve, enhance and integrate landscape features, natural and heritage assets.

 Landscape mitigation measures address the relationship between the Scheme and its surroundings. The mitigation measures have sought to incorporate and retain, as far as possible, existing natural features such as hedgerows, trees, and field patterns. The landscape mitigation measures also incorporate landscape treatment to ensure that the Scheme can be satisfactorily assimilated into the surrounding area. The landscape mitigation measures also look to protect any important local views into, out of or through the Site. Landscape mitigation measures are set out in the LVIA Chapter 8 [APP-046] of the ES.
- j) Footpaths and Bridleways within the Scheme's Order Limits are subject to some disruption during the Construction and Decommissioning phases of the Scheme. Due regard has been taken in order to minimise disruption through mitigation measures. Enhancement measures have been put in place and are explored within the Public Rights of Way Management Plan [REP5-018]. In addition, and in support of the aims of the Public Rights of Way Improvement Plan, there is to be the creation of a new permissive footpath from the track off Sykes Lane along the



		Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub.
		n) The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and the Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] have secured recycling of materials wherever possible. Where materials cannot be recycled, they will look to be reused and where reuse isn't an option, materials will be wasted.
		p) The Scheme, through the Flood Risk Assessment and Drainage Strategy Report [APP-089], has duly considered flood risks and surface water run off risks.
		The Scheme is considered to be in compliance with Policy ST35.
POLICY ST37	Proposals that contribute to the nature and quality of Bassetlaw's landscapes will be supported where it can be demonstrated that: a) it protects and where possible enhances the distinctive qualities of the relevant landscape character policy zone, as identified in the Bassetlaw Landscape Character Assessment 2009 by conserving, restoring, reinforcing or creating relevant landscape forms and features;	The Scheme's impact upon Bassetlaw's Landscape Character has been detailed within ES Chapter 8: Landscape and Visual Assessment [APP-046]. The summary of residual landscape effects is contained within section 8.12 of ES Chapter 8: Landscape and Visual Assessment. A significant share of the Cable Route Corridor is located within Bassetlaw and, given that the Cable Route Corridor is to be below ground, this aspect of the Scheme is not expected to result in harm to Bassetlaw's landscape. During Construction and Decommissioning, there is anticipated to be an element of adverse harm upon the landscape whilst works are undertaken. However, this is to be for a temporary period.
POLICY ST39	The connectivity, quality, multifunctionality, biodiversity and amenity value of the green and blue infrastructure network will be enhanced, extended and managed through:	Only part of the grid connection corridor is located within Bassetlaw District.



- a. protecting and enhancing the landscape character and the distinctiveness of Green Gaps, Registered Parks and Gardens and ornamental parklands, registered Common Lands and Village Greens, and Local Green Spaces;
- b. protecting, enhancing and restoring watercourses, ponds, lakes and water dependent habitats where appropriate;
- c. providing for biodiversity net gain, including reconnecting vulnerable and priority habitats (see policy ST41);
- d. protecting and enhancing ancient and mature woodland and hedgerows, and providing for tree planting to secure recreational benefits and/or to aid carbon offsetting;
- e. making appropriate provision for new green/blue infrastructure in new development including open space, allotments, playing fields and outdoor sports facilities, and natural and semi natural greenspace and bluespace; and/or incorporating and where practicable facilitating the improvement of existing provision through the design of development;
- f. applying climate change mitigation and adaptation measures through new development, including flood risk and watercourse management;
- g. linking walking and cycling routes, bridleways and public rights of way to and through development, where appropriate;
- 2. The function, setting, and biodiversity, landscape, access and recreational value of the following main and minor green corridors, as identified on the Policies Map will be protected and enhanced:
- a) Main green corridors

As stated in Chapter 9: Ecology of the ES [APP-047], The Scheme will not result in the loss of ancient woodland or veteran trees. It will also retain existing hedgerow field boundaries meanwhile the Scheme proposed to 'gap up' hedgerows. Whilst some small loss of hedgerow will be required, this is outweighed by the additional planting that is to be undertaken. Undeveloped buffers will be included to protect all hedgerows, veteran/ancient trees, ponds and ancient woodland during construction and operation. Within some of these buffers, particularly around the ancient woodland, natural regeneration of woodland will create additional scrub and woodland habitat. Other areas will be managed as grassland. Tree Root Protection fencing will be erected around retained trees, in line with British Standard BS 5837: Trees in relation to design, demolition and construction – Recommendations and the undeveloped buffers will be of at least 15m from woodlands, trees and hedgerows with trees and 5m from hedgerows without trees.

The Scheme will protect and enhance biodiversity. A Biodiversity Net Gain (BNG) assessment, using Defra's Metric 3.0, has been provided with the DCO application [EN010132/EX6EX7/WB3.1_GH]. For the purposes of BNG, the Scheme will result in an overall significant net. Measures to enhance the biodiversity value of the Order limits and enhance the quality and connectivity of habitats are set out by the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025].

Footpaths and Bridleways within the Scheme's Order Limits are subject to some disruption during the Construction and Decommissioning phases of the Scheme. Due regard has been taken in order to minimise disruption through mitigation measures. Enhancement measures have been put in place and are explored



	iv. River Trent	within the Public Rights of Way Management Plan [REP5-018]. In addition, and in support of the aims of the Public Rights of Way Improvement Plan, there is to be the creation of a new permissive footpath from the track off Sykes Lane along the Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub.
		The Scheme's Cable Route Corridor will traverse the River Trent, being a Main Green Corridor. In order to protect its function and setting, the Cable Route Corridor is proposed to be HDD across the river.
		The Scheme is considered to comply with this policy.
POLICY ST40	The Council will seek to protect and enhance the biodiversity and geodiversity of Bassetlaw, including: International Sites a) a proposal that may impact on a Special Area of Conservation (SAC), Special Protection Area (SPA), Ramsar site and/or the Sherwood Forest ppSPA will only be supported where it can be demonstrated that there will be no likely significant effects on their integrity, unless there are no alternative solutions, and it is justified by an 'imperative reasons of overriding public interest' assessment under the Habitats Directives. National Designations b) a proposal that may either directly or indirectly adversely impact a Site of Special Scientific Interest (SSSI), National Nature Reserve (NNR) or ancient woodland and their buffer zones will be refused	The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline OEMP [REP5-020] and Outline Decommissioning Strategy [EN010132/EX6/WB7.2_BREP6-023] set out measures to protect the environment during construction, operation and decommissioning. The Scheme will protect and enhance biodiversity. A Biodiversity Net Gain (BNG) assessment, using Defra's Metric 3.0, has been provided with the DCO application [EN010132/EX6EX7/WB3.1_GH]. For the purposes of BNG, the Scheme will result in an overall significant net gain. Measures to enhance the biodiversity value of the Order limits and enhance the quality and connectivity of habitats are set out by the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025]. The Outline Landscape and Ecological Management Plan [EN010132/EX6/WB7.3_EREP6-025] details how the Scheme will manage biodiversity assets for the lifetime of the project (being up to 60 years) and is therefore in conformity with the Policy.



other than in wholly exceptional circumstances. All proposals should seek to protect and enhance these features wherever possible.

c) where it can be demonstrated that housing development within the identified zones of influence of Clumber Park SSSI, the Birklands and Bilhaugh SAC, and Sherwood Forest ppSPA will create adverse recreational impacts on the integrity of these designated sites the development will make provision for appropriate mitigation measures including on the development site, and/or as a financial contribution towards mitigation, management and monitoring at the designated asset.

Local Designations and Locally Important Ecological Features

- d) proposals having a direct or indirect adverse effect on a Local Nature Reserve, Local Wildlife Site or Local Geological Site and their buffer zones or other biodiversity/geodiversity asset, will only be supported where there are no reasonable alternatives; and the case for development clearly outweighs the need to safeguard the ecological, recreational and/or educational value of the site.
- 2. In all cases, where the principle of development is considered appropriate the mitigation hierarchy must be applied so that:
- a) firstly harm is avoided wherever possible; then
- b) appropriate mitigation is provided to ensure no net loss or a net gain of priority habitat and local populations of priority species;
- c) as a last resort, compensation is delivered to offset any residual damage to biodiversity;
- d) they protect, restore, enhance and provide appropriate buffers around wildlife and geological features at a local and wider landscapescale to deliver robust ecological networks, to help deliver priorities in

ES Chapter 9: Ecology and Biodiversity [APP-047] details any designations, whether it be international, national or local designations that are within the Scheme's Order Limits and proposes mitigation measures where needed.

The Scheme is considered to generally comply with this policy.



	the Nottinghamshire Biodiversity Opportunity Model for Bassetlaw and Idle Valley 201814; e) they establish additional ecological links to the Nature Recovery Network. Biodiversity Net Gain	
	3. All new development should make provision for at least 10% net biodiversity gain on site, or where it can be demonstrated that for design reasons this is not practicable, off site through an equivalent financial contribution.	
	4. A commuted sum equivalent to 30 years maintenance will be sought to manage the biodiversity assets in the long term.	
POLICY 41	The Council will protect existing trees, woodland and hedgerows and secure additional planting that increases canopy cover in the interests of biodiversity, amenity and climate change adaptation by: a) retaining, protecting and improving woodland and trees subject to Tree Preservation Orders (TPOs), trees within conservation areas, and 'important' hedgerows as defined by the Hedgerows Regulations 1997; b) making Tree Preservation Orders; c) giving consideration to trees and hedgerows both on individual merit as well as their contribution to amenity and interaction as part of a group within the broader landscape setting; d) resisting the loss or deterioration of ancient woodland and ancient or veteran trees unless there are wholly exceptional reasons and a suitable compensation strategy exists; e) seeking contributions to the national tree planting target to contribute to net zero emissions in accordance with Policy ST50.	Only part of the grid connection corridor is located within Bassetlaw District. As stated in Chapter 9: Ecology and Biodiversity of the ES [APP-047], The Scheme will not result in the loss of ancient woodland or veteran trees. It will also retain existing hedgerow field boundaries. Whilst some limited loss of hedgerow vegetation will be required, this is outweighed by the additional planting that is to be undertaken. Undeveloped buffers will be included to protect all hedgerows and ponds during construction and operation. Within some of these buffers, particularly around the ancient woodland, natural regeneration of woodland will create additional scrub and woodland habitat. Other areas will be managed as grassland. Tree Root Protection fencing will be erected around retained trees, in line with British Standard BS 5837. A detailed management plan for trees, woodlands and hedgerows has been incorporated within the Scheme in the form of the Outline Landscape and Ecological Management Plan



	 2. Where development would adversely affect trees or hedgerows the application must be accompanied by: a) an accurate tree survey and arboriculture assessment, undertaken by an experienced arboriculturist, of all existing trees and hedgerows on site in accordance with BS5837 (Trees in relation to design, demolition and construction – Recommendations) 201217; 	[EN010132/EX6/WB7.3_E].REP6-025]. This plan encompasses the entire lifetime of the Scheme, being up to 60 years, and is therefore compliant with the 10 year requirement. The Scheme therefore demonstrates compliance with this policy.
	b) details of protective measures to be put in place during the development to ensure the health and safety of each specimen and hedgerow to be retained;c) an avoidance and mitigation strategy to include replacement planting	
	for specimens of at least equal amenity and ecological value of a local provenance; and	
	d) a detailed management plan providing details of maintenance arrangements for 10 years.	
POLICY ST42	The historic environment will be conserved and enhanced, sensitively managed, enjoyed and celebrated for its contribution to sustainable communities. Proposals will be supported where they:	The Cable Route Corridor is predominantly located within Bassetlaw. ES Chapter 13: Cultural Heritage [APP-051] does not identify any significant adverse impacts upon designated heritage assets within
	a) give great weight to the conservation and re-use of heritage assets (designated and non-designated) and their settings, including for appropriate temporary use, based on their significance in accordance with national policy 1;	the District. ES Chapter 13: Heritage [APP-051] concludes that there is to be reharm upon Non-Designated Archaeological Remains - AR67 to Al
	b) make a positive contribution to the character and local distinctiveness of the historic environment, including through the use of innovative design;	along the cable route in Bassetlaw. The significant public benefits of the Scheme clearly and demonstrably outweigh the reversible harm to non-designated heritage assets, that may result.
	c) positively conserve or enhance a historic designed landscape;	ES Chapter 13: Heritage [APP-051] has been produced and is considered to comply with section 2 of Policy ST42.



	d) maintain, conserve, sustain or return to beneficial use designated or non-designated assets;	The Scheme is considered to comply with this policy.
	e) capitalise in an appropriate and sensitive manner the regeneration, tourism and energy efficiency potential of heritage assets;	The seneme is considered to comply with this policy.
	f) positively secure the conservation and re-use of 'at risk' heritage assets;	
	g) improve access and enjoyment of the historic environment where appropriate, particularly where they retain, create or facilitate public access to heritage assets to increase understanding of their significance.	
	2. Applicants will be required to submit evidence in line with best practice and relevant national guidance, examining the significance of any heritage assets affected through a Heritage Statement, including any contribution made by their setting. The level of detail should be proportionate to the asset's significance, and the results submitted to the Nottinghamshire Historic Environment Record. In some circumstances, further survey, analysis and/or recording will be made a condition of consent.	
POLICY 43	1. Proposals for development, including change of use, that involve a designated heritage asset, or the setting of a designated heritage asset will be expected to:a) conserve, enhance or better reveal those elements which contribute to the heritage significance and/or its setting;	To address this Policy in order 1. The Scheme, within Bassetlaw, is not considered to effect the setting of a designated heritage asset give that the Cable Route Corridor is the only aspect of the Scheme which is to be located within Bassetlaw and is to be buried underground.
	b) respect any features of special architectural or historic interest, including where relevant the historic curtilage or context, its value within a group and/or its setting, such as the importance of a street frontage, traditional roofscape, or traditional shopfronts;	 As explained in ES Chapter 13: Cultural Heritage [APP-051], the Scheme does not result in the total loss of significance of designated and non-designated heritage assets.



- c) be sympathetic in terms of its siting, size, scale, height, alignment, proportions, design and form, building technique(s), materials and detailing, boundary treatments and surfacing, or are of a high quality contemporary or innovative nature which complements the local vernacular, in order to retain the special interest that justifies its designation;
- d) ensure significant views away from, through, towards and associated with the heritage asset(s) are conserved or enhanced;
- e) in the case of a Conservation Area, to have regard to the established urban grain and ensure that spaces between and around buildings, such as paddocks, greens, gardens and other gaps, are preserved where they contribute to the Conservation Area's character and appearance.
- 2. Proposals that will lead to substantial harm or total loss of significance will be refused unless it can be demonstrated that the substantial harm or loss is necessary to achieve substantial public benefits that outweigh that harm or loss, where it can be demonstrated that:
- a) the nature of the heritage asset prevents all reasonable uses of the site;
- b) no viable use of the heritage asset itself can be found in the medium term through appropriate marketing that will enable its conservation;
- c) conservation by grant-funding or some form of not for profit, charitable or public ownership is demonstrably not possible;
- d) the harm or loss is outweighed by the benefit of bringing the site back into use.

3. ES Chapter 13: Cultural Heritage, Section 13.11 [APP-051] concludes the residual level of harm that is expected following mitigation measures.

The significant public benefits of the Scheme clearly and demonstrably outweigh the reversible, low level, less than substantial harm on designated and non-designated heritage assets.



	3. Proposals that would result in less than substantial harm to the significance of a designated heritage asset will only be supported where it can be demonstrated that the public benefits will outweigh any harm identified.	
Policy ST46	 Delivering Quality, Accessible Open Space 1. The amount, quality, community value, functionality and accessibility of publicly accessible open space and green infrastructure will be protected and enhanced by: c) requiring relevant schemes of 10 dwellings or more (or 0.50ha or more) to contribute to maintaining 1ha of Local Nature Reserve per 1000 people, to bring 95% of people within 1km of a Local Nature Reserve; d) requiring proposals that make provision for new open space to provide a management plan and where appropriate a commuted sum to ensure that the quality of new open space is maintained for the lifetime of the development. 	The Scheme seeks to protect and enhance the green infrastructures such as Public Rights of Way (PRoWs) and Footpaths. The Scheme will enhance the PRoW network within Order limits with an additional permissive path, through Work No.11, which will help to enhance the identity of the local area whilst maximising pedestrian and cycle permeability. In order to protect PRoWs and Footpaths, minimum widths have been incorporated into the Scheme design (between Scheme infrastructure). In all cases the PRoW and new permissive paths will be of typical width, 1.5–3.0m, with at least 5m spacing either side of the centreline of the PRoW and therefore delivering a minimum 10m space. This will avoid the perception of being channelled into narrow passages between PV Arrays. The details of these are explored within the Public Rights of Way Plan [REP4-010].
POLICY 48	 Proposals for development should be designed and constructed to avoid and minimise impacts on the amenity of existing and future users, individually and cumulatively, within the development and close to it. As such, proposals will be expected to: not have a significant adverse effect on the living conditions of existing and new residents and future occupiers of the proposed development through loss of privacy, excessive overshadowing or overbearing impact; and 	The Scheme will not adversely affect neighbour amenity as demonstrated by Section 6.4 Good design, Landscape and Visual Impact (section 6.5), Noise (section 6.11), Glint and Glare (section 6.12), Air Quality (section 6.17) of the Planning Statement [EN010132/EX6EX7/WB7.5_CD]. Adverse impacts have been avoided through continual and considerate design iterations as well as the inclusion of additional mitigation measures. The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline OEMP [REP5-020] and Outline Decommissioning Strategy



- b) not generate a level of activity, noise, light, air quality, odour, vibration or other pollution which cannot be mitigated to an appropriate standard.
- 2. Proposals for development adjacent to, or in the locality of, existing 'bad neighbour' uses such as waste sites, incinerators, chemical production, heavy industry and businesses with out of normal hour (9-5) operations, will need to demonstrate that:
- a) the ongoing use of the neighbouring site is not compromised; and
- b) the amenity of future occupiers of the new development can be achieved in accordance with Part 1 of this policy with the ongoing normal use of the neighbouring site;
- 3. Where the development of a new bad neighbour business or change of use could have a significant adverse effect on residential amenity, appropriate mitigation will be required before the development can be occupied.

[EN010132/EX6/WB7.2_BREP6-023] set out measures to avoid pollution to land air or water in order to ensure effects on living conditions of neighbours are minimised during construction and decommissioning. The policy tests and indicators set out by the NPSs and draft NPSs should inform how "unacceptable impacts" referred to in this policy are defined for this NSIP.

Section 15.11 of Chapter 15: Noise & Vibration of the ES **[APP-053]** concludes that there are no anticipated significant adverse effects on health and quality of life arising from the noise or vibration impacts from the construction, decommissioning or operation of the Scheme, including effects on health and quality of life from noise.

Chapter 17: Air Quality of the ES **[APP-055]** concludes that there are anticipated to be no significant adverse effects on air quality as a result of the construction, operation or decommissioning of the Scheme.

Chapter 21 Other Environmental Matters of the ES [APP-061] considers additional ES matters, such as human heath, in order to fully assess the Scheme's impact upon living conditions of new and existing residents.

The Scheme, through the Cable Route Corridor, is adjacent to the West Burton Power Station which is an existing 'bad neighbour'. The development adjacent to West Burton Power Station constitutes the Cable Route Corridor and through considerate construction and decommissioning activities is not considered to have a significant adverse effect on residential amenity.



POLICY 49	Where development is considered to be on contaminated land and/or unstable land, through an appropriate contamination assessment and/or land instability risk assessment, proposals should:	Two Phase 1 Preliminary Ecological Appraisals (PEA) report have been prepared, covering land within the Order limits, and is available in Appendix 9.2 and 9.4 of the ES [APP-078 and APP-080].
	 a) ensure that all works, including investigation of the nature of any contamination or land instability, and removal of materials can be undertaken without causing unacceptable risk to health, waterways or to the environment; b) identify the nature and extent of existing unstable land and/or contaminated land and the level of risk that contaminants/instability could pose in relation to the proposed development and its users, and adjoining land; c) ensure appropriate mitigation measures are identified and 	The information collected as part of the PEA suggests that there are no significant constraints with regards to contamination of soil and groundwater that would limit the development of the Order limits. The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] ensures that, during construction, there is a scheme of works should contamination be found on Site. Where contamination is found, construction will cease, a report and risk assessment will be conducted prior to any commencement of development.
	implemented which are suitable for the proposed use and that the occupiers and neighbouring uses are not exposed to an unacceptable level of risk;	
	d) demonstrate that the developed site will be suitable for the proposed use without risk from contaminants/instability to people, buildings, services or the environment including the apparatus of statutory undertakers.	
POLICY ST50	 All proposals, including the change of use of existing buildings and spaces, should seek to reduce carbon and energy impacts in their design and construction in accordance with Policy ST35. Proposals should incorporate measures that address issues of climate change mitigation through: a) ensuring no adverse impact on local air quality; 	To address this Policy in turn: The construction of the Scheme has considered the impacts of the resource use and climate change. Mitigation includes the use of lower carbon construction methods, segregation of materials for recycling and the reuse of materials wherever possible. Measures are detailed in the Outline CEMP [EN010132/EX6/WB7.1_D].REP6-021]. The Scheme therefore demonstrates compliance with this aspect of the policy.



- b) directing development towards locations that minimise the need to travel and maximise the ability to make trips by sustainable modes of transport;
- c) incorporating passive and energy efficient materials and/or technologies where appropriate;
- d) requiring compliance with relevant national building standards such as meeting BREEAM very good-excellent standards;
- e) promoting the retrofitting of existing buildings, including incorporating measures to reduce energy consumption;
- f) providing for electric vehicle charging capability and charging infrastructure in new development, and/or providing infrastructure that supports car-free living, particularly in town centres;
- g) ensuring that major development makes an appropriate financial contribution to the Bassetlaw carbon offsetting fund;
- h) making best use of available opportunities to reduce the impact of climate change on biodiversity and the natural environment by providing space for habitats and species to move through the landscape and for the operation of natural processes;
- i) minimising the use of natural resources over the development's lifetime, such as minerals and consumable products, by reuse or recycling of materials in construction, and by making the best use of existing buildings and infrastructure;
- 2. All new development should be designed to improve resilience to the anticipated effects of climate change. Proposals should incorporate measures that address issues of adaptation to climate change through:

The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] has instated measures to reduce air quality impacts, such as through the use of wheel washing facilities before exiting Site.

Given the Scale and nature of the Scheme, the Scheme is located within a rural environment. In order to minimise trips to the Scheme, the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] seeks to encourage car sharing and the implementation of staff minibuses.

The Scheme seeks to retain as many existing trees and hedgerows as possible. In addition, the Scheme looks to 'gap up' hedgerows as well as deliver significant Biodiversity Net Gains. This is detailed within the Design and Access Statement [APP-314 to APP-315].

The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline Operational Environmental Management Plan [REP5-020] and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023] have set out a method in which waste (the demanded use of natural resources) is minimised. Where possible, materials will be reused wherever possible. Where this isn't possible, materials will be recycled appropriately by third parties.

As captured within ES Chapter 7: Climate Change [APP-043], the Scheme has considered an assessment of likely impacts and effects within Section 7.8 of the Chapter. The Scheme has is assessed embedded design mitigation in Section 7.7 and adds additional mitigation measures and enhancement measures in section 7.9.

The Flood Risk Assessment and Drainage Strategy Report [APP-089] has duly considered drainage design to reduce the risk of flooding.



	 a) designing layouts so that the orientation of buildings and spaces take the opportunity to maximise solar gain; b) using appropriate materials that enable buildings to ventilate efficiently by day and night; c) adapting surface materials and drainage design to reduce the risk of flooding to land, property and people as a result of more extreme rainfall in accordance with Policy ST52; 	Green infrastructure enhancements have been central to the Scheme's design given its scale and rural nature. Enhancement measures have been put in place and are explored within the Public Rights of Way Management Plan (green infrastructure linkages) [REP5-018].
	d) promoting water efficiency by residential development meeting the tighter Building Regulations optional requirement of 110 litres per person/per day;	
	e) using integrated water management systems to manage runoff and provide a non-potable water supply;	
	f) providing green/blue infrastructure, and where possible, retaining existing trees and woodlands to reduce the 'urban heating effect' during warmer summers; and	
	g) using urban greening methods within the design of new buildings.	
POLICY ST51	Development that generates, shares, transmits and/or stores zero	Only part of the Cable Corridor is located within Bassetlaw District.
	carbon and/or low carbon renewable energy will be supported in principle at the Area of Best Fit at the former High Marnham power station site, as identified on the Policies Map as a result of the ability of on site development to connect to the on site national electricity	The Scheme's Site has been carefully selected and refined. The Site Selection Assessment [AS-004] outlines the reasons behind selecting the Sites that make up the Order Limits.
	grid infrastructure.	The Site falls outside of the Area of Best Fit and should not be
	2. Proposals for renewable energy development on land at the Area of Best Fit should deliver a scheme in accordance with an agreed masterplan framework, relevant supporting technical assessments, delivery strategy and phasing plan for the site in accordance with Policy ST58, and other relevant policies in this Plan.	considered against this aspect of the Policy. The Statement of Need [APP-320] explains in detail the compelling case for the Scheme in relation to urgently delivering low carbon renewable energy to meet the aim of decarbonising the UK's



	 Outside the Area of Best Fit, development that generates, shares, transmits and/or stores zero carbon and/or low carbon renewable energy including community energy schemes will be supported and expected to demonstrate an operational and/or economic need for the development in that location, and the satisfactory resolution of all relevant site specific and cumulative impacts that the scheme could have on the area, taking into account operational and approved developments, as well as any proposed intensification to operational or approved proposals. An assessment should address cumulative visual and landscape impacts, as well as heritage; hydrology; hydrogeology; ecology; traffic and transport; noise; recreation and local amenity impacts. All renewable energy development will be expected to provide details of the expected power generation based upon yield or local self-consumption to enable effective monitoring of the district's contribution to the national zero carbon targets. A decommissioning programme will be required to demonstrate, the effective restoration of land and/or buildings to their original use (such as agriculture) and condition three years after cessation of operations. 	electricity supplies by 2035; providing security of supply as well as affordability for end consumers. An assessment of the cumulative impacts of the Scheme upon the mentioned subjects has been conducted and concluded within the ES [APP-039 to APP-061]. The Scheme's expected generation is outlined within the Grid Connection Statement [APP-316]. An Outline Decommissioning Statement has been produced [EN010132/EX6/WB7.2_B].REP6-023]. This will be developed upon to produce a Decommissioning Statement. The information within this Statement will inform the restoration of the land to its previous use. The Scheme is considered to comply with the requirements of this policy.
POLICY ST52	1. All proposals are required to consider and, where necessary, mitigate the impacts of the proposed development on flood risk, on-site and off-site, commensurate with the scale and impact of the development. Proposals, including change of use applications, must be accompanied by a Flood Risk Assessment (where appropriate), which demonstrates that the development, including the access and egress, will be safe for	A Flood Risk Assessment and Drainage Strategy Report is provided in appendices 10.1 – 10.5 of the ES [APP-089 to APP-093] . The FRA and DSR provides a detailed assessment of the risk of flooding to and from the Scheme (taking account of climate change) and concludes that the risk of flooding will not be increased as a result of the construction, operation or decommissioning of the Scheme. It is therefore considered that the Scheme is compliant with this policy.



its lifetime, without increasing or exacerbating flood risk elsewhere and where possible will reduce flood risk overall.	
2. Where relevant, proposals must demonstrate that they pass the	
Sequential Test and if necessary the Exceptions Test in Flood Zones 2	
and 3 and ensure that where land is required to manage flood risk, it	
is safeguarded from development.	



1.4 Nottinghamshire Minerals Local Plan (NMLP)

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
SO4	Safeguarding of mineral resources: Protect the County's potential mineral resources of economic importance from development which would prevent or hinder their future use.	ES Chapter 12: Minerals [APP-050] assesses the impact of the Cable Route Corridor upon Nottinghamshire's minerals. The Chapter concludes that the sensitivity of the areas affected is medium. However, given the small and/ or isolated nature of mineral reserves, the magnitude is low and the resulting effect is moderate/minor which is not considered significant.
Policy SP7	Minerals Safeguarding Areas 1. Locally and nationally important mineral resources, permitted reserves, allocated sites and associated minerals infrastructure will be safeguarded from needless sterilisation by non-minerals development through the designation of minerals safeguarding areas as identified on the Policies Map. 2. Non-minerals development within minerals safeguarding areas will have to demonstrate that mineral resources will not be needlessly sterilised as a result of the development and that the development would not pose a serious hindrance to future extraction in the vicinity. 3. Where this cannot be demonstrated, and where there is a clear and demonstrable need for the non-minerals development, prior extraction will be sought where practicable.	Minerals Safeguarding has been considered within section 12.8 of ES Chapter 12: Minerals [APP-050]. Within this Chapter, mineral safeguarding has been considered with mitigation measures being concluded. The proposed cabling connecting the Sites to the Grid are unlikely to sterilise any significant volume of safeguarded mineral. The proposed Cable Route Corridor particularly those in the Trent Valley, however, do have the potential to introduce additional constraints to future mineral working and sever otherwise economic reserves. This impact has been mitigated wherever possible by cable routes following existing infrastructure corridors or edges of significant landscape features. No significant adverse impacts on minerals safeguarding are
	Minerals Consultation Areas 4. District and Borough Councils within Nottinghamshire will consult the County Council as Minerals Planning Authority on proposals for non-	identified within ES Chapter 12: Minerals [APP-050] and as such, the Scheme will not result in an unacceptable impact upon the mineral site or allocation.



minerals development within the designated Mineral Consultation Area, as shown on the Policies Map.	The Scheme is considered to comply with the requirements of Policy SP7.
5. The Minerals Planning Authority will resist inappropriate non- minerals development within the Minerals Consultation Areas.	
6. Where non-minerals development would cause an unacceptable impact on the development, operation or restoration of a permitted minerals site, mineral allocation, or associated minerals infrastructure, suitable mitigation should be provided by the applicant prior to the completion of the development.	



1.5 Lincolnshire Minerals and Waste Local Plan (LMNLP) (Core Strategy & Development Management Policies (June 2016)

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
Policy M2	The County Council will ensure a steady and adequate supply of sand and gravel for aggregate purposes by making provision over the period 2014 - 203 1 (inclusive) for the extraction of 42.66 million tonnes of sand and gravel (2 .37 million tonnes per annum). This will be divided between the three Production Areas (as shown on the Key Diagram) as follows:	This policy is noted, in particular the references to the Active Mining Sites being west of Lincoln and to the north/ south of Lincolnshire.
	 18.00 million tonnes (1.00 million tonnes per annum) from the Lincoln/ Trent Valley Production Area; 	
	 9 .00 million tonnes (0.50 million tonnes per annum) from the Central Lincolnshire Production Area; and 	
	• 15 .66 million tonnes (0.87 million tonnes per annum) from the South Lincolnshire Production Area.	
	The County Council will make provision for the release of sand and gravel reserves in the Site Locations Document. This will give priority to extensions to existing Active Mining Sites. New quarries will be allocated where they are required to replace existing Active Mining Sites that will become exhausted during the Plan period and where they are located in the relevant Areas of Search as shown on the Policies Map (Figure 5), namely:	
	west of Lincoln and north/ south of Gainsborough for the Lincoln/ Trent Valley Production Area;	



	 Tattershall Thorpe for the Central Lincolnshire Production Area; and West Deeping/ Langtoft for the South Lincolnshire Production Area. 	
Policy M4	Sites allocated in the Site locations Document will be granted planning permission for sand and gravel extraction for aggregate purposes provided that: • in the case of an extension to an existing Active Mining Site, extraction would follow on after the cessation of sand and gravel extraction from the existing areas supplying the plant site; and • in the case of a new quarry, it is required to replace an existing Active Mining Site that is nearing exhaustion.	ES Chapter 12: Minerals [APP-050] states that current assessments report that there is no need for new minerals sites to come forward during the Lincolnshire Minerals and Waste Local Plan period up to 2031. Furthermore, on the basis the scheme has a lifespan of up to 60 years and due to the Scheme being decommissioned at the end of its operational life, any minerals would not be permanently sterilised and would be available to exploit if required at a future date. Thus, there is not considered to be any conflict with the mineral safeguarding policy.
	 For sites not allocated in the Site locations Document, planning permission will be granted for sand and gravel extraction for aggregate purposes where the site is required to meet: a proven need that cannot be met from the existing permitted reserves; or a specific shortfall in the landbank of the relevant Production Area and either: (i) forms an extension to an existing Active Mining Site; or (ii) is located in the relevant Area of Search as shown on the Policies Map (Figure 5) and will replace an existing Active 	The proposed cabling connecting the Sites to the Grid are unlikely to sterilise any significant volume of safeguarded mineral. The proposed Cable Route Corridor particularly those in the Trent Valley, however, do have the potential to introduce additional constraints to future mineral working and sever otherwise economic reserves. This impact has been mitigated wherever possible by cable routes following existing infrastructure corridors or edges of significant landscape features. No significant adverse impacts on minerals safeguarding are identified within ES Chapter 12: Minerals [APP-050]. The Scheme is considered to comply with the requirements of Policy M4.
	Mining Site that is nearing exhaustion. In all cases the proposal must accord with all relevant Development Management Policies and Restoration Policies set out in the Plan.	



Policy M11

Sand and gravel, blown sand and limestone resources that are considered to be of current or future economic importance within the Minerals Safeguarding Areas shown on Figure 1, together with potential sources of dimension stone for use in building and restoration projects connected to Lincoln Cathedral/Lincoln Castle within the areas shown on Figure 2, and chalk resources included on Figure 3, will be protected from permanent sterilisation by other development.

Applications for non-minerals development in a minerals safeguarding area must be accompanied by a Minerals Assessment. Planning permission will be granted for development within a Minerals Safeguarding Area provided that it would not sterilise mineral resources within the Mineral Safeguarding Areas or prevent future minerals extraction on neighbouring land. Where this is not the case, planning permission will be granted when:

- the applicant can demonstrate to the Mineral Planning Authority that prior extraction of the mineral would be impracticable, and that the development could not reasonably be sited elsewhere; or
- the incompatible development is of a temporary nature and can be completed and the site restored to a condition that does not inhibit extraction within the timescale that the mineral is likely to be needed; or
- there is an overriding need for the development to meet local economic needs, and the development could not reasonably be sited elsewhere; or
- the development is of a minor nature which would have a negligible impact with respect to sterilising the mineral resource; or

ES Chapter 12: Minerals **[APP-050]** states that current assessments report that there is no need for new minerals sites to come forward during the Lincolnshire Minerals and Waste Local Plan period up to 2031. Furthermore, on the basis the scheme has a lifespan of up to 60 years and due to the Scheme being decommissioned at the end of its operational life, any minerals would not be permanently sterilised and would be available to exploit if required at a future date. Thus, there is not considered to be any conflict with the mineral safeguarding policy.

The proposed Cable Route Corridor connecting the Sites to West Burton Power Station is unlikely to sterilise any significant volume of safeguarded mineral. The proposed Cable Route Corridor particularly those in the Trent Valley, however, do have the potential to introduce additional constraints to future mineral working and sever otherwise economic reserves. This impact has been mitigated wherever possible by cable routes following existing infrastructure corridors or edges of significant landscape features.

No significant adverse impacts on minerals safeguarding are identified within ES Chapter 12: Minerals [APP-050].

The Scheme is considered to comply with the requirements of Policy M11.



	 the development is, or forms part of, an allocation in the Development Plan. 	
	Exemptions	
	This policy does not apply to the following:	
	Applications for householder development	
	 Applications for alterations to existing buildings and for change of use of existing development, unless intensifying activity on site 	
	Applications for Advertisement Consent	
	Applications for Listed Building Consent	
	 Applications for reserved matters including subsequent applications after outline consent has been granted 	
	 Prior Notifications (telecommunications; forestry; agriculture; demolition) 	
	 Certificates of Lawfulness of Existing or Proposed Use or Development (CLEUDs and CLOPUDs) 	
	Applications for Tree Works	
Policy M12	Mineral sites (excluding dormant sites) and associated infrastructure that supports the supply of minerals in the County will be safeguarded against development that would unnecessarily sterilise the sites and infrastructure or prejudice or jeopardise their use by creating incompatible land uses nearby. Exemptions	ES Chapter 12: Minerals [APP-050] states that current assessments report that there is no need for new minerals sites to come forward during the Lincolnshire Minerals and Waste Local Plan period up to 2031. Furthermore, on the basis the scheme has a lifespan of up to 60 years and due to the Scheme being decommissioned at the end of its operational life, any minerals would not be permanently
	This policy does not apply to the following:	sterilised and would be available to exploit if required at a future



- Applications for householder development
- Applications for alterations to existing buildings and for change of use of existing development, unless Intensifying activity on site
- Applications for Advertisement Consent
- Applications for Listed Building Consent
- Applications for reserved matters including subsequent applications after outline consent has been granted
- Prior Notifications (telecommunications; forestry; agriculture; demolition)
- Certificates of Lawfulness of Existing or Proposed Use or Development (CLEUDs and CLOPUDs)
- Applications for Tree Works

date. Thus, there is not considered to be any conflict with the mineral safeguarding policy.

The proposed cabling connecting the individual Sites to each other, and the grid are unlikely to sterilise any significant volume of safeguarded mineral. The proposed Cable Route Corridor particularly those in the Trent Valley, however, do have the potential to introduce additional constraints to future mineral working and sever otherwise economic reserves. This impact has been mitigated wherever possible by cable routes following existing infrastructure corridors or edges of significant landscape features.

No significant adverse impacts on minerals safeguarding are identified within ES Chapter 12: Minerals **[APP-050]**. The Scheme is considered to comply with the requirements of Policy M12.



1.6 Lincolnshire Minerals and Waste Local Plan Site Locations (December 2017)

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
Policy SL2	Allocated sites, as set out in Policy SL1, including an area of 250 metres surrounding each site, will be safeguarded against development that would unnecessarily sterilise the sites or prejudice or jeopardise their use by creating incompatible land uses nearby. Exemptions This policy does not apply to the following: • Applications for householder development • Applications for alterations to existing buildings and for change of use of existing development, unless intensifying activity on site • Applications for Advertisement Consent • Applications for Listed Building Consent • Applications for reserved matters including subsequent applications after outline consent has been granted • Prior Notifications (telecommunications; forestry; agriculture; demolition) • Certificates of Lawfulness of Existing or Proposed Use or Development (CLUEDS and CLOPUDs) • Applications for Tree Works	ES Chapter 12: Minerals [APP-050] states that current assessments report that there is no need for new minerals sites to come forward during the Lincolnshire Minerals and Waste Local Plan period up to 2031. Furthermore, on the basis the scheme has a lifespan of up to 60 years and due to the Scheme being decommissioned at the end of its operational life, any minerals would not be permanently sterilised and would be available to exploit if required at a future date. Thus, there is not considered to be any conflict with the mineral safeguarding policy. The proposed cabling connecting the individual Sites to each other, and the grid are unlikely to sterilise any significant volume of safeguarded mineral. The proposed Cable Route Corridor particularly those in the Trent Valley, however, do have the potential to introduce additional constraints to future mineral working and sever otherwise economic reserves. This impact has been mitigated wherever possible by cable routes following existing infrastructure corridors or edges of significant landscape features. No significant adverse impacts on minerals safeguarding are identified within ES Chapter 12: Minerals [APP-050]. The Statement of Need [APP-320] explains in detail the compelling case for the Scheme in relation to urgently delivering low carbon renewable energy to meet the aim of decarbonising the UK's



	electricity supplies by 2035; providing security of supply as well as affordability for end consumers. This is considered a significant benefit which outweighs any temporary sterilisation of Safeguarded Minerals.
	The Scheme is considered to comply with the requirements of Policy SL2.



2 Neighbourhood Plans

2.1 Saxilby with Ingleby Parish Council (2022). Saxilby with Ingleby Neighbourhood Plan 2019 – 2036 Final Approved Version May 2017. West Lindsey District Council.

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
POLICY 2	 All new development must deliver good quality design. In order to achieve this all new development must: Respect the existing pattern of development in terms of enclosure and definition of streets and spaces. Use materials appropriate to the development's context. Be of an appropriate scale and density in relation to its setting. Have good access to public transport or otherwise help to reduce car dependency, such as promoting active travel (walking and cycling). Take advantage of the local topography, landscape, trees and plants, wildlife habitats, existing buildings and site orientation. Take advantage of views into and out of the site in order to make the development easy to access and to navigate through. Car parking should be integrated within the landscaping of the scheme to minimise its visual impact but it should also serve its intended users and encourage natural surveillance. Provide an environment that contributes to the promotion of health and wellbeing of residents through the provision of meeting place 	As detailed in Section 3 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme has been subject to a detailed and sensitive iterative design process. This has taken account of the context and features of the land within the Order limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts, and provision of environmental and other enhancements, where practicable. The design process and basis of design decisions taken are described in the Chapter 5: Alternatives and Design Evolution of the ES [APP-043].



	both in relation to crime and clear and legible pedestrian routes and high quality open space.	
	i. Incorporate flood resilience and resistance measures including, where appropriate, Sustainable Urban Drainage Systems.	
	2. All new development must demonstrate how the above criteria, Building for Life 12 and the Saxilby with Ingleby Village Character Assessment has been used in the designing of the site through the submission of a written statement.	
POLICY 5	1. Proposed developments will be supported where they preserve or enhance the character or appearance of the Parish, Conservation Area	Section 13.7 of Chapter 13: Cultural Heritage of the ES [APP-051] describes, assesses and outlines the significance of heritage assets.
	architectural or historic interest and other heritage assets as set out in Appendix B. 2. When considering the impact of a proposed development on the significance of a designated heritage asset, great weight will be given to the asset's conservation. The more important the asset, the greater the weight will be. Significance can be harmed or lost through alteration or destruction of the heritage asset or development within its setting. As heritage assets are irreplaceable, any harm or loss will require clear and convincing justification.	Chapter 13 identifies the significance of the Scheme's impacts and proposed design mitigation measures required pertaining to cultural heritage. This includes the provision of stand-offs between the
		Scheme and heritage assets in order to help to preserve their setting during the construction, operational and decommissioning periods. By providing the embedded mitigation and stand-offs the Scheme respects and responds to the local context of heritage assets, in accordance with this policy.
		Chapter 13: Cultural Heritage of the ES [APP-051] concludes that there will be moderate adverse harm upon one Listed Building during the construction phase. Church of St Botolph, Saxilby with Ingleby will experience moderate adverse harm.
POLICY 7	Proposals for new B1/B2/B8 employment developments and/ or redevelopment of sites for B1/B2/B8 uses will be supported in both the existing and proposed employment sites shown on Proposal Map 3 provided the proposed development is of a	A Flood Risk Assessment and Drainage Strategy Report is provided in appendices 10.1 – 10.5 of the ES [APP-089 to APP-093]. The FRA and DSR provides a detailed assessment of the risk of flooding to and from the Scheme (taking account of climate change) and concludes



scale that respects the character of the area and neighbouring land uses.

- 2. All new development must ensure that suitable flood resilience and resistance measures, including, where appropriate, the use of sustainable urban drainage systems, are incorporated into the design of any development.
- 3. Development schemes must include landscaping within sites and along boundaries to ensure that the development is satisfactorily screened from the A57 boundary and to minimise the visual impact on the setting of the village and nearby residential properties.
- 4. Developments must provide or contribute to the provision of the walking and cycling routes to the village shown on Proposal Map 6 and take every opportunity to encourage other means of transport than the car.

that the risk of flooding will not be increased as a result of the construction, operation or decommissioning of the Scheme.

Where appropriate the FRA and DSR'S have provided flood resilience and resistance measures.

As Captured within ES Chapter 8: Landscape and Visual Impact [APP-046], Section 8.5 outlines the Baseline Conditions with regards to visual receptors, including receptors along the A57 as well as residential/ PRoWs receptors in Saxilby and Ingleby. Screening has been incorporated within the Scheme to ameliorate its visual impact. These measures are detailed within Section 8.8 of ES Chapter 8: Landscape and Visual Impact [APP-046] whilst the detailing and location of proposed planting has been captured within the Outline Landscape Ecological Mitigation Plan [EN010132/EX6/WB7.3_EREP6-025].

The Scheme will enhance the PRoW network within Order limits with an additional permissive path which will help to enhance the identity of the local area whilst maximising pedestrian and cycle permeability. A minimum width has been incorporated into the Scheme design for PRoW and permissive paths, as well as the corridor in which they will be provided (between Scheme infrastructure). In all cases the PRoW and new permissive paths will be of typical width, 1.5–3.0m, with at least 5m spacing either side of the centreline of the PRoW and therefore delivering a minimum 10m space. This will avoid the perception of being channelled into narrow passages between PV Arrays. The details of these are explored within the Public Rights of Way Plan [REP4-010].



		It is therefore considered that the Scheme is compliant with this policy.
POLICY 9	1. Proposals to redevelop or change the use of an existing community facility or land or buildings last used as a community facility will only be permitted where one of the following conditions is met: a. A replacement facility of an equivalent or better size, layout and quality is provided in a suitable location; or b. It can be satisfactorily demonstrated that the facility is no longer fit for purpose or economically viable for a new or another community use; or	As detailed within the Draft Development Consent Order [EN010132/EX6EX7/WB3.1_GH], Work No. 9 seeks to create and maintain habitat management areas. The Scope of the Works is considered to result in a replacement of an equivalent size with an improved layout, boundary treatments, drainage provision, means of access and enhancement measures (such as planting).
	c. The alternative use would have significant community benefits for the local community.	
POLICY 11	Development will be supported where it can demonstrate it meets the following criteria: a. Where development protects and enhances existing features in the natural environment.	The Scheme looks to protect existing hedgerows and other landscape features have been incorporated into the Scheme as far as possible as set out in Figure 8.15.1 to Figure 8.18.1 [REP1-026] to Figure 8.18.3 [REP1-030] of the ES.
	 b. Development will be expected to retain well-established landscape features such as mature trees, species-rich hedgerows and ponds. c. The use of boundary treatments that are sympathetic to maintaining and enhancing biodiversity on new or existing developments will be encouraged and supported. 2. If there is significant and unavoidable loss of trees and shrubs as part of development new provision will be expected elsewhere on the site. 	As detailed in Section 3 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme has been subject to a detailed and sensitive iterative design process. The Scheme has sought to retain well-established landscape features such as mature trees, species-rich hedgerows and ponds wherever possible. The Scheme will protect and enhance biodiversity. A Biodiversity Net Gain (BNG) assessment, using Defra's Metric 3.0, has been provided with the DCO application [EN010132/EX6EX7/WB3.1_GH]. For the purposes of BNG, the Scheme will result in an overall significant net gain. Measures to enhance the biodiversity value of the Order limits and enhance the quality and connectivity of habitats are set out by



		the Outline LEMP [EN010132/EX6/WB7.3_E].REP6-025]. It has been concluded that the Scheme delivers a significant net gain in biodiversity of 86.80% gains provided in habitat, 54.71% gains in hedgerow and 33.25% gains in river units as detailed within the Biodiversity Net Gain Assessment [APP-088].
POLICY 12	Developments proposals should plan positively for the protection, enhancement and creation of networks to improve the connectivity between biodiversity and green infrastructure.	The Scheme will enhance the PRoW network within Order limits with an additional permissive path which will help to enhance the identity of the local area whilst maximising pedestrian and cycle permeability. A minimum width has been incorporated into the Scheme design for PRoW and permissive paths, as well as the corridor in which they will be provided (between Scheme infrastructure). In all cases the PRoW and new permissive paths will be of typical width, 1.5–3.0m, with at least 5m spacing either side of the centreline of the PRoW and therefore delivering a minimum 10m space. This will avoid the perception of being channelled into narrow passages between PV Arrays. The details of these are explored within the Public Rights of Way Plan [REP4-010].
POLICY 16	 All development across the Plan area which is directly related to improving or extending non-vehicular routes will be permitted where the proposals: a. Do not detract from the landscape character or ecological value as defined in the most recent Landscape Character Assessment Study. b. Are for enhancing the understanding or enjoyment of the area's biodiversity; and are designed to ensure continued privacy for residents. 	The Scheme will enhance the PRoW network within Order limits to enhance the identity of the local area whilst maximising pedestrian and cycle permeability. A minimum width has been incorporated into the Scheme design for PRoW and permissive paths, as well as the corridor in which they will be provided (between Scheme infrastructure). In all cases the PRoW and new permissive paths will be of typical width, 1.5–3.0m, with at least 5m spacing either side of the centreline of the PRoW and therefore delivering a minimum 10m space. This will avoid the perception of being channelled into narrow passages between PV Arrays. These amendments to the PRoW network is considered to result in
		an improvement to the network.



	New development shall take every opportunity to provide new, or enhance existing, non-vehicular routes including connections with the existing network.	The Scheme, through Work No.11, is to result in the creation of an additional permissive footpath from the track off Sykes Lane along the Codder Lane Belt and then south and west to re-join Sykes Lane opposite Hardwick Scrub [EN010132/EX6EX7/WB3.1_GH]. This Works package of the Scheme is considered to result in an extension of the existing non-vehicular network.
		The full details of these amendments and additions to the non-vehicular network are explored within the Public Rights of Way Plan [REP4-010].
POLICY 17	Development proposals in Saxilby must ensure that any transport impacts of the scheme are identified and acceptable. Any measures needed to deal with the anticipated impacts must be implemented.	The Construction, Operation and Decommissioning impacts of the Scheme upon the transport network have been assessed in Appendix 14.1 [REP1-014]. The appendices concludes that the Construction Vehicle Routes will ensure the sufficient spread of construction and decommissioning traffic and is therefore not expected to have any significant effect on the local highway network.
		During Operation, there is anticipated to be less than one visit per day to the Site for maintenance purposes and as such this is not considered to result in a significant effect on the local highway network.



2.2 Sturton by Stow Parish Council and Stow Parish Council (2022). Sturton by Stow and Stow Neighbourhood Plan 2019 – 2036 Final Approved Version March 2022. Gainsborough: West Lindsey District Council.

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
Policy 1	1. To support and enhance the sustainability of the Parishes of Sturton by Stow and Stow, development will be supported where it is consistent with the following principles as appropriate to the proposal's scale, nature and location within the neighbourhood area: c. any necessary physical or social infrastructure or improvements to such infrastructure that may be required to make a particular development proposal acceptable in planning terms are delivered in association with that development; d. development outside the existing or planned built-up areas of Sturton by Stow and Stow villages will only be supported if it: i. is required for agricultural purposes; or ii. is required to support an existing agricultural or non-agricultural use; or iii. makes sustainable use of a previously developed site; or e. development does not increase the risk of flooding and should reduce such risk where possible; f. developments in Sturton by Stow and Stow are located, designed, constructed and operated so as to be consistent with the national target of bringing the United Kingdom's greenhouse gas emissions to net zero by 2050;	The construction of the Scheme has considered the impacts of the resource use and climate change. Mitigation includes the use of lower carbon construction methods, segregation of materials for recycling and the reuse of materials wherever possible. Measures are detailed in the Outline CEMP [EN010132/EX6/WB7.1_D]-REP6-021]. The Scheme therefore demonstrates compliance with this aspect of the policy. Chapter 7 Climate Change of the ES [REP1-012] presents a lifecycle greenhouse gas (GHG) impact assessment which considers the impact of GHG emissions arising over the lifetime of the Scheme on the climate. This concludes that over a 60 year operational lifetime, the Scheme will produce 31,425,614 MWh of electricity and deliver a reduction of 3,981,049 tCO2e over the lifetime of the Scheme compared to if it did not go ahead. This demonstrates its very low carbon attributes compared to other non-renewable forms of electricity generation, providing an overall major beneficial impact in relation to the UK meeting its carbon reduction targets and therefore represents a major beneficial effect on the climate. A Flood Risk Assessment (FRA) is provided at Appendices 10.1 – 10.6 of ES Chapter 10 [APP-048]. The FRA provides a detailed assessment of the risk of flooding to and from the Scheme (taking account of climate change) and concludes that the risk of flooding will not be



	g. development is located and designed so that any potential negative impact on climate change such as increased carbon emissions or flood risk is mitigated. h. developments should incorporate clear measures for adaptation and resilience to climate change	increased as a result of the construction, operation or decommissioning of the Scheme. It is therefore considered that the Scheme is compliant with this policy. During construction, the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] sets out measures to ensure the safety of staff during construction from flood risk. This includes the appointment of at least one designated Flood Warden who is familiar with the risks and remains vigilant to news reports, Environment Agency flood warnings, relevant weather warnings and water levels of the local waterway.
Policy 5	 As appropriate to their scale, nature and location, developments should demonstrate good quality design and respect the character and appearance of the surrounding area. All development proposals will be assessed to ensure that they effectively address the following matters, as described in detail in each Character Area chapter of the Neighbourhood Profile: siting and layout; density, scale, form and massing; detailed design and materials; landscaping and streetscape. Development proposals will be supported if it is demonstrated that their design solutions: apply principles of good design to ensure that both neighbouring users and occupiers of the proposed development will benefit from reasonable standards of amenity, unimpaired by unacceptable overlooking, loss of privacy, loss of light, pollution (including contaminated land, light pollution or emissions), odour, noise and other forms of disturbance; 	As detailed in Section 3 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme has been subject to a detailed and sensitive iterative design process. This has taken account of the context and features of the land within the Order limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts, and provision of environmental and other enhancements, where practicable. The design process and basis of design decisions taken are described in the Chapter 5: Alternatives and Design Evolution of the ES [APP-043]. The Scheme makes a significant contribution towards limiting climate change and ES Chapter 7: Climate change [REP-045] concludes it will have a significant beneficial effect in terms of climate change.



	c. minimise the waste of resources (e.g. electricity, gas and water) and promote renewable energy generation and energy efficiency, minimise risk of flooding, the design of all aspects of the development should mitigate for climate change impacts and incorporate climate change adaptation and resilience measures that ensure there is no increase in carbon emissions (preferably a reduction), they promote renewable energy generation and energy efficiency and do not increase the risk of local and nearby flooding (including the use of Sustainable Urban Drainage Solutions, permeable surfaces etc).; d. avoid adversely impacting on Heritage Assets listed in Policy 6 and/or the Protected Views of Policy 9 g. promote safe access by vehicles, pedestrians, wheelchair users and cyclists, and promote connectivity across and around the development for pedestrians, pushchairs, wheelchair users, cyclists and mobility vehicles	The panels are predominantly made from recyclable materials. The Applicant refers the parties to Table 20.7 in WB6.2.20 ES Chapter 20 Waste [APP-058] which identifies estimated volumes of waste from decommissioning. Approximately 95% of the panel weight is made from glass and metal frames, which can easily be reused and recycled. The remaining silicon and electrical waste can be partially recycled at Waste Electrical and Electronic Equipment (WEEE) facilities. The Scheme will be adequately served by highways infrastructure and there will be no significant impacts upon highway safety as demonstrated by ES Chapter 14: Transport and Access [APP-052].
Policy 6	 Proposed developments will be supported where they preserve or enhance the character or appearance of the historic settlements, listed buildings and their settings and any features of special architectural or historic interest, including locally important heritage assets, all as identified in Policy Map 6. When considering the impact of a proposed development on the significance of a designated and non-designated heritage asset (as shown on Policy Map 6), great weight will be given to the asset's conservation. The more important the asset, the greater the weight will be. 	The Scheme does not involve any internal or external alterations, or extensions to a listed building or listed structure, nor does it involve change of use of a listed building or listed structure. As assessment of local heritage assets has been conducted and is within the ES Chapter 13: Cultural Heritage [APP-051]. Section 13.8 of Chapter 13: Cultural Heritage of the ES [APP-051] outlines the significance of heritage assets, significance of the Scheme's impacts and proposed design mitigation measures required pertaining to cultural heritage. This includes the provision of stand-offs between the Scheme and heritage assets in order to help to preserve their setting during the construction, operational and decommissioning periods. By providing the embedded



		mitigation and stand-offs the Scheme respects and responds to the local context of heritage assets, in accordance with this policy.
Policy 7	 Proposals for new business premises, or the expansion and regeneration of existing business premises will be supported, subject to the following criteria: a) It can be demonstrated that any proposals protect and, where practicable, enhance: (i) the character of the Parishes – including local heritage assets, as detailed in the Neighbourhood Profile, in line with Policy 5: Delivering Good Design. (ii) the local environment and biodiversity. c) The proposal incorporates measures to mitigate any nuisance from increased traffic, noise, smell, lighting, vibration or other emissions or activities generated by the proposed development. d) The proposal improves the visual amenity of the neighbourhood area where it is practicable to do so and relates directly to the development proposed. 	The Scheme will protect and enhance biodiversity. A Biodiversity Net Gain (BNG) assessment, using Defra's Metric 3.0, has been provided with the DCO application [EN010132/EX6EX7/WB3.1_GH]. For the purposes of BNG, the Scheme will result in an overall significant net gain. Measures to enhance the biodiversity value of the Order limits and enhance the quality and connectivity of habitats are set out by the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025]. Artificial lighting, traffic, noise, smell, and vibration nuisances will be most prevalent during construction and decommissioning. These issues have been addressed within the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] and Outline Decommissioning Statement [EN010132/EX6/WB7.2_BREP6-023]. Mitigation measures such as inward directed lighting, core working hours, no idling of car engines and the forbidding out vehicle reversing sirens. An assessment of the potential landscape and visual impacts associated with the construction, operation and decommissioning of the Scheme has been carried out and is presented in Chapter 8: Landscape and Visual Impact of the ES [APP-046]. Section 8.7 of Chapter 8: Landscape and Visual Impacts of the ES [APP-046] outlines and identifies the likely significant effects of the Scheme before addressing mitigation measures in section 8.8.
Policy 9	The Plan identifies Protected Views as shown on Policy Maps 9.1 and 9.2.	An assessment of protected views relating to the Scheme is presented in Chapter 8: Landscape and Visual Impact of the ES [APP-046]. It is considered that due consideration has been given to all



	Development proposals should be located and designed to take account of the identified Protected Views and, where practicable, to enhance or provide greater accessibility to the views concerned. Development proposals which would have an unacceptable impact on a Protected View will not be supported.	relevant Protected Views and that this application is compliant with this policy.
Policy 11	1. As appropriate to the scale, nature and location, development proposals should: a) contribute to the enhancement and management of existing green corridors and infrastructure assets, where practicable; and	Due regard has been taken in order to minimise disruption through mitigation measures. Enhancement measures have been put in place and are explored within the Public Rights of Way Management Plan (green infrastructure linkages) [REP5-018].
	b) contribute to the provision of new public green spaces and enhance green infrastructure linkages, where practicable.	The Statement of Need [APP-320] explains in detail the compelling case for the Scheme in relation to urgently delivering low carbon
	2. Development proposals that result in an unacceptable impact on the purpose or function of existing green infrastructure will not be supported unless they:	renewable energy to meet the aim of decarbonising the UK's electricity supplies by 2035; providing security of supply as well as affordability for end consumers. This is considered a benefit which
	a) demonstrate that the impact on the purpose or function of the green infrastructure is unavoidable and significantly and demonstrably outweighed by the benefits of the development; and	exceeds any harm caused. As such, the Scheme is considered compliant with this policy.
b) provide for the implementation of alternative solutions, as part of the development, to reinstate the green infrastructure's purpose or function to the previous quality and connectivity.		
	3. Development proposals that result in unacceptable harm to the biodiversity of existing green infrastructure and that cannot be avoided (through locating on an alternative site with less harmful impacts), adequately mitigated, or, as a last resort, compensated for, will not be supported.	



	 Developments that enhance and/or connect existing or create new Green Infrastructure will be supported, in particular where they clearly demonstrate mitigation, adaptation and resilience to climate change. Proposals for development that create/make provision for new green space (in addition to and not a replacement for existing green space) will be supported. Where practicable, such proposals should provide amenity for residents, be of value for wildlife and provide climate change mitigation, adaptation and resilience. 	
Policy 12	 Development proposals will be supported where the primary objective is to conserve or enhance biodiversity or geodiversity of the environment. All developments, projects and activities will be supported which: a. provide a practicable level of protection to legally protected sites and species; b. protect irreplaceable habitats, such as ancient woodlands and ancient or veteran trees, except where there are wholly exceptional reasons and a suitable compensation strategy exists; c. maintain and where practicable enhance conditions for priority habitats; d. maintain and where practicable enhance recognised geodiversity assets; e. maintain and where practicable enhance other sites, features, species; f. identify, protect, maintain and expand as appropriate networks of ecological interest and provide for appropriate management; 	The Scheme will protect and enhance biodiversity. A Biodiversity Net Gain (BNG) assessment, using Defra's Metric 3.0, has been provided with the DCO application [EN010132/EX6EX7/WB3.1_GH]. For the purposes of BNG, the Scheme will result in an overall significant net. Measures to enhance the biodiversity value of the Order limits and enhance the quality and connectivity of habitats are set out by the Outline LEMP [EN010132/EX6/WB7.3_E].REP6-025]. The measures instated within the Scheme seek to ensure species are protected appropriately, irreplaceable habitats are protected, and sites, species and features are enhanced. The Outline CEMP [EN010132/EX6/WB7.1_DREP6-021], Outline OEMP [REP5-020] and Outline Decommissioning Strategy [EN010132/EX6/WB7.2_BREP6-023] set out measures to protect the environment during construction, operation and decommissioning.



	g. identify measures to avoid and/or reduce any potentially adverse impacts on the natural environment to acceptable levels (commensurate with the status of specific sites where applicable); h. mitigate against any necessary impacts through appropriate habitat	
	creation, restoration or enhancement on site or elsewhere.	
	i. seek and exploit opportunity to conserve, augment and reinstate the stock of trees, hedges, woodlands, wetlands and countryside as wildlife habitat and for aesthetic enjoyment, in both the rural and urban environment;	
	3. As appropriate to their scale, nature and location, development proposals should incorporate environmental protection measures, which clearly demonstrate mitigation, adaptation and resilience to climate change.	
Policy 13	1. Development proposals, including those within areas that have experienced flooding, as shown on accredited flood risk maps, should demonstrate that the proposal has considered the risk of flooding from all sources and will not have an unacceptable impact on existing foul and surface water drainage infrastructure. Development proposals should make use of sustainable drainage systems to manage surface water, wherever practicable.	A Flood Risk Assessment (FRA) is provided at Appendices 10.1 – 10.6 of the ES [APP-048]. The FRA provides a detailed assessment of the risk of flooding to and from the Scheme (taking account of climate change) and concludes that the risk of flooding will not be increased as a result of the construction, operation or decommissioning of the Scheme. It is therefore considered that the Scheme is compliant with this policy.
	2. Development proposals should not increase the rates of surface water run-off or increase flood risk in the area.	Adequate buffers between development and watercourses are incorporated into the Scheme. These buffers will be enhanced or
	3. Development proposals that include de-culverting any culverted watercourses within the development boundary will be particularly supported.	allowed to enhance by natural regeneration, in accordance with this policy.
	4. Development proposals for new dwellings should be designed to minimise the discharge of surface water. Proposals that include the	The OOEMP [REP5-020] sets out that Staff on site will undertake regular weather checks to forecast any heavy rain events and to prepare for flooding where necessary. Areas of the Order limits at



	provision of permeable parking spaces and driveways will be particularly supported. 5. Drainage strategies for the management of surface water run-off from new development should incorporate Sustainable Drainage Systems and be designed to incorporate ecological benefits where practicable.	risk of flooding are not expected to be frequently occupied by staff and access to the Solar Farm Site is located in Flood Zone 1.
Policy 15	 Development proposals directly related to improving or extending walking and cycling routes, as identified on Policy Map 15, will be supported where they: a) do not have an unacceptable impact on the landscape character or ecological value, as defined in the Sturton by Stow and Stow Neighbourhood Profile. b) do not have an unacceptable impact on the privacy and amenity of nearby or directly adjoining neighbouring properties. Developments that propose improvements or extensions to the existing public rights of way footpaths, as identified on Policy Map 15, from Sturton by Stow to Stow and other nearby settlements, or the creation of new walking and cycling routes, will be strongly supported. 	An assessment of the potential landscape and visual impacts associated with the construction, operation and decommissioning of the Scheme has been carried out and is presented in Chapter 8: Landscape and Visual Impact of the ES [APP-046]. Section 8.7 of Chapter 8: Landscape and Visual Impacts of the ES [APP-046] outlines and identifies the likely significant effects of the Scheme before addressing mitigation measures in section 8.8. The Scheme will not result in the closure of any PRoW during the operation. PRoW diversions may be required during construction. These would be short in terms of distance and duration. Appendix 14.3 of the ES [REP5-018] .



2.3 Sturton Ward Neighbourhood Plan (2021). Sturton Ward Neighbourhood Plan Review (2021-2037). West Lindsey District Council.

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
Policy 1	Sustainable development, infill and the development boundary: 2. Development proposals outside the defined development boundaries will be carefully controlled in accordance with national and local planning policies.	The Scheme is considered to be compliant with national Policy as has been detailed throughout the Planning Statement [EN010132/EX6EX7/WB7.5_CD] and Planning Statement Appendix C (National Policy Accordance Table).
Policy 2a	Protecting the landscape character, significant green gaps and key views 1. As appropriate to their scale, nature and location, development proposals should protect the positive attributes of the open countryside and landscape character as identified within the Bassetlaw Landscape Character Assessment and the Sturton Ward Design Code 2020. Where appropriate, mitigation planting should include native species recommended for the Mid Nottinghamshire Farmlands or Trent Washlands character areas. 2. The Plan designates the parcels of land shown on Maps 6a, 6b and 6c as Significant Green Gaps. Development proposals should demonstrate how they would safeguard the positive contribution made by the affected Significant Green Gap to the landscape and character of the neighbourhood area, the role and character of a settlement affected and, where appropriate its relationship with settlements around the neighbourhood area. Development proposals which would have an unacceptable impact on the character of an identified Significant Green Gap will not be supported.	The Landscape mitigation measures set out in ES Chapter 8: Landscape and Visual Impact [APP-046] have addressed the intrinsic value of the landscape and townscape, including the setting of settlements. The Scheme will have particular regard to maintaining and responding positively to any natural and man-made features within the landscape and townscape which positively contribute to the character of the area. Measures will respond to historic buildings and monuments, other landmark buildings, topography, trees and woodland, hedgerows, walls, water features, field patterns and intervisibility between rural historic settlements. The Scheme will not adversely affect neighbour amenity as demonstrated by Section 6.4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD]. The aspect of the Scheme which is to pass through this neighbourhood area is the Cable Route Corridor. As such, all development will be below ground and the impacts on views within the area will be limited only to Construction and Decommissioning stages. Through the adoption of an outline Constriction Environmental Management Plan [EN010132/EX6/WB7.1_DREP6-021] and a Decommissioning Statement



	 3. Development proposals should respect the views highlighted on Maps 5a, 5b and 5c by careful attention to their layout, massing and height. Development proposals which would have an unacceptable impact on the identified views will not be supported. 4. Development proposals (excluding householder development), should demonstrate they have regard to the guidance in the Sturton Ward Design Code (2020) and where practicable: a) retain existing trees and hedges; and 	[EN010132/EX6/WB7.2_BREP6-023], the impacts of these stages have been limited wherever possible and pose only temporary impacts upon views. Additionally, a Landscape and Ecology Mitigation and Enhancement Plan [REP1-026 to REP1-031] has been provided to capture where existing planting is to be retained and removed and where proposed planting and waterbodies are to be located. It is considered that the planting proposals result in the creation of strong planting belts within the Site.
	b) use strong planting belts that run horizontal with the contours on sites close to the existing built form; and	
	c) maintain the natural flow of water through water courses and prevent surface water from being connected to the foul sewerage network.	
Policy 2b	1. Proposals which would improve existing environmental assets and enhance biodiversity will be supported. The following enhancements will be particularly supported:	The Scheme will protect and enhance biodiversity. A Biodiversity Net Gain (BNG) assessment, using Defra's Metric 3.0, has been provided with the DCO application [EN010132/EX6EX7/WB3.1_GH]. For the
	a) strengthening hedgerows (gapping up), field boundaries and maintaining the natural flow of water through water courses to provide more robust blue-green habitat 'corridors'; and	purposes of BNG, the Scheme will result in an overall significant net gain. Measures to enhance the biodiversity value of the Order limits and enhance the quality and connectivity of habitats are set out by the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025].
	b) planting wildflower meadows and strips; and	As set out by the Outline LEMP [EN010132/EX6/WB7.3_EREP6-025],
	c) encouraging native tree and shrub planting on suitable sites, especially species that provide good berry or nectar sources; and	hedgerows will be gapped up where there is vegetation loss while adjacent hedgerows will be gapped up.
	d) providing sustainable urban drainage schemes (SuDS) (e.g. rain gardens, ponds and wetland creation) in new schemes and 'retrofitting' where appropriate; and	As outlined within Figures 8.18.1 – 8.18.3, [REP1-026 to REP1-031]], the planting of wildflower meadows and strips as well as native tree and shrubs has been incorporated into the Scheme's design.
		The Flood Risk Assessment and Drainage Strategy Report [APP-089] captures the threats posed by flooding and surface water run-off and



	e) providing habitat improvements (i.e. nest or bat boxes) to benefit all bats and bird species of conservation concern such as swifts, swallow, house martin and house sparrows; and	propose mitigation measures where necessary to ensure that the Scheme does not result in significant adverse harm.
	f) protecting existing dry ditches which are essential to the sustainable management of surface water.	
	2. Proposals that would result in the net loss of biodiversity will not be supported. Development should aim to achieve a net biodiversity gain in accordance with local and national planning policy. If significant ecological impacts are identified, appropriate mitigation or compensation measures will be required. These measures should be targeted to benefit local conservation priorities as identified in the Nottinghamshire Local Biodiversity Action Plan.	
Policy 4	1. All development proposals are required to consider and, when necessary, address the effect of the proposed development on flood risk both on-site and offsite, commensurate with the scale and impact of the development. This should be demonstrated through a flood risk assessment where appropriate and in accordance with national policy.	During construction, the Outline CEMP [EN010132/EX6/WB7.1_DREP6-021] sets out measures to ensure the safety of staff and the Site during construction from flood risk. Given the scale of the Scheme, this includes the appointment of at least one designated Flood Warden who is familiar with the risks and
	2. Proposals for flood management or other infrastructure which lower the risk of flooding will be supported, subject to the proposal not resulting in an increase in flood risk elsewhere.	remains vigilant to news reports, Environment Agency flood warnings, relevant weather warnings and water levels of the local waterway.
	3. Surface water management should be undertaken, where necessary and practicable, through the utilisation of appropriate SuDS techniques which mimic natural drainage patterns and, where appropriate, achieve net gains for nature through the creation of ponds and wetlands near	Flood risks during operation will be managed through the instillation of mitigation measures as explored within ES Chapter 10: Hydrology, Flood Risk and Drainage [APP-048] and the Flood Risk Assessment and Drainage Strategy Report [APP-089].
	watercourses and the introduction or enhancement of blue-green corridors. For SuDS techniques which are designed to encourage infiltration, a site-specific infiltration test will be required to ensure that the water table is low enough.	Planting Schemes, as detailed within Figures 8.18.1 – 8.18.3, [REP1-026 to REP1-031], propose a significant uplift in the planting of native trees and shrubs which is considered to be beneficial to decreasing surface water runoff rates across the Scheme.



	4. All developments should be designed to achieve, where appropriate, a net decrease in surface water run-off rates, including through green infrastructure provision such as the planting of native trees and bushes.	
Policy 6	1. Development proposals will be supported where they preserve or enhance the Conservation Area, listed buildings and other heritage assets as set out in Appendix C and where they comply with the following criteria:	Section 13.5 of the ES Chapter 13: Cultural Heritage [APP-051] includes an assessment of the impact of the Scheme upon conservation areas within 5km of the Order Limits. There are no significant effects upon Conservation Areas.
	a) the development or alteration proposed does not have a detrimental effect on the heritage asset concerned; and	The Scheme does not involve any internal or external alterations, or extensions to a listed building or listed structure, nor does it involve
	b) the heritage asset is sensitively and fully incorporated into the development proposal.	change of use of a listed building or listed structure.
	2. Gardens, open spaces and fields to the south of North and South Wheatley form part of the special interest of the Conservation Area. Development will only be supported on gardens, open spaces between buildings and fields within the Conservation Area where it can be demonstrated that proposals will not harm the character and appearance of the Conservation Area as a whole.	
Policy 8	1. Proposals for commercial, business or service use (Class E), public houses and hot food/takeaway (sui generis), non-residential institutions (Class D1) and assembly and leisure uses (Class D2) will be supported subject to the following criteria:	The rural location is justified due to the scale of the land required to deliver the substantial renewable energy generation capacity that the Scheme will provide, and the need to be in sufficient proximity of the connection point to the National Electricity Transmission System
	a) the site is within an existing development boundary unless it can be demonstrated that the business operation requires a rural location outside a development boundary; and	(NETS). The Scheme could not be located within an urban area or settlement boundary as explained within the Site Selection Assessment [AS-004].
	b) the scale, design and form of the proposed development is in keeping with its surroundings; and	As detailed in Section 3 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme has been subject to a detailed and sensitive iterative design process. This has taken account of the context and features of the land within the Order



	c) the nature of the business concerned does not have a detrimental impact on the amenity of the surrounding area due to unacceptable noise, light, smell, flicker, dust and emissions; and d) the proposal will not have an unacceptable detrimental effect on the built or natural environment of its immediate locality; and e) the proposal will not have an unacceptable detrimental effect on the operation and safety of the local highway network.	limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts upon its surroundings. The design process and basis of design decisions taken are described in the Chapter 5: Alternatives and Design Evolution of the ES [APP-043].
		Chapter 13: Cultural Heritage [APP-051] identifies the significance of the Scheme's impacts and proposed design mitigation measures required pertaining to cultural heritage (built heritage). This includes the provision of stand-offs between the Scheme and heritage assets in order to help to preserve their setting during the construction, operational and decommissioning periods. By providing the embedded mitigation and stand-offs the Scheme respects and responds to the local context of heritage assets, in accordance with this policy.
		As for the natural environment, The Scheme is in an appropriate location as demonstrated within the Site Selection Assessment [AS-004] and will not significantly harm the character and appearance of the surrounding countryside given the Scheme's scale as demonstrated by ES Chapter 8: Landscape and Visual Assessment [APP-046].
		There are no significant adverse impacts on the local highway network as demonstrated by ES Chapter 14: Transport and Access [APP-052].
Policy 12	1. Proposals for the development of low carbon homes that maximise water efficiency and the generation of renewable and low carbon	As detailed within ES Chapter 22 [APP-060] , adverse impacts have been mitigated for through embedded and additional measures.



energy resources will be supported where, either individually or cumulatively, it can be demonstrated that adverse impacts have been mitigated.

- 2. In particular proposals for low carbon homes should:
- a) not have an unacceptably adverse impact on the amenity of residents and visitors (including noise, vibration, views and vistas, shadow flicker, water pollution, odour, air quality, emissions, sensitivity and character of landscape); and
- b) not have a unacceptable adverse effect on any designated site (including SSSI, regionally or locally important geological sites, sites of ecological value, valued landscapes, listed buildings, heritage assets, local green spaces, significant green gaps, Conservation Areas or their settings); and
- c) not result in an unacceptably adverse effect on protected species including migration routes and sites of biodiversity value; and
- d) not result in the unacceptable loss of good-quality agricultural land; and
- e) include details of associated developments including access roads and ancillary buildings. Transmission lines should be located below ground wherever possible to reduce the impact on the open countryside; and
- f) include measures for the removal of structures and the restoration of sites, should sites become non-operational.

The Scheme, as assessed through sections 6.4, 6.11, 6.5, 6.10, 6.12, 6.18 of the Planning Statement, has considered potential adverse impacts on the amenity of residents and visitors (including noise, vibration, views and vistas, shadow flicker, water pollution, odour, air quality, emissions, sensitivity and character of landscape) and concludes with mitigation measures to minimise the significance of the adverse effects.

The Scheme is not considered to have an unacceptable adverse effect on any designated Site.

The Scheme is not considered to result in any unacceptable adverse effects on protected species.

The Scheme's Order Limits contains 26.24% BMV land. Given the Scale of the project, a rural location was necessary and resultingly, through careful design consideration, has sought to minimise the land take of BMV.

Access roads to the Scheme during construction, operation and decommissioning of the Scheme has been detailed in the Construction Traffic Management Plan [REP4-

038].EN010132/EX7/WB6.3.14.2 E]. The Cable Route Corridor is to be located below ground in order to minimise the impact upon the open countryside.

The Scheme is considered to be in conformity with this policy.



2.4 Tresswell and Cottam Neighbourhood Plan (2019). (Referendum Version). 2018 – 2033 (Adopted February 2019). Bassetlaw District Council.

Relevant Paragraph/ Policy Reference	Policy Requirement	Compliance with Policy
Policy 1	 Proposals for rural enterprise will be supported where they can demonstrate that such development would support the economic sustainability of Treswell or Cottam. Developments shall be located within areas at least risk of flooding. Proposals that are located within either flood zones 2 or 3 should undertake a sequential assessment to identify whether there are areas at a lower risk of flooding than the one proposed. All development will be designed having regard to the policies and supporting evidence set out in this Neighbourhood Plan and will be located to ensure that the development does not significantly and adversely affect the: a) amenity of nearby residents; b) character and appearance of the area in which it is located; c) social, built, historic cultural and natural assets of the parish. 	The Scheme will support 296 net direct jobs per annum during the construction period. Of these, 142 jobs per annum will be expected to be taken-up by residents within the combined areas of Bassetlaw District and West Lindsey District. During operation the Scheme would directly generate a gross 12 FTE employees per annum as set out within ES Chapter 18, Socio Economics Tourism and Recreation [APP-056]. As part of the DCO Submission, the Scheme includes a Flood Risk Assessment and Drainage Strategy [APP-089] as well as Flood Risk Assessments and Drainage Strategies for West Burton 1, 2, 3 and the Cable Route Corridor [APP-090 to APP-093]. These FRA's and DS' propose mitigation measures which demonstrably lower the risks of flooding. The Landscape mitigation measures set out in ES Chapter 8: Landscape and Visual Impact [APP-046] have addressed the intrinsic value of the landscape and townscape, including the setting of settlements. The Scheme will have particular regard to maintaining and responding positively to any natural and man-made features within the landscape and townscape which positively contribute to the character of the area. Measures will respond to historic buildings and monuments, other landmark buildings, topography, trees and woodland, hedgerows, walls, water features, field patterns and intervisibility between rural historic settlements. The Scheme will not



		adversely affect neighbour amenity as demonstrated by Section 6.4 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD].
Policy 2	Design Principles 1. Developments should create places or character based upon an appreciation of the site and surrounding area, responding positively to its natural and built context. This policy should be read in conjunction with the most recent Treswell and Cottam Character Assessment (Appendix 1). To achieve this, development proposals will where appropriate consider the following principles: b) Development shall be designed to sustain significant views that contribute to the character and appearance of the area. These views include (but not limited to) are those identified on figures 8 and 5, and applications shall include an assessment of the impact of the proposal on the positive qualities of such views, explaining the rationale of design choices used; j) The plan areas listed buildings and non-designated heritage assets are shown on figures 7 and 10. Proposals affecting the listed buildings and/ or its setting will be expected to preserve and, if possible, enhance the listed building and its setting proposals affecting non-designated heritage assets will be judged against the scale of harm or loss to the significance of the asset. 2. Proposals that do not have regard to the key features of the character area concerned and would create demonstrable harm to its key features and attributes, will not be supported	As detailed in Section 3 of the Planning Statement [EN010132/EX6EX7/WB7.5_CD], the Scheme has been subject to a detailed and sensitive iterative design process. This has taken account of the context and features of the land within the Order limits, nearby sensitive receptors and assets, information emerging from environmental surveys, feedback from stakeholders, and opportunities and constraints in order to develop a good design that balances the need to maximise the energy generation capacity of the Scheme, with the avoidance and mitigation of impacts, and provision of environmental and other enhancements, where practicable. The design process and basis of design decisions taken are described in the Chapter 5: Alternatives and Design Evolution of the ES [APP-043]. The Site is an appropriate location for the Scheme as demonstrated within the Site Selection Assessment [AS-004] and will not significantly harm the character and appearance of the surrounding countryside as demonstrated by ES Chapter 8: Landscape and Visual Assessment [APP-046]. Chapter 13 identifies the significance of the Scheme's impacts and proposed design mitigation measures required pertaining to cultural heritage. This includes the provision of stand-offs between the Scheme and heritage assets in order to help to preserve their setting during the construction, operational and decommissioning periods. By providing the embedded mitigation and stand-offs the Scheme respects and responds to the local context of heritage assets, in accordance with this policy.
Policy 6	Supporting Local Employment Opportunities	As detailed within ES Chapters 14: Transport and Access [APP-052], 15: Noise and Vibration [APP-053], 17: Air Quality [APP-055] and 16:



- 1 Proposals for new, or the expansion of existing businesses and enterprises, within the Neighbourhood Plan Area, will only be supported, where:
- a) it can be demonstrated, to the Local Planning Authority, that there will be no unreasonable impact resulting from increased traffic, noise, smell, lighting, vibration or other emissions or activities generated by the proposed development;
- b) it would have an acceptable impact on the character and scale of the villages and the adjacent landscape in terms of its scale, colour and height;
- c) where relevant, opportunities are taken to secure the re-use of vacant or redundant buildings as part of the development;
- d) it is supporting local employment opportunities;
- e) It is diversifying or supplementing an established existing business to support its continued economic viability.

Glint and Glare **[APP-054]**, the Scheme, through mitigation measures, is not anticipated to result in adverse effects which are considered significant in EIA terms.

The Site is an appropriate location for the Scheme as demonstrated within the Site Selection Assessment [AS-004] and will not significantly harm the character and appearance of the surrounding countryside as demonstrated by ES Chapter 8: Landscape and Visual Assessment [APP-046].

Chapter 13 identifies the significance of the Scheme's impacts and proposed design mitigation measures required pertaining to cultural heritage. This includes the provision of stand-offs between the Scheme and heritage assets in order to help to preserve their setting during the construction, operational and decommissioning periods. By providing the embedded mitigation and stand-offs the Scheme respects and responds to the local context of heritage assets, in accordance with this policy.

The Scheme would have a positive impact on employment generation within the plan period to 2036. This includes the following:

- The Scheme will support 296 net direct jobs per annum during the construction period. Of these, 142 jobs per annum will be expected to be taken-up by residents within the combined areas of Bassetlaw District and West Lindsey District. During operation the Scheme would directly generate a gross 12 FTE employees per annum as set out within ES Chapter 18, Socio Economics Tourism and Recreation [APP-056].
- As detailed in ES Chapter 18, Socio Economics Tourism and Recreation [APP-056], the development will bring in tangible



		economic benefits. The construction phase will result in a Gross Value Added (GVA) figure of £64,100,000 whilst the GVA figure for a 60-year operation phase is £77,400,000 and decommissioning is estimated at £51,300,000.
	-	An outline Skills, Supply chain and Employment plan [APP-319] this seeks to have a positive impact on education and skills attainment in fields such as construction, engineering, and energy technology throughout the operational lifetime of the Scheme.